# Dimension Theory

# RYSZARD ENGELKING

Warsaw University



1978

NORTH-HOLLAND PUBLISHING COMPANY AMSTERDAM · OXFORD · NEW YORK

PWN – POLISH SCIENTIFIC PUBLISHERS WARSZAWA



A revised and enlarged translation of Teoria wymiaru, Warszawa 1977

Translated by the Author

Library of Congress Cataloging in Publication Data

Ryszard, Engelking. Dimension theory.

Translation of Teoria wymiaru. Bibliography: p. 1. Topological spaces. 2. Dimension theory (Topology) I. Title. QA611.3.R9713 514'.3 78-12442 ISBN 0-444-85176-3

Copyright (C) by PWN - Polish Scientific Publishers - Warszawa 1978

All rights reserved. No part of this publication may be reproduced, stored in a retrieval system or transmitted in any form or by any means, electronic, mechanical, photocopying, recording, or otherwise, without the prior written permission of the copyright owner

Published in co-edition with PWN-Polish Scientific Publishers – Warszawa 1978 Distributors for the socialist countries:

ARS POLONA, Krakowskie Przedmieście 7, 00-068 Warszawa, Poland

North-Holland Publishing Company – Amsterdam · New York · Oxford Distributors outside the socialist countries:

Elsevier/North-Holland Inc. 52, Vanderbilt Avenue, New York, N. Y. 10017, U.S.A. Sole distributors for the U.S.A. and Canada

Printed in Poland by D.R.P.

# PREFACE

Dimension theory is a branch of topology devoted to the definition and study of the notion of dimension in certain classes of topological spaces. It originated in the early twenties and rapidly developed during the next fifteen years. The investigations of that period were concentrated almost exclusively on separable metric spaces; they are brilliantly recapitulated in Hurewicz and Wallman's book Dimension Theory, published in 1941. After the initial impetus, dimension theory was at a standstill for ten years or more. A fresh start was made at the beginning of the fifties, when it was discovered that many results obtained for separable metric spaces can be extended to larger classes of spaces, provided that the dimension is properly defined. The last reservation necessitates an explanation. It is possible to define the dimension of a topological space X in three different ways, the small inductive dimension ind X, the large inductive dimension IndX, and the covering dimension dimX. The three dimension functions coincide in the class of separable metric spaces, i.e., indX = IndX $= \dim X$  for every separable metric space X. In larger classes of spaces the dimensions ind, Ind, and dim diverge. At first, the small inductive dimension ind was chiefly used; this notion has a great intuitive appeal and leads quickly and economically to an elegant theory. The dimension functions Ind and dim played an auxiliary role and often were not even explicitly defined. To attain the next stage of development of dimension theory, namely its extension to larger classes of spaces, first and foremost to the class of metrizable spaces, it was necessary to realize that in fact there are three theories of dimension and to decide which is the proper one. The adoption of such a point of view immediately led to the understanding that the dimension ind is practically of no importance outside the class of separable metric spaces and that the dimension dim prevails over the dimension Ind. The greatest achievement in dimension theory during the fifties was the discovery that IndX = dim X for every metric space X and the creation of a satisfactory dimension theory for metrizable spaces. Since that time many important results on dimension of topological spaces

#### Preface

have been obtained; they primarily bear upon the covering dimension dim. Included among them are theorems of an entirely new type, such as the factorization theorems, with no counterpart in the classical theory, and a few quite complicated examples, which finally demarcated the range of applicability of various dimension functions.

The above outline of the history of dimension theory helps to explain the choice and arrangement of the material in the present book. In Chapter 1, which in itself constitutes more than half of the book, the classical dimension theory of separable metric spaces is developed. The purpose of the chapter is twofold: to present a self-contained exposition of the most important section of dimension theory and to provide the necessary geometric background for the rather abstract considerations of subsequent chapters. Chapters 2 and 3 are devoted to the large inductive dimension and the covering dimension, respectively. They contain the most significant results in dimension theory of general topological spaces and exhaustive information on further results. Chapter 4, the last in the book, develops the dimension theory of metrizable spaces. The interdependence of Chapters 2-4 is rather loose. After having read Chapter 1, the reader should be able to continue the reading according to his own interests or needs; in particular, he can read small fragments of Sections 3.1 and 3.2 and pass to Chapter 4 (cf. the introduction to that chapter).

Chapter 1 is quite elementary; the reader is assumed to be familiar only with the very fundamental notions of topology of separable metric spaces. The subsequent chapters are more difficult and demand from the reader some acquaintance with the notions and methods of general topology.

Each section ends with historical and bibliographic notes. Those are followed by problems which aim both at testing the reader's comprehension of the material and at providing additional information; the problems usually contain detailed hints, which, in fact, are outlines of proofs.

The mark  $\square$  indicates the end of a proof or of an example. If it appears immediately after the statement of a theorem, a proposition or a corollary, it means that the statement is obviously valid.

Numbers in square brackets refer to the bibliography at the end of the book. The papers of each author are numbered separately, the number being the year of publication. In referring to my *General Topology* (Engelking [1977]), which is quite often cited in the second half of the present book, the symbol [GT] is used.

In 1971–1973 I gave a two-year course of lectures on dimension theory at the Warsaw University; this book is based on the notes from those lectures. When preparing the present text, I availed myself of the comments of my students and colleagues. Thanks are due to K. Alster, J. Chaber, J. Kaniewski, P. Minc, R. Pol, T. Przymusiński, J. Przytycki and K. Wojtkowska. I am particularly obliged to Mrs. E. Pol, the first reader of this book, for her helpful cooperation, and to J. Krasinkiewicz for his careful reading of Chapter 1.

Ryszard Engelking

Warsaw, February 1977

### CHAPTER 1

# DIMENSION THEORY OF SEPARABLE METRIC SPACES

In the present chapter the classical dimension theory of separable metric spaces is developed. Practically all the results of this chapter were obtained in the years 1920–1940. They constitute a canon on which, in subsequent years, dimension theory for larger classes of spaces was modelled. Similarly, in Chapters 2–4 we shall follow the pattern of Chapter 1 and constantly refer to the classical theory. This arrangement influences our exposition: the classical material is discussed here in relation to modern currents in the theory; in particular, the dimension functions Ind and dim are introduced at an early stage and are discussed simultaneously with the dimension function ind.

To avoid repetitions in subsequent chapters, a few definitions and theorems are stated in a more general setting, not for separable metric but for topological, Hausdorff, regular or normal spaces; this is done only where the generalization does not influence the proof. If the reader is not acquainted with the notions of general topology, he should read "metric space" instead of "topological space", "Hausdorff space", "regular space", and "normal space". Reading the chapter for the first time, one can omit Sections 1.4 and 1.12–1.14, which deal with rather special topics; similarly, the final parts of Sections 1.6, 1.8 and 1.9 can be skipped.

Let us describe briefly the contents of this chapter.

Section 1.1 opens with the definition of the small inductive dimension ind; in the sequel some simple consequences and reformulations of the definition are discussed. Sections 1.2 and 1.3 are devoted to a study of zerodimensional spaces. We prove several important theorems, specified in the titles of the sections, which are generalized to spaces of higher dimension in Sections 1.5, 1.7 and 1.11.

In Section 1.4 we compare the properties of zero-dimensional spaces with the properties of different highly disconnected spaces. From this comparison it follows that the class of zero-dimensional spaces in the sense of the small inductive dimension is the best candidate for the zero level in a classification of separable metric spaces according to their dimension. The results of this section are not used in the sequel of the book.

Section 1.5 contains the first group of basic theorems on *n*-dimensional spaces. As will become clear further on, the theorems in this group depend on the dimension ind, whereas the theorems that follow them depend on the dimension dim. Besides the generalizations of five theorems proved in Sections 1.2 and 1.3 for zero-dimensional spaces, Section 1.5 contains the decomposition and addition theorems.

In Section 1.6 the large inductive dimension Ind and the covering dimension dim are introduced; they both coincide with the small inductive dimension ind in the class of separable metric spaces. In larger classes of spaces the dimensions ind, Ind and dim diverge. This subject will be discussed thoroughly in the following chapters. In particular, it will become evident that the dimension ind, though excellent in the class of separable metric spaces, loses its importance outside this class.

Section 1.7 opens with the compactification theorem. The location of this theorem at such an early stage in the exposition of dimension theory is a novelty which, it seems, permits a clearer arrangement of the material. From the compactification theorem the coincidence of ind, Ind, and dim for separable metric spaces is deduced.

In Section 1.8 we discuss the dimensional properties of Euclidean spaces. We begin with the fundamental theorem of dimension theory, which states that  $\operatorname{ind} R^n = \operatorname{Ind} R^n = \dim R^n = n$ ; then we characterize *n*-dimensional subsets of  $R^n$  as sets with a non-empty interior, and we show that no closed subset of dimension  $\leq n-2$  separates  $R^n$ . This last result is strengthened in Mazurkiewicz's theorem, which is established in the final part of the section with the assistance of Lebesgue's covering theorem.

Section 1.9 opens with the characterization of dimension in terms of extending mappings to spheres from a closed subspace over the whole space. From this characterization the Cantor-manifold theorem is deduced. In the final part of the section we give some information on the cohomological dimension.

In Section 1.10 we characterize *n*-dimensional spaces in terms of mappings with arbitrarily small fibers to polyhedra of geometric dimension  $\leq n$  and develop the technics of nerves and  $\varkappa$ -mappings which are crucial for the considerations of this and the following section.

In Section 1.11 we prove that every *n*-dimensional space can be embedded in  $R^{2n+1}$  and we describe two subspaces of  $R^{2n+1}$  which contain topologically all *n*-dimensional spaces; the second of those is a compact space.

The last three sections are of a more special character. Section 1.12 is devoted to a study of the relations between the dimensions of the domain and the range of a continuous mapping. In Section 1.13 we characterize compact spaces of dimension  $\leq n$  as spaces homeomorphic to the limits of inverse sequences of polyhedra of geometric dimension  $\leq n$ , and in Section 1.14 we briefly discuss the prospects for an axiomatization of dimension theory.

### 1.1. Definition of the small inductive dimension

**1.1.1. Definition.** To every regular space X one assigns the *small inductive dimension* of X, denoted by indX, which is an integer larger than or equal to -1 or the "infinite number"  $\infty$ ; the definition of the dimension function ind consists in the following conditions:

- (MU1) ind X = -1 if and only if  $X = \emptyset$ ;
- (MU2) ind  $X \le n$ , where n = 0, 1, ..., if for every point  $x \in X$  and each neighbourhood  $V \subset X$  of the point x there exists an open set  $U \subset X$  such that

 $x \in U \subset V$  and  $\operatorname{ind} \operatorname{Fr} U \leq n-1$ ;

- (MU3) ind X = n if ind  $X \le n$  and ind X > n-1, i.e., the inequality ind  $X \le n-1$  does not hold;
- (MU4) ind  $X = \infty$  if ind X > n for n = -1, 0, 1, ...

The small inductive dimension ind is also called the Menger-Urysohn dimension.

Applying induction with respect to indX, one can easily verify that whenever regular spaces X and Y are homeomorphic, then indX = indY, i.e., the small inductive dimension is a topological invariant.

In order to simplify the statements of certain results proved in the sequel, we shall assume that the formulas  $n \le \infty$  and  $n + \infty = \infty + n$ =  $\infty + \infty = \infty$  hold for every integer *n*.

Since every subspace M of a regular space X is itself regular, if the

dimension ind is defined for a space X it is also defined for every subspace M of the space X.

**1.1.2. The subspace theorem.** For every subspace M of a regular space X we have  $\operatorname{ind} M \leq \operatorname{ind} X$ .

**Proof.** The theorem is obvious if  $indX = \infty$ , so that one can suppose that  $indX < \infty$ . We shall apply induction with respect to indX. Clearly, the inequality holds if indX = -1.

Assume that the theorem is proved for all regular spaces whose dimension does not exceed  $n-1 \ge -1$ . Consider a regular space X with  $\operatorname{ind} X$ = n, a subspace M of the space X, a point  $x \in M$  and a neighbourhood V of the point x in M. By the definition of the subspace topology, there exists an open subset  $V_1$  of the space X satisfying the equality  $V = M \cap V_1$ . Since  $\operatorname{ind} X \le n$ , there exists an open set  $U_1 \subset X$  such that

 $x \in U_1 \subset V_1$  and  $\operatorname{ind} \operatorname{Fr} U_1 \leq n-1$ .

The intersection  $U = M \cap U_1$  is open in M and satisfies  $x \in U \subset V$ . The boundary  $\operatorname{Fr}_M U$  of the set U in the space M is equal to  $M \cap \overline{M \cap U_1} \cap \overline{M \setminus U_1}$ , where the bar denotes the closure operation in the space X; thus the boundary  $\operatorname{Fr}_M U$  is a subspace of the space  $\operatorname{Fr} U_1$ . Hence, by the inductive assumption,  $\operatorname{ind} \operatorname{Fr}_M U \leq n-1$ , which—together with (MU2)—yields the inequality  $\operatorname{ind} M \leq n = \operatorname{ind} X$ .

Sometimes it is more convenient to apply condition (MU2) in a slightly different form, involving the notion of a partition.

**1.1.3. Definition.** Let X be a topological space and A, B a pair of disjoint subsets of the space X; we say that a set  $L \subset X$  is a *partition between* A and B if there exist open sets U,  $W \subset X$  satisfying the conditions

(1)  $A \subset U$ ,  $B \subset W$ ,  $U \cap W = \emptyset$  and  $X \setminus L = U \cup W$ .

Clearly, the partition L is a closed subset of X.

The notion of a partition is related to the notion of a separator. Let us recall that a set  $T \subset X$  is a separator between A and B, or T separates the space X between A and B, if there exist two sets  $U_0$  and  $W_0$  open in the subspace  $X \setminus T$  and such that  $A \subset U_0$ ,  $B \subset W_0$ ,  $U_0 \cap W_0 = \emptyset$  and  $X \setminus C$  $= U_0 \cup V_0$ . Obviously, a set  $L \subset X$  is a partition between A and B if and only if L is a closed subset of X and L is a separator between A and B. Separators are not to be confused with cuts, a related notion we will refer to in the notes below and in Section 1.8. Let us recall that a set  $T \subset X$  is a *cut between A and B*, or *T cuts the space X between A and B*, if the sets *A*, *B* and *T* are pairwise disjoint and every continuum, i.e., a compact connected space  $C \subset X$ , intersecting both *A* and *B* intersects the set *T*. Clearly, every separator between *A* and *B* is a cut between *A* and *B*. Sut the two notions are not equivalent (see Problems 1.1.D and 1.8.F).

**1.1.4. Proposition.** A regular space X satisfies the inequality  $\operatorname{ind} X \leq n \geq 0$  if and only if for every point  $x \in X$  and each closed set  $B \subset X$  such that  $x \notin B$  there exists a partition L between x and B such that  $\operatorname{ind} L \leq n-1$ .

**Proof.** Let X be a regular space satisfying  $\operatorname{ind} X \leq n \geq 0$ ; consider a point  $x \in X$  and a closed set  $B \subset X$  such that  $x \notin B$ . There exist a neighbourhood  $V \subset X$  of the point x such that  $\overline{V} \subset X \setminus B$  and an open set  $U \subset X$  such that  $x \in U \subset V$  and  $\operatorname{ind} \operatorname{Fr} U \leq n-1$ . One easily sees that the set  $L = \operatorname{Fr} U$  is a partition between x and B; the sets U and  $W = X \setminus \overline{U}$  satisfy conditions (1).



Fig. 1

Now, assume that a regular space X satisfies the condition of the theorem; consider a point  $x \in X$  and a neighbourhood  $V \subset X$  of the point x. Let L be a partition between x and  $B = X \setminus V$  such that  $\operatorname{ind} L \leq n-1$  and let  $U, W \subset X$  be open subsets of X satisfying conditions (1). We have

and

 $\operatorname{Fr} U \subset (X \setminus U) \cap (X \setminus W) = X \setminus (U \cup W) = L,$ 

 $x \in U \subset X \setminus W \subset X \setminus B = V$ 

so that ind Fr  $U \leq n-1$  by virtue of 1.1.2. Hence ind  $X \leq n$ .

Obviously, a regular space X satisfies the inequality  $\operatorname{ind} X \leq n \geq 0$ if and only if X has a base  $\mathscr{B}$  such that ind  $\operatorname{Fr} U \leq n-1$  for every  $U \in \mathscr{B}$ . In the realm of separable metric spaces this observation can be made more precise.

**1.1.5. Lemma.** If a topological space X has a countable base, then every base  $\mathscr{B}$  for the space X contains a countable family  $\mathscr{B}_0$  which is a base for X.

**Proof.** Let  $\mathcal{D} = \{V_i\}_{i=1}^{\infty}$  be a countable base for the space X. For i = 1, 2, ... define

$$\mathscr{B}_i = \{ U \in \mathscr{B} \colon U \subset V_i \};$$

as  $\mathscr{B}$  is a base for X, we have  $\bigcup \mathscr{B}_i = V_i$ . The subspace  $V_i$  of the space X also has a countable base, so that the open cover  $\mathscr{B}_i$  of  $V_i$  contains a countable subcover  $\mathscr{B}_{0,i}$ . The family  $\mathscr{B}_0 = \bigcup_{i=1}^{\infty} \mathscr{B}_{0,i} \subset \mathscr{B}$  is countable and is a base for X; indeed, every non-empty open subset of X can be represented as the union of a subfamily of  $\mathscr{D}$ , and thus can also be represented as the union of a subfamily of  $\mathscr{B}_0$ .

**1.1.6. Theorem.** A separable metric space X satisfies the inequality  $\operatorname{ind} X \leq n \geq 0$  if and only if X has a countable base  $\mathscr{B}$  such that  $\operatorname{ind} \operatorname{Fr} U \leq n-1$  for every  $U \in \mathscr{B}$ .

#### Historical and bibliographic notes

The dimension of simple geometric objects is one of the most intuitive mathematical notions. There is no doubt that a segment, a square and a cube have dimension 1, 2 and 3, respectively. The necessity of a precise definition of dimension became obvious only when it was established that a segment has exactly as many points as a square (Cantor 1878), and that a square has a continuous parametric representation on a segment, i.e., that there exist continuous functions x(t) and y(t) such that points of the form (x(t), y(t)) fill out a square when t runs through a segment (Peano 1890). First and foremost the question arose whether there exists a parametric representation of a square on a segment which is at the same time one-to-one and continuous, i.e., whether a segment and a square are homeomorphic, and—more generally—whether the *n*-cube  $I^n$  and the *m*-cube  $I^m$  are homeomorphic if  $n \neq m$ ; clearly, a negative answer was expected. Between 1890 and 1910 a few faulty proofs of the fact that  $I^n$ 

and  $I^m$  are not homeomorphic if  $n \neq m$  were produced and it was established that I,  $I^2$  and  $I^3$  are all topologically different.

The theorem that  $I^n$  and  $I^m$  are not homeomorphic if  $n \neq m$  was proved by Brouwer in [1911]. The idea suggests itself that to prove this theorem one should define a function d assigning to every space a natural number, expressing the dimension of that space, such that to every pair of homeomorphic spaces the same natural number is assigned and that  $d(I^n) = n$ . It was none too easy, however, to discover such functions; the search for them gave rise to dimension theory. In Brouwer's paper [1911] no function d is explicitly defined, yet an analysis of the proof shows that to differentiate  $I^n$  and  $I^m$  for  $n \neq m$  the author applies the fact that for a sufficiently small positive number  $\varepsilon$  it is impossible to transform the *n*-cube  $I^n \subset R^n$  into a polyhedron  $K \subset R^n$  of geometric dimension less than n by a continuous mapping  $f: I^n \to K$  such that  $\varrho(x, f(x)) < \varepsilon$  for every  $x \in I^n$ . As we shall show in Section 1.10, this property characterizes compact subspaces of  $R^n$  which have dimension equal to n. Another topological property of the *n*-cube I<sup>n</sup> was discovered by Lebesgue in [1911], viz. the fact that  $I^n$  can be covered, for every  $\varepsilon > 0$ , by a finite family of closed sets with diameters less than  $\varepsilon$  such that all intersections of n+2 members of the family are empty, and cannot be covered by a finite family of closed sets with diameters less than 1 such that all intersections of n+1 members of the family are empty. Obviously, Lebesgue's observation implies that  $I^n$  and  $I^m$  are not homeomorphic if  $n \neq m$ . Though the proof outlined by Lebesgue contains a gap (filled by Brouwer in [1913] and by Lebesgue in [1921]), nevertheless the discovery of the new invariant was an important achievement which eventually led to the definition of the covering dimension. Lebesgue's paper [1911] contains one more important discovery. The author formulated the theorem (the proof was given in his paper [1921]) that for every continuous parametric representation  $f(t) = (x_1(t), x_2(t), ..., x_n(t))$ of the *n*-cube  $I^n$  on the closed unit interval I, some fibres of f, i.e., inverse images of one-point sets, have cardinality at least n+1, and that  $I^n$  has a continuous parametric representation on I with fibres of cardinality at most n+1.

A decisive step towards the definition of dimension was made by Poincaré in [1912], where he observed that the dimension is related to the notion of separation and could be defined inductively. Poincaré called attention to the simple fact that solids can be separated by surfaces, surfaces by lines, and lines by points. It was due to the character of the journal for which Poincaré was writing and also to his death in the same year 1912 that Poincaré's important ideas were not presented as a precise definition of dimension.

The first definition of a dimension function was given by Brouwer in [1913], where he defined a topological invariant of compact metric spaces, called Dimensionsgrad, and proved that the Dimensionsgrad of the *n*-cube  $I^n$  is equal to *n*. In conformity with Poincaré's suggestion, the definition is inductive and refers to the notion of a cut: Brouwer defined the spaces with *Dimensionsgrad* 0 as spaces which do not contain any continuum of cardinality larger than one (i.e., as punctiform spaces; cf. Section 1.4), and stated that a space X has *Dimensionsgrad* less than or equal to  $n \ge 1$  if for every pair A, B of disjoint closed subsets of X there exists a closed set  $L \subset X$  which cuts X between A and B and has Dimensionsgrad less than or equal to n-1. Brouwer's notion of dimension is not equivalent to what we now understand by the dimension of a compact metric space; however, the two notions coincide in the realm of locally connected compact metric spaces (the proof is based on the fact that in this class of spaces the notions of a separator and a cut are equivalent for closed subsets; cf. Kuratowski [1968], p. 258). Brouwer did not study the new invariant closely: he only used it to give another proof that  $I^n$ and  $I^m$  are not homeomorphic if  $n \neq m$ .

Referring to the second part of Lebesgue's paper [1911], Mazurkiewicz proved in [1915] that for every continuous parametric representation of the square  $I^2$  on the interval I, some fibres of f have cardinality at least 3, and showed that every continuum  $C \subset R^2$  whose interior in  $R^2$  is empty can be represented as a continuous image of the Cantor set under a mapping with fibres of cardinality at most 2. These results led him to define the dimension of a compact metric space X as the smallest integer n with the property that the space X can be represented as a continuous image of a closed subspace of the Cantor set under a mapping f such that  $|f^{-1}(x)| \leq n+1$  for every  $x \in X$ . As was proved later (cf. Problem 1.7.D), this definition is equivalent to the definition of the small inductive dimension, but Mazurkiewicz's paper had no influence on the development of dimension theory.

The definition of the small inductive dimension ind was formulated by Urysohn in [1922] and by Menger in [1923], both papers contain also Theorem 1.1.2. Menger and Urysohn, working independently, built the framework of the dimension theory of compact metric spaces, but Urysohn was ahead of Menger by a few months and was able to establish a larger number of basic properties of dimension. Urysohn's results are presented

#### Problems

in a two-part paper, [1925] and [1926], published after the author's death in 1924, whereas Menger's results are contained in his papers [1923] and [1924] and in his book [1928]. A generalization of dimension theory to separable metric spaces is due to Tumarkin ([1925] and [1926]) and Hurewicz ([1927] and [1927b]). In [1927] Hurewicz, in a particularly successful way, made use of the inductive character of dimension and greatly simplified the proofs of some important theorems, e.g., the sum theorem and the decomposition theorem. Moreover, owing to his discovery of the compactification theorem, Hurewicz reduced, in a sense, the dimension theory of separable metric spaces to the dimension theory of compact metric spaces.

When the work of Menger and Urysohn drew the attention of mathematicians to the notion of dimension, Brouwer (in [1923], [1924], [1924a] and [1924b]) ascertained that the definition of his Dimensionsgrad was marred by a clerical error and that it should read exactly as the definition of the large inductive dimension (see Section 1.6) and thus should lead to the same notion of dimension for compact metric spaces; he also commented that even the original faulty definition of Dimensionsgrad could serve as a basis for an equally good, although different, dimension theory. Brouwer's arguments do not seem quite convincing. After the publication of Menger's book [1928] a heated discussion arose between Brouwer ([1928]) and Menger ([1929a], [1930], [1933]) concerning priority in defining the notion of dimension; a good account of this discussion is contained in Freudenthal's notes in the second volume of Brouwer's Collected Papers (Brouwer [1976]). The history of the first years of dimension theory and, in particular, an evaluation of the contributions of Menger and Urysohn can be found in Alexandroff [1951].

# Problems

**1.1.A.** Observe that a metric space X satisfies the inequality  $\operatorname{ind} X \leq n \geq 0$  if and only if for every point  $x \in X$  and each positive number  $\varepsilon$  there exists a neighbourhood  $U \subset X$  of the point x such that  $\delta(U) < \varepsilon$  and  $\operatorname{ind} \operatorname{Fr} U \leq n-1$ .

**1.1.B.** To every regular space X and every point  $x \in X$  one assigns the *dimension of X at the point x*, denoted by  $\operatorname{ind}_x X$ , which is an integer larger than or equal to 0 or the infinite number  $\infty$ ; the definition consists

in the following conditions: (1)  $\operatorname{ind}_x X \leq n$  if for each neighbourhood  $V \subset X$  of the point x there exists an open set  $U \subset X$  such that  $x \in U \subset V$  and  $\operatorname{ind} \operatorname{Fr} U \leq n-1$ ; (2)  $\operatorname{ind}_x X = n$  if  $\operatorname{ind}_x X \leq n$  and  $\operatorname{ind}_x X > n-1$ ; (3)  $\operatorname{ind}_x X = \infty$  if  $\operatorname{ind}_x X > n$  for n = 0, 1, ...

(a) Note that  $ind X \leq n$  if and only if  $ind_x X \leq n$  for every  $x \in X$ .

(b) Formulate and prove the counterparts of 1.1.2, 1.1.4 and 1.1.6 for the dimension at a point.

**1.1.C.** Show that whenever regular spaces X and Y are homeomorphic, then indX = indY.

**1.1.D.** Give an example of a subspace X of the plane and of a closed set  $C \subset X$  with the property that for a pair A, B of disjoint closed subsets of X the set C is a cut between A and B but is not a separator between A and B.

#### 1.2. The separation and enlargement theorems for dimension 0

A regular space X satisfying the equality ind X = 0 will be called a zerodimensional space.

To begin with, we shall specialize the contents of the previous section to the case of zero-dimensional spaces.

**1.2.1. Proposition.** A regular space X is zero-dimensional if and only if X is non-empty and for every point  $x \in X$  and each neighbourhood  $V \subset X$  of the point x there exists an open-and-closed set  $U \subset X$  such that  $x \in U \subset V$ .

**1.2.2. Proposition.** Every non-empty subspace of a zero-dimensional space is zero-dimensional.

**1.2.3. Proposition.** A regular space X is zero-dimensional if and only if X is non-empty and for every point  $x \in X$  and each closed set  $B \subset X$  such that  $x \notin B$  the empty set is a partition between x and B.

**1.2.4.** Proposition. A separable metric space X is zero-dimensional if and only if X is non-empty and has a countable base consisting of open-and-closed sets.  $\Box$ 

We shall now discuss a few examples.

**1.2.5. Examples.** The space of irrational numbers  $P \subset R$  is zero-dimensional because it has a countable base consisting of open-and-closed sets, viz., the sets of the form  $P \cap (a, b)$ , where a and b are rational numbers.

Similarly, the space of rational numbers  $Q \subset R$  is zero-dimensional. More generally, if a metric space X satisfies the condition 0 < |X| < c, then  $\operatorname{ind} X = 0$ . Indeed, for every point  $x \in X$  and each neighbourhood  $V \subset X$  of the point x there exist a positive number r such that  $B(x, r) \subset V$  and a positive number t < r such that  $\varrho(x, y) \neq t$  for every  $y \in X$ , where  $\varrho$  is the metric on the space X; the set U = B(x, t) satisfies the condition  $x \in U \subset V$  and is open-and-closed, because  $\operatorname{Fr} U \subset \{y \in X: \varrho(x, y) = t\} = \emptyset$ .

A non-empty subspace X of the real line R is zero-dimensional if and only if it does not contain any interval. Since intervals are connected and no connected space containing at least two points is zero-dimensional, the condition is necessary by virtue of Proposition 1.2.2. The condition is also sufficient; indeed, the sets of the form  $X \cap (a, b)$ , where  $a, b \in R$ and a < x < b, constitute a base for the space X at the point x and for each  $V = X \cap (a, b)$  one can find an open-and-closed set  $U \subset V$  such that  $x \in U \subset V$ , it suffices to define  $U = X \cap (c, d)$ , where  $c \in (a, x) \setminus X$  and  $d \in (x, b) \setminus X$ .

In particular, the subspace C of the real line consisting of all real numbers in the closed unit interval I that have a tryadic expansion in which the digit 1 does not occur, i.e., the set of all numbers of the form  $x = \sum_{i=1}^{\infty} \frac{2x_i}{3^i}$ , where  $x_i$ is equal to 0 or 1 for i = 1, 2, ..., is zero-dimensional. Indeed, the set C does not contain any interval because  $C = \bigcap_{i=1}^{\infty} F_i$ , where  $F_i$  is the subset of I consisting of all numbers having a tryadic expansion in which 1 does not occur as the *j*-th digit for  $j \leq i$ , and  $F_i$  contains no interval of length larger than  $1/3^i$ . One easily sees that the set  $F_1$  is obtained from I by removing the "middle" intervals (1/3, 2/3), the set  $F_2$  is obtained from  $F_1$  by removing the "middle" intervals (1/9, 2/9) and (7/9, 8/9) of both parts of  $F_1$ , and so on. The set  $F_i$  consists of  $2^i$  disjoint intervals of length  $1/3^i$ .

The subspace C of the real line is called the *Cantor set*. Since C is a closed subset of I, the Cantor set is compact.

The Cartesian product  $Q^n \subset R^n$  is zero-dimensional, because it is a countable space. The Cartesian product  $P^n \subset R^n$  is also zero-dimensional; the proof is left to the reader (cf. Theorem 1.3.6).

As shown in the following theorem, zero-dimensional separable metric spaces have a separation property which is much stronger than the property described in Proposition 1.2.3.

**1.2.6.** The first separation theorem for dimension 0. If X is a zero-dimensional separable metric space, then for every pair A, B of disjoint closed subsets of X the empty set is a partition between A and B, i.e., there exists an openand-closed set  $U \subset X$  such that  $A \subset U$  and  $B \subset X \setminus U$ .

**Proof.** For every  $x \in X$  there exists an open-and-closed set  $W_x \subset X$  such that  $x \in W_x$  and

(1) either  $A \cap W_x = \emptyset$  or  $B \cap W_x = \emptyset$ .

The open cover  $\{W_x\}_{x \in X}$  of the space X has a countable subcover  $\{W_{x_i}\}_{i=1}^{\infty}$ . The sets

 $U_i = W_{x_i} \bigvee_{j < i} W_{x_j} \subset W_{x_i}, \text{ where } i = 1, 2, ...,$ 

are open and constitute a cover of the space X. Let us define

$$U = \bigcup \{U_i : A \cap U_i \neq \emptyset\}$$
 and  $W = \bigcup \{U_i : A \cap U_i = \emptyset\};$ 

obviously,  $A \subset U$  and it follows from (1) that  $B \subset W$ . Since the sets  $U_i$  are pairwise disjoint,  $W = X \setminus U$ , which implies that the set U is openand-closed and that  $B \subset X \setminus U$ .

**1.2.7. Remark.** It follows from the above proof that in Theorem 1.2.6 the assumption that X is a separable metric space can be replaced by the weaker assumption that X is a *Lindelöf space*, i.e., a regular space which has the property that every open cover of X has a countable subcover.

Now we are going to prove the second separation theorem, which is still stronger than Theorem 1.2.6. Let us recall that two subsets A and Bof a topological space X are separated if  $A \cap \overline{B} = \emptyset = \overline{A} \cap B$ . One easily sees that the sets A and B are separated if and only if they are disjoint and open (or—equivalently—closed) in their union  $A \cup B$ , i.e., if  $A \cap B = \emptyset$ and the empty set is a partition between A and B in the subspace  $A \cup B$ of X. In particular, two disjoint open sets, and also two disjoint closed sets, are separated. The second separation theorem will be deduced from two lemmas; with a view to further applications, the second lemma is formulated in a more general way than needed in this section.

**1.2.8. Lemma.** For every pair A, B of separated sets in a metric space X there exist open sets  $U, W \subset X$  such that

(2) 
$$A \subset U, \quad B \subset W \quad and \quad U \cap W = \emptyset.$$

**Proof.** Let  $\varrho$  be the metric on the space X and let  $f(x) = \varrho(x, A)$  and  $g(x) = \varrho(x, B)$  denote the distance of the point  $x \in X$  from A and B, respectively. Since the functions f and g are continuous, the sets

$$U = \{x \in X: f(x) - g(x) < 0\} \text{ and } W = \{x \in X: f(x) - g(x) > 0\}$$

are open. The inclusions in (2) follow from the equalities  $f^{-1}(0) = \overline{A}$ and  $g^{-1}(0) = \overline{B}$ ; the equality  $U \cap W = \emptyset$  follows directly from the definition of U and W.

**1.2.9. Lemma.** Let M be a subspace of a metric space X and A, B a pair of disjoint closed subsets of X. For every partition L' in the space M between  $M \cap \overline{V}_1$  and  $M \cap \overline{V}_2$ , where  $V_1$ ,  $V_2$  are open subsets of X such that  $A \subset V_1$ ,  $B \subset V_2$  and  $\overline{V}_1 \cap \overline{V}_2 = \emptyset$ , there exists a partition L in the space X between A and B which satisfies the inclusion  $M \cap L \subset L'$ .

If M is a closed subspace of a metric space X and A, B a pair of disjoint closed subsets of X, then for every partition L' in the space M between  $M \cap A$  and  $M \cap B$  there exists a partition  $L_{q}$  in the space X between A and B which satisfies the inclusion  $M \cap L \subset L'$ .

**Proof.** Let U', W' be open subsets of M satisfying the conditions

 $M \cap \overline{V_1} \subset U', \quad M \cap \overline{V_2} \subset W', \quad U' \cap W' = \emptyset \quad \text{and} \quad M \setminus L' = U' \cup W'.$ Observe that

$$(3) A \cap \overline{W}' = \emptyset = B \cap \overline{U'}.$$

Indeed, since  $V_1 \cap W' = M \cap V_1 \cap W' \subset U' \cap W' = \emptyset$  and since the set  $V_1$  is open, we have  $V_1 \cap \overline{W'} = \emptyset$ , which implies that  $A \cap \overline{W'} = \emptyset$ ; by symmetry of assumptions, also  $B \cap \overline{U'} = \emptyset$ .

The sets U' and W' are disjoint and open in their union  $U' \cup W'$ , and thus they are separated, i.e.,

(4) 
$$U' \cap \overline{W'} = \emptyset = \overline{U'} \cap W'.$$

If follows from (3) and (4) that the sets  $A \cup U'$  and  $B \cup W'$  are also separated. Hence, by Lemma 1.2.8, there exist open sets  $U, W \subset X$  such that

$$A \cup U' \subset U$$
,  $B \cup W' \subset W$  and  $U \cap W = \emptyset$ .

The set  $L = X \setminus (U \cup W)$  is a partition in the space X between A and B. Since

$$M \cap L = M \setminus (U \cup W) \subset M \setminus (U' \cup W') = L',$$

the first part of the lemma is established.



Fig. 3

To prove the second part, consider open subsets  $U_1$ ,  $W_1$  of the space M satisfying the conditions

 $M \cap A \subset U_1$ ,  $M \cap B \subset W_1$ ,  $U_1 \cap W_1 = \emptyset$  and  $M \setminus L' = U_1 \cup W_1$ . Since  $A \cap (M \setminus U_1) = \emptyset$ ,  $B \cap (M \setminus W_1) = \emptyset$  and  $A \cap B = \emptyset$ , there exist open sets  $V_1$ ,  $V_2 \subset X$  such that

$$A \subset V_1 \subset \overline{V}_1 \subset X \setminus (M \setminus U_1), \quad B \subset V_2 \subset \overline{V}_2 \subset X \setminus (M \setminus W_1)$$
  
and  $\overline{V}_1 \cap \overline{V}_2 = \emptyset.$ 

Obviously, L' is a partition in the space M between  $M \cap \overline{V}_1$  and  $M \cap \overline{V}_2$ , so that the partition L exists by the first part of the lemma.

**1.2.10.** Remark. In the proof of Lemma 1.2.9 only the fact that Lemma 1.2.8 holds in metric spaces was applied; as the latter lemma holds in hereditarily normal spaces (see Theorem 2.1.1), Lemma 1.2.9 also holds in hereditarily normal spaces.

**1.2.11.** The second separation theorem for dimension 0. If X is an arbitrary metric space and Z a zero-dimensional separable subspace of X, then for

every pair A, B of disjoint closed subsets of X there exists a partition L between A and B such that  $L \cap Z = \emptyset$ .

**Proof.** Consider open sets  $V_1, V_2 \subset X$  such that  $A \subset V_1, B \subset V_2$  and  $\overline{V}_1 \cap \overline{V}_2 = \emptyset$ . By virtue of Theorem 1.2.6, the empty set is a partition in the space Z between  $Z \cap \overline{V}_1$  and  $Z \cap \overline{V}_2$ . Applying the first part of Lemma 1.2.9 we obtain the required partition L.  $\Box$ 

The last theorem yields a characterization of zero-dimensional subspaces in terms of neighbourhoods in the whole space (cf. the proof of Proposition 1.1.4):

**1.2.12. Proposition.** A separable subspace M of an arbitrary metric space X is zero-dimensional if and only if M is non-empty and for every point  $x \in M$  (or—equivalently—for every point  $x \in X$ ) and each neighbourhood V of the point x in the space X there exists an open set  $U \subset X$  such that  $x \in U \subset V$  and  $M \cap \operatorname{Fr} U = \emptyset$ .  $\Box$ 

Proposition 1.2.12 and Lemma 1.1.5 imply

**1.2.13. Proposition.** A subspace M of a separable metric space X is zerodimensional if and only if M is non-empty and X has a countable base  $\mathscr{B}$ such that  $M \cap \operatorname{Fr} U = \emptyset$  for every  $U \in \mathscr{B}$ .

It is natural to ask at this point whether every zero-dimensional subspace of a given space can be enlarged to a "better" zero-dimensional subspace. The example of the subspace of the real line consisting of rational numbers shows that, generally, zero-dimensional subspaces cannot be enlarged to closed zero-dimensional subspaces; however, as shown in the next theorem, they can always be enlarged to zero-dimensional  $G_{\delta}$ -sets. Let us recall that  $G_{\delta}$ -sets are defined as countable intersections of open sets, and  $F_{\sigma}$ -sets as their complements, i.e., countable unions of closed sets.

**1.2.14.** The enlargement theorem for dimension 0. For every zero-dimensional separable subspace Z of an arbitrary metric space X there exists a  $G_{\delta}$ -set  $Z^*$  in X such that  $Z \subset Z^*$  and the subspace  $Z^*$  of the space X is zero-dimensional.

**Proof.** Since every closed subset of a metric space is a  $G_{\delta}$ -set and a  $G_{\delta}$ -set in a subspace which itself is a  $G_{\delta}$ -set is a  $G_{\delta}$ -set in the space, one can assume

that  $\overline{Z} = X$ . Thus X is separable and, by virtue of Proposition 1.2.13, has a countable base  $\mathscr{B}$  such that  $Z \cap \operatorname{Fr} U = \emptyset$  for every  $U \in \mathscr{B}$ . The union  $F = \bigcup \{\operatorname{Fr} U : U \in \mathscr{B}\}$  is an  $F_{\sigma}$ -set, and its complement  $Z^* = X \setminus F$ is a  $G_{\sigma}$ -set which contains the set Z. From Proposition 1.2.13 it follows that  $Z^*$  is zero-dimensional.  $\Box$ 

As the reader has undoubtedly observed, Theorem 1.2.6 states that in separable metric spaces two properties, viz., the property that the empty set is a partition between any disjoint closed sets A, B, and the property that the empty set is a partition between every point x and each closed set B such that  $x \notin B$ , are equivalent. The question arises whether the property that the empty set is a partition between any distinct points x, y is still the same property. As shown in the following example, the answer to this question is negative (cf. the notion of a totally disconnected space discussed in Section 1.4).

**1.2.15. Erdös' example.** Let us recall that *Hilbert space H* consists of all infinite sequences  $\{x_i\}$  of real numbers such that the series  $\sum_{i=1}^{\infty} x_i^2$  is convergent. For every point  $x = \{x_i\} \in H$  the norm ||x|| of the point *x* is the number  $\sqrt{\sum_{i=1}^{\infty} x_i^2}$ , and the distance between  $x = \{x_i\}$  and  $y = \{y_i\}$  is defined by

$$\varrho(x,y) = \sqrt{\sum_{i=1}^{\infty} (x_i - y_i)^2} ,$$

i.e., is equal to the norm of the difference x-y. The function  $\varrho$  is a metric on H, and H is a separable metric space.

We shall show that, in the subspace  $H_0$  of the space H consisting of the points  $\{x_i\} \in H$  such that  $x_i$  is rational for every *i*, the empty set is a partition between any distinct points x, y, and yet  $H_0$  is not zero-dimensional.

Let us consider a pair  $x = \{x_i\}$ ,  $y = \{y_i\}$  of distinct points of  $H_0$ . There exists a natural number  $i_0$  such that  $x_{i_0} \neq y_{i_0}$ ; without loss of generality one can assume that  $x_{i_0} < y_{i_0}$ . Take an irrational number t such that  $x_{i_0} < t < y_{i_0}$  and define

$$U = \{\{z_i\} \in H_0: z_{i_0} < t\}.$$

One can easily verify that the set U is open-and-closed in  $H_0$ . As  $x \in U \subset H_0 \setminus \{y\}$ , the empty set is a partition between x and y.

Now, let  $x_0 \in H_0$  be the sequence whose terms are all equal to zero and let  $V = B(x_0, 1) = \{x \in H_0: ||x|| < 1\}$ . We shall show that for every neighbourhood U of the point  $x_0$  which is contained in the neighbourhood V of  $x_0$  we have  $\operatorname{Fr} U \neq \emptyset$ .

We shall define inductively a sequence  $a_1, a_2, \ldots$  of rational numbers such that

(5) 
$$x_k = (a_1, a_2, ..., a_k, 0, 0, ...) \in U$$
 and  $\varrho(x_k, H_0 \setminus U) \leq 1/k$ 

for k = 1, 2, ... Conditions (5) are satisfied for k = 1 if we let  $a_1 = 0$ . Suppose that the rational numbers  $a_1, a_2, ..., a_{m-1}$  are already defined and conditions (5) are satisfied for  $k \leq m-1$ . The sequence

$$x_i^m = (a_1, a_2, \dots, a_{m-1}, i/m, 0, 0, \dots)$$

is an element of  $H_0$  for i = 1, 2, ..., m. As  $x_0^m = x_{m-1} \in U$  and  $x_m^m \notin U$ because  $||x_m^m|| \ge 1$ , there exists an  $i_0 < m$  such that  $x_i^m \in U$  and  $x_{i_0+1}^m \notin U$ . One easily sees that conditions (5) are satisfied for k = m if we let  $a_m = i_0/m$ . Thus the sequence  $a_1, a_2, ...$  is defined. From the first part of (5) it follows that  $\sum_{i=1}^{k} a_i^2 < 1$  for k = 1, 2, ..., so that  $\sum_{i=1}^{\infty} a_i^2 \le 1$ . Hence  $a = \{a_i\}$  is a point of  $H_0$  and  $a \in \overline{U}$ . On the other hand, from the second part of (5) it follows that  $a \in \overline{H_0 \setminus U}$ , so that  $a \in \operatorname{Fr} U$  and  $\operatorname{Fr} U \neq \emptyset$ .

Thus we have shown that there is no open-and-closed set  $U \subset H_0$  such that  $x_0 \in U \subset V$ , and this implies that the space  $H_0$  is not zero-dimensional.

To conclude, let us observe that the only open-and-closed bounded subset of the space  $H_0$  is the empty set. Indeed, if there existed a nonempty open-and-closed bounded set  $W \subset H_0$ , then by a suitable translation and contraction of W we could obtain an open-and-closed set  $U \subset H_0$  such that  $x_0 \in U \subset V$ .  $\Box$ 

#### Historical and bibliographic notes

Zero-dimensional spaces were defined by Sierpiński in [1921], before dimension theory was originated. Sierpiński's objective was to compare a few classes of metric spaces which are highly disconnected; a similar comparison will be drawn in Section 1.4. The theorems in the present section are all special cases of theorems which will be proved in Section 1.5 for an arbitrary dimension n; they were at once established in this more general form. Theorem 1.2.6 was proved for compact metric spaces by Menger in [1924] and by Urysohn in [1926] and was extended to separable metric spaces by Tumarkin in [1926] (announcement in [1925]) and by Hurewicz in [1927]; the generalization stated in Remark 1.2.7 was obtained by Vedenissoff in [1939]. Theorem 1.2.11 was proved for compact metric spaces by Menger in [1924] and was extended to separable metric spaces by Menger in [1924]. Theorem 1.2.14 was obtained by Tumarkin in [1926] (announcement [1925]). The space in Example 1.2.15 was described by Erdös in [1940]; the first example of a space with similar properties was given by Sierpiński in [1921].

## Problems

**1.2.A.** Let M be a subspace of a metric space X and let x be a point of M. Prove that  $\operatorname{ind}_{x} M = 0$  if and only if there exists a base  $\{U_{i}\}_{i=1}^{\infty}$  for the space X at the point x such that  $M \cap \operatorname{Fr} U_{i} = \emptyset$  for i = 1, 2, ...

**1.2.B.** Show that a subspace M of a metric space X is zero-dimensional if and only if M is non-empty and for every point  $x \in M$  and each neighbourhood V of the point x in the space X there exists an open set  $U \subset X$  such that  $x \in U \subset V$  and  $M \cap \operatorname{Fr} U = \emptyset$  (cf. Proposition 1.2.12 and Problem 4.1.C).

**1.2.C.** Show by an example that in the second part of Lemma 1.2.9 the assumption that the subspace M is closed cannot be omitted.

**1.2.D.** Check that every countable compact space has isolated points. Note that locally compact countable spaces have the same property.

Hint. Arrange all points of the space into a sequence  $x_1, x_2, ...$  and, assuming that none of the points  $x_i$  is isolated, define a decreasing sequence  $F_1 \supset F_2 \supset ...$  of non-empty closed sets such that  $x_i \notin F_i$  for i = 1, 2, ... One can equally well apply the Baire category theorem.

*Remark.* Every completely metrizable space with no isolated points contains a subspace homeomorphic to the Cantor set and thus is of cardinality at least c (see [GT], Problem 4.5.5(a)).

Ch. 1, § 2] The separation and enlargement theorems for dimension 0

**1.2.E** (Kuratowski [1932a]). Define a mapping f of the Cantor set C to the interval [-1, 1] by letting

$$f(0) = 0$$
 and  $f(x) = \frac{(-1)^{i_1}}{2} + \frac{(-1)^{i_2}}{2^2} + \dots$   
for each  $x = \sum_{i=1}^{\infty} \frac{2x_i}{3^i} \in C \setminus \{0\},$ 

where  $i_1 < i_2 < ...$  is the sequence (finite or infinite) of all *i*'s such that  $x_i = 1$ ; let  $C_0$  denote the set of all points  $x \in C \setminus \{0\}$  for which the sequence  $i_1, i_2, ...$  is infinite and let  $C_1 = C \setminus C_0$ .

(a) Observe that the set  $C_1$  consists of the number 0 and the right end-points of the intervals (1/3, 2/3), (1/9, 2/9), (7/9, 8/9), ... removed from I to obtain the Cantor set; note that  $C_1$  is a countable set.

(b) Check that the function f is continuous at all points of  $C_0$ .

(c) Verify that for every  $x = \frac{2}{3^{i_1}} + \frac{2}{3^{i_2}} + \dots + \frac{2}{3^{i_k}} \in C_1$ , the upper limit  $\overline{f(x)}$  and the lower limit  $\underline{f(x)}$  of the function f at the point x are equal to  $f(x) + \frac{1}{2^k}$  and  $f(x) - \frac{1}{2^k}$ , respectively; observe that the function f is discontinuous at all points of  $C_1$ .

(d) For each point  $x \in C_1$  define two sequences,  $\{x'_n\}$  and  $\{x''_n\}$ , of points in  $C_1$  such that

$$\lim x'_n = x$$
,  $\lim \overline{f(x'_n)} = \overline{f(x)}$ ,  $\lim \underline{f(x'_n)} = f(x)$ 

and

$$\lim x_n'' = x, \quad \lim \underline{f(x_n')} = \underline{f(x)}, \quad \lim \overline{f(x_n')} = f(x).$$

(e) Consider the graph  $K = \{(x, f(x)): x \in C\}$  of the function f and prove that  $\operatorname{ind}_{(x, f(x))} K = 1$  for every  $x \in C_1$ .

*Hint.* Prove that the space  $D = K \cup \bigcup_{x \in C_1} (\{x\} \times [f(x), \overline{f(x)}]) \subset \mathbb{R}^2$  is compact. Assuming that  $\operatorname{ind}_{(x_0, f(x_0))} K = 0$  for an  $x_0 \in C_1$ , show that there exists a partition L in the space D between  $(x_0, f(x_0))$  and  $(x_0, \overline{f(x_0)})$  such that  $L \cap K = \emptyset$ . Show that the set M consisting of all points  $\overline{x} \in C_1$  such that L is a partition between either (x, f(x)) and  $(x, \overline{f(x)})$  or (x, f(x)) and  $(x, \overline{f(x)})$  is contained in the projection of L onto C and contains an isolated point (see Problem 1.2.D); deduce a contradiction of (d).

(f) Observe that  $\inf_{(x,f(x))} K = 1$  for every  $x \in C_0$ .

19

(g) Check that the empty set is a partition between any distinct points of the space K.

(h) Note that the space K is completely metrizable.

*Hint.* Check that K is a  $G_{\delta}$ -set in D and apply Lemma 1.3.12.

# 1.3. The sum, Cartesian product, universal space, compactification and embedding theorems for dimension 0

The theorems enumerated in the title of the section belong to the most important results of dimension theory. For the time being we shall only prove their special cases pertaining to zero-dimensional spaces. We begin with the sum theorem.

**1.3.1. The sum theorem for dimension** 0. If a separable metric space X can be represented as the union of a sequence  $F_1$ ,  $F_2$ , ... of closed zero-dimensional subspaces, then X is zero-dimensional.



Fig. 4

**Proof.** Consider a pair A, B of disjoint closed subsets of the space X. We shall prove that there exist open sets  $U, W \subset X$  such that

(1)  $A \subset U$ ,  $B \subset W$ ,  $U \cap W = \emptyset$  and  $X = U \cup W$ ,

i.e., that the empty set is a partition between A and B.

Let  $U_0$ ,  $W_0$  be open subsets of X such that

(2) 
$$A \subset U_0, \quad B \subset W_0 \quad \text{and} \quad \overline{U}_0 \cap \overline{W}_0 = \emptyset.$$

20

We shall define inductively two sequences  $U_0, U_1, U_2, ...$  and  $W_0, W_1, W_2, ...$  of open subsets of X satisfying for i = 0, 1, 2, ... the conditions:

(3) 
$$U_{i-1} \subset U_i$$
,  $W_{i-1} \subset W_i$  if  $i \ge 1$  and  $\overline{U}_i \cap \overline{W}_i = \emptyset$ .

(4) 
$$F_i \subset U_i \cup W_i$$
, where  $F_0 = \emptyset$ .

Clearly, the sets  $U_0$ ,  $W_0$  defined above satisfy both conditions for i = 0. Assume that the sets  $U_i$ ,  $W_i$  satisfying (3) and (4) are defined for all i < k.

The sets  $\overline{U}_{k-1} \cap F_k$  and  $\overline{W}_{k-1} \cap F_k$  are closed and disjoint; since the space  $F_k$  is zero-dimensional, by virtue of Theorem 1.2.6 there exists an open-and-closed subset V of  $F_k$  such that

(5) 
$$\overline{U}_{k-1} \cap F_k \subset V$$
 and  $\overline{W}_{k-1} \cap F_k \subset F_k \setminus V$ .

The set  $F_k$  being closed in X, the sets V and  $F_k \setminus V$  are also closed in X; from (5) it follows that

$$(\overline{U}_{k-1}\cup V)\cap [\overline{W}_{k-1}\cup (F_k\setminus V)] = (V\cap \overline{W}_{k-1})\cup [\overline{U}_{k-1}\cap (F_k\setminus V)] = \emptyset,$$

so that there exist open sets  $U_k$ ,  $W_k \subset X$  satisfying

$$\overline{U}_{k-1} \cup V \subset U_k$$
,  $\overline{W}_{k-1} \cup (F_k^{\prime} \setminus V) \subset W_k$  and  $\overline{U}_k \cap \overline{W}_k = \emptyset$ .

The sets  $U_k$ ,  $W_k$  satisfy (3) and (4) for i = k; thus, the construction of the sequences  $U_0$ ,  $U_1$ ,  $U_2$ , ... and  $W_0$ ,  $W_1$ ,  $W_2$ , ... is completed. It follows from (2), (3) and (4) that the unions  $U = \bigcup_{i=0}^{\infty} U_i$  and  $W = \bigcup_{i=0}^{\infty} W_i$  satisfy (1).

**1.3.2. Remark.** Undoubtedly, the reader has noted that in the proof of Theorem 1.3.1 only the normality of the space X and the fact that the empty set is a partition between each pair of disjoint closed subsets of the space  $F_k$  were applied. Hence we have proved that if a normal space X can be represented as the union of a sequence  $F_1, F_2, \ldots$  of closed subspaces with the property that for  $i = 1, 2, \ldots$  the empty set is a partition between each pair of disjoint closed subsets of the space  $F_i$ , then the empty set is a partition between each pair of disjoint closed subsets of the space  $F_i$ , then the empty set is a partition between any disjoint closed subsets of the space X.

From Theorem 1.3.1 several corollaries follow:

**1.3.3. Corollary.** If a separable metric space X can be represented as the union of a sequence  $F_1, F_2, \ldots$  of zero-dimensional subspaces, where  $F_i$  is an  $F_{\sigma}$ -set for  $i = 1, 2, \ldots$ , then X is zero-dimensional.

**Proof.** Let us suppose that  $A = \overline{A}$ ; the open set  $X \setminus A \subset B$  is zerodimensional by virtue of the subspace theorem. Since every open subset of a metric space is an  $F_{\sigma}$ -set and since  $X = A \cup (X \setminus A)$ , to complete the proof it suffices to apply Corollary 1.3.3.  $\Box$ 

**1.3.5. Corollary.** If by adjoining a finite number of points to a zero-dimensional separable metric space one obtains a metric space, then the space obtained is zero-dimensional and separable.  $\Box$ 

In connection with the last corollary, let us note that by adjoining countably many points to the space of irrational numbers one can obtain the real line, i.e., a space of positive dimension.

We shall now prove the Cartesian product theorem.

**1.3.6. The Cartesian product theorem for dimension** 0. The Cartesian product  $X = \prod_{i=1}^{\infty} X_i$  of a countable family  $\{X_i\}_{i=1}^{\infty}$  of regular spaces is zero-dimensional if and only if all spaces  $X_i$  are zero-dimensional.

**Proof.** If  $X \neq \emptyset$ , then each space  $X_i$  is homeomorphic to a subspace of X, so that if X is zero-dimensional, then all spaces  $X_i$  are zero-dimensional. To prove the reverse implication, it is enough to consider for i = 1, 2, ... a base  $\mathscr{B}_i$  for the space  $X_i$  consisting of open-and-closed sets and observe that the sets of the form  $U_1 \times U_2 \times ... \times U_k \times \prod_{i=k+1}^{\infty} X_i$ , where  $U_i \in \mathscr{B}_i$  for  $i \leq k$  and k = 1, 2, ..., constitute a base for X and are open-and-closed in X.  $\Box$ 

Theorems 1.1.2 and 1.3.6 yield

**1.3.7. Corollary.** The limit of an inverse sequence  $\{X_i, \pi_j^i\}$  of zero-dimensional spaces is either zero-dimensional or empty.  $\Box$ 

The sum and Cartesian product theorems allow us to increase our stock of zero-dimensional spaces.

**1.3.8. Examples.** For every pair k, n of integers satisfying  $0 \le k \le n \ge 1$  denote by  $Q_k^n$  the subspace of Euclidean n-space  $\mathbb{R}^n$  consisting of all points

which have exactly k rational coordinates. We shall prove that  $Q_k^n$  is a zerodimensional space.

For each choice of k distinct natural numbers  $i_1, i_2, ..., i_k$  not larger than n and each choice of k rational numbers  $r_1, r_2, ..., r_k$ , the Cartesian product  $\prod_{i=1}^{n} R_i$ , where  $R_{ij} = \{r_j\}$  for j = 1, 2, ..., k and  $R_i = R$  for  $i \neq i_j$ , is a closed subspace of  $R^n$ . Hence,  $Q_k^n \cap \prod_{i=1}^{n} R_i$  is a closed subspace of  $Q_k^n$ . Since the space  $Q_k^n \cap \prod_{i=1}^{n} R_i$  is homeomorphic to the subspace of  $R^{n-k}$  consisting of all points with irrational coordinates, it follows from Example 1.2.5 and Theorem 1.3.6 that it is a zero-dimensional space. Theorem 1.3.1 implies that the space  $Q_k^n \cap \prod_{i=1}^{\infty} R_i$  is countable and its union is equal to the whole of  $Q_k^n$ .

It also follows from Theorem 1.3.6 that the space  $Q^{\aleph_0}$ , which is the Cartesian product of  $\aleph_0$  copies of the space of rational numbers, and the space  $P^{\aleph_0}$ , which is the Cartesian product of  $\aleph_0$  copies of the space of irrational numbers, are zero-dimensional.

**1.3.9. Definition.** We say that a topological space X is *universal* for a class  $\mathscr{K}$  of topological spaces if X belongs to  $\mathscr{K}$  and every space in the class  $\mathscr{K}$  is homeomorphic to a subspace of the space X.

We are now going to prove that the Cantor set C and the space P of irrational numbers are universal spaces for the class of all zero-dimensional separable metric spaces; in the proof we shall apply the fact that both C and P can be represented as countable Cartesian products.

**1.3.10. Proposition.** The Cantor set C is homeomorphic to the Cartesian product  $D^{\aleph_0} = \prod_{i=1}^{\infty} D_i$ , where  $D_i$ , for i = 1, 2, ..., is the two-point discrete space  $D = \{0, 1\}$ .

**Proof.** As one readily verifies, for each point  $x \in C$  the representation in the form  $x = \sum_{i=1}^{\infty} \frac{2x_i}{3^i}$ , where  $x_i$ 's are equal to 0 or 1, is unique. Hence, letting

$$f(\lbrace x_i \rbrace) = \sum_{i=1}^{\infty} \frac{2x_i}{3^i} \quad \text{for} \quad \lbrace x_i \rbrace \in D^{\aleph_0},$$

we define a one-to-one mapping of  $D^{\aleph_0}$  onto C. Since the function  $f_i$ :  $D^{\aleph_0} \to I$  defined by  $f_i(\{x_i\}) = \frac{2x_i}{3^i}$  is continuous for i = 1, 2, ... and the series  $\sum_{i=1}^{\infty} f_i$  is uniformly convergent to f, the latter function is continuous. It follows from the compactness of  $D^{\aleph_0}$  that f is a homeomorphism.  $\Box$ 

The proof of the counterpart of Proposition 1.3.10 for the space of irrational numbers requires some calculation to remedy the lack of compactness. Let us recall that if X is a metric space, then by a *metric on the space* X we mean any metric on the set X which is equivalent to the original metric on X, i.e., induces the same convergence as the original metric.

**1.3.11. Lemma.** Let  $\varrho$  be an arbitrary metric on the space P of irrational numbers and  $\varepsilon$  a positive number. For every non-empty open set  $U \subset P$  there exists an infinite sequence  $F_1, F_2, \ldots$  of pairwise disjoint non-empty open-and-closed subsets of P such that  $U = \bigcup_{i=1}^{\infty} F_i$  and the diameters with respect to  $\varrho$  of all sets  $F_i$  are less than  $\varepsilon$ .

**Proof.** Consider an interval  $(a, b) \subset R$  with rational end-points such that  $(a, b) \cap P \subset U$  and divide it into  $\aleph_0$  pairwise disjoint non-empty intervals  $(a_1, b_1), (a_2, b_2), \ldots$  with rational end-points. Thus we have  $(a, b) \cap P = \bigcup_{l=1}^{\infty} A_l$ , where  $A_l = (a_l, b_l) \cap P \neq \emptyset$ ,  $a_l, b_l \in Q$  and  $A_l \cap A_j = \emptyset$  whenever  $i \neq j$ ; in addition let  $A_0 = U \setminus (a, b)$ . The sets  $A_0, A_1, A_2, \ldots$  are open in P and by virtue of Proposition 1.2.4 for  $i = 0, 1, 2, \ldots$  there exist in P open-and-closed sets  $A_{l,1}, A_{l,2}, \ldots$ , all of diameter less than  $\varepsilon$ , such that  $A_l = \bigcup_{j=1}^{\infty} A_{l,j}$ ; letting  $B_{l,j} = A_{l,j} \setminus \bigcup_{k < j} A_{l,k}$  for  $j = 1, 2, \ldots$ , we obtain pairwise disjoint open-and-closed subsets of P whose union is equal to  $A_l$ . To complete the proof it suffices to arrange all non-empty sets  $B_{l,j}$  into a simple sequence  $F_1, F_2, \ldots$ 

The next lemma is an important theorem on complete spaces. In con-

sideration of further applications it is stated in full generality and not merely for the subspace P of the real line R.

**1.3.12. Lemma.** Every  $G_{\delta}$ -set X in a completely metrizable space  $X_0$  is completely metrizable.

**Proof.** Let  $\rho$  be a complete metric on the space  $X_0$  and let  $X = \bigcap_{i=1}^{\infty} G_i$ , where  $G_i$  is open in  $X_0$  for i = 1, 2, ... Define

$$F_i = X_0 \setminus G_i$$
 and  $f_i(x) = 1/\varrho(x, F_i)$  for  $x \in X$  and  $i = 1, 2, ...;$ 

the functions  $f_1, f_2, ...$  from X to the real line R are continuous. One readily sees that the formula  $f(x) = (x, f_1(x), f_2(x), ...)$  defines a homeomorphic embedding  $f: X \to \prod_{i=0}^{\infty} X_i$ , where  $X_i = R$  for  $i \ge 1$ . Since the Cartesian product of countably many completely metrizable spaces and a closed subspace of a completely metrizable space are completely metrizable, to complete the proof it suffices to show that f(X) is a closed subset of  $\prod_{i=0}^{\infty} X_i$ . We shall show that every point  $x = \{x_i\} \in \prod_{i=0}^{\infty} X_i \setminus f(X)$  has a neighbourhood V contained in the complement of f(X).

We first consider the case where  $x_0 \in X$ . As  $x \notin f(X)$ , there exists a k > 0such that  $x_k \neq f_k(x_0)$ . Let  $U_1$  and  $U_2$  be disjoint neighbourhoods of  $x_k$ and  $f_k(x_0)$  in the real line. The functions  $f_k$  being continuous, there exists a neighbourhood  $U_0 \subset X_0$  of the point  $x_0$  such that  $f_k(U_0 \cap X) \subset U_2$ . One easily checks that

(6) 
$$x = \{x_i\} \in V = p_0^{-1}(U_0) \cap p_k^{-1}(U_1) \subset \prod_{i=0}^{\infty} X_i \setminus f(X),$$

where  $p_i$  denotes the projection of  $\prod_{i=0}^{\infty} X_i$  onto  $X_i$ .

Now, consider the case where  $x_0 \notin X$ ; thus we have  $x_0 \in F_k$  for a k > 0. Take a positive number r such that  $x_k+1 < 1/r$  and let  $U_0 = B(x_0, r)$ and  $U_1 = \{x \in \mathbb{R} : x < x_k+1\}$ . One easily checks that formula (6) also holds in this case.  $\Box$ 

**1.3.13. Proposition.** The space of irrational numbers P is homeomorphic to the Cartesian product  $N^{\aleph_0} = \prod_{i=1}^{\infty} N_i$ , where  $N_i$ , for i = 1, 2, ..., is the discrete space of natural numbers N.

**Proof.** By virtue of Lemma 1.3.12, there exists a complete metric  $\varrho$  on the space *P*. Applying Lemma 1.3.11, for every sequence  $k_1, k_2, ..., k_i$  of natural numbers define an open-and-closed subset  $F_{k_1k_2...k_i}$  of *P* such that

(7) 
$$P = \bigcup_{k=1}^{\infty} F_k$$
 and  $F_{k_1k_2...k_l} = \bigcup_{k=1}^{\infty} F_{k_1k_2...k_lk}$ .

(8) 
$$F_{k_1k_2...k_l} \neq \emptyset$$
 and  $\delta(F_{k_1k_2...k_l}) < 1/i$ .

(9) 
$$F_{k_1k_2...k_l} \cap F_{m_1m_2...m_l} = \emptyset$$
 whenever  $(k_1, k_2, ..., k_l) \neq (m_1, m_2, ..., m_l)$ .

It follows from (7) and (8) that for every  $\{k_i\} \in N^{\aleph_0}$  the subsets  $F_{k_1}, F_{k_1k_2}, F_{k_1k_2k_3}, \ldots$  of the space *P* form a decreasing sequence of non-empty closed sets whose diameters converge to zero; hence, by virtue of the Cantor theorem, the set  $\bigcap_{i=1}^{\infty} F_{k_1k_2\dots k_i}$  contains exactly one point, which we shall denote by  $f(\{k_i\})$ . Conditions (7)-(9) imply that by assigning  $f(\{k_i\})$  to  $\{k_i\} \in N^{\aleph_0}$  one defines a one-to-one mapping of  $N^{\aleph_0}$  onto *P*. Since, as one readily verifies,

$$f\Big(\{k_1\}\times\{k_2\}\times\ldots\times\{k_j\}\times\prod_{i=j+1}^{\infty}N_i\Big)=F_{k_1k_2\ldots k_j},$$

the mapping f is a homeomorphism, because the sets on the left-hand side of the last equality form a base for  $N^{\aleph_0}$  and the sets on the right-hand side form a base for P.  $\Box$ 

**1.3.14. Corollary.** The Cantor set is homeomorphic to a subspace of the space of irrational numbers.  $\Box$ 

Now we are ready to prove the universality of C and P.

**1.3.15. The universal space theorem for dimension** 0. The Cantor set and the space of irrational numbers are universal spaces for the class of all zero-dimensional separable metric spaces.

**Proof.** By virtue of Corollary 1.3.14, it suffices to show that for every zero-dimensional separable metric space X there exists a homeomorphic embedding  $f: X \to D^{\aleph_0}$ .

It follows from Proposition 1.2.4 that the space X has a countable base  $\mathscr{B} = \{U_i\}_{i=1}^{\infty}$  consisting of open-and-closed sets. For i = 1, 2, ...define a mapping  $f_i: X \to D_i = D$  by letting

$$f_t(x) = \begin{cases} 1 & \text{for } x \in U_t, \\ 0 & \text{for } x \in X \setminus U_t. \end{cases}$$

Ch. 1, § 3] The compactification and embedding theorems for dimension 0 27

Consider the mapping  $f: X \to D^{\aleph_0}$  defined by the formula  $f(x) = (f_1(x), f_2(x), ...)$ . Since for every natural number k we have

$$f(U_k) = f(X) \cap \left\{ \{ x_i \} \in D^{\aleph_0} \colon x_k = 1 \right\},$$

the mapping f is a homeomorphic embedding.  $\Box$ 

The universal space theorem implies the compactification theorem and the embedding theorem.

**1.3.16.** The compactification theorem for dimension 0. For every zero-dimensional separable metric space X there exists a zero-dimensional compactification  $\tilde{X}$ , i.e., a zero-dimensional compact metric space  $\tilde{X}$  which contains a dense subspace homeomorphic to X.

**Proof.** Let  $f: X \to C$  be a homeomorphic embedding of X in the Cantor set C. Since the Cantor set is compact, so is its closed subspace  $\tilde{X} = f(\overline{X})$ ; the space  $\tilde{X}$  is zero-dimensional by virtue of the subspace theorem.  $\Box$ 

**1.3.17.** The embedding theorem for dimension 0. Every zero-dimensional separable metric space is embeddable in the real line R.  $\Box$ 

Let us conclude this section by observing that in the theory of zerodimensional spaces the key role is played by four theorems, viz., the separation theorems, the sum theorem and the universal space theorem. All the remaining results either are elementary or easily follow from one of the four cited theorems. As the reader shall see later, the situation changes when we pass to higher dimensions.

**1.3.18. Remark.** In Theorems 1.3.6, 1.3.15 and 1.3.16 countability is not essential. In the same way one proves that the Cartesian product  $X = \prod_{s \in S} X_s$  of a family  $\{X_s\}_{s \in S}$  of regular spaces satisfies the equality  $\operatorname{ind} X = 0$  if and only if  $\operatorname{ind} X_s = 0$  for every  $s \in S$ , and that every regular space X satisfying  $\operatorname{ind} X = 0$  is embeddable in the Cantor cube  $D^{\mathfrak{m}}$  (i.e., the Cartesian product of  $\mathfrak{m}$  copies of the two-point discrete space D), which implies that X has a compactification  $\tilde{X} \subset D^{\mathfrak{m}}$  such that  $\operatorname{ind} \tilde{X} = 0$ ; the cardinal number  $\mathfrak{m}$  is the cardinality of a base for the space X consisting of open-and-closed sets.

#### Historical and bibliographic notes

The theorems in the present section are all special cases of theorems which will be proved in Sections 1.5, 1.7 and 1.11 for an arbitrary dimension n. Theorem 1.3.1 was proved for compact spaces (and for an arbitrary dimension n) by Menger in [1924] and by Urysohn in [1926] (announcement in [1922]); it was extended to separable metric spaces by Tumarkin in [1926] (announcement in [1925]) and by Hurewicz in [1927]. Theorem 1.3.6 was established by Kuratowski in [1933] and Theorem 1.3.15 by Sierpiński in [1921].

#### Problems

**1.3.A.** (a) Prove that if a separable metric space X can be represented as the union of a sequence  $F_0, F_1, F_2, \ldots$  of closed subspaces such that  $\operatorname{ind} F_i = 0$  for  $i = 1, 2, \ldots$ , then  $\operatorname{ind}_x X = 0$  for every point  $x \in F_0$  such that  $\operatorname{ind}_x F_0 = 0$ .

(b) Give an example of a separable metric space X which can be represented as the union of two closed subspaces  $F_1$  and  $F_2$  in such a way that for a point  $x \in F_1 \cap F_2$  we have  $\operatorname{ind}_x F_1 = \operatorname{ind}_x F_2 = 0$  and yet  $\operatorname{ind}_x X > 0$ .

**1.3.B.** Note that Theorem 1.3.1 for a subspace X of the real line is a consequence of the Baire category theorem. Deduce from Theorem 1.3.1 the Baire category theorem for the real line.

**1.3.C.** (a) (implicitly, Sierpiński [1928]) Prove that every non-empty closed subset A of a zero dimensional separable metric space X is a *retract* of X, i.e., that there exists a continuous mapping  $r: X \to A$  such that r(x) = x for every  $x \in A$ .

*Hint.* Represent the complement  $X \setminus A$  as the union of a sequence  $F_1, F_2, \ldots$  of pairwise disjoint open-and-closed sets such that  $\lim \delta(F_i) = 0$ . For  $i = 1, 2, \ldots$  choose a point  $x_i \in A$  such that  $\varrho(x_i, F_i) < \varrho(A, F_i) + 1/i$  and let  $r(x) = x_i$  for  $x \in F_i$ .

(b) Note that if a non-empty regular space X has the property that every non-empty closed set  $A \subset X$  is a retract of X, then X is zero-dimensional.

Ch. 1, § 3]

#### Problems

*Remark.* The characterization of zero-dimensional spaces contained in (a) and (b) generalizes to higher dimensions; see Problem 4.1.F.

1.3.D (Alexandroff [1927a] (announcement [1925]), Hausdorff [1927]). Check that letting

$$f(\{x_i\}) = \sum_{i=1}^{\infty} \frac{x_i}{2^i}$$
 for  $\{x_i\} \in D^{\aleph_0}$ ,

one defines a continuous mapping of the Cantor set  $D^{\aleph_0}$  onto the closed interval *I*. Verify that if *y* is an end-point of one of the intervals removed from *I* to obtain the Cantor set, then  $|f^{-1}(y)| = 2$ , and otherwise  $|f^{-1}(y)| = 1$ .

Define a continuous mapping of the Cantor set  $D^{\aleph_0}$  onto the Hilbert cube  $I^{\aleph_0}$  and—applying Problem 1.3.C together with the fact that  $I^{\aleph_0}$ is a universal space for the class of all separable metric spaces—show that every non-empty compact metric space is a continuous image of the Cantor set. Deduce that every non-empty separable metric space is an image of a zero-dimensional separable metric space under a one-to-one continuous mapping.

**1.3.E.** (a) (Mazurkiewicz [1917]) Prove that every  $G_{\delta}$ -set which is dense, and whose complement is also dense, in a completely metrizable separable zero-dimensional space is homeomorphic to the space of irrational numbers.

Hint. Modify the proof of Proposition 1.3.13.

(b) (Alexandroff and Urysohn [1928]) Show that every completely metrizable separable zero-dimensional space which does not contain any non-empty compact open subspace is homeomorphic to the space of irrational numbers.

Hint. Apply (a).

(c) Note that the subspace of the Cantor set C consisting of all points which are not end-points of intervals removed from I to obtain the Cantor set is homeomorphic to the space of irrational numbers.

1.3.F (Brouwer [1910]). Prove that every zero-dimensional compact metric space with no isolated points is homeomorphic to the Cantor set.

*Hint.* Modify the construction in the proof of Proposition 1.3.13 in such a way that the sets  $F_{k_1k_2...k_i}$  will be defined for  $k_1 \leq m_1, k_2 \leq m_2, ...$ ...,  $k_i \leq m_i$ , where  $m_1, m_2, ...$  is a sequence of powers of the number 2. **1.3.G.** (a) (Brouwer [1913a]; implicity, Fréchet [1910]) Prove that for any two countable dense subsets A, B of the real line R there exists a homeomorphism  $f: R \to R$  such that f(A) = B (see Problem 1.8.D).

*Hint.* Let  $A = \{a_1, a_2, ...\}$  and  $B = \{b_1, b_2, ...\}$ ; define inductively a function f from A to R by letting  $f(a_1) = b_1$  and taking as  $f(a_i)$  an element of B, with the smallest possible index, such that the conditions  $a_j < a_k$  and  $f(a_j) < f(a_k)$  are equivalent for  $j, k \leq i$ . Extend f over R and prove that this extension is a homeomorphism.

(b) (Fréchet [1910]) Prove that the space of rational numbers Q is a universal space for the class of all countable metric spaces.

*Hint*. Observe that countable metric spaces are embeddable in the real line; for a countable  $X \subset R$  apply (a) to the sets  $X \cup Q$  and Q.

**1.3.H.** (a) Prove that if  $A_1, A_2, B_1$  and  $B_2$  are countable dense subsets of the real line R satisfying the condition  $A_1 \cap A_2 = \emptyset = B_1 \cap B_2$ , then there exists a homeomorphism  $f: R \to R$  such that  $f(A_1) = B_1$  and  $f(A_2) = B_2$ .

Hint. See Problem 1.3.G(a).

(b) Show that for any two countable dense subsets A, B of the space of irrational numbers P there exists a homeomorphism  $f: P \to P$  such that f(A) = B.

(c) Show that for any two countable dense subsets A, B of the Cantor set C there exists a homeomorphism  $f: C \to C$  such that f(A) = B.

*Hint.* Observe that for every countable set  $A \subset C$  there exists a homeomorphism  $g: C \to C$  such that the set g(A) is disjoint from the set consisting of the end-points of all intervals removed from I to obtain the Cantor set.

(d) (Sierpiński [1920a] (announcement [1915])) Prove that every countable metric space dense in itself is homeomorphic to the space of rational numbers.

*Hint.* By virtue of (b) it suffices to show that every countable metric space X dense in itself is homeomorphic to a dense subspace of the space of irrational numbers P. To that end, embed X in P, consider the closure  $\overline{X} \subset P$ , remove in an appropriate way  $\aleph_0$  points from  $\overline{X} \setminus X$ , and apply Problem 1.3.E(a).

One can equally well use Problem 1.3.F and apply (c).

(e) Note that the subspace of the Cantor set C consisting of the endpoints of all intervals removed from I to obtain the Cantor set is homeomorphic to the space of rational numbers.

## 1.4. Various kinds of disconnectedness

We shall now compare the class of zero-dimensional spaces with three other classes of highly disconnected spaces. It will follow from this comparison that none of these three classes satisfies the counterparts of the theorems proved for zero-dimensional spaces in Sections 1.2 and 1.3. Hence, zero-dimensional spaces form the nicest class of highly disconnected spaces. The dimension functions that one could define inductively, in the way the function ind is defined, starting at the zero-level with another class of highly disconnected spaces instead of the class of zero-dimensional spaces, would not lead to a dimension theory as rich and harmonious as the theory based on the dimension function ind developed in this chapter.

**1.4.1. Definition.** A topological space X is called *totally disconnected* if for every pair x, y of distinct points of X there exists an open-and-closed set  $U \subset X$  such that  $x \in U$  and  $y \in X \setminus U$ , i.e., if the empty set is a partition between any distinct points x, y of the space X.

Clearly, every zero-dimensional space is totally disconnected.

Totally disconnected spaces are characterized by the property that their quasi-components are one-point sets. Let us recall that quasi-components of a topological space X are defined as the minimal non-empty intersections of open-and-closed subsets of X, i.e., a non-empty set  $K \subset X$ is a *quasi-component* of the space X if K can be represented as the intersection of open-and-closed sets and for every open-and-closed set  $U \subset X$ such that  $K \cap U \neq \emptyset$  we have  $K \subset U$ . The quasi-components of a space X constitute a decomposition of X into pairwise disjoint closed subsets.

**1.4.2. Definition.** A topological space X is called *hereditarily disconnected* if X does not contain any connected subspace of cardinality larger than one.

Every totally disconnected space is hereditarily disconnected. Indeed, if X is a totally disconnected space, then for each subspace  $M \subset X$  which contains at least two distinct points x, y the sets  $M \cap U$  and  $M \setminus U$ , where U is an open-and-closed subset of X such that  $x \in U$  and  $y \in X \setminus U$ , form a decomposition of the space M into two non-empty disjoint open subsets, so that the space M is not connected.

Hereditarily disconnected spaces are characterized by the property that their components are one-point sets. Let us recall that components
of a topological space X are defined as the maximal non-empty connected subsets of X, i.e., a non-empty set  $S \subset X$  is a *component* of the space X if S is connected and for every connected set  $A \subset X$  such that  $S \subset A$  we have S = A. The components of a space X constitute a decomposition of X into pairwise disjoint closed subsets.

**1.4.3. Definition.** A topological space X is called *punctiform*, or *discontinuous*, if X does not contain any continuum of cardinality larger than one.

Clearly, every hereditarily disconnected space is punctiform and every compact punctiform space is hereditarily disconnected. As shown in Example 1.4.8 below, there exist connected punctiform spaces of cardinality larger than one.

The reader can easily verify that the above three classes of spaces are closed with respect to the subspace operation.

We shall now show that in the realm of non-empty locally compact spaces the three classes under consideration coincide with the class of zerodimensional spaces.

**1.4.4. Lemma.** In every compact space quasi-components and components coincide.

**Proof.** To begin with, we shall prove that in an arbitrary topological space X quasi-components contain the components. Consider a component S of the space X. Let x be a point in S and K the quasi-component of the space X which contains the point x; we shall show that  $S \subset K$ . Take an open-and-closed set  $U \subset X$  which contains x. Since the sets  $S \cap U$  and  $S \setminus U$  are open in S and disjoint and since  $S \cap U \neq \emptyset$ , it follows from the connectedness of S that  $S \setminus U = \emptyset$ , i.e., that  $S \subset U$ . The set K being the intersection of all open-and-closed subsets of X which contain x, we have  $S \subset K$ .

To complete the proof it suffices to show that the quasi-components of a compact space are connected. Let us consider the decomposition of a quasi-component K of a compact space X into two disjoint closed sets A, B and let us assume that  $A \neq \emptyset$ . By the normality of compact spaces there exist open sets V,  $W \subset X$  such that

 $A \subset V$ ,  $B \subset W$  and  $V \cap W = \emptyset$ .

Denote by  $\mathscr{U}$  a family of open-and-closed subsets of X satisfying  $\bigcap \mathscr{U} = K$ .

Since  $\bigcap \mathscr{U} \subset V \cup W$ , the family  $\mathscr{F} = \{U \setminus (V \cup W): U \in \mathscr{U}\}$  of closed subsets of X has an empty intersection. It follows from the compactness of X that a finite subfamily of  $\mathscr{F}$  also has an empty intersection, i.e., that there exists a finite number of sets  $U_1, U_2, ..., U_k \in \mathscr{U}$  satisfying

$$U = U_1 \cap U_2 \cap \ldots \cap U_k \subset V \cup W.$$

The set U is open-and-closed. Since

$$\overline{V \cap U} \subset \overline{V} \cap U = \overline{V} \cap (V \cup W) \cap U = V \cap U,$$

the set  $V \cap U$  is also open-and-closed. From the relation  $\emptyset \neq A \subset V \cap U$  it follows that  $K \subset V \cap U$ , and so  $B \subset K \cap W \subset V \cap U \cap W = \emptyset$ , which proves that the quasi-component K is connected.  $\square$ 

**1.4.5. Theorem.** Zero-dimensionality, total disconnectedness, hereditary disconnectedness and punctiformness are equivalent in the realm of non-empty locally compact spaces.

**Proof.** It suffices to prove that every non-empty locally compact punctiform space is zero-dimensional. Consider a point  $x \in X$  and a neighbourhood  $V \subset X$  of the point x. By the local compactness of the space X the point x has a neighbourhod  $W \subset X$  such that the closure  $\overline{W}$  is compact. The subspace  $M = \overline{V \cap W}$  of the space X is compact and punctiform, so that it is hereditarily disconnected. By virtue of Lemma 1.4.4, the component  $\{x\}$  of the space M can be represented as the intersection of a family  $\mathscr{U}$  of openand-closed subsets of M. It follows from the compactness of M that there exists a finite number of sets  $U_1, U_2, \ldots, U_k \in \mathscr{U}$  such that the intersection  $U = U_1 \cap U_2 \cap \ldots \cap U_k$  is disjoint from the set  $M \setminus (V \cap W)$ . The set U is closed in M, and thus it is closed in X; on the other hand, the set Uis open in  $V \cap W$ , so that U is an open-and-closed subset of X. As  $x \in U$  $\subset V$ , the space X is zero-dimensional.  $\square$ 

We shall now describe three subspaces of the plane which exhibit the difference between the classes of zero-dimensional, totally disconnected, hereditarily disconnected and punctiform spaces. They are all closely related to the space  $H_0$  described in Example 1.2.15, which is itself a totally disconnected non zero-dimensional space.

**1.4.6. Example.** One readily checks that by assigning to every point  $\{x_i\}$  of the space  $H_0$  described in Example 1.2.15 the same point  $\{x_i\}$  in the Cartesian product  $Q^{\aleph_0}$  of  $\aleph_0$  copies of the space of rational numbers one

defines a one-to-one continuous mapping of  $H_0$  to  $Q^{\aleph_0}$ . Hence, by virtue of Theorems 1.3.6 and 1.3.15, there exists a one-to-one continuous mapping  $f: H_0 \to C$  of the space  $H_0$  to the Cantor set C. Since the Cantor set is homogeneous (i.e., for every pair x, y of distinct points of C there exists a homeomorphism of C onto itself which transforms x to y), one can suppose that  $f(x_0) = 0$ , where  $x_0 \in H_0$  is the sequence whose terms are all equal to zero. Letting

$$h(x) = \left(\frac{f(x)}{\max(1, ||x||)}, \frac{||x||}{1 + ||x||}\right) \quad \text{for } x \in H_0,$$

one defines a continuous mapping  $h: H_0 \to I^2$ . One can prove that h is a homeomorphic embedding (see Problem 1.4.B(a)). This implies that the subspace  $X = h(H_0)$  of the plane is totally disconnected but is not zero-dimensional; however, a direct proof of these properties of the space X is simpler.





To begin with, let us observe that by letting

$$G(y_1, y_2) = y_1 \cdot \max\left(1, \frac{y_2}{1 - y_2}\right)$$
 for  $0 \le y_1 \le 1$  and  $0 \le y_2 < 1$ 

one defines a continuous mapping  $G: I \times [0, 1] \to R$  which to the point  $h(x) \in X$  assigns the point  $f(x) \in C$ , so that the restriction  $g = G|X: X \to C$  is a one-to-one continuous mapping of the space X to the Cantor set C. From the existence of such a mapping it follows that X is a totally disconnected space. Indeed, for every pair x, y of distinct points of X there

exists an open-and-closed set  $V \subset C$  such that  $g(x) \in V$  and  $g(y) \in C \setminus V$ ; the inverse image  $U = g^{-1}(V)$  is an open-and-closed subset of X such that  $x \in U$  and  $y \in X \setminus U$ .

Now, consider an open-and-closed set  $U \subset X$  such that  $U \subset (I \times [0, 1/2)) \cap X$ . The inverse image  $h^{-1}(U) \subset H_0$  is an open-and-closed bounded subset of  $H_0$  and thus, by virtue of the final observation in Example 1.2.15, it is empty. Hence, there is no open-and-closed set  $U \subset X$  such that  $h(x_0) = (0, 0) \in U \subset (I \cap [0, 1/2)) \cap X$ , which shows that the space X is not zero-dimensional.  $\Box$ 

The following two examples are: a space  $Y \subset I^2$  which is hereditarily disconnected but is not totally disconnected and a space  $Z \subset I^2$  which is punctiform and connected; in both examples we shall use the notation introduced in Example 1.4.6.

**1.4.7. Example.** We shall show that the subspace of the plane  $Y = X \cup \{p\}$ , where p = (0, 1/2), is hereditarily disconnected but is not totally disconnected.

Consider a connected subspace A of the space Y. As  $G(p) = 0 \in C$ , the image G(A) is a connected subspace of the Cantor set and thus contains at most one point. It follows that A is contained in a fibre of the mapping G|Y. Since all fibres of G|Y are at most of cardinality 2, the set A either is empty or consists of exactly one point, and this implies that the space Y is totally disconnected.

Consider now an open-and-closed set  $U \subset Y$  such that  $p \in U$ . There exists a number  $a \in I \setminus C$  such that  $([0, a) \times \{1/2\}) \cap Y \subset U$ . One readily sees that the set

$$V = ([0, a) \times [0, 1/2)) \cap Y \setminus U = ([0, a) \times [0, 1/2]) \cap Y \setminus U$$

is open-and-closed in X. The inverse image  $h^{-1}(V) \subset H_0$  is an open-andclosed bounded subset of  $H_0$  and thus is empty. Hence the set V is also empty, which implies that  $h(x_0) = (0, 0) \in U$ . Thus for the pair x = p,  $y = h(x_0)$  of distinct points of Y there exists no open-and-closed set  $U \subset X$ such that  $x \in U$  and  $y \in Y \setminus U$ , i.e., the space Y is not totally disconnected.  $\Box$ 

**1.4.8. Example.** We shall show that the subspace of the plane  $Z = X \cup \{q\}$ , where q = (0, 1), is punctiform and connected.

Consider a continuum  $A \subset Z$ . The difference  $A \setminus \{q\} \subset X$  is an  $F_{\sigma}$ -set in A, and so  $A \setminus \{q\} = \bigcup_{i=1}^{\infty} A_i$ , where  $A_i$  is compact for i = 1, 2, ... For every *i* the restriction  $g_i = g|A_i: A_i \to g_i(A_i) \subset C$  is a homeomorphism, because it is a one-to-one continuous mapping defined on a compact space. Since  $\operatorname{ind} g_i(A_i) \leq 0$ , we have  $\operatorname{ind} A_i \leq 0$  for i = 1, 2, ... and -by virtue of Theorems 1.1.2 and 1.3.1—ind  $A \leq \operatorname{ind} (\{q\} \cup \bigcup_{i=1}^{\infty} A_i) = 0$ . Hence the set A either is empty or consists of exactly one point, and this implies that the space Z is punctiform.

Consider now an open-and-closed set  $U \subset Z$  such that  $q \in U$ . One readily sees that there exists a number  $a \in (0, 1)$  such that  $(I \times (a, 1]) \cap Z$  $\subset U$ . The inverse image  $h^{-1}(V) \subset H_0$ , where  $V = Z \setminus U$ , is an open-andclosed bounded subset of  $H_0$  and thus is empty. Hence U = Z, i.e., the space Z is connected.  $\Box$ 

In the table on p. 37 the basic properties of countable, zero-dimensional, totally disconnected, hereditarily disconnected, and punctiform spaces are compared; a plus means that a theorem holds in the corresponding class, a minus that it does not hold. Formal statements of the results in the table together with hints how to obtain them can be found in the problems below.

## Historical and bibliographic notes

Totally disconnected spaces were introduced by Sierpiński in [1921], hereditarily disconnected spaces-by Hausdorff in [1914], and punctiform spaces-by Janiszewski in [1912]. Theorem 1.4.5 was proved (for compact metric spaces) by Menger in [1923] and by Urysohn in [1925]. The first example of a totally disconnected space which is not zero-dimensional was given by Sierpiński in [1921]; Sierpiński's space is a completely metrizable subspace of the plane. The first example of a hereditarily disconnected space which is not totally disconnected was also given by Sierpiński in [1921]; this space is also a completely metrizable subspace of the plane. Example 1.4.7 is a simplified version of an example described by Roberts in [1956]. The first example of a punctiform space which is not hereditarily disconnected was described by Sierpiński in [1920]; this space is a connected subspace of the plane. An example of a completely metrizable punctiform and connected subspace of the plane was given by Mazurkiewicz in [1920]. A simple modification that leads from spaces described in Examples 1.4.6-1.4.8 to similar spaces which are, moreover, completely metrizable is sketched in Problem 1.4.B.

Punctiform spaces	Hereditarily discon- nected spaces	Totally disconnected spaces	Zero-dimensional spaces	Countable spaces	
I	I	I	+	I	Enlargement to a $G_{\delta}$ -set
+	!	I	+	+	Countable sum
+	I	1	+	+	Finite sum
+	1	I	+	+	Adjoining one point
+	+	+	+	;	Countable Cartesian product
÷	+	+	+	+	Finite Cartesian product
I	1	1	+	+	Universal space
I	I	l	+	1	Compactifi- cation
I	I	l	+	+	Embedding in the real line
i	I	I	+	+	Embedding in an Euclid- ean space

Various kinds of disconnectedness

Various kinds of disconnectedness

1

## Problems

1.4.A. (a) Observe that zero-dimensionality, total disconnectedness, hereditary disconnectedness and punctiformness are equivalent in the realm of regular spaces which can be represented as countable unions of compact subspaces.

(b) Note that zero-dimensionality, total disconnectedness, hereditary disconnectedness and punctiformness are equivalent in the realm of subspaces of the real line. Deduce that there exist totally disconnected spaces which are not embeddable in the real line.

**1.4.B.** (a) (Roberts [1956]) Prove that the mapping  $h: H_0 \to I^2$  defined in Example 1.4.6 is a homeomorphic embedding.

*Hint.* Prove that in Hilbert space H a sequence of points  $x^1, x^2, ...,$  where  $x^m = \{x_i^m\}$  for m = 1, 2, ..., converges to a point  $x = \{x_i\}$  if and only if the sequence  $||x^m||$  converges to ||x|| and the sequence  $x_i^1, x_i^2, ...$  converges to  $x_i$  for i = 1, 2, ... (cf. Example 1.5.17).

(b) Give examples of completely metrizable spaces  $X_1, Y_1, Z_1 \subset I^2$  such that  $X_1$  is totally disconnected but is not zero-dimensional,  $Y_1$  is hereditarily disconnected but is not totally disconnected, and  $Z_1$  is punctiform and connected.

*Hint.* Consider the subspace  $H_1$  of Hilbert space H consisting of the points  $\{x_i\} \in H$  such that  $x_i$  is irrational for every i and suitably modify the constructions in Examples 1.4.6–1.4.8. When proving that  $X_1$  is completely metrizable, apply Lemma 1.3.12. In the proof of complete metrizability of  $Y_1$  and  $Z_1$  use the fact that every completely metrizable subspace of  $I^2$  is a  $G_{\delta}$ -set (see [GT], Theorem 4.3.24).

**1.4.C.** (a) (Knaster and Kuratowski [1921]) Let C be the Cantor set on the interval  $[0, 1] \times \{0\} \subset \mathbb{R}^2$ ; denote by Q the set of the end-points of all intervals removed from  $[0, 1] \times \{0\}$  to obtain the Cantor set and let  $P = C \setminus Q$ . Join every point  $c \in C$  to the point  $q = (1/2, 1/2) \in \mathbb{R}^2$  by a segment  $L_c$  and denote by  $F_c$  the set of all points  $(x, y) \in L_c$ , where y is rational if  $c \in Q$  and y is irrational if  $c \in P$ . The subspace  $F = \bigcup_{c \in C} F_c$ of the plane is called the Knaster-Kuratowski fan.

Prove that the Knaster-Kuratowski fan is connected and punctiform. *Hint*. Let  $r_1, r_2, ...$  be the sequence of all rational numbers in the interval [0, 1/2] and let  $P_i \subset R^2$  be the horizontal line  $y = r_i$ . Suppose

#### Problems

that  $F = (F \cap A) \cup (F \cap B)$ , where the sets A and B are closed in  $\mathbb{R}^2$ ,  $F \cap A \cap B$ =  $\emptyset$  and  $q \in A$ . Consider the sets  $K_i = \{c \in C: A \cap B \cap L_c \cap P_i \neq \emptyset\}$ , check that they are closed and that  $\bigcup_{i=1}^{\infty} K_i \subset P$ . Show that  $B \cap L_c = \emptyset$  for every point  $c \in P \setminus \bigcup_{i=1}^{\infty} K_i$  and apply the Baire category theorem to prove that  $F \cap B = \emptyset$ .

(b) (Knaster and Kuratowski [1921]) Prove that the space  $F \setminus \{q\}$  is hereditarily disconnected but is not totally disconnected.

(c) (E. Pol [1978a]) Prove that every completely metrizable space X which contains a subspace homeomorphic to  $F \setminus \{q\}$  also contains a subspace homeomorphic to the closed unit interval.

*Hint.* Apply the Lavrentieff theorem (see [GT], Theorem 4.3.21) to reduce the problem to the case where X is a  $G_{\delta}$ -set in the plane and  $F \setminus \{q\} \subset X$ . Consider the set  $P \cap \bigcap_{i=1}^{\infty} p(X \cap P_i)$ , where p is the projection from q onto C and  $P_i$  is defined in the hint to part (a).

(d) Observe that modifying the construction of the space F by taking all points  $(x, y) \in L_c$ , where y is irrational if  $c \in Q$  and y is rational if  $c \in P$ , one obtains a zero-dimensional space.

1.4.D. Let K and D be the spaces discussed in Problem 1.2.E.

(a) Show that if  $\operatorname{ind}_z K = 1$ , then there exists a point  $p \in D \setminus K$  such that the empty set is not a partition in the space  $K' = K \cup \{p\}$  between z and p.

Hint. See the hint to Problem 1.2.E(e).

(b) Prove that the space K' defined in (a) is hereditarily disconnected but is not totally disconnected.

(c) Verify that the space K' is completely metrizable.

1.4.E. (a) Note that by adjoining a point to a totally disconnected space one can obtain a space which is not totally disconnected. Deduce that the counterpart of the sum theorem does not hold either for totally disconnected spaces or for hereditarily disconnected spaces, even in the case where the space is represented as the union of two closed subspaces. Show that the counterpart of the sum theorem holds for punctiform spaces.

(b) Check that the Cartesian product of a family of totally disconnected, hereditarily disconnected, or punctiform spaces is a space in the same class. (c) Note that the counterpart of the compactification theorem does not hold for any of the four classes of spaces in the table on p. 37 distinct from the class of zero-dimensional spaces.

**1.4.F** (Hilgers [1937]). (a) Let Z be a topological space and T a subspace of Z whose cardinality is equal to c. Prove that for every family  $\mathscr{G}$  of subspaces of the Cartesian product  $Z \times Z$  such that  $|\mathscr{G}| \leq c$  there exists a set  $H \subset Z \times Z$  satisfying the following conditions:

- (1) The projection of  $Z \times Z$  onto the first axis maps H in a one-to-one way onto the subspace T.
- (2) If  $H \subset G$  for a  $G \in \mathcal{G}$ , then G contains a set homeomorphic to Z.

Hint. Let  $\varphi$  be an arbitrary transformation of T onto  $\mathscr{G}$ ; define a mapping f of T to Z by letting f(t) be a point  $z \in Z$  such that  $(t, z) \in (\{t\} \times Z) \setminus \varphi(t)$  if such points exist, and an arbitrary point  $z \in Z$  otherwise. Consider the set  $H = \{(t, f(t)): t \in T\}$ , i.e., the graph of the mapping f.

(b) Applying Theorem 1.5.11 and the equality  $\operatorname{ind} R^n = n$  (see Theorem 1.8.2), for every natural number n define a separable metric space X such that  $\operatorname{ind} X = n$  and X can be mapped by a continuous and one-to-one mapping onto a zero-dimensional space, Observe that X is totally disconnected and deduce that there exists a totally disconnected separable metric space which cannot be embedded in a Euclidean space.

*Hint.* Consider the space  $Z = R^n$ , a subspace  $T \subset Z$  homeomorphic to the Cantor set, and the family  $\mathscr{G}$  of all  $G_{\delta}$ -sets in the Cartesian product  $R^n \times R^n$ ; then apply (a).

*Remark.* The first example of a totally disconnected separable metric space of an arbitrary dimension  $n \ge 1$  was given by Mazurkiewicz in [1927]; Mazurkiewicz's spaces are completely metrizable. Clearly, such spaces do not contain any compact subspace of positive dimension.

**1.4.G** (R. Pol [1973]). (a) Prove that every separable metric space X which for each punctiform separable metric space Y contains a subspace homeomorphic to Y contains also a subspace homeomorphic to the Hilbert cube.

*Hint.* One can assume that  $X \subset I^{\aleph_0}$ . Consider the space  $Z = I^{\aleph_0}$ , a subspace  $T \subset Z$  homeomorphic to the Cantor set, and the family  $\mathscr{G}$  consisting of all sets of the form  $f^{-1}(X)$ , where f is a continuous mapping defined on a  $G_{\delta}$ -set in the Cartesian product  $I^{\aleph_0} \times I^{\aleph_0}$  and taking values

in  $I^{\kappa_0}$ ; then apply Problem 1.4.F(a) and the Lavrentieff theorem (see [GT], Theorem 4.3.21).

(b) Define a totally disconnected separable metric space X with the property that every completely metrizable separable space which contains a subspace homeomorphic to X also contains a subspace homeomorphic to the Hilbert cube.

*Hint.* Consider the space  $Z = I^{\aleph_0}$ , a subspace  $T \subset Z$  homeomorphic to the Cantor set and the family  $\mathscr{G}$  of all  $G_{\delta}$ -sets in the Cartesian product  $I^{\aleph_0} \times I^{\aleph_0}$ ; then apply Problem 1.4.F(a) and the Lavrentieff theorem (see [GT], Theorem 4.3.21).

# 1.5. The sum, decomposition, addition, enlargement, separation and Cartesian product theorems

We begin with some observations on the dimension of subspaces. The subspace theorem established in Section 1.1 states that for every subspace M of a regular space X we have  $\operatorname{ind} M \leq \operatorname{ind} X$ . In this context it is natural to ask whether among the subspaces of a space X such that  $\operatorname{ind} X = n$  one can find subspaces of all intermediate dimensions between 0 and n-1. As shown in the next theorem, the answer is positive and there even exist closed subspaces of intermediate dimensions.

**1.5.1. Theorem.** If X is a regular space and  $ind X = n \ge 1$ , then for k = 0, 1, ..., n-1 the space X contains a closed subspace M such that ind M = k.

**Proof.** It is enough to show that X contains a closed subspace M such that  $\operatorname{ind} M = n-1$ . As  $\operatorname{ind} X > n-1$ , there exist a point  $x \in X$  and a neighbourhood  $V \subset X$  of the point x such that for every open set  $U \subset X$  satisfying the condition  $x \in U \subset V$  we have  $\operatorname{ind} \operatorname{Fr} U > n-2$ . On the other hand, as  $\operatorname{ind} X \leq n$ , there exists an open set  $U \subset X$  satisfying the above condition and such that  $\operatorname{ind} \operatorname{Fr} U \leq n-1$ . The closed subspace  $M = \operatorname{Fr} U$  of the space X has the required property.  $\Box$ 

The situation is quite different in spaces of dimension  $\infty$ . In Example 1.8.21 we shall describe, applying the continuum hypothesis, a separable metric space of dimension  $\infty$  whose finite-dimensional subspaces are all countable (it turns out that the existence of such a space is equivalent to the continuum hypothesis). Let us observe that spaces with the above

property are rather peculiar; in particular, no such space is completely metrizable, because every uncountable completely metrizable separable space contains a subspace homeomorphic to the Cantor set (see [GT], Problems 1.7.11 and 4.5.5). There also exist compact metric spaces of dimension  $\infty$  whose finite-dimensional subspaces are all zero-dimensional, but examples of such spaces are very complicated.

We now pass to the sum theorem.

**1.5.2. Lemma.** If a separable metric space X can be represented as the union of two subspaces Y and Z such that  $\operatorname{ind} Y \leq n-1$  and  $\operatorname{ind} Z \leq 0$ , then  $\operatorname{ind} X \leq n$ .

**Proof.** Consider a point  $x \in X$  and a neighbourhood  $V \subset X$  of the point x. By virtue of Theorem 1.2.11, there exist disjoint open sets  $U, W \subset X$ such that  $x \in U, X \setminus V \subset W$  and  $[X \setminus (U \cup W)] \cap Z = \emptyset$ . Clearly,  $x \in U$  $\subset V$ ; as  $\operatorname{Fr} U \subset [X \setminus (U \cup W)] \subset X \setminus Z \subset Y$ , we have  $\operatorname{ind} \operatorname{Fr} U \leq n-1$ . Hence  $\operatorname{ind} X \leq n$ .  $\Box$ 

**1.5.3. The sum theorem.** If a separable metric space X can be represented as the union of a sequence  $F_1, F_2, ...$  of closed subspaces such that  $\operatorname{ind} F_i \leq n$  for i = 1, 2, ..., then  $\operatorname{ind} X \leq n$ .

**Proof.** We shall apply induction with respect to the number *n*. For n = 0 the theorem is already proved. Assume that the theorem holds for dimensions less than *n* and consider a space  $X = \bigcup_{i=1}^{\infty} F_i$ , where  $F_i$  is closed and  $\operatorname{ind} F_i \leq n \geq 1$  for i = 1, 2, ... Applying Theorem 1.1.6, choose for i = 1, 2, ..., a countable base  $\mathscr{B}_i$  for the space  $F_i$  such that  $\operatorname{ind} \operatorname{Fr} U \leq n-1$  for every  $U \in \mathscr{B}_i$ , where Fr denotes the boundary operator in the space  $F_i$ . By the inductive assumption the subspace  $Y = \bigcup \{\operatorname{Fr} U: U \in \bigcup_{i=1}^{\infty} \mathscr{B}_i\}$  of the space X satisfies the inequality  $\operatorname{ind} Y \leq n-1$ . Now, Proposition 1.2.13 implies that for i = 1, 2, ... the subspace  $Z_i = F_i \setminus Y$  of the space  $F_i$  satisfies the inequality  $\operatorname{ind} Z_i \leq 0$ ; hence, by the sum theorem for dimension 0, the subspace  $Z = \bigcup_{i=1}^{\infty} Z_i = X \setminus Y$  of the space X also satisfies the inequality  $\operatorname{ind} Z \leq 0$ , because it follows from the relation  $Z_i = F_i \setminus Y = F_i \cap Z$  that all the  $Z_i$ 's are closed in Z. Thus by virtue of Lemma 1.5.2 we have  $\operatorname{ind} X \leq n$ .

As in the case of zero-dimensional spaces, the sum theorem implies three corollaries:

**1.5.4. Corollary.** If a separable metric space X can be represented as the union of a sequence  $F_1, F_2, ...$  of subspaces such that  $\operatorname{ind} F_i \leq n$  and  $F_i$  is an  $F_{\sigma}$ -set for i = 1, 2, ..., then  $\operatorname{ind} X \leq n$ .  $\Box$ 

**1.5.5. Corollary.** If a separable metric space X can be represented as the union of two subspaces A and B, one of them closed, such that  $\operatorname{ind} A \leq n$  and  $\operatorname{ind} B \leq n$ , then  $\operatorname{ind} X \leq n$ .  $\Box$ 

**1.5.6. Corollary.** If by adjoining a finite numbers of points to a separable metric space X such that  $indX \le n$  one obtains a metric space Y, then the space Y satisfies the inequality  $indY \le n$  and is separable.  $\square$ 

Let us observe that the sum theorem plays a key role in the dimension theory of separable metric spaces. Indeed, all the remaining results in this section follow either from the sum theorem or from one of the decomposition theorems which are easy consequences of the sum theorem.

Applying the sum theorem, one readily shows that the condition in Lemma 1.5.2 characterizes separable metric spaces of dimension  $\leq n$ :

**1.5.7. The first decomposition theorem.** A separable metric space X satisfies the inequality  $\operatorname{ind} X \leq n \geq 0$  if and only if X can be represented as the union of two subspaces Y and Z such that  $\operatorname{ind} Y \leq n-1$  and  $\operatorname{ind} Z \leq 0$ .

**Proof.** Consider a separable metric space X such that  $\operatorname{ind} X \leq n \geq 0$ . By virtue of Theorem 1.1.6, the space X has a countable base  $\mathscr{B}$  such that  $\operatorname{ind} \operatorname{Fr} U \leq n-1$  for every  $U \in \mathscr{B}$ . It follows from the sum theorem that the subspace  $Y = \bigcup \{\operatorname{Fr} U: U \in \mathscr{B}\}$  has dimension  $\leq n-1$  and from Proposition 1.2.13 that the subspace  $Z = X \setminus Y$  has dimension  $\leq 0$ . To complete the proof it suffices to apply Lemma 1.5.2.  $\Box$ 

From the first decomposition theorem we obtain by an easy induction

**1.5.8.** The second decomposition theorem. A separable metric space X satisfies the inequality  $ind X \le n \ge 0$  if and only if X can be represented

[Ch. 1, § 5

as the union of n+1 subspaces  $Z_1, Z_2, ..., Z_{n+1}$  such that  $\operatorname{ind} Z_i \leq 0$  for i = 1, 2, ..., n+1.  $\Box$ 

Let us at once explain that, as will be shown in Section 1.8 (see Theorem 1.8.20), the Hilbert cube cannot be represented as a countable union of zero-dimensional subspaces, or—equivalently—of finite-dimensional subspaces. Hence, the second decomposition theorem does not extend to separable metric spaces of dimension  $\infty$ .

**1.5.9. Examples.** For every point x in the real line R or in the circle  $S^1$  and each neighbourhood V of the point x there exists a neighbourhood U of x such that  $U \subset V$  and the boundary  $\operatorname{Fr} U$  is a two-point set. Hence,  $\operatorname{ind} R \leq 1$  and  $\operatorname{ind} S^1 \leq 1$ . Since  $\operatorname{ind} I > 0$  by virtue of Example 1.2.5, the subspace theorem implies that  $\operatorname{ind} R = \operatorname{ind} S^1 = \operatorname{ind} I = 1$ .

For every point x in Euclidean *n*-space  $\mathbb{R}^n$  or in the *n*-sphere  $\mathbb{S}^n$  and each neighbourhood V of the point x there exists a neighbourhood U of x such that  $U \subset V$  and the boundary  $\operatorname{Fr} U$  is homeomorphic to  $\mathbb{S}^{n-1}$ . Hence, as shown by an inductive argument,  $\operatorname{ind} \mathbb{R}^n \leq n$ ,  $\operatorname{ind} \mathbb{S}^n \leq n$  and  $\operatorname{ind} I^n \leq n$  for every natural number n.

The small inductive dimension of  $\mathbb{R}^n$ ,  $\mathbb{S}^n$  and  $\mathbb{I}^n$  is indeed equal to n, but the proof of this fact is much more difficult than the above evaluations; it will be given in Section 1.8. The equality ind  $\mathbb{R}^n = n$  is of utmost importance for dimension theory. In a sense, it justifies the definition of the dimension function by showing that this definition yields a notion conforming to geometric intuition. The fact that  $\operatorname{ind} \mathbb{R}^n = n$  is sometimes called the fundamental theorem of dimension theory.

The decomposition of  $\mathbb{R}^n$  into n+1 zero-dimensional subspaces following from the second decomposition theorem can be defined—according to Example 1.3.8—by the equality

$$R^n = Q_0^n \cup Q_1^n \cup \ldots \cup Q_n^n.$$

The last formula yields another proof of the inequality ind  $\mathbb{R}^n \leq n$ .

For every pair k, n of integers satisfying  $0 \le k \le n \ge 1$  define

$$N_k^n = Q_0^n \cup Q_1^n \cup \ldots \cup Q_k^n \quad \text{and} \quad L_k^n = Q_k^n \cup Q_{k+1}^n \cup \ldots \cup Q_n^n;$$

thus  $N_k^n$  is the subspace of Euclidean *n*-space  $\mathbb{R}^n$  consisting of all points which have at most k rational coordinates and  $L_k^n$  is the subspace of  $\mathbb{R}^n$  consisting of all points which have at least k rational coordinates. From the second decomposition theorem it follows that  $\operatorname{ind} N_k^n \leq k$  and  $\operatorname{ind} L_k^n \leq n-k$ ; we shall show in Section 1.8, applying the equality  $\operatorname{ind} \mathbb{R}^n = n$ ,

that  $\operatorname{ind} N_k^n = k$  and  $\operatorname{ind} L_k^n = n-k$ . The space  $N_n^{2n+1} \subset \mathbb{R}^{2n+1}$  will play a particularly important role in the sequel: it turns out to be a universal space for the class of all separable metric spaces of dimension  $\leq n$  (see Theorem 1.11.5).  $\Box$ 

We shall now state further consequences of the sum and decomposition theorems. Let us begin with the addition theorem, which follows immediately from the second decomposition theorem.

**1.5.10.** The addition theorem. For every pair X, Y of separable subspaces of a metric space we have

 $\operatorname{ind}(X \cup Y) \leq \operatorname{ind} X + \operatorname{ind} Y + 1. \square$ 

**1.5.11. The enlargement theorem.** For every separable subspace M of an arbitrary metric space X satisfying the inequality ind  $M \leq n$  there exists a  $G_{\delta}$ -set  $M^*$  in X such that  $M \subset M^*$  and ind  $M^* \leq n$ .

**Proof.** By the second decomposition theorem  $M = Z_1 \cup Z_2 \cup \ldots \cup Z_{n+1}$ , where  $\operatorname{ind} Z_i \leq 0$  for  $i = 1, 2, \ldots, n+1$ . Applying Theorem 1.2.14, enlarge each  $Z_i$  to a  $G_{\delta}$ -set  $Z_i^*$  in X such that  $\operatorname{ind} Z_i^* \leq 0$ . The union  $M^* = Z_1^* \cup \cup Z_2^* \cup \ldots \cup Z_{n+1}^*$  has the required properties.  $\Box$ 

**1.5.12. The first separation theorem.** If X is a separable metric space such that  $\operatorname{ind} X \leq n \geq 0$ , then for every pair A, B of disjoint closed subsets of X there exists a partition L between A and B such that  $\operatorname{ind} L \leq n-1$ .

**Proof.** By the first decomposition theorem  $X = Y \cup Z$ , where  $\operatorname{ind} Y \leq n-1$  and  $\operatorname{ind} Z \leq 0$ . Applying Theorem 1.2.11, we obtain a partition L between A and B such that  $L \cap Z = \emptyset$ . As  $L \subset X \setminus Z \subset Y$ , we have  $\operatorname{ind} L \leq n-1$  by the subspace theorem.  $\Box$ 

In a similar way, applying the first decomposition theorem to the subspace M, we obtain

**1.5.13.** The second separation theorem. If X is an arbitrary metric space and M is a separable subspace of X such that  $\operatorname{ind} M \leq n \geq 0$ , then for every pair A, B of disjoint closed subsets of X there exists a partition L between A and B such that  $\operatorname{ind}(L \cap M) \leq n-1$ .  $\Box$  Clearly, the first separation theorem is a special case of the second separation theorem. On the other hand, the second separation theorem easily follows (cf. the proof of Theorem 1.2.11) from the first separation theorem and Lemma 1.2.9, which is an elementary topological fact; hence, both separation theorems are in a sense equivalent.

The second separation theorem yields a characterization of the dimension of subspaces in terms of neighbourhoods in the whole space, which generalizes Proposition 1.2.12:

**1.5.14. Proposition.** A separable subspace M of an arbitrary metric space X satisfies the inequality  $\operatorname{ind} M \leq n \geq 0$  if and only if for every point  $x \in M$  (or—equivalently—for every point  $x \in X$ ) and each neighbourhood V of the point x in the space X there exists an open set  $U \subset X$  such that  $x \in U \subset V$  and  $\operatorname{ind}(M \cap \operatorname{Fr} U) \leq n-1$ .  $\Box$ 

Proposition 1.5.14 and Lemma 1.1.5 imply the following generalization of Proposition 1.2.13.

**1.5.15. Proposition.** A subspace M of a separable metric space X satisfies the inequality  $\operatorname{ind} M \leq n \geq 0$  if and only if X has a countable base  $\mathscr{B}$  such that  $\operatorname{ind}(M \cap \operatorname{Fr} U) \leq n-1$  for every  $U \in \mathscr{B}$ .  $\Box$ 

The general Cartesian product theorem reads as follows:

**1.5.16.** The Cartesian product theorem. For every pair X, Y of separable metric spaces of which at least one is non-empty we have

$$\operatorname{ind}(X \times Y) \leq \operatorname{ind} X + \operatorname{ind} Y.$$

**Proof.** The theorem is obvious if one of the spaces has dimension  $\infty$ , and so we can suppose that  $k(X, Y) = \operatorname{ind} X + \operatorname{ind} Y$  is finite. We shall apply induction with respect to that number. If k(X, Y) = -1, then either  $X = \emptyset$  or  $Y = \emptyset$ , and our inequality holds. Assume that the inequality is proved for every pair of separable metric spaces the sum of the dimensions of which is less than  $k \ge 0$  and consider separable metric spaces X and Y such that  $\operatorname{ind} X = n \ge 0$ ,  $\operatorname{ind} Y = m \ge 0$  and n+m = k. Let (x, y) be a point of  $X \times Y$  and  $W \subset X \times Y$  a neighbourhood of (x, y). There exist neighbourhoods  $U', U \subset X$  of the point x and  $V, V' \subset Y$  of the point y such that  $U' \times V' \subset W, U \subset U', V \subset V'$ ,  $\operatorname{ind} \operatorname{Fr} U \le n-1$  and  $\operatorname{ind} \operatorname{Fr} V \le m-1$ . Since

$$\operatorname{Fr}(U \times V) \subset (X \times \operatorname{Fr} V) \cup (\operatorname{Fr} U \times Y),$$

by virtue of the inductive assumption and the sum theorem we have  $\operatorname{ind} \operatorname{Fr}(U \times V) \leq k-1$ . Hence  $\operatorname{ind}(X \times Y) \leq k$  and the proof is completed.  $\Box$ 

The inequality in the Cartesian product theorem cannot be replaced by an equality. In the next example we shall describe a separable metric space X such that indX = 1, and yet  $ind(X \times X) = 1$ , because X is homeomorphic to the square  $X \times X$  (cf. the remark to Problem 1.5.C). There exist even compact metrizable spaces the dimension of the Cartesian product of which is less than the sum of their dimensions, but they are more complicated, and in checking their properties one has to apply the methods of algebraic topology; let us note that such spaces are necessarily of dimension  $\ge 2$ , because for every pair X, Y of compact metrizable spaces X, Y such that indY = 1 we have  $ind(X \times Y) = indX + 1 = indX + indY$ (see Problem 1.9.E(b)).

**1.5.17. Example.** We shall show that the space  $H_0$  defined in Example 1.2.15 has the required properties.

To establish the equality  $\operatorname{ind} H_0 = 1$  it is enough to prove that  $\operatorname{ind} H_0 \leq 1$ . Since every point  $x \in H_0$  can be transformed by a suitable translation to the point  $x_0 \in H_0$ , it suffices to show that for every natural number n the boundary  $F_n = \{x \in H_0: ||x|| = 1/n\}$  of the 1/n-ball  $U_n = \{x \in H_0: ||x|| < 1/n\}$  about  $x_0$  is zero-dimensional. This, however, is a consequence of the final paragraph of Example 1.3.8 and the fact that the mapping  $h: F_n \to h(F_n) \subset Q^{\aleph_0}$  defined by letting  $h(\{x_i\}) = \{x_i\}$  is a homeomorphism, which, in its turn, is implied by the fact that both in  $Q^{\aleph_0}$  and in  $F_n$  a sequence  $x_1^1 = \{x_i^1\}, x_i^2 = \{x_i^2\}, \dots$  converges to  $x = \{x_i\}$  if and only if the sequence  $x_i^1, x_i^2, \dots$  converges to  $x_i$  for  $i = 1, 2, \dots$  The last equivalence is well known to hold in the Cartesian product  $Q^{\aleph_0}$ ; it does not generally hold in Hilbert space H, but it does hold in the subspace  $F_n$ , because all points of  $F_n$  have the same norm (cf. the hint to Problem 1.4.B(a)).

Now, to show that  $H_0$  is homeomorphic to the square  $H_0 \times H_0$  it suffices to note that by assigning to a point  $(x, y) = (\{x_i\}, \{y_i\}) \in H_0 \times H_0$  the point  $(x_1, y_1, x_2, y_2, ...) \in H_0$  one defines a homeomorphism of  $H_0 \times H_0$ onto  $H_0$ .

To conclude, let us observe that the Cartesian products  $H_0^2$ ,  $H_0^3$ , ... are all homeomorphic to  $H_0$ , and thus are one-dimensional spaces. Since the countable Cartesian product  $X^{\aleph_0}$  is homeomorphic to the limit of an inverse sequence consisting of finite Cartesian products  $X^i$ , and since the limit of an inverse sequence consisting of separable metric spaces of dimension  $\leq n$  is itself of dimension  $\leq n$  (see Theorem 1.13.4), the countable Cartesian product  $H_{0}^{s_0}$  is also one-dimensional. Let us add that every infinite Cartesian product of compact metric spaces of finite positive dimension is infinite-dimensional. Indeed, every compact metric space of finite dimension > 0 contains a compact subspace of dimension one, and—as noted above—multiplying by such spaces raises the dimension by one.  $\Box$ 

## Historical and bibliographic notes

An example of a compact metric space of dimension  $\infty$  whose finitedimensional subspaces are all zero-dimensional was announced by Walsh in [1978]. In [1965] Henderson constructed a compact metric space of dimension  $\infty$  whose finite-dimensional closed subspaces are all zerodimensional (simpler, but still very difficult examples of such spaces were given by Henderson in [1967] and by Zarelua in [1972]). Theorem 1.5.3 was proved for compact metric spaces by Menger in [1924] and by Urysohn in [1926] (announcement in [1922]) and was extended to separable metric spaces by Tumarkin in [1926] (announcement in [1925]) and by Hurewicz [1927] (the latter gave the simple proof reproduced here). Theorems 1.5.7, 1.5.8 and 1.5.10 were established for compact metric spaces by Urysohn in [1926] (announcement in [1922]) and were extended to separable metric spaces by Tumarkin and Hurewicz in the above quoted papers. Theorem 1.5.11 was proved by Tumarkin in [1926] (announcement in [1925]). As the reader will see in the next section, Theorem 1.5.12 states, in substance, that for every separable metric space X we have the equality ind X = Ind X. For compact metric spaces this equality was announced by Brouwer in [1924] while he was discussing relationships between his Dimensionsgrad and the small inductive dimension ind (Brouwer commented that the equality was also known to Urysohn); the proof was given by Menger in [1924] and by Urysohn in [1926]. For separable metric spaces, the equality of ind and Ind was established by Tumarkin in [1926] (announcement in [1925]) and by Hurewicz in [1927]. Theorem 1.5.13 was proved by Menger in [1924] for compact metric spaces, and extended by Hurewicz in [1927] to separable metric spaces. Theorem 1.5.16

#### Problems

was obtained by Menger in [1928]. Example 1.5.17 was described by Erdös in [1940]; let us add that Anderson and Keisler described in [1967] a space  $K(n) \subset \mathbb{R}^n$ , where n = 2, 3, ..., such that  $\operatorname{ind} K(n) = \operatorname{ind} [K(n)]^{\aleph_0} = n-1$ .

The first example of two-dimensional compact metric spaces X and Y such that the Cartesian product  $X \times Y$  is three-dimensional was given by Pontrjagin in [1930]; examples of such spaces can be found in Kodama [1970]. No "geometric" characterization of compact metric spaces satisfying the equality  $\operatorname{ind}(X \times Y) = \operatorname{ind} X + \operatorname{ind} Y$  is known; in particular, it is an open question if this equality holds for all absolute neighbourhood retracts (as shown by Borsuk in [1936], it holds if X and Y are absolute neighbourhood retracts satisfying condition ( $\Delta$ )). The question is connected with the problem of delineating the class of spaces in which the small inductive dimension ind coincides with the cohomological dimension dim<sub>Zp</sub> with respect to the group  $Z_p$  of integers modulo p (cf. the final part of Section 1.9), because for every pair X, Y of locally compact spaces and every prime number p we have dim<sub>Zp</sub>(X × Y) = dim<sub>Zp</sub>X + dim<sub>Zp</sub>Y.

## Problems

**1.5.A** (de Groot and Nagata [1969]; announcement Hurewicz [1928]). Prove that if a completely metrizable separable space X of dimension  $\infty$  can be represented as the union of countably many finite-dimensional subspaces, then for n = 0, 1, 2, ... the space X contains a closed subspace M such that  $\operatorname{ind} M = n$ .

ł

*Hint.* Let  $X = \bigcup_{i=1}^{\infty} Z_i$ , where  $\operatorname{ind} Z_i = 0$ . Assuming that X does not contain any closed subspace of dimension n and applying Theorem 1.5.1, define a point whose all sufficiently small neighbourhoods have infinite-dimensional boundaries; consider such a boundary  $F_1$  satisfying  $\delta(F_1) < 1$  and  $F_1 \cap Z_1 = \emptyset$ . Iterating this procedure obtain a contradiction to the Cantor theorem.

**1.5.B.** (a) Observe that if a separable metric space X can be represented as the union of a family  $\{F_s\}_{s\in S}$  of closed subspaces such that every point  $x \in X$  has a neighbourhood which meets at most countably many sets  $F_s$  and  $\inf F_s \leq n$  for every  $s \in S$ , then  $\inf X \leq n$ .

(b) Prove that if a separable metric space X can be represented as the union of a sequence  $F_0, F_1, F_2, ...$  of closed subspaces such that  $\operatorname{ind} F_i \leq n$  for i = 1, 2, ..., then  $\operatorname{ind}_x X \leq n$  for every point  $x \in F_0$  such that  $\operatorname{ind}_x F_0 \leq n$ .

**1.5.C.** Let X be a separable metric space such that  $ind X = n \ge 1$ ; the set  $\{x \in X : ind_x X = n\}$  is called the *dimensional kernel* of the space X.

(a) (Menger [1924], Urysohn [1926]) Check that the dimensional kernel is an  $F_{q}$ -set.

(b) (Menger [1926]) Show that the dimensional kernel of a separable metric space X such that  $ind X = n \ge 1$  has dimension  $\ge n-1$ .

*Hint*. Represent the complement of the kernel as the union of two subspaces Y and Z such that  $\operatorname{ind} Y \leq n-2$ ,  $\operatorname{ind} Z \leq 0$  and Y is an  $F_{\sigma}$ -set in X.

(c) (Menger [1926]) Prove that the dimensional kernel of a compact metric space X such that  $ind X = n \ge 1$  has dimension n at each point (cf. Theorem 1.9.8).

Hint. Suppose that for a point x of the dimensional kernel M the inequality  $\operatorname{ind}_x M \leq n-1$  holds, and for every positive number  $\varepsilon$  define a neighbourhood U of the point x in the space X such that  $\delta(U) < \varepsilon$  and ind  $\operatorname{Fr} U \leq n-2$ . To that end take a neighbourhood  $U_0$  of the point x in the space X such that  $\delta(U_0) < \varepsilon/2$  and  $\operatorname{ind}(M \cap \operatorname{Fr} U_0) \leq n-2$ . Then enlarge  $M \cap \operatorname{Fr} U_0$  to an (n-2)-dimensional  $G_{\delta}$ -set  $M^*$  in X. Let  $U = U_0$ , if  $\operatorname{Fr} U_0$  $\subset M^*$ , and if  $\operatorname{Fr} U_0 \setminus M^* \neq \emptyset$ , define a countable family  $\{U_i\}_{i=1}^{\infty}$  of open subsets of X such that

 $\delta(U_i) < \varepsilon/2, \quad \text{ind} \operatorname{Fr} U_i \leq n-2, \quad U_i \cap (\operatorname{Fr} U_0 \setminus M^*) \neq \emptyset$  for i = 1, 2, ..., and

$$\operatorname{Fr} U_0 \setminus M^* \subset \bigcup_{i=1}^{\infty} U_i \subset \bigcup_{i=1}^{\infty} U_i \subset \operatorname{Fr} U_0 \cup \bigcup_{i=1}^{\infty} \overline{U}_i,$$
  
and let  $U = \bigcup_{i=1}^{\infty} U_i$ ; note that  $\operatorname{Fr} U \subset (\operatorname{Fr} U_0 \setminus \bigcup_{i=1}^{\infty} U_i) \cup \bigcup_{i=1}^{\infty} \operatorname{Fr} U_i.$ 

Remark. A separable metric space X such that  $ind X = n \ge 1$  and the dimensional kernel of X has dimension n-1 is called a *weakly n-dimensional* space. Clearly, a weakly *n*-dimensional space contains no compact subspace of dimension *n*. The space K described in Problem 1.2.E is weakly one-dimensional; the first example of such a space was given by Sierpiński in [1921]. First examples of weakly *n*-dimensional spaces for n = 2, 3, ... were given by Mazurkiewicz in [1929]; Mazurkiewicz's spaces are completely metrizable. A simpler construction of weakly *n*-dimensional spaces

for n = 1, 2, ... was described by Tomaszewski in [1979], where it is also shown that if X is a weakly *n*-dimensional space and Y is a weakly *m*-dimensional space, then  $ind(X \times Y) \le n+m-1 = indX+indY-1$ .

**1.5.D.** Prove that a subspace M of a metric space X satisfies the inequality ind  $M \le n \ge 0$  if and only if for every point  $x \in M$  and each neighbourhood V of the point x in the space X there exists an open set  $U \subset X$  such that  $x \in U \subset V$  and  $\operatorname{ind}(M \cap \operatorname{Fr} U) \le n-1$  (cf. Proposition 1.5.14 and Problem 4.1.C).

Hint. Apply Lemma 1.2.9.

**1.5.E** (Menger [1928]). (a) Show that if  $\operatorname{ind}_x X \leq 0$  and  $\operatorname{ind}_y Y \leq 0$ , then  $\operatorname{ind}_{(x,y)}(X \times Y) \leq 0$ .

(b) Applying the equality  $\operatorname{ind} I^2 = 2$ , give an example of two subspaces X and Y of the real line such that for some points  $x \in X$  and  $y \in Y$  we have  $\operatorname{ind}_x X = 0$  and  $\operatorname{ind}_y Y = 1$ , and yet  $\operatorname{ind}_{(x,y)}(X \times Y) = 2$ .

**1.5.F.** Give an example of a completely metrizable separable space X such that indX = 1 and X is homeomorphic to the square  $X \times X$ .

# 1.6. Definitions of the large inductive dimension and the covering dimension. Metric dimension

The first separation theorem, established in the preceding section, suggests a modification in the definition of the small inductive dimension consisting in replacing the point x by a closed set A. In this way we are led to the notion of the large inductive dimension Ind, defined for all normal spaces. Both dimensions coincide in the realm of separable metric spaces. They diverge, however, in the class of metric spaces and also in the class of compact spaces; let us make clear at once that examples in point are very difficult and will not be discussed in this book. The theory of the dimension function Ind will be developed in Chapter 2, and in Chapter 4 it will be shown that in the realm of all metric spaces this theory is quite similar to the theory of the dimension function Ind, just as the covering dimension dim discussed later in this section, will play an auxiliary role: introducing these dimension functions leads to a simplification of the theory.

We pass to the formal definition of the dimension Ind.

**1.6.1. Definition.** To every normal space X one assigns the *large inductive* dimension of X, denoted by IndX, which is an integer larger than or equal to -1 or the "infinite number  $\infty$ "; the definition of the dimension function Ind consists in the following conditions:

(BČ1) IndX = -1 if and only if X = Ø;
(BČ2) IndX ≤ n, where n = 0, 1, ..., if for every closed set A ⊂ X and each open set V ⊂ X which contains the set A there exists an open set U ⊂ X such that

 $A \subset U \subset V$  and  $\operatorname{Ind} \operatorname{Fr} U \leq n-1$ ;

(BČ3)  $\operatorname{Ind} X = n$  if  $\operatorname{Ind} X \leq n$  and  $\operatorname{Ind} X > n-1$ ; (BČ4)  $\operatorname{Ind} X = \infty$  if  $\operatorname{Ind} X > n$  for n = -1, 0, 1, ...

The large inductive dimension Ind is also called the *Brouwer-Čech* dimension.

Applying induction with respect to IndX, one can easily verify that whenever normal spaces X and Y are homeomorphic, then IndX = IndY, i.e., the large inductive dimension is a topological invariant.

Modifying slightly the proof of Proposition 1.1.4, one obtains

**1.6.2. Proposition.** A normal space X satisfies the inequality  $\operatorname{Ind} X \leq n \geq 0$  if and only if for every pair A, B of disjoint closed subsets of X there exists a partition L between A and B such that  $\operatorname{Ind} L \leq n-1$ .  $\Box$ 

Applying induction with respect to IndX, one can easily prove the following theorem, which justifies the names of the small and the large inductive dimensions.

**1.6.3. Theorem.** For every normal space X we have  $indX \leq IndX$ .  $\Box$ 

Both dimensions coincide in the realm of separable metric spaces.

**1.6.4.** Theorem. For every separable metric space X we have ind X = Ind X.

**Proof.** If suffices to show that  $\operatorname{Ind} X \leq \operatorname{ind} X$ ; clearly, one can suppose that  $\operatorname{ind} X < \infty$ . We shall apply induction with respect to  $\operatorname{ind} X$ . The inequality holds if  $\operatorname{ind} X = 1$ . Assume that the inequality is proved for all

52

separable metric spaces of small inductive dimension less than  $n \ge 0$  and consider a separable metric space X such that  $\operatorname{ind} X = n$ . Let A, B be a pair of disjoint closed subsets of X. By virtue of the first separation theorem, there exists a partition L between A and B such that  $\operatorname{ind} L \le n-1$ . It follows from the inductive assumption that  $\operatorname{Ind} L \le n-1$ , so that  $\operatorname{Ind} X \le n$  by Proposition 1.6.2. Hence  $\operatorname{Ind} X \le \operatorname{ind} X$  and the proof is completed.  $\Box$ 

Using the dimension function Ind one can reformulate Theorem 1.2.6 in the following form (cf. Remark 1.2.7):

**1.6.5. Theorem.** For every Lindelöf space X the conditions ind X = 0 and Ind X = 0 are equivalent.  $\Box$ 

Besides the inductive dimensions ind and Ind, in dimension theory one studies another dimension function, namely the covering dimension dim defined for all normal spaces. In the following section we shall prove that the dimensions ind and dim coincide in the realm of separable metric spaces. Later on the reader will see that they diverge in the class of metric spaces and also in the class of compact spaces (see Remark 4.1.6 and Example 3.1.31). On the other hand, the dimensions Ind and dim coincide in the realm of all metric spaces (see Theorem 4.1.3) and diverge in the class of compact spaces (see Example 3.1.31). The reason why we introduce the covering dimension now is that this notion comes out in a natural way in proofs of the compactification, embedding and universal space theorems.

Let us sum up. There are three ways of defining the dimension of separable metric spaces. They are all equivalent and equally natural, but they are based on different geometric properties of spaces. Outside the class of separable metric spaces the dimensions ind, Ind, and dim diverge and three different dimension theories arise, all poorer than the dimension theory of separable metric spaces. The dimensions Ind and dim lead to much more interesting results than the dimension ind; as a matter of fact, the latter is practically of no importance outside the class of separable metric spaces. Finally, in the dimension theory of separable metric spaces some theorems depend—roughly speaking—on the dimension ind, and other ones depend on the dimension dim; so far we have discussed theorems of the first group, in the subsequent sections we shall discuss those of the second group. In the definition of the covering dimension the notion of the order of a family of sets will be applied.

**1.6.6. Definition.** Let X be a set and  $\mathscr{A}$  a family of subsets of X. By the order of the family  $\mathscr{A}$  we mean the largest integer n such that the family  $\mathscr{A}$  contains n+1 sets with a non-empty intersection; if no such integer exists, we say that the family  $\mathscr{A}$  has order  $\infty$ . The order of a family  $\mathscr{A}$  is denoted by ord  $\mathscr{A}$ .

Thus, if the order of a family  $\mathscr{A} = \{A_s\}_{s\in S}$  equals *n*, then for each n+2 distinct indexes  $s_1, s_2, \ldots, s_{n+2} \in S$  we have  $A_{s_1} \cap A_{s_2} \cap \ldots \cap A_{s_{n+2}} = \emptyset$ . In particular, a family of order -1 consists of the empty set alone, and a family of order 0 consists of pairwise disjoint sets which are not all empty.

Let us recall that a cover  $\mathscr{B}$  is a *refinement* of another cover  $\mathscr{A}$  of the same space, in other words  $\mathscr{B}$  refines  $\mathscr{A}$ , if for every  $B \in \mathscr{B}$  there exists an  $A \in \mathscr{A}$  such that  $B \subset A$ . Clearly, every subcover  $\mathscr{A}_0$  of  $\mathscr{A}$  is a refinement of  $\mathscr{A}$ .

**1.6.7. Definition.** To every normal space X one assigns the covering dimension of X, denoted by dimX, which is an integer larger that or equal to -1 or the "infinite number  $\infty$ "; the definition of the dimension function dim consists in the following conditions:

- (ČL1) dim  $X \le n$ , where n = -1, 0, 1, ..., if every finite open cover of the space X has a finite open refinement of order  $\le n$ ;
- (ČL2) dim X = n if dim  $X \le n$  and dim X > n-1;
- (ČL3) dim $X = \infty$  if dimX > n for n = -1, 0, 1, ...

The covering dimension dim is also called the *Čech-Lebesgue dimension*.

One readily sees that whenever normal spaces X and Y are homeomorphic, then dim  $X = \dim Y$ , i.e., the covering dimension is a topological invariant. Clearly, dim X = -1 if and only if  $X = \emptyset$ .

The next proposition contains two useful characterizations of the covering dimension; in the second one the notion of a shrinking is used.

**1.6.8. Definition.** By a *shrinking* of the cover  $\{A_s\}_{s\in S}$  of a topological space X we mean any cover  $\{B_s\}_{s\in S}$  of the space X such that  $B_s \subset A_s$  for every  $s \in S$ . A shrinking is *open* (closed) if all its members are open (closed) subsets of the space X.

Clearly, every shrinking  $\mathscr{B}$  of a cover  $\mathscr{A}$  is a refinement of  $\mathscr{A}$  and satisfies the inequality ord  $\mathscr{B} \leq \operatorname{ord} \mathscr{A}$ .

**1.6.9.** Proposition. For every normal space X the following conditions are equivalent:

- (a) The space X satisfies the inequality  $\dim X \leq n$ .
- (b) Every finite open cover of the space X has an open refinement of order  $\leq n$ .
- (c) Every finite open cover of the space X has an open shrinking of order  $\leq n$ .

**Proof.** The implications (a)  $\Rightarrow$  (b) and (c)  $\Rightarrow$  (a) are obvious. Consider a normal space X which satisfies (b). Let  $\{U_i\}_{i=1}^k$  be a finite open cover of the space X and  $\mathscr{V}$  an open refinement of this cover such that  $\operatorname{ord} \mathscr{V} \leq n$ . For every  $V \in \mathscr{V}$  choose an  $i(V) \leq k$  such that  $V \subset U_{i(V)}$  and define  $V_i = \bigcup \{V: i(V) = i\}$ . One readily verifies that  $\{V_i\}_{i=1}^k$  is a shrinking of  $\{U_i\}_{i=1}^k$  and has order  $\leq n$ , so that (b)  $\Rightarrow$  (c).  $\Box$ 

We shall now show that when checking the inequality  $\dim X \leq n$  it suffices to consider (n+2)-element covers.

**1.6.10. Theorem.** A normal space X satisfies the inequality dim  $X \le n$  if and only if every (n+2)-element open cover  $\{U_i\}_{i=1}^{n+2}$  of the space X has an open shrinking  $\{W_i\}_{i=1}^{n+2}$  of order  $\le n$ , i.e., such that  $\bigcap_{i=1}^{n+2} W_i = \emptyset$ .

**Proof.** It suffices to show that every normal space X such that dim X > n has an (n+2)-element open cover  $\{U_i\}_{i=1}^{n+2}$  with the property that each open shrinking  $\{W_i\}_{i=1}^{n+2}$  of  $\{U_i\}_{i=1}^{n+2}$  satisfies the condition  $\bigcap_{i=1}^{n+2} W_i \neq \emptyset$ .

Since dim X > n, by virtue of Proposition 1.6.9 there exists an open cover  $\mathscr{V} = \{V_i\}_{i=1}^k$  of the space X which has no open shrinking of order  $\leq n$ . Moreover, one can assume—replacing, if necessary,  $\mathscr{V}$  by a suitable shrinking—that if  $\mathscr{V}' = \{V'_i\}_{i=1}^k$  is an open shrinking of  $\{V_i\}_{i=1}^k$ , then

(1) 
$$V'_{i_1} \cap V'_{i_2} \cap \ldots \cap V'_{i_m} \neq \emptyset$$
 whenever  $V_{i_1} \cap V_{i_2} \cap \ldots \cap V_{i_m} \neq \emptyset$ ,

where  $i_1, i_2, ..., i_m$  is a sequence of natural numbers less than or equal to k. Indeed, if  $\mathscr{V}$  has an open shrinking  $\mathscr{V}'$  which does not satisfy (1), one replaces  $\mathscr{V}$  by  $\mathscr{V}'$  and one continues this procedure until an open

cover with the required property is obtained; as the number of intersections in (1) is finite, the process will come to an end after finitely many steps. Since ord  $\mathcal{V} > n$ , rearranging if need be the members of  $\mathcal{V}$ , we have

(2) 
$$\bigcap_{i=1}^{n+2} V_i \neq \emptyset.$$

We shall show that the (n+2)-element open cover  $\{U_i\}_{i=1}^{n+2}$  of the space X, where  $U_i = V_i$  for  $i \le n+1$  and  $U_{n+2} = \bigcup_{\substack{i=n+2\\i=n}}^{k} V_i$ , has the required property. Consider an open shrinking  $\{W_i\}_{i=1}^{n+2}$  of  $\{U_i\}_{i=1}^{n+2}$ . The cover

 $\{W_1, W_2, \ldots, W_{n+1}, W_{n+2} \cap V_{n+2}, W_{n+2} \cap V_{n+3}, \ldots, W_{n+2} \cap V_k\}$ 

of the space X is an open shrinking of  $\mathscr{V}$ , so that by (1) and (2) we have

$$\bigcap_{i=1}^{n+2} W_i \supset \left(\bigcap_{i=1}^{n+1} W_i\right) \cap (W_{n+2} \cap V_{n+2}) \neq \emptyset. \square$$

Let us note that the last theorem immediately yields

**1.6.11. Theorem.** For every normal space X the conditions Ind X = 0 and dim X = 0 are equivalent.

In the realm of compact metric spaces, the covering dimension can be characterized in terms of a metric, viz., by the condition that the space has finite covers of order  $\leq n$  by open sets of arbitrarily small diameter. Let us recall that the *mesh* of a family  $\mathscr{A}$  of subsets of a metric space X, denoted by mesh $\mathscr{A}$ , is defined as the least upper bound of the diameters of all members of  $\mathscr{A}$ , i.e.,

$$\operatorname{mesh} \mathscr{A} = \sup \{ \delta(A) \colon A \in \mathscr{A} \};$$

the mesh is either a non-negative real number or the "infinite number"  $\infty$ .

**1.6.12. Theorem.** For every compact metric space X the following conditions are equivalent:

- (a) The space X satisfies the inequality  $\dim X \leq n$ .
- (b) For every metric  $\varrho$  on the space X and for every positive number  $\varepsilon$  there exists a finite open cover  $\mathcal{U}$  of the space X such that mesh $\mathcal{U} < \varepsilon$  and ord $\mathcal{U} \leq n$ .
- (c) There exists a metric  $\varrho$  on the space X with the property that for every positive number  $\varepsilon$  there exists a finite open cover  $\mathcal{U}$  of the space X such that mesh $\mathcal{U} < \varepsilon$  and  $\operatorname{ord} \mathcal{U} \leq n$ .

Ch. 1, § 6]

**Proof.** Let X be a compact metric space satisfying  $\operatorname{ind} X \leq n$ ; consider a metric  $\varrho$  on the space X and a positive number  $\varepsilon$ . By the compactness of X, the open cover  $\{B(x, \varepsilon/3)\}_{x\in X}$  has a finite open subcover; by applying (ČL1) to this subcover we obtain a finite open cover  $\mathscr{U}$  such that  $\operatorname{mesh} \mathscr{U}$  $< \varepsilon$  and  $\operatorname{ord} \mathscr{U} \leq n$ . Hence (a)  $\Rightarrow$  (b).

The implication (b)  $\Rightarrow$  (c) being obvious, to conclude the proof it suffices to show that (c)  $\Rightarrow$  (a). Let  $\varrho$  be a metric on the space X which has the property stated in (c). Consider a finite open cover  $\{U_i\}_{i=1}^k$  of the space X and denote by  $\varepsilon$  a Lebesgue number for the cover  $\{U_i\}_{i=1}^k$ , i.e., a positive number such that every subset of X which has diameter less than  $\varepsilon$  is contained in one of the sets  $U_i$ . The cover  $\mathscr{U}$  in condition (c) is a refinement of  $\{U_i\}_{i=1}^k$ , so that dim $X \leq n$ .  $\Box$ 

The attempts to extend the last theorem to separable metric spaces led to the notion of the metric dimension, with a discussion of which we shall conclude this section. Before that, let us briefly comment upon conditions (b) and (c) in separable metric spaces. To begin with, observe that if for a metric  $\rho$  on a space X and for every positive number  $\varepsilon$  there exists a finite cover of the space X with mesh less than  $\varepsilon$ , then the metric  $\rho$  is totally bounded. Hence, when passing to separable metric spaces, we have to replace condition (b) by the following condition:

(b') For every totally bounded metric  $\varrho$  on the space X and for every positive number  $\varepsilon$  there exists a finite open cover  $\mathcal{U}$  of the space X such that mesh $\mathcal{U} < \varepsilon$  and ord $\mathcal{U} \leq n$ .

Now, one proves that conditions (a) and (b') are equivalent for every separable metric space X (see Problem 1.6.B), whereas conditions (a) and (c) are generally not equivalent for such spaces (see Example 1.10.23).

To every metric space  $(X, \varrho)$  one assigns the *metric dimension* of  $(X, \varrho)$ , denoted by  $\mu \dim(X, \varrho)$ , which is an integer larger than or equal to -1or the "infinite number"  $\infty$ ; the definition follows the pattern of the definition of the covering dimension dim except that condition (ČL1) is replaced by the condition that for every positive number  $\varepsilon$  there exists an open cover  $\mathscr{U}$  of the space X such that mesh $\mathscr{U} < \varepsilon$  and ord $\mathscr{U} \leq n$ . Clearly, if  $(X, \varrho)$  is a compact space, then  $\mu \dim(X, \varrho) = \dim X$ .

From the discussion in the penultimate paragraph it follows that in the realm of separable metric spaces the metric dimension is not a topological invariant; in general, the number  $\mu \dim(X, \varrho)$  depends upon the metric  $\varrho$ on the space X. Nevertheless, a theory of the metric dimension  $\mu \dim$  can be developed, which shows a resemblance to the theory of the covering dimension dim. Let us observe that—as the reader can readily verify—if the metrics  $\rho$  and  $\sigma$  on a space X are uniformly equivalent, i.e., the identity mapping of  $(X, \rho)$  to  $(X, \sigma)$  and also the identity mapping of  $(X, \sigma)$  to  $(X, \rho)$  are uniformly continuous, then  $\mu \dim(X, \rho) = \mu \dim(X, \sigma)$ ; this means that the metric dimension is a uniform invariant.

Let us also note that in the definition of the metric dimension one has to consider an arbitrary open cover  $\mathscr{U}$ , because the restriction to finite open covers would imply the restriction of the definition to totally bounded spaces. However, in the case where  $(X, \varrho)$  is a totally bounded metric space, the restriction to finite open covers yields an equivalent definition (see Problem 1.6.C).

From Problem 1.7.E below it follows that for every separable metric space  $(X, \varrho)$  we have  $\mu \dim(X, \varrho) \leq \dim X$ ; by virtue of an important characterization of the covering dimension, to be established in Chapter 3 (see Theorem 3.2.1), this inequality extends to all metric spaces. On the other hand, for every metric space  $(X, \varrho)$  we have  $\dim X \leq 2\mu \dim(X, \varrho)$ ; a proof is sketched out in the hint to Problem 1.6.D.

We shall return briefly to the metric dimension in Section 1.10, where a characterization of this dimension function in terms of mappings to polyhedra will be given (see Problem 1.10.L).

## Historical and bibliographic notes

As we have already observed in the notes to Section 1.1, the notion of the large inductive dimension Ind is related to Brouwer's notion of *Dimensionsgrad*. A formal definition of the dimension function Ind in the class of normal spaces was first given by Čech in [1931], which was a short announcement of results in his paper [1932] devoted to a study of the large inductive dimension. Theorem 1.6.4 is a restatement of Theorem 1.5.12, its history is described in the notes to Section 1.5. The covering dimension dim was formally introduced and discussed in Čech's paper [1933]; it is related to a property of covers of the *n*-cube  $I^n$  discovered by Lebesgue in [1911] (see the notes to Section 1.1). Theorem 1.6.10 was proved by Hemmingsen in [1946], and Theorem 1.6.11 by Vedenissoff in [1939].

The notion of the metric dimension was introduced by Alexandroff around 1930. As a definition he used the characterization given here in

#### Problems

Problem 1.10.L, which is connected with his famous theorem on  $\varepsilon$ -translations to polyhedra (see Theorem 1.10.19). Alexandroff's question whether the metric dimension coincides with the covering dimension in the realm of separable metric spaces was solved in the negative by Sitnikov in [1953] (see Example 1.10.23). This last paper called the topologists' attention back to the notion of the metric dimension. The basic properties of the dimension  $\mu$ dim were established by Smirnov in [1956] and by Egorov in [1959]. Besides the notion of the metric dimension discussed in this section, which is a natural geometric notion with a sound intuitive background, a few other metric dimension functions have recently been studied; they are all obtained by replacing topological conditions by the corresponding metrical ones in various characterizations of the dimension dim. A discussion of this topic can be found in Nagami's book [1970].

## **Problems**

1.6.A. Give a direct proof of Theorem 1.6.11.

**1.6.B** (Hurewicz [1930]). Prove that a separable metric space X satisfies the inequality dim  $X \le n$  if and only if for every totally bounded metric  $\varrho$  on the space X and for every positive number  $\varepsilon$  there exists a finite open cover  $\mathscr{U}$  of the space X such that mesh $\mathscr{U} \stackrel{*}{\le} \varepsilon$  and  $\operatorname{ord} \mathscr{U} \le n$ .

Hint. For a finite open cover  $\{U_i\}_{i=1}^k$  of the space X define a metric  $\varrho$ on the space X with the property that every subset of X which has diameter less than 1 is contained in one of the sets  $U_i$ . To that end define continuous functions  $f_1, f_2, \ldots, f_k$  from X to I such that  $f_i(X \setminus U_i) \subset \{0\}$  for  $i = 1, 2, \ldots$  $\ldots, k$  and  $\sum_{i=1}^k f_i^{-1}(1) = X$ ; observe that by adding to the original distance of x and y the sum  $\sum_{i=1}^k |f_i(x) - f_i(y)|$  one obtains a metric on the space X. **1.6.C** (Egorov [1959]). Prove that a totally bounded metric space  $(X, \varrho)$ 

satisfies the inequality  $\mu \dim(X, \varrho) \leq n$  if and only if for every positive number  $\varepsilon$  there exists a finite open cover  $\mathscr{U}$  of the space X such that mesh $\mathscr{U}$  $< \varepsilon$  and  $\operatorname{ord} \mathscr{U} \leq n$ .

*Hint*. Consider a finite subset  $\{x_1, x_2, ..., x_k\}$  of the space X with the property that for every point  $x \in X$  there exists an  $i \leq k$  such that  $\varrho(x, x_i) < \varepsilon/4$  and observe that every subset of X which has diameter less than  $\varepsilon/4$  is contained in a member of the cover  $\{B(x_i, \varepsilon/2)\}_{i=1}^k$ .

**1.6.D** (Katětov [1958]). Prove that for every metric space  $(X, \varrho)$  we have  $\dim X \leq 2\mu \dim(X, \varrho)$ .

Hint. For i = 1, 2, ... choose an open cover  $\mathscr{U}_i$  of the space X such that mesh $\mathscr{U}_i < 1/3^i$  and  $\operatorname{ord} \mathscr{U}_i \leq n = \mu \dim(X, \varrho)$ . Consider a finite open cover  $\{U_j\}_{j=1}^k$  of the space X and denote by  $\mathscr{V}_i$  the family of all sets  $U \in \mathscr{U}_i$  such that  $\varrho(U, X \setminus U_j) \ge 1/3^i$  for a certain  $j \leq k$ . Check that the sets  $V_1, V_2, ...$ , where  $V_i = \bigcup \mathscr{V}_i$ , form a cover of the space X and that  $\overline{V}_i \subset V_{i+1}$  for i = 1, 2, ... Consider the sets  $F_i = \overline{V}_{i-1} \cup \overline{C}_i$ , where  $V_0 = \emptyset$  and  $C_i$  consists of all points of X which belong to n+1 members of  $\mathscr{V}_i$ , and the families  $\mathscr{W}_i = \{V \setminus F_{i-1}: V \in \mathscr{V}_i\}$ , where  $F_0 = \emptyset$ ; define  $\mathscr{W} = \bigcup_{i=1}^{\infty} \mathscr{W}_i$  and  $W_i = \bigcup \mathscr{W}_i$ . Show that  $\overline{C}_i \subset V_i$  for i = 1, 2, ... and deduce that  $\mathscr{W}$  is a cover of the space X; observe that it is a refinement of  $\{U_j\}_{j=1}^k$ . Check that  $W_i \cap W_m = \emptyset$  whenever  $m \ge i+2$  and deduce that ord  $\mathscr{W} \le 2n$ .

*Remark.* The evaluation of dimX in Problem 1.6.D cannot be improved (see Example 1.10.23 and Problem 1.10.J).

# 1.7. The compactification and coincidence theorems. Characterization of dimension in terms of partitions

The compactification theorem belongs to the group of theorems depending on the dimension dim, and in its proof covers are used in an essential way; accordingly, we formulate this theorem in terms of the covering dimension dim. The compactification theorem is an important step towards the proof of the coincidence theorem, which states that the dimensions ind, Ind and dim coincide in the realm of separable metric spaces.

We begin with introducing three simple operations on covers.

If  $\mathscr{A}_1, \mathscr{A}_2, ..., \mathscr{A}_k$  are covers of a topological space X, then the family of all intersections  $A_1 \cap A_2 \cap ... \cap A_k$ , where  $A_i \in \mathscr{A}_i$  for i = 1, 2, ..., k, is a cover of the space X. We denote this cover by  $\mathscr{A}_1 \wedge \mathscr{A}_2 \wedge ... \wedge \mathscr{A}_k$ ; obviously, it is a refinement of  $\mathscr{A}_i$  for i = 1, 2, ..., k.

If  $f: X \to Y$  is a continuous mapping of a topological space X to a topological space Y and  $\mathscr{A}$  is a cover of the space Y, then the family of all inverse images  $f^{-1}(A)$ , where  $A \in \mathscr{A}$ , is a cover of the space X. We denote this cover by  $f^{-1}(\mathscr{A})$ .

If M is a subspace of a topological space X and  $\mathscr{A}$  is a cover of the space X, then the family of all intersections  $M \cap A$ , where  $A \in \mathscr{A}$ , is a cover of the subspace M. We denote this cover by  $\mathscr{A}|M$ .

One readily sees that the above operations applied to open (closed) covers yield open (closed) covers, and applied to finite covers yield finite covers.

**1.7.1. Lemma.** Let  $(X, \varrho)$  be a totally bounded metric space such that dim  $X \leq n$ . For every finite sequence  $f_1, f_2, ..., f_k$  of continuous functions from X to I and for every positive number  $\varepsilon$  there exists a finite open cover  $\mathcal{U}$  of the space X such that mesh $\mathcal{U} < \varepsilon$ , ord  $\mathcal{U} \leq n$ , and  $|f_i(x) - f_i(y)| < \varepsilon$  for i = 1, 2, ..., k whenever x and y belong to the same member of  $\mathcal{U}$ .

**Proof.** Let  $\mathscr{V}$  be a finite open cover of the space X such that mesh  $\mathscr{V} < \varepsilon$ and let  $\mathscr{W}$  be a finite open cover of the interval I such that mesh  $\mathscr{W} < \varepsilon$ . The reader can readily check that any finite open refinement  $\mathscr{U}$  of the cover  $\mathscr{V} \wedge f_1^{-1}(\mathscr{W}) \wedge f_2^{-1}(\mathscr{W}) \wedge \ldots \wedge f_k^{-1}(\mathscr{W})$  such that  $\operatorname{ord} \mathscr{U} \leq n$  has the required properties.  $\Box$ 

**1.7.2. The compactification theorem.** For every separable metric space X there exists a dimension preserving compactification, i.e., a compact metric space  $\tilde{X}$  which contains a dense subspace homeomorphic to X and satisfies the inequality dim $\tilde{X} \leq \dim X$ .

More exactly, for every totally bounded metric  $\varrho$  on the space X there exists an equivalent metric  $\tilde{\varrho}$  on X such that  $\varrho(x, y) \leq \tilde{\varrho}(x, y)$  for  $x, y \in X$  and the completion  $\tilde{X}$  of the space X with respect to the metric  $\tilde{\varrho}$  is a compactification of X which satisfies the inequality dim  $\tilde{X} \leq \dim X$ .

**Proof.** We can suppose that  $\dim X = n < \infty$ . For m = 1, 2, ... we shall define a finite open cover  $\mathscr{U}_m = \{U_{m,k}\}_{k=1}^{k_m}$  of the space X such that mesh $\mathscr{U}_m < 1/2^m$ , ord  $\mathscr{U}_m \leq n$ , and

(1)  $|f_{i,j}(x) - f_{i,j}(y)| < 1/2^m$  for i < m and  $j = 1, 2, ..., k_i$ whenever x and y belong to the same member of  $\mathcal{U}_m$ , where

(2) 
$$f_{i,j}(x) = \frac{\varrho(x, X \setminus U_{i,j})}{\sum_{l=1}^{k_i} \varrho(x, X \setminus U_{i,l})} \quad \text{for } x \in X.$$

To obtain  $\mathscr{U}_1$  it suffices to apply the total boundedness of  $\varrho$  and the inequality dim  $X \leq n$ . When the covers  $\mathscr{U}_1, \mathscr{U}_2, \ldots, \mathscr{U}_{m-1}$  are defined, it suffices to apply Lemma 1.7.1 to obtain  $\mathscr{U}_m$ .

[Ch. 1, § 7

Let us now arrange all pairs (i, j), where i = 1, 2, ... and j = 1, 2, ......,  $k_i$ , into a simple infinite sequence and denote by n(i, j) the place of (i, j) in this sequence. Define a new metric  $\tilde{\varrho}$  on the set X by letting

(3) 
$$\tilde{\varrho}(x, y) = \varrho(x, y) + \sum_{n(i,j)=1}^{\infty} \frac{1}{2^{n(i,j)}} |f_{i,j}(x) - f_{i,j}(y)|;$$

one readily verifies that the metric  $\varrho$  and  $\tilde{\varrho}$  are equivalent.

We shall show that the sequence  $\operatorname{mesh} \mathscr{U}_m$ , where  $\operatorname{mesh}$  denotes the mesh with respect to the metric  $\tilde{\varrho}$ , converges to zero. Consider an arbitrary positive number  $\varepsilon$ . Let N be a natural number satisfying the inequality  $1/2^N < \varepsilon/3$ , and M a natural number such that  $M \ge N$ ,

$$\frac{1}{2^M}\sum_{l=1}^N\frac{1}{2^l}<\varepsilon/3,$$

and M > i whenever  $n(i, j) \leq N$ . From (1) and (3) it follows by a simple computation that  $\widetilde{\text{mesh}}\mathcal{U}_m < \varepsilon$  if  $m \geq M$ , i.e.,  $\limsup_{m \to \infty} \mathcal{U}_m = 0$ . In particular, the space  $(X, \tilde{\varrho})$  is totally bounded, so that the completion  $\tilde{X}$  of the space X with respect to the metric  $\tilde{\varrho}$  is a compactification of X.

Since the functions  $f_{m,k}: X \to I$  are uniformly continuous with respect to  $\tilde{\varrho}$ , they can be extended to continuous functions  $\tilde{f}_{m,k}: \tilde{X} \to I$ . From (2) it follows that  $\sum_{k=1}^{k_m} \mathcal{f}_{m,k}(x) = 1$  for every  $x \in X$ . Hence  $\sum_{k=1}^{k_m} \tilde{f}_{m,k}(x) = 1$ for every  $x \in \tilde{X}$ , which implies that the family  $\tilde{\mathcal{U}}_m = \{\tilde{U}_{m,k}\}_{k=1}^{k_m}$ , where  $\tilde{U}_{m,k} = \tilde{f}_{m,k}^{-1}((0, 1])$ , is an open cover of the space  $\tilde{X}$  for m = 1, 2, ... Now, by the density of X in  $\tilde{X}$  and since  $X \cap \tilde{U}_{m,k} = U_{m,k}$  for m = 1, 2, ... and  $k = 1, 2, ..., k_m$ , we have  $\lim \widetilde{\text{mesh}} \tilde{\mathcal{U}}_m = 0$  and  $\operatorname{ord} \tilde{\mathcal{U}}_m \leq n$ , so that  $\dim \tilde{X} \leq n = \dim X$  by virtue of Theorem 1.6.12.  $\Box$ 

A variant of the above proof of the compactification theorem is outlined in Problem 1.7.B.

Let us observe that from the equality  $\operatorname{ind} \tilde{X} = \operatorname{dim} \tilde{X}$ , which is a consequence of Lemmas 1.7.4 and 1.7.6, and from Theorem 1.1.2 and Lemma 1.7.4 it follows that  $\operatorname{dim} \tilde{X} = \operatorname{dim} X$  in the compactification theorem.

We now pass to the coincidence theorem.

**1.7.3. Lemma.** Let X be a metric space and M a subspace of X. For every family  $\{F_i\}_{i=1}^k$  of pairwise-disjoint closed subsets of M there exists a family  $\{W_i\}_{i=1}^k$  of pairwise-disjoint open subsets of X such that  $F_i \subset W_i$  for i = 1, 2, ..., k.

Ch. 1, § 7] The compactification and coincidence theorems

Proof. The sets

$$W_i = \bigcap_{j \neq i} \{ x \in X \colon \varrho(x, F_i) < \varrho(x, F_j) \}$$

have all the required properties.  $\Box$ 

**1.7.4. Lemma.** For every separable metric space X we have  $\dim X \leq \operatorname{ind} X$ .

**Proof.** We can suppose that  $\operatorname{ind} X < \infty$ . If  $\operatorname{ind} X = -1$ , we clearly have  $\dim X \leq \operatorname{ind} X$ . Consider the case where  $\operatorname{ind} X = 0$ . Let  $\mathscr{U} = \{U_i\}_{i=1}^k$  be a finite open cover of the space X. By virtue of Proposition 1.2.4, the cover  $\mathscr{U}$  has a refinement  $\{V_i\}_{i=1}^\infty$  consisting of open-and-closed subsets of X. The sets

$$W_1 = V_1, \quad W_2 = V_2 \setminus W_1, \quad \dots,$$
$$W_i = V_i \setminus (W_1 \cup W_2 \cup \dots \cup W_{i-1}), \quad \dots$$

are open-and-closed and pairwise disjoint, and form a cover of the space X which refines the cover  $\mathcal{U}$ . From Proposition 1.6.9 it follows that dim  $X \leq 0$ , so that again dim  $X \leq ind X$ .

Now, consider the case where  $\operatorname{ind} X = n > 0$ . Let  $\mathscr{U} = \{U_i\}_{i=1}^k$  be a finite open cover of the space X. By virtue of the second decomposition theorem

$$X = Z_1 \cup Z_2 \cup ... \cup Z_{n+1}$$
, where  $ind Z_j \leq 0$  for  $j = 1, 2, ..., n+1$ .

It follows from the already proved special case of our lemma that  $\dim Z_j \leq 0$  for j = 1, 2, ..., n+1. Hence the cover  $\mathscr{U}|Z_j$  of the space  $Z_j$  has a shrinking  $\{F_{j,i}\}_{i=1}^k$  consisting of pairwise disjoint open-and-closed subsets of  $Z_j$ . Applying Lemma 1.7.3, we obtain a family  $\{W_{j,i}\}_{i=1}^k$  of pairwise disjoint open subsets of X such that  $F_{j,i} \subset W_{j,i}$  for i = 1, 2, ..., k. The sets  $V_{j,i} = W_{j,i} \cap U_i$ , where i = 1, 2, ..., k and j = 1, 2, ..., n+1, form an open cover of the space X which refines the cover  $\mathscr{U}$ ; the order of this cover is not larger than n, because any n+2 sets  $V_{j,i}$  include at least two with the same index j, and each two of such sets have an empty intersection. Thus  $\dim X \leq n$ , i.e.,  $\dim X \leq \inf X$ .

Let us observe that in the case where indX = 0 the inequality  $dimX \le indX$  follows from Theorems 1.6.4 and 1.6.11, but the argument given above is much simpler.

1.7.5. Remark. It follows from the second part of the last proof that for every separable subspace M of a metric space X such that  $ind M \le n$ 

and for every family  $\{U_i\}_{i=1}^k$  of open subsets of X such that  $M \subset \bigcup_{i=1}^k U_i$ there exists a finite family  $\mathscr{V}$  of open subsets of X satisfying the conditions  $M \subset \bigcup \mathscr{V}$  and  $\operatorname{ord} \mathscr{V} \leq n$  and such that each of its members is contained in a set  $U_i$ .

# **1.7.6. Lemma.** For every compact metric space X we have $\operatorname{ind} X \leq \dim X$ .

**Proof.** We can suppose that  $\dim X < \infty$ . We shall apply induction with respect to  $\dim X$ . If  $\dim X = -1$ , we clearly have  $\operatorname{ind} X \leq \dim X$ . Assume that our inequality holds for all compact metric spaces with covering dimension  $\leq n-1$  and consider a compact metric space X such that  $\dim X = n \geq 0$ , a point  $x \in X$ , and a closed set B such that  $x \notin B$ . It suffices to define open sets  $K, M \subset X$  which, together with the set  $L = X \setminus (K \cup M)$ , satisfy the conditions

(4) 
$$x \in K$$
,  $B \subset M$ ,  $K \cap M = \emptyset$  and  $\dim L \leq n-1$ ;

indeed, the set L is then a partition between x and B, and  $\operatorname{ind} L \leq n-1$  by virtue of the inductive assumption. To that end we shall define two sequences  $K_0, K_1, K_2, \ldots$  and  $M_0, M_1, M_2, \ldots$  of closed subsets of X satisfying for  $i = 1, 2, \ldots$  the following conditions:

(5) 
$$x \in K_{i-1} \subset \operatorname{Int} K_i$$
,  $B \subset M_{i-1} \subset \operatorname{Int} M_i$  and  $K_i \cap M_i = \emptyset$ .

(6) The set  $L_i = X \setminus (K_i \cup M_i)$  has a finite open cover with mesh < 1/iand order  $\leq n-1$ .

Let  $K_0 = \{x\}$ ,  $M_0 = B$ , and assume that the sets  $K_i$ ,  $M_i$  are already defined for i < j and satisfy (5) and (6) for 0 < i < j. Since in a compact metric space the distance of two disjoint closed sets is positive, there exists a finite open cover  $\mathcal{U}_j$  of the space X such that mesh $\mathcal{U}_j < \min(1/j, \varrho(K_{j-1}, M_{j-1}))$  and  $\operatorname{ord} \mathcal{U}_j \leq n$ . Let  $K_j = X \setminus H_j$  and  $M_j = X \setminus G_j$ , where

$$G_j = \bigcup \{ U \in \mathscr{U}_j \colon U \cap M_{j-1} = \emptyset \}$$

and

$$H_i = \bigcup \{ U \in \mathcal{U}_i : \overline{U} \cap M_{i-1} \neq \emptyset \}.$$

As the closure of no member of  $\mathcal{U}_j$  meets both  $K_{j-1}$  and  $M_{j-1}$ , it follows from the definitions of  $G_j$  and  $H_j$  that

$$\overline{G}_{j} \cap M_{j-1} = \emptyset = \overline{H}_{j} \cap K_{j-1},$$

which implies that  $K_{j-1} \subset X \setminus \overline{H_j} = \operatorname{Int} K_j$  and  $M_{j-1} \subset X \setminus \overline{G_j} = \operatorname{Int} M_j$ ; moreover,  $K_j \cap M_j = 0$ , because  $G_j \cup H_j = X$ . Thus, condition (5) is satisfied for i = j. The family  $\mathscr{W}_j = \{U \cap L_j: U \in \mathscr{U}_j \text{ and } \overline{U} \cap M_{j-1} \neq \emptyset\}$ is an open cover of the set  $L_j = X \setminus (K_j \cup M_j) = G_j \cap H_j$  and mesh $\mathscr{W}_j$ < 1/j. Since every point  $x \in L_j \subset G_j$  belongs to at least one set  $U \in \mathscr{U}_j$ such that  $\overline{U} \cap M_{j-1} = \emptyset$ , the order of  $\mathscr{W}_j$  is not larger than n-1. Thus condition (6) is also satisfied for i = j, so that the construction of the sequences  $K_0, K_1, K_2, \ldots$  and  $M_0, M_1, M_2, \ldots$  is completed.



Fig. 6

The open sets  $K = \bigcup_{i=0}^{\infty} K_i$  and  $M = \bigcup_{i=0}^{\infty} M_i$  are disjoint and contain x and B respectively, and the set  $L = X \setminus (K \cup M) = \bigcap_{i=1}^{\infty} L_i$  satisfies, by virtue of (6) and Theorem 1.6.12, the inequality dim  $L \le n-1$ ; hence conditions (4) hold and the proof of the theorem is completed.  $\Box$ 

**1.7.7. The coincidence theorem.** For every separable metric space X we have ind X = Ind X = dim X.

**Proof.** By virtue of Theorem 1.6.4 and Lemma 1.7.4 it suffices to show that  $\operatorname{ind} X \leq \dim X$ . Apply the compactification theorem to obtain a compactification  $\tilde{X}$  of the space X such that  $\dim \tilde{X} \leq \dim X$ . It follows from Lemma 1.7.6 that  $\operatorname{ind} \tilde{X} \leq \dim \tilde{X}$ , so that  $\operatorname{ind} X \leq \dim X$  by virtue of the subspace theorem.  $\Box$ 

Let us now make some comments on the last theorem. To prove that for every separable metric space X we have indX = dimX one has to establish two inequalities:  $dimX \le indX$  and  $indX \le dimX$ . The proof of the former is fairly easy, and the proof of the latter much more difficult. Usually, we prove the inequality  $\operatorname{ind} X \leq \dim X$  for compact metric spaces, which greatly simplifies the arguments, and then apply the compactification theorem to extend the inequality over all separable metric spaces. In all the existing proofs of the inequality  $\operatorname{ind} X \leq \dim X$  one can detect an auxiliary integer-valued invariant d(X) for which the inequalities ind X  $\leq d(X)$  and  $d(X) \leq \dim X$  are established. In our proof the invariant d(X), defined for every compact metric space X, was the smallest integer  $n \ge 0$  such that the space X has finite open covers of order  $\le n$  with arbitrarily small meshes; the inequality  $d(X) \leq \dim X$  was trivial and the proof reduced to showing that  $ind X \leq d(X)$ . In Section 1.11 we shall, incidentally, give another proof of the inequality  $\operatorname{ind} X \leq \operatorname{dim} X$  for compact metric spaces. In that proof the invariant d(X) will be the smallest integer  $n \ge 0$  such that X is embeddable in the space  $N_n^{2n+1} \subset \mathbb{R}^{2n+1}$ ; since ind X  $\leq d(X)$  by virtue of Example 1.5.9, the proof will reduce to showing that  $d(X) \leq \dim X$ . One more proof of the inequality  $\operatorname{ind} X \leq \dim X$  is sketched out in Problem 1.7.D.

Let us also note that in Section 4.1 the proof of Lemma 1.7.6 will reappear almost verbatim as part of the proof that the inequality  $IndX \le dim X$  holds for every metric space X.

We conclude this section with a characterization of dimension stated in terms of partitions. At the basis of this characterization of dimension lies an interesting geometric property of the *n*-cube  $I^n$ ; viz., the property that if  $L_i$  is a partition between the pair of opposite faces  $A_i = \{\{x_j\} \in I^n : x_i = 0\}$  and  $B_i = \{\{x_j\} \in I^n : x_i = 1\}$  of  $I^n$  for i = 1, 2, ..., n, then  $\bigcap_{i=1}^n L_i \neq \emptyset$ ; this property is closely related to the fixed-point property of  $I^n$  (see Theorem 1.8.1 and Problem 1.8.B). Let us note that the importance of the theorem on partitions consists in the fact that it provides an internal characterization of *n*-dimensional spaces which, in effect, is but a reformulation of an important external characterization of such spaces, namely of the characterization in terms of mappings to the *n*-sphere  $S^n$  (see Theorem 1.9.3), which in turn is very close to the cohomological characterization of dimension (see the final part of Section 1.9). The theorem on partitions will also be used in the proof of the fundamental equality ind  $R^n = n$  (see Theorem 1.8.2).

Let us begin with an auxiliary theorem which will often be used in our study of the covering dimension.

**1.7.8. Theorem.** Every finite open cover  $\{U_i\}_{i=1}^k$  of a normal space X has a closed shrinking  $\{F_i\}_{i=1}^k$ .

**Proof.** We shall apply induction with respect to k beginning with k = 2. In this case, applying the definition of normality to disjoint closed subsets  $A = X \setminus U_1$  and  $B = X \setminus U_2$  of the space X, we obtain disjoint open sets  $V_1, V_2 \subset X$  such that  $A \subset V_1$  and  $B \subset V_2$ . The sets  $F_1 = X \setminus V_1$  and  $F_2 = X \setminus V_2$  form the required shrinking. Assume that the theorem is proved for every natural number  $k < m \ge 3$  and consider an *m*-element open cover  $\{U_i\}_{i=1}^m$  of the space X. Define

$$U'_i = U_i \text{ for } i \leq m-2 \text{ and } U'_{m-1} = U_{m-1} \cup U_{m-2};$$

applying the theorem to the cover  $\{U'_i\}_{i=1}^{m-1}$  we obtain a closed shrinking  $\{F'_i\}_{i=1}^{m-1}$ . The closed subspace  $F'_{m-1}$  of the space X is normal; applying the theorem again, this time to the two-element open cover  $\{F'_{m-1} \cap U_{m-1}, F'_{m-1} \cap U_m\}$  of the space  $F'_{m-1}$ , we obtain a closed cover  $\{F_{m-1}, F_m\}$  of  $F'_{m-1}$  such that  $F_{m-1} \subset U_{m-1}$  and  $F_m \subset U_m$ . One readily checks that the family  $\{F_i\}_{i=1}^m$ , where  $F_i = F'_i$  for  $i \leq m-2$ , is the required shrinking of the cover  $\{U_i\}_{i=1}^m$  of the space X.  $\square$ 

**1.7.9. Theorem on partitions.** A separable metric space X satisfies the inequality ind  $X \le n \ge 0$  if and only if for every sequence  $(A_1, B_1), (A_2, B_2), \ldots, (A_{n+1}, B_{n+1})$  of n+1 pairs of disjoint closed subsets of X there exist closed sets  $L_1, L_2, \ldots, L_{n+1}$  such that  $L_i$  is a partition between  $A_i$  and  $B_i$  and  $\bigcap_{i=1}^{n+1} L_i = \emptyset$ .

**Proof.** If  $ind X \le n \ge 0$ , then—applying the second separation theorem we can define, one by one, partitions  $L_1, L_2, ..., L_{n+1}$  such that  $ind(L_1 \cap$ 

$$\cap L_2 \cap \ldots \cap L_i$$
  $\leqslant n-i$  for  $i = 1, 2, ..., n+1$ ; clearly  $\bigcap_{i=1}^{n+1} L_i = \emptyset$ .

We shall now show that if X satisfies the condition in the theorem, then dim  $X \leq n$  which, by the coincidence theorem, will complete the proof. We are going to apply Theorem 1.6.10; consider thus an (n+2)-element open cover  $\{U_i\}_{i=1}^{n+2}$  of the space X. By Theorem 1.7.8, the cover  $\{U_i\}_{i=1}^{n+2}$ has a closed shrinking  $\{B_i\}_{i=1}^{n+2}$ ; let  $A_i = X \setminus U_i$  for i = 1, 2, ..., n+1. The sequence  $(A_1, B_1), (A_2, B_2), ..., (A_{n+1}, B_{n+1})$  consists of n+1 pairs of disjoint closed subsets of X. Hence, there exist closed sets  $L_1, L_2, ..., L_{n+1}$
such that  $L_i$  is a partition between  $A_i$  and  $B_i$  and  $\bigcap_{i=1}^{n+1} L_i = \emptyset$ . Let us consider open sets  $V_i$ ,  $W_i \subset X$  such that

(7) 
$$A_i \subset V_i$$
,  $B_i \subset W_i$ ,  $V_i \cap W_i = \emptyset$  and  $X \setminus L_i = V_i \cup W_i$   
for  $i = 1, 2, ..., n+1$ . Observe that

(8)  $\bigcup_{i=1}^{n+1} V_i \cup \bigcup_{i=1}^{n+1} W_i = \bigcup_{i=1}^{n+1} (V_i \cup W_i) = \bigcup_{i=1}^{n+1} (X \setminus L_i) = X \setminus \bigcap_{i=1}^{n+1} L_i = X.$ 

From (8), (7) and the inclusion  $B_{n+2} \subset U_{n+2}$  it follows that

$$\bigcup_{i=1}^{n+1} W_i \cup [U_{n+2} \cap \bigcup_{i=1}^{n+1} V_i] = [\bigcup_{i=1}^{n+1} W_i \cup U_{n+2}] \cap [\bigcup_{i=1}^{n+1} W_i \cup \bigcup_{i=1}^{n+1} V_i] \supset \bigcup_{i=1}^{n+2} B_i = X,$$

so that the family  $\{W_i\}_{i=1}^{n+2}$ , where  $W_{n+2} = U_{n+2} \cap \bigcup_{i=1}^{n+1} V_i$ , is an open shrinking of the cover  $\{U_i\}_{i=1}^{n+2}$ . It follows from (7) that

$$\bigcap_{i=1}^{n+2} W_i = \bigcap_{i=1}^{n+1} W_i \cap [U_{n+2} \cap \bigcup_{i=1}^{n+1} V_i] \subset \bigcap_{i=1}^{n+1} W_i \cap \bigcup_{i=1}^{n+1} V_i = \emptyset,$$

therefore dim  $X \leq n$  by virtue of Theorem 1.6.10.  $\Box$ 

**1.7.10. Remark.** Let us note that in the second part of the above proof only the normality of the space X was applied; hence, we have shown that if for every sequence  $(A_1, B_1), (A_2, B_2), \ldots, (A_{n+1}, B_{n+1})$  of n+1 pairs of disjoint closed subsets of a normal space X there exist closed sets  $L_1, L_2, \ldots, L_{n+1}$  such that  $L_i$  is a partition between  $A_i$  and  $B_i$  and  $\bigcap_{i=1}^{n+1} L_i = \emptyset$ , then dim $X \leq n$ .

Let us call the reader's attention to the structure of the last proof. We started with the inequality  $ind X \le n$ , showed that this inequality implies a property of the space X, then proved that this property implies the inequality dim $X \le n$ , and, finally, applied the coincidence theorem. Thus, we incidentally gave another proof of Lemma 1.7.4; obviously, the original proof of Lemma 1.7.4 is more perspicuous. The proof of Theorem 1.9.3 below has a similar structure. The conditions in both theorems characterize the covering dimension dim in the realm of normal spaces; however, in the realm of separable metric spaces the proofs are considerably shortened by applying the coincidence theorem (cf. Theorems 3.2.6 and 3.2.10).

[Ch. 1, § 7

### Problems

## Historical and bibliographic notes

Theorem 1.7.2 was established by Hurewicz in [1927b] (another proof in [1930]). This theorem permitted or facilitated the extension of many theorems of dimension theory from compact metric spaces to separable metric spaces; it was an important achievement in the development of the theory. Lemma 1.7.4 was proved for metric compact spaces by Menger in [1924] and by Urysohn in [1926] (announcement in [1922]) and was extended to separable metric spaces by Hurewicz in [1927b]. Lemma 1.7.6 was obtained by Urysohn in [1926]. The equality ind $X = \dim X$  in Theorem 1.7.7 was established by Hurewicz in [1927b], the history of the equality ind $X = \operatorname{Ind} X$  is described in the notes to Section 1.5. Theorem 1.7.9 was proved by Eilenberg and Otto in [1938].

## Problems

1.7.A. Observe that the enlargement theorem readily follows from the compactification theorem and the Lavrentieff theorem (see [GT], Theorem 4.3.21).

**1.7.B.** Let X be a separable metric space such that dimX = n and  $f_{i,j}: X \to I$ , where i = 1, 2, ... and  $j = 1, 2, ..., k_i$ , the functions defined in the proof of Theorem 1.7.2. Show that by assigning to every point  $x \in X$  the point f(x) in the Hilbert cube  $I^{\aleph_0}$ , the n(i, j)-th coordinate of which is equal to  $f_{i,j}(x)$ , one defines a homeomorphic embedding  $f: X \to I^{\aleph_0}$  such that dim $\overline{f(X)} \leq n$ .

**1.7.C.** (a) (Engelking [1960], Forge [1961]) Show that for every separable metric space X and for every sequence of continuous functions  $f_1, f_2, ...$ , where  $f_i: X \to I$  for i = 1, 2, ..., there exists a compactification  $\tilde{X}$  of the space X such that dim $\tilde{X} \leq \dim X$  and each  $f_i$  is extendable to a continuous function  $\tilde{f_i}: \tilde{X} \to I$ .

(b) (Engelking [1960]; for n = 0, de Groot and McDowell [1960]) Prove that for every separable metric space X and for every sequence of continuous mappings  $g_1, g_2, ...$ , where  $g_i: X \to X$  for i = 1, 2, ..., there exists a compactification  $\tilde{X}$  of the space X such that dim $\tilde{X} \leq \dim X$  and each  $g_i$  is extendable to a continuous mapping  $\tilde{g}_i: \tilde{X} \to \tilde{X}$ .

*Hint*. One can assume that  $g_1 = id_x$  and that for every pair *i*, *j* there exists a k such that  $g_jg_i = g_k$ . As in the proof of Theorem 1.7.2, for m

= 1, 2, ... define a finite open cover  $\mathscr{U}_m$  which moreover is a refinement of  $g_1^{-1}(\mathscr{U}_{m-1}) \wedge g_2^{-1}(\mathscr{U}_{m-1}) \wedge \ldots \wedge g_{m-1}^{-1}(\mathscr{U}_{m-1})$ , where  $\mathscr{U}_0 = \{X\}$ . Consider the metric  $\tilde{\varrho}$  on the space X defined by letting

$$\tilde{\varrho}(x, y) = \sum_{i=1}^{\infty} \frac{1}{2^{i}} \varrho(g_{i}(x), g_{i}(y)) + \sum_{n(i,j,k)=1}^{\infty} \frac{1}{2^{n(i,j,k)}} |f_{i,j}g_{k}(x) - f_{i,j}g_{k}(y)|,$$

where n(i, j, k) is the place of the triple (i, j, k) in an infinite sequence consisting of all triples of natural numbers.

**1.7.D.** (a) (Kuratowski [1932]) Prove that for every compact metric space X with no isolated point such that  $0 \leq \dim X \leq n$  there exists a continuous mapping  $f: C \to X$  of the Cantor set C onto the space X with fibres of cardinality at most n+1. Deduce the topological characterization of the Cantor set stated in Problem 1.3.F.

Hint. Observe that the subspace  $\Phi = \{f \in X^C : f(C) = X\}$  of the function space  $X^C$  is non-empty and closed in  $X^C$ , and hence is completely metrizable (see Problem 1.3.D and the beginning of Section 1.11). For k = 1, 2, ... consider the subset  $\Psi_k$  of  $\Phi$  consisting of all functions  $f \in X^C$ which have the property that for some n+2 points  $x_1, x_2, ..., x_{n+2}$  of the Cantor set C such that  $|x_i - x_j| > 1/k$  whenever  $i \neq j$  we have the equality  $f(x_1) = f(x_2) = ... = f(x_{n+2})$ . Prove that the sets  $\Psi_k$  are closed and nowhere dense in  $\Phi$ ; then apply the Baire category theorem. When proving that  $\Psi_k$  is nowhere dense, observe that every finite open cover of the space X consisting of non-empty sets has a closed shrinking of order  $\leq n$  consisting of non-empty sets; then apply Problem 1.3.D.

(b) (Kuratowski [1932]) Prove that for every compact metric space X such that dim  $X \le n \ge 0$  there exists a continuous mapping  $f: A \to X$  of a closed subspace A of the Cantor set C onto the space X with fibres of cardinality at most n+1.

*Hint*. Applying the definition of dim, prove that  $\dim(X \times C) \leq n$ .

(c) (Hurewicz [1926]) Prove that if for a compact metric space X there exists a continuous mapping  $f: A \to X$  of a closed subspace A of the Cantor set C onto the space X with fibres of cardinality at most n+1, then  $ind X \leq n$  (cf. Theorem 1.12.2).

*Hint.* Apply induction with respect to *n*. Assuming that (c) holds for every n < m, consider a continuous mapping  $f: A \to X$  of a closed subspace *A* of the Cantor set *C* onto the space *X* with fibres of cardinality at most m+1, a point  $x \in X$ , and a neighbourhood  $V \subset X$  of the point *x*; then take an open-and-closed set  $W \subset A$  such that  $f^{-1}(y) \subset W \subset f^{-1}(V)$  and

Ch. 1, § 8] Dimensional properties of Euclidean space and the Hilbert cube 71

show that the open set  $U = X \setminus f(A \setminus W) \subset X$  satisfies the conditions  $y \in U \subset V$  and  $\operatorname{Fr} U \subset f(W) \cap f(A \setminus W)$ .

(d) (Kuratowski [1932]) Observe that (b) and (c) imply that for every compact metric space X we have  $ind X \leq dim X$ .

*Remark.* Hurewicz proved in [1926] that a separable metric space X satisfies the inequality  $\operatorname{ind} X \leq n \geq 0$  if and only if there exists a closed mapping (see the beginning of Section 1.12)  $f: Z \to X$  of a zero-dimensional separable metric space Z onto the space X with fibres of cardinality at most n+1. This fact can be deduced from the compactification theorem, the coincidence theorem, part (b) of the present problem, and the hint to part (c); obviously, Hurewicz's original proof was a direct one.

Let us also note that, as proved by Nagata in [1960] (announcement by Hurewicz in [1928]), a separable metric space X is a continuous image of a zero-dimensional separable metric space under a closed mapping with finite fibres if and only if X can be represented as the union of countably many finite-dimensional subspaces.

**1.7.E.** Show that a separable metric space X satisfies the inequality  $\dim X \leq n$  if and only if every open cover of the space X has an open refinement of order  $\leq n$  or—equivalently —if every open cover of the space X has an open shrinking of order  $\leq n$  (cf. Proposition 3.2.2).

*Hint*. When proving that every open cover of a separable metric space X satisfying dim  $X \le n$  has an open refinement of order  $\le n$ , first reduce the problem to countable covers, then consider the special case where n = 0, and finally apply the second decomposition theorem and Lemma 1.2.8.

# 1.8. Dimensional properties of Euclidean spaces and the Hilbert cube. Infinite-dimensional spaces

The main result in this section is the fundamental theorem of dimension theory, which states that Euclidean *n*-space  $\mathbb{R}^n$  has dimension *n*, i.e., that ind  $\mathbb{R}^n = \operatorname{Ind} \mathbb{R}^n = \dim \mathbb{R}^n = n$  for n = 1, 2, ... This theorem justifies the definitions of our three dimension functions, because any dimension function assigning to  $\mathbb{R}^n$  a number distinct from *n* would contradict the intuitive notion of dimension and thus would not be acceptable. It readily follows from the evaluation of dimensions of Euclidean *n*-space  $\mathbb{R}^n$  that the dimensions ind, Ind and dim of the *n*-cube  $\mathbb{I}^n$  and the *n*-sphere  $S^n$  are also equal to *n*. The spaces  $R^n$ ,  $I^n$  and  $S^n$  are our first examples of spaces of dimension larger then one; so far we have not shown that such spaces exist.

The proof that Euclidean *n*-space  $\mathbb{R}^n$  has dimension *n* requires a deeper insight into the structure of this space; by the nature of things, some combinatorial or algebraic arguments must appear in it. In this book, instead of producing a complete proof of the fundamental theorem of dimension theory, we shall deduce this result from the Brouwer fixed-point theorem, which states that for every continuous mapping  $g: \mathbb{I}^n \to \mathbb{I}^n$  there exists a point  $x \in \mathbb{I}^n$  such that g(x) = x. The latter theorem is certainly well known to the reader; let us note that it is closely related to the "geometric versions" of the fundamental theorem of dimension theory, i.e., to Theorems 1.8.1 and 1.8.15 (see Problem 1.8.B).

We begin with a theorem reflecting an interesting geometric property of the *n*-cube  $I^n$ .

**1.8.1. Theorem.** Let  $A_i$  and  $B_i$ , where i = 1, 2, ..., n, be the subsets of the n-cube  $I^n$  defined by the conditions

$$A_{i} = \{\{x_{j}\} \in I^{n}: x_{i} = 0\} \quad and \quad B_{i} = \{\{x_{j}\} \in I^{n}: x_{i} = 1\},\$$

*i.e.*, the pairs of opposite faces of  $I^n$ . If  $L_i$  is a partition between  $A_i$  and  $B_i$  for i = 1, 2, ..., n, then  $\bigcap_{i=1}^n L_i \neq \emptyset$ .

**Proof.** Let us consider open sets  $U_i$ ,  $W_i \subset I^n$  such that  $A_i \subset U_i$ ,  $B_i \subset W_i$ ,  $U_i \cap W_i = \emptyset$  and  $I^n \setminus L_i = U_i \cup W_i$  for i = 1, 2, ..., n. Since  $(I^n \setminus W_i) \cap \cap (I^n \setminus U_i) = I^n \setminus (U_i \cup W_i) = L_i$ , the formulas

(1) 
$$f_i(x) = \begin{cases} \frac{1}{2} \frac{\varrho(x, L_i)}{\varrho(x, L_i) + \varrho(x, A_i)} + \frac{1}{2} & \text{for} \quad x \in I^n \setminus W_i, \\ -\frac{1}{2} \frac{\varrho(x, L_i)}{\varrho(x, L_i) + \varrho(x, B_i)} + \frac{1}{2} & \text{for} \quad x \in I^n \setminus U_i \end{cases}$$

define for i = 1, 2, ..., n a continuous function  $f_i: I^n \to I$ . Clearly, we have

(2) 
$$f_i^{-1}(1/2) = L_i$$
,  $f_i(A_i) = \{1\}$  and  $f_i(B_i) = \{0\}$ .

Assume that  $\bigcap_{i=1}^{n} L_i = \emptyset$ ; it follows from the first part of (2) that the continuous mapping  $f: I^n \to I^n$  defined by letting  $f(x) = (f_1(x), f_2(x), \dots, f_n(x))$  for  $x \in I^n$  does not assume the value  $a = (1/2, 1/2, \dots)$ 

...,  $1/2 
ightarrow I^n$ . The composition  $g: I^n \to I^n$  of the mapping f and the projection p of  $I^n \setminus \{a\}$  from the point a onto the boundary of  $I^n$ , i.e., onto the set  $B = \bigcup_{i=1}^n (A_i \cup B_i)$ , satisfies the inclusion  $g(I^n) \subset B$ ; by the second and the third part of (2), we have  $g(A_i) \subset B_i$  and  $g(B_i) \subset A_i$ . The last three inclusions show that  $g(x) \neq x$  for every  $x \in I^n$ , which contradicts the Brouwer fixed-point theorem. Hence  $\bigcap_{i=1}^n L_i \neq \emptyset$ .  $\Box$ 

**1.8.2.** The fundamental theorem of dimension theory. For every natural number n we have

$$\operatorname{ind} R^n = \operatorname{Ind} R^n = \dim R^n = n.$$

**Proof.** By virtue of the inequality  $\operatorname{ind} \mathbb{R}^n \leq n$  established in Example 1.5.9 and by the coincidence theorem, it suffices to show that  $\operatorname{ind} \mathbb{R}^n \geq n$ ; the latter inequality follows immediately from Theorem 1.8.1 and the theorem on partitions.  $\Box$ 

### 1.8.3. Corollary. For every natural number n we have

 $\operatorname{ind} I^n = \operatorname{Ind} I^n = \dim I^n = n = \dim S^n = \operatorname{Ind} S^n = \operatorname{ind} S^n$ .  $\Box$ 

**1.8.4. Corollary.** For the Hilbert cube  $I^{\aleph_0}$  we have

ind 
$$I^{\aleph_0} = \operatorname{Ind} I^{\aleph_0} = \dim I^{\aleph_0} = \infty$$
.  $\Box$ 

**1.8.5. Theorem.** For the subspace  $N_k^n$  of Euclidean n-space  $\mathbb{R}^n$  consisting of all points which have at most k rational coordinates and the subspace  $L_k^n$  of  $\mathbb{R}^n$  consisting of all points which have at least k rational coordinates we have

 $\operatorname{ind} N_k^n = k$  and  $\operatorname{ind} L_k^n = n - k$ .

**Proof.** In Example 1.5.9 we have shown that  $\operatorname{ind} N_k^n \leq k$  and  $\operatorname{ind} L_k^n \leq n-k$ . Since  $R^n = N_k^n \cup L_{k+1}^n = N_{k-1}^n \cup L_k^n$ , the reverse inclusions follow from the equality  $\operatorname{ind} R^n = n$  and the addition theorem.  $\Box$ 

Looking closer at the above proof of the fundamental theorem of dimension theory we can see that first the inequalities  $\operatorname{ind} R^n \leq n$  and  $\operatorname{Ind} R^n \geq n$  were established, then—by applying Theorem 1.6.4—it was deduced that  $\operatorname{ind} R^n = \operatorname{Ind} R^n = n$ , and finally Lemmas 1.7.4 and 1.7.6 were applied to obtain the equality  $\dim R^n = n$ . The equality  $\dim I^n = n$ 

[Ch. 1, § 8

can be also obtained in a direct way. Theorem 1.7.8 and Theorem 1.8.15 below imply that dim  $I^n \ge n$ , and the reverse inequality is obtained by defining a finite open cover of  $I^n$  which has order  $\le n$  and arbitrarily small mesh. This can be done by dividing  $I^n$  into small "bricks" and then enlarging the bricks to open sets in such a way that the order of the family does not increase; Fig. 7 illustrates the procedure in the case of n = 2.



Fig. 7

Our next task is to characterize *n*-dimensional subspaces of Euclidean *n*-space  $\mathbb{R}^n$ ; we shall show that they are subsets of  $\mathbb{R}^n$  which have a nonempty interior. We start with an auxiliary geometric result, interesting in itself, which states that if a subset C of  $\mathbb{R}^n$  has an empty interior, then C is homeomorphic to a subset D of  $\mathbb{R}^n$  contained together with its closure in  $N_{n-1}^n$ ; in particular, each such subset is homeomorphic to a nowhere dense subset of  $\mathbb{R}^n$ .

**1.8.6. Lemma.** If a subset C of  $\mathbb{R}^n$  has an empty interior and is dense in  $\mathbb{R}^n$ , then every subset D of  $\mathbb{R}^n$  which is homeomorphic to C has an empty interior.

**Proof.** Let  $f: D \to C$  be a homeomorphism of D onto C. Suppose that Ind  $D \neq \emptyset$  and consider a non-empty open subset V of  $\mathbb{R}^n$  such that the closure  $\overline{V}$  of V in  $\mathbb{R}^n$  is compact and contained in D. The image f(V) is open in C and its closure in C equals  $f(\overline{V})$ ; the last set being compact, the closure  $\overline{f(V)}$  of f(V) in  $\mathbb{R}^n$  also equals  $f(\overline{V})$ . Now, let U be an open subset of  $\mathbb{R}^n$ such that  $f(V) = C \cap U$ . As the set C is dense in  $\mathbb{R}^n$ , we have  $\emptyset \neq U \subset \overline{U}$  $= \overline{C \cap U} = \overline{f(V)} = f(\overline{V}) \subset C$ , which contradicts the assumption that Int  $C = \emptyset$ .  $\square$  Ch. 1, § 8] Dimensional properties of Euclidean spaces and the Hilbert cube 75

**1.8.7. Lemma.** Let A be a subset of  $\mathbb{R}^n$  and a an arbitrary point of  $\mathbb{R}^n \setminus A$ . For every positive number r there exists a homeomorphism  $f: A \to f(A) \subset \mathbb{R}^n$  such that for all  $x, y \in A$  we have

 $\varrho(x, f(x)) \leq r, \quad \varrho(x, y) \leq \varrho(f(x), f(y)) \quad and \quad B(a, r) \cap f(A) = \emptyset.$ 

**Proof.** It suffices to assign to every  $x \in A$  the point f(x) situated on the ray starting from a and passing through x, which satisfies the condition  $\varrho(a, f(x)) = \varrho(a, x) + r$ .  $\Box$ 

**1.8.8. Theorem.** For every subset C of  $\mathbb{R}^n$  which has an empty interior there exists a subset D of  $\mathbb{R}^n$  homeomorphic to C and such that  $\overline{D} \subset \mathbb{N}_{n-1}^n$ .

**Proof.** The set C can easily be enlarged to a dense subset of  $\mathbb{R}^n$  which has an empty interior, and it suffices to prove the theorem for this larger set. Hence, without loss of generality, one can assume that the set C is dense in  $\mathbb{R}^n$ . Let us arrange all points of the complement  $\mathbb{R}^n \setminus \mathbb{N}_{n-1}^n$ , i.e., all points of  $\mathbb{R}^n$  with rational coordinates, into a sequence  $a_1, a_2, \ldots$  We shall define inductively a sequence  $C_0, C_1, C_2, \ldots$  of subsets of  $\mathbb{R}^n$ , where  $C_0 = C$ , and a sequence of homeomorphisms  $f_0, f_1, f_2, \ldots$ , where  $f_i: C_i \to C_{i+1}$ , satisfying the following conditions:

(3) 
$$\varrho(x, f_i(x)) \leq 1/3^{i+1}$$
 for every  $x \in C_i$ .

(4) 
$$\varrho(x, y) \leq \varrho(f_i(x), f_i(y))$$
 for all  $x, y \in C_i$ .

(5)  $B(a_i, 1/4 \cdot 3^i) \cap C_{i+1} = \emptyset \quad \text{for } i \ge 1.$ 

Conditions (3)-(5) are satisfied for i = 0 if we let  $C_1 = C$  and  $f_0 = id_C$ . Assume that the sets  $C_0, C_1, C_2, ..., C_k$  and the homeomorphisms  $f_0, f_1, f_2, ..., f_{k-1}$  are defined and satisfy (3)-(5) for  $i < k \ge 1$ . The set  $C_k = f_{k-1}(C_{k-1})$  is homeomorphic to  $C_0 = C$ , so that  $\operatorname{Int} C_k = \emptyset$  by virtue of Lemma 1.8.6; thus there exists a point  $a \in B(a_k, 1/4 \cdot 3^{k+1}) \setminus C_k$ . One readily checks that Lemma 1.8.7 applied to the set  $A = C_k$ , the point a, and the number  $r = 1/3^{k+1}$  yields a homeomorphism  $f_k = f: C_k \to f(C_k)$  which together with the set  $C_{k+1} = f(C_k)$  satisfies conditions (3)-(5) for i = k. Thus our construction is completed.

For i = 1, 2, ... consider the composition  $h_i = g_i f_{i-1} ... f_1 f_0: C \to \mathbb{R}^n$ , where  $g_i$  denotes the embedding of  $C_i$  in  $\mathbb{R}^n$ . By virtue of (3), we have for  $k \ge 1$ 

(6) 
$$\varrho(h_i(x), h_{i+k}(x))$$
  
 $\leq \varrho(h_i(x), h_{i+1}(x)) + \varrho(h_{i+1}(x), h_{i+2}(x)) + \dots + \varrho(h_{i+k-1}(x), h_{i+k}(x))$   
 $\leq 1/3^{i+1} + 1/3^{i+2} + \dots + 1/3^{i+k} < 1/2 \cdot 3^i \text{ for every } x \in C;$ 

hence, for every  $x \in C$  the sequence  $h_1(x), h_2(x), ...$  is a Cauchy sequence and thus converges to a point  $h(x) \in \mathbb{R}^n$ . In this way a mapping h of C to  $\mathbb{R}^n$  is defined. Passing in (6) to the limit with respect to k, we see that the sequence of mappings  $h_1, h_2, ...$  uniformly converges to the mapping h, so that h is a continuous mapping. Condition (4) implies that  $\varrho(x, y) \le \varrho(h_i(x), h_i(y))$  for all  $x, y \in C$  and i = 1, 2, ... Hence,  $\varrho(x, y) \le \varrho(h(x), h(y))$  for all  $x, y \in C$ , which implies that h is a one-to-one mapping and that the inverse mapping  $h^{-1}$  is continuous, i.e., that h is a homeomorphism of the set C onto the set  $D = h(C) \subset \mathbb{R}^n$ .

Now, consider a point  $a_i \in \mathbb{R}^n \setminus \mathbb{N}_{n-1}^n$  and an arbitrary point  $z \in D$ ; let  $h^{-1}(z) = x \in C$ . It follows from (5) that

(7) 
$$\varrho(a_i, h_{i+1}(x)) \ge 1/4 \cdot 3^i.$$

By virtue of (6) we have  $\rho(h_{i+1}(x), h_{i+k}(x)) < 1/2 \cdot 3^{l+1}$  for k = 1, 2, ..., so that

(8) 
$$\varrho(h_{i+1}(x), z) \leq 1/2 \cdot 3^{i+1}.$$

Relations (7) and (8) yield the inequality  $\varrho(a_i, z) \ge 1/4 \cdot 3^{i+1}$ ; hence  $B(a_i, 1/4 \cdot 3^{i+1}) \cap D = \emptyset$ , which implies that  $a_i \notin \overline{D}$ . Thus the inclusion  $\overline{D} \subset N_{n-1}^n$  is established.  $\Box$ 

Theorem 1.8.8 yields

**1.8.9. Theorem.** Every subset of  $\mathbb{R}^n$  which has an empty interior is homeomorphic to a nowhere dense subset of  $\mathbb{R}^n$ .  $\Box$ 

**1.8.10. Theorem.** A subspace M of Euclidean n-space  $\mathbb{R}^n$  satisfies the condition ind M = n if and only if the interior of M in  $\mathbb{R}^n$  is non-empty.

**Proof.** If  $\operatorname{Int} M \neq \emptyset$ , then M contains a subspace homeomorphic to  $I^n$ , so that  $\operatorname{ind} M = n$ . On the other hand, if  $\operatorname{Int} M = \emptyset$ , then—as follows from Theorem 1.8.8—the space M is homeomorphic to a subspace of  $N_{n-1}^n$ , so that  $\operatorname{ind} M \leq n-1$  by virtue of Theorem 1.8.5.  $\square$ 

**1.8.11. Corollary.** A subspace M of the n-cube  $I^n$ , or the n-sphere  $S^n$ , satisfies the condition ind M = n if and only if the interior of M is non-empty.  $\Box$ 

**1.8.12. Theorem.** Let X be Euclidean n-space  $\mathbb{R}^n$ , the n-cube  $\mathbb{I}^n$ , or the n-sphere  $S^n$ . If a set  $F \subset X$  is the boundary of a non-empty open subset of X which is not dense in X, then  $\operatorname{ind} F = n-1$ .

**Proof.** Consider first a subspace F of  $S^n$ , and let  $F = \operatorname{Fr} U$ , where U is an open subset of  $S^n$  such that  $U \neq \emptyset \neq S^n \setminus \overline{U}$ . From Corollary 1.8.11 it follows that  $\operatorname{ind} F \leq n-1$ ; suppose that  $\operatorname{ind} F \leq n-2$ . Let x be a point of U. Since  $\overline{U} \neq S^n$ , for every positive number  $\varepsilon$  one can readily define a homeomorphism f of the *n*-sphere  $S^n$  onto itself such that f(x) = x and  $f(\overline{U}) \subset B(x, \varepsilon)$ . Hence, the point x has arbitrarily small neighbourhoods with (n-2)-dimensional boundaries. This contradiction of Theorem 1.8.2 shows that  $\operatorname{ind} F = n-1$ .

Since  $\mathbb{R}^n$  is homeomorphic to the open subspace  $S^n \setminus \{a\}$  of  $S^n$  and for every open set  $U \subset \mathbb{R}^n$  the boundary of U in  $\mathbb{R}^n$  and the boundary of its counterpart in  $S^n$  differ topologically by at most one point, the validity of our theorem for subspaces of  $\mathbb{R}^n$  follows from its validity for subspaces of  $S^n$  and Corollary 1.5.6.

The case of  $X = I^n$  is left to the reader.

Let us recall that a set  $T \subset X$  separates a topological space X if T separates X between a pair of points x and y, i.e., if the complement  $X \setminus T$  is disconnected.

**1.8.13. Theorem.** Let X be Euclidean n-space  $\mathbb{R}^n$ , the n-cube  $\mathbb{I}^n$ , or the n-sphere  $\mathbb{S}^n$ . If a closed subset L of X satisfies the inequality ind  $L \leq n-2$ , then L does not separate the space X, i.e., the complement  $X \setminus L$  is connected.

**Proof.** Suppose that  $X \ L = U \cup V$ , where U, V are non-empty disjoint open sets. Clearly  $\operatorname{Fr} U \subset L$ , so that  $\operatorname{ind} \operatorname{Fr} U \leq n-2$ , which contradicts Theorem 1.8.12.  $\Box$ 

The last theorem leads to the notion of a Cantor-manifold, which will be discussed in the following section.

We shall now pass to a study of deeper dimensional properties of Euclidean spaces. We aim at Mazurkiewicz's theorem, which is much stronger than Theorem 1.8.13 and states that if a subset M of a region  $G \,\subset\, \mathbb{R}^n$  (i.e., of a connected open subspace G of  $\mathbb{R}^n$ ) satisfies the inequality ind  $M \leq n-2$ , then M does not cut G (cf. Problems 1.8.E and 1.8.F). Let us recall that a set  $T \subset X$  cuts a topological space X if T cuts X between a pair of points x and y, i.e., if each continuum  $C \subset X$  which contains x and y meets the set T. We start with Lebesgue's covering theorem, reflecting an interesting geometric property of the *n*-cube  $I^n$ , which is closely related to the equality dim  $I^n = n$ . As stated in the notes to Section 1.1, this theorem played an important role in the formation of dimension theory. **1.8.14. Lemma.** Let  $A_i$ ,  $B_i$ , where i = 1, 2, ..., n, be the pairs of opposite faces of  $I^n$ . If  $I^n = L'_0 \supset L'_1 \supset ... \supset L'_n$  is a descending sequence of closed sets such that  $L'_i$  is a partition in  $L'_{i-1}$  between  $L'_{i-1} \cap A_i$  and  $L'_{i-1} \cap B_i$  for i = 1, 2, ..., n, then  $L'_n \neq \emptyset$ .

**Proof.** By virtue of the second part of Lemma 1.2.9, for i = 2, 3, ..., n there exists a partition  $L_i$  in  $I^n$  between  $A_i$  and  $B_i$  such that

(9)  $L'_{i-1} \cap L_i \subset L'_i$  for i = 2, 3, ..., n; let, additionally,  $L_1 = L'_1$ .

From (9) we obtain, one by one, the inclusions  $L_1 \subset L'_1, L_1 \cap L_2 \subset L'_2, \dots$ ...,  $L_1 \cap L_2 \cap \dots \cap L_n \subset L'_n$ , so that  $L'_n \neq \emptyset$  by Theorem 1.8.1.  $\square$ 

**1.8.15. Lebesgue's covering theorem.** If  $\mathcal{F}$  is a finite closed cover of the *n*-cube  $I^n$  no member of which meets two opposite faces of  $I^n$ , then ord  $\mathcal{F} \ge n$ .

**Proof.** Let  $A_i$ ,  $B_i$ , where i = 1, 2, ..., n, be the pairs of opposite faces of  $I^n$ . For i = 1, 2, ..., n define

$$\mathscr{F}_i = \{F \in \mathscr{F} \colon F \cap A_i \neq \emptyset\}$$

and consider the families

$$\mathscr{K}_1 = \mathscr{F}_1, \quad \mathscr{K}_2 = \mathscr{F}_2 \backslash \mathscr{K}_1, \quad \dots, \quad \mathscr{K}_n = \mathscr{F}_n \backslash (\mathscr{K}_1 \cup \dots \cup \mathscr{K}_{n-1})$$

and

and

$$\mathscr{K}_{n+1} = \mathscr{F} \setminus (\mathscr{K}_1 \cup \mathscr{K}_2 \cup \ldots \cup \mathscr{K}_n).$$

It follows from the assumptions of the theorem that the closed sets  $K_i = \bigcup \mathscr{K}_i$  satisfy the inclusions

(10) 
$$A_i \subset I^n \setminus (K_{i+1} \cup \dots \cup K_{n+1})$$
 and  $B_i \subset I^n \setminus K_i$   
for  $i = 1, 2, \dots, n$ .

The sets  $L'_1, L'_2, \ldots, L'_n$ , where

$$L'_i = K_1 \cap K_2 \cap \ldots \cap K_i \cap (K_{i+1} \cup K_{i+2} \cup \ldots \cup K_{n+1}),$$

are closed and form a descending sequence. Observe that  $I^n \setminus L'_i = U_i \cup W_i$ , where

 $U_i = I^n \setminus (K_{i+1} \cup K_{i+2} \cup \dots \cup K_{n+1})$  $W_i = I^n \setminus (K_1 \cap K_2 \cap \dots \cap K_i).$ 



Fig. 8

Inclusions (10) yield

 $L'_{i-1} \cap A_i \subset L'_{i-1} \cap U_i$  and  $L'_{i-1} \cap B_i \subset L'_{i-1} \cap W_i$ 

for i = 1, 2, ..., n, where  $L'_0 = I^n$ . One readily checks that  $(L'_{i-1} \cap U_i) \cap (L'_{i-1} \cap W_i) = \emptyset$ ; moreover,

$$L'_{i-1} \setminus [(L'_{i-1} \cap U_i) \cup (L'_{i-1} \cap W_i)] = L'_{i-1} \setminus (U_i \cup W_i) = L'_{i-1} \cap L'_i = L'_i,$$

so that  $L'_i$  is a partition in  $L'_{i-1}$  between  $L'_{i-1} \cap A_i$  and  $L'_{i-1} \cap B_i$  for i = 1, 2, ..., n. By virtue of Lemma 1.8.14, we have  $L'_n = K_1 \cap K_2 \cap ... \cap K_{n+1} \neq \emptyset$ . Since every member of the cover  $\mathscr{F}$  belongs to exactly one family  $\mathscr{K}_i$ , the last inequality implies that  $\operatorname{ord} \mathscr{F} \ge n$ .  $\Box$ 

If L is a partition in  $I^n$  between  $A_1$  and  $B_1$ , then letting  $L'_1 = L$  and defining, as in the above proof, the families  $\mathcal{F}_i, \mathcal{K}_i$  and the sets  $K_i, L'_i$  for i = 2, 3, ..., n, we obtain the following generalization of Lebesgue's covering theorem:

**1.8.16. Theorem.** If L is a partition between a pair of opposite faces of the *n*-cube  $I^n$  and  $\mathcal{F}$  is a finite closed cover of L no member of which meets two opposite faces of  $I^n$ , then ord  $\mathcal{F} \ge n-1$ .  $\Box$ 

We are ready now to prove Mazurkiewicz's theorem. The theorem will be preceded by two lemmas; the second lemma is a special case of the theorem, to which the general situation will be reduced by a relatively simple argument. **1.8.17. Lemma.** Let M be a subspace of a totally bounded metric space X. If there exists a positive number  $\varepsilon$  with the property that every finite family  $\mathscr{V}$  of open subsets of X such that mesh  $\mathscr{V} < \varepsilon$  and  $M \subset \bigcup \mathscr{V}$  satisfies the inequality ord  $\mathscr{V} \ge n \ge 0$ , then ind  $M \ge n$ .

**Proof.** Suppose that  $\operatorname{ind} M \leq n-1$  and consider a finite cover  $\{U_i\}_{i=1}^k$  of the space X by  $\varepsilon/3$ -balls. By virtue of Remark 1.7.5, there exists a finite family  $\mathscr{V}$  of open subsets of X such that mesh  $\mathscr{V} \leq 2\varepsilon/3 < \varepsilon$ ,  $M \subset \bigcup \mathscr{V}$  and  $\operatorname{ord} \mathscr{V} \leq n-1$ , which is impossible. Hence  $\operatorname{ind} M \geq n$ .  $\Box$ 

**1.8.18. Lemma.** Let A, B be a pair of opposite faces of the n-cube  $I^n$ . If M is a subspace of  $X = I^n \setminus (A \cup B)$  which meets every continuum  $C \subset I^n$  such that  $A \cup B \subset C$ , then ind  $M \ge n-1$ .

**Proof.** Consider a finite family  $\mathscr{V}$  of open subsets of X such that mesh  $\mathscr{V} < 1$ and  $M \subset V = \bigcup \mathscr{V}$ ; obviously,  $A \cup B \subset I^n \setminus V$ . Let  $S_A$  and  $S_B$  be the components of the compact space  $I^n \setminus V$  which contain A and B, respectively. Since  $S_A$  and  $S_B$  are both continua and  $A \cup B \subset S_A \cup S_B \subset I^n \setminus M$ , it follows from the assumption about M that  $S_A \cap S_B = \emptyset$ . By virtue of Lemma 1.4.4 there exist disjoint open-and-closed subsets  $U_1$ ,  $W_1$  of  $I^n \setminus V$  such that

(11) 
$$A \subset S_A \subset U_1$$
,  $B \subset S_B \subset W_1$  and  $U_1 \cup W_1 = I^n \setminus V$ .

The sets  $U_1$ ,  $W_1$  being closed in  $I^n$ , there exists open sets  $U, W \subset I^n$  such that

$$U_1 \subset U, \quad W_1 \subset W \quad \text{and} \quad U \cap W = \emptyset.$$

By virtue of (11), the set  $L = I^n \setminus (U \cup W)$  is contained in V and is a partition in  $I^n$  between A and B. From Theorem 1.7.8 it follows that the cover  $\mathscr{V}|L$  of the space L has a closed shrinking  $\mathscr{F}$ . As mesh  $\mathscr{V} < 1$ , no member of  $\mathscr{F}$  meets two opposite faces of  $I^n$ , and Theorem 1.8.16 implies that  $\operatorname{ord} \mathscr{F} \ge n-1$ . Since  $\operatorname{ord} \mathscr{V} \ge \operatorname{ord} \mathscr{F}$ , it follows from Lemma 1.8.17 that  $\operatorname{ind} M \ge n-1$ .  $\Box$ 

**1.8.19.** Mazurkiewicz's theorem. If a subset M of a region  $G \subset \mathbb{R}^n$  satisfies the inequality ind  $M \leq n-2$ , then M does not cut G, i.e., for every pair of points  $x, y \in G \setminus M$  there exists a continuum  $C \subset G \setminus M$  which contains x and y.

**Proof.** Let us start with the special case where  $G = R^n$ . Consider a pair of points  $x, y \in R^n \setminus M$  and denote by K the closed ball in  $R^n$  whose centre coincides with the centre of the segment with end-points x and y and whose radius is equal to  $\frac{1}{2}\varrho(x, y)$ . Let  $f: I^n \to K$  be a continuous mapping of  $I^n$ onto K which maps a pair A, B of opposite faces of  $I^n$  to x and y, respectively, and has the property that the restriction  $g = f|[I^n \setminus (A \cup B)]$  is a homeomorphism of  $I^n \setminus (A \cup B)$  onto  $K \setminus \{x, y\}$ ; such a mapping can easily be obtained by an application of Problem 1.8.A and Lemma 1.8.7. The set  $M' = f^{-1}(K \cap M) \subset I^n \setminus (A \cup B)$  satisfies the inequality  $\operatorname{ind} M' \leq n-2$ . Hence by Lemma 1.8.18 there exists a continuum  $C' \subset I^n$  such that  $A \cup B$  $\subset C'$  and  $C' \cap M' = \emptyset$ . The set  $C = f(C') \subset R^n \setminus M$  is a continuum which contains x and y, so that Mazurkiewicz's theorem is proved for  $G = R^n$ .

Now, consider an arbitrary region  $G \subset \mathbb{R}^n$  and a pair of points  $x, y \in G \setminus M$ . Let  $B_1, B_2, \ldots, B_k$  be a sequence of open balls in  $\mathbb{R}^n$  such that  $x \in B_1, y \in B_k, B_i \subset G$  for  $i = 1, 2, \ldots, k$  and  $B_i \cap B_{i+1} \neq \emptyset$  for  $i \leq k-1$ . The existence of such a sequence follows from the connectedness of G, because the set of all points in G that can be joined to the point x in a similar way is open-and-closed. Since the set M has an empty interior, for  $i = 1, 2, \ldots, k-1$  there exists a point  $z_i \in B_i \cap B_{i+1} \setminus M$ ; let, additionally,  $z_0 = x$  and  $z_k = y$ . By the special case of Mazurkiewicz's theorem established above, for  $i = 1, 2, \ldots, k$  there exists a continuum  $C_i \subset B_i \setminus M$  which contains  $z_{i-1}$  and  $z_i$ . The union  $C = C_1 \cup C_2 \cup \ldots \cup C_k \subset G \setminus M$  is a continuum which contains x and y.  $\Box$ 

We conclude this section with a theorem and an example announced in Section 1.5 (cf. Problem 1.8.G).

**1.8.20. Theorem.** The Hilbert cube  $I^{\aleph_0}$  cannot be represented as a countable union of finite-dimensional subspaces.

**Proof.** By virtue of the second decomposition theorem it is enough to show that for every sequence  $Z_1, Z_2, ...$  of zero-dimensional subspaces of  $I^{\aleph_0}$  we have  $\bigcup_{i=1}^{\infty} Z_i \neq I^{\aleph_0}$ . Let  $A_i = \{\{x_j\} \in I^{\aleph_0}: x_i = 0\}$  and  $B_i = \{\{x_j\} \in I^{\aleph_0}: x_i = 1\}$ 

for i = 1, 2, ... Applying Theorem 1.2.11, we can find a partition  $L_i$  between  $A_i$  and  $B_i$  such that

(12) 
$$L_i \cap Z_i = \emptyset$$
 for  $i = 1, 2, ...$ 

Define  $I_n = \{\{x_j\} \in I^{\aleph_0}: x_j = 0 \text{ for } j > n\}$ ; the intersection  $L_i \cap I_n$  is a partition in  $I_n$  between  $A_i \cap I_n$  and  $B_i \cap I_n$  for i = 1, 2, ..., n. From Theorem 1.8.1 it follows that  $\bigcap_{i=1}^n L_i \supset \bigcap_{i=1}^n L_i \cap I_n \neq \emptyset$ . Hence, the family  $\{L_i\}_{i=1}^\infty$  of closed subsets of  $I^{\aleph_0}$  has the finite intersection property. The space  $I^{\aleph_0}$  being compact,  $\bigcap_{i=1}^\infty L_i \neq \emptyset$ . It follows from (12) that  $\bigcap_{i=1}^\infty L_i \subset I^{\aleph_0} \setminus \bigcup_{i=1}^\infty Z_i$ , so that  $\bigcup_{i=1}^\infty Z_i \neq I^{\aleph_0}$ .  $\Box$ 

**1.8.21. Example.** We shall now describe, applying the *continuum hypothesis*, i.e., the equality  $\aleph_1 = c$ , a space  $X \subset I^{\aleph_0}$  of dimension  $\infty$  whose finite-dimensional subspaces are all countable.

One readily checks that the family of all  $G_{\delta}$ -sets in  $I^{\aleph_0}$  has cardinality c; let us arrange—applying the continuum hypothesis—all zero-dimensional members of this family into a transfinite sequence

$$Z_1, Z_2, \ldots, Z_{\alpha}, \ldots, \alpha < \omega_1,$$

where  $\omega_1$  denotes the first uncountable ordinal number, i.e., the initial number of cardinality  $\aleph_1$ . As all one-point subsets of  $I^{\aleph_0}$  are among the sets  $Z_{\alpha}$ , we have

(13) 
$$\bigcup_{\alpha < \omega_1} Z_{\alpha} = I^{\aleph_0}$$

From Theorem 1.8.20 it follows that  $\bigcup_{\alpha < \gamma} Z_{\alpha} \neq I^{\aleph_0}$  for every  $\gamma < \omega_1$ , so that by virtue of (13) there exists an uncountable set  $\Gamma$  of countable ordinal numbers such that  $Z_{\gamma} \setminus \bigcup_{\alpha < \gamma} Z_{\alpha} \neq \emptyset$  for  $\gamma \in \Gamma$ . Let us choose a point  $x_{\gamma} \in Z_{\gamma} \setminus \bigcup_{\alpha < \gamma} Z_{\alpha}$  for every  $\gamma \in \Gamma$  and consider the subspace  $X = \bigcup_{\gamma \in \Gamma} \{x_{\gamma}\}$ of  $I^{\aleph_0}$ . By Theorem 1.2.14, for each zero-dimensional subspace  $Z \subset X$  $\subset I^{\aleph_0}$  there exists a zero-dimensional  $G_{\delta}$ -set  $Z^* \subset I^{\aleph_0}$  such that  $Z \subset Z^*$ , i.e., there exists an index  $\alpha < \omega_1$  such that  $Z \subset Z_{\alpha}$ . Thus we have Z $\subset \bigcup_{\gamma \leq \alpha} \{x_{\gamma}\}$ , which implies that the subspace Z is countable. It follows from the second decomposition theorem that each finite-dimensional subspace of X is also countable. As the space X itself is uncountable, ind  $X = \infty$ .

Conversely, the existence of a space X with the above properties implies the continuum hypothesis. Indeed, since every non-empty metric space of cardinality < c is zero-dimensional (see Example 1.2.5),  $|X| \ge c$  and every subset of X which has cardinality  $< \mathfrak{c}$  is countable; considering a set  $A \subset X$  such that  $|A| = \aleph_1$ , we conclude that  $\aleph_1 = \mathfrak{c}$ . As the continuum hypothesis is independent of the standard axioms of set theory, the existence of a metric space X such that  $\operatorname{ind} X = \infty$  and all finite-dimensional subspaces of X are countable is also independent of the standard axioms of set theory.  $\Box$ 

In the light of Theorem 1.8.20 we see that among all infinite-dimensional separable metric spaces there are "weakly" infinite-dimensional spaces, for example the union  $\bigcup_{n=1}^{\infty} X_n$ , where  $X_n$  is homeomorphic to the *n*-cube  $I^n$  and  $X_n \cap X_m = \emptyset$  whenever  $n \neq m$ , and "strongly" infinite-dimensional spaces, for example the Hilbert cube  $I^{\aleph_0}$ . The classification of infinitedimensional spaces into spaces which can be represented as countable unions of finite-dimensional spaces (such spaces are called countabledimensional) and spaces which cannot be represented in such a way, suggested by Theorem 1.8.20, is not the only possible classification into "weakly" and "strongly" infinite-dimensional spaces; several classifications of this kind are defined and studied in the literature. It is also possible to extend the inductive definition of ind and Ind from natural numbers to ordinal numbers; in this way one obtains the transfinite small inductive dimension trind and the transfinite large inductive dimension trInd, which satisfy the inequality trind  $X \leq$  trInd X. It turns out, however, that there exist separable metric spaces, even countable-dimensional spaces, with trind larger than any given ordinal number. E.g., the subspace X of the Hilbert cube  $I^{\aleph_0}$  consisting of all points which have at most finitely many coordinates distinct from zero does not satisfy the inequality trind  $X \leq \alpha$ , and, a fortiori, the inequality trInd  $X \leq \alpha$ , for any ordinal number  $\alpha$ ; so that for the space X neither trind nor trInd is defined. One can also prove that for the union  $X = \bigcup_{n=1}^{\infty} X_n$ , where  $X_n$  is homeomorphic to the *n*-cube I<sup>n</sup> and  $X_n \cap X_m = \emptyset$  whenever  $n \neq m$ , trInd is not defined, although, as can easily be seen, trind  $X = \omega_0$ . Finally, there exist compact metric spaces for which both trind and trInd are defined but are distinct ordinal numbers.

The theory dealing with transfinite dimensions and different notions of "weak" infinite-dimensionality now forms quite an extensive part of dimension theory. It includes various characterizations of "weakly" infinite-dimensional spaces (see, e.g., the remark to Problem 1.7.D), variations on addition, sum, enlargement, compactification and universal space theorems, and also many interesting examples. Let us note that in this domain several natural and interesting questions are still unanswered. In consideration of the elementary character of the present book we confine ourselves to calling the reader's attention to these topics; in the notes below we list the most important items of the bibliography of infinitedimensional spaces.

## Historical and bibliographic notes

A proof of the Brouwer fixed-point theorem can be found in the appendix to Section 7.3 of [GT]. Theorem 1.8.1 is implicit in Eilenberg and Otto's paper [1938]. As already mentioned in the notes to Section 1.1, the equality  $\dim R^n = n$  (more exactly, Theorem 1.8.15) was discovered by Lebesgue in [1911] and proved by Brouwer in [1913]; the gap in Lebesgue's original outline of proof, given in [1911], was filled in his later paper [1921]. Brouwer's paper [1913] contains also a proof of the equality  $\operatorname{Ind} R^n = n$ , which is reduced to the equality dim  $R^n = n$ . The equality ind  $R^n = n$  was established—also by a reduction to the equality dim  $R^n = n$  and an application of Lebesgue's result-by Menger in [1924] and by Urysohn in [1925] (announcement in [1922]). Theorem 1.8.9 was given by Sierpiński in [1922]; Theorem 1.8.8 is obtained by a small modification of Sierpiński's proof. Theorems 1.8.10, 1.8.12 and 1.8.13 were obtained by Menger in [1924] and by Urysohn in [1925] (announcement in [1922]). Urysohn in his proof of Theorem 1.8.10 applied an earlier result of Fréchet and Brouwer (see Problem 1.8.D), and Menger showed that if a subspace M of  $R^n$  has an empty interior, then for every point  $x \in \mathbb{R}^n$  and each positive number  $\varepsilon$ there exists a neighbourhood  $U \subset \mathbb{R}^n$  of the point x such that  $\delta(U) < \varepsilon$ , the boundary  $\operatorname{Fr} U$  is homeomorphic to  $S^{n-1}$  and the intersection  $M \cap \operatorname{Fr} U$ has an empty interior in the space Fr U. Theorem 1.8.16 was proved in Lebesgue's paper [1921]. Mazurkiewicz established Theorem 1.8.19 in [1929]. Theorem 1.8.20 and Example 1.8.21 were given by Hurewicz in [1928] and [1932], respectively.

Hurewicz was the first to hint, in [1928], at the possibility of a classification of infinite-dimensional spaces; he defined there countably-dimensional spaces. A comprehensive exposition of the theory of infinite-dimensional spaces can be found in Alexandroff and Pasynkov's book [1973]; some information is contained in Nagata's book [1965]. Further results were obtained by Arhangel'skii in [1963], Lelek in [1965], Nagami in [1965], Nagami and Roberts in [1965], Schurle in [1969], Shmuely in [1972], and Ljuksemburg in [1973] and [1973a].

### Ch. 1, § 8]

#### Problems

## Problems

**1.8.A.** Let us recall that a subset A of Euclidean *n*-space  $\mathbb{R}^n$  is convex if for each pair x, y of points of A the segment with end-points x and y is contained in A.

Show that every convex compact subspace  $A \subset \mathbb{R}^n$  which has a non-empty interior is homeomorphic to the *n*-ball  $B^n$  and its boundary  $\operatorname{Fr} A$  is homeomorphic to the (n-1)-sphere  $S^{n-1}$ . Note that, in particular,  $I^n$  and  $B^n$ , and also  $\operatorname{Fr} I^n$  and  $S^{n-1}$ , are homeomorphic to each other for n = 1, 2, ...

*Hint*. Consider a point  $x \in IntA$  and prove that every ray starting from x meets FrA at exactly one point.

**1.8.B.** Show that the Brouwer fixed-point theorem follows easily both from Theorem 1.8.1 and from Theorem 1.8.15.

*Hint*. The Brouwer fixed-point theorem is equivalent to the statement that  $S^{n-1}$  is not a retract of  $B^n$ .

**1.8.C** (Nöbeling [1932]). Prove that if the projection of a compact subspace X of  $\mathbb{R}^m$  onto the Cartesian product of each n coordinate axes of  $\mathbb{R}^m$ , where 0 < n < m, has an empty interior, then the subspace X is embeddable in  $\mathbb{N}_{n-1}^m$ . Deduce that if an  $F_{\sigma}$ -set  $X \subset \mathbb{R}^m$  satisfies the inequality ind  $X \ge n > 0$ , then there exists a set of n coordinate axes of  $\mathbb{R}^m$  such that the projection of X onto the Cartesian product of these axes has dimension n (the assumption that X is an  $F_{\sigma}$ -set cannot be omitted; see Example 1.10.23).

*Hint.* Let  $H_1, H_2, ...$  be the sequence of all linear (m-n)-varieties in  $\mathbb{R}^m$  defined by conditions of the form  $x_{i_1} = r_1, x_{i_2} = r_2, ..., x_{i_n} = r_n$ , where  $1 \le i_1 < i_2 < ... < i_n \le m$  and  $r_1, r_2, ..., r_n$  are arbitrary rational numbers. Observe that for i = 1, 2, ... the set  $A_i \subset \mathbb{R}^m$  of all points  $a \in \mathbb{R}^m$ such that  $\{x+a: x \in X\} \cap H_i \neq \emptyset$  is closed and has an empty interior.

**1.8.D** (Brouwer [1913a]; implicitly, Fréchet [1910]). Prove that for any two countable dense subsets A, B of Euclidean *n*-space  $\mathbb{R}^n$  there exists a homeomorphism  $f: \mathbb{R}^n \to \mathbb{R}^n$  such that f(A) = B. Note that this result yields Theorem 1.8.10.

*Hint.* A set  $A \subset \mathbb{R}^n$  is in general position with respect to the coordinate axes if for every pair of distinct points  $x = \{x_i\}, y = \{y_i\} \in A$  the difference  $x_i - y_i$  does not vanish for i = 1, 2, ..., n. Prove first that for every countable set  $A \subset \mathbb{R}^n$  there exists a homeomorphism of  $\mathbb{R}^n$  onto itself

that maps A onto a set in general position with respect to the coordinate axes. Then show that the elements of two countably infinite sets in general position with respect to the coordinate axes can be arranged into two sequences  $x_1, x_2, ...$  and  $y_1, y_2, ...$ , where  $x_j = \{x_i^j\}$  and  $y_j = \{y_i^j\}$  for j = 1, 2, ..., in such a way that the differences  $x_i^j - x_i^k$  and  $y_i^j - y_i^k$  have the same sign for  $j, k = 1, 2, ..., j \neq k$ , and i = 1, 2, ..., n (cf. the hint to Problem 1.3.G(a)).

*Remark.* It was proved by Fort in [1962] that the Hilbert cube also has the above strong homogeneity property; more general results were obtained by Bessaga and Pełczyński in [1970] and by Bennett in [1972].

**1.8.E.** (a) Deduce from Theorem 1.8.13 that if a subset M of a region  $G \subset \mathbb{R}^n$  satisfies the inequality ind  $M \leq n-2$ , then M does not separate G, i.e., the complement  $G \setminus M$  is connected.

*Hint.* Observe first that every set T which separates a topological space X between a pair of points x and y contains a partition between x and y. Then reduce the problem to the special case where  $G = R^n$  (cf. the proof of Theorem 1.8.19).

(b) Give an example of a one-dimensional subset of the plane  $R^2$  which does not separate any region  $G \subset R^2$ .

*Remark.* Sitnikov gave in [1954] an example of a two-dimensional subset of  $R^3$  which does not separate any region  $G \subset R^3$ . On the other hand, every (n-1)-dimensional closed subset of  $R^n$  does separate a region  $G \subset R^n$  (see Alexandroff [1930], Frankl and Pontrjagin [1930], and Frankl [1930]).

1.8.F. Give an example of a subset of the plane  $R^2$  which cuts  $R^2$  but does not separate it.

**1.8.G** (Smirnov [1958]). Show that every non-empty separable metric space can be represented as the union of an increasing transfinite sequence of type  $\omega_1$  consisting of zero-dimensional subspaces.

*Hint.* Show that the interval I can be represented in this way, and apply the universality of the Hilbert cube for the class of all separable metric spaces.

*Remark.* Related results can be found in Arhangel'skii's paper [1963] and Nagami's paper [1965].

In the preceding sections we chiefly studied the internal properties of *n*-dimensional spaces. In the present one, and the next two to follow, we shall study the external properties; more exactly, we shall discuss the relations of *n*-dimensional spaces to standard spaces such as spheres, polyhedra and Euclidean spaces.

We begin with a characterization of dimension in terms of extending mappings to spheres from a closed subspace over the whole space.

Let us recall that a continuous mapping  $f: A \to Y$  defined on a subspace A of a space X is continuously extendable over X if there exists a continuous mapping  $F: X \to Y$  such that F|A = f, i.e. F(x) = f(x) for every  $x \in X$ ; the mapping F is called a continuous extension of f. One of the most important results on extending mappings is the Tietze-Urysohn theorem, which states that every continuous function from a closed subspace A of a normal space X to the closed interval I is continuously extendable over X. Urysohn's lemma is a special case of this theorem; it states that for every pair A, B of disjoint closed subsets of a normal space X there exists a continuous function  $f: X \to I$  such that  $f(A) \subset \{0\}$  and  $f(B) \subset \{1\}$ . When X is a metric space, the function f in Urysohn's lemma can be obtained in a particularly simple way, viz., by defining

$$f(x) = \frac{\varrho(x, A)}{\varrho(x, A) + \varrho(x, B)}$$
 for  $x \in X$ .

The Tietze-Urysohn theorem implies that for every continuous mapping  $f: A \to I^n$  of a closed subspace A of a normal space X to the *n*-cube  $I^n$  there exists a continuous extension  $F: X \to I^n$  over X; indeed, it suffices to choose for i = 1, 2, ..., n a continuous extension  $F_i: X \to I$  of the composition  $p_i f$ , where  $p_i: I^n \to I$  is the projection of  $I^n$  onto the *i*-th coordinate axis, and define  $F(x) = (F_1(x), F_2(x), ..., F_n(x))$ .

On the other hand, for every continuous mapping  $f: A \to S^n$  of a closed subspace A of a normal space X to the *n*-sphere  $S^n$  there exists an open set  $U \subset X$  containing A and such that f is continuously extendable over U. Indeed, the two spaces being homeomorphic (see Problem 1.8.A), one can replace the sphere  $S^n$  by the boundary  $S_I^n$  of the (n+1)-cube  $I^{n+1}$  in  $R^{n+1}$ and, applying the above observation, obtain a continuous mapping  $F_0$ :  $X \to I^{n+1}$  such that  $F_0(x) = f(x)$  for every  $x \in A$ ; now, the mapping f is continuously extendable over the open set  $U = F_0^{-1}(I^{n+1} \setminus \{a\}) \subset X$ , where a = (1/2, 1/2, ..., 1/2); the composition F of the restriction  $F_0|U$ and the projection p of  $I^{n+1} \setminus \{a\}$  from the point a to  $S_I^n$  is a continuous extension of f over U.

It is well known (cf. the hint to Problem 1.8.B) that in general continuous mappings to the *n*-sphere  $S^n$  are not continuously extendable from a closed subspace to the whole space. We shall now show that the extendability of such mappings depends exclusively on the dimension of the complement of the subspace under consideration.

**1.9.1. Lemma.** Let X be a separable metric space and A a closed subspace of X such that  $ind(X \setminus A) \le n \ge 0$ . For every pair  $A_1, A_2$  of closed subsets of A such that  $A_1 \cup A_2 = A$  there exist closed subspaces  $X_1, X_2$  of the space X which satisfy the conditions

(1)  $X = X_1 \cup X_2$ ,  $A_1 \subset X_1$ ,  $A_2 \subset X_2$ ,  $A_1 \cap X_2 = A_1 \cap A_2 = X_1 \cap A_2$ and

(2) 
$$\operatorname{ind}[(X_1 \cap X_2) \setminus (A_1 \cap A_2)] \leq n-1.$$

**Proof.** The sets  $A_1 \ A_2 = A_1 \ (A_1 \cap A_2)$  and  $A_2 \ A_1 = A_2 \ (A_1 \cap A_2)$ are disjoint and closed in the subspace  $X \ (A_1 \cap A_2)$  of the space X. By virtue of the second separation theorem, there exists a partition L in  $X \ (A_1 \cap A_2)$  between  $A_1 \ A_2$  and  $A_2 \ A_1$  such that  $\operatorname{ind}[L \cap (X \setminus A)]$  $\leq n-1$ ; since  $L \cap A = [L \cap (A_1 \ A_2)] \cup [(L \cap (A_2 \ A_1)] \cup [L \cap A_1 \cap A_2] = \emptyset$ , the last inequality can be rewritten as  $\operatorname{ind} L \leq n-1$ .



Fig. 9

Consider a pair of sets  $U, V \subset X \setminus (A_1 \cap A_2)$  which are open in  $X \setminus (A_1 \cap A_2)$  and satisfy the conditions

 $A_1 \setminus A_2 \subset U, \ A_2 \setminus A_1 \subset V, \ U \cap V = \emptyset \text{ and } [X \setminus (A_1 \cap A_2)] \setminus L = U \cup V.$ 

The sets U and V are open in X; we shall verify that their complements  $X_1 = X \setminus V$  and  $X_2 = X \setminus U$  satisfy (1) and (2).

First of all,

$$X_1 \cup X_2 = (X \setminus V) \cup (X \setminus U) = X \setminus (U \cap V) = X.$$

Next,

$$A_1 = (A_1 \setminus A_2) \cup (A_1 \cap A_2) \subset U \cup [X \setminus (U \cup V)] \subset X \setminus V = X_1,$$

and—similarly— $A_2 \subset X_2$ . Then,

$$A_1 \cap A_2 \subset A_1 \cap X_2 = A_1 \setminus U \subset A_1 \setminus (A_1 \setminus A_2) = A_1 \cap A_2,$$

so that  $A_1 \cap A_2 = A_1 \cap X_2$ , and—similarly— $A_1 \cap A_2 = X_1 \cap A_2$ . Finally, since  $X_1 \cap X_2 = X \setminus (U \cup V) = L \cup (A_1 \cap A_2)$ , we have  $[(X_1 \cap X_2) \setminus (A_1 \cap A_2)]$ = L, so that  $\operatorname{ind}[(X_1 \cap X_2) \setminus (A_1 \cap A_2)] \leq n-1$ .  $\square$ 

**1.9.2. Theorem.** If X is a separable metric space and A a closed subspace of X such that  $ind(X \setminus A) \le n \ge 0$ , then for every continuous mapping f:  $A \to S^n$  there exists a continuous extension F:  $X \to S^n$  of f over X.

**Proof.** We shall apply induction with respect to *n*. When n = 0, the mapping *f* takes values in the two-point space  $S^0 = \{-1, 1\}$  and it follows from Lemma 1.9.1, applied to the sets  $A_1 = f^{-1}(1)$  and  $A_2 = f^{-1}(-1)$ , that there exist closed subspaces  $X_1, X_2$  of the space X such that  $X = X_1 \cup \bigcup X_2$ ,  $A_1 \subset X_1, A_2 \subset X_2$  and  $X_1 \cap X_2 = \emptyset$ . The mapping  $F: X \to S^0$  defined by letting

$$F(x) = \begin{cases} 1 & \text{for } x \in X_1, \\ -1 & \text{for } x \in X_2 \end{cases}$$

is then the required continuous extension of f.

Consider now an  $n \ge 1$  and assume that the theorem holds for continuous mappings to the (n-1)-sphere  $S^{n-1}$ . Let  $f: A \to S^n$  be a continuous mapping defined on a closed subspace A of a separable metric space Xsuch that  $\operatorname{ind}(X \setminus A) \le n$ . Denote by  $S_+^n$  and  $S_-^n$  the upper and the lower hemisphere of  $S^n$ , respectively. Let  $A_1 = f^{-1}(S_+^n)$  and  $A_2 = f^{-1}(S_-^n)$ ; since  $S_+^n \cap S_-^n = S^{n-1}$ , the restriction  $g = f|A_1 \cap A_2$  maps  $A_1 \cap A_2$  to  $S^{n-1}$ . Applying Lemma 1.9.1, we obtain closed subspaces  $X_1, X_2 \subset X$  which satisfy conditions (1) and (2). From (2) and the inductive assumption it follows that the mapping  $g: A_1 \cap A_2 \to S^{n-1}$  has a continuous extension  $G: X_1 \cap X_2 \to S^{n-1}$  over the subspace  $X_1 \cap X_2$  of the space X. Since  $A_1 \cap X_2$   $= A_1 \cap A_2 = X_1 \cap A_2$ , the formulas

$$f_1(x) = \begin{cases} f(x) & \text{for } x \in A_1, \\ G(x) & \text{for } x \in X_1 \cap X_2, \end{cases} \quad f_2(x) = \begin{cases} f(x) & \text{for } x \in A_2, \\ G(x) & \text{for } x \in X_1 \cap X_2 \end{cases}$$

define continuous mappings

$$f_1: A_1 \cup (X_1 \cap X_2) \to S_+^n$$
 and  $f_2: A_2 \cup (X_1 \cap X_2) \to S_-^n$ .

The hemispheres  $S_{+}^{n}$  and  $S_{-}^{n}$  being homeomorphic to  $I^{n}$ , it follows from the Tietze-Urysohn theorem that  $f_{1}$  and  $f_{2}$  are continuously extendable to mappings  $F_{1}: X_{1} \to S_{+}^{n}$  and  $F_{2}: X_{2} \to S_{-}^{n}$ . Letting

$$F(x) = \begin{cases} F_1(x) & \text{for } x \in X_1, \\ F_2(x) & \text{for } x \in X_2, \end{cases}$$

we define a continuous extension  $F: X \to S^n$  of the mapping f.  $\Box$ 

**1.9.3. Theorem on extending mappings to spheres.** A separable metric space X satisfies the inequality  $\operatorname{ind} X \leq n \geq 0$  if and only if for every closed subspace A of the space X and each continuous mapping  $f: A \to S^n$  there exists a continuous extension  $F: X \to S^n$  of f over X.

**Proof.** By virtue of Theorem 1.9.2, it suffices to show that extendability of mappings implies the inequality ind  $X \le n$ . We shall apply the theorem on partitions.

Let  $(A_1, B_1), (A_2, B_2), ..., (A_{n+1}, B_{n+1})$  be a sequence of n+1 pairs of disjoint closed subsets of X. Define  $A = \bigcup_{i=1}^{n+1} (A_i \cup B_i)$  and, for i = 1, 2, ......, n+1, consider a continuous function  $f_i: A \to I$  such that

$$f_i(A_i) \subset \{0\}$$
 and  $f_i(B_i) \subset \{1\}$ .

Letting  $f(x) = (f_1(x), f_2(x), \dots, f_{n+1}(x))$  for  $X \in A$ , we define a continuous mapping  $f: A \to S_I^n$  of A to the boundary  $S_I^n$  of the (n+1)-cube  $I^{n+1}$  in  $\mathbb{R}^{n+1}$ . As the space  $S_I^n$  is homeomorphic to  $S^n$ , the mapping f has a continuous extension  $F: X \to S_I^n$  over X. The composition  $F_i: X \to I$  of F and the projection of  $S_I^n$  onto the *i*-th coordinate axis is an extension of  $f_i$  for  $i = 1, 2, \dots, n+1$ , so that the set  $L_i = F_1^{-1}(1/2)$  is a partition between  $A_i$ and  $B_i$ . Since  $\bigcap_{i=1}^{n+1} L_i = \emptyset$ , we have  $\operatorname{ind} X \leq n$  by virtue of the theorem on partitions.  $\Box$  Ch. 1, § 9]

#### Cantor-manifolds

**1.9.4. Remark.** Let us note that in the second part of the above proof only the normality of the space X was applied; hence, we have shown that if a normal space X has the property that for every closed subspace A of X and each continuous mapping  $f: A \to S^n$  there exists a continuous extension  $F: X \to S^n$  of f over X, then for every sequence  $(A_1, B_1)$ ,  $(A_2, B_2), \ldots, (A_{n+1}, B_{n+1})$  of n+1 pairs of disjoint closed subsets of X there exist closed sets  $L_1, L_2, \ldots, L_{n+1}$  such that  $L_i$  is a partition between  $A_i$  and  $B_i$  and  $\bigcap_{i=1}^{n+1} L_i = \emptyset$ .

The characterization of dimension in terms of mappings to spheres will now be applied to establish the Cantor-manifold theorem; another important application of this characterization will be given in Section 1.12 (see Theorem 1.12.4).

**1.9.5. Definition.** A compact metric space X such that  $\operatorname{ind} X = n \ge 1$  is an *n*-dimensional Cantor-manifold if no closed subset L of X satisfying the inequality  $\operatorname{ind} L \le n-2$  separates the space X, i.e., if for every such set the complement  $X \setminus L$  is connected.

Clearly, every one-dimensional metric continuum is a one-dimensional Cantor-manifold, and from Theorem 1.8.13 it follows that for every  $n \ge 1$  the *n*-cube  $I^n$  and the *n*-sphere  $S^n$  are *n*-dimensional Cantor-manifolds. On the other hand, for every  $n \ge 2$  the union of two copies of the *n*-cube  $I^n$  with exactly one point in common is an example of an *n*-dimensional compact metric space which is not a Cantor-manifold. Let us observe that if X is an *n*-dimensional Cantor-manifold, then the dimension of X at each point  $x \in X$  (see Problem 1.1.B) is equal to *n*, but—as shown by the last example—a compact metric space of dimension *n* at each point need not be a Cantor-manifold.

As we have already observed (see remarks to Problems 1.4.F(b) and 1.5.C), for every  $n \ge 1$  there exist separable metric spaces of dimension n which contain no compact subspace of dimension n, and, *a fortiori*, no *n*-dimensional Cantor-manifold. However, every compact metric space of dimension  $n \ge 1$  does contain an *n*-dimensional Cantor-manifold. The proof of this important theorem is preceded by two lemmas.

Let us recall that continuous mappings  $f_0, f_1$  of a topological space X to a topological space Y are *homotopic* if there exists a continuous mapping  $F: X \times I \to Y$  such that  $F(x, i) = f_i(x)$  for i = 0, 1 and  $x \in X$ ; the mapping F is called a *homotopy* between  $f_0$  and  $f_1$ . **1.9.6. Lemma.** Let  $f, g: X \to S^n$  be continuous mappings of a separable metric space X to the n-sphere  $S^n$ . If the set

$$D(f,g) = \{x \in X \colon f(x) \neq g(x)\}$$

satisfies the inequality  $\operatorname{ind} D(f, g) \leq n-1$ , then the mappings f and g are homotopic.

**Proof.** Consider the space  $Y = X \times I$ , the closed set  $A = (X \times \{0, 1\}) \cup \cup (X \setminus D(f, g)) \times I \subset Y$  and the continuous mapping  $h: A \to S^n$  defined by letting

and

$$h(x, 0) = f(x), \quad h(x, 1) = g(x) \text{ for } x \in X$$

h(x, t) = f(x) = g(x) for  $x \in X \setminus D(f, g)$  and  $t \in I$ .

As  $Y \setminus A \subset D(f, g) \times I$ , we have  $\operatorname{ind}(Y \setminus A) \leq n$ , and by virtue of Theorem 1.9.2 there exists a continuous extension  $H: Y \to S^n$  of the mapping  $h: A \to S^n$ ; clearly, H is a homotopy between f and g.  $\Box$ 

The next lemma describes an important property of mappings to spheres; it is called the *Borsuk homotopy extension theorem*.

**1.9.7. Lemma.** Let X be a topological space such that the Cartesian product  $X \times I$  is normal (in particular, a metric or a compact space), A a closed subspace of X, and f, g:  $A \to S^n$  a pair of homotopic continuous mappings of A to the n-sphere  $S^n$ . If f is continuously extendable over X, then g is also continuously extendable over X; moreover, for every extension  $F: X \to S^n$  of f there exists an extension  $G: X \to S^n$  of g such that F and G are homotopic.

**Proof.** Let  $h: A \times I \to S^n$  be a homotopy between f and g. The continuous mapping  $h': (X \times \{0\}) \cup (A \times I) \to S^n$  defined by letting

$$h'(x, 0) = F(x)$$
 for  $x \in X$ 

and

$$h'(x, t) = h(x, t)$$
 for  $x \in A$  and  $t \in I$ 

can be extended to a continuous mapping  $H': U \to S^n$  defined on an open set  $U \subset X \times I$  which contains  $(X \times \{0\}) \cup (A \times I)$ . It follows from the compactness of I and the definition of the Cartesian product topology that every point  $a \in A$  has a neighbourhood  $V_a \subset X$  such that  $V_a \times I \subset U$ ; the union  $V = \bigcup_{a \in A} V_a$  is an open subset of X such that  $A \times I \subset V \times I \subset U$ . The space X, being homeomorphic to the closed subspace  $X \times \{0\}$  of the normal space  $X \times I$ , is itself normal, so that by Urysohn's lemma there Ch. 1, § 9]

Cantor-manifolds

exists a continuous function  $u: X \to I$  such that

$$u(X \setminus V) \subset \{0\}$$
 and  $u(A) \subset \{1\}$ .

One readily verifies that the formula

 $H(x, t) = H'(x, t \cdot u(x)) \quad \text{for } (x, t) \in X \times I$ 

defines a continuous mapping  $H: X \times I \to S^n$  such that H(x, 0) = F(x)for  $x \in X$  and H(x, 1) = g(x) for  $x \in A$ . Hence, the mapping  $G: X \to S^n$ defined by letting G(x) = H(x, 1) for  $x \in X$  is a continuous extension of g, and H is a homotopy between F and G.  $\square$ 

**1.9.8. The Cantor-manifold theorem.** Every compact metric space X such that  $ind X = n \ge 1$  contains an n-dimensional Cantor-manifold.

**Proof.** By virtue of Theorem 1.9.3, there exists a closed subspace A of the space X and a continuous mapping  $f: A \to S^{n-1}$  which cannot be continuously extended over X. Denote by  $\mathscr{C}$  the family consisting of all closed sets  $C \subset X$  such that the mapping f cannot be continuously extended over  $A \cup C$ , and define an order  $\leq$  in the family  $\mathscr{C}$  by letting  $C_1 \leq C_2$  whenever  $C_2 \subset C_1$ ; since  $X \in \mathscr{C}$ , the family  $\mathscr{C}$  is non-empty.

Now, consider a subfamily  $\mathscr{C}_0$  of  $\mathscr{C}$  which is linearly ordered by  $\leq$ . We shall show that the closed set  $C_0 = \bigcap \mathscr{C}_0$  is a member of  $\mathscr{C}$ . Assume the contrary, i.e., that the mapping f is continuously extendable over  $A \cup C_0$ . There exists then an open set  $U \subset X$  containing  $A \cup C_0$  such that f is continuously extendable over U. Since  $C_0 = \bigcap \mathscr{C}_0 \subset U$ , from the compactness of X follows the existence of a finite number of sets  $C_1, C_2, ...$ ...,  $C_k \in \mathscr{C}_0$  such that  $C_1 \cap C_2 \cap \ldots \cap C_k \subset U$ . The family  $\mathscr{C}_0$  being linearly ordered by  $\leq$ , there exists an  $i_0 \leq k$  such that  $C_{i_0} \subset C_t$  for every  $i \leq k$ . Hence,  $C_{i_0} \subset U$  and the function f is continuously extendable over  $A \cup$  $\cup C_{t_0} \subset U$ ; but this is impossible, because  $C_{t_0}$  is a member of  $\mathscr{C}$ . The contradiction shows that  $C_0 \in \mathscr{C}$ ; clearly  $C \leq C_0$  for every  $C \in \mathscr{C}_0$ . Applying the Kuratowski-Zorn lemma, we obtain a maximal set  $M \in \mathscr{C}$ , i.e., a closed set  $M \subset X$  such that the mapping f cannot be continuously extended over  $A \cup M$ , and yet for every proper closed subset M' of M it is continuously extendable over  $A \cup M'$ . We shall show that M is an n-dimensional Cantor-manifold.

As ind  $M \leq \text{ind} X = n$ , it suffices to show that if  $M = M_1 \cup M_2$ , where  $M_1$  and  $M_2$  are proper closed subsets of M, then  $\text{ind}(M_1 \cap M_2) \geq n-1$ . Assume that there exist proper closed subsets  $M_1, M_2$  of the set M such that  $M = M_1 \cup M_2$  and  $\text{ind}(M_1 \cap M_2) \leq n-2$ ; consider the sets  $A_1 = A \cup M_1$  and  $A_2 = A \cup M_2$ . For i = 1, 2 the mapping f can be extended to a continuous mapping  $f_i: A_i \to S^{n-1}$ . The restrictions  $f_1 | B$  and  $f_2 | B$  of  $f_1$  and  $f_2$  to the set  $B = A \cup (M_1 \cap M_2)$  differ on a subset of  $M_1 \cap M_2$ , so that by virtue of Lemma 1.9.6 they are homotopic. Since  $f_2 | B$  is continuously extendable over  $A \cup M_2$ , it follows from Lemma 1.9.7 that  $f_1 | B$ is also extendable to a continuous mapping  $f'_1: A \cup M_2 \to S^{n-1}$ . From the equality  $(A \cup M_1) \cap (A \cup M_2) = B$  it follows that letting

$$F(x) = \begin{cases} f_1(x) & \text{for } x \in A \cup M_1, \\ f_1'(x) & \text{for } x \in A \cup M_2 \end{cases}$$

we define a continuous mapping  $F: A \cup M \to S^{n-1}$  which extends the mapping f. This contradiction concludes the proof.  $\square$ 

**1.9.9. Corollary.** Every n-dimensional compact metric space X has an n-dimensional component.

The characterization of dimension in terms of mappings to spheres leads, via Hopf's extension theorem, to the cohomological characterization of dimension. We shall describe briefly this process.

It is a well-known fact that the Čech cohomology groups with coefficients in the group of integers, based on all open covers, satisfy the Eilenberg-Steenrod axioms. In particular, for every topological space X and every closed subset A of the space X the cohomology sequence

$$H^{0}(X, A) \xrightarrow{j^{*}} \dots \xrightarrow{\delta} H^{n}(X, A) \xrightarrow{j^{*}} H^{n}(X) \xrightarrow{i^{*}} H^{n}(A) \xrightarrow{\delta} H^{n}(X, A) \xrightarrow{j^{*}} \dots$$

of the pair (X, A) is an exact sequence, and for every continuous mapping  $f: X \to S^n$  an element  $f^*(s^n)$  of the cohomology group  $H^n(X)$  is defined, the image of a fixed generator of the group  $H^n(S^n)$  under the induced homomorphism  $f^*: H^n(S^n) \to H^n(X)$ . Hopf's extension theorem, which was mentioned above, states that

(H) A continuous mapping  $f: A \to S^n$  defined on a closed subspace A of a paracompact space X such that  $\dim X \leq n+1$  is continuously extendable over X if and only if  $f^*(s^n) \in i^*(H^n(X))$ .

It follows from the definition of cohomology groups that if X is a paracompact space such that dim  $X \leq n$ , then  $H^{n+1}(X, A) = 0$  for every closed set  $A \subset X$  (cf. Proposition 3.2.2). On the other hand, if a finite-dimensional separable metric space X satisfies the inequality dim X > n, then by virtue

of Theorem 1.5.1 there exists a closed subspace X' of X such that  $\dim X'$ = n+1, and by virtue of Theorem 1.9.3 there exists a closed subspace A of X' and a continuous mapping  $f: A \to S^n$  which cannot be continuously extended over X'; from Hopf's extension theorem and the elementary properties of cohomology groups it follows that  $f^*(s^n) \notin i^*(H^n(X))$ , so that  $H^{n+1}(X, A) \neq 0$  by the exactness of the cohomology sequence of the pair (X, A). Thus we obtain the cohomological characterization of dimension, which states that a finite-dimensional separable metric space X satisfies the inequality dim  $X \leq n \geq 0$  if and only if  $H^{n+i}(X, A) = 0$  for every closed set  $A \subset X$  and for i = 1, 2, ... This characterization remains valid in the class of all paracompact spaces (cf. Theorem 3.2.10). Let us observe that for a metric space X it would be enough to consider the group  $H^{n+1}(X, A)$ , but for arbitrary paracompact spaces it is necessary to take into account all groups  $H^{n+1}(X, A)$ , because no counterpart of Theorem 1.5.1 holds in paracompact spaces for the covering dimension dim. Let us also note that it is an open question whether the assumption of finitedimensionality of X in the cohomological characterization of dimension is essential.

The cohomological characterization of dimension was the point of departure for cohomological dimension theory, a section of dimension theory on the border-line of point-set and algebraic topology, which studies the notion of the cohomological dimension with respect to an abelian group. For a fixed abelian group  $G \neq 0$ , to every topological space X one assigns the cohomological dimension of X with respect to G, denoted by  $\dim_G X$ , which is an integer larger than or equal to -1 or the "infinite number"  $\infty$ ; the definition of the dimension function  $\dim_G$  consists in the following conditions:

- (CD1)  $\dim_G X = -1$  if and only if  $X = \emptyset$ ;
- (CD2) dim<sub>G</sub>  $X \leq n$ , where n = 0, 1, ..., if  $H^{n+i}(X, A; G) = 0$  for every closed set  $A \subset X$  and for i = 1, 2, ...;
- (CD3)  $\dim_G X = n$  if  $\dim_G X \leq n$  and  $\dim_G X > n-1$ ;
- (CD4)  $\dim_G X = \infty$  if  $\dim_G X > n$  for n = -1, 0, 1, ...

One proves that the functions  $\dim_G$ , although distinct for different groups G, have many properties of dimension. In particular, for every abelian group  $G \neq 0$  and every natural number n we have  $\dim_G R^n = \dim_G I^n$  $= \dim_G S^n = n$ . One also proves that, under suitable assumptions on the space X and the group G, cohomological dimensions satisfy the counterparts of the subspace, sum, Cartesian product and compactification theorems. In the proofs of these theorems methods of algebraic topology and homological algebra are largely applied.

To conclude, let us note that the dimension of compact spaces can also be characterized in terms of Čech homology groups with coefficients in the group  $R_1$  of real numbers modulo 1. One proves that a finite-dimensional compact space X satisfies the inequality dim $X \le n \ge 0$  if and only if  $H_{n+i}(X, A; R_1) = 0$  for every closed set  $A \subset X$  and for i = 1, 2, ...In the proof, the exactness of the homology sequence and the homological counterpart of Hopf's extension theorem are used; since compactness is crucial for the validity of both these results, when passing to larger classes of spaces one has to replace homology with cohomology.

## Historical and bibliographic notes

Theorem 1.9.2 was established by Hurewicz in [1935a]. The same paper contains Theorem 1.9.3 for compact metric spaces. It is a restatement of an earlier result of Alexandroff (see Problem 1.9.A); Hurewicz's contribution was to find a clever formulation and a simple proof (cf. Problem 1.9.C). The extension of Theorem 1.9.3 to separable metric spaces was given by Hurewicz and Wallman in [1941]. The notion of a Cantor-manifold was introduced by Urysohn in [1925]; Theorem 1.9.8 was proved independently by Hurewicz and Menger in [1928] and by Tumarkin in [1928]. The original proofs were much more involved than the one presented here, discovered by Hurewicz in [1937] and by Kuratowski in [1937] (a similar proof was given by Freudenthal in [1937]). Homological dimension theory was originated by Alexandroff in [1932]. Complete proofs of homological and cohomological characterizations of dimension in the realm of compact metric spaces can be found in Hurewicz and Wallman's book [1941]. A comprehensive exposition of cohomological dimension theory was developed by Kuz'minov in [1968] and by Kodama in [1970].

## Problems

**1.9.A** (Alexandroff [1932] for compact metric spaces). A continuous mapping  $f: X \to B^{n+1}$  of a topological space X to the (n+1)-ball in  $\mathbb{R}^{n+1}$  is essential if there is no continuous mapping  $g: X \to B^{n+1}$  such that

#### Problems

 $g|f^{-1}(S^n) = f|f^{-1}(S^n)$  and  $B^{n+1} g(X) \neq \emptyset$ ; essential mappings with values in sets homeomorphic to balls are defined in a similar way.

Show that a separable metric space X satisfies the inequality  $\operatorname{ind} X \leq n \geq 0$  if and only if no continuous mapping  $f: X \to B^{n+1}$  is essential.

**1.9.B** (Hurewicz and Wallman [1941]). A point  $y \in f(X)$  is an unstable value of a continuous mapping  $f: X \to Y$  of a topological space X to a metric space Y if for every positive number  $\varepsilon$  there exists a continuous mapping  $g: X \to Y$  such that  $\varrho(f(x), g(x)) < \varepsilon$  for every  $x \in X$  and  $y \notin g(X)$ ; otherwise,  $y \in f(X)$  is a stable value of f.

(a) Show that  $y \in f(X)$  is an unstable value of a continuous mapping  $f: X \to I^n$  if and only if for every neighbourhood U of the point y there exists a continuous mapping  $g: X \to I^n$  such that g(x) = f(x) whenever  $f(x) \notin U$ ,  $g(x) \in U$  whenever  $f(x) \in U$ , and  $y \notin g(X)$ .

(b) Show that a separable metric space X satisfies the inequality  $\operatorname{ind} X \leq n \geq 0$  if and only if for every continuous mapping  $f: X \to I^{n+1}$  all points in f(X) are unstable values of f. Observe that instead of  $I^{n+1}$  one can consider  $S^{n+1}$  or  $R^{n+1}$ .

**1.9.C** (Hurewicz [1935a]). (a) Let Y be a complete metric space and Z a totally bounded metric space. Prove that for every open set  $U \subset Y \times Z$  which is dense in  $Y \times Z$  there exists a set  $A \subset Y$  which is dense in Y and such that for every  $a \in A$  the set  $\{z \in Z : (a, z) \in U\}$  is dense in Z.

(b) Prove that if X is a compact metric space and the set  $\{f \in (\mathbb{R}^{n+1})^X : f(x) \neq (0, 0, ..., 0) \text{ for every } x \in X\}$  is dense in the function space  $(\mathbb{R}^{n+1})^X$ , then  $\operatorname{ind} X \leq n$  (see the beginning of Section 1.11).

*Hint.* Apply induction with respect to n; use the equality  $(R^{n+1})^X = (R^n)^X \times R^X$  and part (a).

(c) Apply part (b) to show that if for every closed subspace A of a compact metric space X and each continuous mapping  $f: A \to S^n$  there exists a continuous extension  $F: X \to S^n$  of f over X, then  $ind X \leq n$ .

**1.9.D** (Borsuk [1937]; for a compact metric space X, Eilenberg [1936]). Prove that for every continuous mapping  $f: A \to S^k$  defined on a closed subspace A of a separable metric space X such that  $ind(X \setminus A) \leq n$ , where  $0 \leq k \leq n$ , there exists a closed subspace B of the space X such that  $A \cap B = \emptyset$ , ind  $B \leq n-k-1$ , and the mapping f has a continuous extension F:  $X \setminus B \to S^k$  over  $X \setminus B$ .

*Hint*. Apply induction with respect to k+n; modify the proof of Theorem 1.9.2.

1.9.E. (a) Let  $p: R \to S^1$  be the continuous mapping of the real line to the circle defined by letting  $p(t) = e^{2\pi i t}$ . Applying the fact that for every continuous mapping  $g: X \to S^1$  of a metric space X to  $S^1$  which is homotopic to a constant mapping there exists a continuous mapping  $\tilde{g}:$  $X \to R$  such that  $g = p\tilde{g}$  (see Spanier [1966], p. 67), show that for every one-dimensional compact metric space Z there exists a closed set  $M \subset Z$ such that the quotient space Z/M can be mapped onto  $S^1$  by a mapping which is not homotopic to the constant mapping.

*Hint.* Let M be a closed subspace of the space Z with the property that there exists a continuous mapping  $f: M \to S^0$  which cannot be continuously extended over Z, and let  $F: Z \to I$  be a continuous mapping such that F(x) = f(x) for every  $x \in M$ . Consider the quotient space Z/M and the continuous mapping  $g: Z/M \to I/S^0 = S^1$  induced by F.

(b) (Hurewicz [1935]) Applying the fact that for every *n*-dimensional compact metric space Z, where n = 1, 2, ..., there exists a closed set  $M \subset Z$  such that the quotient space Z/M can be mapped onto  $S^n$  by a mapping which is not homotopic to the constant mapping (see Dowker [1947], p. 235), show that for every compact metric space X and every one-dimensional separable metric space Y we have  $ind(X \times Y) = indX + 1 = indX + indY$ .

*Hint.* Let  $\operatorname{ind} X = n$ ; reduce the problem to the special case where there exists a continuous mapping  $g: X \to S^n$  which is not homotopic to the constant mapping. Then consider a pair A, B of disjoint closed subsets of Y such that the empty set is not a partition between A and Band, assuming that  $\operatorname{ind}(X \times Y) \leq n$ , extend the mapping  $f: X \times (A \cup B) \to S^n$ defined by letting

f(x, y) = (1, 0, 0, ..., 0) for  $y \in A$  and f(x, y) = g(x) for  $y \in B$ 

to a continuous mapping  $F: X \times Y \to S^n$ ; examine the set of those  $y \in Y$  for which the restriction  $F|(X \times \{y\})$  is homotopic to the constant mapping.

1.9.F (Kuratowski and Otto [1939]). Let  $\{X_s\}_{s\in S}$  be a family of *n*-dimensional Cantor-manifolds contained in an *n*-dimensional compact metric space X. Prove that if there exists an  $s_0 \in S$  such that the inequality  $\operatorname{ind}(X_s \cap X_{s_0}) \ge n-1$  holds for all  $s \in S$ , then the subspace  $\bigcup_{s\in S} X_s$  of X is an *n*-dimensional Cantor-manifold.

*Remark.* As proved by Borsuk in [1951], the Cartesian product of two Cantor-manifolds is not necessarily a Cantor-manifold.

**1.9.G.** (a) (Anderson and Keisler [1967]) Let X be an open subspace of an *n*-dimensional Cantor-manifold. Prove that if a subset M of X meets every continuum  $C \subset X$  which has cardinality larger than one, then ind  $M \ge n-1$  (cf. Theorem 1.8.19).

*Hint.* Assuming that  $\operatorname{ind} M \leq n-2$ , define inductively a decreasing sequence  $X_1 \supset X_2 \supset \ldots \supset X_{n-1}$  of subsets of X such that  $X_k$  is an (n-k)-dimensional Cantor-manifold and  $\operatorname{ind}(M \cap X_k) \leq n-k-2$  for  $k = 1, 2, \ldots$   $\ldots, n-1$ .

(b) Give an example of a two-dimensional Cantor-manifold of which a one-point subset is a cut.

1.9.H. (a) (Alexandroff [1932]) Show that every *n*-dimensional Cantormanifold contained in an *n*-dimensional compact metric space X can be enlarged to a maximal Cantor-manifold contained in X, i.e., to a Cantor manifold which is not a proper subset of another Cantor manifold contained in X; maximal *n*-dimensional Cantor-manifolds contained in an *n*-dimensional compact metric space X are called *dimensional components* of X. Check that dimensional components of a one-dimensional compact metric space coincide with components of the space X which have cardinality larger than one. Note that the union of all dimensional components of a compact metric space X is not necessarily equal to X. Observe that the intersection of two dimensional components of an *n*-dimensional compact metric space has dimension  $\leq n-2$ .

(b) (Mazurkiewicz [1933]) Prove that if A is a dimensional component of an *n*-dimensional compact metric space, and B is the union of all the remaining dimensional components of X, then  $ind(A \cap B) \leq n-2$ .

Hint. Define a decreasing transfinite sequence  $X = F_1 \supset F_2 \supset ... \supset F_{\alpha}$  $\supset ..., \alpha < \omega_1$  of closed sets containing A such that if  $F_{\alpha} \setminus A \neq \emptyset$ , then  $F_{\alpha} = F_{\alpha+1} \cup \overline{(F_{\alpha} \setminus F_{\alpha+1})}$ , where  $\operatorname{ind}[F_{\alpha+1} \cap \overline{(F_{\alpha} \setminus F_{\alpha+1})}] \leq n-2$ , and  $F_{\lambda}$  $= \bigcap_{\alpha < \lambda} F_{\alpha}$  for every limit number  $\lambda < \omega_1$ . Applying the fact that there exists an  $\alpha_0 < \omega_1$  such that  $F_{\alpha} = F_{\alpha_0}$  for every  $\alpha \ge \alpha_0$  (see [GT], Problem 3.12.7(b)), show that  $F_{\alpha_0} = A$  and  $X = A \cup \bigcup_{\alpha < \alpha_0} (F_{\alpha} \setminus F_{\alpha+1})$ .

## 1.10. Characterization of dimension in terms of mappings to polyhedra

The characterization of dimension which is the object of the present section will be formulated in terms of mappings with "arbitrarily small fibres" to polyhedra of geometric dimension  $\leq n$ .

We begin by recalling the notions of a simplex, a simplicial complex, and a polyhedron.

Let  $p = \{p_i\}$  and  $q = \{q_i\}$  be points of Euclidean *m*-space  $R^m$ . The sum p+q of the points p and q and the product  $\lambda p$  of the point p by the real number  $\lambda$  are defined by the formulas

$$p+q = r = \{r_i\}, \text{ where } r_i = p_i + q_i,$$

and

$$\lambda p = s = \{s_i\}, \text{ where } s_i = \lambda p_i.$$

The point of  $\mathbb{R}^m$  which has all coordinates equal to zero, i.e., the origin of  $\mathbb{R}^m$ , will be denoted by the symbol 0; the distance  $\varrho(0, x)$  will be denoted by ||x||. One can readily verify that

$$||p-q|| = \varrho(p,q), \quad ||\lambda p|| = |\lambda| \cdot ||p|| \text{ and } ||p+q|| \le ||p|| + ||q||;$$

the last inequality is a reformulation of the triangle inequality in  $\mathbb{R}^m$ .

A finite system of points  $p_0, p_1, ..., p_k \in \mathbb{R}^m$  is linearly independent if for each sequence  $\lambda_0, \lambda_1, ..., \lambda_k$  of k+1 real numbers the conditions

$$\lambda_0 p_0 + \lambda_1 p_1 + \dots + \lambda_k p_k = 0$$
 and  $\lambda_0 + \lambda_1 + \dots + \lambda_k = 0$ 

imply that  $\lambda_i = 0$  for i = 0, 1, ..., k.

Let  $p_0, p_1, ..., p_n$  be a linearly independent system of n+1 points in  $\mathbb{R}^m$ ; the subset of  $\mathbb{R}^m$  consisting of all points

(1) 
$$p = \lambda_0 p_0 + \lambda_1 p_1 + \ldots + \lambda_n p_n,$$

where

(2) 
$$\lambda_0 + \lambda_1 + \ldots + \lambda_n = 1$$
 and  $\lambda_i \ge 0$  for  $i = 0, 1, \ldots, n$ ,

is called an *n-simplex* spanned by the points  $p_0, p_1, \ldots, p_n$  and is denoted by  $p_0 p_1 \ldots p_n$ . Clearly, the simplex  $p_0 p_1 \ldots p_n$  does not depend on the ordering of points  $p_0, p_1, \ldots, p_n$ , it depends on the set  $\{p_0, p_1, \ldots, p_n\}$ only. One proves that the simplex  $p_0 p_1 \ldots p_n$  determines the points  $p_0, p_1, \ldots, p_n$  and can be characterized as the smallest convex subset of  $R^m$  which contains these points; one also proves that the diameter of the simplex  $p_0 p_1 \ldots p_n$  is equal to the diameter of the set  $\{p_0, p_1, \ldots, p_n\}$ (see Problem 1.10.A).

Consider an *n*-simplex  $p_0p_1 \dots p_n \subset \mathbb{R}^m$ ; for each choice of k+1 distinct non-negative integers  $i_0, i_1, \dots, i_k$  not larger than *n*, where  $0 \leq k \leq n$ , the points  $p_{i_0}, p_{i_1}, \dots, p_{i_k}$  form a linearly independent system, so that the *k*-simplex  $p_{i_0}p_{i_1} \dots p_{i_k}$  is well defined. Every simplex of this form is called a *k*-face of the simplex  $p_0p_1 \dots p_n$ ; 0-faces  $p_0, p_1, \dots, p_m$  are also called vertices of the simplex  $p_0p_1 \dots p_n$ . One easily sees that the *k*-face  $p_{i_0}p_{i_1} \dots p_{i_k}$ 

100

consists of all points of form (1) satisfying (2) and such that

 $\lambda_i = 0$  whenever  $i \neq i_j$  for j = 0, 1, ..., k.

To denote the fact that  $S_0 = p_{i_0}p_{i_1} \dots p_{i_k}$  is a face of  $S = p_0p_1 \dots p_n$  we write symbolically  $S_0 \leq S$ . For each *n*-simplex S the union of all k-faces of S with k < n is called the *boundary* of the simplex S; the complement of the boundary is called the *interior* of the simplex S.

It follows from the linear independence of the set of vertices, that every point p of the simplex spanned by the points  $p_0, p_1, ..., p_n \in \mathbb{R}^m$  can be represented in the form (1), under conditions (2), in a unique way. The coefficients  $\lambda_0, \lambda_1, ..., \lambda_n$  in (1) are called the *barycentric coordinates* of the point p; they will also be denoted by  $\lambda_0(p), \lambda_1(p), ..., \lambda_n(p)$ .

One readily checks that every simplex  $S = p_0 p_1 \dots p_n \subset \mathbb{R}^m$  is a compact subspace of  $\mathbb{R}^m$  and that the barycentric coordinates  $\lambda_0, \lambda_1, \dots, \lambda_n$ are continuous functions from S to I. This implies, in particular, that any two *n*-simplexes are homeomorphic. Hence (see Problems 1.8.A and 1.10.A(b)), every *n*-simplex is homeomorphic to the *n*-cube  $I^n$ . It follows from Corollary 1.8.3, that for every *n*-simplex S we have ind S = Ind S = dim S = n.

A simplicial complex, or, briefly, a complex, is an arbitrary finite family  $\mathscr{K}$  of simplexes in a Euclidean space such that if  $S \in \mathscr{K}$  and  $S_0 \leq S$ then  $S_0 \in \mathscr{K}$ , and if  $S_1, S_2 \in \mathscr{K}$  then the intersection  $S_1 \cap S_2$  either is empty or is a face of both  $S_1$  and  $S_2$ ; all 0-simplexes that belong to  $\mathscr{K}$  are called *vertices* of the complex  $\mathscr{K}$ . Every subfamily  $\mathscr{K}_0$  of a complex  $\mathscr{K}$  which itself is a complex, i.e. which together with a simplex  $S \in \mathscr{K}$  contains all faces of S, is called a *subcomplex* of  $\mathscr{K}$ .

Let  $\mathscr{K}$  be a simplicial complex consisting of simplexes in Euclidean *m*-space  $\mathbb{R}^m$ . The union  $|\mathscr{K}| = \bigcup \{S: S \in \mathscr{K}\} \subset \mathbb{R}^m$  is the *underlying polyhedron* of the complex  $\mathscr{K}$ ; it is a compact subspace of  $\mathbb{R}^m$ . By a *polyhedron* we mean a subspace of a Euclidean space which is the underlying polyhedron  $|\mathscr{K}|$  of a simplicial complex  $\mathscr{K}$ ; clearly, the representation of a polyhedron as the underlying polyhedron of a complex is not unique. One can prove (see Problem 1.10.B) that for every non-empty polyhedron  $K = |\mathscr{K}|$ , the largest integer *n* such that the complex  $\mathscr{K}$  contains *n*-simplexes does not depend on the complex  $\mathscr{K}$  but on the polyhedron *K* only. The number *n* is called the *geometric dimension* of the polyhedron *K*; the geometric dimension of an empty polyhedron is equal to -1. From the sum theorem and the fact that an *n*-simplex is an *n*-dimensional space it follows that the geometric dimension of a polyhedron coincides with its topological dimensions ind, Ind and dim. Let us note that from the same two premises it follows that the integer n discussed above depends on the polyhedron K only. However, since one of the aspects of the characterization of dimension in terms of mappings to polyhedra is that it reduces the topological notion of dimension to the elementary notion of geometric dimension, it is not devoid of importance that the correctness of the definition of geometric dimension can be checked in an elementary way. In the sequel we shall abbreviate "geometric dimension" to "dimension" and call a polyhedron whose geometric dimension equals n an n-dimensional polyhedron.

Every point p of the underlying polyhedron  $|\mathscr{K}| = K$  of a simplicial complex  $\mathscr{K}$  with vertices  $p_0, p_1, \ldots, p_k$  can be represented in the form

(3) 
$$p = \lambda_0 p_0 + \lambda_1 p_1 + \ldots + \lambda_k p_k,$$

where  $\lambda_0 + \lambda_1 + \ldots + \lambda_k = 1$  and  $\lambda_i \ge 0$  for  $i = 0, 1, \ldots, k$ ; moreover, if  $p \in p_{i_0} p_{i_1} \ldots p_{i_1} \in \mathcal{K}$ , then  $\lambda_i = 0$  whenever  $i \ne i_j$  for  $j = 0, 1, \ldots, l$ . It follows from the definition of a simplicial complex that the above representation is unique, so that the coefficients in (3) can be written as  $\lambda_0(p), \lambda_1(p), \ldots, \lambda_k(p)$ ; by the continuity of barycentric coordinates,  $\lambda_0, \lambda_1, \ldots, \lambda_k$  are continuous functions from K to I. For every vertex  $p_i$  of the complex  $\mathcal{K}$  the *star* of  $p_i$  is a subset of the underlying polyhedron defined by

$$\operatorname{St}_{\mathscr{K}}(p_i) = |\mathscr{K}| \setminus \bigcup \{ S \in \mathscr{K} \colon p_i \notin S \};$$

one readily checks that  $\operatorname{St}_{\mathscr{K}}(p_i) = \{p \in |\mathscr{K}| : \lambda_i(p) > 0\}$ , so that the stars of vertices of  $\mathscr{K}$  are open subsets of  $|\mathscr{K}|$ .

We are now going to prepare tools that will be applied later in this section to prove theorems on mappings to polyhedra and in the next one to prove general embedding and universal space theorems.

**1.10.1. Definition.** A finite system of points  $p_1, p_2, ..., p_k \in \mathbb{R}^m$  is in general position if for each sequence  $i_0 < i_1 < ... < i_l \leq k$  of l+1 natural numbers, where  $l \leq m$ , the system  $p_{i_0}, p_{i_1}, ..., p_{i_l}$  is linearly independent.

Since no system of m+2 points in  $\mathbb{R}^m$  is linearly independent, general position means the minimum of bonds between points in a system.

**1.10.2. Theorem.** For every finite system of points  $q_1, q_2, ..., q_k \in \mathbb{R}^m$  and every positive number  $\alpha$  there exists a system of points  $p_1, p_2, ..., p_k \in \mathbb{R}^m$  in general position such that  $\varrho(p_i, q_i) < \alpha$  for i = 1, 2, ..., k. If, moreover, a system of points  $r_1, r_2, ..., r_l \in \mathbb{R}^m$  in general position is given, one can

chose the points  $p_1, p_2, ..., p_k$  in such a way that the whole system  $r_1, r_2, ...$ ...,  $r_1, p_1, p_2, ..., p_k$  is in general position.

**Proof.** The points  $p_1, p_2, ..., p_k$  will be defined by induction. Let  $p_1 = q_1$ ; assume that for  $a_j \leq k$  the points  $p_1, p_2, ..., p_{j-1}$  in general position are defined in such a way that  $\varrho(p_i, q_i) < \alpha$  for i = 1, 2, ..., j-1. For every system of points  $p_{i_0}, p_{i_1}, ..., p_{i_n}$ , where  $i_0 < i_1 < ... < i_n \leq j-1$  and  $n \leq m-1$ , the linear *n*-variety determined by these points is nowhere dense in  $\mathbb{R}^m$ ; hence the union of all such linear varieties is also nowhere dense in  $\mathbb{R}^m$  and there exists a point  $p_j$  outside this union such that  $\varrho(p_j, q_j)$  $< \alpha$ . Clearly the system of points  $p_1, p_2, ..., p_j$  is in general position; thus the first part of the theorem is proved. The proof of the second part is much the same, only—when defining the point  $p_j$ —one has to consider all systems of  $n \leq m-1$  points in the set  $\{r_1, r_2, ..., r_1, p_1, p_2, ..., p_{j-1}\}$ .  $\Box$ 

**1.10.3. Definition.** Let X be a topological space and  $\mathscr{U} = \{U_i\}_{i=1}^k$  a finite open cover of X. By a *nerve* of the cover  $\mathscr{U}$  we understand an arbitrary simplicial complex  $\mathscr{N}(\mathscr{U})$  which has the property that its vertices can be arranged into a sequence  $p_1, p_2, \ldots, p_k$  in such a way that

 $p_{i_0}p_{i_1}\dots p_{i_m} \in \mathcal{N}(\mathcal{U})$  if and only if  $U_{i_0} \cap U_{i_1} \cap \dots \cap U_{i_m} \neq \emptyset$ .

When discussing a nerve  $\mathcal{N}(\mathcal{U})$  of a finite open cover  $\mathcal{U} = \{U_i\}_{i=1}^k$ , we shall always assume that the vertices  $p_1, p_2, \ldots, p_k$  of  $\mathcal{N}(\mathcal{U})$  are arranged in such a way that the above equivalence holds. The underlying polyhedron of the nerve  $\mathcal{N}(\mathcal{U})$  will be denoted by  $N(\mathcal{U})$ . One readily sees that if  $\operatorname{ord} \mathcal{U} \leq n$ , then the dimension of the polyhedron  $N(\mathcal{U})$  is not larger than *n*. Note that every finite open cover has a nerve, although not uniquely determined (cf. Problem 1.10.C). Indeed, for a given open cover  $\{U_i\}_{i=1}^k$  one can, for example, consider an arbitrary linearly independent system  $p_1, p_2, \ldots, p_k$  of k points in  $\mathbb{R}^{k-1}$  and define  $\mathcal{N}(\mathcal{U})$  as the simplicial complex consisting of all faces  $p_{i_0}p_{i_1}\ldots p_{i_m}$  of the simplex  $p_1p_2\ldots p_k$  such that  $U_{i_0} \cap U_{i_1} \cap \ldots \cap U_{i_m} \neq \emptyset$ . It turns out, however, that nerves can be defined in a more economical way.

**1.10.4. Theorem.** Let  $\mathcal{U} = \{U_i\}_{i=1}^k$  be a finite open cover of a topological space X. If  $\operatorname{ord} \mathcal{U} \leq n$ , then there exists a nerve  $\mathcal{N}(\mathcal{U})$  of the cover  $\mathcal{U}$  consisting of simplexes contained in Euclidean (2n+1)-space  $\mathbb{R}^{2n+1}$ . If, moreover, a linear n-variety  $H \subset \mathbb{R}^{2n+1}$ , a system  $q_1, q_2, \ldots, q_k$  of points in  $\mathbb{R}^{2n+1}$ , and a positive number  $\alpha > 0$  are given, one can choose the nerve
$\mathcal{N}(\mathcal{U})$  in such a way that  $H \cap \mathcal{N}(\mathcal{U}) = \emptyset$  and the vertices  $p_1, p_2, ..., p_k$ of  $\mathcal{N}(\mathcal{U})$  satisfy the inequality  $\varrho(p_i, q_i) < \alpha$  for i = 1, 2, ..., k.

**Proof.** Let  $r_1, r_2, \ldots, r_{n+1}$  be a linearly independent system of points in  $\mathbb{R}^{2n+1}$  which spans the linear variety H. By virtue of Theorem 1.10.2, there exists a system of points  $p_1, p_2, \ldots, p_k \in \mathbb{R}^{2n+1}$  such that  $\varrho(p_i, q_i) < \alpha$ for i = 1, 2, ..., k and the system  $r_1, r_2, ..., r_{n+1}, p_1, p_2, ..., p_k$  is in general position. Since ord  $\mathcal{U} \leq n$ , for each sequence  $i_0 < i_1 < \ldots < i_l \leq k$ of l+1 natural numbers such that  $U_{i_0} \cap U_{i_1} \cap \ldots \cap U_{i_l} \neq \emptyset$  we have  $l \leq n < 2n+1$ , so that the points  $p_{i_0}, p_{i_1}, \dots, p_{i_l}$  form a linearly independent system and the simplex  $p_{i_0}p_{i_1} \dots p_{i_l} \subset R^{2n+1}$  is well defined. Let us denote by  $\mathcal{N}(\mathcal{U})$  the family of all simplexes  $p_{i_0}p_{i_1} \dots p_{i_l} \subset \mathbb{R}^{2n+1}$  obtained in this way. To complete the proof it suffices to show that  $\mathcal{N}(\mathcal{U})$  is a simplicial complex and that  $H \cap N(\mathcal{U}) = \emptyset$ . Since each face of a simplex in  $\mathcal{N}(\mathcal{U})$ is in  $\mathcal{N}(\mathcal{U})$ , to show that  $\mathcal{N}(\mathcal{U})$  is a complex it remains to check that if  $S_1 = p_{i_0} p_{i_1} \dots p_{i_l}$  and  $S_2 = p_{j_0} p_{j_1} \dots p_{j_m}$  belong to  $\mathcal{N}(\mathcal{U})$ and the intersection  $S_1 \cap S_2$  is non-empty, then  $S_1 \cap S_2$  is a face of both  $S_1$  and  $S_2$ . Obviously, it is enough to prove that if  $p \in S_1 \cap S_2$  and in the representation  $p = \sum_{s=0}^{l} \lambda_s p_{i_s}$ , where  $\sum_{s=0}^{l} \lambda_s = 1$ , we have  $\lambda_{s_0} > 0$ , then  $p_{i_{s_0}}$  is a vertex of  $S_2$ . Since  $p \in S_2$ ,  $p = \sum_{i=0}^m \lambda'_i p_{j_i}$ , where  $\sum_{i=0}^m \lambda'_i = 1$ ; we have

$$\sum_{s=0}^{l} \lambda_s p_{i_s} - \sum_{t=0}^{m} \lambda_t' p_{j_t} = 0 \quad \text{and} \quad \sum_{s=0}^{l} \lambda_s - \sum_{t=0}^{m} \lambda_t' = 0$$

As observed before,  $l \leq n$ ; similarly,  $m \leq n$ . Hence, the points  $p_{i_0}, p_{i_1}, ...$ ...  $p_{i_1}, p_{j_0}, p_{j_1}, ..., p_{j_m}$  form a linearly independent system, because its cardinality is not larger than  $l+m+2 \leq 2n+2$  and the whole system  $r_1, r_2, ...$ ...,  $r_{n+1}, p_1, p_2, ..., p_k$  is in general position. It follows that  $p_{i_{s_0}}$  occurs among the points  $p_{j_0}, p_{j_1}, ..., p_{j_m}$ , i.e.,  $p_{i_{s_0}}$  is a vertex of  $S_2$ . In a similar way one shows that all simplexes in  $\mathcal{N}(\mathcal{U})$  are disjoint from the linear *n*-variety H.  $\Box$ 

For each finite open cover  $\mathscr{U} = \{U_i\}_{i=1}^k$  of a metric space X and a sequence of points  $p_1, p_2, \ldots, p_k$  in Euclidean *m*-space  $\mathbb{R}^m$  a continuous mapping of X to  $\mathbb{R}^m$  is defined in a natural way.

**1.10.5. Definition.** Let  $\mathscr{U} = \{U_i\}_{i=1}^k$  be a finite open cover of a metric space X and  $p_1, p_2, \ldots, p_k$  a sequence of points in Euclidean *m*-space  $\mathbb{R}^m$ .

The continuous mapping  $\varkappa: X \to R^m$  defined by letting

(4) 
$$\varkappa(x) = \varkappa_1(x)p_1 + \varkappa_2(x)p_2 + \ldots + \varkappa_k(x)p_k,$$

where

(5) 
$$\varkappa_{i}(x) = \frac{\varrho(x, X \setminus U_{i})}{\sum_{j=1}^{k} \varrho(x, X \setminus U_{j})} \quad \text{for } i = 1, 2, ..., k,$$

is called the *x*-mapping determined by the cover  $\mathscr{U}$  and the points  $p_1, p_2, ..., p_k$ . Let us observe that the denominator in (5) does not vanish, because  $\varrho(x, X \setminus U_j) > 0$  whenever  $x \in U_j$ . Let us also note that  $\sum_{i=1}^k \varkappa_i(x) = 1$  for every  $x \in X$ .

As explained in the following theorem, all continuous mappings of a metric space to a Euclidean space can be approximated by *x*-mappings.

**1.10.6. Theorem.** Let  $f: X \to \mathbb{R}^m$  be a continuous mapping of a metric space X to Euclidean m-space  $\mathbb{R}^m$  and let  $\delta$  be a positive number. If  $\mathscr{U} = \{U_i\}_{i=1}^k$  is a finite open cover of the space X and  $p_1, p_2, \ldots, p_k \in \mathbb{R}^m$  is a sequence of points such that

(6) 
$$\delta(\{p_i\} \cup f(U_i)) < \delta \quad \text{for } i = 1, 2, ..., k,$$

then the  $\varkappa$ -mapping  $\varkappa: X \to \mathbb{R}^m$  determined by the cover  $\mathscr{U}$  and the points  $p_1, p_2, \ldots, p_k$  has the property that  $\varrho(f(x), \varkappa(x)) < \delta$  for every  $x \in X$ .

**Proof.** Consider a point  $x \in X$  and let  $U_{i_0}, U_{i_1}, ..., U_{i_l}$  be all members of the cover  $\mathscr{U}$  that contain the point x. By virtue of (5),  $\varkappa_i(x) = 0$  whenever  $i \neq i_j$  for j = 0, 1, ..., l. It follows from (6) that  $||f(x) - p_{i_j}|| = \varrho(f(x), p_{i_j}) < \delta$  for j = 0, 1, ..., l; applying (4), we obtain

$$\varrho(f(x), \varkappa(x)) = ||f(x) - \varkappa(x)|| = \left\| \sum_{j=0}^{l} \varkappa_{ij}(x) f(x) - \sum_{j=0}^{l} \varkappa_{ij}(x) p_{ij} \right\|$$
  
$$\leq \sum_{j=0}^{l} \varkappa_{ij}(x) ||f(x) - p_{ij}|| < \sum_{j=0}^{l} \varkappa_{ij}(x) \delta = \delta \sum_{j=0}^{l} \varkappa_{ij}(x) = \delta. \square$$

The notion of a  $\varkappa$ -mapping determined by a finite open cover  $\mathscr{U} = \{U_i\}_{i=1}^k$  of a metric space and points  $p_1, p_2, \ldots, p_k \in \mathbb{R}^m$  proves particularly useful in the case where the points  $p_1, p_2, \ldots, p_k$  are the vertices of a nerve  $\mathscr{N}(\mathscr{U})$  of the cover  $\mathscr{U}$ .

**1.10.7. Theorem.** If  $\mathcal{U} = \{U_i\}_{i=1}^k$  is a finite open cover of a metric space X and  $\mathcal{N}(\mathcal{U})$  a nerve of  $\mathcal{U}$  with vertices  $p_1, p_2, \ldots, p_k \in \mathbb{R}^m$ , then the  $\varkappa$ -mapping  $\varkappa: X \to \mathbb{R}^m$  determined by the cover  $\mathcal{U}$  and the points  $p_1, p_2, \ldots, p_k$  satisfies the conditions

(7) 
$$\varkappa(X) \subset N(\mathscr{U})$$

anď

(8) 
$$\kappa^{-1}(\operatorname{St}_{\mathscr{N}(\mathscr{U})}(p_i)) = U_i \quad for \ i = 1, 2, ..., k.$$

**Proof.** Consider a point  $x \in X$  and let  $U_{i_0}, U_{i_1}, ..., U_{i_l}$  be all members of the cover  $\mathscr{U}$  that contain the point x. Since  $U_{i_0} \cap U_{i_1} \cap ... \cap U_{i_l} \neq \emptyset$ , the simplex  $p_{i_0}p_{i_1} ... p_{i_l}$  belongs to the nerve  $\mathscr{N}(\mathscr{U})$ . It follows from (5) and (4) that  $\varkappa(x) \in p_{i_0}p_{i_1} ... p_{i_l}$ , so that (7) is satisfied.

Since the representation of points of  $N(\mathcal{U})$  in form (3) is unique, we have  $\lambda_i(\varkappa(x)) = \varkappa_i(x)$  for i = 1, 2, ..., k. Thus

$$\begin{aligned} \varkappa^{-1} \big( \operatorname{St}_{\mathscr{N}(\mathscr{U})}(p_i) \big) &= \{ x \in X \colon \varkappa(x) \in \operatorname{St}_{\mathscr{N}(\mathscr{U})}(p_i) \} = \{ x \in X \colon \lambda_i \big( \varkappa(x) \big) > 0 \} \\ &= \{ x \in X \colon \varkappa_i(x) > 0 \} = U_i \quad \text{for } i = 1, 2, \dots, k, \end{aligned}$$

i.e., (8) is also satisfied.

**1.10.8. Definition.** Let  $\mathscr{E}$  be an open cover of a topological space X and  $f: X \to Y$  a continuous mapping of X to a topological space Y; we say that f is an  $\mathscr{E}$ -mapping if there exists an open cover  $\mathscr{U}$  of the space Y such that the cover  $f^{-1}(\mathscr{U})$  is a refinement of  $\mathscr{E}$ .

**1.10.9. Definition.** Let  $\varepsilon$  be a positive number and  $f: X \to Y$  a continuous mapping of a metric space X to a topological space Y; we say that f is an  $\varepsilon$ -mapping if  $\delta(f^{-1}(y)) < \varepsilon$  for every  $y \in Y$ . Obviously, if  $\mathscr{E}$  is an open cover of a metric space and mesh  $\mathscr{E} < \varepsilon$ , then each  $\mathscr{E}$ -mapping is an  $\varepsilon$ -mapping.

**1.10.10. Theorem.** If X is a compact metric space, then for every open cover  $\mathscr{E}$  of the space X there exists a positive number  $\varepsilon$  such that each  $\varepsilon$ -mapping of X to a Hausdorff space is an  $\mathscr{E}$ -mapping.

**Proof.** Let  $\varepsilon$  be a Lebesgue number for the cover  $\mathscr{E}$ . Consider an  $\varepsilon$ -mapping  $f: X \to Y$  of X to a Hausdorff space Y. For every  $y \in Y$  there exists a  $V_y \in \mathscr{E}$  such that  $f^{-1}(y) \subset V_y$ . Since every continuous mapping of a compact space to a Hausdorff space maps closed sets onto closed sets, the set  $U_y$ 

=  $Y \setminus f(X \setminus V_y)$  is open for every  $y \in Y$ ; one readily checks that  $y \in U_y$ and  $f^{-1}(U_y) \subset V_y$ . Hence  $\mathscr{U} = \{U_y\}_{y \in Y}$  is an open cover of the space Y such that the cover  $f^{-1}(\mathscr{U})$  is a refinement of  $\mathscr{E}$ , i.e., f is an  $\mathscr{E}$ -mapping.  $\Box$ 

**1.10.11. Theorem.** If for every finite open cover  $\mathscr{E}$  of a normal space X there exists an  $\mathscr{E}$ -mapping  $f: X \to Y$  of X to a compact space Y such that dim  $Y \leq n$ , then dim  $X \leq n$ .

**Proof.** Consider an arbitrary finite open cover  $\mathscr{E}$  of the space X and an  $\mathscr{E}$ -mapping  $f: X \to Y$  of X to a compact space Y such that dim  $Y \leq n$ . Let  $\mathscr{U}$  be an open cover of the space Y such that the cover  $f^{-1}(\mathscr{U})$  is a refinement of  $\mathscr{E}$ . As the space Y is compact,  $\mathscr{U}$  has a finite open refinement  $\mathscr{V}$  which, by virtue of the inequality dim  $Y \leq n$ , has in its turn a finite open refinement  $\mathscr{W}$  such that ord  $\mathscr{W} \leq n$ . Now, the cover  $f^{-1}(\mathscr{W})$  of the space X is a finite open cover of order  $\leq n$  which refines  $\mathscr{E}$ , so that dim  $X \leq n$ .  $\Box$ 

Let us observe that if the space Y in the last theorem is a polyhedron, one can slightly modify the above proof so as to use only the geometric dimension of Y. Indeed, in this case the existence of the refinement  $\mathscr{W}$ of the cover  $\mathscr{V}$  such that  $\operatorname{ord} \mathscr{W} \leq n$  follows from the elementary fact that every polyhedron of geometric dimension  $\leq n$  has finite covers of order  $\leq n$  by open sets of arbitrarily small diameter.

Theorems 1.10.10 and 1.10.11 imply

**1.10.12. Theorem.** If X is a compact metric space and for every positive number  $\varepsilon$  there exists an  $\varepsilon$ -mapping  $f: X \to Y$  of X to a compact space Y such that dim  $Y \le n$ , then dim  $X \le n$ .  $\Box$ 

We are now ready to characterize dimension in terms of mappings to polyhedra.

**1.10.13. Theorem on E-mappings.** A metric space X satisfies the inequality  $\dim X \leq n$  if and only if for every finite open cover  $\mathscr{E}$  of the space X there exists an  $\mathscr{E}$ -mapping of X to a polyhedron of dimension  $\leq n$ .

**Proof.** The theorem is obvious if  $\dim X = -1$ . Consider a metric space X such that  $0 \leq \dim X \leq n$  and a finite open cover  $\mathscr{E}$  of the space X. Let  $\mathscr{U} = \{U_i\}_{i=1}^k$  be an open refinement of  $\mathscr{E}$  such that  $\operatorname{ord} \mathscr{U} \leq n$  and let  $\mathscr{N}(\mathscr{U})$  be a nerve of  $\mathscr{U}$  with vertices  $p_1, p_2, \ldots, p_k \in \mathbb{R}^m$ . From Theorem 1.10.7 it follows that the  $\varkappa$ -mapping  $\varkappa: X \to N(\mathscr{U})$  determined by the cover  $\mathscr{U}$  and the points  $p_1, p_2, \ldots, p_k$  is an  $\mathscr{E}$ -mapping, because  $\{\operatorname{St}_{\mathscr{N}(\mathscr{U})}(p_i)\}_{i=1}^k$  is an open cover of the underlying polyhedron  $N(\mathcal{U})$ . The inequality  $\operatorname{ord} \mathcal{U} \leq n$  implies that  $N(\mathcal{U})$  has dimension  $\leq n$ . To complete the proof it suffices to apply Theorem 1.10.11.  $\Box$ 

Let us note that, referring in the above proof to Theorem 1.10.11, we used the coincidence of the geometric dimension and the topological dimension of polyhedra. The observation following Theorem 1.10.11 shows how this can be eliminated.

**1.10.14. Theorem on**  $\varepsilon$ -mappings. A compact metric space X satisfies the inequality dim  $X \leq n$  if and only if for every positive number  $\varepsilon$  there exists an  $\varepsilon$ -mapping of X to a polyhedron of dimension  $\leq n$ .

**Proof.** Consider a compact metric space X such that  $\dim X \leq n$  and a positive number  $\varepsilon$ . Let  $\mathscr{E}$  be a finite open refinement of the open cover  $\{B(x, \varepsilon/2)\}_{x \in X}$  of the space X. By virtue of the theorem on  $\mathscr{E}$ -mappings, there exists an  $\mathscr{E}$ -mapping of X to a polyhedron of dimension  $\leq n$ ; one readily checks that this is an  $\varepsilon$ -mapping. To complete the proof it suffices to apply Theorem 1.10.12.  $\Box$ 

The comment following the theorem on  $\mathscr{E}$ -mappings applies as well to the above proof. Let us also observe that in the theorem on  $\varepsilon$ -mappings the assumption of compactness is essential; indeed, there exist separable metric spaces of dimension larger than one that can be mapped to the interval I by a one-to-one continuous mapping (see Problem 1.4.F(b)).

The reader has undoubtedly noted that in the proofs of Theorems 1.10.13 and 1.10.14 only the existence of a nerve was applied and not the much stronger Theorem 1.10.4. The latter theorem will be applied in the next section, where it will prove to be, together with Theorem 1.10.6, the core of the proofs of the embedding and the universal space theorems.

We shall now show that the theorems on  $\mathscr{E}$ -mappings and on  $\varepsilon$ -mappings can be somewhat strengthened, viz., that the existence of mappings onto polyhedra can be established. In the proof we shall apply the following auxiliary theorem, which states that every subset A of the underlying polyhedron  $|\mathscr{K}|$  of a complex  $\mathscr{K}$  can be swept out of interiors of all simplexes in  $\mathscr{K}$  which are not contained in A.

**1.10.15. The sweeping out theorem.** For every simplicial complex  $\mathcal{K}$  and each subspace A of the underlying polyhedron  $|\mathcal{K}|$  there exist a subcomplex  $\mathcal{K}_0$  of  $\mathcal{K}$  and a continuous mapping  $f: A \to |\mathcal{K}_0|$  such that  $f(A) = |\mathcal{K}_0|$  and  $f(A \cap S) \subset S$  for every  $S \in \mathcal{K}$ .

Ch. 1, § 10]

**Proof.** Let K be the collection of all subcomplexes  $\mathscr{K}'$  of  $\mathscr{K}$  for which there exists a continuous mapping  $f: A \to |\mathscr{K}'|$  such that

(9) 
$$f(A \cap S) \subset S$$
 for every  $S \in \mathscr{K}$ .

As the embedding  $f: A \to |\mathscr{K}|$  of the subspace A in  $|\mathscr{K}|$  satisfies (9), the complex  $\mathscr{K}$  itself belongs to the collection K. Hence the collection K is non-empty and, being finite, contains a subcomplex  $\mathscr{K}_0$  of  $\mathscr{K}$  such that no proper subcomplex of  $\mathscr{K}_0$  belongs to K. Consider a continuous mapping  $f: A \to |\mathscr{K}_0|$  which satisfies (9); we shall show that  $f(A) = |\mathscr{K}_0|$ .

Suppose that there exists a point  $x_0 \in |\mathscr{K}_0| \setminus f(A)$ . Let  $S_0 = p_0 p_1 \dots p_k$  be the intersection of all simplexes in  $\mathscr{K}_0$  which contain the point  $x_0$ . Define

$$\mathscr{S}_0 = \{ S \in \mathscr{K}_0 \colon S_0 \leqslant S \}$$

and

$$\mathscr{B}_0 = \{T \in \mathscr{K}_0 : T \notin \mathscr{S}_0 \text{ and } T \leqslant S \text{ for an } S \in \mathscr{S}_0\};$$

clearly,  $\mathscr{K}_1 = \mathscr{K}_0 \setminus \mathscr{S}_0$  and  $\mathscr{B}_0$  are subcomplexes of  $\mathscr{K}_0$ . The set  $G = \bigcap_{i=0}^k \operatorname{St}_{\mathscr{K}_0}(p_i)$  is open in  $|\mathscr{K}_0|$ ; one readily checks that

$$G = \bigcup \mathscr{G}_0 = G \cup B_0$$
, where  $B_0 = |\mathscr{B}_0|$ .



Fig. 10

Let p denote the projection of  $f(A) \cap \overline{G}$  from the point  $x_0$  onto  $B_0$ . The restriction

(10) 
$$p|f(A) \cap S: f(A) \cap S \to B_0 \cap S$$

is continuous for every  $S \in \mathscr{G}_0$ ; thus  $p: f(A) \cap \overline{G} \to B_0$  is a continuous mapping. Since p(x) = x for  $x \in f(A) \cap B_0$ , the formula

$$g(x) = \begin{cases} p(x) & \text{for } x \in f(A) \cap G, \\ x & \text{for } x \in f(A) \setminus G \end{cases}$$

defines a continuous mapping  $g: f(A) \to |\mathscr{K}_1|$ ; let  $f_1 = gf: A \to |\mathscr{K}_1|$ . It follows from (10) that  $f_1(A \cap S) \subset S$  for every  $S \in \mathscr{K}$ ; hence  $\mathscr{K}_1 \in K$ , which contradicts the definition of  $\mathscr{K}_0$ . The contradiction shows that  $f(A) = |\mathscr{K}_0|$ .  $\square$ 

**1.10.16. Theorem.** A metric (compact metric) space X satisfies the inequality  $\dim X \leq n$  if and only if for every finite open cover  $\mathscr{E}$  of the space X (for every positive number  $\varepsilon$ ) there exists an  $\mathscr{E}$ -mapping (an  $\varepsilon$ -mapping) of X onto a polyhedron of dimension  $\leq n$ .

**Proof.** It suffices to show that if a metric space X satisfies the inequality  $0 \leq \dim X \leq n$ , then for every finite open cover  $\mathscr{E}$  of the space X there exists an  $\mathscr{E}$ -mapping of X onto a polyhedron of dimension  $\leq n$ . Let  $\mathscr{U} = \{U_i\}_{i=1}^k$  be an open refinement of  $\mathscr{E}$  such that  $\operatorname{ord} \mathscr{U} \leq n$  and let  $\varkappa: X \to |\mathscr{K}|$  be the  $\varkappa$ -mapping of X to the underlying polyhedron of a nerve  $\mathscr{K} = \mathscr{N}(\mathscr{U})$  of the cover  $\mathscr{U}$ . By virtue of the sweeping out theorem, there exists a continuous mapping f of the subspace  $A = \varkappa(X)$  of  $|\mathscr{K}|$  onto the underlying polyhedron  $|\mathscr{K}_0|$  of a subcomplex  $\mathscr{K}_0$  of  $\mathscr{K}$  which satisfies (9). Let us note that, for every  $x \in A$  and every vertex  $p_i$  of  $\mathscr{K}_0$ , if  $f(x) \in \operatorname{St}_{\mathscr{K}_0}(p_i)$ , then  $x \in \operatorname{St}_{\mathscr{K}}(p_i)$ , i.e.,

(11) 
$$f^{-1}(\operatorname{St}_{\mathscr{K}_0}(p_i)) \subset \operatorname{St}_{\mathscr{K}}(p_i).$$

Indeed, if  $x \notin \operatorname{St}_{\mathscr{H}}(p_i)$ , then the intersection S of all simplexes in  $\mathscr{H}$  that contain the point x does not contain  $p_i$ ; since  $f(x) \in S$  by virtue of (9), it follows that  $S \in \mathscr{H}_0$ , and thus  $f(x) \notin \operatorname{St}_{\mathscr{H}_0}(p_i)$ . Inclusions (8) and (11) show that the composition  $f_{\mathscr{H}}: X \to |\mathscr{H}_0|$  is an  $\mathscr{E}$ -mapping of X onto  $|\mathscr{H}_0|.\square$ 

The final part of this section will be devoted to subspaces of Euclidean spaces. We start by introducing the notion of an  $\varepsilon$ -translation.

**1.10.17. Definition.** Let  $\varepsilon$  be a positive number, A, B subspaces of a metric space X, and  $f: A \to B$  a continuous mapping of A to B; we say that f is an  $\varepsilon$ -translation if  $\varrho(x, f(x)) < \varepsilon$  for every  $x \in A$ . Obviously, each  $\varepsilon$ -translation is a  $3\varepsilon$ -mapping and an  $\varepsilon$ -translation defined on a compact subspace of X is a  $2\varepsilon$ -mapping.

For subspaces of Euclidean spaces it is much more interesting to discuss  $\varepsilon$ -translations to polyhedra rather than  $\varepsilon$ -mappings. Clearly, every subspace of  $\mathbb{R}^m$  which can be mapped to a polyhedron by an  $\varepsilon$ -translation is bounded. It turns out that for bounded subspaces of Euclidean spaces we have a theorem on  $\varepsilon$ -translations which strictly parallels the theorem on  $\varepsilon$ -map-

pings. In the proof of that theorem we shall apply the elementary fact that, for every finite family  $\mathscr{S}$  of simplexes in  $\mathbb{R}^m$ , the union  $L = \bigcup \mathscr{S}$  is a polyhedron whose dimension is equal to the largest integer *n* such that  $\mathscr{S}$  contains an *n*-simplex; more exactly, we shall assume that there exists a simplicial complex  $\mathscr{L}$  such that  $L = |\mathscr{L}|$  and every simplex of  $\mathscr{L}$  is contained in a member of  $\mathscr{S}$ . A proof of this fact is outlined in Problem 1.10.I.

**1.10.18. Theorem on**  $\varepsilon$ -translations. If X is a bounded subspace of Euclidean m-space  $\mathbb{R}^m$ , and X satisfies the inequality dim  $X \leq n$ , then for every positive number  $\varepsilon$  there exists an  $\varepsilon$ -translation  $f: X \to K$  of X onto a polyhedron  $K \subset \mathbb{R}^m$  of dimension  $\leq n$ .

**Proof.** Without loss of generality one can suppose that  $0 \le n \le m$ . Let  $\mathscr{U} = \{U_i\}_{i=1}^k$  be a finite open cover of the space X such that mesh $\mathscr{U} < \varepsilon/4$  and  $U_i \ne \emptyset$  for i = 1, 2, ..., k; since dim  $X \le n$ , one can assume—replacing  $\mathscr{U}$ , if necessary, by a refinement—that ord  $\mathscr{U} \le n$ . For i = 1, 2, ..., k choose a point  $q_i \in U_i$  and apply Theorem 1.10.2 to obtain a system of points  $p_1, p_2, ..., p_k \in \mathbb{R}^m$  in general position such that

(12) 
$$\delta(\{p_i\} \cup U_i) < \varepsilon/4 \quad \text{for } i = 1, 2, ..., k.$$

Since  $\operatorname{ord} \mathscr{U} \leq n$ , for each sequence  $i_0 < i_1 < \ldots < i_l \leq k$  of l+1 natural numbers such that  $U_{i_0} \cap U_{i_1} \cap \ldots \cap U_{i_l} \neq \emptyset$  we have  $l \leq n \leq m$ , so that the system  $p_1, p_2, \ldots, p_k$  being in general position—the simplex  $p_{i_0}p_{i_1} \ldots p_{i_l} \subset \mathbb{R}^m$  is well defined; by virtue of (12),  $\delta(p_{i_0}p_{i_1} \ldots p_{i_l}) = \delta(\{p_{i_0}, p_{i_1}, \ldots, \dots, p_{i_l}\}) < \varepsilon/2$ . Let  $\mathscr{S}$  be the family of all simplexes obtained in this way; clearly, mesh  $\mathscr{S} < \varepsilon/2$ . The union  $L = \bigcup \mathscr{S}$  is a polyhedron of dimension  $\leq n$ ; moreover, one can assume that  $L = |\mathscr{L}|$ , where  $\mathscr{L}$  is a simplicial complex every simplex of which is contained in a member of  $\mathscr{S}$ , so that we have mesh  $\mathscr{L} < \varepsilon/2$ . It follows from (12) and Theorem 1.10.6 applied to the embedding of X in  $\mathbb{R}^m$  that the  $\varkappa$ -mapping  $\varkappa: X \to \mathbb{R}^m$  determined by the cover  $\mathscr{U}$  and the points  $p_1, p_2, \ldots, p_k$  has the property that

(13) 
$$\varrho(x, \varkappa(x)) < \varepsilon/4$$
 for every  $x \in X$ .

Moreover, as one easily sees,  $A = \varkappa(X) \subset L$ . By virtue of the sweeping out theorem, there exists a continuous mapping  $g: A \to K$  of A onto a polyhedron  $K \subset L$  of dimension  $\leq n$  such that

(14) 
$$\varrho(\varkappa(x), g\varkappa(x)) < \varepsilon/2$$
 for every  $x \in X$ .

It follows from (13) and (14) that the composition  $f = g \varkappa \colon X \to K$  is the required  $\varepsilon$ -translation.

Theorems 1.10.12 and 1.10.18 yield the following

**1.10.19. Theorem.** A compact subspace X of Euclidean m-space  $\mathbb{R}^m$  satisfies the inequality dim  $X \leq n$  if and only if for every positive number  $\varepsilon$  there exists an  $\varepsilon$ -translation  $f: X \to K$  of X onto a polyhedron  $K \subset \mathbb{R}^m$  of dimension  $\leq n$ .  $\Box$ 

We shall now describe a two-dimensional subspace X of the cube  $I^3$ such that for every positive number  $\varepsilon$  there exists an  $\varepsilon$ -translation  $f: X \to K$ of X to a polyhedron  $K \subset R^3$  of dimension  $\leq 1$ . In this way it will be proved that in the theorem on  $\varepsilon$ -translations the assumption of compactness is essential. Let us at once note that the space X has finite open covers of order  $\leq 1$  by sets with arbitrarily small diameters (cf. Theorem 1.6.12).

We start with a lemma on decompositions of continua, which will be applied to evaluate the dimension of the space X. The lemma states an important topological fact and is known as *Sierpiński's theorem*; it is preceded by two technical lemmas.

**1.10.20. Lemma.** If A is a closed subspace of a continuum X such that  $\emptyset \neq A \neq X$ , then for every component C of the space A we have  $C \cap \operatorname{Fr} A \neq \emptyset$ , where  $\operatorname{Fr} A$  is the boundary of A in X.

**Proof.** Assume that  $C \cap \operatorname{Fr} A = \emptyset$  and consider the family  $\{U_s\}_{s \in S}$  of all open-and-closed subsets of A which contain the component C; it follows from Lemma 1.4.4 that  $\bigcap_{s \in S} U_s = C$ . The subspace  $F = \operatorname{Fr} A$  of the space X is compact, and the family  $\{F \setminus U_s\}_{s \in S}$  is an open cover of F; thus there exists a finite number of indexes  $s_1, s_2, \ldots, s_k \in S$  such that  $F \subset \bigcup_{i=1}^k (F \setminus U_{s_i}) = F \setminus \bigcap_{i=1}^k U_{s_i}$ . The set  $U = \bigcap_{i=1}^k U_{s_i}$  is disjoint from F, i.e.,  $U \subset \operatorname{Int} A$ ; being open-and-closed in A, the set U is open-and-closed in the continuum X. Now,  $\emptyset \neq C \subset U$ , so that U = X, and thus  $\operatorname{Fr} A = \emptyset$ , which contradicts the connectedness of X.  $\Box$ 

**1.10.21. Lemma.** If a continuum X is represented as the union of a sequence  $X_1, X_2, \ldots$  of pairwise disjoint closed sets of which at least two are nonempty, then for every natural number i there exists a continuum  $C \subset X$ such that  $C \cap X_i = \emptyset$  and at least two sets in the sequence  $C \cap X_1, C \cap X_2, \ldots$ are non-empty. **Proof.** If  $X_i = \emptyset$ , we let C = X; hence, we can assume that  $X_i \neq \emptyset$ . Consider a natural number  $j \neq i$  such that  $X_j \neq \emptyset$  and a pair  $U, V \subset X$  of disjoint open sets such that  $X_i \subset U$  and  $X_j \subset V$ . Let x be a point in  $X_j$  and C the component of the space  $\overline{V}$  which contains the point x. Clearly, C is a continuum,  $C \cap X_i = \emptyset$  and  $C \cap X_j \neq \emptyset$ . Since, by virtue of Lemma 1.10.20,  $C \cap \operatorname{Fr} \overline{V} \neq \emptyset$ , and since  $X_j \subset \operatorname{Int} \overline{V}$ , there exists a natural number  $k \neq j$  such that  $C \cap X_k \neq \emptyset$ .  $\Box$ 

**1.10.22. Lemma.** If a continuum X is represented as the union of a sequence  $X_1, X_2, ...$  of pairwise disjoint closed sets, then at most one of the sets  $X_i$  is non-empty.

**Proof.** Assume that  $X = \bigcup_{i=1}^{\infty} X_i$ , where the sets  $X_i$  are closed,  $X_i \cap X_j = \emptyset$ whenever  $i \neq j$ , and at least two of the sets  $X_i$  are non-empty. From Lemma 1.10.21 it follows that there exists a decreasing sequence  $C_1 \supset C_2 \supset ...$ of non-empty continua contained in X such that  $C_i \cap X_i = \emptyset$  for i = 1, 2, ...Thus  $\bigcap_{i=1}^{\infty} C_i = (\bigcap_{i=1}^{\infty} C_i) \cap (\bigcup_{i=1}^{\infty} X_i) \stackrel{!}{=} \emptyset$ , which contradicts the compactness of X.  $\Box$ 

**1.10.23.** Sitnikov's example. For every natural number *i* consider the family of all planes in  $\mathbb{R}^3$  determined by equations of the form  $x_j = z/i$ , where j = 1, 2, 3 and z is an arbitrary integer. The planes yield a decomposition of  $\mathbb{R}^3$  into congruent cubes whose edges have length 1/i and whose interiors are pairwise disjoint; denote by  $\mathscr{K}_i$  the family of all the cubes thus obtained and by  $A_i$  the union of all the edges of cubes in  $\mathscr{K}_i$ .

Let  $B_1 = A_1$  and for i = 2, 3, ... translate the set  $A_i$  as a rigid body to obtain a set  $B_i$  disjoint from the union  $B_1 \cup B_2 \cup ... \cup B_{l-1}$  of all the sets previously defined. Clearly, the sets  $B_i$  are closed in  $\mathbb{R}^3$  and the union  $B = \bigcup_{i=1}^{\infty} B_i$  is dense in  $\mathbb{R}^3$ . We shall show that the subspace  $X = I^3 \setminus B$ of the cube  $I^3$  is two-dimensional and yet for every positive number  $\varepsilon$ there exists an  $\varepsilon$ -translation  $f: X \to K$  of X to a polyhedron  $K \subset \mathbb{R}^3$  of dimension  $\leq 1$ .

Since the set  $I^3 \setminus X = I^3 \cap B$  is dense in  $I^3$ , it follows from Corollary 1.8.11 that ind  $X \leq 2$ . Assume that ind  $X \leq 1$  and consider arbitrary points

x and y in the interior G of  $I^3$  in  $R^3$  such that  $x \in B_2$  and  $y \in B_3$ . By virtue of Mazurkiewicz's theorem there exists a continuum  $C \subset G \setminus X = G \cap B$ which contains x and y; the continuum C is the union of the sequence  $C \cap B_1, C \cap B_2, ...$  of pairwise disjoint closed sets of which at least two are non-empty. The contradiction of Sierpiński's theorem shows that indX = 2.



Now, let  $\varepsilon$  be an arbitrary positive number. Consider a natural number *i* such that  $4/i < \varepsilon$  and translate the set  $X \cup B_i$  as a rigid body to make the joints of  $B_i$  coincide with the centres of the cubes in  $\mathcal{K}_i$ . One can assume that the translation g:  $X \cup B_i \to R^3$  does not shift points by more than 2/i, i.e., that  $\rho(x, g(x)) \leq 2/i$  for  $x \in X$ ; clearly  $g(X) \subset \mathbb{R}^3 \setminus g(B_i)$ . Consider now a fixed cube  $T \in \mathscr{K}_i$ . One readily checks that the projection from the centre of T onto the boundary of T maps the set  $T \cap g(X)$  onto a subset of the boundary which does not contain the centres of the faces of T. On the boundary of T the projection coincides with the identity mapping, so that by performing such projections simultaneously on all the cubes in  $\mathcal{K}_i$  we obtain a continuous mapping  $p_1$  that sends g(X) to the union of all faces of cubes in  $\mathcal{K}_i$  in such a way that the centres of the faces do not belong to  $p_1g(X)$ . Now, consider a fixed face S of T. The projection from the centre of S onto the boundary of S maps the set  $S \cap p_1 g(X)$  onto a subset of  $A_i$  and coincides with the identity mapping on the boundary of S. By performing such projections simultaneously on all faces of cubes in  $\mathscr{K}_i$  we obtain a continuous mapping  $p_2$  that sends  $p_1g(X)$  to  $A_i$ . Since the points g(x) and  $p_2 p_1 g(x)$  lie in the same cube of  $\mathcal{K}_i$ , we have  $\varrho(g(x))$ ,  $p_2p_1g(x) > 2/i$ . Hence  $\varrho(x, p_2p_1g(x)) < 4/i < \varepsilon$  for every  $x \in X$  and the set  $p_2 p_1 g(X)$  is contained in a one-dimensional polyhedron  $K \subset A_i$ ;

#### Problems

thus the required  $\varepsilon$ -translation  $f: X \to K$  is defined by letting  $f(x) = p_2 p_1 g(x)$  for every  $x \in X$ .

Though dim  $X = \operatorname{ind} X = 2$ , the space X has, for every positive number  $\varepsilon$ , a finite open cover  $\mathscr{U}$  such that mesh  $\mathscr{U} < \varepsilon$  and  $\operatorname{ord} \mathscr{U} \leq 1$ . Indeed, if  $f: X \to K$  is an  $\varepsilon/3$ -translation of X to a polyhedron  $K \subset \mathbb{R}^3$  of dimension  $\leq 1$ , then the family  $\mathscr{U} = f^{-1}(\mathscr{V})$ , where  $\mathscr{V}$  is a finite open cover of K such that mesh  $\mathscr{V} < \varepsilon/3$  and  $\operatorname{ord} \mathscr{V} \leq 1$ , has the required properties.  $\Box$ 

# Historical and bibliographic notes

Nerves of covers were introduced and studied by Alexandroff in [1927], and  $\varkappa$ -mappings—by Kuratowski in [1933a]. The discovery of these two notions was a turning point in the development of dimension theory, and even of the whole of topology; it made possible the combining of the point-set methods of general topology and the combinatorial methods of traditional algebraic topology. Theorem 1.10.13 was proved by Kuratowski in [1933a] and Theorem 1.10.14 by Alexandroff in [1928]. Theorem 1.10.15 for compact subspaces of polyhedra was obtained by Alexandroff in [1928] (cf. Problem 1.10.H); the extension to arbitrary subspaces was given by Kuratowski in [1933a]. Theorems 1.10.18 and 1.10.19 were proved by Alexandroff in [1928] (announcement in [1926]). Example 1.10.23 was described by Sitnikov in [1953].

# Problems

1.10.A. (a) Check that the diameter of the simplex  $p_0p_1 \dots p_n$  is equal to the diameter of the set  $\{p_0, p_1, \dots, p_n\}$  of its vertices.

(b) Prove that the simplex  $p_0p_1 \dots p_n$  is the smallest convex set which contains the points  $p_0, p_1, \dots, p_n$ .

(c) Show that every simplex determines its vertices, i.e., that if  $p_0p_1 \dots p_n = q_0q_1 \dots q_n$ , then  $\{p_0, p_1, \dots, p_n\} = \{q_0, q_1, \dots, q_n\}$ .

*Hint*. For every point x of a simplex S which is not a vertex of S the set  $S \setminus \{x\}$  is not convex.

**1.10.B.** Prove that if  $\mathscr{K}$  and  $\mathscr{L}$  are simplicial complexes such that  $|\mathscr{K}| = |\mathscr{L}|$  and  $\mathscr{K}$  contains an *n*-simplex, then  $\mathscr{L}$  also contains an *n*-simplex.

*Hint.* If m < n, then every *m*-simplex contained in an *n*-simplex S is nowhere dense in S.

1.10.C. Show that if the simplicial complexes  $\mathcal{N}_1$  and  $\mathcal{N}_2$  are nerves of the same finite open cover, then the underlying polyhedra  $|\mathcal{N}_1|$  and  $|\mathcal{N}_2|$  are homeomorphic.

**1.10.D.** Applying the fact that the one-dimensional polyhedron in Fig. 14 on p. 127 cannot be embedded in the plane (see Example 1.11.8), define a finite open cover  $\mathscr{U}$  of the interval I such that  $\operatorname{ord} \mathscr{U} = 1$  and yet  $\mathscr{U}$  has no nerve in  $\mathbb{R}^2$ .

**1.10.E.** (a) Observe that if for every finite open cover  $\mathscr{E}$  of a normal space X there exists an  $\mathscr{E}$ -mapping  $f: X \to K$  to a zero-dimensional polyhedron such that the cover  $\{f^{-1}(y)\}_{y \in Y}$  is a refinement of  $\mathscr{E}$ , then dim  $X \leq 0$ .

(b) For every natural number  $n \ge 2$  define a separable metric space X such that dim X = n and for every finite open cover  $\mathscr{E}$  of the space X there exists an  $\mathscr{E}$ -mapping  $f: X \to I$  such that the cover  $\{f^{-1}(t)\}_{t \in I}$  is a refinement of  $\mathscr{E}$ .

Hint. Apply Problem 1.4.F(b).

1.10.F (Alexandroff [1932]). (a) Let  $f: X \to B^{n+1}$  be an essential mapping of a compact metric space X to the (n+1)-ball (cf. Problem 1.9.A) and  $\varepsilon$ a positive number less than 1. Show that if a continuous mapping  $g: X \to B^{n+1}$  satisfies  $\varrho(f(x), g(x)) < \varepsilon$  for every  $x \in f^{-1}(S^n)$ , then the image g(X) contains the ball of radius  $1-\varepsilon$  concentric with  $B^{n+1}$ .

(b) Apply (a) and Theorem 1.10.6 to show that if a compact metric space X satisfies the inequality dim  $X \le n \ge 0$ , then no continuous mapping  $f: X \to B^{n+1}$  is essential (see Problem 1.9.A).

(c) Deduce from the theorem on  $\mathscr{E}$ -mappings that if a compact metric space X satisfies the equality dim  $X = n \ge 0$ , then there exists an essential mapping  $f: X \to B^n$ .

1.10.G (Alexandroff [1928a], Chogoshvili [1938]). (a) Prove that a compact subspace X of Euclidean m-space  $\mathbb{R}^m$  satisfies the inequality dim  $X \leq n$ if and only if for every polyhedron  $K \subset \mathbb{R}^m$  of dimension m-n-1 and every positive number  $\varepsilon$  there exists an  $\varepsilon$ -translation  $f: X \to \mathbb{R}^m$  such that  $f(X) \cap K = \emptyset$ , or—equivalently—if and only if for every linear (m-n-1)variety  $H \subset \mathbb{R}^m$  and every positive number  $\varepsilon$  there exists an  $\varepsilon$ -translation  $f: X \to \mathbb{R}^m$  such that  $f(X) \cap H = \emptyset$ .

#### Problems

*Hint.* Show that if X is a compact subspace of  $\mathbb{R}^m$  and for every linear (m-n-1)-variety  $H \subset \mathbb{R}^m$  and every positive number  $\varepsilon$  there exists an  $\varepsilon$ -translation  $f: X \to \mathbb{R}^m$  such that  $f(X) \cap H = \emptyset$ , then X is embeddable in  $\mathbb{N}_n^m$ .

(b) Show that in (a) the assumption of compactness of X is essential.

(c) Applying the fact that in  $\mathbb{R}^3$  there exists an Antoine set, i.e., a subspace A homeomorphic to the Cantor set such that for a circle  $S \subset \mathbb{R}^3 \setminus A$  the embedding of S in  $\mathbb{R}^3 \setminus A$  is not homotopic to the constant mapping of S to a point of  $\mathbb{R}^3 \setminus A$  (see Rushing [1973], p. 71), show that in (a) the words "every polyhedron  $K \subset \mathbb{R}^m$ " cannot be replaced by "every compact subspace K of  $\mathbb{R}^m$ ".

Hint. Consider the disk bounded by S.

**1.10.H.** Note that in the case where A is a compact subspace of  $|\mathcal{K}|$ , the sweeping out theorem can be proved in a simpler way.

*Hint.* Sweep out A consecutively from the interiors of all simplexes in  $\mathcal{K}$  which are not contained in A starting with the simplexes of highest dimension.

1.10.1. Prove that the union and the intersection of a finite family of polyhedra contained in a Euclidean space also are polyhedra.

Hint. A bounded subset of  $R^m$  which can be represented as the intersection of a finite family of half-spaces in  $R^m$  is a geometric cell. Define the interior and the boundary of a geometric cell and the geometric dimension of a geometric cell. Prove that polyhedra can be defined as finite unions of geometric cells. To this end, by analogy to the notion of a simplicial complex, introduce the notion of a cellular complex, observe that every polyhedron can be represented as the union of all cells in a cellular complex, and—applying induction with respect to the maximal geometric dimension of cells—prove that every cellular complex has a subdivision which is a simplicial complex (all the details are worked out in Alexandroff and Hopf's book [1935], pp. 124ff.).

1.10.J (Sitnikov [1953]). Modify the construction in Example 1 10.23 to obtain, for every natural number  $n \ge 4$ , an (n-1)-dimensional subspace X of the *n*-cube  $I^n$  such that for every positive number  $\varepsilon$  there exists an  $\varepsilon$ -translation  $f: X \to K$  of X to a polyhedron  $K \subset \mathbb{R}^n$  of dimension  $\le k$ , where k = n/2 if n is even, and k = (n-1)/2 if n is odd.

**1.10.K.** Prove that if X is a bounded subspace of Euclidean *m*-space  $R^m$  and X satisfies the inequality dim  $X \leq n$ , then for every positive number  $\varepsilon$  there exists a finite open cover  $\mathscr{U}$  of the space X such that mesh $\mathscr{U}$  $< \varepsilon$ , ord  $\mathscr{U} \leq n$ , and  $\mathscr{U}$  has a nerve in  $\mathbb{R}^m$ .

1.10.L (Smirnov [1956]). Show that a bounded subspace X of Euclidean *m*-space  $\mathbb{R}^m$  satisfies the inequality  $\mu \dim(X, \varrho) \leq n$ , where  $\mu \dim$  is the metric dimension defined in Section 1.6 and  $\rho$  is the natural metric on  $\mathbb{R}^m$ , if and only if for every positive number  $\varepsilon$  there exists an  $\varepsilon$ -translation f:  $X \to K$  of X onto a polyhedron  $K \subset \mathbb{R}^m$  of dimension  $\leq n$ .

Hint. Apply Problem 1.6.C.

# 1.11. The embedding and universal space theorems

The considerations in the present section largely apply the notion of a function space. Let us recall that if  $(X, \sigma)$  is a non-empty compact metric space and  $(Y, \varrho)$  an arbitrary metric space, then by letting

$$\hat{\varrho}(f,g) = \sup_{x \in X} \varrho(f(x),g(x)) \quad \text{for } f,g \in Y^X$$

we define a metric  $\hat{\varrho}$  on the set  $Y^{X}$  of all continuous mappings of X to Y; the metric space  $(Y^{X}, \hat{\rho})$  obtained in this way is a function space. One readily shows that if  $(X, \sigma)$  is a compact metric space and  $(Y, \rho)$  is a complete metric space, then the function space  $(Y^{\mathbf{x}}, \hat{\rho})$  is complete. As the reader will see, this simple observation leads, via the Baire category theorem, to important applications of function spaces. We begin with three lemmas on function spaces.

**1.11.1. Lemma.** For every positive number  $\varepsilon$  the set of all  $\varepsilon$ -mappings is open in the function space  $Y^{X}$ .

**Proof.** Consider an  $\varepsilon$ -mapping  $f: X \to Y$ . The closed subspace

$$A = \{(x, x') \in X \times X: \ \sigma(x, x') \ge \varepsilon\}$$

of the Cartesian product  $X \times X$  is compact, so that, since for each pair  $(x, x') \in A$  the inequality  $\varrho(f(x), f(x')) > 0$  holds, there exists a positive number  $\delta$  such that

(1) 
$$\varrho(f(x), f(x')) \ge \delta$$
 for each  $(x, x') \in A$ .

To complete the proof it suffices to observe that every continuous mapping  $g: X \to Y$  such that  $\hat{\varrho}(f, g) < \delta/2$  is an  $\varepsilon$ -mapping. Indeed, if g(x) = g(x'), then  $\varrho(f(x), f(x')) < \delta$  which implies by virtue of (1) that  $(x, x') \notin A$ , i.e.,  $\sigma(x, x') < \varepsilon$ ; thus  $\delta(g^{-1}(y)) < \varepsilon$  for every  $y \in Y$ .  $\Box$ 

**1.11.2. Lemma.** For every closed set  $F \subset Y$  the set  $\{f \in Y^X : f(X) \cap F = \emptyset\}$  is open in the function space  $Y^X$ .

**Proof.** Consider a continuous mapping  $f: X \to Y$  such that  $f(X) \cap F = \emptyset$ . As f(X) is a compact subspace of Y, the distance  $\delta = \varrho(f(X), F)$  is positive. To complete the proof it suffices to note that for every continuous mapping  $g: X \to Y$  such that  $\hat{\varrho}(f, g) < \delta$  the relation  $g(X) \cap F = \emptyset$  holds.  $\Box$ 

**1.11.3. Lemma.** If X is a compact metric space such that  $0 \leq \dim X \leq n$ and H is a linear n-variety in  $\mathbb{R}^{2n+1}$ , then for every positive number  $\varepsilon$  the set of all  $\varepsilon$ -mappings of X to  $\mathbb{R}^{2n+1}$  whose values miss H is dense in the function space  $(\mathbb{R}^{2n+1})^X$ ; in particular, the set of all  $\varepsilon$ -mappings of X to  $\mathbb{R}^{2n+1}$ is dense in the function space  $(\mathbb{R}^{2n+1})^X$ .

**Proof.** Consider an arbitrary continuous mapping  $f: X \to Y$  and a positive number  $\delta$ . It follows from the compactness of X that the mapping f is uniformly continuous; thus there exists a positive number  $\eta$  such that  $\delta(f(A)) < \delta$  whenever  $\delta(A) < \eta$ . By virtue of Theorem 1.6.12 there exists a finite open cover  $\mathscr{U} = \{U_i\}_{i=1}^k$  of the space X such that mesh $\mathscr{U}$  $< \min(\varepsilon, \eta)$ ,  $\operatorname{ord} \mathscr{U} \leq n$  and the sets  $U_i$  are non-empty; obviously,  $\delta(f(U_i))$  $< \delta$  for i = 1, 2, ..., k. Choose a point  $q_i \in f(U_i)$  for i = 1, 2, ..., k, and apply Theorem 1.10.4 to obtain a nerve  $\mathscr{N}(\mathscr{U})$  of the cover  $\mathscr{U}$ , consisting of simplexes contained in  $\mathbb{R}^{2n+1}$  and such that  $H \cap N(\mathscr{U}) = \mathscr{O}$ and the vertices  $p_1, p_2, ..., p_k$  of  $\mathscr{N}(\mathscr{U})$  satisfy the inequality  $\varrho(p_i, q_i) < \alpha$ for i = 1, 2, ..., k, where  $\alpha = \min\{\delta - \delta(f(U_i)): i = 1, 2, ..., k\}$ . We have

$$\delta(\{p_i\} \cup f(U_i)) \leq \varrho(p_i, q_i) + \delta(f(U_i)) < \alpha + \delta(f(U_i)) \leq \delta$$
  
for  $i = 1, 2, ..., k$ .

From Theorem 1.10.6 it follows that the  $\varkappa$ -mapping  $\varkappa: X \to \mathbb{R}^{2n+1}$  determined by the cover  $\mathscr{U}$  and the points  $p_1, p_2, \ldots, p_k$  satisfies the inequality  $\hat{\varrho}(f, \varkappa) < \delta$ . To conclude the proof it suffices to observe that by virtue of Theorem 1.10.7 we have  $\varkappa(X) \subset N(\mathscr{U})$  and  $\varkappa^{-1}(\operatorname{St}_{\mathscr{N}(\mathscr{U})}(p_i)) = U_i$  for  $i = 1, 2, \ldots, k$ , so that  $\varkappa(X) \cap H = \emptyset$  and  $\varkappa$  is an  $\varepsilon$ -mapping.  $\Box$ 

The above lemmas will be applied in the proof of the embedding theorem and in the proof of the first universal space theorem.

**1.11.4. The embedding theorem.** Every separable metric space X such that  $0 \leq \dim X \leq n$  is embeddable in Euclidean (2n+1)-space  $R^{2n+1}$ ; if, moreover, the space X is compact, then all homeomorphic embeddings of X in  $R^{2n+1}$  form a  $G_8$ -set dense in the function space  $(R^{2n+1})^X$ .

**Proof.** First, consider a compact metric space X such that  $0 \leq \dim X \leq n$ . Let  $\Phi_i$  denote for i = 1, 2, ... the subset of the function space  $(R^{2n+1})^X$  consisting of all (1/i)-mappings; it follows from Lemmas 1.11.1 and 1.11.3 that the sets  $\Phi_i$  are open and dense in  $(R^{2n+1})^X$ . By virtue of the Baire category theorem, the  $G_{\delta}$ -set  $\Phi = \bigcap_{i=1}^{\infty} \Phi_i$  is dense in the function space  $(R^{2n+1})^X$ . Since the space X is compact, a continuous mapping  $f: X \to R^{2n+1}$  is a homeomorphic embedding if and only if it is a one-to-one mapping, i.e., if f is an  $\varepsilon$ -mapping for every positive number  $\varepsilon$ . Thus  $\Phi$  is the set of all homeomorphic embeddings of X in  $R^{2n+1}$ .

Now, consider a separable metric space X such that  $0 \leq \dim X \leq n$ . By virtue of the compactification theorem, there exists a compact metric space  $\tilde{X}$  which contains a dense subspace homeomorphic to X and satisfies the inequality  $\dim \tilde{X} \leq n$ . As established above,  $\tilde{X}$  is embeddable in  $\mathbb{R}^{2n+1}$ , and so X is also embeddable in  $\mathbb{R}^{2n+1}$ .  $\Box$ 

We shall show in Example 1.11.8 below that the exponent 2n+1 in the embedding theorem cannot be lowered.

**1.11.5. The first universal space theorem.** The subspace  $N_n^{2n+1}$  of Euclidean (2n+1)-space  $R^{2n+1}$  consisting of all points which have at most n rational coordinates is a universal space for the class of all separable metric spaces whose covering dimension is not larger than n.

**Proof.** It follows from Theorem 1.8.5 and the coincidence theorem that  $\dim N_n^{2n+1} = n$ ; hence—by virtue of the compactification theorem—it suffices to prove that every compact metric space X such that  $0 \leq \dim X \leq n$  is embeddable in  $N_n^{2n+1}$ .

The complement  $R^{2n+1} N_n^{2n+1} = L_{n+1}^{2n+1}$  can be represented as the union of a countable family of linear *n*-varieties in  $R^{2n+1}$ , viz., of all sets defined by conditions of the form  $x_{i_1} = r_1$ ,  $x_{i_2} = r_2$ , ...,  $x_{i_{n+1}} = r_{n+1}$ , where  $1 \le i_1 < i_2 < ... < i_{n+1} \le 2n+1$  and  $r_1, r_2, ..., r_{n+1}$  are arbitrary

rational numbers. Arrange all linear *n*-varieties in the family under consideration into a sequence  $H_1, H_2, ...$  and for i = 1, 2, ... denote by  $\Phi_i$  the subset of the function space  $(R^{2n+1})^X$  consisting of all (1/i)-mappings whose values miss  $H_i$ . It follows from Lemmas 1.11.1-1.11.3 that the intersection  $\Phi = \bigcap_{i=1}^{\infty} \Phi_i$  is a  $G_{\delta}$ -set dense in the function space  $(R^{2n+1})^X$ , and from the compactness of X it follows that  $\Phi$  consists of homeomorphic embeddings.  $\Box$ 

Let us observe that the last paragraph incidentally yields another proof of the inequality  $\operatorname{ind} X \leq \dim X$  for compact metric spaces; as announced in Section 1.7, the auxiliary invariant d(X) is the smallest integer  $n \geq 0$  such that X is embeddable in the space  $N_n^{2n+1}$ .

The space  $N_n^{2n+1}$  is called Nöbeling's universal n-dimensional space. Obviously, Nöbeling's universal 0-dimensional space is the space of irrational numbers; thus the last theorem extends to higher dimensions our earlier result that the space of irrational numbers is universal for the class of all zero-dimensional separable metric spaces. As the reader remembers, the Cantor set is another universal space for the same class of spaces. We shall now describe the *n*-dimensional counterpart of the latter universal space, i.e., Menger's universal n-dimensional space  $M_n^{2n+1}$ . The construction will be carried out under more general circumstances: for every pair n, m of integers satisfying  $0 \le n \le m \ge 1$  we shall define a compact subspace  $M_n^m$  of Euclidean m-space  $R^m$ .

For i = 0, 1, 2, ... let  $\mathcal{K}_i$  be the family of  $3^{mi}$  congruent cubes obtained by dividing the *m*-cube  $I^m$  by all linear (m-1)-varieties in  $\mathbb{R}^m$  determined by equations of the form  $x_j = k/3^i$ , where j = 1, 2, ..., m and  $0 \le k \le 3^i$ . For every family  $\mathcal{K}$  of cubes let

 $|\mathscr{K}| = \bigcup \{K: K \in \mathscr{K}\}$  and  $\mathscr{G}_n(\mathscr{K}) = \bigcup \{\mathscr{G}_n(K): K \in \mathscr{K}\},\$ 

where  $\mathscr{S}_n(K)$  is the family of all faces of K which have dimension  $\leq n$ ; moreover, for  $\mathscr{K} \subset \mathscr{K}_i$  let

$$\mathscr{K}' = \{ K \in \mathscr{K}_{i+1} \colon K \subset |\mathscr{K}| \}.$$

For every pair *n*, *m* of integers satisfying  $0 \le n \le m \ge 1$  define inductively a sequence  $\mathscr{F}_0, \mathscr{F}_1, \ldots$  of finite collections of cubes, where  $\mathscr{F}_i \subset \mathscr{K}_i$  for  $i = 0, 1, 2, \ldots$ , and a decreasing sequence  $F_0 \supset F_1 \supset F_2 \supset \ldots$  of closed subsets of  $I^m$  by letting  $\mathscr{F}_0 = \{I^m\}, F_0 = |\mathscr{F}_0| = I^m$  and

$$F_i = |\mathscr{F}_i|, \text{ where } \mathscr{F}_i = \{K \in \mathscr{F}'_{i-1} \colon K \cap |\mathscr{S}_n(\mathscr{F}_{i-1})| \neq \emptyset\},\$$

for i = 1, 2, ... The intersection

$$M_n^m = \bigcap_{i=0}^\infty F_i \subset I^m$$

is a compact subspace of  $I^m$ . One readily sees that the construction of the Cantor set described in Example 1.2.5 is a special case of the above construction, viz., the case where m = 1 and n = 0; thus the space  $M_0^1$  is the Cantor set. One proves that the spaces  $M_0^m$  are all homeomorphic to the Cantor set (see Problem 1.11.D(a)); obviously,  $M_m^m = I^m$  for m = 1, 2, ... In Figs. 12 and 13 the first three steps in constructing  $M_1^2$  and  $M_1^3$  are exhibited.



Let us observe that  $\dim M_n^m = n$ . Indeed, the inequality  $\dim M_n^m \ge n$ follows directly from the inclusion  $|\mathscr{S}_n(\mathscr{F}_0)| \subset M_n^m$  and the reverse inequality is a consequence of Theorem 1.10.12, because the set  $F_i$ , and, a fortiori, the set  $M_n^m$ , can be translated by a  $(\sqrt{m/2} \cdot 3^{i-1})$ -mapping to the space  $|\mathscr{S}_n(\mathscr{F}_{i-1})|$  which has dimension  $\le n$ . The construction of such a mapping is left to the reader; it should be defined separately on each set of the form  $F_i \cap K$ , where  $K \in \mathscr{F}_{i-1}$ .

122

To prove that the space  $M_n^{n+1}$  is a universal space for the class of all separable metric spaces of dimension  $\leq n$ , rather laborious computations are necessary; some of them will be left to the reader. Since the same argumentation yields an interesting theorem exhibiting a relationship between the spaces  $N_n^m$  and  $M_n^m$ , we shall prove this theorem, and then deduce from it the universality property of  $M_n^{2n+1}$ .

**1.11.6. Theorem.** Every compact subspace X of the space  $N_n^m$  is embeddable in the space  $M_n^m$ .

**Proof.** We shall consider on  $I^m$  the metric  $\sigma$  defined by letting

$$\sigma(x, y) = \max\{|x_j - y_j|: j = 1, 2, ..., m\},\$$
  
where  $x = \{x_j\}$  and  $y = \{y_j\};\$ 

obviously, the metric  $\sigma$  is equivalent to the natural metric on  $I^m$ . Let us note that if two points x and y are contained in the same cube  $K \in \mathscr{K}_i$ , then  $\sigma(x, y) \leq 1/3^i$ . For i = 1, 2, ... denote by  $S_i$  the subset of  $I^m$  consisting of all points which have, at least n+1 coordinates of the form  $(k/3^i)+1/2 \cdot 3^i$ , where  $0 \leq k \leq 3^{i-1}$ , and by  $T_i$  the subset of  $I^m$  consisting of all points which have at least m-n coordinates of the form  $k/3^i$ , where  $0 \leq k \leq 3^i$ ; obviously,  $T_i = |\mathscr{S}_n(\mathscr{K}_i)|$ . One readily checks that for i= 0, 1, 2, ...

(2) 
$$\sigma(S_i, T_i) = 1/2 \cdot 3^i$$

and

(3) 
$$F_{i+1} = F_i \setminus B(S_i, 1/2 \cdot 3^{i+1}),$$

where B(A, r) denotes the r-ball about A with respect to the metric  $\sigma$ .

The proof consists in defining by induction a sequence  $f_0, f_1, f_2, ...$ of homeomorphisms of  $I^m$  onto itself which transform the intersection  $I^m \cap N_n^m$  to  $N_n^m$  and map X consecutively to  $F_0, F_1, F_2, ...$ , and which uniformly converge to a homeomorphism that maps X to the intersection  $\bigcap_{i=0}^{\infty} F_i = M_n^m$ . In the inductive step one observes that the set  $f_i(X)$  is disjoint from  $S_i$ , and one modifies  $f_i$  to  $f_{i+1}$  by sweeping out the set  $f_i(X)$ from the ball  $B(S_i, 1/2 \cdot 3^{i+1})$ . The modification of  $f_i$  to  $f_{i+1}$  is performed separately on each coordinate axis and is described by a piecewise linear homeomorphism  $h_i$  of I onto itself. We shall now define inductively a sequence  $h_0, h_1, h_2, ...$  of homeomorphisms of I onto itself and a sequence  $f_0, f_1, f_2, ...$  of homeomorphisms of  $I^m$  onto itself such that for i = 0, 1, 2, ... the following conditions will be satisfied:

(4) The interval I can be divided into finitely many closed intervals with pairwise disjoint interiors in such a way that on each of these intervals h<sub>i</sub> is a linear function with slope ≥ 2/3.

(5) 
$$h_i(k/2\cdot 3^i) = k/2\cdot 3^i \quad \text{for } 0 \le k \le 2\cdot 3^i.$$

(6) 
$$f_{i+1} = H_i f_i$$
, where  $H_i((x_1, x_2, ..., x_m)) = (h_i(x_1), h_i(x_2), ..., h_i(x_m))$ .

(7) 
$$f_i(I^m \cap N_n^m) \subset N_n^m$$

$$(8) f_i(X) \subset F_i$$

Without loss of generality one can assume that  $X \subset I^m$ . Hence, if we let  $f_0 = id_{I^m}$ , conditions (7) and (8) are satisfied for i = 0. Assume that the homeomorphisms  $f_0, f_1, \ldots, f_i$  and  $h_0, h_1, \ldots, h_{i-1}$  with all the required properties are already defined. We shall first define a homeomorphism  $h_i$  which satisfies (4) and (5) and then show that the homeomorphism  $f_{i+1}$  defined in (6) satisfies (7) and (8) with *i* replaced by i+1.

The relations  $f_i(X) \subset f_i(I \cap N_n^m) \subset N_n^m$  and  $S_i \cap N_n^m = \emptyset$  imply that  $f_i(X) \cap S_i = \emptyset$ ; since  $f_i(X)$  is a compact space, there exists a positive rational number  $\varepsilon$  such that

$$0 < \varepsilon < \min\left(\sigma(f_i(X), S_i), \frac{1}{2 \cdot 3^i} - \frac{1}{3^{i+1}}\right).$$

Consider the division of the interval I into closed intervals with pairwise disjoint interiors determined by the points

$$0 = b_0 < a_0 < b_1 < a_1 < \dots < a_{3^{i}-1} < b_{3^{i}} < a_{3^{i}} = 1,$$

where

$$a_k = \frac{k}{3^i} + \frac{1}{2 \cdot 3^i} - \varepsilon$$
 and  $b_{k+1} = \frac{k}{3^i} + \frac{1}{2 \cdot 3^i} + \varepsilon$ 

for  $k = 0, 1, ..., 3^{i} - 1$ . The reader can easily check that the functions  $g'_{0}, g'_{1}, ..., g'_{3^{i}}$ , where

$$g'_k(t) = \frac{1}{3^{i+1}} \left( \frac{1}{2 \cdot 3^i} - \varepsilon \right)^{-1} \left( \frac{1}{t} - \frac{k}{3^i} \right) + \frac{k}{3^i} \quad \text{for } b_k \leq t \leq a_k,$$

and the functions  $g''_0, g''_1, \dots, g''_{3^{t-1}}$ , where

$$g_k''(t) = \varepsilon^{-1} \left( \frac{1}{2 \cdot 3^i} - \frac{1}{3^{i+1}} \right) \left[ t - \left( \frac{k}{3^i} + \frac{1}{2 \cdot 3^i} \right) \right] + \frac{k}{3^i} + \frac{1}{2 \cdot 3^i}$$
  
for  $a_k \le t \le b_{k+1}$ ,

satisfy the equalities:

$$g'_k(a_k) = g''_k(a_k) = \frac{k}{3^i} + \frac{1}{3^{i+1}},$$

(9).

$$g_k''(b_{k+1}) = g_{k+1}'(b_{k+1}) = \frac{k}{3^i} + \frac{1}{3^i} - \frac{1}{3^{i+1}}$$

for  $k = 0, 1, ..., 3^{i} - 1$ . As  $g'_{0}(0) = 0$  and  $g'_{3i}(1) = 1$ , the functions  $g'_{k}$  and  $g''_{k}$  determine a homeomorphism  $h_{i}$  of *I* onto itself which, as one easily checks, satisfies (4) and (5).

Consider now the homeomorphism  $f_{i+1}$  defined in (6). Since  $\varepsilon$  is a rational number, the homeomorphism  $h_i$  transforms each rational number in *I* into a rational number, so that by virtue of (7) we have  $f_{i+1}(I^m \cap N_n^m) \subset N_n^m$ .

Let x be a point in X; by virtue of (8),  $f_i(x) = (t_1, t_2, ..., t_m) \in F_i$ . Since  $\sigma(f_i(X), S_i) > \varepsilon$ , there exist m-n coordinates of the point  $(t_1, t_2, ..., t_m)$ , say  $t_{j_1}, t_{j_2}, ..., t_{j_{m-n}}$ , and m-n non-negative integers  $k_1, k_2, ..., k_{m-n} \leq 3^i$  such that

$$\left| t_{j_l} - \frac{k_l}{3^i} \right| < \frac{1}{2 \cdot 3^i} - \varepsilon \quad \text{for } l = 1, 2, ..., m - n$$

The last inequality means that  $b_{k_l} \leq t_{j_l} \leq a_{k_l}$ , which together with (9) yield

$$\frac{k_l}{3^i} - \frac{1}{3^{i+1}} \le h_i(t_{j_1}) \le \frac{k_l}{3^i} + \frac{1}{3^{i+1}} \quad \text{for } l = 1, 2, ..., m - n,$$

so that

(10) 
$$f_{i+1}(x) = (h_i(t_1), h_i(t_2), \dots, h_i(t_m)) \in B(T_i, 1/3^{i+1}).$$

From (2) and (10) it follows that

(11) 
$$f_{i+1}(x) \notin B(S_i, 1/2 \cdot 3^{i+1})$$

Since, by virtue of (5), the homeomorphism  $H_i$  maps each cube in  $\mathscr{K}_i$  onto itself,  $f_{i+1}(x) \in F_i$ ; the last relation together with (11) and (3) show that  $f_{i+1}(x) \in F_{i+1}$ . Hence we have  $f_{i+1}(X) \subset F_{i+1}$ .

It remains to check that the sequence of homeomorphisms  $f_0, f_1, f_2, ...$ uniformly converges to a homeomorphism f of  $I^m$  onto itself; the inclusion  $f(X) \subset M_n^m$  will then follow from (8). This amounts to checking that the sequence  $h_0, h_1 h_0, h_2 h_1 h_0, ...$  of homeomorphisms of I onto itself uniformly converges to a homeomorphism of I onto itself. The last fact can be deduced from (4) and (5) by a straightforward computation, which we leave to the reader.  $\Box$ 

**1.11.7. The second universal space theorem.** The compact subspace  $M_n^{2n+1}$  of Euclidean (2n+1)-space  $R^{2n+1}$  is a universal space for the class of all separable metric spaces whose covering dimension is not larger than n.

**Proof.** As observed above,  $\dim M_n^{2n+1} = n$ ; hence—by virtue of the compactification theorem—it suffices to prove that every compact metric space X such that  $\dim X \leq n$  is embeddable in  $M_n^{2n+1}$ . This is, however, an immediate consequence of Theorems 1.11.5 and 1.11.6.  $\Box$ 

In connection with the above universal space theorems one can ask whether there exists a universal space for the class of all subspaces of  $R^m$ which have dimension  $\leq n$ . It is an old hypothesis that  $M_n^m$  is such a space. Quite recently, it was proved that  $M_n^m$  is indeed a universal space for the class of all compact subspaces of  $R^m$  which have dimension  $\leq n$ ; it is a very deep and difficult result. Hence, the hypothesis on the universality of  $M_n^m$  is now reduced to the question whether every subspace X of Euclidean m-space  $R^m$  has a dimension preserving compactification embeddable in  $R^m$ . Let us observe that in some special cases the universality of the space  $M_n^m$  can be deduced from our earlier results. Indeed, the universality of  $M_0^m$  and  $M_m^m$  is obvious, and the universality of  $M_n^{2n+1}$  is a consequence of Theorem 1.11.7; finally, Theorems 1.8.10, 1.8.8, and 1.11.6, together with the simple observation that each closed subset of  $N_{m-1}^m$  has a compactification embeddable in  $N_{m-1}^m$ , imply that the space  $M_{m-1}^m$  is universal for the class of all subspaces of  $R^m$  which have dimension  $\leq m-1$  (a direct proof of this fact is outlined in the hint to Problem 1.11.D(c)).

The considerations of the preceding paragraph imply, in particular, that the space  $N_n^m$  is embeddable in the space  $M_n^m$  for n = 0, n = m, m = 2n+1 and n = m-1; the problem whether  $N_n^m$  is always embeddable in  $M_n^m$  is still open. On the other hand, the space  $M_n^m$  is embeddable in the space  $N_n^m$  for every pair of integers n, m satisfying  $0 \le n \le m \ge 1$  (see Problem 1.11.E). Hence the question if  $N_n^m$  is a universal space for the class of all subspaces of  $\mathbb{R}^m$  which have dimension  $\leq n$  is a weaker version of the problem of the universality of the space  $M_n^m$ .

To conclude, let us note that from the above discussion of special cases it follows that for every pair of integers n, m satisfying  $0 \le n \le m \le 3$  and  $m \ge 1$ , the spaces  $M_n^m$  and  $N_n^m$  are universal for the class of all subspaces of  $R^m$  which have dimension  $\le n$ ; in particular, for every such pair n, m, the space  $N_n^m$  is embeddable in the space  $M_n^m$  and each *n*-dimensional subspace of  $R^m$  has an *n*-dimensional compactification embeddable in  $R^m$ .

We close this section by showing that the exponent 2n + 1 in the embedding theorem cannot be lowered.

**1.11.8. Example.** Let  $K_1$  be the union of all 1-faces of the 4-simplex  $p_0p_1p_2p_3p_4$  (see Fig. 14). Applying the Jordan curve theorem, which states that every simple closed curve (i.e., a set homeomorphic to  $S^1$ ) in the plane  $R^2$  separates  $R^2$  into two regions, we shall show that the one-dimensional polyhedron  $K_1$  cannot be embedded in  $R^2$ .



Assume that there exists a homeomorphic embedding  $f: K_1 \to R^2$ and define  $a_i = f(p_i)$  for  $0 \le i \le 4$ . It follows from the Jordan curve theorem that the simple closed curve  $S_1 = f(p_0p_1 \cup p_1p_2 \cup p_2p_0)$  separates  $R^2$  into two regions. One can suppose that the point  $a_3$  belongs to the bounded component U of  $R^2 \setminus S_1$ , otherwise one should replace the homeomorphic embedding f by the composition of f and a suitable inversion. Since the points  $a_3$  and  $a_4$  can be joined in  $f(K_1)$  by an arc (i.e., a set homeomorphic to I) disjoint to  $S_1$ , we have  $a_4 \in U$ . The Jordan curve theorem implies that the set  $U \setminus T$ , where  $T = f(p_0p_3 \cup p_1p_3 \cup p_2p_3)$ , can be represented as the union of three pairwise disjoint regions  $U_0$ ,  $U_1$ ,  $U_2$  such that  $a_i \notin U_i$  for i = 0, 1, 2. As  $a_4 \notin T$ , we have  $a_4 \in U \setminus T$  and without loss of generality we can suppose that  $a_4 \in U_0$ . Now, the set  $f(p_0p_4)$  is an arc joining the points  $a_0, a_4$  and disjoint to the simple closed curve  $S_2 = f(p_1p_2 \cup p_1p_3 \cup p_2p_3)$ . Since the points  $a_0$  and  $a_4$  are contained in distinct components of  $R^2 \setminus S_2$ , the assumption that  $K_1$  is embeddable in  $R^2$  yields a contradiction.  $\Box$ 



One can prove that the union  $K_n$  of all *n*-faces of a (2n+2)-simplex cannot be embedded in  $\mathbb{R}^{2n}$  for any natural number *n*; a proof of this fact, based on the Borsuk-Ulam antipodal theorem, is outlined in the hint to Problem 1.11.F.

### Historical and bibliographic notes

The first part of Theorem 1.11.4 was formulated, for compact metric spaces, by Menger in [1926] and was proved there for n = 1. In [1928] Menger again proved the theorem for n = 1 and hinted at the modifications in the proof that should permit us to obtain the theorem in full generality. For an arbitrary n the first part of the embedding theorem was proved simultaneously by Nöbeling in [1931], Pontrjagin and Tolstowa in [1931], and Lefschetz in [1931]; the three proofs consisted in constructing a sequence of continuous mappings uniformly converging to a homeomorphic embedding and were rather involved. The present proof was given by Hurewicz in [1933] (announcement in [1931]); application of function spaces yielded the stronger result about the set of all homeomorphic embeddings. The consideration of function spaces and resorting to the Baire category theorem (proofs by category method) proved very useful in the dimension theory of separable metric spaces. This idea, originated by Hurewicz in [1931], was repeatedly exploited by Hurewicz, Kuratowski

### Problems

(cf. Problem 1.7.D) and their followers. Theorem 1.11.5 was established by Nöbeling in [1931]. The spaces  $M_n^m$  were introduced by Menger in [1926]. They are generalizations of the Cantor set and of Sierpiński's universal curve, i.e., the space  $M_1^2$  described by Sierpiński in [1916], where it was also proved that  $M_1^2$  is a universal space for the class of all compact subspaces of the plane which have an empty interior (in [1922] Sierpiński observed that the assumption of compactness is not essential). In [1926] Menger proved that the space  $M_1^3$  is universal for the class of all compact metric spaces of dimension  $\leq 1$ , observed that Sierpiński's argument yields the universality of  $M_{m-1}^m$  for the class of all compact subspaces of  $\mathbb{R}^m$  which have dimension  $\leq m-1$ , announced Theorem 1.11.7 for compact spaces and put forward the hypothesis that the space  $M_n^m$  is universal for the class of all compact subspaces of  $R^m$  which have dimension  $\leq n$ . Theorem 1.11.6 was proved by Bothe in [1963]. In [1931] Lefschetz defined for every pair n, m of integers satisfying  $0 \le n \le m \ge 1$  a compact subspace  $S_n^m$  of  $R^m$  which is very much like  $M_n^m$  (the difference consists in considering simplexes rather than cubes) and proved that  $S_n^{2n+1}$  is a universal space for the class of all separable metric spaces whose covering dimension is not larger than n; he also proved there that the space  $S_n^m$ is embeddable in the space  $N_n^m$ . It is a general belief that the spaces  $M_n^m$ and  $S_n^m$  are homeomorphic, but no proof was ever produced. Hence, Lefschetz is considered to be the author of Theorem 1.11.7, although-formally-the theorem was first proved by Bothe in [1963]. Let us mention, by way of digression, that  $M_n^m$  and  $S_n^m$  are obviously homeomorphic if n = 0 or n = m, that  $M_1^2$  is homeomorphic to  $S_1^2$  and  $M_1^3$  is homeomorphic to  $S_1^3$  by virtue of topological characterizations of  $M_1^2$  and  $M_1^3$  given by Whyburn in [1958] and by Anderson in [1958], respectively, and finally that, as proved by Cannon in [1973],  $M_{m-1}^m$  and  $S_{m-1}^m$  are homeomorphic if  $m \neq 4$ . The theorem stating that  $M_n^m$  is a universal space for the class of all compact subspaces of  $\mathbb{R}^m$  which have dimension  $\leq n$  was proved by Štan'ko in [1971]. A proof of the Jordan curve theorem can be found in Kuratowski's book [1968], p. 510.

### Problems

**1.11.A** (Kuratowski [1937a], Hurewicz and Wallman [1941]). (a) Check that if  $(X, \sigma)$  is a non-empty metric space and  $(Y, \varrho)$  is a compact metric space then the function  $\hat{\varrho}$  defined at the beginning of this section is

a metric on the set  $Y^x$  of all continuous mappings of X to Y and the function space  $(Y^x, \hat{\varrho})$  is complete. Verify that for every open cover  $\mathscr{E}$  of the space X the set of all  $\mathscr{E}$ -mappings is open in the function space  $Y^x$ .

(b) Show that for every separable metric space X there exists a sequence  $\mathscr{E}_1, \mathscr{E}_2, \ldots$  of finite open covers of X such that every continuous mapping  $f: X \to Y$  of X onto a topological space Y which is an  $\mathscr{E}_i$ -mapping for  $i = 1, 2, \ldots$  is a homeomorphism.

*Hint*. Consider a countable base  $\mathscr{B}$  for the space X and arrange into a sequence all covers of the form  $\{W, X \setminus \widetilde{U}\}$ , where  $U, W \in \mathscr{B}$  and  $\widetilde{U} \subset W$ .

(c) Prove that for every finite open cover  $\mathscr{E}$  of a separable metric space X such that  $0 \leq \dim X \leq n$  the set of all  $\mathscr{E}$ -mappings of X to  $I^{2n+1}$  is dense in the function space  $(I^{2n+1})^X$ .

(d) Deduce from (a), (b) and (c) that every separable metric space X such that  $0 \leq \dim X \leq n$  is embeddable in the (2n+1)-cube  $I^{2n+1}$ ; observe that the set of all homeomorphic embeddings of X in  $I^{2n+1}$  contains a  $G_{\delta}$ -set dense in the function space  $(I^{2n+1})^X$ .

*Remark.* As opposed to the case where X is a compact space, the set of all homeomorphic embeddings of a separable metric space X such that  $0 \leq \dim X \leq n$  in  $I^{2n+1}$  is generally not a  $G_{\delta}$ -set (see Roberts [1948]).

**1.11.B** (Kuratowski [1937a], Hurewicz and Wallman [1941]). (a) Check that for every closed subset F of a compact metric space Y the set  $\{f \in Y^X: f(X) \cap F = \emptyset\}$  is open in the function space  $Y^X$ .

(b) Prove that for every finite open cover  $\mathscr{E}$  of a separable metric space X such that  $0 \leq \dim X \leq n$  and every linear *n*-variety H in  $\mathbb{R}^{2n+1}$  the set of all  $\mathscr{E}$ -mappings of X to  $I^{2n+1}$  whose values miss H is dense in the function space  $(I^{2n+1})^X$ .

(c) Prove that for every separable metric space X such that  $\dim X \leq n \geq 0$  there exists a homeomorphic embedding  $f: X \to I^{2n+1}$  of X in  $I^{2n+1}$  which satisfies the inclusion  $\overline{f(X)} \subset N_n^{2n+1}$ . Observe that this fact implies the compactification theorem.

**1.11.C.** (a) Show that for every continuous mapping  $f: X \to Y$  of a separable metric space X to a separable metric space Y and for every compact metric space  $\tilde{Y}$  that contains Y there exists a compact metric space  $\tilde{X}$  that contains X as a dense subset and a continuous mapping  $\tilde{f}: \tilde{X} \to \tilde{Y}$  such that  $\tilde{f}|X = f$  (cf. Lemma 1.13.3).

### Problems

*Hint.* Consider the completion  $\tilde{X}$  of the space X with respect to the metric  $\tilde{\varrho}$  defined by letting  $\tilde{\varrho}(x, y) = \varrho(x, y) + \sigma(f(x), f(y))$ , where  $\varrho$  is a totally bounded metric on the space X and  $\sigma$  is an arbitrary metric on the space  $\tilde{Y}$ .

(b) Let  $S = \{X_i, \pi_i^i\}$  be an inverse sequence of completely metrizable separable spaces. Prove that if  $\pi_i^{i+1}(X_{i+1})$  is a dense subset of  $X_i$  for i = 1, 2, ..., then  $\lim S \neq \emptyset$ .

*Hint.* Apply (a) to define an inverse sequence  $\tilde{S} = \{\tilde{X}_i, \tilde{\pi}_j^i\}$  of compact metric spaces such that  $X_i$  is a dense subset of  $\tilde{X}_i$  and  $\tilde{\pi}_i^{i+1}|X_{i+1} = \pi_i^{i+1}$  for i = 1, 2, ... Show that for every *i* the inverse image  $G_i = \tilde{\pi}_i^{-1}(X_i) \subset \tilde{X} = \underline{\lim} \tilde{S}$ , where  $\tilde{\pi}_i \colon \tilde{X} \to \tilde{X}_i$  denotes the projection, is a  $G_0$ -set dense in the space  $\tilde{X}$ ; use the fact that  $\tilde{\pi}_i$  maps  $\tilde{X}$  onto  $\tilde{X}_i$  (see [GT], Corollary 3.2.15).

(c) Prove that for every continuous mapping  $f: X \to R^{2n+1}$  of a separable metric space X such that  $\dim X \leq n$  to  $R^{2n+1}$  and for every positive number  $\varepsilon$  there exists a homeomorphic embedding  $g: X \to R^{2n+1}$  such such that  $\varrho(f(x), g(x)) < \varepsilon$  for every  $x \in X$ .

Hint. Let  $Z = R^{2n+1} \cup \{\omega\}$  be the one-point compactification of Euclidean (2n+1)-space  $R^{2n+1}$ . Define a compact metric space  $\tilde{X}$  that contains X as a dense subset and satisfies the inequality dim  $\tilde{X} \leq n$ , and a continuous mapping  $\tilde{f}: \tilde{X} \to Z$  such that  $\tilde{f}|X = f$ . For i = 1, 2, ... let  $X_i$  $= \tilde{f}^{-1}(B_i) \subset \tilde{X}$ , where  $B_i = \overline{B(0, i)} \subset R^{2n+1}$ ; assume that  $X_i \neq \emptyset$  and consider the subspace  $\Phi_i$  of the function space  $(R^{2n+1})^{X_i}$  consisting of all homeomorphic embeddings  $g: X_i \to R^{2n+1}$  such that  $\varrho(g(x), \tilde{f}(x)) < \varepsilon$ for every  $x \in X_i$ . Apply (b) to the inverse sequence  $S = \{\Phi_i, \pi_j^i\}$ , where  $\pi_j^i(g) = g|X_j$  for every pair i, j of natural numbers satisfying  $j \leq i$ .

**1.11.D.** (a) Show that for every natural number  $m \ge 1$  the space  $M_0^m$  is homeomorphic to the Cantor set.

Hint. Apply Problem 1.3.F.

(b) (Sierpiński [1916] and [1922]) Prove that  $M_1^2$  is a universal space for the class of all subspaces of the plane which have dimension  $\leq 1$ .

*Hint*. For a compact one-dimensional subspace X of the plane define a subspace of  $R^2$  which contains X and is homeomorphic to  $M_1^2$ . To this end, consider a rectangle containing X and remove from it smaller rectangles disjoint from X in the same way as one removes squares from  $I^2$  to obtain  $M_1^2$  (see Fig. 16, where X has the shape of the letter a). Apply Theorem 1.8.9 to extend the result to all one-dimensional subspaces of the plane.

(c) (Menger [1926] for compact spaces) Prove that the space  $M_{m-1}^m$  is a universal space for the class of all subspaces of  $R^m$  which have dimen-

sion  $\leq m-1$ .

Hint. Modify the construction described in the hint to part (b).



**1.11.E.** Prove that the space  $M_n^m$  is embeddable in the space  $N_n^m$  for every pair of integers m, n satisfying  $0 \le n \le m \ge 1$ .

Hint. Apply Problem 1.8.C.

1.11.F (Flores [1935]). (a) Let  $p_0, p_1, \ldots, p_{2n+2}$  be the vertices of a regular (2n+2)-simplex  $T^{2n+2}$  inscribed in the (2n+1)-sphere  $S^{2n+1} \subset R^{2n+2}$  and let  $q_i = -p_i$  for  $i = 0, 1, \ldots, 2n+2$ . Check that for each sequence  $i_0 < i_1 < \ldots < i_{2n+1} \leq 2n+2$  of 2n+2 non-negative integers, the system of points  $p_{i_0}, p_{i_1}, \ldots, p_{i_n}, q_{i_{n+1}}, \ldots, q_{i_{2n+1}} \in R^{2n+2}$  is linearly independent and the linear (2n+1)-variety in  $R^{2n+2}$  spanned by these points does not contain the origin. Denote by  $S_1^{2n+1}$  the union of all (2n+1)simplexes of the form  $p_{i_0}p_{i_1} \ldots p_{i_n}q_{i_{n+1}} \ldots q_{i_{2n+1}}$  and show that the projection p of  $S_1^{2n+1}$  from the origin onto  $S^{2n+1}$  is a homeomorphism. Observe that p(-x) = -p(x) for every  $x \in S_1^{2n+1}$ .

(b) Let  $\mathscr{K}_n$  be the family of all faces of  $T^{2n+2}$  which have dimension  $\leq n$  and let  $K_n = |\mathscr{K}_n|$ . Consider the cone  $C(K_n)$  over  $K_n$  with vertex at the origin, i.e., the subset of  $R^{2n+2}$  consisting of all points of the form tx, where  $x \in K_n$  and  $0 \leq t \leq 1$ , and the subspace  $S_2^{2n+1}$  of the Cartesian product  $C(K_n) \times C(K_n)$  consisting of all pairs (x, ty) and (tx, y), where  $0 \leq t \leq 1$ ,  $x \in T_1$ ,  $y \in T_2$  and  $T_1$ ,  $T_2$  are disjoint members of  $\mathscr{K}_n$ . Show that by mapping, in a linear way, every segment with end-points (x, 0) and (x, y) contained in  $S_2^{2n+1}$  onto the segment with end-points (0, y) and (x, y) contained in  $S_2^{2n+1}$  onto the segment with end-points -y and  $\frac{1}{2}(x-y)$  contained in  $S_2^{2n+1}$ , one obtains a homeomorphism h of  $S_2^{2n+1}$  onto  $S_1^{2n+1}$ . Observe that h(x, y) = -h(y, x).

(c) Applying the Borsuk-Ulam antipodal theorem, i.e., the fact that

132

133

for every continuous mapping  $g: S^n \to R^n$  there exists a point  $x \in S^n$  such that g(x) = g(-x) (see Spanier [1966], p. 266), prove that the *n*-dimensional polyhedron  $K_n$  defined in (b) cannot be embedded in  $R^{2n}$  for any  $n \ge 1$ .

*Hint.* Assume that there exists a homeomorphic embedding  $f: K_n \to R^{2n}$ . Observe that f determines a homeomorphic embedding  $f_1: C(K_n) \to R^{2n+1}$ , consider the mapping  $f_2: S_2^{2n+1} \to R^{2n+1}$  defined by letting  $f_2(x, y) = f_1(y) - -f_1(x)$  and the composition  $g = f_2 h^{-1} p^{-1}: S^{2n+1} \to R^{2n+1}$ .

# 1.12. Dimension and mappings

We shall now study the relations between the dimensions of the domain and the range of a continuous mapping. Let us begin with the observation that since one-to-one continuous mappings onto can arbitrarily raise or lower the dimension (see Problems 1.3.C and 1.4.F(b)), to obtain sound results we have to restrict ourselves to special classes of mappings. We find that for closed mappings and open mappings many interesting results can be obtained.

Let us recall that a continuous mapping  $f: X \to Y$  is closed (open) if for every closed (open) set  $A \subseteq X$ , the image f(A) is closed (open) in Y. One readily checks that if  $f: X \to Y$  is a closed (an open) mapping, then for every closed (open) subset A of X the restriction  $f|A: A \to f(A) \subset Y$  is a closed (an open) mapping; similarly, if  $f: X \to Y$  is a closed (an open) mapping, then for an arbitrary subset B of Y the restriction  $f_B: f^{-1}(B) \to B$ is a closed (an open) mapping. Clearly, a mapping  $f: X \to Y$  is closed if and only if  $f(\overline{A}) = \overline{f(A)}$  for every  $A \subset X$ , so that each continuous mapping of a compact space to a Hausdorff space is closed.

We shall first discuss closed mappings and begin with the theorem on dimension-raising mappings. In the lemma to this theorem the notion of a network appears; a family  $\mathcal{N}$  of subsets of a topological space Xis a *network* for X if for every point  $x \in X$  and each neighbourhood U of the point x there exists an  $M \in \mathcal{N}$  such that  $x \in M \subset U$ . The definition of a network imitates the definition of a base, only one does not require the members of a network to be open sets. Clearly, every base for a topological space is a network for that space. The family of all one-point subsets is another example of a network.

**1.12.1. Lemma.** A separable metric space X satisfies the inequality  $\operatorname{ind} X \leq n \geq 0$  if and only if X has a countable network  $\mathcal{N}$  such that  $\operatorname{ind} \operatorname{Fr} M \leq n-1$  for every  $M \in \mathcal{N}$ .

**Proof.** By virtue of Theorem 1.1.6, it suffices to show that if a separable metric space X has a network  $\mathcal{N} = \{M_i\}_{i=1}^{\infty}$  such that  $\operatorname{ind} \operatorname{Fr} M_i \leq n-1$  for i = 1, 2, ..., then  $\operatorname{ind} X \leq n$ . Let  $Y = \bigcup_{i=1}^{\infty} \operatorname{Fr} M_i$  and  $Z = X \setminus Y$ . It follows from the sum theorem that  $\operatorname{ind} Y \leq n-1$ ; we shall show that  $\operatorname{ind} Z \leq 0$ . For an arbitrary point  $x \in Z$  and a neighbourhood  $V \subset X$  of the point x there exists an  $M_i \in \mathcal{N}$  such that  $x \in M_i \subset V$ . Since  $x \in X \setminus Y$   $\subset X \setminus \operatorname{Fr} M_i$ , we have  $x \in U = \operatorname{Int} M_i \subset V$ . The inclusion  $\operatorname{Fr} \operatorname{Int} M_i$   $\subset \operatorname{Fr} M_i$  implies that  $Z \cap \operatorname{Fr} U = \emptyset$  and thus we have  $\operatorname{ind} Z \leq 0$ . The inequality  $\operatorname{ind} X \leq n$  now follows from Lemma 1.5.2.  $\Box$ 

**1.12.2. Theorem on dimension-raising mappings.** If  $f: X \to Y$  is a closed mapping of a separable metric space X onto a separable metric space Y and there exists an integer  $k \ge 1$  such that  $|f^{-1}(y)| \le k$  for every  $y \in Y$ , then ind  $Y \le ind X - (k-1)$ .

**Proof.** We can suppose that  $0 \le \operatorname{ind} X < \infty$ . We shall apply induction with respect to the number n+k, where  $n = \operatorname{ind} X$ . If n+k = 1, we have k = 1, so that f is a homeomorphism and the theorem holds. Assume that the theorem holds whenever  $n+k < m \ge 2$  and consider a closed mapping  $f: X \to Y$  such that f(X) = Y and n+k = m.

Let  $\mathscr{B}$  be a countable base for X such that  $\operatorname{ind} \operatorname{Fr} U \leq n-1$  for every  $U \in \mathscr{B}$ . Consider an arbitrary  $U \in \mathscr{B}$ ; by the closedness of f we have

(1) 
$$\operatorname{Fr} f(U) = \overline{f(U)} \cap \overline{Y \setminus f(U)} \subset f(\overline{U}) \cap f(X \setminus U)$$
$$= [f(U) \cup f(\operatorname{Fr} U)] \cap f(X \setminus U) \subset f(\operatorname{Fr} U) \cup B,$$

where  $B = f(U) \cap f(X \setminus U)$ . Since the restriction  $f | \operatorname{Fr} U$ :  $\operatorname{Fr} U \to f(\operatorname{Fr} U)$  is a closed mapping, it follows from the inductive assumption that

$$ind f(Fr U) \leq (n-1) + (k-1) = n+k-2.$$

Assume that  $B \neq \emptyset$ . Consider the restriction  $f_B: f^{-1}(B) \to B$  and the restriction  $f' = f_B|(X \setminus U): (X \setminus U) \cap f^{-1}(B) \to B$ ; both  $f_B$  and f' are closed, and the fibres of f' all have cardinality  $\leq k-1$ , because  $f^{-1}(y) \cap U \neq \emptyset$  for every  $y \in B$ . It follows from the inductive assumption that

$$ind B \leq n + (k-1) - 1 = n + k - 2.$$

As U is an  $F_{\sigma}$ -set in X, both f(U) and B are  $F_{\sigma}$ -sets in Y; applying Corollary 1.5.4, we obtain the inequality  $\operatorname{ind}[f(\operatorname{Fr} U) \cup B] \leq n+k-2$ . From the last inequality and from (1) it follows that  $\operatorname{ind}\operatorname{Fr} f(U) \leq n+k-2$  for

every  $U \in \mathscr{B}$ ; the same inequility holds if  $B = \emptyset$ . One readily checks that the family  $\mathscr{N} = \{f(U): U \in \mathscr{B}\}$  is a network for the space Y, so that  $\operatorname{ind} Y \leq n+k-1 = \operatorname{ind} X+k-1$  by virtue of Lemma 1.12.1.  $\Box$ 

We now pass to the theorem on dimension-lowering mappings. The theorem will be preceded by a lemma which, roughly speaking, shows that in condition (MU2), in the definition of the dimension function ind, points can be replaced by closed sets of small dimension.

**1.12.3. Lemma.** If a separable metric space X has a closed cover  $\{A_s\}_{s \in S}$  such that ind  $A_s \leq m \geq 0$  for each  $s \in S$  and if for every  $s \in S$  and each open set  $V \subset X$  that contains  $A_s$  there exists an open set  $U \subset X$  such that

 $A_s \subset U \subset \widehat{U} \subset V$  and  $\operatorname{ind} \operatorname{Fr} U \leq m-1$ ,

then  $\operatorname{ind} X \leq m$ .

**Proof.** By virtue of Theorem 1.9.3 it suffices to show that for every closed subspace A of the space X and each continuous mapping  $f: A \to S^m$  there exists a continuous extension  $F: X \to S^m$  of f over X. It follows from Theorem 1.9.2 that for each  $s \in S$  the mapping f is continuously extendable over  $A \cup A_s$ , so that there exists an open set  $V_s \subset X$  containing  $A \cup A_s$  such that f is continuously extendable over  $V_s$ . Consider an open set  $U_s \subset X$  satisfying

(2) 
$$A_s \subset U_s \subset \overline{U_s} \subset V_s$$
 and  $\operatorname{ind} \operatorname{Fr} U_s \leq m-1$ ;

obviously, f is continuously extendable over  $A \cup \overline{U}_s$ . The open cover  $\{U_s\}_{s \in S}$ of the space X has a countable subcover  $\{U_{s_j}\}_{j=1}^{\infty}$ . We shall inductively define a sequence  $F_1, F_2, \ldots$  of continuous mappings, where  $F_i: A \cup \bigcup_{j=1}^{i} \overline{U}_{s_j} \rightarrow S^m$ , such that

(3) 
$$F_i|(A \cup \bigcup_{j=1}^{i-1} \overline{U}_{s_j}) = F_{i-1} \text{ for } i > 1.$$

Let  $F_1$  be an arbitrary continuous extension of f over  $A \cup \overline{U}_{s_1}$ . Assume that the mappings  $F_i$  satisfying (3) are defined for i < k. The set  $A \cup \bigcup_{j=1}^k \overline{U}_{s_j}$ can be represented as the union of two closed sets

$$A' = A \cup \bigcup_{j=1}^{k-1} \overline{U}_{s_j}$$
 and  $A'' = A \cup (\overline{U}_{s_k} \setminus \bigcup_{j=1}^{k-1} U_{s_j}).$ 

The mapping f is extendable to a continuous mapping  $f'': A'' \to S^m$  and

$$D = \{x \in A' \cap A'' \colon F_{k-1}(x) \neq f''(x)\} \subset \bigcup_{j=1}^{k-1} \overline{U}_{s_j} \cap (X \setminus \bigcup_{j=1}^{k-1} U_{s_j}) \subset \bigcup_{j=1}^{k-1} \operatorname{Fr} U_{s_j},$$

so that  $\operatorname{ind} D \leq m-1$  by virtue of (2) and the sum theorem. It follows from Lemma 1.9.6 that the mappings  $F_{k-1}|A' \cap A''$  and  $f''|A' \cap A''$  are homotopic. Since the mapping  $f''|A' \cap A''$  is continuously extendable over A'', it follows from Lemma 1.9.7 that the mapping  $F_{k-1}|A' \cap A''$  is extendable to a continuous mapping  $F'': A'' \to S^m$ .

Letting

$$F_k(x) = \begin{cases} F_{k-1}(x) & \text{for } x \in A', \\ F''(x) & \text{for } x \in A'', \end{cases}$$

we define a continuous mapping  $F_k$  of  $A' \cup A'' = A \cup \bigcup_{j=1}^k \overline{U}_{s_j}$  to  $S^m$ , which satisfies (3) for i = k.

As  $X = \bigcup_{i=1}^{\infty} U_{s_i}$ , the formula

$$F(x) = F_i(x)$$
 for  $x \in U_{s_i}$ 

defines a continuous mapping  $F: X \to S^m$ , which is the required extension of f over X.  $\Box$ 

**1.12.4. Theorem on dimension-lowering mappings.** If  $f: X \to Y$  is a closed mapping of a separable metric space X to a separable metric space Y and there exists an integer  $k \ge 0$  such that  $\operatorname{ind} f^{-1}(y) \le k$  for every  $y \in Y$ , then  $\operatorname{ind} X \le \operatorname{ind} Y + k$ .

**Proof.** We can suppose that  $\operatorname{ind} Y < \infty$ . We shall apply induction with respect to  $n = \operatorname{ind} Y$ . If n = -1, we have  $Y = \emptyset$  and  $X = \emptyset$ , and so the theorem holds. Assume that the theorem holds for closed mappings to spaces of dimension less than  $n \ge 0$  and consider a closed mapping  $f: X \to Y$  to a space Y such that  $\operatorname{ind} Y = n$ .

We shall show that the closed cover  $\{f^{-1}(y)\}_{y \in Y}$  of the space X satisfies the conditions of Lemma 1.12.3 for m = n+k. Clearly,  $\operatorname{ind} f^{-1}(y) \leq k \leq m$ for each  $y \in Y$ . Consider now an  $y \in Y$  and an open set  $V \subset X$  which contains  $f^{-1}(y)$ . The set  $W = Y \setminus f(X \setminus V)$  is a neighbourhood of the point y; since  $\operatorname{ind} Y = n$ , there exists an open set  $U' \subset Y$  such that

$$y \in U' \subset U' \subset W$$
 and  $\operatorname{ind} \operatorname{Fr} U' \leq n-1$ .

136

137

Applying the inductive assumption to the restriction  $f_{FrU'}$ :  $f^{-1}(FrU') \rightarrow FrU'$ , we obtain the inequality  $\operatorname{ind} f^{-1}(FrU') \leq n+k-1$ . The open set  $U = f^{-1}(U')$  satisfies the conditions

$$f^{-1}(y) \subset U \subset \overline{U} = \overline{f^{-1}(U')} \subset f^{-1}(\overline{U'}) \subset f^{-1}(W) \subset V$$

and

$$\operatorname{Fr} U = \operatorname{Fr} f^{-1}(U') = \overline{f^{-1}(U')} \setminus f^{-1}(U')$$
  
$$\subset f^{-1}(\overline{U'}) \setminus f^{-1}(U') \subset f^{-1}(\operatorname{Fr} U'),$$

so that  $\operatorname{ind} \operatorname{Fr} U \leq n+k-1$ . Lemma 1.12.3 now implies that  $\operatorname{ind} X \leq n+k$ =  $\operatorname{ind} Y+k$ .  $\Box$ 

Let us note that in Theorem 1.12.4 the assumption that f is a closed mapping cannot be replaced by the assumption that f is open (see Problem 1.12.C). On the other hand, Theorem 1.12.2 holds for open mappings as well; we shall show below that even more is true: open mappings with finite fibres do not change dimension. We shall also show that open mappings with countable fibres defined on locally compact spaces do not change dimension. In the proofs of both theorems the following lemma will be applied; we recall that the symbol  $A^d$  denotes the set of all accumulation points of the set A, i.e., the set of all points x such that  $x \in \overline{A \setminus \{x\}}$ .

**1.12.5. Lemma.** Let  $f: X \to Y$  be an open mapping of a metric space X onto a metric space Y. For every base  $\mathscr{B} = \{U_s\}_{s\in S}$  for the space X there exists a family  $\{A_s\}_{s\in S}$  of subsets of X such that  $A_s \subset U_s$  for each  $s \in S$  and (i)  $A_s$  and  $f(A_s)$  are  $F_{\sigma}$ -sets in X and Y, respectively, (ii)  $f|A_s: A_s \to f(A_s)$  is a homeomorphism, (iii)  $X = (\bigcup_{s\in S} A_s) \cup (\bigcup_{y\in Y} [f^{-1}(y)]^d)$ .

**Proof.** For each  $s \in S$  let

$$A_s = \{x \in U_s: U_s \cap f^{-1}f(x) = \{x\}\}.$$

Observe first that the set  $f(U_s) \setminus f(A_s)$  is open in Y. Indeed, for every point  $y \in f(U_s) \setminus f(A_s)$  there exist two distinct points  $x_1, x_2 \in U_s$  such that  $f(x_1) = f(x_2) = y$ , and—as one readily checks—the open set  $V = f(W_1) \cap f(W_2)$ , where  $W_1, W_2$  are disjoint open subsets of  $U_s$  which contain  $x_1$  and  $x_2$ , respectively, contains the point y and is contained in the set  $f(U_s) \setminus f(A_s)$ . As  $f(A_s) \subset f(U_s)$ ,

$$f(A_s) = f(U_s) \setminus [f(U_s) \setminus f(A_s)],$$

so that the set  $f(A_s)$ , being the intersection of an open set and a closed set, is an  $F_{\sigma}$ -set in Y. From the obvious equality

$$(4) A_s = U_s \cap f^{-1} f(A_s)$$

it follows that  $A_s$  is an  $F_{\sigma}$ -set in X; thus (i) is proved.

The restriction  $f|U_s: U_s \to f(U_s)$  is an open mapping, and so is its restriction  $(f|U_s)_{f(A_s)}: U_s \cap f^{-1}f(A_s) \to f(A_s)$ . By virtue of (4) the last mapping coincides with  $f|A_s: A_s \to f(A_s)$ . Thus the mapping  $f|A_s: A_s \to f(A_s)$  is open and, by the definition of  $A_s$ , one-to-one, i.e., we have (ii).

To prove (iii), it suffices to observe that if  $x \notin [f^{-1}f(x)]^d$ , then there exists a member  $U_s$  of the base  $\mathscr{B}$  such that  $U_s \cap f^{-1}(x) = \{x\}$ , and thus  $x \in A_s$ .  $\Box$ 

**1.12.6. Theorem.** If  $f: X \to Y$  is an open mapping of a separable metric space X onto a separable metric space Y such that for every  $y \in Y$  the fibre  $f^{-1}(y)$  has an isolated point, then  $\operatorname{ind} Y \leq \operatorname{ind} X$ .

**Proof.** Let  $\{U_i\}_{i=1}^{\infty}$  be a countable base for the space X; consider a family  $\{A_i\}_{i=1}^{\infty}$  of subsets of X which satisfy (i)-(iii) in Lemma 1.12.5. By the assumption on the fibres,  $Y = \bigcup_{i=1}^{\infty} f(A_i)$ , and since the subspaces  $A_i \subset X$  and  $f(A_i) \subset Y$  are homeomorphic,  $\operatorname{ind} f(A_i) = \operatorname{ind} A_i \leq \operatorname{ind} X$  for  $i = 1, 2, \ldots$  Hence we have  $\operatorname{ind} Y \leq \operatorname{ind} X$  by virtue of Corollary 1.5.4.  $\Box$ 

**1.12.7. Theorem.** If  $f: X \to Y$  is an open mapping of a separable metric space X onto a separable metric space Y such that for every  $y \in Y$  the fibre  $f^{-1}(y)$  is a discrete subspace of X, then  $\operatorname{ind} X = \operatorname{ind} Y$ .

**Proof.** Let  $\{U_i\}_{i=1}^{\infty}$  be a countable base for the space X; consider a family  $\{A_i\}_{i=1}^{\infty}$  of subsets of X which satisfy (i)-(iii) in Lemma 1.12.5. By the assumption on the fibres,  $X = \bigcup_{i=1}^{\infty} A_i$ , and since the subspaces  $A_i \subset X$  and  $f(A_i) \subset Y$  are homeomorphic,  $\operatorname{ind} A_i = \operatorname{ind} f(A_i) \leq \operatorname{ind} Y$  for  $i = 1, 2, \ldots$  Hence we have  $\operatorname{ind} X \leq \operatorname{ind} Y$  by virtue of Corollary 1.5.4; to complete the proof it suffices to apply Theorem 1.12.6.  $\Box$ 

**1.12.8.** Alexandroff's theorem. If  $f: X \to Y$  is an open mapping of a separable locally compact metric space X onto a separable metric space Y such that  $|f^{-1}(y)| \leq \aleph_0$  for every  $y \in Y$ , then  $\operatorname{ind} X = \operatorname{ind} Y$ .

**Proof.** Since each closed subspace of a locally compact space is locally compact, the fibres of f are locally compact. It follows from the Baire category theorem and complete metrizability of locally compact spaces (or from Problem 1.2.D) that for every  $y \in Y$  the fibre  $f^{-1}(y)$ , being countable, has an isolated point. Hence we have  $\operatorname{ind} Y \leq \operatorname{ind} X$  by Theorem 1.12.6.

Let  $\{F_i\}_{i=1}^{\infty}$  be a countable cover of X consisting of compact subspaces. For every *i* the restriction  $f|F_i: F_i \to f(F_i)$  is a closed mapping with zerodimensional fibres, so that  $\operatorname{ind} F_i \leq \operatorname{inf} f(F_i) \leq \operatorname{ind} Y$  by virtue of the theorem on dimension-lowering mappings. Hence we have  $\operatorname{ind} X \leq \operatorname{ind} Y$  by virtue of the sum theorem.  $\Box$ 

It is possible to define open mappings with countable fibres which arbitrarily raise or lower dimension (see Problems 1.12.E and 1.12.F). Open mappings with zero-dimensional fibres defined on compact spaces can also arbitrarily raise dimension, but examples of such mappings are very complicated and will not be discussed here. Let us note, however, that the space  $M_1^3$  defined in the last section can be mapped onto every locally connected metric continuum by an open mapping whose fibres are all homeomorphic to the Cantor set.

We conclude this section with a discussion of *open-and-closed* mappings, i.e., mappings which are both open and closed. It follows from the last paragraph that open-and-closed mappings with zero-dimensional fibres can arbitrarily raise dimension. We shall now show that open-andclosed mappings with countable fibres preserve dimension.

**1.12.9.** Vainštein's lemma. If  $f: X \to Y$  is a closed mapping of a metric space X onto a metric space Y, then for every  $y \in Y$  the boundary  $\operatorname{Fr} f^{-1}(y)$  of the fibre  $f^{-1}(y)$  is a compact subspace of X.

**Proof.** It suffices to show that every countably infinite subset  $A = \{x_1, x_2, ...\}$  of the boundary  $\operatorname{Fr} f^{-1}(y)$  of an arbitrary fibre  $f^{-1}(y)$  has an accumulation point. Let  $\{V_i\}_{i=1}^{\infty}$  be a base for the space Y at the point y. For i = 1, 2, ... choose a point  $x'_i \in f^{-1}(V_i) \setminus f^{-1}(y)$  satisfying  $\varrho(x_i, x'_i) < 1/i$ ; such a choice is possible, because the intersection  $B(x_i, 1/i) \cap \cap f^{-1}(V_i)$  is a neighbourhood of  $x_i$ . Consider the set  $B = \{x'_1, x'_2, ...\}$   $\subset X$ . We have  $y \in \overline{f(B)} \setminus f(B)$  so that  $\overline{B} \neq B$ , i.e.,  $B^d \neq \emptyset$ . Now, since  $\varrho(x_i, x'_i) < 1/i$  for  $i = 1, 2, ..., A^d = B^d \neq \emptyset$  and the proof is completed.  $\Box$
**1.12.10. Theorem.** If  $f: X \to Y$  is an open-and-closed mapping of a separable metric space X onto a separable metric space Y such that  $|f^{-1}(y)| \leq \aleph_0$  for every  $y \in Y$ , then  $\operatorname{ind} X = \operatorname{ind} Y$ .

**Proof.** Let us note first that Theorem 1.12.4 yields the inequality  $\operatorname{ind} X \leq \operatorname{ind} Y$ .

Now, let  $Y_0 = \{y \in Y: \operatorname{Int} f^{-1}(y) \neq \emptyset\}$ ,  $Y_1 = Y \setminus Y_0$ , and  $X_1 = f^{-1}(Y_1)$ . The mapping f being open, for every  $y \in Y_0$  the one-point set  $\{y\} = f(\operatorname{Int} f^{-1}(y))$  is open, which implies that the set  $Y_0$  is open in Y and  $\operatorname{ind} Y_0 \leq 0$ . The restriction  $f_1 = f_{Y_1}: X_1 \to Y_1$  is also an open mapping, and since  $f_1^{-1}(y) = \operatorname{Fr} f^{-1}(y) \subset f^{-1}(y)$ , the fibres of  $f_1$  are countable and compact, the latter by virtue of Vaĭnšteĭn's lemma; in particular, for every  $y \in Y$  the fibre  $f_1^{-1}(y)$  has an isolated point. It then follows from Theorem 1.12.6 that  $\operatorname{ind} Y_1 \leq \operatorname{ind} X_1 \leq \operatorname{ind} X$ ; as  $Y = Y_0 \cup Y_1$ , Corollary 1.5.5 yields the inequality  $\operatorname{ind} Y \leq \operatorname{ind} X$ .  $\Box$ 

#### Historical and bibliographic notes

Theorem 1.12.2 was proved by Hurewicz in [1927a]. The same paper contains Theorem 1.12.4 for continuous mappings defined on compact metric spaces; the extension of this theorem to separable metric spaces was given by Hurewicz and Wallman in [1941]. Theorem 1.12.6 was first stated by Taimanov in [1955]; however, it is implicitly contained in Alexandroff's paper [1936]. Theorem 1.12.7 was proved by Hodel in [1963] (for the special case of a mapping with finite fibres it was proved earlier by Nagami, namely in [1960]). Alexandroff established Theorem 1.12.8 in [1936]. The first example of a dimension-raising open mapping with zerodimensional fibres defined on a compact metric space was described by Kolmogoroff in [1937]. Keldyš defined in [1954] an open mapping with zero-dimensional fibres which maps a one-dimensional compact metric space onto the square  $I^2$ ; a detailed description of Keldyš' example can be found in Alexandroff and Pasynkov's book [1973]. The fact that  $M_1^3$ can be mapped onto every locally connected metric continuum by an open mapping whose fibres are all homeomorphic to the Cantor set was established by Wilson in [1972]. Theorem 1.12.10 was given by Vainšteĭn in [1949]. Lelek's paper [1971] contains a comprehensive discussion of the topic of the present section and a good bibliography. We shall return to this subject in Section 4.3, where two theorems of a more special character will be proved (see Theorems 4.3.9 and 4.3.12).

#### Problems

**1.12.A** (Hurewicz [1926]). Prove that if  $f: X \to Y$  is a closed mapping of a separable metric space X onto a separable metric space Y and  $|f^{-1}(y)| = k < \infty$  for every  $y \in Y$ , then  $\operatorname{ind} X = \operatorname{ind} Y$ .

*Hint.* Consider a countable base  $\mathscr{B}$  for the space X and the family of all intersections  $\bigcap_{i=1}^{k} f(U_i)$ , where  $U_i \in \mathscr{B}$  for i = 1, 2, ..., k and  $\overline{U}_i \cap \overline{U}_j$ =  $\emptyset$  whenever  $i \neq j$ .

1.12.B (Hurewicz [1937]). Observe that, under the additional hypothesis that X is a finite-dimensional compact space, the theorem on dimension-lowering mappings is a direct consequence of the Cantor-manifold theorem.

*Hint*. Assume that X is a Cantor-manifold and apply induction with respect to ind Y.

1.12.C. Give an example of an open mapping with zero-dimensional fibres which maps a one-dimensional separable metric space onto the Cantor set.

Hint. Use the Knaster-Kuratowski fan.

*Remark.* It follows from Problem 1.12.F that there even exist such mappings with countable fibres (cf. Problem 1.12.G(b)).

**1.12.D.** (a) Observe that in Theorem 1.12.2 the inequality  $|f^{-1}(y)| \le k$  can be replaced by the weaker inequality  $|\operatorname{Fr} f^{-1}(y)| \le k$ .

*Hint.* Consider the restriction  $f|X_1$ , where  $X_1$  is obtained by adjoining to the union  $\bigcup_{y \in Y} \operatorname{Fr} f^{-1}(y)$  one point from each fibre  $f^{-1}(y)$  which has an empty boundary.

(b) Observe that in Theorem 1.12.6 it suffices to assume that for every  $y \in Y$  the boundary  $\operatorname{Fr} f^{-1}(y)$  either has an isolated point or is empty.

1.12.E (Hausdorff [1934]). Show that every separable metric space X can be represented as the image of a subspace Z of the space P of irrational numbers under an open mapping.

*Hint.* Consider a countable base  $\{U_k\}_{k=1}^{\infty}$  for the space X and the subspace Z of the Cartesian product  $N^{\aleph_0} = \prod_{i=1}^{\infty} N_i$ , where  $N_i = N$  for i = 1, 2, ..., consisting of all points  $\{k_i\}$  such that the family  $\{U_{k_i}\}_{i=1}^{\infty}$  is a base for X at a point x; assign the point x to the point  $\{k_i\}$ .

1.12.F (Roberts [1947]). Show that for every open mapping  $f: X \to Y$  of a separable metric space X onto a separable metric space Y there exists a set  $X_1 \subset X$  such that the restriction  $f|X_1$  is an open mapping of  $X_1$  onto Y and has countable fibres.

*Hint.* Consider a countable base  $\{U_i\}_{i=1}^{\infty}$  for the space X and choose one point from each non-empty intersection of the form  $U_i \cap f^{-1}(y)$ , where  $y \in Y$ .

**1.12.G.** (a) Observe that if  $f: X \to Y$  is an open mapping of a complete separable metric space X onto a separable metric space Y such that  $|f^{-1}(y)| \leq \aleph_0$  for every  $y \in Y$ , then ind  $Y \leq \operatorname{ind} X$ .

(b) Give an example of an open mapping with countable fibres which maps a one-dimensional complete separable metric space onto the Cantor set.

*Hint*. For every sequence  $i_1, i_2, ..., i_m$  consisting of zeros and ones, let

$$C(i_1, i_2, ..., i_m) = \left\{ x \in C \colon \sum_{k=1}^m \frac{2i_k}{3^k} \le x \le \sum_{k=1}^m \frac{2i_k}{3^k} + \frac{1}{3^m} \right\},\$$

and let  $C'(i_1, i_2, ..., i_m)$  be a subspace of I which is homeomorphic to the Cantor set and is contained in the interval

$$\left\{ x \in I \colon \sum_{k=1}^{m} \frac{2i_k}{3^k} + \frac{1}{3^{m+1}} < x < \sum_{k=1}^{m} \frac{2i_k}{3^k} + \frac{2}{3^{m+1}} \right\}$$

removed from I in the process of constructing the Cantor set. Consider a countable set  $\{a_1, a_2, ...\}$  dense in the interval [-1, 1] and define

$$X = K \cup \bigcup_{m=1}^{\infty} \bigcup_{i_1, \dots, i_m} \bigcup_{k=1}^{m} [C'(i_1, i_2, \dots, i_m) \times \{a_k\}] \subset X \times [-1, 1],$$

where K is the space in Problem 1.2.E; show that the mapping  $f: X \to C$ , where f|K is the projection of K onto C and  $f|[C'(i_1, i_2, ..., i_m) \times \{a_k\}]$ is an arbitrary homeomorphism of  $C'(i_1, i_2, ..., i_m) \times \{a_k\}$  onto  $C(i_1, i_2, ..., i_m)$ , has the required properties.

#### 1.13. Dimension and inverse sequences of polyhedra

Besides theorems on  $\epsilon$ -mappings and  $\epsilon$ -translations there is one more characterization of dimension in the realm of compact metric spaces through their relations to polyhedra. It has been discovered that the class of compact metric spaces which have dimension  $\leq n$  coincides with the class of spaces which are homeomorphic to the limits of inverse sequences of polyhedra which have dimension  $\leq n$ . We shall deduce this characterization from two theorems which we are going to prove: the theorem on expansion in an inverse sequence and the theorem on the dimension of the limit of an inverse sequence.

As the subject of this section is more specific than our previous considerations, we shall assume here that the reader is familiar with the basic definitions and theorems in the theory of inverse systems (see, e.g., [GT], pp. 135-140, 188 and 189). We shall only recall that an inverse sequence  $\{X_i, \pi_j^i\}$  is an inverse system  $\{X_i, \pi_j^i, N\}$  where N is the set of natural numbers directed by its natural order.

In Lemma 1.13.1 and in the proof of Theorem 1.13.2 below the symbol  $T^m$  will denote the *m*-simplex in  $R^{m+1}$  spanned by the points  $p_1 = (1, 0, ..., 0)$ ,  $p_2 = (0, 1, ..., 0), ..., p_{m+1} = (0, 0, ..., 1)$ . We shall consider on  $T^m$  the metric  $\sigma$  defined by letting

$$\sigma(x,y) = |x-y|, \text{ where } |z| = \sum_{i=1}^{m+1} |\lambda_i| \text{ for } z = (\lambda_1, \lambda_2, \dots, \lambda_{m+1}) \in \mathbb{R}^{m+1};$$

obviously, the metric  $\sigma$  is equivalent to the natural metric on  $T^m$ .

**1.13.1. Lemma.** Let a set  $A_j \subset \{1, 2, ..., n+1\}$  be given for j = 1, 2, ..., land let  $\pi$  be the mapping of  $T^{l-1}$  to  $T^n = p_1 p_2 \dots p_{n+1}$  defined by the formula

$$\pi((\lambda_1, \lambda_2, ..., \lambda_l)) = \sum_{j=1}^l \frac{\lambda_j}{n_j} \sum_{i \in A_j} p_i, \quad \text{where } n_j = |A_j|.$$

If  $A_1 \cap A_2 \cap \ldots \cap A_l \neq \emptyset$ , then  $\sigma(\pi(x), \pi(y)) \leq \frac{n}{n+1} \sigma(x, y)$  for all  $x, y \in T^{l-1}$ .

**Proof.** Consider arbitrary points  $x = (\lambda_1, \lambda_2, ..., \lambda_l)$  and  $y = (\mu_1, \mu_2, ..., \mu_l)$  in  $T^{l-1}$ . Let  $\alpha_j = \lambda_j - \mu_j$  for j = 1, 2, ..., l,  $B = \{j: \alpha_j \ge 0\}$  and

$$C = \{j: \alpha_j < 0\}.$$
 From the equality  $\sum_{j=1}^{l} \alpha_j = 0$  it follows that  $\sum_{j \in B} |\alpha_j| = \sum_{j \in C} |\alpha_j| = \frac{1}{2}\sigma(x, y).$  One easily checks that

$$\left|\sum_{j=1}^{l} \frac{\alpha_j}{n_j}\right| = \sum_{j=1}^{l} \frac{|\alpha_j|}{n_j} - 2\min(|b|, |c|), \quad \text{where } b = \sum_{j \in B} \frac{\alpha_j}{n_j}, \ c = \sum_{j \in C} \frac{\alpha_j}{n_j}.$$

Since  $n_j \leq n+1$  for j = 1, 2, ..., l, we have

$$|b| = \sum_{j \in B} \frac{|\alpha_j|}{n_j} \ge \frac{1}{n+1} \sum_{j \in B} |\alpha_j| = \frac{1}{2} \frac{\sigma(x, y)}{n+1};$$

similarly  $|c| \ge \frac{1}{2} \frac{\sigma(x, y)}{n+1}$ , so that  $\min(|b|, |c|) \ge \frac{1}{2} \frac{\sigma(x, y)}{n+1}$ . Choosing arbitrarily an  $i_0 \in A_1 \cap A_2 \cap \ldots \cap A_l$  and letting  $A'_j = A_j \setminus \{i_0\}$  for  $j = 1, 2, \ldots$ ..., l we have

$$\sigma(\pi(x), \pi(y)) = \left| \sum_{j=1}^{l} \frac{\alpha_j}{n_j} \sum_{i \in A_j} p_i \right| = \left| \left( \sum_{j=1}^{l} \frac{\alpha_j}{n_j} \right) p_{i_0} + \sum_{j=1}^{l} \frac{\alpha_j}{n_j} \sum_{i \in A'_j} p_i \right|$$
  
$$\leq \left| \sum_{j=1}^{l} \frac{\alpha_j}{n_j} \right| + \sum_{j=1}^{l} \frac{|\alpha_j| (n_j - 1)}{n_j} = \sum_{j=1}^{l} \frac{|\alpha_j|}{n_j} + \sum_{j=1}^{l} \frac{|\alpha_j| (n_j - 1)}{n_j} - 2\min(|b|, |c|)$$
  
$$= \sum_{j=1}^{l} |\alpha_j| - 2\min(|b|, |c|) = \sigma(x, y) - 2\min(|b|, |c|)$$
  
$$\leq \sigma(x, y) - \frac{\sigma(x, y)}{n+1} = \frac{n}{n+1} \sigma(x, y). \square$$

**1.13.2.** Theorem on expansion in an inverse sequence. For every compact metric space X such that dim $X \leq n$  there exists an inverse sequence  $\{K_i, \pi_j^i\}$  consisting of polyhedra of dimension  $\leq n$  whose limit is homeomorphic to X; moreover, one can assume that, for  $i = 1, 2, ..., K_i$  is the underlying polyhedron of a nerve  $\mathcal{K}_i$  of a finite open cover of the space X, and that for every  $j \leq i$  the bonding mapping  $\pi_j^i$  is linear on each simplex in  $\mathcal{K}_i$ .

**Proof.** We can suppose that dim  $X \ge 0$ . Consider a sequence  $\mathscr{U}_1, \mathscr{U}_2, ...$  of finite open covers of the space X, where  $\mathscr{U}_i = \{U_{i,k}\}_{k=1}^{k}$  and  $U_{i,k} \ne \emptyset$  for  $k = 1, 2, ..., k_m$ , such that

(1) 
$$\operatorname{ord} \mathscr{U}_i \leq n$$

and

(2) 
$$\operatorname{mesh} \mathscr{U}_{i+1} \leq \min(1/i+1, \varepsilon_i/2),$$

where  $\varepsilon_i$  is a Lebesgue number for the cover  $\mathscr{U}_i$ . It follows from (2) that

(3) if 
$$U_{i+1,j_1} \cap U_{i+1,j_2} \cap \dots \cap U_{i+1,j_l} \neq \emptyset$$
, then there exists a  $k \leq k_i$  such that  $U_{i+1,j_m} \subset U_{i,k}$  for  $m = 1, 2, ..., l$ .

For i = 1, 2, ... let  $\mathscr{K}_t = \mathscr{N}(\mathscr{U}_t)$  be a nerve of the cover  $\mathscr{U}_t$  consisting of faces of the simplex  $T^{k_t-1}$ . By virtue of (1), the underlying polyhedra  $K_t = |\mathscr{K}_t|$  all have dimension  $\leq n$ .

We shall now define continuous mappings  $\pi_i^{i+1}$ :  $K_{i+1} \to K_i$  for i = 1, 2, ... Let  $p_{i+1,j}$  be a vertex of the complex  $\mathscr{K}_{i+1}$ . Consider the member  $U_{i+1,j}$  of the cover  $\mathscr{U}_{i+1}$  which corresponds to  $p_{i+1,j}$ . By virtue of (3) the family

$$\mathscr{U}_{i,j} = \{ U \in \mathscr{U}_i \colon U_{i+1,j} \subset U \}$$

is non-empty. Since  $\bigcap \mathscr{U}_{i,j} \neq \emptyset$ , the vertices of  $\mathscr{K}_i$  which correspond to the members of  $\mathscr{U}_{i,j}$  span a simplex  $S_{i,j} \in \mathscr{K}_i$ ; we let

(4) 
$$\pi_i^{i+1}(p_{i+1,j}) = b(S_{i,j}),$$

where b(S) denotes the barycentre of S.

We shall prove that

(5) for every simplex  $S \in \mathscr{H}_{i+1}$ , the images of vertices of S under  $\pi_i^{i+1}$  are contained in a simplex  $T \in \mathscr{H}_i$ .

Indeed, if  $S = p_{i+1,j_1} p_{i+1,j_2} \dots p_{i+1,j_l}$ , then

 $\emptyset \neq U_{i+1, j_1} \cap U_{i+1, j_2} \cap \dots \cap U_{i+1, j_l} \subset \{ \ \ \ \mathscr{U}_{i, j_1} \cap \bigcap \mathscr{U}_{i, j_2} \cap \dots \cap \bigcap \mathscr{U}_{i, j_l} \}$ so that the vertices of  $\mathscr{K}_i$  which correspond to the members of the union  $\mathscr{U}_{i, j_1} \cup \mathscr{U}_{i, j_2} \cup \dots \cup \mathscr{U}_{i, j_l}$  span a simplex  $T \in \mathscr{K}_i$  which contains the point  $\pi_i^{i+1}(p_{i+1, j_m})$  for m = 1, 2, ..., l.

It follows from (5) that the mapping  $\pi_i^{i+1}$  defined on the set of all vertices of  $\mathscr{K}_{i+1}$  can be extended over each simplex  $S = p_{i+1, j_1} p_{i+1, j_2} \cdots \dots p_{i+1, j_i} \in \mathscr{K}_{i+1}$  by letting

$$\pi_{i}^{i+1}\left(\sum_{m=1}^{l}\lambda_{j_{m}}p_{i+1,j_{m}}\right)=\sum_{j=1}^{l}\lambda_{j_{m}}\pi_{i}^{i+1}(p_{i+1,j_{m}});$$

in this way a continuous mapping  $\pi_i^{i+1}$ :  $K_{i+1} \to K_i$  is defined for i = 1, 2, ...One easily checks that by defining, for every pair i, j of natural numbers satisfying  $j \leq i$ ,

$$\pi_j^i = \pi_j^{j+1} \pi_{j+1}^{j+2} \dots \pi_{i-1}^i \quad \text{if} \quad j < i \quad \text{and} \quad \pi_i^i = \mathrm{id}_{K_i}$$

one obtains continuous mappings  $\pi_j^i$ :  $K_i \to K_j$  linear on each simplex in  $\mathcal{K}_i$ .

Thus the inverse sequence  $S = \{K_i, \pi_j^i\}$  is defined. It remains to show that the space X is homeomorphic to the limit K of this inverse sequence.

By virtue of (3) and (5) we can apply Lemma 1.13.1 to restrictions of  $\pi_i^{i+1}$  to simplexes in  $\mathcal{K}_{i+1}$ , so that

(6) 
$$\delta(\pi_i^{i+1}(F)) \leq \frac{n}{n+1} \delta(F)$$
 for every  $F \subset S \in \mathscr{K}_{i+1}$ .

Let us observe that for every choice of a point  $y_i$  in  $K_i$  for i = 1, 2, ...

(7) if  $\emptyset \neq F_i = \overline{F_i} \subset \bigcup \{S \in \mathscr{K}_i: y_i \in S\}$  and  $\pi_i^{i+1}(F_{i+1}) \subset F_i$  for i = 1, 2, ..., then the limit L of the inverse sequence  $\{F_i, \pi_j^i | F_i\}$  is a one-point set.

Indeed, the set L is non-empty as the limit of an inverse sequence of nonempty compact spaces; since for  $j = 1, 2, ..., \pi_j(L) \subset \bigcap_{i=j}^{\infty} \pi_j^i(F_i)$ , where  $\pi_j: K \to K_i$  is the projection, and since—by virtue of (6) and the inequality  $\delta(F_i \cap S) \leq \delta(S) \leq 2 - \delta(\pi_j^i(F_i)) \leq 4(n/n+1)^{i-j}$  whenever  $j \leq i$ , the sets  $\pi_j(L)$  are all one-point sets, which implies that L is a one-point set.

Now, consider a point  $x \in X$ ; for every natural number *i* let

$$\mathscr{U}_i(x) = \{ U \in \mathscr{U}_i \colon x \in U \}$$

and denote by  $K_i(x)$  the simplex  $\mathcal{H}_i$  spanned by the vertices of  $\mathcal{H}_i$  which correspond to the members of  $\mathcal{U}_i(x)$ . Let us note that

(8) 
$$\pi_i^{i+1}(K_{i+1}(x)) \subset K_i(x)$$
 for  $i = 1, 2, ...$ 

Indeed, if  $U_{i+1,j} \in \mathscr{U}_{i+1}(x)$ , then  $\mathscr{U}_{i,j} \subset \mathscr{U}_i(x)$ , so that the images of vertices of  $K_{i+1}(x)$  under  $\pi_i^{i+1}$  are contained in the simplex  $K_i(x)$ , and this implies (8). It follows from (8) that  $K(x) = \{K_i(x), \pi_j^i | K_i(x)\}$  is an inverse sequence. By virtue of (7) the limit of this inverse sequence contains exactly one point; let it be denoted by f(x). Clearly,  $f(x) \in K$ ; a mapping f of X to K is thus defined.

We shall prove that f is a continuous mapping. Obviously, it is enough to show that for every natural number j the composition  $f_j = \pi_j f$  is continuous. Consider a point  $x_0 \in X$ , a positive number  $\varepsilon$  and a natural number i such that  $i \ge j$  and  $(n/n+1)^{i-j} < \varepsilon/2$ . The set  $U = \bigcap \mathcal{U}_i(x_0)$  is a neighbourhood of  $x_0$  and for every  $x \in U$  the inclusion  $\mathcal{U}_i(x_0) \subset \mathcal{U}_i(x)$  holds, so that  $K_i(x_0) \subset K_i(x)$  and y

$$f_j(x_0) \in \pi_j^j(K_i(x_0)) \subset \pi_j^i(K_i(x)) \quad \text{for } x \in U.$$

Since by virtue of (6)

$$\delta(\pi j(K_i(x))) \leq 2(n/n+1)^{i-j} < \epsilon$$

and since  $f_j(x) \in \pi_j^l(K_i(x))$ ,

$$\sigma(f_j(x_0), f_j(x)) < \varepsilon \quad \text{for } x \in U,$$

i.e., the mapping  $f_j$  is continuous.

As for each pair x, x' of distinct points of X there exists —by virtue of (2)—a natural number *i* such that  $\mathscr{U}_i(x) \cap \mathscr{U}_i(x') = \emptyset$ , i.e.,  $K_i(x) \cap \cap K_i(x') = \emptyset$ , the mapping  $f: X \to K$  is one-to-one. As X is a compact space, to complete the proof it suffices to show that f(X) = K.

Consider a point  $y = \{y_i\} \in K$ . For i = 1, 2, ... let

$$A_i = \bigcup \{ S \in \mathscr{K}_i : y_i \in S \} \subset K_i \text{ and } B_i = f_i^{-1}(A_i) \subset X.$$

The sets  $B_i$  are non-empty. Indeed, if  $S_i = p_{i,j_1} p_{i,j_2} \dots p_{i,j_l}$  is a maximal simplex in  $\mathcal{K}_i$  which contains  $y_i$ , and  $x_i$  is a point in  $U_{i,j_1} \cap U_{i,j_2} \cap \dots \cap U_{i,j_l}$ , then  $K_i(x_i) = S_i \subset A_i$ , which implies that  $x_i \in B_i$ .

Since the image of each simplex  $S \in \mathscr{K}_{i+1}$  under  $\pi_i^{i+1}$  is contained in a simplex  $T \in \mathscr{K}_i$ , and since  $\pi_i^{i+1}(y_{i+1}) = y_i$ ,

(9) 
$$\pi_i^{i+1}(A_{i+1}) \subset A_i \quad \text{for } i = 1, 2, ...$$

The last inclusion implies that

$$B_{i+1} = f_{i+1}^{-1}(A_{i+1}) \subset f_{i+1}^{-1}(\pi_i^{i+1})^{-1}(A_i)$$
  
=  $(\pi_i^{i+1}f_{i+1})^{-1}(A_i) = f_i^{-1}(A_i) = B_i$ 

for i = 1, 2, ..., so that, by the compactness of X, there exists a point  $x \in \bigcap_{i=1}^{\infty} B_i$ . Let  $F_i = A_i \cap K_i(x)$  for i = 1, 2, ...; as  $f_i(x) \in F_i$ , the sets  $F_i$  are non-empty. By virtue of (9) and (8)

$$\pi_i^{i+1}(F_{i+1}) = \pi_i^{i+1}(A_{i+1} \cap K_{i+1}(x)) \subset \pi_i^{i+1}(A_{i+1}) \cap \pi_i^{i+1}(K_{i+1}(x))$$
$$\subset A_i \cap K_i(x) = F_i \quad \text{for } i = 1, 2, ...,$$

so that  $\{F_i, \pi_j^i | F_i\}$  is an inverse sequence; by (7), the limit of it is a onepoint set  $\{z\} \subset K$ . From (9) and (7) it follows that  $\{A_i, \pi_j^i | A_i\}$  is an inverse sequence whose limit is the one-point set  $\{y\}$ . By the definition of f, the one-point set  $\{f(x)\}$  is the limit of the inverse sequence K(x). Since  $\{z\} \subset \{y\} \cap \{f(x)\}$ , we have f(x) = y.

Let us note that another proof of Theorem 1.13.2 (see the hint to Problem 1.13.G(a)) leads to an inverse sequence  $\{K_i, \pi_j^i\}$ , where, for  $i = 1, 2, ..., K_i$  is the underlying polyhedron of a nerve  $\mathcal{K}_i$  of a finite open cover of the space X, and for every  $j \leq i$  the bonding mapping  $\pi_j^i$  is a quasi-simplicial mapping of  $K_i$  onto  $K_j$ , i.e., it is the linear extension of a simplicial mapping of a barycentric subdivision of  $\mathcal{K}_i$  onto a barycentric subdivision of  $\mathcal{K}_j$  (cf. Problem 1.13.C).

Let us also note that from Problem 1.10.K it follows that every compact subspace X of Euclidean *m*-space  $R^m$  such that dim  $X \le n \ge 0$  can be represented as the limit of an inverse sequence consisting of polyhedra of dimension  $\le n$  which are all contained in  $R^m$ . The converse does not hold; simple examples show that the limit of an inverse sequence of polyhedra contained in  $R^m$  need not be embeddable in  $R^m$  (see Problem 1.13.B).

We now pass to the theorem on the dimension of the limit of an inverse sequence. In the proof we shall apply the following lemma, a slight strengthening of the compactification theorem, which allows the reduction of the problem to the special case where the inverse sequence consists of compact spaces.

**1.13.3. Lemma.** For every continuous mapping  $f: X \to Z$  of a separable metric space X to a compact metric space Z there exists a compact metric space  $\tilde{X}$  containing X as a dense subspace and such that  $\dim \tilde{X} \leq \dim X$  and f is extendable to a continuous mapping  $\tilde{f}: \tilde{X} \to Z$ .

**Proof.** Let  $\rho_0$  be a totally bounded metric on the space X and  $\sigma$  an arbitrary metric on the space Z. One easily checks that the formula

$$\varrho(x, y) = \varrho_0(x, y) + \sigma(f(x), f(y)) \quad \text{for } x, y \in X$$

defines a metric equivalent to  $\varrho_0$  and such that the mapping f is uniformly continuous with respect to  $\varrho$  and  $\sigma$ . The metric  $\varrho$  is totally bounded. Indeed, for every positive number  $\varepsilon$  there exists a finite cover  $\mathscr{A}$  of the space  $(X, \varrho_0)$ such that mesh  $\mathscr{A} < \varepsilon/2$  and a finite cover  $\mathscr{B}$  of the space  $(Y, \sigma)$  such that mesh  $\mathscr{B} < \varepsilon/2$ ; one can readily check that the mesh of the cover  $\mathscr{A} \wedge f^{-1}(\mathscr{B})$  of the space  $(X, \varrho)$  is less than  $\varepsilon$ . It follows from the second part of the compactification theorem that on X there exists a metric  $\tilde{\varrho}$ equivalent to  $\varrho$  such that  $\varrho(x, y) \leq \tilde{\varrho}(x, y)$  for  $x, y \in X$ , and that the completion  $\tilde{X}$  of the space X with respect to  $\tilde{\varrho}$  is a compact space such that dim $\tilde{X}$   $\leq \dim X$ . Obviously, the mapping f is uniformly continuous with respect to  $\tilde{\varrho}$  and  $\sigma$ , which implies that f is extendable to a continuous mapping  $\tilde{f}: \tilde{X} \to Z$ .  $\Box$ 

**1.13.4.** Theorem on the dimension of the limit of an inverse sequence. If the inverse sequence  $S = \{X_i, \pi_j^i\}$  consists of separable metric spaces  $X_i$  such that  $\dim X_i \leq n$  for i = 1, 2, ..., then the limit  $X = \lim_{i \to \infty} S$  satisfies the inequality  $\dim X \leq n$ .

**Proof.** Let  $\tilde{X}_1$  be a compact metric space which contains  $X_1$  as a dense subspace and such that dim $\tilde{X}_1 \leq n$ . An inductive construction applying Lemma 1.13.3 yields for i = 2, 3, ... a compact metric space  $\tilde{X}_i$  which contains  $X_i$  as a dense subspace and such that dim $\tilde{X}_i \leq n$  and  $\pi_{i-1}^l: X_i \rightarrow \tilde{X}_{i-1}$  is extendable to a continuous mapping  $\tilde{\pi}_{i-1}^l: \tilde{X}_i \rightarrow \tilde{X}_{i-1}$ . Letting

$$\tilde{\pi}_j^i = \tilde{\pi}_j^{j+1} \tilde{\pi}_{j+1}^{j+2} \dots \tilde{\pi}_{i-1}^i$$
 for  $j < i$  and  $\tilde{\pi}_i^i = \operatorname{id}_{\tilde{X}_i}$ ,

one obtains an inverse sequence  $\tilde{S} = {\{\tilde{X}_i, \tilde{\pi}_j\}}$  consisting of compact metric spaces  $\tilde{X}_i$  such that dim  $\tilde{X}_i \leq n$  for i = 1, 2, ... As  $\lim S \subset \lim \tilde{S}$ , it suffices to prove the theorem under the additional assumption that all spaces  $X_i$  are compact.

Let  $\mathscr{U}$  be a finite open cover of the space X. Since the family of all sets  $\pi_i^{-1}(U_i)$ , where  $\pi_i: X \to X_i$  is the projection and  $U_i$  is an open subset of  $X_i$ , is a base for the space X, and since the space X is compact, as is the limit of an inverse sequence of compact spaces, the cover  $\mathscr{U}$  has a finite refinement  $\{\pi_{i_k}^{-1}(U_{i_k})\}_{k=1}^m$ , where  $i_1 \leq i_2 \leq \ldots \leq i_m$  and  $U_{i_k}$  is an open subset of  $X_{i_k}$  for  $k = 1, 2, \ldots, m$ . The family  $\{\pi_{i_m}^{-1}(V_k)\}_{k=1}^m$ , where  $V_k = (\pi_{i_k}^{i_m})^{-1}(U_{i_k})$ , is also a refinement of  $\mathscr{U}$ . Consider a finite open refinement  $\mathscr{V}$  of the cover  $\{V_k\}_{k=1}^m$  of the space  $M = \bigcup_{k=1}^m V_k \subset X_{i_m}$  such that  $\operatorname{ord} \mathscr{V} \leq n$ ; clearly,  $\pi_{i_m}^{-1}(\mathscr{V})$  is a finite open refinement of  $\mathscr{U}$  and has order  $\leq n$ .

Theorems 1.13.2 and 1.13.4 yield the characterization of dimension which was announced at the beginning of the present section.

**1.13.5. Theorem on inverse sequences.** A compact metric space X satisfies the inequality dim  $X \leq n$  if and only if X is homeomorphic to the limit of an inverse sequence consisting of polyhedra of dimension  $\leq n$ .  $\Box$ 

One readily sees that the theorem on  $\varepsilon$ -mappings is an easy consequence of the theorem on expansion in an inverse sequence (see Problem 1.13.A).

The study of relations between these two theorems led to the notion of a  $\Pi$ -like space. Let  $\Pi$  be a family of polyhedra; we say that a compact metric space X is  $\Pi$ -like if for every positive number  $\varepsilon$  there exists an  $\varepsilon$ -mapping of X onto a polyhedron  $K \in \Pi$ . The main theorem on  $\Pi$ -like spaces states that if either  $\Pi$  is a family of connected polyhedra (more generally, the space X below is a continuum) or  $\Pi$  is a hereditary family of polyhedra (i.e., together with the underlying polyhedron of a simplicial complex  $\mathscr{K}$ contains the underlying polyhedra of all subcomplexes of  $\mathcal{K}$ ), then a compact metric space X is  $\Pi$ -like if and only if X is homeomorphic to the limit of an inverse sequence  $\{K_i, \pi_i^i\}$ , where  $K_i \in \Pi$  for i = 1, 2, ... and  $\pi_i^i$  maps  $X_i$  onto  $X_j$  for every  $j \leq i$ . Obviously, this is a generalization of the theorem on inverse sequences, because the family  $\Pi$  of all polyhedra of dimension  $\leq n$  is a hereditary family of polyhedra and, for this family, the class of  $\Pi$ -like spaces coincides with the class of all compact metric spaces whose covering dimension is not larger than n. Various families  $\Pi$  yield interesting classes of compact metric spaces. Thus, for the family  $\Pi$  consisting of the interval I alone, one obtains the class of snake-like continua. Clearly, each snake-like continuum is one-dimensional; one proves that snake-like continua are embeddable in the plane (see the remark to Problem 1.13.B) and that there exists a universal space for the class of all snake-like continua. In a more general setting, one can prove that if  $\Pi$  is a hereditary family of polyhedra which, moreover, is additive (i.e., together with each pair K, L of polyhedra contains the disjoint sum of K and L), then there exists a universal space for the class of all  $\Pi$ -like spaces. Obviously, this implies the existence of a universal space for the class of all compact metric spaces whose covering dimension is not larger than n. In a less direct way, this also implies the existence of a universal snake-like continuum.

#### Historical and bibliographic notes

Theorem 1.13.2 was proved by Freudenthal in [1937]. Freudenthal's proof leads to an inverse sequence of polyhedra with quasi-simplicial mappings onto which, moreover, are irreducible (see Problem 1.13.F); the simpler proof given here was outlined by Isbell in [1959]. Theorem 1.13.4 was proved by Nagami in [1959]; for compact metric spaces it is implicitly contained in Freudenthal's paper [1937]. The notion of a  $\Pi$ -like space was introduced by Mardešić and Segal in [1963]. In the same paper Mardešić and Segal proved that if  $\Pi$  is a family of connected polyhedra, then a compact metric space is  $\Pi$ -like if and only if it is homeomorphic

#### Problems

to the limit of an inverse sequence of polyhedra in the family  $\Pi$ ; Pasynkov in [1966] showed that this is likewise true under the assumption that  $\Pi$ is a hereditary family of polyhedra. It was also shown by Pasynkov in [1966] that if  $\Pi$  is a hereditary and additive family of polyhedra, then there exists a universal space for the family of all  $\Pi$ -like spaces; a somewhat stronger result was obtained by McCord in [1966]. Snake-like continua were introduced by Bing in [1951]; among other things, Bing proved that each snake-like continuum is embeddable in the plane. The existence of a universal space for the class of all snake-like continua was established by Shori in [1965].

## Problems

**1.13.A.** Let  $X \neq \emptyset$  be the limit of an inverse sequence  $\{X_i, \pi_j^i\}$  of compact metric spaces and let  $\varepsilon_i = \operatorname{mesh}(\{\pi_i^{-1}(x)\}_{x \in X_i})$ , where  $\pi_i \colon X \to X_i$  is the projection. Show that the sequence  $\varepsilon_1, \varepsilon_2, \ldots$  converges to zero.

**1.13.B.** Define an inverse sequence  $S = \{X_i, \pi_j^i\}$ , where  $X_i = I$  for i = 1, 2, ... and  $\pi_j^i$  maps  $X_i$  onto  $X_j$  for every  $j \leq i$ , such that the limit of S cannot be embedded in the real line.

*Remark.* As shown by Isbell in [1959], the limit of an inverse sequence of compact subspaces of  $R^m$  is embeddable in  $R^{2m}$ .

1.13.C (Isbell [1964]). Let  $S = \{|\mathscr{K}_i|, \pi_j^i\}$  be an inverse sequence consisting of polyhedra of dimension  $\geq 1$ , where for  $j \leq i$  the bonding mapping  $\pi_j^i$  maps  $|\mathscr{K}_i|$  onto  $|\mathscr{K}_j|$  and is the linear extension of a simplicial mapping of  $\mathscr{K}_i$  to  $\mathscr{K}_j$ . Prove that if the limit of S contains more than one point, then it contains a subspace homeomorphic to the interval I. Applying the fact that there exist one-dimensional continua with no subspace homeomorphic to the interval I (see, Kuratowski [1968], p. 206), observe that there exist one-dimensional compact metric spaces which are not homeomorphic to the limit of an inverse sequence  $\{|\mathscr{K}_i|, \pi_j^i\}$  of one-dimensional polyhedra, where for  $j \leq i$  the bonding mapping  $\pi_j^i$  is the linear extension of a simplicial mapping of  $\mathscr{K}_i$  to  $\mathscr{K}_i$ .

**1.13.D.** Define an inverse sequence  $S = \{X_i, \pi_j^i\}$  of one-dimensional compact metric spaces, where for  $j \leq i$  the bonding mapping  $\pi_j^i$  maps  $X_i$  onto  $X_j$ , such that the limit of S is homeomorphic to the Cantor set.

**1.13.E** (Mardešić and Segal [1963]). Let  $\Pi$  be the family consisting of all polyhedra which are unions of a one-simplex and a finite number of zero-simplexes. Observe that the Cantor set is a  $\Pi$ -like space, and yet is not homeomorphic to the limit of an inverse sequence  $\{K_i, \pi_j^i\}$ , where  $K_i \in \Pi$  for i = 1, 2, ... and  $\pi_i^i$  maps  $X_i$  onto  $X_j$  for every  $j \leq i$ .

**1.13.F** (Freudenthal [1937]). Let  $f: X \to |\mathcal{K}|$  be a continuous mapping of a topological space X to the underlying polyhedron of a simplicial complex  $\mathcal{K}$ . A continuous mapping  $g: X \to |\mathcal{K}|$  is a modification of f if, for every  $S \in \mathcal{K}$ ,  $g(x) \in S$  whenever  $f(x) \in S$ ; the mapping f is *irreducible* if there is no modification  $g: X \to |\mathcal{K}|$  of f such that  $f(X) \setminus g(X) \neq \emptyset$ .

(a) Show that if F is a closed subset of X and  $g': F \to |\mathcal{K}|$  is a modification of the restriction  $f|F: F \to |\mathcal{K}|$  of a continuous mapping  $f: X \to |\mathcal{K}|$ , then there exists a modification  $g: X \to |\mathcal{K}|$  of f such that g|F = g'.

(b) Check that if  $f: X \to |\mathcal{K}|$  is an irreducible mapping, then for every subcomplex  $\mathcal{K}_0$  of the complex  $\mathcal{K}$  the restriction  $f_{|\mathcal{K}_0|}: f^{-1}(|\mathcal{K}_0|) \to |\mathcal{K}_0|$  also is irreducible.

(c) Let  $\mathscr{S}$  be the simplicial complex consisting of all faces of a simplex and let  $\mathscr{S}_0$  be the subcomplex of  $\mathscr{S}$  consisting of all proper faces of the simplex under consideration. Prove that if for a continuous mapping  $f: X \to |\mathscr{S}|$  there exists a modification  $g': X \to |\mathscr{S}|$  of f such that an interior point p of  $|\mathscr{S}|$ , i.e., a point  $p \in |\mathscr{S}| \setminus |\mathscr{S}_0|$ , does not belong to g'(X), then there also exists a modification  $g: X \to |\mathscr{S}|$  of f such that  $g|f^{-1}(|\mathscr{S}_0|) = f|f^{-1}(|\mathscr{S}_0|)$  and  $p \in |\mathscr{S}| \setminus g(X)$ .

*Hint.* There exists an open set  $U \subset X$  containing  $f^{-1}(|\mathscr{S}_0|)$  and such that the segment with end-points f(x) and g(x) does not contain the point p for any  $x \in U$ .

(d) Show that a continuous mapping  $f: X \to |\mathcal{K}|$  is irreducible if and only if, for every  $S \in \mathcal{K}$  such that an interior point of S belongs to f(X), the restriction  $f_S: f^{-1}(S) \to S$  is an essential mapping (see Problem 1.9.A).

(e) Prove that for every continuous mapping  $f: X \to |\mathcal{K}|$  there exist an irreducible modification  $g: X \to |\mathcal{K}|$  of f and a subcomplex  $\mathcal{K}_0$  of  $\mathcal{K}$  such that  $g(X) = |\mathcal{K}_0|$ .

Hint. To begin with, apply (c) to observe that if for a simplex  $S \in \mathcal{K}$  the restriction  $f_S: f^{-1}(S) \to S$  is essential, then for every face T of S the restriction  $f_T: f^{-1}(T) \to T$  also it essential; then modify the mapping f on all simplexes for which the corresponding restriction is not essential, beginning with the simplexes of the highest dimension.

#### Problems

**1.13.G.** (a) (Pasynkov [1966]) Prove that if  $\Pi$  is a hereditary family of polyhedra, then a compact metric space X is  $\Pi$ -like if and only if X is homeomorphic to the limit of an inverse sequence  $\{K_i, \pi_j^i\}$ , where  $K_i \in \Pi$  for i = 1, 2, ... and  $\pi_j^i$  is a quasi-simplicial mapping of  $K_i$  onto  $K_j$  for every  $j \leq i$ .

*Hint.* Apply Problem 1.13. F(e) to define a simplicial complex  $\mathscr{K}_1$ , a barycentric subdivision  $\mathscr{P}_1$  of  $\mathscr{K}_1$ , and an irreducible mapping  $f_1: X \to |\mathscr{P}_1|$  of X onto  $|\mathscr{P}_1|$  in such a way that  $|\mathscr{K}_1| \in \mathbf{\Pi}$  and the inverse images of stars of vertices of  $\mathscr{P}_1$  under  $f_1$  all have diameters less than 1/2.

Assume that for each j < i a simplicial complex  $\mathscr{K}_j$ , a barycentric subdivision  $\mathscr{P}_j$  of  $\mathscr{K}_j$  and an irreducible mapping  $f_j: X \to |\mathscr{P}_j|$  of X onto  $|\mathscr{P}_j|$  are defined in such a way that  $|\mathscr{K}_j| \in \mathbf{\Pi}$  and the inverse images of stars of vertices of  $\mathscr{P}_j$  under  $f_j$  all have diameters less than  $1/2^j$ ; assume, moreover, that for each pair k, j of integers satisfying  $k \leq j < i$  a quasi-simplicial mapping  $\pi_k^j$  of  $|\mathscr{P}_j|$  onto  $|\mathscr{P}_k|$  is defined in such a way that  $\pi_k^k = \mathrm{id}_{|\mathscr{P}_k|}, \ \pi_l^k \pi_k^j = \pi_l^j$  whenever  $l \leq k$ , and  $\varrho(\pi_k^{j-1}f_{j-1}(x), \pi_k^j f_j(x)) < 1/2^j$  for  $x \in X$  and  $k \leq j-1$ .

Let  $\mathscr{P}$  be a barycentric subdivision of  $\mathscr{P}_{i-1}$  such that mesh $(\{\pi_k^{i-1}(S): S \in \mathscr{P}\}) < 1/2^i$  for every  $k \leq i-1$ ; consider the cover  $\mathscr{U}$  of the space X consisting of inverse images of stars of vertices of  $\mathscr{P}$  under  $f_{i-1}$  and let  $\delta_i = \min(\varepsilon_i, 1/2^i)$ , where  $\varepsilon_i$  is a Lebesgue number for the cover  $\mathscr{U}$ . Define a simplicial complex  $\mathscr{K}_i$ , a barycentric subdivision  $\mathscr{P}_i$  of  $\mathscr{K}_i$  and an irreducible mapping  $f_i: X \to |\mathscr{P}_i|$  of X onto  $|\mathscr{P}_i|$  in such a way that  $|\mathscr{K}_i| \in \Pi$  and the inverse images of stars of vertices of  $\mathscr{P}_i$  all have diameters less than  $\delta_i$ . Observe that by assigning to each vertex  $p \in \mathscr{P}_i$  a vertex  $q \in \mathscr{P}$  such that  $f_i^{-1}(\operatorname{St}_{\mathscr{P}_i}(p)) \subset f_{i-1}^{-1}(\operatorname{St}_{\mathscr{P}_i}(q))$  one defines a simplicial mapping  $\pi_{i-1}^i$ :  $|\mathscr{P}_i| \to |\mathscr{P}_{i-1}|$ . Check that if  $f_{i-1}(x) \in S \in \mathscr{P}$ , then  $\pi_{i-1}^i f_i(x) \in S$ ; deduce that  $\pi_{i-1}^i(|\mathscr{P}_i|) = |\mathscr{P}_{i-1}|$  and that  $\varrho(\pi_k^{-1}f_{i-1}(x), \pi_k^{-1}f_i(x)) < 1/2^i$  for  $x \in X$  and  $k \leq i-1$ , where  $\pi_k^i = \pi_k^{-1}\pi_{i-1}^i$ .

Note that for every natural number k the sequence of compositions  $\pi_k^{k+1}f_{k+1}, \pi_k^{k+2}f_{k+2}, \dots$  uniformly converges to a continuous mapping  $g_k$ :  $X \to |\mathscr{P}_k|$  and that  $\pi_k^{k+1}g_{k+1} = g_k$ . Check that X is homeomorphic to the limit of the inverse sequence  $\{|\mathscr{K}_i|, \pi_j^i\}$ .

(b) (Freudenthal [1937]) Prove that a compact metric space X satisfies the inequality dim  $X \le n$  if and only if X is homeomorphic to the limit of an inverse sequence  $\{K_i, \pi_j^i\}$ , where  $K_i$  is a polyhedron of dimension  $\le n$  for i = 1, 2, ... and  $\pi_j^i$  is a quasi-simplicial mapping of  $K_i$  onto  $K_j$ for every  $j \le i$ .

## 1.14. Dimension and axioms

Since the origin of dimension theory attempts have been made to characterize dimension functions by a few simple properties that could serve as a basis for an axiomatization of the theory. However, no satisfactory set of axioms has been proposed so far; the main drawbacks are that the axioms include properties which are either somewhat artificial or too close to the definition of dimension and that no part of dimension theory, no matter how small, can be deduced from the axioms. As the problem of axiomatization of dimension theory is of secondary importance, we shall confine ourselves to a rather sketchy discussion of this topic.

We shall consider a class  $\mathscr{K}$  of topological spaces, which together with each space X contains all closed subspaces of X, and a function d defined on  $\mathscr{K}$ , having values which are integers larger than or equal to -1 or the "infinite number"  $\infty$ , and such that d(X) = d(Y) for each pair of homeomorphic spaces X,  $Y \in \mathscr{K}$ . By assuming that the function f satisfies some simple conditions which are known to be satisfied by the function dim we shall obtain three sets of axioms for dimension theory.

We begin with Alexandroff's axioms; in this instance  $\mathscr{K}$  is the class of all compact subspaces of Euclidean spaces and the function d satisfies the following conditions:

- (A1)  $d(\emptyset) = -1$ ,  $d(\{0\}) = 0$  and  $d(I^n) = n$  for n = 1, 2, ...
- (A2) If a space  $X \in \mathcal{K}$  is represented as the union of two closed subspaces  $X_1$  and  $X_2$ , then  $d(X) = \max(d(X_1), d(X_2))$ .
- (A3) For every space  $X \in \mathcal{K}$  there exists a positive number  $\varepsilon$  such that if  $f: X \to Y$  is an  $\varepsilon$ -mapping of X onto a space  $Y \in \mathcal{K}$ , then  $d(X) \leq d(Y)$ .
- (A4) For every space  $X \in \mathcal{K}$  of cardinality larger than one there exists a closed set  $L \subset X$  separating X and such that d(L) < d(X).

**1.14.1. Theorem.** The covering dimension dim is the only function d which satisfies conditions (A1)–(A4) in the class  $\mathcal{K}$  of all compact subspaces of Euclidean spaces.

**Proof.** Clearly, the function  $d = \dim$  satisfies conditions (A1)-(A4). Consider now a function d which satisfies (A1)-(A4). It follows from (A1) and (A2) that if K is a polyhedron, then  $d(K) = \dim K$ , so that by virtue of (A3) and the theorem on  $\varepsilon$ -mappings,  $d(X) \leq \dim X$  for every  $X \in \mathcal{K}$ .

Assume that there exists a space  $X \in \mathscr{K}$  such that  $d(X) < \dim X$ . Let d(X) = k and  $\dim X = n$ ; it follows from (A1) that  $n \ge 1$ . Without loss

of generality one can suppose that each  $Y \in \mathscr{K}$  such that d(Y) < k satisfies the equality  $d(Y) = \dim Y$ . By virtue of Theorems 1.9.8 and 1.7.7, the space X contains an n-dimensional Cantor manifold M. From (A2) it follows that  $d(M) \leq d(X) < n = \dim M$ ; since  $d(M) \neq \dim Y$ , we have d(M) = k. Now apply (A4) to obtain a closed set  $L \subset M$  separating M and such that d(L) < d(M). Thus we have  $\dim L = d(L) < k < n$ , so that  $\dim L$  $\leq n-2$ , which contradicts the definition of a Cantor manifold. Hence  $d(X) = \dim X$  for every  $X \in \mathscr{K}$ .  $\Box$ 

It turns out that by replacing condition (A2) with

(A2') If a space  $X \in \mathcal{K}$  is represented as the union of a sequence  $X_1, X_2, ...$ of closed subspaces, then  $d(X) = \sup \{ d(X_i) : i = 1, 2, ... \}$ 

one obtains a set of conditions which characterizes the covering dimension dim in the class  $\mathscr{K}$  of all subspaces of Euclidean spaces (see Problem 1.14.C).

We now pass to Nishiura's axioms; in this instance  $\mathcal{K}$  is the class of all separable metric spaces and the function d satisfies the following conditions:

(N1)  $d(\{0\}) = 0.$ 

(N2) If Y is a subspace of a space  $X \in \mathcal{K}$ , then  $d(Y) \leq d(X)$ .

- (N3) If a space  $X \in \mathcal{K}$  is represented as the union of a sequence  $X_1, X_2, ...$ of closed subspaces, then  $d(X) = \sup \{d(X_i): i = 1, 2, ...\}$ .
- (N4) If a space  $X \in \mathscr{K}$  is represented as the union of two subspaces  $X_1$  and  $X_2$ , then  $d(X) \leq d(X_1) + d(X_2) + 1$ .
- (N5) For every space  $X \in \mathcal{K}$  there exists a compactification  $\tilde{X} \in \mathcal{K}$  such that  $d(\tilde{X}) = d(X)$ .
- (N6) If a non-empty space  $X \in \mathcal{K}$  satisfies the inequality  $d(X) < \infty$ , then for every point  $x \in X$  and each neighbourhood  $V \subset X$  of the point xthere exists an open set  $U \subset X$  such that

 $x \in U \subset V$  and  $d(\operatorname{Fr} U) \leq d(X) - 1$ .

One can prove that the covering dimension dim is the only function d which satisfies conditions (N1)–(N6) in the class  $\mathcal{K}$  of all separable metric spaces (see Problem 1.14.B).

We conclude with *Menger's axioms*, chronologically the earliest set of axioms for dimension theory; in this instance  $\mathscr{K}$  is the class of all subspaces of Euclidean *m*-space  $\mathbb{R}^m$  and the function *d* satisfies the following conditions:

- (M1)  $d(\emptyset) = -1$ ,  $d(\{0\}) = 0$  and  $d(\mathbb{R}^n) = n$  for n = 1, 2, ..., m.
- (M2) If Y is a subspace of a space  $X \in \mathcal{K}$ , then  $d(Y) \leq d(X)$ .
- (M3) If a space  $X \in \mathcal{H}$  is represented as the union of a sequence  $X_1, X_2, ...$ of closed subspaces, then  $d(X) \leq \sup \{d(X_i): i = 1, 2, ...\}$ .
- (M4) For every space  $X \in \mathcal{K}$  there exists a compactification  $\tilde{X} \in \mathcal{K}$  such that  $d(\tilde{X}) = d(X)$ .

Menger put forward the hypothesis that for every natural number m the covering dimension dim is the only function d which satisfies conditions (M1)-(M4) in the class of all subspaces of Euclidean m-space  $R^m$  and showed that the hypothesis is valid for  $m \leq 2$  (see Problem 1.14.D). The problem whether the hypothesis is valid for m > 2 is still open. Let us recall (cf. the discussion in the final part of Section 1.11) that for m > 3 it is not even known whether the function  $d = \dim$  satisfies condition (M4). Clearly, the covering dimension dim satisfies conditions (M1)-(M4) in the class  $\mathscr{K}$  of all subspaces of Euclidean spaces. We find, however, that dim is not the only function with this property; each cohomological dimension dim<sub>G</sub> with respect to a finitely generated abelian group G also satisfies conditions (M1)-(M4) in the class  $\mathscr{K}$  of all subspaces of Euclidean spaces.

# Historical and bibliographic notes

Theorem 1.14.1 was proved by Alexandroff in [1932]. Ščepin announced in [1972] that by replacing condition (A2) by the stronger condition (A2') one obtains a set of axioms that characterizes the covering dimension dim in the class of all subspaces of Euclidean spaces; the proof was published in Alexandroff and Pasynkov's book [1973], where a stronger result, also due to Ščepin, is announced, viz., that the same set of axioms characterizes dim in the class of all metric spaces whose covering dimension is finite. Nishiura proved in [1966] that the axioms (N1)–(N6) characterize the covering dimension dim in the class of all separable metric spaces. Sakai in [1968] and Aarts in [1971] modified Nishiura's axioms to obtain a set of axioms that characterizes dim in the class of all metric spaces. The fact that conditions (M1)–(M4) characterize the covering dimension dim in the class of all subspaces of the plane, and also in the class of all subspaces of the real line, was established by Menger in [1929]. The theorem that each cohomological dimension with respect to a finitely generated

#### Problems

abelian group satisfies conditions (M1)-(M4) in the class of all subspaces of Euclidean spaces was proved by I. Švedov; the proof was first published in Kuz'minov's paper [1968]. A set of axioms characterizing the covering dimension dim in the class of all (not necessarily metric) compact spaces whose dimension dim is finite was given by Lokucievskiĭ in [1973].

#### Problems

1.14.A (Ščepin, cited in Alexandroff and Pasynkov [1973]; announcement Ščepin [1972]). Verify that the axioms (A1)-(A4) are independent.

**1.14.B** (Nishiura [1966]). (a) Prove that the covering dimension dim is the only function d which satisfies conditions (N1)-(N6) in the class  $\mathscr{K}$  of all separable metric spaces.

(b) Verify that the axioms (N1)-(N6) are independent.

*Hint.* To verify that (N2) is independent of the remaining axioms, observe that every separable metric space which is not compact has an infinite-dimensional compactification.

1.14.C (Sčepin, cited in Alexandroff and Pasynkov [1973]; announcement Ščepin [1972]). (a) Show that for every separable metric space X such that dim $X = n \ge 0$  there exists a separable metric space  $X_*$  $= \{x, y\} \cup \bigcup_{i=1}^{\infty} X_i$ , where for  $i = 1, 2, ..., X_i$  is a closed subspace of  $X_*$ homeomorphic to X, with the property that no closed set  $L \subset X_*$  satisfying the inequality dim $L \le n-2$  separates the space  $X_*$  between x and y. (b) Show that for every separable metric space X such that

dim  $X = n \ge 0$  there exists a separable metric space  $X^* = \bigcup_{i=1}^{\infty} X_i$ , where, for  $i = 1, 2, ..., X_i$  is a closed subspace of  $X^*$  homeomorphic to X, with the property that no closed set  $L \subset X^*$  satisfying the inequality dim  $L \le n-2$  separates the space  $X^*$ .

(c) Prove that the covering dimension dim is the only function d which satisfies conditions (A1), (A2'), (A3) and (A4) in the class  $\mathscr{K}$  of all subspaces of Euclidean spaces.

(d) Verify that (A2') is independent of the axioms (A1)-(A4).

**1.14.D.** (a) (Menger [1929]) Check that the covering dimension dim is the only function d which satisfies conditions (M1)–(M4) in the class  $\mathcal{K}$  of all subspaces of the real line.

(b) (Kuratowski and Menger [1930]) Applying the Denjoy-Riesz theorem, i.e., the fact that every zero-dimensional compact subspace of the plane is contained in an arc  $L \subset R^2$  (see Kuratowski [1968], p. 539), and the Moore theorem, i.e., the fact that if there exists a continuous mapping  $f: S^2 \to X$  of the two-sphere onto a space X such that the fibres of f are connected and do not separate  $S^2$ , then the space X is homeomorphic to  $S^2$  (see Kuratowski [1968], p. 533), prove that every zero-dimensional  $F_{\sigma}$ -set in the plane is contained in the union of a sequence of arcs which are pairwise disjoint and have diameters converging to zero.

*Hint.* Show that if a zero-dimensional compact set A is contained in an open set  $U \subset S^2$ , then for every  $\varepsilon > 0$  there exists a sequence  $L_1, L_2, \ldots$  of arcs such that  $L_i \cap L_j = \emptyset$  whenever  $i \neq j$ ,  $\lim \delta(L_i) = 0$ ,  $\delta(L_i) < \varepsilon$  for  $i = 1, 2, \ldots$ , and  $A \subset \bigcup_{i=1}^{\infty} L_i \subset \bigcup_{i=1}^{\infty} L_i \subset U$ . Consider first the case where the set U is connected, observe that no component of  $S^2 \setminus U$ separates  $S^2$  and apply the Moore theorem.

(c) (Menger [1929], Kuratowski and Menger [1930]) Applying (b) and the Moore theorem, prove that the covering dimension dim is the only function d which satisfies conditions (M1)-(M4) in the class  $\mathcal{K}$  of all subspaces of the plane.

Hint. Let  $\mathcal{D}_k = \{X \subset R^2 : d(X) \leq k\}$  for k = -1, 0, 1, 2; it suffices to show that if the family  $\mathcal{D}_k$  contains an *n*-dimensional space, then  $\mathcal{D}_k$  contains all *n*-dimensional subspaces of the plane. Only the case of n = 1 and k = 0, 1 is non-trivial. Let  $X \in \mathcal{D}_k$  be a one-dimensional space; by virtue of Corollary 1.9.9 and condition (M4) one can assume that X is a continuum. Place a copy of the continuum X in a square K in such a way that the four corners of K are the only points of the boundary of K which belong to X. Then divide K into 9 congruent squares and place in the same way a copy of X in each of these smaller squares. Continue the procedure, dividing K consecutively into smaller and smaller squares, and consider the union X\* of countably many smaller and smaller copies of X placed in these squares. Prove that  $d(X^*) = k$  and  $ind(K \setminus X^*)$ = 0. Apply the Lavrentieff theorem (see [GT], Theorem 4.3.21) and condition (M4) to obtain a  $G_{\delta}$ -set Y such that  $X^* \subset Y \subset K$  and d(Y) = k. Then apply (b) to the zero-dimensional  $F_{\alpha}$ -set Int $K \setminus Y$  and denote by

#### Problems

 $L_1, L_2, ...$  the arcs in (b). Deduce from the Moore theorem that the space obtained from  $R^2$  by identifying to points the set  $R^2 \setminus \text{Int } K$  and each of the arcs  $L_i$  is homeomorphic to  $S^2$ . Show that the space Y contains a subspace homeomorphic to the complement of a countable dense subset of  $R^2$  and deduce from Problem 1.8.D that Y is a universal space for the class of all one-dimensional subspaces of the plane.

(d) (Menger [1929]) Verify that the axioms (M1)–(M4) are independent in the class  $\mathcal{K}$  of all subspaces of the plane.

# CHAPTER 2

# THE LARGE INDUCTIVE DIMENSION

Outside the class of separable metric spaces the dimensions ind, Ind and dim generally do not coincide. Nevertheless, a number of theorems established in Chapter 1 extend beyond this class of spaces. In larger classes they hold either for the dimension Ind, or for the dimension dim, or else for both Ind and dim. The dimension ind is practically of no importance outside the class of separable metric spaces and from now on will reappear here only occasionally. Slightly exaggerating, one could say that ind is a satisfactory dimension function only when it is equal to Ind. Thus, for general spaces we have two separate dimension theories: the theory of the large inductive dimension Ind and the theory of the covering dimension dim. They are both poorer and less harmonius than the dimension theory of separable metric spaces, yet they contain many interesting theorems and shed light on classical dimension theory. It should be noted that, while the dimension dim behaves properly in the class of all normal spaces, the dimension Ind does so only in the more restricted class of strongly hereditarily normal spaces. The present chapter and the next are devoted to a closer study of Ind and of dim, respectively. In the final chapter it will be proved that the dimensions Ind and dim coincide in the class of all metrizable spaces and that a dimension theory can be developed in that class which is by no means inferior to the dimension theory of separable metric spaces.

Section 2.1 contains supplementary information about hereditarily normal spaces and an investigation of the class of strongly hereditarily normal spaces.

In Section 2.2 those rare theorems on the dimension Ind are proved which hold either in all normal spaces or in all hereditarily normal spaces. In the final part of the section two important examples are described, showing that neither the subspace theorem nor the sum theorem holds for the dimension Ind in the class of all normal spaces. The large inductive dimension

Section 2.3 is crucial for the present chapter; we develop in it a dimension theory for Ind in the class of strongly hereditarily normal spaces. The main results of this theory are the subspace theorem and a group of sum theorems.

The last section is devoted to a study of the relations between ind and Ind and to the Cartesian product theorems for the dimension Ind.

# 2.1. Hereditarily normal and strongly hereditarily normal spaces

The large inductive dimension Ind is defined for all normal spaces (see Definition 1.6.1). It turns out however, that in such an extensive class of spaces the dimension Ind develops some pathological properties. As the reader will see in Section 2.2, there exist a compact space Z and a normal subspace X of Z such that IndX > IndZ. There also exists a compact space X with Ind X = 2 which can be represented as the union of two closed subspaces  $F_1$  and  $F_2$  such that  $IndF_1 = IndF_2 = 1$ . Finally, there exist compact spaces X and Y such that  $Ind(X \times Y) > IndX + IndY$ . Besides, since a subspace of a normal space is not necessarily normal (see [GT], Example 2.3.36 or 3.2.7), it may happen that the dimension Ind is defined for a space X, and yet is not defined for a subspace M of X. From all the adduced phenomena one gathers that to develop a satisfactory theory of the large inductive dimension Ind one has to restrict the class of spaces under consideration. As the spaces in all the examples cited above are not hereditarily normal, we might expect that no such pathological phenomena can occur in the class of hereditarily normal spaces. Still, as has recently been shown, the dimension Ind is not monotonic in the latter class, so that a further restriction of the class of spaces is necessary.

A natural class of spaces where a satisfactory theory of the large inductive dimension can be developed is the class of strongly hereditarily normal spaces; it is contained in the class of all hereditarily normal spaces and constitutes a common extension of the class of perfectly normal spaces and the class of hereditarily paracompact spaces. The present section is devoted to a study of the topological properties of hereditarily normal spaces and strongly hereditarily normal spaces.

Let us recall that a space X is *hereditarily normal* if every subspace of X is normal. We begin with two simple characterizations of hereditarily

normal spaces; in the second one appears the notion of separated subsets of a topological space introduced in Section 1.2.

**2.1.1. Theorem.** For every  $T_1$ -space X the following conditions are equivalent:

- (a) The space X is hereditarily normal.
- (b) Every open subspace of X is normal.
- (c) For every pair A, B of separated sets in X there exist open sets  $U, V \subset X$ such that  $A \subset U$ ,  $B \subset V$  and  $U \cap V = \emptyset$ .

**Proof.** The implication (a)  $\Rightarrow$  (b) is obvious. We shall show that (b)  $\Rightarrow$  (c). Consider a pair A, B of separated sets in a space X which satisfies (b) and let  $M = X \setminus (\overline{A} \cap \overline{B})$ . Obviously, M is an open subspace of the space X and A,  $B \subset M$ . The closures of A and B in M are disjoint, so that by the normality of M there exist sets U,  $V \subset M$  open in M and such that  $A \subset U$ ,  $B \subset V$  and  $U \cap V = \emptyset$ . The subspace M being open in X, the sets U, V are open in X, so that the space X satisfies (c).

To complete the proof it remains to show that  $(c) \Rightarrow (a)$ . Consider an arbitrary subspace M of a space X which satisfies (c) and a pair A, Bof disjoint closed subsets of M. Clearly A and B are separated in X, so that there exist open sets  $U, V \subset X$  such that  $A \subset U, B \subset V$  and  $U \cap V$  $= \emptyset$ . The intersections  $M \cap U$  and  $M \cap V$  are open in M and disjoint, and contain A and B, respectively, which means that the space X satisfies (a).  $\square$ 

Let us recall that a family  $\{A_s\}_{s\in S}$  of subsets of a topological space X is *point-finite* (*point-countable*) if for every point  $x \in X$  the set  $\{s \in S: x \in A_s\}$  is finite (countable).

**2.1.2. Definition.** A topological space X is called *strongly hereditarily normal* if X is a  $T_1$ -space and for every pair A, B of separated sets in X there exist open sets  $U, V \subset X$  such that  $A \subset U, B \subset V, U \cap V = \emptyset$  and U and V can be represented as the union of a point-finite family of open  $F_{\sigma}$ -sets in X.

Obviously, every strongly hereditarily normal space is hereditarily normal; moreover, every subspace M of a strongly hereditarily normal space is strongly hereditarily normal, because any sets A, B separated in M are also separated in X.

Besides hereditary normality one considers another strengthening of normality, namely perfect normality. Let us recall that a space X is *perfectly normal* if X is a normal space and every open subset of X is an  $F_{\sigma}$ -set in X. Perfect normality is a hereditary property; in particular, every perfectly normal space is hereditarily normal (cf. Lemma 3.1.22 below). The latter fact and definition 2.1.2 imply

2.1.3. Theorem. Every perfectly normal space is strongly hereditarily normal.

**2.1.4. Theorem.** Every hereditarily paracompact space is strongly hereditarily normal.

**Proof.** Since every paracompact space is normal (see [GT], Theorem 5.1.5), every hereditarily paracompact space is hereditarily normal. Thus to complete the proof it suffices to show that every open subset U of a hereditarily paracompact space X can be represented as the union of a point-finite family of open  $F_{\sigma}$ -sets in X.

For every  $x \in U$  consider a neighbourhood  $U_x$  of the point x such that  $\overline{U}_x \subset U$ . The family  $\mathscr{U} = \{U_x\}_{x \in U}$  is an open cover of the subspace U of X. Since the space U is paracompact, there exists a locally finite partition of unity  $\{f_s\}_{s \in S}$  on U subordinated to  $\mathscr{U}$  (see [GT], Theorem 5.1.9). For every  $s \in S$  the set  $U_s = f_s^{-1}((0, 1])$  is an open  $F_\sigma$ -set in U. As U is an open subspace of X, the set  $U_s$  is also open in X. Furthermore,  $U_s \subset U_x \subset \overline{U}_x \subset U$  for a certain  $x \in U$ , so that  $U_s$  is an  $F_\sigma$ -set in the closed subspace  $\overline{U}_x$  of X which implies that  $U_s$  is an  $F_\sigma$ -set in X. Finally, the family  $\{U_s\}_{s \in S}$  is point-finite and  $U = \bigcup_{x \in S} U_s$ .  $\Box$ 

We shall now slightly generalize the last theorem. Let us recall that a topological space X is weakly paracompact<sup>1)</sup> if X is a Hausdorff space and every open cover of the space X has a point-finite open refinement. Weakly paracompact spaces are not necessarily normal (see [GT], Example 5.3.4 or Exercise 5.3.B(b)).

**2.1.5. Theorem.** Every hereditarily weakly paracompact hereditarily normal space is strongly hereditarily normal.

**Proof.** It suffices to show that every open subset U of a hereditarily weakly paracompact hereditarily normal space X can be represented as the union of a point-finite family of open  $F_{\sigma}$ -sets in X.

1) The terms metacompact and point-paracompact are also used.

164

For every  $x \in U$  consider a neighbourhood  $U_x$  of the point x such that  $\overline{U}_x \subset U$ . The family  $\mathscr{U} = \{U_x\}_{x \in U}$  is an open cover of the subspace U of X. Since the space U is weakly paracompact, the cover  $\mathscr{U}$  has a point-finite open refinement  $\mathscr{V} = \{V_s\}_{s \in S}$ . The cover  $\mathscr{V}$ , as any point-finite open cover of a normal space (see [GT], Theorem 1.5.18), has a closed shrinking  $\{F_s\}_{s \in S}$ . By virtue of Urysohn's lemma for every  $s \in S$  there exists a continuous function  $f_s: U \to I$  such that  $f_s(U \setminus V_s) \subset \{0\}$  and  $f_s(F_s) \subset \{1\}$ . For every  $s \in S$  the set  $U_s = f_s^{-1}((0, 1])$  is an open  $F_{\sigma}$ -set in U. Obviously, the set  $U_s$  is open in X. Furthermore,  $U_s \subset V_s \subset U_x \subset \overline{U}_x \subset U$  for a certain  $x \in U$ , so that  $U_s$  is an  $F_{\sigma}$ -set in X. Finally, the family  $\{U_s\}_{s \in S}$  is point-finite and  $U = \bigcup_{s \in S} U_s$ .  $\Box$ 

We conclude this section with two examples: an example of a compact strongly hereditarily normal space which is neither perfectly normal nor hereditarily weakly paracompact and an example of a compact hereditarily normal space which is not strongly hereditarily normal.

**2.1.6. Example.** Let W be the set of all ordinal numbers less than or equal to the first uncountable ordinal number  $\omega_1$ . The set W is well-ordered by the natural order <. Consider on W the topology obtained by taking as a base all sets of the form

(1) 
$$(\alpha, \omega_1] = \{x: \alpha < x\}, \quad [0, \beta) = \{x: x < \beta\}$$
and 
$$(\alpha, \beta) = \{x: \alpha < x < \beta\},$$

where  $\alpha < \beta \leq \omega_1$ . One easily sees that W is a Hausdorff space. We shall show that W is compact.

Let  $\{U_s\}_{s\in S}$  be an open cover of the space W and let A consist of all  $\alpha \in W$  such that the set  $[0, \alpha] = \{x \colon x \leq \alpha\}$  is contained in the union of finitely many members of the cover under consideration. It suffices to show that  $W \setminus A = \emptyset$ .

Assume that  $W \setminus A \neq \emptyset$  and denote by  $x_0$  the smallest element of this set. Choose an  $s_0 \in S$  such that  $x_0 \in U_{s_0}$ ; since  $x_0 > 0$ , there exists an  $x < x_0$  such that  $(x, x_0] \subset U_{s_0}$ . By the definition of  $x_0$ , the point x belongs to A, so that  $[0, x] \subset \bigcup_{i=1}^{k} U_{s_i}$ . It follows that  $[0, x_0] \subset \bigcup_{i=0}^{k} U_{s_i}$ , and we have a contradiction.

We shall now prove that every open subspace U of W is normal, i.e., that the space W is hereditarily normal. We shall say that a set  $C \subset W$ is *convex* if  $(\alpha, \beta) \subset C$  whenever  $\alpha, \beta \in C$ . One readily sees that the union of a family of convex sets is convex provided that the intersection of the family is non-empty. Hence, considering for each point  $x \in U$  the union of all convex subsets of U which contain that point, we obtain a decomposition of the set U into pairwise disjoint convex sets  $\{U_s\}_{s\in S}$ , which will be called the *convex components* of the set U. The set U being open, all  $U_s$ 's are open subsets of W, so that  $U = \bigoplus_{s\in S} U_s$ , where the symbol  $\bigoplus$ denotes the sum of topological spaces (see [GT], Section 2.2). Since the set W is well-ordered by <, every open and convex proper subset of W is of form (1). Thus, to prove that U is normal it suffices to show that all subspaces of form (1) are normal. The subspaces of the form  $(\alpha, \omega_1]$  $= [\alpha + 1, \omega_1]$ , where  $\alpha < \omega_1$ , are normal as closed subspaces of the normal space W. The subspaces of the form  $[0, \beta)$  and  $(\alpha, \beta)$ , where  $\beta < \omega_1$ , are regular second-countable spaces and thus are metrizable (see [GT], Theorem 4.2.9) and, *a fortiori*, normal. It remains to prove that the subspace  $W_0 = W \setminus \{\omega_1\}$  of the space W is normal.

We shall show more, viz., that for every pair A, B of disjoint closed subsets of  $W_0$  the closures  $\overline{A}$  and  $\overline{B}$  of A and B in the space W are disjoint. This follows from the fact that  $\omega_1$  belongs to at most one of the sets  $\overline{A}$  and  $\overline{B}$ . Indeed, if we had  $\omega_1 \in \overline{A} \cap \overline{B}$ , we could define inductively two sequences,  $\alpha_1, \alpha_2, \ldots$  and  $\beta_1, \beta_2, \ldots$ , of countable ordinal numbers satisfying

$$\alpha_i < \beta_i < \alpha_{i+1}, \quad \alpha_i \in A, \quad \beta_i \in B \quad \text{for } i = 1, 2, ...;$$

then the smallest ordinal number  $\gamma$  larger than all  $\alpha_i$ 's and  $\beta_i$ 's would belong to  $\overline{A \cap B}$ , which is impossible, because  $\gamma < \omega_1$  since the set  $W_0$ contains no countable cofinal subset. Let us recall that a subset K of an ordered set X is *cofinal* in X if for every  $\alpha \in X$  there exists a  $\beta \in K$  such that  $\alpha \leq \beta$ ; a set  $K \subset X$  is *bounded* in X, if it is not cofinal in X.

Since the space W is hereditarily normal, in proving that W is a strongly hereditarily normal space it suffices to consider a pair of open separated sets, i.e., disjoint open sets  $U, V \subset X$ . Let  $U = \bigoplus_{s \in S} U_s$  and  $V = \bigoplus_{t \in T} V_t$ be the decompositions of U and V into convex components. If all the convex components  $U_s$  and  $V_t$  are bounded in W, they are countable and, a fortiori, they are  $F_{\sigma}$ -sets in W. So, in this case U and V can themselves be represented as the union of a point-finite family of open  $F_{\sigma}$ -sets in W. On the other hand, if one of the convex components, say the convex component  $U_{s_0}$ , of the set U is cofinal in W, then the set V is bounded in W. In this case the sets U and V are contained, respectively, in disjoint open sets  $U \cup \{\omega_1\}$ and V which can be represented as the union of a point-finite family of open  $F_{\sigma}$ -sets in W. Every closed subset of W which is contained in  $W_0$  is bounded in W. so that every  $F_{\sigma}$ -set in W which is contained in  $W_0$  also is bounded in W. It follows that the open subset  $W_0$  of the space W is not an  $F_{\sigma}$ -set in W. Thus the space W is not perfectly normal.

To prove that the space W is not hereditarily weakly paracompact it is enough to show that if  $\mathscr{U} = \{U_s\}_{s \in S}$  is an open cover of the subspace  $W_0$  of W and all  $U_s$ 's are bounded in  $W_0$ , then  $\mathscr{U}$  is not point-finite. Suppose that  $\mathscr{U}$  is point-finite. Hence for every  $x \in W_0$  the set  $St(x, \mathscr{U})$  $= \bigcup \{U_s: x \in U_s\}$ , i.e., the star of the point x with respect to the cover  $\mathscr{U}$ , is bounded; thus one can define inductively an increasing sequence  $\alpha_1$  $< \alpha_2 < \ldots$  of countable ordinal numbers satisfying

(2)  $\alpha_{i+1} \notin \operatorname{St}(\alpha_i, \mathscr{U})$  for i = 1, 2, ...

The smallest ordinal number  $\gamma$  larger than all  $\alpha_i$ 's belongs to a member  $U_{s_0}$  of the cover  $\mathscr{U}$ . The set  $U_{s_0}$  is open and thus contains almost all  $\alpha_i$ 's. This contradiction of (2) shows that the cover  $\mathscr{U}$  is not point-finite.  $\Box$ 

We now turn to the example of a compact hereditarily normal space which is not strongly hereditarily normal.

2.1.7. Example. Let W' be a topological space homeomorphic to the space W described in Example 2.1.6 and such that  $W' \cap W = \emptyset$ , and let  $W'_0$ and  $\omega'_1$  denote the counterparts of  $W_0$  and  $\omega_1$  in W'. The sum  $W \oplus W'$  is hereditarily normal and so is the space X obtained from  $W \oplus W'$  by identifying the points  $\omega_1$  and  $\omega'_1$ , i.e., the quotient space determined by the decomposition of  $W \oplus W'$  into the set  $\{\omega_1, \omega'_1\}$  and all one-point sets  $\{x\}$ with  $x \in W_0 \cup W'_0$ . However, the space X is not strongly hereditarily normal, because for its separated subsets  $q(W_0)$  and  $q(W'_0)$ , where  $q: W \oplus W' \to X$ is the natural quotient mapping, there exist no disjoint open sets U, V, which can be represented as the union of a point-finite family of open  $F_{\sigma}$ -sets in X, such that  $q(W_0) \subset U$  and  $q(W'_0) \subset V$ . Indeed, the only disjoint open sets  $U, V \subset X$  that contain  $q(W_0)$  and  $q(W'_0)$ , respectively, are  $U = q(W_0)$  and  $V = q(W'_0)$ . Now, if  $q(W_0)$  could be represented as the union of a point-finite family of open  $F_{\sigma}$ -sets in X, the subspace  $W_0$  of W would have a point-finite open cover by sets bounded in  $W_0$ , whereas, by the last part of Example 2.1.6, no such cover exists.

#### Historical and bibliographic notes

Various restrictions of the class of normal spaces to a class where a satisfactory theory of the large inductive dimension Ind can be developed

167

have been proposed more than once. The first step was made by Čech in [1932], who developed the theory of the large inductive dimension in the class of perfectly normal spaces. Then, Dowker in [1953] introduced the class of totally normal spaces and extended the theory to that class. Let us recall here that a topological space X is totally normal, if X is a normal space and every open subset U of X can be represented as the union of a locally finite in U family of open  $F_{\sigma}$ -sets in X. Clearly, every perfectly normal space is totally normal and so is every hereditarily paracompact space; the fact that every totally normal space is hereditarily normal is by no means obvious (see Problem 2.1.C). Subsequently, Pasynkov in [1967] defined Dowker spaces as hereditarily normal spaces in which every open set can be represented as the union of a point-finite family of open  $F_{\sigma}$ -sets, and announced extensions of some theorems on Ind to this class of spaces. Proofs of the announced theorems, together with further results, were published by Lifanov and Pasynkov in [1970]. Finally, Nishiura in [1977] introduced the class of super normal spaces and correspondingly extended the theory of the large inductive dimension; according to Nishiura, a topological space X is called super normal if X is a  $T_1$ -space and for every pair A, B of separated sets in X there exist open sets  $U, V \subset X$  such that  $A \subset U$ ,  $B \subset V$ ,  $U \cap V = \emptyset$  and U and V can be represented as the union of a locally finite, in U and V respectively, family of open  $F_{\sigma}$ -sets in X. Our class of strongly hereditarily normal spaces is obtained by amalgamating the ideas of Pasynkov and Nishiura. Let us add that in the process of extending the theory of the large inductive dimension from the class of totally normal spaces to the larger classes mentioned above, only slight modifications in Dowker's original arguments were necessary.

#### Problems

**2.1.A.** Deduce from Urysohn's lemma that a subset A of a normal space X is an open  $F_{\sigma}$ -set (a closed  $G_{\delta}$ -set) if and only if there exists a continuous function  $f: X \to I$  such that  $A = f^{-1}((0, 1])$  (such that  $A = f^{-1}(0)$ ).

**2.1.B.** (a) Show that a topological space X is strongly hereditarily normal if and only if X is hereditarily normal and every open domain in X can be represented as the union of a point-finite family of open  $F_{\sigma}$ -sets in X. Let us recall that a subset U of a topological space X is an open domain in X if  $U = \text{Int } \overline{U}$ .

(b) Applying the fact that for every pair U, V of disjoint open sets

#### Problems

in the Cantor cube  $D^{\mathfrak{c}} = \prod_{t \in I} D_t$ , where  $D_t = D$  for  $t \in I$ , there exists a countable set  $I_0 \subset I$  such that the projections of U and V onto  $\prod_{t \in I_0} D_t$ are disjoint (see [GT], Problem 2.7.12 (b)), show that every open domain in  $D^{\mathfrak{c}}$  is an  $F_{\sigma}$ -set. Observe, using Remark 1.3.18, that the Cantor cube  $D^{\mathfrak{c}}$  is not hereditarily normal.

**2.1.C** (Dowker [1953]). (a) Show that if a space X can be represented as the union of a locally finite family  $\{F_s\}_{s\in S}$  of closed subspaces each of which is normal, then X is a normal space.

*Hint*. Map the sum  $\bigoplus_{s \in S} F_s$  onto X and apply the fact that normality is an invariant of closed mappings.

(b) Show that if a space X can be represented as the union of a sequence  $F_1, F_2, \ldots$  of closed normal subspaces such that  $F_i \subset \text{Int} F_{i+1}$  for  $i = 1, 2, \ldots$ , then X is a normal space.

*Hint.* Note that the family  $\{A_i\}_{i=1}^{\infty}$ , where  $A_1 = F_1$  and  $A_i = F_i \setminus \text{Int} F_{i-1}$  for i > 1, is locally finite.

(c) Show that every totally normal space is hereditarily normal and deduce that every subspace of a totally normal space is totally normal.

Hint. Apply (a), (b) and Problem 2.1.A.

2.1.D. (a) Prove that every paracompact totally normal space is hereditarily paracompact.

(b) Prove that every weakly paracompact Dowker space is hereditarily weakly paracompact.

**2.1.E.** (a) Prove that a  $T_1$ -space X is normal if and only if for every closed set  $F \subset X$  and each open set  $W \subset X$  that contains F there exists a sequence  $W_1, W_2, \ldots$  of open subsets of X such that  $F \subset \bigcup_{i=1}^{\infty} W_i$  and  $\overline{W_i} \subset W$  for  $i = 1, 2, \ldots$ 

*Hint.* Let A and B be disjoint closed subsets of a  $T_1$ -space X which has the property under consideration. Define sequences  $W_1, W_2, \ldots$  and  $V_1, V_2, \ldots$  of open subsets of X such that

$$A \subset \bigcup_{i=1}^{\infty} W_i, \ B \subset \bigcup_{i=1}^{\infty} V_i \text{ and } B \cap \overline{W_i} = \emptyset = A \cap \overline{V_i} \text{ for } i = 1, 2, ...$$

Verify that the sets  $U = \bigcup_{i=1}^{\infty} G_i$  and  $V = \bigcup_{i=1}^{\infty} H_i$ , where  $G_i = W_i \setminus \bigcup_{j \le i} \overline{V}_j$ 

[Ch. 2, § 2

and  $H_i = V_i \bigvee_{j \in I} \overline{W}_j$ , satisfy the conditions  $A \subset U$ ,  $B \subset V$  and  $U \cap V = \emptyset$ .

(b) Prove that a  $T_1$ -space X is perfectly normal if and only if for every open set  $W \subset X$  there exists a sequence  $W_1, W_2, ...$  of open subsets of X such that  $W = \bigcup_{i=1}^{\infty} W_i$  and  $\overline{W_i} \subset W$  for i = 1, 2, ...

# 2.2. Basic properties of the dimension Ind in normal and hereditarily normal spaces

Among the theorems of dimension theory established in Chapter 1 only a few are valid for the dimension Ind in normal or hereditarily normal spaces. As noted above, the dimension Ind is not monotonic in hereditarily normal spaces and the sum theorem for the dimension Ind does not hold in normal spaces, so that one cannot think of developing a dimension theory for Ind in those classes of spaces. In the following section such a theory will be developed in the more restricted class of strongly hereditarily normal spaces. In the present section we merely clear the ground for the considerations of the next one.

The definition of the large inductive dimension Ind was stated in Section 1.6; let us recall that  $\operatorname{Ind} X = -1$  if and only if  $X = \emptyset$ , and that a normal space X satisfies the inequality  $\operatorname{Ind} X \leq n \geq 0$  if and only if for every closed set  $A \subset X$  and each open set  $V \subset X$  which contains the set A there exists an open set  $U \subset X$  such that  $A \subset U \subset V$  and  $\operatorname{Ind} \operatorname{Fr} U \leq n-1$ . In other words,  $\operatorname{Ind} X \leq n \geq 0$  if and only if for every pair A, B of disjoint closed subsets of X there exists a partition L between A and B such that  $\operatorname{Ind} L \leq n-1$ .

Since normality is not a hereditary property, it may happen that the dimension Ind is defined for a space X and yet is not defined for a subspace  $M \subset X$ . Still, normality being hereditary with respect to closed subsets, Ind M is defined for every closed subspace  $M \subset X$ . Moreover in much the same way as Theorem 1.1.2 one obtains

**2.2.1. Theorem.** For every closed subspace M of a normal space X we have  $\operatorname{Ind} M \leq \operatorname{Ind} X$ .  $\Box$ 

The counterpart of Theorem 1.5.1 reads as follows

**2.2.2. Theorem.** If X is a normal space and  $\operatorname{Ind} X = n \ge 1$ , then for k = 0, 1, ..., n-1 the space X contains a closed subspace M such that  $\operatorname{Ind} M = k.\Box$ 

In Example 2.2.11 we shall define a compact space Z which contains a normal subspace X such that  $\operatorname{Ind} X > \operatorname{Ind} Z$ . Hence, in Theorem 2.2.1 the assumption that M is a closed subspace of X cannot be replaced by the weaker assumption that  $\operatorname{Ind} M$  is defined. Recently, a much stronger result was obtained: one defined a hereditarily normal space X such that  $\operatorname{Ind} X = \dim X = 0$ , and yet X contains, for every natural number n, a subspace  $A_n$  with  $\operatorname{Ind} A_n = \dim A_n = n$ . The latter example, however, is too difficult to be described in this book.

We shall now show that for subspaces of a fixed hereditarily normal space X monotonicity of the dimension Ind is equivalent to its being monotonic with respect to open subspaces.

**2.2.3. Proposition.** For every hereditarily normal space X the following conditions are equivalent:

- (a) For each subspace  $Y \subset X$  and every subspace M of Y we have  $\operatorname{Ind} M \leq \operatorname{Ind} Y$ .
- (b) For each subspace  $Y \subset X$  and every open subspace U of Y we have Ind  $U \leq \text{Ind } Y$ .

**Proof.** The implication (a)  $\Rightarrow$  (b) is obvious. Suppose that X satisfies (b). Condition (a) is satisfied if  $\operatorname{Ind} Y = \infty$ , so that it suffices to consider subspaces  $Y \subset X$  with  $\operatorname{Ind} Y < \infty$ . We shall apply induction with respect to  $\operatorname{Ind} Y$  to show that  $\operatorname{Ind} M \leq \operatorname{Ind} Y$  whenever  $M \subset Y$ . Clearly, the inequality holds if  $\operatorname{Ind} Y = -1$ . Assume that the inequality is proved for all subspaces of X the large inductive dimension of which does not exceed  $n-1 \geq -1$  and consider a subspace  $Y \subset X$  with  $\operatorname{Ind} Y = n$  and an arbitrary subspace M of Y. Let A and B be disjoint closed subsets of M. As  $U = Y \setminus (\overline{A} \cap \overline{B})$  is an open subspace of Y, we have  $\operatorname{Ind} U \leq \operatorname{Ind} Y = n$ by virtue of (b). The intersections  $U \cap \overline{A}$  and  $U \cap \overline{B}$  are disjoint closed subsets of U; therefore there exists a partition L in U between  $U \cap \overline{A}$  and  $U \cap \overline{B}$  such that  $\operatorname{Ind} L \leq n-1$ . Since  $M \subset U$ ,  $U \cap \overline{A} \cap M = A$ , and  $U \cap \overline{B} \cap$  $\cap M = B$ , the set  $L \cap M$  is a partition in M between A and B. By the inductive assumption  $\operatorname{Ind}(L \cap M) \leq \operatorname{Ind} L \leq n-1$ , so that  $\operatorname{Ind} M \leq n = \operatorname{Ind} Y$ . Thus X satisfies condition (a).  $\Box$ 

The separation and addition theorems for the dimension Ind hold

in hereditarily normal spaces. From Lemma 1.2.9, Remark 1.2.10 and Theorem 2.2.1 one easily obtains the following theorem (cf. the proof of Theorem 1.2.11):

**2.2.4. The separation theorem for** Ind. If X is a hereditarily normal space and M is a subspace of X such that  $\operatorname{Ind} M \leq n \geq 0$ , then for every pair A, B of disjoint closed subsets of X there exists a partition L between A and B such that  $\operatorname{Ind}(L \cap M) \leq n-1$ .  $\Box$ 

**2.2.5. The addition theorem for** Ind. For every pair X, Y of subspaces of a hereditarily normal space we have  $\operatorname{Ind}(X \cup Y) \leq \operatorname{Ind} X + \operatorname{Ind} Y + 1.$ 

**Proof.** The theorem is obvious if one of the subspaces has dimension  $\infty$ , so that we can suppose that  $m(X, Y) = \operatorname{Ind} X + \operatorname{Ind} Y$  is finite. We shall apply induction with respect to that number. If m(X, Y) = -2, then  $X = \emptyset = Y$  and our inequality holds. Assume that the inequality is proved for every pair of subspaces the sum of the large inductive dimensions of which is less than  $n \ge -1$  and consider subspaces X and Y such that m(X, Y) = n; clearly, we can suppose that  $\operatorname{Ind} X \ge 0$ . Let A and B be disjoint closed subsets of  $X \cup Y$ . By virtue of the separation theorem there exists a partition L in  $X \cup Y$  between A and B such that  $\operatorname{Ind}(L \cap X) \le \operatorname{Ind} X - -1$ . Since  $m(L \cap X, L \cap Y) \le \operatorname{Ind} X + \operatorname{Ind} Y - 1 = n - 1$ , we have  $\operatorname{Ind} L \le n$  by the inductive assumption. This implies that  $\operatorname{Ind}(X \cup Y) \le n+1 = \operatorname{Ind} X + +\operatorname{Ind} Y + 1$ .  $\Box$ 

The addition theorem yields.

**2.2.6. Corollary.** If a hereditarily normal space X can be represented as the union of n+1 subspaces  $Z_1, Z_2, ..., Z_{n+1}$  such that  $\operatorname{Ind} Z_i \leq 0$  for i = 1, 2, ..., n+1, then  $\operatorname{Ind} X \leq n$ .  $\Box$ 

Let us note that it is an open problem whether every normal (or even hereditarily normal) space X with  $IndX \le n$  can be represented as the union of n+1 subspaces the large inductive dimension of which does not exceed zero.

Remark 1.3.2 implies that Theorem 1.3.1 can be restated as follows:

**2.2.7. Theorem.** If a normal space X can be represented as the union of a sequence  $F_1, F_2, \ldots$  of closed subspaces such that  $\operatorname{Ind} F_i \leq 0$  for  $i = 1, 2, \ldots$ , then  $\operatorname{Ind} X \leq 0$ .

Ch. 2, § 2] Dimension Ind in normal and hereditarily normal spaces

Example 2.2.13 below shows that in normal spaces the dimension Ind satisfies only the sum theorem for dimension 0, i.e., Theorem 2.2.7. It is an open problem whether the situation improves in hereditarily normal spaces.

A very weak version of the sum theorem for the dimension Ind in normal spaces reads as follows:

**2.2.8. Proposition.** Let  $\{X_s\}_{s \in S}$  be a family of normal spaces and let  $X = \bigoplus_{s \in S} X_s$ . The inequality  $\operatorname{Ind} X \leq n$  holds if and only if  $\operatorname{Ind} X_s \leq n$  for every  $s \in S$ .

Let us note that a few, rather specialized, results related to Theorems 2.2.5 and 2.2.7 are stated in Problem 2.2.C.

The status of the Cartesian product theorem for Ind in normal spaces is similar to that of the sum theorem. There exist compact spaces X and Y such that indX = IndX = 1, indY = IndY = 2, and yet  $Ind(X \times Y) \ge ind(X \times Y) \ge 4$  as well as a normal space Z, whose square  $Z \times Z$  is also normal, such that IndZ = 0 and yet  $Ind(Z \times Z) > 0$ . The descriptions of these examples are very difficult and cannot be reproduced in this book.

We now turn to a discussion of dimension preserving compactifications. As the reader certainly knows, among the compactifications of a completely regular space X a particular role is played by the Čech-Stone compactification  $\beta X$  (see [GT], Section 3.6), which can be characterized by the property that every continuous function  $f: X \to I$  is continuously extendable over  $\beta X$  (we assume here that X is actually a subspace of  $\beta X$ ). In the realm of normal spaces the Čech-Stone compactification can also be characterized by the property that every pair of disjoint closed subsets of X has disjoint closures in  $\beta X$ . Hence, for every closed subspace M of a normal space X the closure  $\overline{M}$  of M in  $\beta X$  is the Čech-Stone compactification of the space M. We shall show that the Čech-Stone compactification preserves the dimension Ind.

# **2.2.9. Theorem.** For every normal space X we have $\operatorname{Ind} \beta X = \operatorname{Ind} X$ .

**Proof.** To begin with, we shall prove that  $\operatorname{Ind} X \leq \operatorname{Ind} \beta X$ . The inequality is obvious if  $\operatorname{Ind} \beta X = \infty$ , so that we can suppose that  $\operatorname{Ind} \beta X < \infty$ . We shall apply induction with respect to  $\operatorname{Ind} \beta X$ . If  $\operatorname{Ind} \beta X = -1$ , then  $\beta X$  $= \emptyset = X$  and our inequality holds. Assume that the inequality holds for all normal spaces the dimension Ind of the Čech-Stone compactification of which is less than  $n \ge 0$  and consider a normal space X such that  $\operatorname{Ind} \beta X = n$ . Let *A* and *B* be disjoint closed subsets of *X*. The sets  $\overline{A}$  and  $\overline{B}$ , where the bar denotes the closure operator in  $\beta X$ , are disjoint, so that there exists a partition *L* in  $\beta X$  between  $\overline{A}$  and  $\overline{B}$  such that  $\operatorname{Ind} L \leq n-1$ . Clearly,  $L_0 = L \cap X$  is a partition in *X* between *A* and *B*. Since  $\beta L_0 = \overline{L}_0 \subset L$ , it follows from Theorem 2.2.1 and the inductive assumption that  $\operatorname{Ind} L_0 \leq n-1$ , so that  $\operatorname{Ind} X \leq n = \operatorname{Ind} \beta X$ .

Now, we shall prove that  $\operatorname{Ind} \beta X \leq \operatorname{Ind} X$ . As in the first part of the proof, we shall suppose that  $\operatorname{Ind} X < \infty$  and apply induction with respect to  $\operatorname{Ind} X$ . Our equality holds if  $\operatorname{Ind} X = -1$ . Assume that it is proved for all normal spaces the dimension Ind of which is less than  $n \geq 0$  and consider a normal space X such that  $\operatorname{Ind} X = n$ . Let A and B be disjoint closed subsets of  $\beta X$ . There exist open sets  $V_1, V_2 \subset \beta X$  such that

$$A \subset V_1, \quad B \subset V_2 \quad \text{and} \quad \overline{V_1} \cap \overline{V_2} = \emptyset.$$

The sets  $A_0 = X \cap \overline{V_1}$  and  $B_0 = X \cap \overline{V_2}$  are closed in X and disjoint, so that there exists a partition  $L_0$  in X between  $A_0$  and  $B_0$  such that  $\operatorname{Ind} L_0 \leq n-1$ . Let  $U_0$ ,  $W_0$  be open subsets of X satisfying

 $A_0 \subset U_0$ ,  $B_0 \subset W_0$ ,  $U_0 \cap W_0 = \emptyset$  and  $X \setminus L_0 = U_0 \cup W_0$ . We shall show that

(1) 
$$\overline{U_0} \cap \overline{W}_0 \subset \overline{L}_0.$$

Consider a point  $x \in \overline{U}_0 \cap \overline{W}_0$  and an arbitrary neighbourhood  $G \subset \beta X$ of the point x. Let  $H \subset \beta X$  be an open set such that  $x \in H \subset \overline{H} \subset G$ . One easily sees that

$$x \in \overline{W}_0 \cap H \subset \overline{W_0 \cap H}$$
 and  $x \in \overline{U}_0 \cap H \subset \overline{U_0 \cap H};$ 

therefore, by the above-mentioned characterization of the Čech-Stone compactification in the realm of normal spaces, the closures of  $W_0 \cap H$  and  $U_0 \cap H$  in X intersect. Hence

$$\emptyset \neq (W_0 \cup L_0) \cap (U_0 \cup L_0) \cap \overline{H} = [(U_0 \cap W_0) \cup L_0] \cap \overline{H} = L_0 \cap \overline{H} \subset L_0 \cap G,$$

which shows that every neighbourhood of x meets  $L_0$ , i.e., that  $x \in \overline{L_0}$ . Thus inclusion (1) is established.

By virtue of (1) the sets

$$U = \beta X \setminus (\overline{W}_0 \cup \overline{L}_0)$$
 and  $W = \beta X \setminus (\overline{U}_0 \cup \overline{L}_0)$ 

satisfy

(2) 
$$U \cup W = \beta X \setminus [(\overline{W}_0 \cup \overline{L}_0) \cap (\overline{U}_0 \cup \overline{L}_0)] = \beta X \setminus [(\overline{W}_0 \cap \overline{U}_0) \cup \overline{L}_0]$$
$$= \beta X \setminus \overline{L}_0;$$

Ch. 2, § 2] Dimension Ind in normal and hereditarily normal spaces

on the other hand,

(3) 
$$U \cap W = \beta X \setminus [(\overline{W}_0 \cup \overline{L}_0) \cup (\overline{U}_0 \cup \overline{L}_0)] = \beta X \setminus (\overline{U}_0 \cup \overline{W}_0 \cup \overline{L}_0)$$
$$= \beta X \setminus \overline{X} = \emptyset.$$

Since the sets  $A_0 = X \cap \overline{V_1}$  and  $W_0 \cup L_0$  are closed in X and disjoint, we have  $\overline{V_1} \cap (\beta X \setminus U) = \overline{X \cap \overline{V_1}} \cap (\overline{W_0 \cup L_0}) = \emptyset$ , i.e.,  $\overline{V_1} \subset U$ ; similarly,  $\overline{V_2} \subset W$ . Thus  $A \subset U$  and  $B \subset W$ , which together with (2) and (3) shows that  $\overline{L_0}$  is a partition in  $\beta X$  between A and B. As  $\overline{L_0} = \beta L_0$ , by the inductive assumption  $\operatorname{Ind} \overline{L_0} \leq \operatorname{Ind} L_0 \leq n-1$ , so that  $\operatorname{Ind} \beta X \leq n = \operatorname{Ind} X$ .  $\Box$ 

**2.2.10. Corollary.** For every normal space X and a dense normal subspace  $M \subset X$  which has the property that every continuous function  $f: M \to I$  is continuously extendable over X we have  $\operatorname{Ind} M = \operatorname{Ind} X$ .

In other words,  $\operatorname{Ind} Y = \operatorname{Ind} X$  for every normal space X and every normal subspace Y of  $\beta X$  which contains X.

**Proof.** From the extendability of every continuous function  $f: M \to I$  over X it follows that  $\beta M = \beta X$ .  $\Box$ 

Let us observe that the counterpart of Theorem 2.2.9 for the dimension ind does not hold (see Problem 2.2.E).

As the Čech-Stone compactification generally raises the weight of spaces, it is natural to ask if there exist compactifications preserving both the dimension Ind and weight. One proves that for every normal space X there exists a compactification  $\tilde{X}$  such that  $\operatorname{Ind} \tilde{X} \leq \operatorname{Ind} X$  and  $w(\tilde{X}) = w(X)$ , where w denotes the weight of a space, i.e., the infimum of the cardinalities of the bases for that space. The construction of such a compactification is rather difficult and will not be given here. Let us note that even more is true, viz., that for every integer  $n \geq 0$  and every cardinal number  $\mathfrak{m} \geq \aleph_0$  there exists a compact space  $B_{\mathfrak{m}}^n$  such that  $\operatorname{Ind} B_{\mathfrak{m}}^n = n$ ,  $w(B_{\mathfrak{m}}^n) = \mathfrak{m}$ , and every normal space X which satisfies the conditions  $\operatorname{Ind} X \leq n$  and  $w(X) \leq \mathfrak{m}$  is homeomorphic to a subspace of the space  $B_{\mathfrak{m}}^n$ . In other words, there exists a compact universal space for the class of all normal spaces whose large inductive dimension is not larger than n and whose weight is not larger than  $\mathfrak{m}$ .

We conclude this section with the counter-examples announced above. The first is a compact space Z with IndZ = 0 which contains a normal subspace X such that IndX > 0. **2.2.11.** Dowker's example. Let Q denote the subspace of the interval I consisting of all rational numbers in I (clearly, Q is homeomorphic to the space of rational numbers). By letting

*xEy* if and only if 
$$|x-y| \in Q$$

we define an equivalence relation E on the set I. Since each equivalence class of E is countable, the family of all equivalence classes has cardinality c. Let us choose from it a subfamily of cardinality  $\aleph_1$  which does not contain the equivalence class Q and let us arrange the members of this subfamily into a transfinite sequence  $Q_0, Q_1, \dots, Q_{\alpha}, \dots, \alpha < \omega_1$ .

For every  $\gamma < \omega_1$  the set  $P_{\gamma} = I \setminus \bigcup_{\alpha \ge \gamma} Q_{\alpha}$  satisfies the equality ind  $P_{\gamma} = 0$ ; indeed  $Q \subset P_{\gamma}$ , and since the sets  $Q_{\alpha}$  are dense in *I*, the set  $P_{\gamma}$  contains no interval. Let *W* be the space of all ordinal numbers  $\leq \omega_1$  and let  $W_0$  $= W \setminus \{\omega_1\}$  (see Example 2.1.6). For every  $\alpha < \omega_1$  the subspace  $M_{\alpha}$  $= [0, \alpha] = [0, \alpha+1)$  is open-and-closed in *W*. Consider the Cartesian product  $W \times I$  and its subspaces

$$X_{\alpha} = \bigcup_{\gamma \leq \alpha} (\{\gamma\} \times P_{\gamma}), \quad X = \bigcup_{\alpha < \omega_1} X_{\alpha} \quad \text{and} \quad X^* = X \cup (\{\omega_1\} \times I).$$

As noted in Example 2.1.6, the subspaces  $M_{\alpha}$  are metrizable. Being countable, they are zero-dimensional, so that  $\operatorname{ind}(M_{\alpha} \times P_{\alpha}) = 0$  by virtue of Theorem 1.3.6. Since for every  $\alpha < \omega_1$  the set  $X_{\alpha}$  is open-and-closed in X and  $\operatorname{ind} X_{\alpha} = 0$  in view of the inclusion  $X_{\alpha} \subset M_{\alpha} \times P_{\alpha}$ , we have  $\operatorname{ind} X = 0$ . From Remark 1.3.18 and Theorem 1.6.5 it follows that there exists a compact space Z with  $\operatorname{Ind} Z = 0$  which contains X as a subspace.

We shall show now that the space  $X^*$  is normal which is the first step towards establishing normality of the space X. As  $X^*$  is a subspace of  $W \times I$ , it is a  $T_1$ -space. Consider a pair A, B of disjoint closed subsets of  $X^*$ . The sets

$$F_1 = \{x \in I: (\omega_1, x) \in A\}$$
 and  $F_2 = \{x \in I: (\omega_1, x) \in B\}$ 

are disjoint and closed in I so that there exist open sets  $H_1, H_2 \subset I$  such that  $F_1 \subset H_1, F_2 \subset H_2$  and  $H_1 \cap H_2 = \emptyset$ . For every  $x \in I \setminus H_1$  there exists a neighbourhood  $U_x \subset I$  of the point x and an ordinal number  $\alpha(x) < \omega_1$  such that  $[(W \setminus M_{\alpha(x)}) \times U_x] \cap A = \emptyset$ . The set  $I \setminus H_1$  being closed in I, we can choose a finite number of points  $x_1, x_2, \ldots, x_k \in I \setminus H_1$  such that  $I \setminus H_1 \subset \bigcup_{i=1}^k U_{x_i}$ . For  $\alpha_1 = \max(\alpha(x_1), \alpha(x_2), \ldots, \alpha(x_k)) < \omega_1$  we have  $[(W \setminus M_{\alpha_1}) \times (I \setminus H_1)] \cap A = \emptyset$ , i.e., (4)  $A \cap (X^* \setminus X_{\alpha_1}) \subset (W \setminus M_{\alpha_1}) \times H_1$ .
In a similar way we determine an  $\alpha_2 < \omega_1$  such that

(5)  $B \cap (X^* \setminus X_{\alpha_2}) \subset (W \setminus M_{\alpha_2}) \times H_2;$ 

without loss of generality we can assume that  $\alpha_1 \leq \alpha_2$ . The sets  $A \cap X_{\alpha_2}$ and  $B \cap X_{\alpha_2}$  are disjoint and closed in  $X_{\alpha_2}$ , so that there exist open sets  $U_1, V_1 \subset X_{\alpha_2}$  such that

(6) 
$$A \cap X_{a_1} \subset U_1, \quad B \cap X_{a_2} \subset V_1 \quad \text{and} \quad U_1 \cap V_1 = \emptyset;$$

since  $X_{\alpha_2}$  is an open subset of  $X^*$ , the sets  $U_1$  and  $V_1$  are open in  $X^*$ . The sets  $U = U_1 \cup \{[(W \setminus M_{\alpha_2}) \times H_1] \cap X^*\}$  and  $V = V_1 \cup \{[(W \setminus M_{\alpha_2}) \times H_2] \cap \cap X^*\}$  are open in  $X^*$ . By virtue of (4)-(6)

 $A \subset U$ ,  $B \subset V$  and  $U \cap V = \emptyset$ ,

so that the space  $X^*$  is normal.

To prove that the space X is normal, it suffices to show that, for every pair A, B of disjoint closed subsets of X, the closures  $\overline{A}$  and  $\overline{B}$  of A and B in the space X\* are disjoint. Suppose that there exists a point  $(\omega_1, x) \in \overline{A} \cap \overline{B}$ . It follows from the definition of  $P_{\gamma}$  that there exists an  $\alpha_0 < \omega_1$  such that  $x \in P_{\alpha}$  for every  $\alpha \ge \alpha_0$ . We can readily define by induction two sequences  $\alpha_1, \alpha_2, \ldots$  and  $\beta_1, \beta_2, \ldots$  of countable ordinal numbers and two sequences  $x_1, x_2, \ldots$  and  $y_1, y_2, \ldots$  of real numbers in I satisfying

$$\begin{aligned} \alpha_0 < \alpha_i < \beta_i < \alpha_{i+1}, & |x-x_i| < 1/i, & |x-y_i| < 1/i, \\ (\alpha_i, x_i) \in A, & (\beta_i, y_i) \in B \end{aligned}$$

for i = 1, 2, ... Now, for the smallest ordinal number  $\gamma$  larger than all  $\alpha_i$ 's we have  $(\gamma, x) \in A \cap B$ , which is impossible. Thus  $\overline{A} \cap \overline{B} = \emptyset$  and X is a normal space.

It remains to show that  $\operatorname{Ind} X > 0$ . Assume that  $\operatorname{Ind} X = 0$  and consider the pair  $W_0 \times \{0\}$ ,  $W_0 \times \{1\}$  of disjoint closed subsets of the space X. Then there exists an open-and-closed set  $U \subset X$  such that  $W_0 \times \{0\} \subset U$  and  $W_0 \times \{1\} \subset X \setminus U$ . As established in the preceding paragraph, the closures  $\overline{A}$  and  $\overline{B}$  of the sets A = U and  $B = X \setminus U$  in the space  $X^*$  are disjoint. Since  $X = A \cup B$  and  $\overline{X} = X^*$ , we have  $\overline{A} \cup \overline{B} = X^*$ . The sets  $A_1 = \{x \in I: (\omega_1, x) \in \overline{A}\}$  and  $B_1 = \{x \in I: (\omega_1, x) \in \overline{B}\}$  are disjoint and closed in I; moreover,  $A_1 \cup B_1 = I$  and  $A_1 \neq \emptyset \neq B_1$ , because  $0 \in A_1$  and  $1 \in B_1$ . Thus our assumption contradicts the connectedness of the interval I. Hence  $\operatorname{Ind} X > 0$ .  $\Box$ 

We now turn to the counterexample to the sum theorem for Ind in normal spaces. To begin with, we shall describe two auxiliary spaces L and  $L_0$ , which are known as the long segment and the long line.

[Ch. 2, § 2

**2.2.12. Example.** Let  $W_0$  be the set of all countable ordinal numbers and let  $L_0 = W_0 \times [0, 1)$ . Define a linear order < in the set  $L_0$  by letting  $(\alpha_1, t_1) < (\alpha_2, t_2)$  whenever  $\alpha_1 < \alpha_2$  or  $\alpha_1 = \alpha_2$  and  $t_1 < t_2$ . Furthermore, let  $L = L_0 \cup \{\omega_1\}$  and extend the linear order < over L by letting  $x < \omega_1$  for every  $x \in L_0$ . By assigning to each  $\alpha \in W_0$  the point  $(\alpha, 0) \in L_0$  we define a one-to-one mapping of  $W_0$  onto the set  $W_0 \times \{0\} \subset L_0$ ; in the sequel we shall identify  $\alpha$  with  $(\alpha, 0)$  and we shall consider  $W_0$  as a subset of  $L_0$ . The set W of all ordinal numbers  $\leq \omega_1$  is a subset of L.

Consider on L the topology obtained by taking as a base all sets of the form

$$(x_0, \omega_1] = \{x: x_0 < x\}, \quad [0, x_1) = \{x: x < x_1\}$$
  
and  $(x_0, x_1) = \{x: x_0 < x < x_1\},$ 

where  $x_0, x_1 \in L$  and  $x_0 < x_1$ . One easily sees that L is a connected compact space and that the subspace topology on  $W \subset L$  coincides with the topology on W defined in Example 2.1.6. For every  $x_0 \in L_0$  the closed subspace  $[0, x_0] = \{x: x \leq x_0\}$  of L has a countable base, viz., the family of all sets  $((\alpha_0, t_0), (\alpha_1, t_1))$ , where  $\alpha_0, \alpha_1 \leq x_0$  and  $t_0, t_1$  are rational numbers in I, and thus is a compact metric space. The space L is called the *long* segment; the subspace  $L_0$  of L is called the *long line*.  $\Box$ 

We shall now describe a compact space X with  $\operatorname{Ind} X \ge 2$  which contain closed subspaces  $F_1, F_2$  such that  $F_1 \cup F_2 = X$  and  $\operatorname{Ind} F_1 = \operatorname{Ind} F_2 = 1$ .

2.2.13. Lokucievskii's example. Let L be the long segment and C the Cantor set. The subspace of the Cantor set consisting of the end-points of all intervals removed from I to obtain the Cantor set will be denoted by Q (cf. Problem 1.3.H(e)). Since L and C are compact, the Cartesian product  $L \times C$  is a compact space. We shall now establish a property of open subsets of  $L \times C$  which will prove crucial for the evaluation of IndX. Namely

- (7) for every open set  $U \subset L \times C$  such that  $U \cap (\{\omega_1\} \times C) \neq \emptyset$  and  $t_v = \sup\{t: (\omega_1, t) \in U\}$  belongs to  $(C \setminus Q) \setminus \{1\}$  either
  - (i) there exists an  $x' \in L_0$  such that  $(x', \omega_1] \times \{t_v\} \subset \operatorname{Fr} U$
  - or
  - (ii) there exist a  $t' \in (t_v, 1]$  and a set  $L' \subset L_0$  cofinal in  $L_0$  such that  $L' \times ([t_v, t'] \cap C) \subset U$ .

Consider a sequence  $t_1, t_2, ...$  converging to  $t_v$  and such that  $(\omega_1, t_i) \in U$  for i = 1, 2, ... The set U being open, there exists a sequence  $\alpha_1, \alpha_2, ...$  of countable ordinal numbers such that  $(\alpha_i, \omega_1] \times t_i \subset U$  for i = 1, 2, ...

For the smallest ordinal number  $\alpha_0 \in W_0 \subset L_0$  larger than all  $\alpha_i$ 's we have  $(\alpha_0, \omega_1] \times \{t_v\} \subset \overline{U}$ . If (i) does not hold, then there exists a set  $L_1 \subset L_0$  cofinal in  $L_0$  and such that  $L_1 \times \{t_v\} \subset U$ . For every  $x \in L_1$  we can find a  $t_x \in (t_v, 1]$  satisfying  $\{x\} \times ([t_v, t_x] \cap C) \subset U$ . Clearly, there exists a natural number k such that the set  $L(k) = \{x \in L_1: t_x - t_v > 1/k\}$  is cofinal in  $L_0$ , and thus (ii) holds with  $t' = t_v + 1/k$  and L' = L(k).

Let  $f: C \to I$  be the continuous mapping of C onto I consisting in matching the end-points of each interval removed from I to obtain the Cantor set (see Problem 1.3.D) and let E be the equivalence relation on the space  $L \times C$  corresponding to the decomposition of  $L \times C$  into one-point subsets of  $L_0 \times C$  and the sets  $\{\omega_1\} \times f^{-1}(t)$ , where  $t \in I$ . Since the equivalence relation E is closed, the quotient space  $Y = (L \times C)/E$  is compact (see [GT], Theorem 3.2.11). The image of the set  $\{\omega_1\} \times C \subset L \times C$  under the natural quotient mapping will be denoted by I and will be identified with the interval [0, 1]; the image of  $\{\omega_1\} \times Q$  will be denoted by K.

We shall prove that  $\operatorname{ind} Y = \operatorname{Ind} Y = 1$ . To begin with, let us observe that  $\operatorname{ind} Y \leq 1$ . Indeed, every neighbourhood of a point  $x \in Y \setminus I$  contains a neighbourhood of this point with a boundary homeomorphic to  $C \oplus C$ and every neighbourhood of a point  $x \in I$  contains a neighbourhood of this point with a boundary homeomorphic to  $C \oplus B$ , where B is a discrete space of cardinality  $\leq 2$ ; the last fact is a consequence of the density of K in I. Now we shall show that  $\operatorname{Ind} Y \leq 1$ . Consider a closed set  $A \subset Y$ and an open set  $V \subset Y$  which contains the set A. For every  $x \in A$  there exists a neighbourhood  $U_x$  such that  $U_x \subset V$  and  $\operatorname{Fr} U_x$  is a zero-dimensional compact metric space. The subspace A of Y being compact, we can choose a finite number of points  $x_1, x_2, \ldots, x_k$  such that  $A \subset U = \bigcup_{i=1}^k U_{x_i} \subset V$ . The subspace  $F = \bigcup_{i=1}^k \operatorname{Fr} U_{x_i}$  of Y is normal, so that by virtue of Theorems 1.6.4 and 2.2.7 we have  $\operatorname{Ind} F = 0$ . Since  $\operatorname{Fr} U \subset F$ , it follows from Theorem 2.2.1 that  $\operatorname{Ind} \operatorname{Fr} U \leq 0$ . Hence  $\operatorname{Ind} Y \leq 1$ ; as the space Y contains the

interval I, we have  $\operatorname{ind} Y = \operatorname{Ind} Y = 1$ .

Let us note that (7) yields

(8) for every open set  $U \subset Y$  such that  $U \cap I \neq \emptyset$  and  $\sup\{t: t \in U \cap I\}$  belongs to  $(I \setminus K) \setminus \{1\}$  we have ind  $\operatorname{Fr} U \ge 1$ .

Consider two disjoint copies  $Y_1$  and  $Y_2$  of the space Y, and denote by  $I_i$  and  $K_i$  the counterparts of I and K in  $Y_i$  for i = 1, 2. Let  $K_3 \subset I_2 \setminus K_2$ be a countable set dense in  $I_2$  which does not contain the end-points of  $I_2$ . It follows from Problem 1.3.G(a) that there exists a homeomorphism  $h: I_1 \to I_2$  such that  $h(K_1) = K_3 \subset I_2 \setminus K_2$ . Let S be the equivalence relation on the space  $Y_1 \oplus Y_2$  corresponding to the decomposition of  $Y_1 \oplus Y_2$  into one-point subsets of  $(Y_1 \setminus I_1) \cup (Y_2 \setminus I_2)$  and the sets  $\{x, h(x)\}$ , where  $x \in I_1$ . Since the equivalence relation S is closed, the quotient space  $X = (Y_1 \oplus Y_2)/S$  is compact. Roughly speaking, the space X is obtained by matching  $Y_1$  with  $Y_2$  along  $I_1$  and  $I_2$  in such a way that no points of  $K_1$  and  $K_2$  are matched to each other. For i = 1, 2 the image  $F_i$  of the set  $Y_i$  under the natural quotient mapping  $q: Y_1 \oplus Y_2 \to X$  is closed in X and homeomorphic to Y, so that  $\operatorname{Ind} F_1 = \operatorname{Ind} F_2 = 1$ ; moreover  $F_1 \cup F_2 = X$ .

It remains to show that  $\operatorname{Ind} X \ge 2$ . Consider the mid-point x of the interval obtained by matching  $I_1$  with  $I_2$  and a neighbourhood  $V \subset X$  of the point x which does not contain the end-points of this interval. From (8) it follows that for every open set  $U \subset X$  such that  $x \in U \subset V$  we have ind  $\operatorname{Fr} U \ge 1$ , because  $\sup\{t: t \in q^{-1}(U) \cap I_i\}$  belongs to  $I_i \setminus K_i$  either for i = 1 or for i = 2. Thus  $\operatorname{Ind} X \ge \operatorname{ind} X \ge 2$ ; one can show that  $\operatorname{Ind} X$  =  $\operatorname{ind} X = 2$  (see Problem 2.2.C(c)).  $\Box$ 

### Historical and bibliographic notes

Theorem 2.2.1 was noted by Čech in [1932]. An example of a hereditarily normal space X with Ind X = dim X = 0 which for every natural number n contains a subspace  $A_n$  with  $Ind A_n = dim A_n = n$  was described by E. Pol and R. Pol in [1979] (in [1977] the same authors gave an example to show that Ind and dim are not monotonic in hereditarily normal spaces); under additional set theoretic assumptions such an example was defined by Filippov in [1973]. Theorem 2.2.3 was proved by Dowker in [1953]. Theorem 2.2.5 was given by Smirnov in [1951]. Theorem 2.2.7 is implicit in Čech's paper [1933] (cf. Theorems 1.6.11 and 3.1.8); it was first formulated by Vedenissoff in [1939]. An example of compact spaces Xand Y such that indX = IndX = 1, indY = IndY = 2 and  $Ind(X \times Y)$  $\geq$  ind $(X \times Y) \geq 4$  was described by Filippov in [1972] and an example of a normal space Z, whose square  $Z \times Z$  is also normal, such that IndZ= 0 and yet  $Ind(Z \times Z) > 0$  was given by Wage in [1977]. In his original construction Wage applied the continuum hypothesis; Przymusiński in [1977] noted that the continuum hypothesis can be avoided by a modification of Wage's construction. Theorem 2.2.9 was established by

#### Problems

Vedenissoff in [1939]. The existence of compactifications preserving both the dimension Ind and the weight of normal spaces and the corresponding universal space theorem were announced by Pasynkov in [1971]; proofs were given in Alexandroff and Pasynkov's book [1973]. Example 2.2.11 was described by Dowker in [1955], and Example 2.2.13—by Lokucievskiĭ in [1949].

## Problems

**2.2.A** (Urysohn [1925]). (a) Prove that a subspace M of a hereditarily normal space X satisfies the inequality  $\operatorname{ind} M \leq n \geq 0$  if and only if for every point  $x \in M$  and each neighbourhood V of the point x in the space X there exists an open set  $U \subset X$  such that  $x \in U \subset V$  and  $\operatorname{ind}(M \cap \operatorname{Fr} U) \leq n-1$ .

(b) Show that for every pair X, Y of subspaces of a hereditarily normal space we have

$$\operatorname{ind}(X \cup Y) \leq \operatorname{ind} X + \operatorname{ind} Y + 1$$
.

**2.2.B** (Aarts and Nishiura [1971]). Prove that for every continuous mapping  $f: A \to S^k$  defined on a closed subspace A of a hereditarily normal space X such that  $\operatorname{Ind}(X \setminus A) \leq n$ , where  $0 \leq k \leq n$ , there exists a closed subspace B of the space X such that  $A \cap B = \emptyset$ ,  $\operatorname{Ind} B \leq n-k-1$ , and the mapping f has a continuous extension  $F: X \setminus B \to S^k$  over  $X \setminus B$ .

Hint. See Problem 1.9.D.

**2.2.C.** (a) Show that if a normal space X can be represented as the union of two subspaces M and F such that M is normal and non-empty and F is closed in X, then for every pair A, B of disjoint closed subsets of X there exist a partition L in X between A and B and a partition L' in M between  $M \cap A$  and  $M \cap B$  such that  $L \setminus F = L' \setminus F$  and  $\operatorname{Ind} L' \leq \operatorname{Ind} M - 1$ .

(b) Deduce from (a) that if a normal space X can be represented as the union of two subspaces  $X_1$  and F such that  $X_1$  is normal and non-empty, F is closed, and  $\operatorname{Ind} F = 0$ , then  $\operatorname{Ind} X \leq \operatorname{Ind} X_1$ .

(c) Show that if a normal space can be represented as the union of two closed subspaces  $F_1$  and  $F_2$ , then  $\operatorname{Ind} X \leq \operatorname{Ind} F_1 + \operatorname{Ind} F_2$ .

(d) Show that if a normal space can be represented as the union of two normal subspaces X and Y of which one is either closed or open, then  $Ind(X \cup Y) \leq IndX + IndY + 1$ .

(e) Show that if a hereditarily normal space X contains a closed subspace F such that  $\operatorname{Ind} F \leq n$  and  $\operatorname{Ind}(X \setminus F) \leq n$ , then  $\operatorname{Ind} X \leq n$  (cf. Theorem 2.3.1).

**2.2.D.** Check that if X is a metric space and a dense subspace  $M \subset X$  has the property that every continuous function  $f: M \to I$  is continuously extendable over X, then M = X.

**2.2.E.** Note that for the space X described in Example 2.2.11 we have  $nd\beta X \neq indX$ .

**2.2.F** (Dowker [1955]). (a) Prove that the space X described in Example 2.2.11 satisfies the equality IndX = 1.

(b) Show that to the space X described in Example 2.2.11 one point can be adjoined either in such a way that one obtains a normal space  $X_1$  with  $\operatorname{ind} X_1 > 0$  or in such a way that one obtains a normal space  $X_2$  with  $\operatorname{Ind} X_2 = 0$ .

**2.2.G** (Smirnov [1958]). Modify the construction of the space X in Example 2.2.11 to obtain a compact space Z with Ind Z = 0 which contains a normal subspace Y such that  $\text{Ind} Y = \infty$ .

*Hint*. Replace the interval by the Hilbert cube and the sets  $P_{\gamma}$  by the Cartesian products  $P_{\gamma}^{\aleph_0}$ .

**2.2.H.** (a) Show that the long segment L is a strongly hereditarily normal space.

(b) Prove that for every point  $x_0$  in the long line  $L_0$  the subspace  $[0, x_0]$  of L is homeomorphic to the interval I.

*Hint*. Define a countable dense subset of  $[0, x_0]$  which is ordered similarly to the set of all rational numbers in *I*.

(c) Check that for the subspace W of the long segment L there exists no  $G_{\delta}$ -set  $W^*$  in L such that  $W \subset W^*$  and  $\operatorname{Ind} W^* = \operatorname{Ind} W$ .

# 2.3. Basic properties of the dimension Ind in strongly hereditarily normal spaces

Strongly hereditarily normal spaces constitute a relatively wide class of spaces where two of the most important theorems of dimension theory, viz., the subspace theorem and the countable sum theorem, hold for the dimension Ind. Both theorems are tightly connected; we shall prove them simultaneously. For more clarity the proof is divided into several lemmas.

Let us consider the following properties of a hereditarily normal space X:

- $(\mu_n)$  For each subspace  $Y \subset X$  and every open subspace U of Y, if  $\operatorname{Ind} Y \leq n$ , then  $\operatorname{Ind} U \leq n$ .
- $(\sigma_n)$  For each subspace  $Y \subset X$  and every sequence  $F_1, F_2, \ldots$  of closed subspaces of Y such that  $Y = \bigcup_{i=1}^{\infty} F_i$ , if  $\operatorname{Ind} F_i \leq n$  for  $i = 1, 2, \ldots$ , then  $\operatorname{Ind} Y \leq n$ .

Clearly, every hereditarily normal space X has property  $(\mu_{-1})$ , and thus to prove that all strongly hereditarily normal spaces have properties  $(\mu_n)$  and  $(\sigma_n)$  for n = -1, 0, 1, ..., it suffices to show that the implications  $(\mu_{n-1}) \Rightarrow (\mu_n)$  and  $(\mu_n) \Rightarrow (\sigma_n)$  hold for every strongly hereditarily normal space X. The second implication holds for all hereditarily normal spaces; it will be deduced from the following version of the sum theorem.

**2.3.1. Theorem.** If a hereditarily normal space X can be represented as the union of a sequence  $K_1, K_2, ...$  of pairwise disjoint subspaces such that  $\operatorname{Ind} K_i \leq n$  and the union  $\bigcup_{j \leq i} K_j$  is closed for i = 1, 2, ..., then  $\operatorname{Ind} X \leq n$ .

**Proof.** We shall apply induction with respect to the number *n*. For n = -1 the theorem is obvious. Assume that the corresponding statements hold for dimensions less than *n* and consider a hereditarily normal space X which satisfies the assumptions of our theorem. Let  $F_i = \bigcup_{j \le i} K_j$  for i = 1, 2, ...

Consider a pair A, B of disjoint closed subsets of X. Let  $U_0$ ,  $V_0$  be open subsets of X such that

(1)  $A \subset U_0, \quad B \subset V_0 \quad \text{and} \quad \overline{U}_0 \cap \overline{V}_0 = \emptyset,$ 

and let  $K_0 = F_0 = L_0 = \emptyset$ . We shall define inductively two sequences  $U_0, U_1, \ldots$  and  $V_0, V_1, \ldots$  of open subsets of X and a sequence  $L_0, L_1, \ldots$  of subsets of X satisfying for  $i = 0, 1, \ldots$  the following conditions:

(2) 
$$L_i \subset K_i$$
 and  $\operatorname{Ind} L_i \leq n-1$ .

(3) The set 
$$E_i = \bigcup_{j \le i} L_j$$
 is closed.

The large inductive dimension

(4) 
$$F_i \subset U_i \cup V_i \cup E_i.$$

(5) 
$$U_i \cup V_i \subset X \setminus E_i \text{ and } \overline{U_i} \cap \overline{V_i} \subset E_i.$$

(6) 
$$U_{i-1} \subset U_i$$
 and  $V_{i-1} \subset V_i$  for  $i > 0$ .

The sets  $U_0$ ,  $V_0$  and  $L_0$  satisfying (2)-(6) for i = 0 are already defined. Let us assume that the sets  $U_i$ ,  $V_i$  and  $L_i$  satisfying (2)-(6) are defined for i < m > 0. The sets  $\overline{U}_{m-1} \cap K_m$  and  $\overline{V}_{m-1} \cap K_m$  are closed in  $K_m$  and disjoint, because by virtue of (5) and (2) with i = m-1

$$\overline{U}_{m-1} \cap \overline{V}_{m-1} \subset E_{m-1} \subset \bigcup_{j < m} K_j$$

and the last set is disjoint from  $K_m$ . Since  $\operatorname{Ind} K_m \leq n$ , there exists a partition  $L_m$  in  $K_m$  between  $\overline{U}_{m-1} \cap K_m$  and  $\overline{V}_{m-1} \cap K_m$  which satisfies (2) with i = m; there also exist open subsets G, H of  $K_m$  such that

(7) 
$$\overline{U}_{m-1} \cap K_m \subset G, \quad \overline{V}_{m-1} \cap K_m \subset H, \quad G \cap H = \emptyset$$
  
and  $K_m \setminus L_m = G \cup H.$ 

From (7) it follows that  $L_m \cap (U_{m-1} \cup V_{m-1}) = \emptyset$ ; the union  $U_{m-1} \cup V_{m-1}$ being open in X, we have  $\overline{L}_m \cap (U_{m-1} \cup V_{m-1}) = \emptyset$ . Since  $L_m$  is closed in  $K_m$  and  $F_m$  is closed in X, the last equality and (4) with i = m-1 yield the inclusion

$$\overline{L}_m \subset L_m \cup [F_{m-1} \setminus (U_{m-1} \cup V_{m-1})] \subset E_m,$$

which shows that (3) holds for i = m.

Since  $G \cap V_{m-1} = \emptyset = H \cap U_{m-1}$ , we have  $\overline{G} \cap V_{m-1} = \emptyset = \overline{H} \cap U_{m-1}$ , which implies that

(8) 
$$\overline{G} \cap \overline{H} \subset L_m \cup [F_{m-1} \setminus (U_{m-1} \cup V_{m-1})] \subset E_m,$$

because  $\overline{G} \cap \overline{H} \subset \overline{K}_m \subset \overline{F}_m = F_m$ . The same equalities  $\overline{G} \cap V_{m-1} = \emptyset$ =  $\overline{H} \cap U_{m-1}$  together with (7), the equality  $U_{m-1} \cap V_{m-1} = \emptyset$ , which is a consequence of (5) with i = m-1, and (4) with i = m-1 yield

(9) 
$$G \cap V_{m-1} = [G \cap (V_{m-1} \setminus V_{m-1})] \setminus K_m \subset F_{m-1} \cap (V_{m-1} \setminus V_{m-1}) \subset E_{m-1}$$
  
and

$$\overline{H} \cap \overline{U}_{m-1} = [\overline{H} \cap (\overline{U}_{m-1} \setminus U_{m-1})] \setminus K_m \subset F_{m-1} \cap (\overline{U}_{m-1} \setminus U_{m-1}) \subset E_{m-1}.$$
(10)

Relations (8)-(10) and the second part of (5) with i = m-1 show that the sets  $(\overline{U}_{m-1} \cup \overline{G}) \setminus E_m$  and  $(\overline{V}_{m-1} \cup \overline{H}) \setminus E_m$  are disjoint. As these sets are closed in  $X \setminus E_m$ , from the hereditary normality of X it follows that

184

[Ch. 2, § 3

Ch. 2, § 3] Dimension Ind in strongly hereditarily normal spaces

there exist disjoint sets  $U_m$ ,  $V_m$  open in  $X \ E_m$ , and consequently open in X, such that

$$\langle \overline{U}_{m-1} \cup \overline{G} \rangle \setminus E_m \subset U_m, \quad (\overline{V}_{m-1} \cup \overline{H}) \setminus E_m \subset V_m,$$

and (5) is satisfied for i = m. The last two inclusions imply that (4) and (6) also are satisfied for i = m, because  $L_m \cap (U_{m-1} \cup V_{m-1}) = \emptyset$  by virtue of (7). Hence the construction of the sets  $U_i$ ,  $V_i$  and  $L_i$  satisfying (2)-(6) for i = 1, 2, ... is completed.

Let  $U = \bigcup_{i=1}^{\infty} U_i$ ,  $V = \bigcup_{i=1}^{\infty} V_i$ , and  $L = \bigcup_{i=1}^{\infty} L_i$ ; it follows from (4) that  $X = U \cup V \cup L$ . The sets U and V are open and, by virtue of (6) and the equality  $U_i \cap V_i = \emptyset$ , which is a consequence of (5), disjoint. From (5), (6) and the inclusion  $E_{i-1} \subset E_i$  it follows that  $U \cap L = V \cap L = \emptyset$ , so that  $X \setminus L = U \cup V$ , which together with (1) shows that L is a partition between A and B. By virtue of (2) and (3) the inductive assumption can be applied to the space L and the sequence  $L_1, L_2, ...$ ; thus  $\operatorname{Ind} L \leq n-1$ , which shows that  $\operatorname{Ind} X \leq n$ .  $\Box$ 

**2.3.2. Corollary.** If a hereditarily normal space X has property  $(\mu_n)$ , then it also has property  $(\sigma_n)$ .

**Proof.** Consider a subspace  $Y \subset X$  and a sequence  $F_1, F_2, ...$  of closed subspaces of Y such that  $Y = \bigcup_{i=1}^{\infty} F_i$  and  $\operatorname{Ind} F_i \leq n$  for i = 1, 2, ... By virtue of  $(\mu_n)$  the set  $K_i = F_i \setminus \bigcup_{j < i} F_j$  satisfies the inequality  $\operatorname{Ind} K_i \leq n$ for i = 1, 2, ... Applying Theorem 2.3.1 to the space Y and the sequence  $K_1, K_2, ...,$  we obtain the inequality  $\operatorname{Ind} Y \leq n$ .  $\Box$ 

We now turn to the implication  $(\mu_{n-1}) \Rightarrow (\mu_n)$ . To begin with, we shall establish a simple property of point-finite open covers, which will be applied here and in the following chapter.

**2.3.3. Lemma.** Let  $\{U_s\}_{s\in S}$  be a point-finite open cover of a topological space X. For i = 1, 2, ... denote by  $K_i$  the set of all points of the space X which belong to exactly i members of the cover  $\{U_s\}_{s\in S}$  and by  $\mathcal{T}_i$  the family of all subsets of S that have exactly i elements. Then

(11) 
$$X = \bigcup_{i=1}^{\infty} K_i, \ K_i \cap K_j = \emptyset$$
 whenever  $i \neq j$ , and  
the union  $F_i = \bigcup_{j \leq i} K_j$  is closed for  $i = 1, 2, ...;$ 

moreover,

(12) 
$$K_i = \bigcup_{T \in \mathcal{T}_i} K_T$$
, where the sets  $K_T$ , defined by letting  $K_T = K_i \cap \bigcap_{s \in T} U_s$   
for  $T \in \mathcal{T}_i$ , are open in  $K_i$  and pairwise disjoint.

**Proof.** The first two equalities in (11) follow directly from the definition of the sets  $K_i$ . If  $x \notin F_i$ , then  $x \in U_{s_1} \cap U_{s_2} \cap \ldots \cap U_{s_{i+1}} \subset X \setminus F_i$ , where  $s_1, s_2, \ldots, s_{i+1}$  are distinct elements of S, so that the sets  $F_i$  are closed. To establish (12) it suffices to note that  $K_T \subset K_i$  for  $T \in \mathcal{T}_i$  and that whenever T, T' are distinct members of  $\mathcal{T}_i$ , then  $K_T \cap K_{T'} = \emptyset$ , because the union  $T \cup T'$  contains at least i+1 elements of S.  $\Box$ 

We shall now prove a lemma which, together with Corollary 2.3.2, yields the implication  $(\mu_{n-1}) \Rightarrow (\mu_n)$  for every perfectly normal space X; afterwards, we shall deduce from this lemma that the implication  $(\mu_{n-1}) \Rightarrow (\mu_n)$  holds for every strongly hereditarily normal space X.

**2.3.4. Lemma.** If a hereditarily normal space X has property  $(\sigma_{n-1})$  and if  $\operatorname{Ind} X \leq n$ , then  $\operatorname{Ind} U \leq n$  for every open subspace U of X which can be represented as the union of a point-finite family of open  $F_{\sigma}$ -sets in X.

**Proof.** Let us first consider the special case where U is an open  $F_{\sigma}$ -set in X. Then there exists a continuous function  $f: X \to I$  such that  $U = f^{-1}((0, 1])$  (see Problem 2.1.A). The sets  $B_i = f^{-1}([1/i, 1])$  are closed in X and satisfy

(13)  $B_i \subset \operatorname{Int} B_{i+1}$ ,  $\operatorname{Ind} B_i \leq n$  for i = 1, 2, ... and  $U = \bigcup_{i=1}^{\infty} B_i$ .

Consider an arbitrary set  $A \subset U$  which is closed in U and an open set  $V \subset U$  which contains the set A. For i = 1, 2, ... let

(14) 
$$A_i = A \cap (B_i \setminus \operatorname{Int} B_{i-1})$$
 and  $V_i = V \cap (\operatorname{Int} B_{i+1} \setminus B_{i-2})$ ,

where  $B_{-1} = B_0 = \emptyset$ ; clearly,  $A_i \subset V_i \subset B_{i+1}$ ,  $A_i$  is closed in  $B_{i+1}$ , and  $V_i$  is open in  $B_{i+1}$ . Since  $\operatorname{Ind} B_{i+1} \leq n$ , there exists an open subset  $U_i$  of  $B_{i+1}$  such that

(15) 
$$A_i \subset U_i \subset \overline{U}_i \subset V_i$$
 and  $\operatorname{Ind} \operatorname{Fr} U_i \leq n-1$ ,

where the closure and boundary operators refer to the space U. Indeed, as  $B_{i+1}$  is a closed subset of U and  $U_i \subset \text{Int} B_{i+1}$ , the closure and the boundary of  $U_i$  in  $B_{i+1}$  and in U coincide. From the inclusion  $\overline{U}_i \subset U \setminus B_{i-2}$  it follows that the family  $\{\overline{U}_i\}_{i=1}^{\infty}$  is locally finite in the space U, so that for the set  $U_0 = \bigcup_{i=1}^{\infty} U_i$  we have  $\overline{U}_0 = \bigcup_{i=1}^{\infty} \overline{U}_i$  and  $\operatorname{Fr} U_0 \subset \bigcup_{i=1}^{\infty} \operatorname{Fr} U_i$ . Applying (13)-(15) and  $(\sigma_{n-1})$ , we obtain

$$A = A \cap \bigcup_{i=1}^{\infty} B_i \subset \bigcup_{i=1}^{\infty} A_i \subset U_0 \subset \overline{U}_0 = \bigcup_{i=1}^{\infty} \overline{U}_i \subset V$$
  
and Ind Fr  $U_0 \leq n-1$ ,

which shows that  $\operatorname{Ind} U \leq n$ .

Let us now pass to the general situation and consider a point-finite family  $\{U_s\}_{s\in S}$  of open  $F_{\sigma}$ -sets in X such that  $U = \bigcup_{s\in S} U_s$ . Applying Lemma 2.3.3 to the cover  $\{U_s\}_{s\in S}$  of the space U, we obtain the sets  $K_i$  which satisfy (11) and (12) with X = U. Since for each  $T \in \mathcal{T}_i$  we have  $K_T = K_i \cap$  $\cap \bigcap_{s\in T} U_s = F_i \cap \bigcap_{s\in T} U_s$ , the sets  $K_T$  are  $F_{\sigma}$ -sets in X.

The subspace  $X_i = (X \setminus U) \cup F_i$  is closed in X, so that  $\operatorname{Ind} X_i \leq n$ . Since the set  $K_i = F_i \setminus F_{i-1} = X_i \cap (U \setminus F_{i-1})$  is open in  $X_i$ , the sets  $K_T$  are open  $F_\sigma$ -sets in  $X_i$ . Applying the already established special case of our theorem to the space  $X_i$  and the set  $K_T$ , we conclude that  $\operatorname{Ind} K_T \leq n$  for every  $T \in \mathcal{T}_i$ . Thus  $\operatorname{Ind} K_i \leq n$  for  $i = 1, 2, \ldots$  by virtue of (12) and Proposition 2.2.8. Theorem 2.3.1 and (11) imply that  $\operatorname{Ind} U \leq n$ .  $\Box$ 

**2.3.5. Lemma.** If a strongly hereditarily normal space X has property  $(\mu_{n-1})$ , then it also has property  $(\mu_n)$ .

**Proof.** Since strong hereditary normality and property  $(\mu_{n-1})$  are both hereditary properties, it suffices to show that if  $\operatorname{Ind} X \leq n$ , then  $\operatorname{Ind} U \leq n$ for every open subspace U of X. Consider a pair A, B of disjoint closed subsets of the space U. Let  $U' = X \setminus (\overline{A} \cap \overline{B})$ ,  $A' = U' \cap \overline{A}$  and  $B' = U' \cap \overline{B}$ . Obviously,  $U \subset U'$ ,  $A \subset A'$ ,  $B \subset B'$  and the sets A', B' are disjoint and closed in U'. It is enough to show that there exists a partition L' in U' between A' and B' such that  $\operatorname{Ind} L' \leq n-1$ , because then  $L = U \cap L'$ will be a partition in U between A and B satisfying, by virtue of  $(\mu_{n-1})$ , the inequality  $\operatorname{Ind} L \leq n-1$ . Since  $\overline{A'} = \overline{A}$  and  $\overline{B'} = \overline{B}$ , we have U'  $= X \setminus (\overline{A'} \cap \overline{B'})$ . Thus without loss of generality we can suppose that U  $= \overline{X} \setminus (\overline{A \cap B})$  and define the required partition in U.

As the sets A and B are separated in X, there exist disjoint open sets  $V, W \subset X$  such that  $A \subset V, B \subset W$  and V and W can be represented

[Ch. 2, § 3

as the union of a point-finite family of open  $F_{\sigma}$ -sets in X. From the equality  $\overline{V} \cap W = \emptyset = V \cap \overline{W}$  it follows that  $\overline{A} \cap W = \emptyset = V \cap \overline{B}$ , so that  $(\overline{A} \cap \overline{B}) \cap (V \cup W) = \emptyset$ , i.e.,  $V \cup W \subset U$ . The space U being normal there exists an open set  $O \subset U$  such that  $A \subset O \subset U \cap \overline{O} \subset V$ . The sets A and  $F = V \setminus O$  are disjoint and closed in V; since  $\operatorname{Ind} V \leq n$  by virtue of Corollary 2.3.2 and Lemma 2.3.4, there exists a partition L in V between A and F such that  $\operatorname{Ind} L \leq n-1$ . Thus there exist open sets G,  $H \subset X$  such that

 $A \subset G$ ,  $F \subset H$ ,  $G \cap H = \emptyset$  and  $V \setminus L = G \cup H$ .

The set  $H \cup (U \setminus \overline{O})$  is open in X, disjoint from G, and contains B, because  $B \subset W \subset U \setminus V \subset U \setminus \overline{O}$ ; moreover,

$$G \cup [H \cup (U \setminus \overline{O})] = (V \setminus L) \cup (U \setminus \overline{O}) = U \setminus L.$$

Thus L is a partition in U between A and B.  $\Box$ 

From Corollary 2.3.2 and Lemma 2.3.5 it follows that all strongly hereditarily normal spaces have properties  $(\mu_n)$  and  $(\sigma_n)$  for n = -1, 0, 1, ..., which, together with Theorems 2.2.3 and 2.3.1, yields the following two results.

**2.3.6.** The subspace theorem for Ind. For every subspace M of a strongly hereditarily normal space X we have  $\operatorname{Ind} M \leq \operatorname{Ind} X$ .  $\Box$ 

**2.3.7. Proposition.** If a strongly hereditarily normal space X can be represented as the union of a sequence  $K_1, K_2, ...$  of subspaces such that  $\operatorname{Ind} K_i \leq n$  and the union  $\bigcup_{j \leq i} K_j$  is closed for i = 1, 2, ..., then  $\operatorname{Ind} X \leq n$ .  $\Box$ 

The last proposition yields

**2.3.8. The countable sum theorem for** Ind. If a strongly hereditarily normal space X can be represented as the union of a sequence  $F_1, F_2, ...$  of closed subspaces such that  $\operatorname{Ind} F_i \leq n$  for i = 1, 2, ..., then  $\operatorname{Ind} X \leq n$ .  $\Box$ 

In the dimension theory of general spaces there occur also sum theorems of a different kind, where instead of countable covers one considers locally finite ones. Such theorems are not discussed in the classical dimension theory of separable metric spaces, because—as is easy to verify—every locally finite cover of a separable metric space is countable, and thus in such spaces the locally finite sum theorem is only a particular case of the countable sum theorem. Exactly as in the case of the countable sum theorem, the locally finite sum theorem for Ind in strongly hereditarily normal spaces shall be deduced from a version of this theorem which holds in all hereditarily normal spaces. Let us note that in countably paracompact strongly hereditarily normal spaces (in particular, in perfectly normal spaces and in hereditarily weakly paracompact hereditarily normal spaces; cf. [GT], Sections 5.2 and 5.3), the locally finite sum theorem is an easy consequence of Theorems 2.3.1, 2.3.6 and 2.3.8 (see Problem 2.3.B).

**2.3.9. Theorem.** If a hereditarily normal space X can be represented as the union of a transfinite sequence  $K_1, K_2, ..., K_{\alpha}, ..., \alpha < \xi$  of pairwise disjoint subspaces such that  $\operatorname{Ind} K_{\alpha} \leq n$  and the union  $\bigcup_{\beta < \alpha} K_{\beta}$  is closed for  $\alpha < \xi$ , and the family  $\{K_{\alpha}\}_{\alpha < \xi}$  is locally finite, then  $\operatorname{Ind} X \leq n$ .

**Proof.** We shall apply induction with respect to the number *n*. For n = -1 the theorem is obvious. Assume that the corresponding statements hold for dimensions less than *n* and consider a hereditarily normal space X which satisfies the assumptions of our theorem. Let  $F_{\alpha} = \bigcup_{\beta \leq \alpha} K_{\beta}$  for  $\alpha < \xi$ .

Consider a pair A, B of disjoint closed subsets of X. We shall define inductively three transfinite sequences  $U_1, U_2, ..., U_{\alpha}, ..., \alpha < \xi, V_1, V_2, ...$  $..., V_{\alpha}, ..., \alpha < \xi$ , and  $L_1, L_2, ..., L_{\alpha}, ..., \alpha < \xi$  of subsets of X satisfying for  $\alpha < \xi$  the following conditions:

(16) 
$$L_{\alpha} \subset K_{\alpha}$$
 and  $\operatorname{Ind} L_{\alpha} \leq n-1$ .

(17) The sets  $U_{\alpha}$  and  $V_{\alpha}$  are open in  $F_{\alpha}$  and  $U_{\alpha} \cup V_{\alpha} = F_{\alpha} \setminus E_{\alpha}$ ,

where  $E_{\alpha} = \bigcup_{\beta \leq \alpha} L_{\beta}$ .

(18) 
$$A \cap F_{\alpha} \subset U_{\alpha}, \quad B \cap F_{\alpha} \subset V_{\alpha} \text{ and } U_{\alpha} \cap V_{\alpha} = \emptyset.$$

(19) 
$$U_{\beta} = F_{\beta} \cap U_{\alpha}, \quad V_{\beta} = F_{\beta} \cap V_{\alpha} \quad \text{for} \quad \beta < \alpha.$$

Since  $F_1 = K_1$  and  $\operatorname{Ind} K_1 \leq n$ , there exist sets  $U_1$ ,  $V_1$  and  $L_1$  satisfying (16)-(19) for  $\alpha = 1$ . Let us assume that the sets  $U_{\alpha}$ ,  $V_{\alpha}$  and  $L_{\alpha}$  satisfying (16)-(19) are defined for  $\alpha < \gamma > 1$ . Let

$$F'_{\gamma} = \bigcup_{\alpha < \gamma} K_{\alpha}, \quad U'_{\gamma} = \bigcup_{\alpha < \gamma} U_{\alpha}, \quad V'_{\gamma} = \bigcup_{\alpha < \gamma} V_{\alpha} \quad \text{and} \quad E'_{\gamma} = \bigcup_{\alpha < \gamma} L_{\alpha};$$

note that if  $\gamma = \gamma_0 + 1$ , then the sets defined above are equal to  $F_{\gamma_0}$ ,  $U_{\gamma_0}$ ,  $V_{\gamma_0}$  and  $E_{\gamma_0}$ , respectively. From conditions (17), (18) and (19) with  $\alpha < \gamma$ 

it follows that

(20)  $A \cap F'_{\gamma} \subset U'_{\gamma}, \quad B \cap F'_{\gamma} \subset V'_{\gamma}, \quad U'_{\gamma} \cap V'_{\gamma} = \emptyset$ and  $F'_{\gamma} \setminus E'_{\gamma} = U'_{\gamma} \cup V'_{\gamma}.$ 

We shall show that the sets  $U'_{\gamma}$ ,  $V'_{\gamma} \subset F'_{\gamma}$  are open in  $F'_{\gamma}$ ; clearly, it suffices to consider the case where  $\gamma$  is a limit number.

Consider an arbitrary point  $x \in U'_{\gamma}$ . Since the family  $\{K_{\alpha}\}_{\alpha < \gamma}$  is locally finite, there exist a neighbourhood  $W \subset X$  of the point x and an ordinal number  $\alpha_0 < \gamma$  such that  $W \cap K_{\alpha} = \emptyset$  whenever  $\alpha_0 < \alpha < \gamma$ , i.e., such that  $W \cap F'_{\gamma} = W \cap F_{\alpha_0}$ . The set  $U_{\alpha_0}$  being open in  $F_{\alpha_0}$ , there exists an open set  $W' \subset X$  such that  $U_{\alpha_0} = F_{\alpha_0} \cap W'$ . From (19) it follows that  $F_{\alpha_0} \cap U'_{\gamma}$  $= U_{\alpha_0}$ , so that

$$x \in W \cap U'_{\gamma} = W \cap F'_{\gamma} \cap U'_{\gamma} = W \cap F_{\alpha_0} \cap U'_{\gamma} = W \cap U_{\alpha_0} \subset W \cap W',$$

i.e., the set  $W \cap W'$  is a neighbourhood of x in X. As

$$F'_{\gamma} \cap W \cap W' = W \cap F_{\alpha_0} \cap W' = W \cap U_{\alpha_0} \subset U_{\alpha_0} \subset U'_{\gamma},$$

the set  $U'_{\gamma}$  is open in  $F'_{\gamma}$ . From symmetry of our assumptions it follows that the set  $V'_{\gamma}$  is also open in  $F'_{\gamma}$ . By virtue of (20) the set  $E'_{\gamma}$  is closed in  $F'_{\gamma}$ , which implies that  $E'_{\gamma}$  is closed in X, and  $(A \cup B) \cap E'_{\gamma} = \emptyset$ . The sets  $A \cup U'_{\gamma}$  and  $B \cup V'_{\gamma}$  are disjoint and closed in  $X \setminus E'_{\gamma}$ . From the hereditary normality of X it follows that there exist disjoint sets G,  $H \subset X \setminus E'_{\gamma}$  open in  $X \setminus E'_{\gamma}$ , and consequently open in X, such that

(21) 
$$A \cup U'_{\gamma} \subset G, \quad B \cup V'_{\gamma} \subset H \text{ and } \overline{G} \cap \overline{H} \subset E'_{\gamma}.$$

By virtue of the last inclusion in (21) and by equality  $E'_{\gamma} \cap K_{\gamma} = \emptyset$ , the sets  $G \cap K_{\gamma}$  and  $H \cap K_{\gamma}$  have disjoint closures in  $K_{\gamma}$ . Hence there exists a partition  $L_{\gamma}$  in  $K_{\gamma}$  between  $G \cap K_{\gamma}$  and  $H \cap K_{\gamma}$  which satisfies (16) with  $\alpha = \gamma$ ; there also exist open sets  $G', H' \subset K_{\gamma}$  such that

(22) 
$$G \cap K_{\gamma} \subset G', \quad H \cap K_{\gamma} \subset H', \quad G' \cap H' = \emptyset$$
  
and  $K_{\gamma} \setminus L_{\gamma} = G' \cup H'.$ 

Since the set  $K_{\gamma} = F_{\gamma} \setminus F'_{\gamma}$  is open in  $F_{\gamma}$ , the sets G' and H' are open in  $F_{\gamma}$ . From (20), (21) and (22) it follows that the sets

$$U_{\gamma} = (G \cap F_{\gamma}) \cup G'$$
 and  $V_{\gamma} = (H \cap F_{\gamma}) \cup H'$ 

satisfy conditions (17)-(19) with  $\alpha = \gamma$ . Hence the construction of the sets  $U_{\alpha}$ ,  $V_{\alpha}$  and  $L_{\alpha}$  satisfying (16)-(19) for  $\alpha < \xi$  is completed.

It follows from (20) that the set  $L = E'_{\xi}$  is a partition in  $F'_{\xi} = X$  between A and B. Applying the inductive assumption to the space L and the se-

quence  $L_1, L_2, ..., L_{\alpha}, ..., \alpha < \xi$ , we obtain the inequality  $\operatorname{Ind} L \leq n-1$ , which shows that  $\operatorname{Ind} X \leq n$ .  $\Box$ 

Theorems 2.3.9 and 2.3.6 yield

**2.3.10. The locally finite sum theorem for** Ind. If a strongly hereditarily normal space X can be represented as the union of a locally finite family  $\{F_s\}_{s\in S}$  of closed subspaces such that  $\operatorname{Ind} F_s \leq n$  for  $s \in S$ , then  $\operatorname{Ind} X \leq n$ .  $\Box$ 

The following two theorems are common generalizations of the countable and the locally finite sum theorems. Let us recall that a family  $\{A_s\}_{s\in S}$ of subsets of a topological space X is *locally countable* if for every point  $x \in X$  there exists a neighbourhood U such that the set  $\{s \in S: U \cap A_s \neq \emptyset\}$ is countable.

**2.3.11. Theorem.** If a strongly hereditarily normal space X can be represented as the union of a  $\sigma$ -locally finite family  $\{F_s\}_{s\in S}$  of closed subspaces such that  $\operatorname{Ind} F_s \leq n$  for  $s \in S$ , then  $\operatorname{Ind} X \leq n$ .

**Proof.** The family  $\{F_s\}_{s\in S}$  decomposes into countably many locally finite families the union of each of which is closed in X and—by virtue of the locally finite sum theorem—has dimension not larger than n. To complete the proof it suffices to apply the countable sum theorem.  $\square$ 

**2.3.12. Theorem.** If a strongly hereditarily normal space X can be represented as the union of a transfinite sequence  $F_1, F_2, \ldots, F_{\alpha}, \ldots, \alpha < \xi$  of closed subspaces such that  $\operatorname{Ind} F_{\alpha} \leq n$  and the family  $\{F_{\beta}\}_{\beta < \alpha}$  is locally finite for  $\alpha < \xi$ , and the family  $\{F_{\alpha}\}_{\alpha < \xi}$  is locally countable, then  $\operatorname{Ind} X \leq n$ .

**Proof.** If the set of all ordinal numbers less than  $\xi$  contains a countable cofinal subset, then the family  $\{F_{\alpha}\}_{\alpha < \xi}$  is  $\sigma$ -locally-finite and the theorem follows from Theorem 2.3.11.

Assume now that the set of all ordinal numbers less than  $\xi$  contains no countable cofinal subset. To complete the proof it suffices to show that under this assumption the family  $\{F_{\alpha}\}_{\alpha<\xi}$  is locally finite. Consider an arbitrary point  $x \in X$  and a neighbourhood U of this point such that the set  $\{\alpha < \xi : U \cap F_{\alpha} \neq \emptyset\}$  is countable. By our assumption there exists an  $\alpha_0 < \xi$  such that  $U \cap F_{\alpha} = \emptyset$  for  $\alpha \ge \alpha_0$ . The family  $\{F_{\alpha}\}_{\alpha<\alpha_0}$  being locally finite, there exists a neighbourhood V of the point x such that the set  $\{\alpha < \alpha_0 : V \cap F_{\alpha} \neq \emptyset\}$  is finite. The intersection  $W = U \cap V$  is a neighbourhood of the point x which meets only finitely many sets  $F_{\alpha}$ .

The next result is still another sum theorem. It will yield two further sum theorems which hold in the class of weakly paracompact strongly hereditarily normal spaces. Let us observe that in this class Theorem 2.3.13 generalizes the locally finite sum theorem (see Problem 2.3.F) and Theorem 2.3.15 generalizes both the countable and the locally finite sum theorems.

**2.3.13. The point-finite sum theorem for** Ind. If a strongly hereditarily normal space X can be represented as the union of a family  $\{F_s\}_{s\in S}$  of closed subspaces such that  $\operatorname{Ind} F_s \leq n$  for  $s \in S$ , and if there exists a point-finite open cover  $\{U_s\}_{s\in S}$  of the space X such that  $F_s \subset U_s$  for  $s \in S$ , then  $\operatorname{Ind} X \leq n$ .

**Proof.** Consider the decomposition of the space X described in Lemma 2.3.3. From the definition of the sets  $K_T$  it follows that  $K_T \subset \bigcup_{s \in T} F_s$  for  $T \in \mathcal{T}_i$ , so that  $\operatorname{Ind} K_T \leq n$  by virtue of Theorems 2.3.8 and 2.3.6. Theorem 2.2.8 and (12) imply that  $\operatorname{Ind} K_i \leq n$  for i = 1, 2, ... To complete the proof it suffices to apply Theorem 2.3.1.  $\Box$ 

**2.3.14. Theorem.** If a weakly paracompact strongly hereditarily normal space X can be represented as the union of a family  $\{U_s\}_{s\in S}$  of open subspaces such that  $\operatorname{Ind} U_s \leq n$  for  $s \in S$ , then  $\operatorname{Ind} X \leq n$ .

**Proof.** The space X being weakly paracompact, one can assume that the cover  $\{U_s\}_{s\in S}$  is point-finite, and thus has a closed shrinking  $\{F_s\}_{s\in S}$  (see [GT], Theorem 1.5.18). To complete the proof it suffices to apply Theorems 2.2.1 and 2.3.13.

**2.3.15. Theorem.** If a weakly paracompact strongly hereditarily normal space X can be represented as the union of a locally countable family  $\{F_s\}_{s \in S}$  of closed subspaces such that  $\operatorname{Ind} F_s \leq n$  for  $s \in S$ , then  $\operatorname{Ind} X \leq n$ .

**Proof.** For every point  $x \in X$  there exist a neighbourhood  $U_x$  and a countable set  $S(x) \subset S$  such that  $U_x \cap F_s = \emptyset$  for  $s \in S \setminus S(x)$ . From the last relation it follows that  $U_x \subset \bigcup \{F_s : s \in S(x)\}$ , so that by virtue of Theorems 2.3.8 and 2.3.6 we have  $\operatorname{Ind} U_x \leq n$  for  $x \in X$ . To complete the proof it suffices to apply Theorem 2.3.14.  $\Box$ 

We close this section with a characterization of the dimension Ind in strongly hereditarily normal spaces which will be applied in Section 2.4.

**2.3.16. Lemma.** If for a pair A, B of disjoint closed subsets of a topological space X there exists a  $\sigma$ -locally finite open cover  $\mathscr{V}$  of the space X which has the property that for every  $V \in \mathscr{V}$  either  $\overline{V} \cap A = \emptyset$  or  $\overline{V} \cap B = \emptyset$ , then there exists a partition L between A and B such that

(23) 
$$L \subset \bigcup \{ \operatorname{Fr} V \colon V \in \mathscr{V} \}.$$

**Proof.** Let  $\mathscr{V} = \bigcup_{i=1}^{\infty} \mathscr{V}_i$ , where the families  $\mathscr{V}_i$  are locally finite. For i = 1, 2, ... define  $\mathscr{W}_i = \{ V \in \mathscr{V}_i : \overline{V} \cap A = \emptyset \}$  and  $\mathscr{U}_i = \mathscr{V}_i \setminus \mathscr{W}_i$ ; consider the sets

 $U_i = \bigcup \mathscr{U}_i$  and  $W_i = \bigcup \mathscr{W}_i$ .

The family  $\mathscr{V}_i$  being locally finite,

(24) 
$$\overline{U}_i \cap B = \emptyset = \overline{W}_i \cap A \quad \text{for } i = 1, 2, ...;$$

moreover

(25) 
$$A \subset \bigcup_{i=1}^{\infty} U_i \text{ and } B \subset \bigcup_{i=1}^{\infty} W_i.$$

Let

(26) 
$$G_i = U_i \setminus \bigcup_{j < i} \overline{W_i}$$
 and  $H_i = W_i \setminus \bigcup_{j \leq i} \overline{U_j}$ .

From (24)-(26) it follows that the open sets  $U = \bigcup_{i=1}^{\infty} G_i$  and  $W = \bigcup_{i=1}^{\infty} H_i$ satisfy the conditions  $A \subset U$ ,  $B \subset W$  and  $U \cap W = \emptyset$ . From the local finiteness of  $\mathscr{V}_i$  it follows that  $\operatorname{Fr} U_i \cup \operatorname{Fr} W_i \subset \bigcup \{\operatorname{Fr} V: V \in \mathscr{V}_i\}$ , so that to complete the proof it suffices to show that the partition  $L = X \setminus (U \cup W)$ between A and B satisfies the inclusion

(27) 
$$L \subset \bigcup_{i=1}^{\infty} \operatorname{Fr} U_i \cup \bigcup_{i=1}^{\infty} \operatorname{Fr} W_i.$$

Consider a point  $x \in L$  and denote by F the first element of the sequence  $\overline{U_1}, \overline{W_1}, \overline{U_2}, \overline{W_2}, \dots$  that contains the point x. If  $F = \overline{U_i}$ , then  $x \in \operatorname{Fr} U_i$ =  $\overline{U_i} \setminus U_i$ , because  $x \notin G_i$  and  $x \notin \overline{W_j}$  for j < i. On the other hand, if  $F = \overline{W}_i, \text{ then } x \in \operatorname{Fr} \overline{W}_i \approx \overline{W}_i \setminus W_i, \text{ because } x \notin H_i \text{ and } x \notin \overline{U}_j \text{ for } j \leqslant i.$ Hence in both cases  $x \in \bigcup_{i=1}^{\infty} \operatorname{Fr} U_i \cup \bigcup_{i=1}^{\infty} \operatorname{Fr} W_i.$ 

**2.3.17. Theorem.** For every strongly hereditarily normal space X and each integer  $n \ge 0$  the following conditions are equivalent:

- (a) The space X satisfies the inequality  $Ind X \leq n$ .
- (b) Every locally finite open cover of the space X has a locally finite open refinement  $\mathscr{V}$  such that  $\operatorname{Ind} \operatorname{Fr} V \leq n-1$  for  $V \in \mathscr{V}$ .
- (c) Every two-element open cover of the space X has a  $\sigma$ -locally finite open refinement  $\mathscr{V}$  such that  $\operatorname{Ind} \operatorname{Fr} V \leq n-1$  for every  $V \in \mathscr{V}$ .

**Proof.** Let  $\{U_s\}_{s\in S}$  be a locally finite open cover of a normal space X such that  $\operatorname{Ind} X \leq n \geq 0$ . Consider a closed shrinking  $\{F_s\}_{s\in S}$  of the cover  $\{U_s\}_{s\in S}$  (see [GT], Theorem 1.5.18) and for every  $s \in S$  choose an open set  $V_s \subset X$  such that

$$F_s \subset V_s \subset U_s$$
 and  $\operatorname{Ind} \operatorname{Fr} V_s \leq n-1$ .

Clearly, the family  $\mathscr{V} = \{V_s\}_{s\in S}$  is a locally finite open refinement of the cover  $\{U_s\}_{s\in S}$ . Hence (a)  $\Rightarrow$  (b) for every normal space X.

The implication (b)  $\Rightarrow$  (c) being obvious, to complete the proof it suffices to show that (c)  $\Rightarrow$  (a). Let X be a strongly hereditarily normal space which satisfies (c). Consider a pair A, B of disjoint closed subsets of X. There exist open sets U,  $W \subset X$  such that  $A \subset U$ ,  $B \subset W$  and  $\overline{U} \cap \overline{W} = \emptyset$ . The two-element open cover  $\{X \setminus \overline{U}, X \setminus \overline{W}\}$  of the space X has a  $\sigma$ -locally finite open refinement  $\mathscr{V}$  such that  $\operatorname{Ind} \operatorname{Fr} V \leq n-1$  for every  $V \in \mathscr{V}$ . Since for every  $V \in \mathscr{V}$  either  $\overline{V} \cap A = \emptyset$  or  $\overline{V} \cap B = \emptyset$ , by virtue of Lemma 2.3.16 there exists a partition L between A and B such that  $L \subset \bigcup {\operatorname{Fr} V \colon V \in \mathscr{V}}$ . From Theorem 2.3.11 it follows that  $\operatorname{Ind} L \leq n-1$ , so that  $\operatorname{Ind} X \leq n$ .  $\Box$ 

## Historical and bibliographic notes

The study of relations between properties  $(\mu_n)$  and  $(\sigma_n)$  was originated by Dowker in [1953]. Dowker's paper contains Theorem 2.3.1, Corollary 2.3.2, Lemma 2.3.4 with "point-finite" replaced by "locally finite", as well as theorems 2.3.6, 2.3.7 and 2.3.8 for totally normal spaces (Theorems

### Problems

2.3.6 and 2.3.8 for perfectly normal spaces were established by Čech in [1932]). Lemma 2.3.3 and the present version of Lemma 2.3.4 were given by Lifanov and Pasynkov in [1970]; the same paper contains Theorems 2.3.6, 2.3.7 and 2.3.8 for Dowker spaces (announced in Pasynkov [1967]). The last three theorems were extended to super normal spaces by Nishiura in [1977]. Theorem 2.3.9 was established by Lifanov and Pasynkov in [1970]. Theorem 2.3.10 was proved by Kimura in [1967] for totally normal spaces (under the additional assumption of countable paracompactness in [1963]) and was extended to super normal spaces by Nishiura in [1977]. Theorems 2.3.12, 2.3.14 and 2.3.15 were proved by Lifanov and Pasynkov in [1970] for Dowker spaces; for totally normal spaces, Theorem 2.3.14 was given by Dowker in [1955], and Theorem 2.3.16—by Kimura in [1967] (implicitly). Theorem 2.3.17 was proved by Nagami in [1969] and [1960a], respectively for totally normal spaces and hereditarily paracompact spaces.

## Problems

**2.3.A** (Smirnov [1951]). Let X be a normal space with the property that for each closed subspace  $Y \subset X$  and every finite sequence  $F_1, F_2, \ldots, F_k$  of closed subspaces of Y such that  $Y = \bigcup_{i=1}^{k} F_i$ , if  $\operatorname{Ind} F_i \leq n$  for  $i = 1, 2, \ldots, k$ , then  $\operatorname{Ind} Y \leq n$ . Prove that  $\operatorname{ind} \beta X = \operatorname{Ind} X$ .

Deduce that for every strongly hereditarily normal space X we have ind  $\beta X = \text{Ind } X$ .

**2.3.B.** Show that under the additional hypothesis that X is a countably paracompact space Theorem 2.3.10 is an easy consequence of Theorems 2.3.1, 2.3.6 and 2.3.8.

Hint. For i = 1, 2, ... denote by  $K_i$  the set of all points of the space X which belong to exactly *i* members of the cover  $\{F_s\}_{s\in S}$  and by  $\mathcal{T}_i$  the family of all subsets of S that have exactly *i* elements. Note that  $K_i = \bigcup_{T \in \mathcal{T}_i} K_T$ , where  $K_T = K_i \cap \bigcap_{s \in T} F_s$ . Show that the sets  $U_i = K_1 \cup K_2 \cup ... \cup K_i$  are open in X and that Ind  $U_i \leq n$  for i = 1, 2, ... Apply the fact that the open cover  $\{U_i\}_{i=1}^{\infty}$  of the space X has a closed shrinking (see [GT], Theorem 5.2.3). **2.3.C.** (a) Observe that Theorem 2.3.7 is a consequence of Theorem 2.3.8 and Problem 2.2.C(e).

(b) Deduce from Theorem 2.3.9 that if a strongly hereditarily normal space X can be represented as the union of a transfinite sequence  $K_1, K_2, ...$ ...,  $K_{\alpha}, ..., \alpha < \xi$  of subspaces such that  $\operatorname{Ind} K_{\alpha} \leq n$  and the union  $\bigcup_{\beta < \alpha} K_{\beta}$  is closed for  $\alpha < \xi$ , and the family  $\{K_{\alpha}\}_{\alpha < \xi}$  is locally finite, then  $\operatorname{Ind} X \leq n$ . Observe that the latter fact is also a consequence of Theorem 2.3.10 and Problem 2.2.C(e).

**2.3.D** (Lifanov and Pasynkov [1970]). Prove that if a hereditarily normal space X can be represented as the union of a transfinite sequence  $K_1, K_2, ..., K_{\alpha}, ..., \alpha < \xi$  of pairwise disjoint subspaces such that  $\operatorname{Ind} K_{\alpha} \leq n$ , the union  $\bigcup_{\beta < \alpha} K_{\beta}$  is closed and the family  $\{K_{\beta}\}_{\beta < \alpha}$  is locally finite for  $\alpha < \xi$ , and the family  $\{K_{\alpha}\}_{\alpha < \xi}$  is locally countable, then  $\operatorname{Ind} X \leq n$ .

**2.3.E.** (a) (Nagata [1965]) Prove that if a strongly hereditarily normal space X can be represented as the union of a transfinite sequence  $F_1, F_2, \ldots$ ...,  $F_{\alpha}, \ldots, \alpha < \xi$  of closed subspaces such that  $\operatorname{Ind} F_{\alpha} \leq n$  for  $\alpha < \xi$ , and if there exists a transfinite sequence  $U_1, U_2, \ldots, U_{\alpha}, \ldots, \alpha < \xi$  of open subsets of X such that  $F_{\alpha} \subset U_{\alpha}$  and the family  $\{U_{\beta}\}_{\beta<\alpha}$  is locally finite for  $\alpha < \xi$ , then  $\operatorname{Ind} X \leq n$ .

*Hint*. Observe that if the set of all ordinal numbers less than  $\xi$  contains no countable cofinal subset, then the family  $\{U_{\alpha}\}_{\alpha < \xi}$  is point-finite.

(b) (Nagata [1967]) Prove that if a strongly hereditarily normal space X can be represented as the union of a transfinite sequence  $K_1, K_2, ...$ ...,  $K_{\alpha}, ..., \alpha < \xi$  of subspaces such that  $\operatorname{Ind} K_{\alpha} \leq n$  and the union  $\bigcup_{\beta < \alpha} K_{\beta}$  is closed for  $\alpha < \xi$ , and if there exists a transfinite sequence  $U_1, U_2, ...$ ...,  $U_{\alpha}, ..., \alpha < \xi$  of open subsets of X such that  $K_{\alpha} \subset U_{\alpha}$  and the family  $\{U_{\beta}\}_{\beta < \alpha}$  is locally finite for  $\alpha < \xi$ , then  $\operatorname{Ind} X \leq n$ .

**2.3.F** (Smith and Krajewski [1971]). Prove that for every locally finite family  $\{F_s\}_{s\in S}$  of closed subsets of a weakly paracompact space X there exists a point-finite family  $\{U_s\}_{s\in S}$  of open sets such that  $F_s \subset U_s$  for  $s \in S$ .

*Hint.* For every point  $x \in X$  choose a neighbourhood  $U_x$  which meets only finitely many sets  $F_s$ . Consider a point-finite open refinement  $\mathscr{V}$  of the cover  $\{U_x\}_{x\in X}$  and let  $U_s = \bigcup \{V \in \mathscr{V} : V \cap F_s \neq \emptyset\}$ .

# 2.4. Relations between the dimensions ind and Ind. Cartesian product theorems for the dimension Ind

As shown in Section 1.6, for every separable metric space X we have ind X = IndX. In the present section this equality will be extended to a somewhat larger class of spaces. However, we should make it clear at once that the class of all spaces whose small inductive and large inductive dimensions coincide is rather restricted. Indeed, there exists a first-countable compact space X such that indX = 2 and IndX = 3, as well as a completely metrizable space X, known as *Roy's space*, such that indX = 0and  $IndX = \dim X = 1$ . The description of the above two spaces is quite complicated and the computation of their dimensions is rather difficult, so that they will not be reproduced in this book. The only space with non-coinciding inductive dimensions discussed here in detail is the space X described in Example 2.2.11, which satisfies the equalities indX = 0 and IndX = 1.

Let us recall that a topological space X is strongly paracompact<sup>1)</sup> if X is a Hausdorff space and every open cover of the space X has a star-finite open refinement (a family of sets  $\mathcal{A}$  is star-finite if every set  $A \in \mathcal{A}$  meets only finitely many members of  $\mathcal{A}$ ). It immediately follows from the definitions that every strongly paracompact space is paracompact. One proves that every Lindelöf space is strongly paracompact (see [GT], Corollary 5.3.11).

In the considerations of this section we shall apply the fact that starfinite covers decompose in a natural way into countable families of sets. Let us recall that the *component* of a member  $A_0$  of a family of sets  $\mathcal{A}$ is the subfamily  $\mathcal{A}_0 \subset \mathcal{A}$  consisting of all sets  $A \in \mathcal{A}$  for which there exists a finite sequence  $A_1, A_2, \ldots, A_k$  of members of  $\mathcal{A}$  such that  $A_k = A$ and  $A_i \cap A_{i-1} \neq \emptyset$  for  $i = 1, 2, \ldots, k$ . The components of two distinct members of  $\mathcal{A}$  either coincide or are disjoint, so that

$$\mathscr{A} = \bigcup_{s \in S} \mathscr{A}_s$$
, where  $\mathscr{A}_s \cap \mathscr{A}_{s'} = \emptyset$  for  $s \neq s'$ 

and for every  $s \in S$  the family  $\mathscr{A}_s$  is the component of a member of  $\mathscr{A}$ . The representation of  $\mathscr{A}$  as the union of the families  $\mathscr{A}_s$  will be called the decomposition of  $\mathscr{A}$  into components; clearly  $(\bigcup \mathscr{A}_s) \cap (\bigcup \mathscr{A}_{s'}) = \varnothing$ for  $s \neq s'$ . Let us observe that all components of a star-finite family  $\mathscr{A}$ are countable. Indeed, the component  $\mathscr{A}_0$  of the set  $\mathscr{A}_0 \in \mathscr{A}$  can be rep-

<sup>&</sup>lt;sup>1)</sup> The term hypocompact is also used.

resented as the union of families  $\mathscr{A}_1, \mathscr{A}_2, ...,$  where  $\mathscr{A}_k$  consists of all sets  $A \in \mathscr{A}$  for which there exist a sequence  $A_1, A_2, ..., A_k$  of k members of  $\mathscr{A}$  such that  $A_k = A$  and  $A_i \cap A_{i-1} \neq \emptyset$  for i = 1, 2, ..., k, and—as can easily be verified—each of the families  $\mathscr{A}_k$  is countable.

**2.4.1. Lemma.** If X is a strongly paracompact space such that  $\operatorname{ind} X \leq n \geq 0$ , then for every pair A, B of disjoint closed subsets of X there exists a partition L between A and B which can be represented as the union of a sequence  $L_1, L_2, \ldots$  of closed subspaces of X such that  $\operatorname{ind} L_i \leq n-1$  for  $i = 1, 2, \ldots$ 

**Proof.** For every point  $x \in X$  there exists a neighbourhood  $U_x$  such that ind Fr  $U_x \leq n-1$  and either  $\overline{U}_x \cap A = \emptyset$  or  $\overline{U}_x \cap B = \emptyset$ . Consider a starfinite open refinement  $\mathscr{U}$  of the cover  $\{U_x\}_{x\in X}$  of the space X. Let  $\mathscr{U} = \bigcup_{s\in S} \mathscr{U}_s$ be the decomposition of the cover  $\mathscr{U}$  into components; as the cover  $\mathscr{U}$ is star-finite, the components are countable, i.e.,  $\mathcal{U}_s = \{U_{s,i}\}_{i=1}^{\infty}$  for  $s \in S$ . Let  $X_s = \bigcup_{i=1}^{s} U_{s,i}$  for every  $s \in S$ ; the sets  $X_s$  are open and pairwise disjoint, so that  $X = \bigoplus_{s \in S} X_s$ . For each set  $U_{s,i}$  choose a point  $x(s,i) \in X$ such that  $U_{s,i} \subset U_{x(s,i)}$  and let  $V_{s,i} = U_{x(s,i)} \cap X_s$ ; clearly,  $X_s = \bigcup_{i=1}^{\infty} V_{s,i}$ and we have either  $\overline{V}_{s,i} \cap A = \emptyset$  or  $\overline{V}_{s,i} \cap B = \emptyset$ . Since for i = 1, 2, ...the family  $\{V_{s,i}\}_{s\in S}$  is discrete, the family  $\mathscr{V}$  consisting of all the sets  $V_{s,i}$ is a  $\sigma$ -locally finite open cover of the space X. By virtue of Lemma 2.3.16 there exists a partition L between A and B such that  $L \subset \bigcup_{i=1}^{i} F_i$ , where  $F_i = \bigcup_{s \in S} \operatorname{Fr} V_{s,i}$ . Let  $L_i = L \cap F_i$  for i = 1, 2, ... To complete the proof it suffices to show that  $L_i$  is closed in X and  $ind L_i \leq n-1$  for i = 1, 2, ...Since  $\operatorname{Fr} V_{s,i} \subset X_s$  for  $s \in S$ ,  $F_i = \bigoplus_{s \in S} \operatorname{Fr} V_{s,i}$  and  $F_i$  is a closed subset of X; thus  $L_i$  is closed in X for i = 1, 2, ... On the other hand,  $\operatorname{Fr} V_{s,i} \subset \operatorname{Fr} U_{x(s,i)} \cup$  $\cup$  Fr $X_s =$  Fr $U_{x(s,i)}$ , so that ind Fr $V_{s,i} \leq n-1$  for  $s \in S$ ; thus ind  $F_i \leq n-1$ and  $\operatorname{ind} L_i \leq n-1$  for  $i = 1, 2, \dots$ 

From Lemma 2.4.1 we immediately obtain the following generalization of Theorem 1.6.5.

**2.4.2. Theorem.** For every strongly paracompact space X the conditions ind X = 0 and Ind X = 0 are equivalent.  $\Box$ 

The next theorem shows that one step forward is possible.

**2.4.3. Theorem.** For every strongly paracompact space X the conditions ind X = 1 and Ind X = 1 are equivalent.

**Proof.** In view of Theorem 2.4.2 it suffices to show that for every strongly paracompact space X such that indX = 1 we have IndX = 1. Consider a pair A, B of disjoint closed subsets of the space X. By virtue of Lemma

2.4.1 there exists a partition L between A and B such that  $L = \bigcup_{i=1}^{n} L_i$ , where  $L_i$  is closed in X and  $\operatorname{ind} L_i \leq 0$  for i = 1, 2, ... As every closed subspace of a strongly paracompact space is strongly paracompact, it follows from Theorem 2.4.2 that  $\operatorname{Ind} L_i \leq 0$  for i = 1, 2, ..., so that  $\operatorname{Ind} L \leq 0$  by virtue of Theorem 2.2.7. Thus  $\operatorname{Ind} X \leq 1$ , and from the inequality  $\operatorname{ind} X \leq \operatorname{Ind} X$  it follows that  $\operatorname{Ind} X = 1$ .  $\Box$ 

**2.4.4. Theorem.** For every strongly paracompact strongly hereditarily normal space X we have ind X = Ind X.

**Proof.** It suffices to show that  $\operatorname{Ind} X \leq \operatorname{ind} X$ ; clearly, one can suppose that  $\operatorname{ind} X < \infty$ . We shall apply induction with respect to  $\operatorname{ind} X$ . The inequality holds if  $\operatorname{ind} X = -1$ . Assume that the inequality is proved for all strongly paracompact strongly hereditarily normal spaces of small inductive dimension less than  $n \geq 0$  and consider a strongly paracompact strongly hereditarily normal space X such that  $\operatorname{ind} X = n$ . Let A, B be a pair of disjoint closed subsets of X. By virtue of Lemma 2.4.1 and by the inductive assumption there exists a partition L between A and B such that  $L = \bigcup_{i=1}^{\infty} L_i$ , where  $L_i$  is closed in X and  $\operatorname{Ind} L_i \leq n-1$  for i = 1, 2, ...

By virtue of Theorem 2.3.8 we have  $\operatorname{Ind} L \leq n$ , so that  $\operatorname{Ind} X \leq n = \operatorname{ind} X$ .  $\Box$ 

Let us note that in the realm of Lindelöf spaces Lemma 2.4.1 is an immediate consequence of Lemma 2.3.16, so that the fact that Theorems 2.4.2, 2.4.3 and 2.4.4 hold for Lindelöf spaces can be obtained independently of the above-mentioned result that every strongly paracompact space is a Lindelöf space.

Let us also note that an obvious modification of the proof of Theorem 2.4.4 shows that indX = IndX for every strongly paracompact space X in each closed subspace of which the countable sum theorem holds either for ind or for Ind.

The large inductive dimension

In the problems listed at the end of this section Theorems 2.4.2, 2.4.3 and 2.4.4 are extended to some classes of spaces larger than the class of strongly paracompact spaces. However, the definitions of these classes of spaces are a little less natural than the definition of strongly paracompact spaces and, to some degree, are inspired by the methods of proofs.

We now turn to a study of the behaviour of the dimension Ind under Cartesian multiplication. Let us begin with recalling that, as stated in Section 2.2, there exist compact spaces X and Y such that  $Ind(X \times Y) > IndX +$  $+ \operatorname{Ind} Y$ . Thus, one can see that the Cartesian product theorem for Ind, i.e., the inequality  $Ind(X \times Y) \leq IndX + IndY$ , requires rather strong assumptions on X and Y. Several theorems of this type were proved under various assumptions. Usually, one assumes that X and Y have some properties related to paracompactness and that the Cartesian product  $X \times Y$ is totally normal; generally, this last assumption can be weakened to the assumption that  $X \times Y$  is strongly hereditarily normal or to the assumption that the finite sum theorem for Ind holds in closed subspaces of Xand Y, i.e., that, for every closed subspace Z of either X or Y,  $\operatorname{Ind} Z \leq n$ whenever Z can be represented as the union of a finite number of closed subspaces  $F_1, F_2, ..., F_k$  such that  $\operatorname{Ind} F_i \leq n$  for i = 1, 2, ..., k. Among the Cartesian product theorems for Ind so far discovered there is no strongest result. We shall quote two such theorems, which are relatively strong. Thus, the inequality  $\operatorname{Ind}(X \times Y) \leq \operatorname{Ind} X \times \operatorname{Ind} Y$  holds for every pair X, Y of normal spaces of which at least one is non-empty provided that either

- (i) the finite sum theorem for Ind holds in closed subspaces of X and Y the Cartesian product  $X \times Y$  is normal and one of the factors is compact (more generally: the projection onto one of the factors is a closed mapping),
- or
- (ii) the finite sum theorem for Ind holds in closed subspaces of X and Y, the Cartesian product  $X \times Y$  is normal, one of the factors is metrizable and the other is countably paracompact (more generally: one of the factors can be mapped to a metrizable space by a perfect mapping and the other is countably paracompact).

Let us observe that the proofs of Cartesian product theorems for Ind are fairly difficult; yet the difficulties are chiefly connected with the complicated structure of subsets of Cartesian products, so that—roughly speaking—they are of "topological" rather than "dimensional" nature. As a sample we shall prove here two Cartesian product theorems for Ind which are among the simplest.

Arguing as in the proof of Theorem 1.5.16, we obtain

**2.4.5. Lemma.** For every pair X, Y of regular spaces of which at least one is non-empty such that the finite sum theorem for ind holds in closed subspaces of the Cartesian product  $X \times Y$  we have

 $\operatorname{ind}(X \times Y) \leq \operatorname{ind} X + \operatorname{ind} Y.$ 

Theorem 2.4.4 and Lemma 2.4.5 yield

**2.4.6. Theorem.** For every pair X, Y of normal spaces of which at least one is non-empty such that the Cartesian product  $X \times Y$  is strongly paracompact and strongly hereditarily normal we have

$$\operatorname{Ind}(X \times Y) \leq \operatorname{Ind} X + \operatorname{Ind} Y.$$

**2.4.7. Lemma.** The Cartesian product  $X \times Y$  of a perfectly normal space X and a metrizable space Y is perfectly normal.

**Proof.** As  $X \times Y$  is a  $T_1$ -space it suffices to show (see Problem 2.1.E (b)) that for every open set  $W \subset X \times Y$  there exists a sequence  $W_1, W_2, \ldots$ of open subsets of  $X \times Y$  such that  $W = \bigcup_{i=1}^{\infty} W_i$  and  $\overline{W}_i \subset W$  for  $i = 1, 2, \ldots$ Let  $\mathscr{V}$  be a base for the space Y which can be represented as the union of locally finite families  $\mathscr{V}_1, \mathscr{V}_2, \ldots$  (see [GT], Corollary 4.4.4). Consider the family  $\mathscr{W}$  of all sets  $U \times V$ , where U is an open subset of X and  $V \in \mathscr{V}$ , such that  $\overline{U} \times \overline{V} \subset W$ ; clearly  $W = \bigcup \mathscr{W}$ . For every  $V \in \mathscr{V}$  the union  $U(V) = \bigcup \{U: U \times V \in \mathscr{W}\}$  is an open  $F_{\sigma}$ -set in X, therefore U(V) $= \bigcup_{j=1}^{\infty} U_j(V)$ , where  $U_j(V)$  is open and  $\overline{U_j(V)} \subset U(V)$  for  $j = 1, 2, \ldots$ The family  $\mathscr{W}_{j,k} = \{U_j(V) \times V: V \in \mathscr{V}_k\}$  is locally finite in  $X \times Y$  for  $j, k = 1, 2, \ldots$ , so that the set  $W_{j,k} = \bigcup \mathscr{W}_{j,k}$  satisfies the inclusion  $\overline{W}_{j,k} \subset W$ . Since  $\bigcup_{j,k=1}^{\infty} W_{j,k} = \bigcup \mathscr{W} = W$ , to complete the proof it suffices to arrange all the sets  $W_{j,k}$  into a sequence  $W_1, W_2, \ldots$ 

[Ch. 2, § 4

**2.4.8. Theorem.** For every perfectly normal space X and every metrizable space Y of which at least one is non-empty we have

$$\operatorname{Ind}(X \times Y) \leq \operatorname{Ind} X + \operatorname{Ind} Y.$$

**Proof.** The theorem is obvious if the dimension Ind of one of the spaces X, Y equals  $\infty$ , so that we can suppose that  $k(X, Y) = \operatorname{Ind} X + \operatorname{Ind} Y$  is finite. We shall apply induction with respect to that number. If k(X, Y) = -1, then our inequality holds. Assume that the inequality is proved for every perfectly normal space and every metrizable space such that at least one of them is non-empty and the sum of large inductive dimensions of which is less than  $k \ge 0$ , and consider a perfectly normal space X and a metrizable space Y such that  $\operatorname{Ind} X = n \ge 0$ ,  $\operatorname{Ind} Y = m \ge 0$  and n+m = k. Let  $\varrho$  be a metric on the space Y. Since every metrizable space is paracompact, the space Y has for i = 1, 2, ... a locally finite open cover  $\mathcal{U}_i$  such that  $\operatorname{Ind} \operatorname{Fr} V \le m-1$  for every  $\infty$ 

 $V \in \mathscr{V}_i$ . Obviously, the family  $\mathscr{V} = \bigcup_{i=1}^{\infty} \mathscr{V}_i$  is a base for the space Y. Consider now an arbitrary two-element open cover  $\{G, H\}$  of the Cartesian product  $X \times Y$ ; let  $\mathscr{W}$  be a refinement of the cover  $\{G, H\}$  consisting of sets of the form  $U \times V$ , where U is an open subset of X and  $V \in \mathscr{V}$ .

For every  $V \in \mathscr{V}$  the unions

$$G(V) = \bigcup \{ U: \ U \times V \in \mathscr{W} \text{ and } U \times V \subset G \} \text{ and}$$
$$H(V) = \bigcup \{ U: \ U \times V \in \mathscr{W} \text{ and } U \times V \subset H \}$$

are open  $F_{\sigma}$ -sets in X therefore

$$G(V) = \bigcup_{j=1}^{\infty} G_j(V)$$
 and  $H(V) = \bigcup_{j=1}^{\infty} H_j(V)$ ,

where  $G_j(V)$ ,  $H_j(V)$  are open and

Ind  $\operatorname{Fr} G_j(V) \leq n-1$  and  $\operatorname{Ind} \operatorname{Fr} H_j(V) \leq n-1$  for j = 1, 2, ...The family

$$\mathcal{W}_{j,i} = \{G_j(V) \times V \colon V \in \mathcal{V}_i\} \cup \{H_j(V) \times V \colon V \in \mathcal{V}_i\}$$

is locally finite in  $X \times Y$  for j, i = 1, 2, ..., so that the union  $\bigcup_{j,i=1}^{\infty} \mathscr{W}_{j,i}$  is a  $\sigma$ -locally finite open refinement of the cover  $\{G, H\}$ .

From Theorem 2.1.3 and Lemma 2.4.7 it follows that the Cartesian product  $X \times Y$  is strongly hereditarily normal. Thus, by virtue of Theorem 2.3.17, to complete the proof it suffices to show that  $\operatorname{Ind} \operatorname{Fr} W \leq k-1$  for every  $W \in \mathcal{W}_{j,i}$  and j, i = 1, 2, ... The last inequality is a consequence of the inclusions

$$\operatorname{Fr}(G_{j}(V) \times V) \subset (X \times \operatorname{Fr} V) \cup (\operatorname{Fr} G_{j}(V) \times Y)$$

and

$$\operatorname{Fr}(H_j(V) \times V) \subset (X \times \operatorname{Fr} V) \cup (\operatorname{Fr} H_j(V) \times Y),$$

the inductive assumption, and Theorem 2.3.8.

# Historical and bibliographic notes

The first example of a compact space X such that  $indX \neq IndX$  was defined by Filippov in [1969]; the example is discussed in detail in Filippov's paper [1970b] and in Pears' book [1975]. Simpler, but still quite complicated examples of such spaces were described by Filippov in [1970]. Pasynkov and Lifanov in [1970], and Pasynkov in [1970]; these last spaces, moreover, are first-countable. The example of a completely metrizable space X such that ind X = 0 and Ind X = dim X = 1 was outlined by Roy in [1962]; a detailed discussion of this example is contained in Roy's paper [1968] and Pears' book [1975]. The first example of a normal space with non-coinciding inductive dimensions was given by Smirnov in [1951]. Theorems 2.4.2 and 2.4.3 were proved by Vedenissoff in [1939] under the stronger assumption that X is a Lindelöf space. Theorems 2.4.4 and 2.4.6 were given by Katuta in [1966] with strong hereditary normality replaced by total normality; an important special case of Theorem 2.4.4, viz., the equality ind X = Ind X for every strongly paracompact metrizable space X, was proved by Morita in [1950a] (see also notes to Theorem 1.5.13). The fact that the inequality  $Ind(X \times Y) \leq IndX + IndX$  holds for every pair X, Y of spaces satisfying either (i) or (ii) was proved by Filippov in [1979] (announcement [1973]); part (ii) was announced independently by Pasynkov in [1973]. In the original formulation of (ii) Filippov and Pasynkov assume that the Cartesian product  $X \times Y$  is normal and countably paracompact; since Rudin and Starbird established in [1975] that if the Cartesian product of a metrizable space and a countably paracompact space is normal, then it is also countably paracompact, the assumption of the countable paracompactness of  $X \times Y$  can be dropped. Theorem 2.4.8 was obtained by Kimura in [1963]; under the additional assumption that X is a paracompact space it was proved by Nagami in [1960a]. Further information on Cartesian product theorems for Ind can be found in Kimura [1963], Nagata [1967], Lifanov [1968], Nagami [1969], Pasynkov [1969], van Dalen [1972] and Pears [1975].

# Problems

2.4.A. (a) (Nagata [1957]) Prove that the Cartesian product of the open unit interval (0, 1) and the Baire space  $B(\aleph_1)$  (see Example 4.1.23) is not strongly paracompact. Deduce that the class of all strongly paracompact metrizable spaces is not hereditary with respect to  $F_{\sigma}$ -sets and is not finitely multiplicative. Show that a metrizable space which can be represented as the union of a locally finite countable family of strongly paracompact closed subspaces is not necessarily strongly paracompact.

(b) Show that every paracompact space X such that IndX = 0 is strongly paracompact.

(c) Show that the Cartesian product  $X \times Y$  of a strongly paracompact space X and a compact space Y is strongly paracompact.

(d) (Morita [1954]) Prove that the Cartesian product  $I^{\aleph_0} \times B(\mathfrak{m})$  is a universal space for the class of all strongly paracompact metrizable spaces whose weight is not larger than  $\mathfrak{m} \ge \aleph_0$ .

**2.4.B** (Zarelua [1963] (announcement [1961])). A topological space X is called *completely paracompact* if X is a regular space and for every open cover  $\mathscr{U}$  of the space X there exists a sequence  $\mathscr{V}_1, \mathscr{V}_2, \ldots$  of star-finite open covers of X such that the union  $\bigcup_{i=1}^{\infty} \mathscr{V}_i$  contains a refinement of  $\mathscr{U}$ .

(a) Observe that every completely paracompact space is paracompact and that every strongly paracompact space is completely paracompact. Show that in the realm of connected spaces complete paracompactness is equivalent to the Lindelöf property. Prove that complete paracompactness is hereditary with respect to  $F_{\sigma}$ -sets.

(b) Prove that a metrizable space is completely paracompact if and only if it has a base which can be represented as the union of countably many star-finite covers. Deduce that the class of completely paracompact metrizable spaces is hereditary and countably multiplicative. Show that

### Problems

a metrizable space which can be represented as the union of a countable family of completely paracompact closed subspaces is not necessarily completely paracompact.

(c) Prove that the Cartesian product  $I^{\aleph_0} \times B(\mathfrak{m})$  is a universal space for the class of all completely paracompact metrizable spaces the weight of which is not larger than  $\mathfrak{m} \ge \aleph_0$ .

(d) Give an example of a completely paracompact metrizable space which cannot be represented as the union of a  $\sigma$ -locally finite family of strongly paracompact closed subspaces.

*Hint*. Consider the Cartesian product  $\mathbb{R}^{\aleph_0} \times \mathbb{B}(\aleph_1)$ .

(e) Check that Theorems 2.4.2, 2.4.3 and 2.4.4 remain valid for completely paracompact spaces.

**2.4.C.** (a) (Nagami [1969]) A topological space X is called  $\sigma$ -totally paracompact if X is a regular space and for every base  $\mathscr{B}$  for the space X there exists a  $\sigma$ -locally finite open cover  $\mathscr{V}$  of X such that for each  $V \in \mathscr{V}$  one can find a  $U \in \mathscr{B}$  satisfying  $V \subset U$  and  $\operatorname{Fr} V \subset \operatorname{Fr} U$ .

Show that the class of  $\sigma$ -totally paracompact spaces is hereditary with respect to  $F_{\sigma}$ -sets. Check that Theorems 2.4.2, 2.4.3 and 2.4.4 remain valid for  $\sigma$ -totally paracompact spaces.

(b) (Zarelua [1963] (announcement [1961])) Observe that every  $\sigma$ -totally paracompact space is paracompact and that every completely paracompact space is  $\sigma$ -totally paracompact.

**2.4.D.** (a) (Fitzpatrick and Ford [1967]) A topological space X is order totally paracompact if X is a regular space and for every base  $\mathscr{B}$  for the space X there exists an open cover  $\{V_s\}_{s\in S}$ , where the set S is linearly ordered by a relation <, such that for every  $s_0 \in S$  the family  $\{V_{s_0} \cap V_s\}_{s < s_0}$  is locally finite in the space  $V_{s_0}$  and for each  $s \in S$  one can find a  $U \in \mathscr{B}$  satisfying  $V_s \subset U$  and  $\operatorname{Fr} V_s \subset \operatorname{Fr} U$ .

Show that the class of order totally paracompact spaces is hereditary with respect to closed subspaces.

(b) Prove that Theorem 2.4.2 remains valid for order totally paracompact spaces.

*Hint.* Check that if for a pair A, B of disjoint closed subsets of a topological space X there exists an open cover  $\{V_s\}_{s\in S}$  of X, where the set S is linearly ordered by a relation <, such that for every  $s_0 \in S$  either  $\overline{V}_{s_0} \cap A = \emptyset$  or  $\overline{V}_{s_0} \cap B = \emptyset$  and the family  $\{V_{s_0} \cap V_s\}_{s < s_0}$  is locally finite in the

space  $V_{s_0}$ , then there exists a partition L between A and B such that L  $\subset \bigcup_{s\in S} \operatorname{Fr} V_s.$ 

Remark. It is not known if Theorems 2.4.3 and 2.4.4 remain valid for order totally paracompact spaces.

(c) (Fitzpatrick and Ford [1967]) Prove that for every order totally paracompact metrizable space X we have indX = IndX.

*Hint*. Apply the hint to (b); use the fact that X has a  $\sigma$ -locally finite base.

(d) (Katuta [1967]) Prove that a regular space X is paracompact if and only if every open cover  $\mathscr{U}$  of X has an open refinement  $\{V_s\}_{s\in S}$ , where the set S is linearly ordered by a relation <, such that for every  $s \in S_0$ the family  $\{V_{s_0} \cap V_s\}_{s < s_0}$  is locally finite in the space  $V_{s_0}$ .

*Hint*. To prove that the space X is paracompact it suffices to show that every open cover of X has a locally finite refinement consisting of arbitrary sets (see [GT], Theorem 5.1.11).

(e) Deduce from (d) that every order totally paracompact space is paracompact and note that every  $\sigma$ -totally paracompact space is order totally paracompact.

2.4.E. Show that if a metrizable space X can be represented as the union of a locally countable family  $\{F_s\}_{s\in S}$  of closed subspaces such that  $\inf F_s$ =  $\operatorname{Ind} F_s$  for  $s \in S$ , then  $\operatorname{ind} X = \operatorname{Ind} X$ .

## CHAPTER 3

# THE COVERING DIMENSION

Chapter 2 was devoted to an examination of the question which results of the classical dimension theory of separable metric spaces hold in more general classes of spaces for the large inductive dimension Ind. In the present chapter, the covering dimension dim becomes the subject of similar considerations. It will be seen that outside the class of separable metric spaces the dimension dim behaves somewhat better than the dimension Ind, i.e., that for the dimension dim a larger number of theorems of the classical theory can be extended to topological spaces and that the extensions hold under weaker assumptions.

Section 3.1 is primarily devoted to the question of monotonicity of dim in general spaces and to a study of sum theorems for dim. We also discuss the relations between the covering dimension dim and the inductive dimensions ind and Ind; in particular, we prove that  $\dim X \leq \operatorname{Ind} X$  for every normal space X.

Section 3.2 begins with several characterizations of the dimension dim in normal spaces. They include generalizations of three important theorems of the classical theory of dimension, namely the theorems on partitions, on extending mappings to spheres and on  $\mathscr{E}$ -mappings. The final part of the section is devoted to a discussion of Cartesian product theorems for the covering dimension.

In Section 3.3 we establish the compactification and the universal space theorems for dim and characterize compact spaces whose covering dimension does not exceed n as limits of inverse systems of compact metric spaces of dimension  $\leq n$ .

We shall return briefly to the topic of this chapter in Section 4.3, where some information on relations between the covering dimension of the domain and the range of a closed mapping will be given.

# 3.1. Basic properties of the dimension dim in normal spaces. Relations between the dimensions ind, Ind and dim

The definition of the covering dimension dim was stated in Section 1.6; let us recall that a normal space X satisfies the inequality  $\dim X \leq n$  if and only if every finite open cover of the space X has a finite open refinement of order  $\leq n$ . We shall begin this section with a characterization of the dimension dim which demonstrates that instead of finite open refinements of order  $\leq n$  one can equally well consider finite closed refinements of order  $\leq n$ . The characterization is preceded by an auxiliary theorem on swellings of finite families of closed subsets of normal spaces which will be applied in its proof.

**3.1.1. Definition.** By a *swelling* of the family  $\{A_s\}_{s\in S}$  of subsets of a topological space X we mean any family  $\{B_s\}_{s\in S}$  of subsets of the space X such that  $A_s \subset B_s$  for every  $s \in S$  and for every finite set of indices  $s_1, s_2, \ldots, s_m \in S$ 

 $A_{s_1} \cap A_{s_2} \cap \ldots \cap A_{s_m} \neq \emptyset$  if and only if  $B_{s_1} \cap B_{s_2} \cap \ldots \cap B_{s_m} \neq \emptyset$ ;

a swelling is open if all its members are open subsets of the space X.

Clearly, every swelling  $\mathscr{B}$  of a family  $\mathscr{A}$  satisfies the equality ord $\mathscr{B}$  = ord  $\mathscr{A}$ .

The following theorem is, in a sense, dual to Theorem 1.7.8, which deals with shrinkings of finite open covers of normal spaces.

**3.1.2. Theorem.** Every finite family  $\{F_i\}_{i=1}^k$  of closed subsets of a normal space X has an open swelling  $\{U_i\}_{i=1}^k$ . If, moreover, a family  $\{V_i\}_{i=1}^k$  of open subsets of X satisfying  $F_i \subset V_i$  for i = 1, 2, ..., k is given, then the swelling can be defined in such a way that  $\overline{U}_i \subset V_i$  for i = 1, 2, ..., k.

**Proof.** The union  $E_1$  of all intersections  $F_{i_1} \cap F_{i_2} \cap \ldots \cap F_{i_m}$  satisfying the equality  $F_1 \cap F_{i_1} \cap F_{i_2} \cap \ldots \cap F_{i_m} = \emptyset$  is a closed set disjoint from  $F_1$ , so that there exist an open set  $U_1$  such that  $F_1 \subset U_1$  and  $\overline{U_1} \cap E_1 = \emptyset$ . One readily sees that the family  $\{\overline{U_1}, F_1, \ldots, F_k\}$  is a swelling of the family  $\{F_i\}_{i=1}^k$ .

Assume that for i = 1, 2, ..., n-1 an open set  $U_i$  is defined in such a way that  $F_i \subset U_i$  and the family  $\{\overline{U}_1, \overline{U}_2, ..., \overline{U}_{n-1}, F_n, ..., F_k\}$  is a swelling of the family  $\{F_i\}_{i=1}^k$ . The union  $E_n$  of all intersections of members of the family  $\{\overline{U}_1, \overline{U}_2, ..., \overline{U}_{n-1}, F_n, ..., F_k\}$  which are disjoint from  $F_n$  is a closed set disjoint from  $F_n$ , so that there exists an open set  $U_n$  such that  $F_n \subset U_n$  and  $\overline{U}_n \cap E_n = \emptyset$ . The family  $\{\overline{U}_1, \overline{U}_2, ..., \overline{U}_n, F_{n+1}, ..., F_k\}$  is a swelling of the family  $\{F_i\}_{i=1}^k$ . In this way we obtain open sets  $U_1, U_2, ...$ ...,  $U_k$  such that  $F_i \subset U_i$  for i = 1, 2, ..., k and the family  $\{\overline{U}_i\}_{i=1}^k$  is a swelling of the family  $\{F_i\}_{i=1}^k$ . Clearly, the family  $\{U_i\}_{i=1}^k$  is the required open swelling. The second part of the theorem is obvious.  $\Box$ 

Theorems 1.7.8 and 3.1.2 yield

**3.1.3. Proposition.** For every normal space X the following conditions are equivalent:

- (a) The space X satisfies the inequality  $\dim X \leq n$ .
- (b) Every finite open cover of the space X has a closed shrinking of order  $\leq n$ .
- (c) Every finite open cover of the space X has a finite closed refinement of order  $\leq n$ .  $\Box$

Since normality is not a hereditary property, it may happen that the dimension dim is defined for a space X and yet is not defined for a subspace  $M \subset X$ . Still, normality being hereditary with respect to closed subsets, Ind M is defined for every closed subspace  $M \subset X$ . Moreover, the following counterpart of Theorem 2.2.1 holds.

**3.1.4. Theorem.** For every closed subspace M of a normal space X we have  $\dim M \leq \dim X$ .

**Proof.** The theorem is obvious if dim  $X = \infty$ , so that we can assume that dim  $X = n < \infty$ . Consider a finite open cover  $\mathscr{U} \doteq \{U_i\}_{i=1}^k$  of the space M. For i = 1, 2, ..., k let  $W_i$  be an open subset of X such that  $U_i = M \cap W_i$ . The family  $\{X \setminus M\} \cup \{W_i\}_{i=1}^k$  is an open cover of the space X and, since dim  $X \leq n$ , it has a finite open refinement  $\mathscr{V}$  of order  $\leq n$ . One easily sees that the family  $\mathscr{V}|M$  is a finite open cover of the space M, refines  $\mathscr{U}$  and has order  $\leq n$ , so that dim  $M \leq n = \dim X$ .  $\Box$ 

From Theorem 1.6.11 it follows that the compact space Z and its normal subspace X defined in Example 2.2.11 satisfy the relation  $0 < \dim X$  $> \dim Z = 0$ . Hence, in Theorem 3.1.4 the assumption that M is a closed subspace of X cannot be replaced by the weaker assumption that dim M is defined. In Section 2.2 an even stronger example is cited which shows that the dimension dim is not monotonic in hereditarily normal spaces. Let us observe that the theorem on subspaces of intermediate dimensions, i.e., the counterpart of Theorems 1.5.1 and 2.2.2, does not hold for the dimension dim in normal spaces. More exactly, for every natural number  $n \ge 0$  there exists a compact space  $X_n$  such that dim $X_n = n$  and for every closed subspace  $M \subset X_n$  we have either dim  $M \le 0$  or dim M = n. The description of spaces  $X_n$  and verification of their properties are too difficult to be reproduced in this book.

We shall now show that for a fixed hereditarily normal space X the monotonicity of the dimension dim is equivalent to its being monotonic with respect to open subspaces.

**3.1.5. Proposition.** For every hereditarily normal space X the following conditions are equivalent:

(a) For every subspace M of X we have  $\dim M \leq \dim X$ .

(b) For every open subspace U of X we have  $\dim U \leq \dim X$ .

**Proof.** The implication (a)  $\Rightarrow$  (b) is obvious. Suppose that X satisfies (b). Condition (a) is satisfied if dim  $X = \infty$ , so that we can assume that dim  $X = n < \infty$ . Consider a subspace M of X and a finite open cover  $\mathscr{U} = \{U_i\}_{i=1}^k$  of the space M. For i = 1, 2, ..., k let  $W_i$  be an open subset of X such that  $U_i = M \cap W_i$ . Since for the open subspace  $W = \bigcup_{i=1}^k W_i$  of X we have dim  $W \leq n$ , the cover  $\{W_i\}_{i=1}^k$  of the space W has a finite open refinement  $\mathscr{V}$  of order  $\leq n$ . One easily sees that the family  $\mathscr{V}|M$  is a finite open cover of the space M, refines  $\mathscr{U}$  and has order  $\leq n$ , so that dim  $M \leq n = \dim X$ . Thus X satisfies condition (a).  $\square$ 

We shall return to the question of the monotonicity of dim later in this section and we shall show that the dimension dim is monotonic in the class of strongly hereditarily normal spaces; the proof of this important fact depends on the countable sum theorem for dim.

We now turn to a study of sum theorems for the covering dimension.

**3.1.6. Lemma.** If a normal space X can be represented as the union of a sequence  $K_1, K_2, \ldots$  of subspaces such that dim  $Z \leq n$  for every closed subspace Z of the space X contained in a set  $K_i$  and the union  $\bigcup_{j \leq i} K_j$  is closed for  $i = 1, 2, \ldots$ , then dim  $X \leq n$ .

**Proof.** Consider a finite open cover  $\mathscr{U} = \{U_j\}_{j=1}^k$  of the space X. We shall define inductively a sequence  $\mathscr{U}_0, \mathscr{U}_1, \ldots$  of open covers of the space X, where  $\mathscr{U}_i = \{U_{i,j}\}_{j=1}^k$ , satisfying the conditions:

(1)  $\overline{U}_{i,j} \subset U_{i-1,j}$  if  $i \ge 1$  and  $U_{0,j} \subset U_j$  for j = 1, 2, ..., k.

(2) ord
$$(\{F_i \cap U_{i,j}\}_{j=1}^k) \leq n$$
, where  $F_i = K_1 \cup K_2 \cup \ldots \cup K_i$ 

if 
$$i \ge 1$$
 and  $F_0 = \emptyset$ .

Both conditions are satisfied for i = 0 if we define  $U_{0,j} = U_j$  for j = 1, 2, ..., k. Assume that the coverings  $\mathscr{U}_i$  satisfying (1) and (2) are defined for all  $i < m \ge 1$ . Consider the set  $A = \bigcup_{T \in \mathscr{T}} \bigcap_{j \in T} \overline{U}_{m-1,j}$ , where  $\mathscr{T}$  is the family of all subsets of the set 1, 2, ..., k which have exactly n+2 elements. From (2) with i = m-1 it follows that  $A \cap F_{m-1} = \emptyset$ . The intersection  $Z = A \cap F_m$  is a closed subspace of the space X and is contained in  $F_m \setminus F_{m-1} \subset K_m$ , so that dim  $Z \le n$ . By virtue of Proposition 1.6.9 the cover  $\{Z \cap U_{m-1,j}\}_{j=1}^k$  of the space Z has an open shrinking  $\{V_j\}_{j=1}^k$ , where  $W_j = (U_{m-1,j} \setminus Z) \cup V_j \subset U_{m-1,j}$ , is an open cover of the space X and that ord $(\{F_m \cap W_j\}_{j=1}^k) \le n$ . By Theorem 1.7.8 and the normality of X there exists an open shrinking  $\mathscr{U}_m = \{U_{m,j}\}_{j=1}^k$  of the cover  $\{W_j\}_{j=1}^k$  such that  $\overline{U}_{m,j} \subset W_j$  for j = 1, 2, ..., k. Clearly, the cover  $\mathscr{U}_m$  satisfies conditions (1) and (2) with i = m. Hence the construction of the covers  $\mathscr{U}_i$  satisfying (1) and (2) for i = 0, 1, ... is completed.

For every point  $x \in X$  there exists an  $j(x) \leq k$  such that x belongs to infinitely many sets  $U_{i, j(x)}$ ; it follows from (1) that  $x \in \bigcap_{i=1}^{\infty} U_{i, j(x)}$ . Applying (1) and (2) we readily see that the family  $\{\bigcap_{i=1}^{\infty} \overline{U}_{i, j}\}_{j=1}^{k}$  is a closed shrinking of the cover  $\{U_j\}_{j=1}^{k}$  and has order  $\leq n$ . Therefore we have dim  $X \leq n$  by virtue of Proposition 3.1.3.  $\Box$ 

From Theorem 3.1.4 and Lemma 3.1.6 we obtain

**3.1.7. Proposition.** If a normal space X can be represented as the union of a sequence  $K_1, K_2, ...$  of normal subspaces such that  $\dim K_i \leq n$  and the union  $\bigcup_{j \leq i} K_j$  is closed for i = 1, 2, ..., then  $\dim X \leq n$ .  $\Box$ 

The last proposition yields

**3.1.8. The countable sum theorem for** dim. If a normal space X can be represented as the union of a sequence  $F_1, F_2, ...$  of closed subspaces such that dim  $F_i \leq n$  for i = 1, 2, ..., then dim  $X \leq n$ .  $\Box$ 

We now pass to the locally finite sum theorem for dim. The theorem will be deduced from a lemma which is formulated here in a more general form in view of an application in the following section.

**3.1.9. Lemma.** Let  $\mathcal{U} = \{U_s\}_{s\in S}$  be an open cover of a normal space X. If the space X has a locally finite closed cover  $\mathcal{F}$  each member of which has covering dimension not larger than n and meets only finitely many sets  $U_s$ , then the cover  $\mathcal{U}$  has an open shrinking of order  $\leq n$ .

**Proof.** Let us adjoin the set  $F_0 = \emptyset$  to the cover  $\mathscr{F}$  and let us arrange the members of this cover into a transfinite sequence  $F_0, F_1, \ldots, F_{\alpha}, \ldots$ ,  $\alpha \leq \xi$  of type  $\xi+1$ . We shall define inductively a transfinite sequence  $\mathscr{U}_0, \mathscr{U}_1, \ldots, \mathscr{U}_{\alpha}, \ldots, \alpha \leq \xi$  of open covers of the space X, where  $\mathscr{U}_{\alpha} = \{U_{\alpha,s}\}_{s\in S}$ , satisfying the conditions:

(3) 
$$U_{\alpha,s} \subset U_{\beta,s}$$
 if  $\alpha > \beta \ge 0$  and  $U_{0,s} \subset U_s$  for  $s \in S$ .

(4) 
$$\operatorname{ord}(\{F_{\alpha} \cap U_{\alpha,s}\}_{s \in S}) \leq n.$$

(5) 
$$U_{\beta,s} \setminus U_{\alpha,s} \subset \bigcup_{\beta \leq \gamma \leq \alpha} F_{\gamma}$$
 for  $\beta < \alpha$  and  $s \in S$ .

All conditions are satisfied for  $\alpha = 0$  if we define  $U_{0,s} = U_s$  for  $s \in S$ . Assume that the coverings  $\mathscr{U}_{\alpha}$  satisfying (3)–(5) are defined for all  $\alpha < \alpha_0 \ge 1$ . To begin with, we shall show that the family  $\mathscr{U}'_{\alpha_0} = \{U'_{\alpha_0,s}\}_{s\in S}$ , where

$$U'_{\alpha_0,s} = \bigcap_{\alpha < \alpha_0} U_{\alpha,s} \quad \text{for } s \in S,$$

is an open cover of the space X. This is clear if  $\alpha_0 = \alpha_1 + 1$ , because then  $U'_{\alpha_0} = U_{\alpha_1}$ ; thus, we can assume that  $\alpha_0$  is a limit number.

Consider an arbitrary point  $x \in X$ . Since the family  $\mathscr{F}$  is locally finite, there exist a neighbourhood  $U \subset X$  of the point x and an ordinal number  $\beta < \alpha_0$  such that  $U \cap F_{\gamma} = \emptyset$  whenever  $\beta \leq \gamma < \alpha_0$ . The family  $\mathscr{U}_{\beta}$  being a cover of X, there exists an  $s \in S$  such that  $x \in U_{\beta,s}$ . It follows from (5) that  $x \in U_{\alpha,s}$  whenever  $\beta < \alpha < \alpha_0$ , so that  $x \in U'_{\alpha_0,s}$ . Hence  $\mathscr{U}'_{\alpha_0}$  is a cover of the space X.

To show that the sets  $U'_{\alpha_0,s}$  are open it suffices to consider, for an arbitrary point  $x \in U'_{\alpha_0,s}$ , a neighbourhood  $U \subset X$  of the point x and an ordinal number  $\beta < \alpha_0$  such that  $U \cap F_{\gamma} = \emptyset$  whenever  $\beta \leq \gamma < \alpha_0$ ,
213

to choose a set  $U_{\beta,s}$  which contains x, and to observe that  $x \in U \cap U_{\beta,s}$  $\subset U'_{\alpha_0,s}$  by virtue of (5).

The open cover  $\{F_{\alpha_0} \cap U'_{\alpha_0,s}\}_{s \in S}$  of the subspace  $F_{\alpha_0} \subset X$  has an open shrinking  $\{V_s\}_{s \in S}$  of order  $\leq n$ , because—by (3) and the assumption on the family  $\mathscr{F}$ —only finitely many members of that cover are non-empty. One readily observes that the family  $\mathscr{U}_{\alpha_0} = \{U_{\alpha_0,s}\}_{s \in S}$ , where  $U_{\alpha_0,s}$  $= (U'_{\alpha_0,s} \setminus F_{\alpha_0}) \cup V_s$ , is an open cover of the space X satisfying conditions (3)-(5) with  $\alpha = \alpha_0$ . Hence the construction of the covers  $\mathscr{U}_{\alpha}$  satisfying (3)-(5) for  $\alpha \leq \xi$  is completed.

Now, it follows from (4) that  $\operatorname{ord} \mathscr{U}_{\xi} \leq n$ ; as  $\mathscr{U}_{\xi}$  is, by virtue of (3), a shrinking of  $\mathscr{U}$ , the lemma is established.  $\Box$ 

Lemma 3.1.9 yields

**3.1.10. The locally finite sum theorem for** dim. If a normal space X can be represented as the union of a locally finite family  $\{F_s\}_{s\in S}$  of closed subspaces such that dim  $F_s \leq n$  for  $s \in S$ , then dim  $X \leq n$ .  $\Box$ 

The following two theorems are common generalizations of the countable and the locally finite sum theorems. The proofs, parallel to the proofs of Theorems 2.3.11 and 2.3.12, are left to the reader.

**3.1.11. Theorem.** If a normal space X can be represented as the union of a  $\sigma$ -locally finite family  $\{F_s\}_{s\in S}$  of closed subspaces such that dim  $F_s \leq n$  for  $s \in S$ , then dim  $X \leq n$ .  $\Box$ 

**3.1.12. Theorem.** If a normal space X can be represented as the union of a transfinite sequence  $F_1, F_2, ..., F_{\alpha}, ..., \alpha < \xi$  of closed subspaces such that dim  $F_{\alpha} \leq n$  and the family  $\{F_{\beta}\}_{\beta < \alpha}$  is locally finite for  $\alpha < \xi$ , and the family  $\{F_{\alpha}\}_{\alpha < \xi}$  is locally countable, then dim  $X \leq n$ .  $\Box$ 

The next result is the point-finite sum theorem for dim. It will yield two further sum theorems which hold in the class of weakly paracompact normal spaces. Let us note that in this last class Theorem 3.1.13. generalizes the locally finite sum theorem (see Problem 2.3.F) and Theorem 3.1.15 generalizes both the countable and the locally finite sum theorems.

**3.1.13. The point-finite sum theorem for** dim. If a normal space X can be represented as the union of a family  $\{F_s\}_{s\in S}$  of closed subspaces such that dim  $F_s \leq n$  for  $s \in S$ , and if there exists a point-finite open cover  $\{U_s\}_{s\in S}$  of the space X such that  $F_s \subset U_s$  for  $s \in S$ , then dim  $X \leq n$ .

**Proof.** Consider the decomposition of the space X described in Lemma 2.3.3, i.e., let

(6) 
$$X = \bigcup_{i=1}^{\infty} K_i$$
, where the union  $\bigcup_{j \le i} K_j$  is closed for  $i = 1, 2, ...$ 

and

(7) 
$$K_i = \bigcup_{T \in \mathcal{F}_i} K_T$$
, where the sets  $K_T = K_i \cap \bigcap_{s \in T} U_s$  are open in  $K_i$   
and pairwise disjoint

Let Z be a closed subspace of the space X contained in a set  $K_i$ . It follows from (7) that for every  $T \in \mathcal{F}_i$  the set  $Z \cap K_T$  is closed in X. Since  $Z \cap K_T \subset \bigcup_{s \in T} F_s$ , by virtue of Theorems 3.1.8 and 3.1.4 we have dim $(Z \cap K_T)$  $\leq n$  for every  $T \in \mathcal{F}_i$ ; and this implies that dim $Z \leq n$ . In view of (6), to conclude the proof it suffices to apply Lemma 3.1.6.  $\Box$ 

**3.1.14. Theorem.** If a weakly paracompact normal space X can be represented as the union of a family  $\{U_s\}_{s\in S}$  of open subspaces such that  $\dim \overline{U_s} \leq n$  for  $s \in S$ , then  $\dim X \leq n$ .

**Proof.** The space X being weakly paracompact, one can assume that the cover  $\{U_s\}_{s\in S}$  is point-finite and thus has a closed shrinking  $\{F_s\}_{s\in S}$  (see [GT], Theorem 1.5.18). To complete the proof it suffices to apply Theorems 3.1.4 and 3.1.13.  $\Box$ 

A variant of the last theorem which closely parallels Theorem 2.3.14 is given in Problem 3.1.D(a).

**3.1.15. Theorem.** If a weakly paracompact normal space X can be represented as the union of a locally countable family  $\{F_s\}_{s\in S}$  of closed subspaces such that dim  $F_s \leq n$  for  $s \in S$ , then dim  $X \leq n$ .  $\Box$ 

**Proof.** For every point  $x \in X$  there exist a neighbourhood  $U_x$  and a countable set  $S(x) \subset S$  such that  $\overline{U}_x \cap F_s = \emptyset$  for  $s \in S \setminus S(x)$ . From this relation it follows that  $\overline{U}_x \subset \bigcup \{F_s: s \in S(x)\}$ , so that by virtue of Theorems 3.1.4 and 3.1.8 we have dim  $\overline{U}_x \leq n$  for  $x \in X$ . To complete the proof it suffices to apply Theorem 3.1.14.  $\square$ 

We now pass to the addition theorems for dim.

**3.1.16. Proposition.** If a normal space X can be represented as the union of two subspaces A and B such that A is normal,  $\dim A \leq n$ , and  $\dim Z \leq m$  for every closed subspace Z of the space X contained in B, then  $\dim X \leq n + +m+1$ .

**Proof.** By virtue of Lemma 3.1.6 it suffices to show that  $\dim \overline{A} \leq n+m+1$ . Consider a finite open cover  $\mathscr{U} = \{U_i\}_{i=1}^k$  of the space  $\overline{A}$ . Since  $\dim A \leq n$ , there exists a family  $\mathscr{V} = \{V_i\}_{i=1}^k$  of open subsets of  $\overline{A}$  such that

$$A \cap V_i \subset A \cap U_i \quad \text{for } i = 1, 2, ..., k, \quad A \subset \bigcup_{i=1}^n V_i$$
  
and 
$$\operatorname{ord}(\{A \cap V_i\}_{i=1}^k) \leq n;$$

without loss of generality one can assume that  $V_i \subset U_i$  for i = 1, 2, ..., k. The set  $Z = \overline{A} \setminus \bigcup_{i=1}^k V_i$  is closed in X and contained in B, so that dim  $Z \leq m$ , which implies that there exists a closed cover  $\{F_i\}_{i=1}^k$  of Z such that  $F_i \subset Z \cap U_i$  for i = 1, 2, ..., k and  $\operatorname{ord}(\{F_i\}_{i=1}^k) \leq m$ . By virtue of Theorem 3.1.2 the family  $\{F_i\}_{i=1}^k$  of closed subsets of  $\overline{A}$  has an open swelling  $\mathcal{W} = \{W_i\}_{i=1}^k$  in the space  $\overline{A}$  such that  $W_i \subset U_i$  for i = 1, 2, ..., k. Thus

$$Z = \overline{A} \setminus \bigcup_{i=1}^{k} V_i \subset \bigcup_{i=1}^{k} W_i \quad \text{and} \quad \operatorname{ord}(\{W_i\}_{i=1}^k) \leq m.$$

The union  $\mathscr{V} \cup \mathscr{W}$  covers the space  $\overline{A}$  and refines the cover  $\mathscr{U}$ . As  $\operatorname{ord}(\mathscr{V} \cup \mathscr{W}) \leq n+m+1$ ,  $\dim \overline{A} \leq n+m+1$ .  $\square$ 

Theorem 3.1.16 implies

**3.1.17.** The addition theorem for dim. For every pair X, Y of subspaces of a hereditarily normal space we have

 $\dim(X \cup Y) \leq \dim X + \dim Y + 1. \square$ 

The addition theorem yields

**3.1.18. Corollary.** If a hereditarily normal space X can be represented as the union of n+1 subspaces  $Z_1, Z_2, ..., Z_{n+1}$  such that dim  $Z_i \leq 0$  for i = 1, 2, ..., n+1, then dim  $X \leq n$ .  $\Box$ 

Let us note that the implication in the last corollary cannot be reversed even in the class of compact perfectly normal spaces. Indeed, applying the continuum hypothesis one can define a compact perfectly normal space X such that  $\dim X = 1$  and  $\operatorname{ind} X = \operatorname{Ind} X = 2$ ; now, if the space X could be represented as the union  $Z_1 \cup Z_2$ , where dim  $Z_1 = \dim Z_2 = 0$ , then by virtue of Theorems 1.6.11 and 2.2.5 we would have  $\operatorname{Ind} X = 1$ . The description of the space X is too complicated to be discussed in this book.

Let us return to the question of the monotonicity of dim. We are now able to prove

**3.1.19. The subspace theorem for** dim. For every subspace M of a strongly hereditarily normal space X we have dim  $M \leq \dim X$ .

**Proof.** We can assume that  $\dim X = n < \infty$ . To begin with, let us observe that from Theorems 3.1.13, 3.1.8 and 3.1.4 it follows that  $\dim U \le n$  for every open subspace U of X which can be represented as the union of a point-finite family of open  $F_{\sigma}$ -sets in X.

Now, consider an arbitrary subspace M of X and a finite open cover  $\{U_i\}_{i=1}^k$  of the space M; let  $\{F_i\}_{i=1}^k$  be a closed shrinking of this last cover. For i = 1, 2, ..., k the sets  $F_i$  and  $M \setminus U_i$  are separated in X, so that there exists an open set  $V_i \subset X$  such that  $F_i \subset V_i$ ,  $M \cap V_i \subset U_i$ , and  $V_i$  can be represented as the union of a point-finite family of open  $F_{\sigma}$ -sets in X. By the above observation the set  $V = \bigcup_{i=1}^k V_i$  satisfies the inequality dim  $V \leq n$ ; therefore there exists an open shrinking  $\{W_i\}_{i=1}^k$  of the cover  $\{V_i\}_{i=1}^k$  of the space V such that  $\operatorname{ord}(\{W_i\}_{i=1}^k) \leq n$ . Since  $M = \bigcup_{i=1}^k F_i \subset \bigcup_{i=1}^k V_i = V$  and  $M \cap V_i \subset U_i$ , the family  $\{M \cap W_i\}_{i=1}^k$  is an open shrinking of the cover  $\{U_i\}_{i=1}^k$ . As  $\operatorname{ord}(\{M \cap W_i\}_{i=1}^k) \leq n$ , dim  $M \leq n$ .  $\Box$ 

The last theorem together with Theorems 2.1.3 and 2.1.5 yield the following two corollaries, which can also be deduced directly from Proposition 3.1.5 and Theorem 3.1.8 and 3.1.14.

**3.1.20. Corollary.** For every subspace M of a perfectly normal space X we have dim  $M \leq \dim X$ .  $\Box$ 

**3.1.21. Corollary.** For every subspace M of a hereditarily weakly paracompact hereditarily normal space X we have dim  $M \leq \dim X$ .  $\Box$ 

We shall now establish a theorem on the monotonicity of dim which slightly differs in nature from our previous theorems of this type: here,

216

the assumptions are about the internal topological properties of the subspace M rather than the position of the subspace M in the space X.

**3.1.22. Lemma.** Every subspace M of a normal space X which is an  $F_{\sigma}$ -set in X is normal.

**Proof.** It suffices to show (see Problem 2.1.E(a) or [GT], Lemma 1.5.14) that for every closed set  $F \subset M$  and each open set  $W \subset M$  there exists a sequence  $W_1, W_2, ...$  of open subsets of M such that  $F \subset \bigcup_{i=1}^{\infty} W_i$  and  $M \cap \overline{W}_i \subset W$  for i = 1, 2, ..., where the bar denotes the closure operator in X. Let  $F_1, F_2, ...$  be closed subsets of X such that  $M = \bigcup_{i=1}^{\infty} F_i$  and U an open subset of X such that  $W = M \cap U$ . The space X being normal, for i = 1, 2, ... there exists an open set  $U_i \subset X$  such that  $F \cap F_i \subset U_i \subset \overline{U}_i \subset U$ . The sets  $W_i = M \cap U_i$  satisfy all the required conditions.  $\Box$ 

**3.1.23. Theorem.** For every strongly paracompact subspace M of a normal space X we have dim  $M \leq \dim X$ .

**Proof.** We can assume that  $\dim X = n < \infty$ . Consider a finite open cover  $\{U_i\}_{i=1}^k$  of the space *M*. For i = 1, 2, ..., k let  $W_i$  be an open subset of X such that  $U_i = M \cap W_i$  and let  $W = \bigcup_{i=1}^{n} W_i$ . For every point  $x \in M$ choose a neighbourhood  $V_x$  of the point x in X such that  $x \in V_x \subset \overline{V_x} \subset W$ . The open cover  $\{M \cap V_x\}_{x \in M}$  of the space M has a star-finite open refinement  $\mathscr{V}$ . Consider the decomposition  $\{\mathscr{V}_s\}_{s\in S}$  of the cover  $\mathscr{V}$  into components; as was established in Section 2.4, the components  $\mathscr{V}_s$  are countable and the sets  $V_s = \bigcup \mathscr{V}_s$  are pairwise disjoint. Let  $F_s$  be the union of the closures in X of all sets which belong to the family  $\mathscr{V}_s$ . Clearly  $V_s \subset F_s$  $\subset W$ , and by virtue of Lemma 3.1.22 the subspace  $F_s$  of X is normal; from the countable sum theorem it follows that dim  $F_s \leq n$ . Thus, for every  $s \in S$  there exists an open cover  $\{W_{s,i}\}_{i=1}^k$  of the space  $F_s$  such that  $W_{s,i} \subset W_i$  for i = 1, 2, ..., k and  $\operatorname{ord}(\{W_{s,i}\}_{i=1}^k) \leq n$ . One readily sees that the family  $\{V_i\}_{i=1}^k$ , where  $V_i = M \cap \bigcup_{s \in S} (V_s \cap W_{s,i})$ , is an open shrinking of the cover  $\{U_i\}_{i=1}^k$  of the space M. As  $\operatorname{ord}(\{V_i\}_{i=1}^k) \leq n$ ,  $\dim M \leq n$  $= \dim X.$ 

The covering dimension

Applying Theorem 3.1.14 we can strengthen the last theorem as follows (we recall that a space X is *locally strongly paracompact* if every point of X has a neighbourhood whose closure is strongly paracompact).

**3.1.24. Theorem.** For every weakly paracompact locally strongly paracompact normal subspace M of a normal space X we have dim  $M \leq \dim X$ .  $\Box$ 

Let us observe that in Theorem 3.1.23 the assumption that X is strongly paracompact cannot be replaced by the weaker assumption that X is paracompact (cf. Problem 3.2.G(b)). Indeed, by virtue of Remark 1.3.18, Roy's space X, cited in Section 2.4, which satisfies  $\operatorname{ind} X = 0$  and  $\operatorname{dim} X = 1$ , is embeddable in a Cantor cube  $D^{\mathrm{m}}$ ; since  $\operatorname{dim} D^{\mathrm{m}} = 0$  (see Theorems 1.6.5 and 1.6.11), we have  $\operatorname{dim} X > \operatorname{dim} D^{\mathrm{m}}$ .

We shall now show that the Čech-Stone compactification preserve the dimension dim. In Section 3.3 it will be proved that for every normal space there exist compactifications preserving both the dimension dim and the weight.

## **3.1.25. Theorem.** For every normal space X we have $\dim \beta X = \dim X$ .

**Proof.** To begin with, we shall prove that  $\dim X \leq \dim \beta X$ . The inequality is obvious if  $\dim \beta X = \infty$ , so that we can suppose that  $\dim \beta X = n < \infty$ . Consider a finite open cover  $\{U_i\}_{i=1}^k$  of the space X. By virtue of Theorem 1.7.8 there exists a closed shrinking  $\{F_i\}_{i=1}^k$  of the cover  $\{U_i\}_{i=1}^k$ , and by Urysohn's lemma for i = 1, 2, ..., k there exists a continuous function  $f_i: X \to I$  such that

(8) 
$$f_i(X \setminus U_i) \subset \{0\} \text{ and } f_i(F_i) \subset \{1\};$$

let  $\overline{f_i}$ :  $\beta X \to I$  be the continuous extension of  $f_i$  over  $\beta X$ . By virtue of (8) the family  $\{W_i\}_{i=1}^k$ , where  $W_i = \overline{f_i}^{-1}((0, 1])$ , is an open cover of the space  $\beta X$  and

(9) 
$$X \cap W_i \subset U_i \quad \text{for } i = 1, 2, ..., k.$$

Since dim $\beta X \leq n$ , the cover  $\{W_i\}_{i=1}^k$  has an open shrinking  $\{V_i\}_{i=1}^k$  of order  $\leq n$ . From (9) it follows that the family  $\{X \cap V_i\}_{i=1}^k$  is an open shrinking of the cover  $\{U_i\}_{i=1}^k$  of the space X. As  $\operatorname{ord}(\{X \cap V_i\}_{i=1}^k) \leq n$ , dim $X \leq n$ .

Now, we shall prove that  $\dim \beta X \leq \dim X$ . As in the first part of the proof, we shall suppose that  $\dim X = n < \infty$ . Consider a finite open cover

 $\{U_i\}_{i=1}^k$  of the space  $\beta X$ . By virtue of Theorem 1.7.8 there exists an open shrinking  $\{W_i\}_{i=1}^k$  of the cover  $\{U_i\}_{i=1}^k$  such that  $\overline{W}_i \subset U_i$  for i = 1, 2, ..., k. Since dim X = n, the cover  $\{X \cap W_i\}_{i=1}^k$  of the space X has an open shrinking  $\{V_i\}_{i=1}^k$  of order  $\leq n$  which, in turn, has a closed shrinking  $\{E_i\}_{i=1}^k$ . For i = 1, 2, ..., k let  $F_i = \overline{E}_i$ , where the bar denotes the closure operator in  $\beta X$ . The family  $(F_i)_{i=1}^k$  is a cover of the space  $\beta X$ ; since  $F_i = \overline{E}_i \subset \overline{V}_i \subset \overline{X \cap W_i} = \overline{W}_i \subset U_i$  for i = 1, 2, ..., i it is a closed shrinking of the cover  $\{U_i\}_{i=1}^k$ . To complete the proof it suffices to show that  $\operatorname{ord}(\{F_i\}_{i=1}^k) \leq n$ .

For i = 1, 2, ..., k define a continuous function  $f_i: X \to I$  such that  $f_i(E_i) \subset \{0\}$  and  $f_i(X \setminus V_i) \subset \{1\}$ ; let  $\overline{f_i}: \beta X \to I$  be the continuous extension of  $f_i$  over  $\beta X$ . Consider an arbitrary subfamily  $\{F_{i_1}, F_{i_2}, ..., F_{i_m}\}$  of the cover  $\{F_i\}_{i=1}^k$  such that  $F_{i_1} \cap F_{i_2} \cap ... \cap F_{i_m} \neq \emptyset$ . Let  $\overline{f} = \max(\overline{f_{i_1}}, \overline{f_{i_2}}, ..., \overline{f_{i_m}})$ . The set  $U = \overline{f^{-1}}(([0, 1]))$  is open in  $\beta X$ , and since  $F_{i_1} \cap F_{i_2} \cap ... \cap F_{i_m} \subset U$ , we have  $U \neq \emptyset$ , which implies that  $U \cap X \neq \emptyset$ . One readily checks that  $U \cap X \subset V_{i_1} \cap V_{i_2} \cap ... \cap V_{i_m}$ ; thus the relation  $\operatorname{ord}(\{V_i\}_{i=1}^k) \leq n$ .  $\Box$ 

**3.1.26. Corollary.** For every normal space X and a dense normal subspace  $M \subset X$  which has the property that every continuous function  $f: M \to I$  is continuously extendable over X we have dim  $M = \dim X$ .

In other words,  $\dim Y = \dim X$  for every normal space X and every normal subspace Y of  $\beta X$  which contains X.

**Proof.** From the extendability of every continuous function  $f: M \to I$  over X it follows that  $\beta M = \beta X$ .  $\Box$ 

The final part of this section is devoted to a study of relations between ind, Ind and dim.

Let us begin with reminding the reader that in Section 1.7, when proving that the equality  $\operatorname{ind} X = \dim X$  holds for every separable metric space X, we ascertained that the proof of the inequality  $\dim X \leq \operatorname{ind} X$  was much easier than the proof of the reverse inequality. We shall now show that the inequality  $\dim X \leq \operatorname{ind} X$  holds for every strongly paracompact space X and that the related inequality  $\dim X \leq \operatorname{Ind} X$  holds for every normal space X. Both results will be deduced from a common lemma.

**3.1.27. Lemma.** If for every pair A, B of disjoint closed subsets of a normal space X there exists a partition L between A and B such that dim  $L \leq n-1$ , then dim  $X \leq n$ .

[Ch. 3, § 1

**Proof.** Consider a finite open cover  $\{U_i\}_{i=1}^k$  of the space X; let  $\{F_i\}_{i=1}^k$  be a closed shrinking of  $\{U_i\}_{i=1}^k$ . For i = 1, 2, ..., k there exists a partition  $L_i$  between  $F_i$  and  $X \setminus U_i$  such that dim  $L_i \leq n-1$ ; let  $W_i$  be an open set satisfying  $F_i \subset W_i \subset U_i$  and  $\operatorname{Fr} W_i \subset L_i$ . Since  $L = \bigcup_{i=1}^k L_i$  is a normal space, dim  $L \leq n-1$  by virtue of Theorem 3.1.8. By shrinking the open cover  $\{L \cap U_i\}_{i=1}^k$  of the space L to a closed cover of order  $\leq n$  and then swelling the latter cover in conformity with Theorem 3.1.2, we obtain a family  $\{V_i\}_{i=1}^k$  of open subsets of X such that  $\overline{V_i} \subset U_i$  for i = 1, 2, ..., k,

(10) 
$$L \subset V = \bigcup_{i=1}^{k} V_i$$
 and  $\operatorname{ord}(\{\overline{V}_i\}_{i=1}^k) \leq n-1$ .

The sets  $\overline{V}_1, \overline{V}_2, \dots, \overline{V}_k$  together with the sets  $Z_1, Z_2, \dots, Z_k$ , where

$$Z_i = \overline{W}_i \setminus (V \cup \bigcup_{j < i} W_j),$$

constitute a closed refinement of  $\{U_i\}_{i=1}^k$ . To complete the proof it suffices to observe that this refinement has order  $\leq n$ , which, however, follows immediately from the second part of (10) and the fact that for  $j < i \leq k$  we have

$$Z_j \cap Z_i \subset \overline{W}_j \cap [\overline{W}_i \setminus (V \cup W_j)] \subset (X \setminus V) \cap \operatorname{Fr} W_j = \emptyset. \square$$

From the last lemma, by applying induction with respect to Ind X, we obtain

**3.1.28. Theorem.** For every normal space X we have  $\dim X \leq \operatorname{Ind} X$ .

In the next chapter we shall show that for every metrizable space X we have the equality Ind X = dim X. From example 3.1.31 below it follows that the equality does not hold in compact spaces. The commentary to Corollary 3.1.18 above shows that it does not hold in perfectly normal compact spaces either.

From Lemmas 3.1.27, 2.4.1 and Theorem 3.1.8, by applying induction with respect to ind X, we obtain the following

**3.1.29. Theorem.** For every strongly paracompact space X we have dim  $X \leq ind X$ .  $\Box$ 

As shown by Roy's space cited above, in the last theorem the assumption that X is strongly paracompact cannot be replaced by the weaker

assumption that X is paracompact (cf. Problem 3.1.F). The same example shows that the equality ind X = dim X does not hold in metric spaces. That is does not hold in compact spaces either is shown in Example 3.1.31.

To conclude, let us observe that Theorems 2.4.2 and 1.6.11 yield the following

**3.1.30. Theorem.** For every strongly paracompact space X the conditions  $\operatorname{ind} X = 0$ ,  $\operatorname{Ind} X = 0$  and  $\dim X = 0$  are equivalent.  $\Box$ 

**3.1.31. Example.** In Example 2.2.13 we described a compact space X with  $\operatorname{Ind} X \ge \operatorname{ind} X \ge 2$  which contains closed subspaces  $F_1, F_2$  such that  $F_1 \cup F_2 = X$  and  $\operatorname{Ind} F_1 = \operatorname{Ind} F_2 = 1$ . From Theorems 3.1.28, 3.1.8 and 1.6,11 it follows that  $\dim X = 1$ . Thus X is a compact space such that  $\dim X \ne \operatorname{Ind} X$  and  $\dim X \ne \operatorname{ind} X$ .  $\Box$ 

### Historical and bibliographic notes

Proposition 3.1.3 and Theorem 3.1.4 were given by Čech in [1933]. It was proved by Fedorčuk in [1973] that for every natural number  $n \ge 2$ there exists a compact space  $X_n$  such that  $\dim X_n = n$  and for every closed subspace  $M \subset X_n$  we have either dim  $M \leq 0$  or dim M = n. Theorem 3.1.5 and Lemma 3.1.6 in the case where  $X = K_1 \cup K_2$  (cf. Problem 3.1.B (a)) were obtained by Dowker in [1955]. Theorem 3.1.8 was established by Čech in [1933], and Theorem 3.1.10 independently by Morita in [1950a] and Katětov in [1952]; Lemma 3.1.9 appeared in Katětov's paper [1952]. Theorem 3.1.14 was proved by Dowker in [1955] and by Nagami in [1955] for paracompact spaces; Theorem 3.1.15, also for paracompact spaces, was given by Nagami in [1955]. Proposition 3.1.16 was established by Zarelua in [1963a]; its particular case formulated as Theorem 3.1.17 was obtained by Smirnov in [1951]. An example of a compact perfectly normal space X such that  $\dim X = 1$  and  $\operatorname{ind} X = \operatorname{Ind} X = 2$  was described under the assumption of the continuum hypothesis by Filippov in [1970a]; the first example of such a space was constructed under the joint assumption of the continuum hypothesis and the existence of a Souslin space (see [GT], Remark to Problem 2.7.9(f)) by Lifanov and Filippov in [1970]. Theorem 3.1.19 for totally normal spaces was proved by Dowker in [1955]; it was extended to Dowker spaces by Lifanov and Pasynkov in [1970] (announcement in Paşynkov [1967]) and to super normal spaces by Nishiura in [1977]. Corollary 3.1.20 was established by Čech in [1933]. Theorem 3.1.23 was proved by Morita in [1953]; its strengthening stated as Theorem 3.1.24 was given by Lifanov and Pasynkov in [1970] (an intermediate result was obtained by Pupko in [1961]). Theorem 3.1.25 was established by Wallman in [1938]. Theorem 3.1.28 was proved by Vedenissoff in [1941] and Theorem 3.1.29 by Morita in [1950a] (the latter theorem for Lindelöf spaces was obtained independently by Morita in [1950] and by Smirnov in [1951]; for compact spaces it was proved by Alexandroff in [1941]). Example 3.1.31 was given by Lokucievskiĭ in [1949]; the first example of a compact space X such that dim $X \neq indX$  was described by Lunc in [1949].

As observed by Katetov in [1950], the definition of the covering dimension dim can be slightly modified so as to lead to a notion of dimension which behaves relatively well in completely regular spaces. The modification consists in replacing condition ( $\check{C}L1$ ) by

(ČL1') dim  $X \le n$ , where n = -1, 0, 1, ..., if every finite functionally open cover of the space X has a finite functionally open refinement of order  $\le n$ .

Let us recall that a subset A of a topological space X is functionally open (functionally closed)<sup>1)</sup> if there exists a continuous function  $f: X \to I$  such that  $A = f^{-1}((0, 1])$  (such that  $A = f^{-1}(0)$ ); a family of subsets of a topological space is functionally open (closed) if all its members are functionally open (closed) sets. As noted in Problem 2.1.A, in normal spaces functionally open (closed) sets coincide with open  $F_{\sigma}$ -sets (closed  $G_{\delta}$ -sets). From Theorem 3.1.2 and Proposition 3.1.3 it easily follows that the modified definition of dim is equivalent to the original one in the realm of normal spaces. When applied to completely regular spaces, the modified definition yields the covering dimension for completely regular spaces, which is also denoted by dim. Some theorems proved in this section hold in the larger class of completely regular spaces for the covering dimension thus extended (see Problems 3.1.H, 3.1.I, and 3.2.H-3.2.K).

Occasionally, the covering dimension for completely regular spaces, or even for larger classes of spaces, was defined just by conditions ( $\check{C}L1$ )-( $\check{C}L3$ ) with no modifications. The dimension function obtained in this way satisfies the counterparts of a few theorems proved in this section

<sup>&</sup>lt;sup>1)</sup> The term *cozero-set* (zero-set) is also used.

#### Problems

(see Ostrand [1971]). It seems, however, that such a definition is not quite consistent with the geometrical conception of dimension. The notion obtained displays some undesirable features of a separation axiom; for example, one can easily check that every  $T_1$ -space whose dimension in the sense discussed equals zero is normal.

### Problems

**3.1.A** (Morita [1950a]). Prove that if for a family  $\{F_s\}_{s\in S}$  of closed subsets of a normal space X there exists a locally finite family  $\{V_s\}_{s\in S}$  of open subsets of X such that  $F_s \subset V_s$  for every  $s \in S$ , then the family  $\{F_s\}_{s\in S}$  has an open swelling  $\{U_s\}_{s\in S}$  such that  $\overline{U_s} \subset V_s$  for  $s \in S$  (cf. Problem 4.2.B(a)).

*Hint*. Apply transfinite induction; modify the proof of Theorem 3.1.2. *Remark*. For every locally finite family  $\{F_s\}_{s\in S}$  of closed subsets of a countably paracompact collectionwise normal space X there exists a locally finite family of open subsets of X such that  $F_s \subset V_s$  for every  $s \in S$  (see [GT], Problem 5.5.18(a)).

**3.1.B.** (a) Observe that Lemma 3.1.6 is a consequence of its special case, viz., the case where  $X = K_1 \cup K_2$ , and Theorem 3.1.8.

(b) Prove that if a normal space X can be represented as the union of a transfinite sequence  $K_1, K_2, ..., K_{\alpha}, ..., \alpha < \xi$  of subspaces such that dim  $Z \leq n$  for every closed subspace Z of the space X contained in a set  $K_{\alpha}$ , the union  $\bigcup_{\beta < \alpha} K_{\beta}$  is closed for  $\alpha < \xi$ , and the family  $\{K_{\alpha}\}_{\alpha < \xi}$  is locally finite, then dim  $X \leq n$ .

*Hint*. Apply transfinite induction; use Lemma 3.1.6 and Theorem 3.1.10.

(c) Prove that if a normal space X can be represented as the union of a transfinite sequence  $K_1, K_2, ..., K_{\alpha}, ..., \alpha < \xi$  of normal subspaces such that dim  $K_{\alpha} \leq n$  and the union  $\bigcup_{\beta < \alpha} K_{\beta}$  is closed for  $\alpha < \xi$ , and the family  $\{K_{\alpha}\}_{\alpha < \xi}$  is locally finite, then dim  $X \leq n$ .

**3.1.C.** (a) (Nagata [1965]) Prove that if a normal space X can be represented as the union of a transfinite sequence  $F_1, F_2, \ldots, F_{\alpha}, \ldots, \alpha < \xi$  of closed subspaces such that dim  $F_{\alpha} \leq n$  for  $\alpha < \xi$ , and if there exists

a transfinite sequence  $U_1, U_2, ..., U_{\alpha}, ..., \alpha < \xi$  of open subsets of X such that  $F_{\alpha} \subset U_{\alpha}$  and the family  $\{U_{\beta}\}_{\beta < \alpha}$  is locally finite for  $\alpha < \xi$ , then  $\dim X \leq n$ .

*Hint*. See the hint to Problem 2.3.E(a).

(b) Prove that if a normal space X can be represented as the union of a transfinite sequence  $K_1, K_2, ..., K_{\alpha}, ..., \alpha < \xi$  of subspaces such that dim  $Z \leq n$  for every closed subspace Z of the space X contained in a set  $K_{\alpha}$  and the union  $\bigcup_{\beta < \alpha} K_{\beta}$  is closed for  $\alpha < \xi$ , and if there exists a transfinite sequence  $U_1, U_2, ..., U_{\alpha}, ..., \alpha < \xi$  of open subsets of X such that  $K_{\alpha} \subset U_{\alpha}$  and the family  $\{U_{\beta}\}_{\beta < \alpha}$  is locally finite for  $\alpha < \xi$ , then dim  $X \leq n$ .

**3.1.D.** (a) Show that if a weakly paracompact normal space X can be represented as the union of a family  $\{U_s\}_{s\in S}$  of normal open subspaces such that dim  $U_s \leq n$  for  $s \in S$ , then dim  $X \leq n$ .

(b) Show, applying Problem 3.1.I(a) below, that the assumption of normality in (a), as well as in Proposition 3.1.7, can be omitted if by dim one understands the covering dimension for completely regular spaces as defined in the notes to this section.

3.1.E (Dowker [1955]). Note that the space X described in Example 2.2.11 satisfies the equality  $\dim X = 1$ .

3.1.F (Zarelua [1963] (announcement [1961])). Show that for every completely paracompact space X we have  $\dim X \leq \operatorname{ind} X$  (see Problem 2.4.B).

**3.1.G.** (a) Check that the union and the intersection of finitely many functionally open (closed) sets are functionally open (closed). Show that the union (intersection) of countably many functionally open (closed) sets is functionally open (closed). Prove that for every pair A, B of disjoint functionally closed subsets of a topological space X there exists a continuous function  $f: X \to I$  such that  $A \subset f^{-1}(0)$  and  $B \subset f^{-1}(1)$ .

(b) Prove that every finite functionally open cover  $\{U_i\}_{i=1}^k$  of a topological space X has shrinkings  $\{F_i\}_{i=1}^k$  and  $\{W_i\}_{i=1}^k$ , which are, respectively, functionally closed and functionally open, and such that  $F_i \subset W_i \subset \overline{W_i} \subset U_i$  for i = 1, 2, ..., k.

(c) Prove that every finite family  $\{F_i\}_{i=1}^k$  of functionally closed subsets of a topological space X has a functionally open swelling  $\{U_i\}_{i=1}^k$ . Observe

Ch. 3, § 1]

#### Problems

that if, moreover, a family  $\{V_i\}_{i=1}^k$  of functionally open subsets of X satisfying  $F_i \subset V_i$  for i = 1, 2, ..., k is given, then the swelling can be defined in such a way that  $\overline{U}_i \subset V_i$  for i = 1, 2, ..., k.

3.1.H. Show that for every completely regular space X the following conditions are equivalent (see the notes to this section):

- (i) The space X satisfies the inequality  $\dim X \leq n$ .
- (ii) Every finite functionally open cover of the space X has a functionally open shrinking of order  $\leq n$ .
- (iii) Every finite functionally open cover of the space X has a functionally closed shrinking of order  $\leq n$ .
- (iv) Every finite functionally open cover of the space X has a finite functionally closed refinement of order  $\leq n$ .

3.1.I (Katětov [1950]). (a) Show that if a subspace M of a completely regular space X has the property that every continuous function  $f: M \to I$  is continuously extendable over X, then dim  $M \leq \dim X$ .

(b) Prove that for every completely regular space X we have  $\dim \beta X = \dim X$ .

(c) Deduce from (b) that if a completely regular space X can be represented as the union of a sequence  $A_1, A_2, ...$  of subspaces such that dim  $A_i \leq n$  and every continuous function  $f: A_i \to I$  is continuously extendable over X for i = 1, 2, ..., then dim  $X \leq n$ .

Remark. Terasawa defined in [1977] a completely regular space X with  $\dim X > 0$  which can be represented as the union of a functionally closed subspace F with  $\dim F = 0$  and an open discrete subspace of cardinality  $\aleph_0$ . As shown by E. Pol in [1978] (announcement in [1976]), there exists a completely regular space X with  $\dim X > 0$  which can be represented as the union of two functionally closed subspaces  $F_1$  and  $F_2$  such that  $\dim F_1 = \dim F_2 = 0$ . It is an open problem whether every completely regular space X which can be represented as the union of a locally finite family  $\{A_s\}_{s\in S}$  of subspaces such that  $\dim A_s \leq n$  and every continuous function  $f: A_s \to I$  is continuously extendable over X for  $s \in S$  satisfies the inequality  $\dim X \leq n$ .

# 3.2. Characterizations of the dimension dim in normal spaces. Cartesian product theorems for the dimension dim

In this section several characterizations of the covering dimension in normal spaces are established. They split into two groups. The first consists of characterizations which are outwardly close to the definition of dim. The second is made up of generalizations of three important theorems proved in Chapter 1 for separable metric spaces, viz., of the theorems on partitions, on extending mappings to spheres, and on  $\mathscr{E}$ -mappings. In the final part of the section we apply one of the characterizations of the first group to obtain two theorems on the dimension dim of Cartesian products.

**3.2.1. Dowker's theorem.** For every normal space X the following conditions are equivalent:

- (a) The space X satisfies the inequality  $\dim X \leq n$ .
- (b) Every locally finite open cover of the space X has an open shrinking of order  $\leq n$ .
- (c) Every locally finite open cover of the space X has an open refinement of order  $\leq n$ .

**Proof.** We shall show first that (a)  $\Rightarrow$  (b). Consider a normal space X such that dim $X \leq n$ . Let  $\mathscr{U} = \{U_s\}_{s \in S}$  be an arbitrary locally finite open cover of the space X. Denote by  $\mathscr{T}$  the family of all non-empty finite subsets of S and for every  $T \in \mathscr{T}$  define

$$F_T = \bigcap_{s \in T} \overline{U}_s \cap \bigcap_{s \notin T} (X \setminus U_s);$$

by virtue of Theorem 3.1.4 dim  $F_T \leq n$ . The family  $\mathscr{F} = \{F_T\}_{T \in \mathscr{F}}$  is a closed cover of X each member of which meets only finitely many sets  $U_s$ . The cover  $\mathscr{U}$  being locally finite, for every point  $x \in X$  there exists a neighbourhood U and a finite set  $S_0 \subset S$  such that  $U \cap U_s = \emptyset$ , and consequently  $U \cap \overline{U}_s = \emptyset$ , for  $s \in S \setminus S_0$ . From the definition of the sets  $F_T$  it follows that if  $U \cap F_T \neq \emptyset$ , then  $T \subset S_0$ ; thus the cover  $\mathscr{F}$  is locally finite. By virtue of Lemma 3.1.9 the cover  $\mathscr{U}$  has an open shrinking of order  $\leq n$ , so that the space X satisfies condition (b).

To complete the proof it suffices to observe that the implication (b)  $\Rightarrow$  (c) is obvious and the implication (c)  $\Rightarrow$  (a) follows from Proposition 1.6.9.

The last theorem yields

**3.2.2. Proposition.** For every paracompact space X the following conditions are equivalent:

- (a) The space X satisfies the inequality  $\dim X \leq n$ .
- (b) Every open cover of the space X has an open shrinking of order  $\leq n$
- (c) Every open cover of the space X has an open refinement of order  $\leq n$ .

Conditions (a) and (c) in the last proposition are not equivalent in the realm of all normal spaces. Indeed, each cover of finite order is point-finite, so that every space X which satisfies (c) is weakly paracompact, whereas there exist normal spaces which are not weakly paracompact (see Example 2.1.6). One can show that conditions (a) and (c) are not equivalent even in the realm of all weakly paracompact spaces but the example is more difficult (see [GT], Problem 5.5.3(c)). On the other hand, conditions (b) and (c) are clearly equivalent for every normal space X.

Obviously, every family of sets which can be represented as the union of n+1 families of order  $\leq 0$ , i.e., consisting of pairwise disjoint sets has order  $\leq n$ . We are now going to strengthen Theorem 3.2.1 by proving that every locally finite open cover of a normal space X such that dim  $X \leq n$ has an open shrinking of this last form. This result is in a sense a substitute for the decomposition theorem for dim.

**3.2.3. Lemma.** For every locally finite open cover  $\mathcal{U} = \{U_s\}_{s\in S}$  of a normal space X such that  $\operatorname{ord} \mathcal{U} \leq n \geq 0$  there exists an open cover  $\mathscr{V}$  of the space X which can be represented as the union of n+1 families  $\mathscr{V}_1, \mathscr{V}_2, \ldots, \mathscr{V}_{n+1}$ , where  $\mathscr{V}_i = \{V_{i,s}\}_{s\in S}$ , such that  $\operatorname{ord} \mathscr{V}_i \leq 0$  and  $V_{i,s} \subset U_s$  for  $s \in S$  and  $i = 1, 2, \ldots, n+1$ .

**Proof.** We shall apply induction with respect to n. The lemma is obvious if n = 0. Assume that the lemma is proved for all normal spaces and all locally finite open covers of order  $\langle n \rangle = 1$  and consider a normal space X and a locally finite open cover  $\mathscr{U} = \{U_s\}_{s \in S}$  of X such that  $\operatorname{ord} \mathscr{U} \leq n$ . Let  $\{W_s\}_{s \in S}$  and  $\{F_s\}_{s \in S}$  be, respectively, an open and a closed shrinking of  $\mathscr{U}$  such that  $W_s \subset F_s$  for every  $s \in S$  (cf. [GT], Theorem 1.5.18). Denote by  $\mathscr{T}$  the family of all subsets of S that have exactly n+1 elements and for every  $T \in \mathscr{T}$  define

$$U_T = \bigcap_{s \in T} U_s, \quad W_T = \bigcap_{s \in T} W_s \text{ and } F_T = \bigcap_{s \in T} F_s.$$

From the local finiteness of  $\mathscr{U}$  it follows that the family  $\{F_r\}_{r\in\mathscr{F}}$  is locally finite. The inequality  $\operatorname{ord} \mathscr{U} \leq n$  implies that  $U_r \cap U_{r'} = \emptyset$  whenever  $T \neq T'$ .

For every  $T \in \mathcal{T}$  choose arbitrarily an  $s(T) \in T$  and let

$$V_{n+1,s} = \bigcup \{ U_r \colon s(T) = s \} \quad \text{for } s \in S.$$

The family  $\mathscr{V}_{n+1} = \{V_{n+1,s}\}_{s\in S}$  has order  $\leqslant 0$  and  $\mathscr{V}_{n+1,s} \subset U_s$  for  $s \in S$ ; moreover

(1) 
$$\bigcup_{s\in S} V_{n+1,s} = \bigcup_{T\in\mathcal{F}} U_T.$$

Consider now two subspaces of the space X

$$Y = X \setminus \bigcup_{T \in \mathcal{F}} W_T$$
 and  $Z = X \setminus \bigcup_{T \in \mathcal{F}} F_T;$ 

the subspace Y is closed in X, the subspace Z is open in X, and  $Z \subset Y$ . From the definition of the sets  $W_r$  it follows that the locally finite open cover  $\{Y \cap W_s\}_{s \in S}$  of the space Y has order  $\leq n-1$ . Since Y is a normal space, by virtue of the inductive assumption there exists an open cover  $\mathscr{V}'$  of the space Y which can be represented as the union of n families  $\mathscr{V}'_1, \mathscr{V}'_2, \ldots, \mathscr{V}'_n$ , where  $\mathscr{V}'_i = \{V'_{i,s}\}_{s \in S}$ , such that  $\operatorname{ord} \mathscr{V}'_i \leq 0$  and  $V'_{i,s}$  $\subset Y \cap W_s \subset U_s$  for  $s \in S$  and i = 1, 2, ..., n.

For i = 1, 2, ..., n let  $\mathscr{V}_i = \{V_{i,s}\}_{s \in S}$ , where  $V_{i,s} = Z \cap V'_{i,s}$ . One readily sees that the sets  $V_{i,s}$  are open in X, that ord  $\mathscr{V}_i \leq 0$  for i = 1, 2, ..., nand that  $V_{i,s} \subset U_s$  for  $s \in S$  and i = 1, 2, ..., n; moreover,  $Z \subset \bigcup_{i=1}^n \bigcup_{s \in S} V_{i,s}$ . Since  $F_T \subset U_T$  for  $T \in \mathscr{T}$ , we have

$$X \setminus Z = \bigcup_{T \in \mathscr{F}} F_T \subset \bigcup_{T \in \mathscr{F}} U_T,$$

so that by virtue of (1), the union  $\mathscr{V} = \bigcup_{i=1}^{m} \mathscr{V}_i$  is an open cover of the space X.  $\Box$ 

Theorem 3.2.1 and Lemma 3.2.3 yield

**3.2.4. Ostrand's theorem.** A normal space X satisfies the inequality dim  $X \le n \ge 0$  if and only if for every locally finite open cover  $\mathcal{U} = \{U_s\}_{s\in S}$  of the space X there exists an open cover  $\mathcal{V}$  of the space X which can be represented as the union of n+1 families  $\mathcal{V}_1, \mathcal{V}_2, \dots, \mathcal{V}_{n+1}$ , where  $\mathcal{V}_i = \{V_{i,s}\}_{s\in S}$ , such that  $\operatorname{ord} \mathcal{V}_i \le 0$  and  $V_{i,s} \subset U_s$  for  $s \in S$  and i = 1, 2, ..., n+1.

Ch. 3, § 2] Characterizations of the dimension dim in normal spaces

The next theorem is a strong version of the theorem on partitions.

**3.2.5. Morita's theorem.** A normal space X satisfies the inequality dim  $X \leq n \geq 0$  if and only if for every locally finite family  $\{U_s\}_{s\in S}$  of open subsets of X and every family  $\{F_s\}_{s\in S}$  of closed subsets of X such that  $F_s \subset U_s$  for  $s \in S$  there exist families  $\{W_s\}_{s\in S}$  and  $\{V_s\}_{s\in S}$  of open subsets of X such that  $F_s \subset V_s \subset \overline{V_s} \subset W_s \subset \overline{W_s} \subset U_s$  for  $s \in S$  and  $\operatorname{ord}(\{\overline{W_s} \setminus V_s\}_{s\in S}) \leq n-1$ .

**Proof.** First we shall prove that every normal space X with dim  $X \le n$  satisfies the condition in the theorem. Consider a locally finite family  $\{U_s\}_{s\in S}$  of open subsets of X and a family  $\{F_s\}_{s\in S}$  of closed subsets of X such that  $F_s \subset U_s$  for  $s \in S$ . Denote by  $\mathscr{T}$  the family of all finite subsets of S, define for every non-empty  $T \in \mathscr{T}$ 

(2) 
$$G_T = \bigcap_{s \in T} U_s \cap \bigcap_{s \notin T} (X \setminus F_s),$$

and let  $G_{\sigma} = X \bigcup_{s \in S} F_s$ . The family  $\{G_r\}_{r \in \mathcal{F}}$  is a locally finite open cover of the space X. By virtue of Dowker's theorem (cf. Remark 3.2.7) the cover  $\{G_r\}_{r \in \mathcal{F}}$  has an open shrinking  $\{H_r\}_{r \in \mathcal{F}}$  of order  $\leq n$ , and, in turn, the cover  $\{H_r\}_{r \in \mathcal{F}}$  has a closed shrinking  $\{A_r\}_{r \in \mathcal{F}}$ . From (2) it follows that

(3) if 
$$A_T \cap F_s \neq \emptyset$$
, then  $s \in T$ .

For every  $T \in \mathcal{T}$  and each  $s \in T$  we can define open sets  $W_T(s)$  and  $V_T(s)$  such that

(4) 
$$A_T \subset V_T(s) \subset \overline{V_T(s)} \subset W_T(s) \subset \overline{W_T(s)} \subset H_T$$
 for  $s \in T$ 

(5) if  $s, s' \in T$  and  $s' \neq s$ , then either  $\overline{W_r(s)} \subset V_r(s')$  or  $\overline{W_r(s')} \subset V_r(s)$ .

Moreover, define  $W_T(s) = V_T(s) = \emptyset$  for  $s \in S \setminus T$ .

$$W_s = \bigcup_{T \in \mathcal{F}} W_T(s)$$
 and  $V_s = \bigcup_{T \in \mathcal{F}} V_T(s)$  for  $s \in S$ .

The sets  $W_s$  and  $V_s$  are open; from the local finiteness of the cover  $\{H_T\}_{T\in\mathcal{F}}$ it follows that  $\overline{V_s} \subset W_s$  and  $\overline{W_s} \subset U_s$ , because (2) implies that  $\overline{W_T(s)} \subset U_s$ . Now, for every point  $x \in F_s$  there exists a  $T \in \mathcal{T}$  such that  $x \in A_T$ ; as  $s \in T$  by virtue of (3), it follows from (4) that  $x \in V_T(s) \subset V_s$ , so that  $F_s \subset V_s$ . To complete the first part of the proof it suffices to show that  $\operatorname{ord}(\{\overline{W_s} \setminus V_s\}_{s \in S}) \leq n-1$ . The covering dimension

Consider a sequence  $s_1, s_2, ..., s_k$  of distinct elements of the set S such that  $\bigcap_{i=1}^{k} (\overline{W}_{s_i} \setminus V_{s_i}) \neq \emptyset$ . Let x be a point in the last intersection. There exist sets  $T_1, T_2, ..., T_k \in \mathcal{T}$  such that  $s_i \in T_i$  for i = 1, 2, ..., kand  $x \in \bigcap_{i=1}^{k} (\overline{W}_{T_i}(s_i) \setminus V_{T_i}(s_i))$ ; by virtue of (5)  $T_i \neq T_j$  for  $i \neq j$ . From (4) it follows that  $x \notin \bigcup_{i=1}^{k} A_{T_i}$ , and since  $\{A_T\}_{T \in \mathcal{T}}$  is a cover of X, there exists a  $T_0 \in \mathcal{T}$  such that  $x \in A_{T_0} \subset H_{T_0}$ ; clearly  $T_0 \neq T_i$  for i = 1, 2, ......, k. Thus  $x \in \bigcap_{i=0}^{k} H_{T_i}$ , and as  $\operatorname{ord}(\{H_T\}_{T \in \mathcal{T}}) \leq n$  we have  $k \leq n$ , which shows that  $\operatorname{ord}(\{\overline{W}_s \setminus V_s\}_{s \in S}) \leq n-1$ .

Consider now a normal space X which satisfies the condition in the theorem. By virtue of Remark 1.7.10, to prove that dim  $X \leq n$  it suffices to show that for every sequence  $(A_1, B_1), (A_2, B_2), \dots, (A_{n+1}, B_{n+1})$  of n+1 pairs of disjoint closed subsets of X there exist closed sets  $L_1, L_2, \dots$  $\dots, L_{n+1}$  such that  $L_i$  is a partition between  $A_i$  and  $B_i$  and  $\bigcap_{i=1}^{n+1} L_i = \emptyset$ . Define

 $U_i = X \ B_i$  and  $F_i = A_i$  for i = 1, 2, ..., n+1.

Applying the condition in the theorem to  $S = \{1, 2, ..., n+1\}$  and the sets  $U_i$  and  $F_i$  defined above, we obtain a family  $\{V_i\}_{i=1}^{n+1}$  of open subsets of X such that

(6) 
$$A_i \subset V_i \subset \overline{V}_i \subset X \setminus B_i$$
 for  $i = 1, 2, ..., n+1$  and  
ord $(\{\overline{V}_i \setminus V_i\}_{i=1}^{n+1}) \leq n-1.$ 

By virtue of the first part of (6) the set  $L_i = \overline{V}_i \setminus V_i$  is a partition between  $A_i$  and  $B_i$  for i = 1, 2, ..., n+1; the second part of (6) means that  $\bigcap_{i=1}^{n+1} L_i$ =  $\emptyset$ .  $\Box$ 

Theorem 3.2.5 and Remark 1.7.10 yield

**3.2.6. Theorem on partitions.** A normal space X satisfies the inequality  $\dim X \le n \ge 0$  if and only if for every sequence  $(A_1, B_1), (A_2, B_2), \ldots$  $\ldots, (A_{n+1}, B_{n+1})$  of n+1 pairs of disjoint closed subsets of X there exist closed sets  $L_1, L_2, \ldots, L_{n+1}$  such that  $L_i$  is a partition between  $A_i$  and  $B_i$  and  $\bigcap_{i=1}^{n+1} L_i = \emptyset$ .  $\Box$  **3.2.7. Remark.** Let us note that if the family  $\{U_s\}_{s\in S}$  in Morita's theorem is finite then in the proof of this theorem the cover  $\{H_r\}_{r\in\mathcal{F}}$  can be obtained by applying Theorem 1.7.8 rather than Dowker's theorem. In particular, the theorem on partitions can be proved without resorting to Dowker's theorem.

We now turn to the theorem on extending mappings to spheres.

**3.2.8. Lemma.** Let  $f, g: X \to S_I^n$  be continuous mappings of a topological space X to the boundary  $S_I^n$  of the (n+1)-cube  $I^{n+1}$  in  $\mathbb{R}^{n+1}$ . If for every point  $x \in X$  the points f(x) and g(x) belong to the same face of  $I^{n+1}$ , then the mappings f and g are homotopic.

**Proof.** By assigning to every point  $x \in X$  and each number  $t \in I$  the point h(x, t) which divides the interval with end-points f(x) and g(x) in the ratio of t to 1-t one defines a homotopy  $h: X \times I \to S_I^n$  between f and  $g.\square$ 

**3.2.9. Theorem.** If X is a normal space and A is a closed subspace of X such that dim  $Z \le n \ge 0$  for every closed subspace Z of the space X contained in  $X \setminus A$ , then for every continuous mapping  $f: A \to S^n$  there exists a continuous extension  $F: X \to S^n$  of f over X.

**Proof.** There exists an open set  $W \subset X$  containing A and such that f has a continuous extension  $\overline{f}: W \to S^n$  over W. Consider an open set V such that  $A \subset V \subset \overline{V} \subset W$  and let  $Z = X \setminus V$  and  $B = Z \cap \overline{V}$ . To show that f is continuously extendable over X it suffices to prove that for the mapping  $h = \overline{f} | B: B \to S^n$  there exists a continuous extension  $H: Z \to S^n$  over Z. Indeed, the mapping  $F: X \to S^n$  defined by letting

$$F(x) = f(x)$$
 for  $x \in V$  and  $F(x) = H(x)$  for  $x \in Z$ 

will then be a continuous extension of f over X. Since dim $Z \leq n$ , the above observation shows that with no loss of generality one can assume that the space X satisfies the inequality dim $X \leq n$ . Moreover, one can assume that the space X is compact, because dim $\beta X \leq n$  by Theorem 3.1.25 and the mapping f is continuously extendable over the closure of the set A in  $\beta X$ , this last closure being the Čech-Stone compactification  $\beta A$  of the space A (see [GT], Corollary 3.6.8). Finally, instead of mappings to the *n*-sphere  $S^n$  one can consider mappings to the boundary  $S_I^n$  of the (n+1)-cube  $I^{n+1}$  in  $\mathbb{R}^{n+1}$ , which is homeomorphic to  $S^n$ .

[Ch. 3, § 2

Consider therefore a continuous mapping  $f: A \to S_I^n$  defined on a closed subspace A of a compact space X such that dim  $X \leq n$ . For i = 1, 2, ..., n+1 denote by  $p_i: I^{n+1} \to I$  the projection of the (n+1)-cube  $I^{n+1}$  onto the *i*-th coordinate axis and let

 $f_t = p_i f: A \to I, \quad A_i = f_i^{-1}(0) \quad \text{and} \quad B_t = f_i^{-1}(1);$  obviously

(7) 
$$A = \bigcup_{i=1}^{n+1} A_i \cup \bigcup_{i=1}^{n+1} B_i.$$

By virtue of Theorem 3.2.6 there exist closed sets  $L_1, L_2, ..., L_{n+1}$  such that  $L_i$  is a partition between  $A_i$  and  $B_i$  and  $\bigcap_{i=1}^{n+1} L_i = \emptyset$ . Consequently, there exist open sets  $U_i, W_i \subset X$ , where i = 1, 2, ..., n+1, such that

$$A_i \subset U_i, \quad B_i \subset W_i, \quad U_i \cap W_i = \emptyset \quad \text{and} \quad X \setminus L_i = U_i \cup W_i.$$

From Theorem 3.1.2 it follows that there exist open sets  $V_1, V_2, ..., V_{n+1}$  satisfying

$$L_i \subset V_i \subset X \setminus (A_i \cup B_i)$$
 for  $i = 1, 2, ..., n+1$  and  $\bigcap_{i=1}^{n+1} V_i = \emptyset$ .

By virtue of Urysohn's lemma for i = 1, 2, ..., n+1 there exist continuous functions

$$g'_i: U_i \cup L_i \to [0, 1/2]$$
 and  $g''_i: W_i \cup L_i \to [1/2, 1]$ 

such that

$$g'_i(U_i \setminus V_i) \subset \{0\}, \quad g'_i(L_i) \subset \{1/2\}$$

and

$$g''_i(W_i \setminus V_i) \subset \{1\}, \quad g''_i(L_i) \subset \{1/2\}.$$

Letting

$$g_i(x) = \begin{cases} g'_i(x) & \text{for } x \in U_i \cup L_i, \\ g''_i(x) & \text{for } x \in W_i \cup L_i, \end{cases}$$

we define continuous functions  $g_i: X \to I$  such that

(8) 
$$g_i(A_i) \subset \{0\}, \quad g_i(B_i) \subset \{1\}$$
 and  $g_i^{-1}(1/2) \subset V_i$   
for  $i = 1, 2, ..., n+1$ 

Since  $\bigcap_{i=1}^{n+1} V_i = \emptyset$ , the continuous mapping  $g: X \to I^{n+1}$  defined by letting  $g(x) = (g_1(x), g_2(x), \dots, g_{n+1}(x))$  does not assume the value  $a = (1/2, 1/2, \dots, 1/2) \in I^{n+1}$ . The composition of the mapping g and the projection

p of  $I^{n+1} \setminus \{a\}$  from the point a onto the boundary  $S_I^n$  of  $I^{n+1}$  is a continuous mapping  $G: X \to S_I^n$ . From (7) and (8) it follows that  $g(A) \subset S_I^n$ , so that G|A = g|A, i.e., the mapping G is a continuous extension of the restriction g|A over X. Now, for every point  $x \in A$  there exists an  $i \leq n+1$ such that either  $f_i(x) = g_i(x) = 0$  or  $f_i(x) = g_i(x) = 1$ , so that for every point  $x \in A$  the points f(x) and g(x) belong to the same face of  $I^{n+1}$ . From Lemma 3.2.8 it follows that the mappings f and g|A are homotopic, and Lemma 1.9.7 implies that there exists a continuous extension  $F: X \to S_I^n$ of the mapping f over X.  $\Box$ 

Theorem 3.2.9, Remark 1.9.4 and Theorem 3.2.6 yield the following

**3.2.10. Theorem on extending mappings to spheres.** A normal space X satisfies the inequality dim  $X \le n \ge 0$  if and only if for every closed subspace A of the space X and each continuous mapping  $f: A \to S^n$  there exists a continuous extension  $F: X \to S^n$  of f over X.  $\Box$ 

The last characterization of the covering dimension to be established in this section uses the notion of an  $\mathscr{E}$ -mapping (see Definition 1.10.8).

**3.2.11. Theorem on &-mappings.** A normal space X satisfies the inequality  $\dim X \leq n$  if and only if for every finite open cover  $\mathscr{E}$  of the space X there exists an  $\mathscr{E}$ -mapping of X to a polyhedron of dimension  $\leq n$ .

**Proof.** By virtue of Theorem 1.10.11, it suffices to show that for every finite open cover  $\mathscr{E} = \{U_i\}_{i=1}^k$  of a normal space X with  $0 \leq \dim X \leq n$  there exists an  $\mathscr{E}$ -mapping of X to a polyhedron of dimension  $\leq n$ .

Consider an open shrinking  $\mathscr{V} = \{V_i\}_{i=1}^k$  of the cover  $\mathscr{E}$  such that ord  $\mathscr{V} \leq n$  and a closed shrinking  $\{F_i\}_{i=1}^k$  of  $\mathscr{V}$ . Let  $\mathscr{N}(\mathscr{V})$  be a nerve of  $\mathscr{V}$  with vertices  $p_1, p_2, \ldots, p_k \in \mathbb{R}^m$ . By Urysohn's lemma for  $i = 1, 2, \ldots$  $\ldots, k$  there exists a continuous function  $f_i: X \to I$  such that  $f_i(X \setminus V_i)$  $\subset \{0\}$  and  $f_i(F_i) \subset \{1\}$ . One readily checks (cf. the proof of Theorem 1.10.7) that the formula

$$\varkappa(x) = \varkappa_1(x)p_1 + \varkappa_2(x)p_2 + \ldots + \varkappa_k(x)p_k,$$

where

$$\varkappa_i(x) = \frac{f_i(x)}{f_1(x) + f_2(x) + \dots + f_k(x)} \quad \text{for } i = 1, 2, \dots, k,$$

defines a continuous mapping  $\kappa \colon X \to N(\mathscr{V})$  of X to the underlying polyhedron of the nerve  $\mathscr{N}(\mathscr{V})$  which satisfies the conditions

$$\varkappa^{-1}(\operatorname{St}_{\mathscr{N}(\mathscr{V})}(p_i)) = V_i \subset U_i \quad \text{for } i = 1, 2, \dots, k.$$

As  $\{St_{\mathscr{N}(\mathscr{P})}(p_i)\}_{i=1}^k$  is an open cover of  $N(\mathscr{V})$ , the mapping  $\varkappa$  is an  $\mathscr{E}$ -mapping. The inequality ord  $\mathscr{V} \leq n$  implies that  $N(\mathscr{V})$  has dimension  $\leq n$ .  $\Box$ 

Applying Theorem 1.10.15 and arguing as in the proof of Theorem 1.10.16, one obtains the following strengthening of the theorem on  $\mathscr{E}$ -mappings.

**3.2.12. Theorem.** A normal space X satisfies the inequality  $\dim X \leq n$  if and only if for every finite open cover  $\mathscr{E}$  of the space X there exists an  $\mathscr{E}$ -mapping of X onto a polyhedron of dimension  $\leq n$ .

We now turn to a study of the behaviour of the dimension dim under Cartesian multiplication. First of all, let us recall that in Section 2.2 we cited an example of a normal space Z, whose square  $Z \times Z$  is also normal, such that  $\operatorname{Ind} Z = 0$  and yet  $\operatorname{Ind}(Z \times Z) > 0$ ; in view of Theorem 1.6.11, this example shows that the inequality  $\dim(X \times Y) \leq \dim X + \dim Y$  does not hold under the sole assumption of the normality of the Cartesian product  $X \times Y$ . Just as in the case of the dimension Ind, several theorems determining conditions for that inequality have been discovered, but there is no strongest result among them. We shall quote two such theorems which are relatively strong. Thus, the inequality  $\dim(X \times Y) \leq \dim X +$  $+\dim Y$  holds for every pair X, Y of normal spaces of which at least one is non-empty provided that either

- (i) the Cartesian product  $X \times Y$  is normal and one of the factors is compact (more generally: the projection onto one of the factors is a closed mapping),
- or
- (ii) the Cartesian product  $X \times Y$  is normal, one of the factors is metrizable and the other is countably paracompact (more generally: one of the factors can be mapped to a metrizable space by a perfect mapping and the other is countably paracompact).

Let us add that the proofs of Cartesian product theorems for dim are fairly difficult. As a sample we shall prove here two Cartesian product theorems for dim which are among the simplest.

**3.2.13. Theorem.** For every pair X, Y of compact spaces of which at least one is non-empty we have

$$\dim(X \times Y) \leqslant \dim X + \dim Y.$$

**Proof.** We can assume that dim X = n and dim Y = m, where n and m are non-negative integers. Consider an arbitrary sequence  $(A_1, B_1)$ ,  $(A_2, B_2), ..., (A_{n+m+1}, B_{n+m+1})$  of n+m+1 pairs of disjoint closed subsets of the Cartesian product  $X \times Y$ . The sets  $A_i$  being compact, for i = 1, 2, ..., n+m+1 there exist closed sets  $E_{i,j} \subset X$  and  $F_{i,j} \subset Y$ and open sets  $U_{i,j} \subset X$  and  $V_{i,j} \subset Y$ , where  $j = 1, 2, ..., k_i$ , such that

(9) 
$$A_{i} \subset \bigcup_{j=1}^{k_{i}} (E_{i,j} \times F_{i,j})$$
  
and  $E_{i,j} \times F_{i,j} \subset U_{i,j} \times V_{i,j} \subset (X \times Y) \setminus B_{i}$ 

By virtue of Theorem 3.2.5 there exist open sets  $G_{i,j} \subset X$  and  $H_{i,j} \subset Y$  such that

(10) 
$$E_{i,j} \subset G_{i,j} \subset \overline{G}_{i,j} \subset U_{i,j}$$
 and  $F_{i,j} \subset H_{i,j} \subset \overline{H}_{i,j} \subset V_{i,j}$   
and

(11) ord({Fr 
$$G_{i,j}$$
}<sub>(i,j)\in S</sub>)  $\leq n-1$  and ord({Fr  $H_{i,j}$ }<sub>(i,j)\in S</sub>)  $\leq m-1$ ,

where S denotes the set of all pairs (i, j) with i = 1, 2, ..., n+m+1 and  $j = 1, 2, ..., k_i$ . For i = 1, 2, ..., n+m+1 let  $W_i = \bigcup_{j=1}^{k_i} (G_{i,j} \times H_{i,j})$ . From (9) and (10) it follows that  $A_i \subset W_i \subset \overline{W_i} \subset (X \times Y) \setminus B_i$ , so that the set  $L_i = \operatorname{Fr} W_i$  is a partition between  $A_i$  and  $B_i$ . Consider the family

$$\mathscr{C}_{l} = \{\operatorname{Fr} G_{l,j} \times \overline{H}_{l,j}\}_{j=1}^{k_{l}} \cup \{\overline{G}_{l,j} \times \operatorname{Fr} H_{l,j}\}_{j=1}^{k_{l}}$$

of subsets of  $X \times Y$ , denote by  $C_i$  the union  $\bigcup \mathscr{C}_i$ , and let  $\mathscr{C} = \mathscr{C}_1 \cup \mathscr{C}_2 \cup \cup \cup \dots \cup \mathscr{C}_{n+m+1}$ . Since  $L_i \subset \bigcup_{j=1}^{k_i} \operatorname{Fr}(G_{i,j} \times H_{i,j}) \subset C_i$ , by virtue of Theorem 3.2.6 to complete the proof it suffices to show that  $\bigcap_{i=1}^{n+m+1} C_i = \emptyset$ . Now,

the last equality follows from the inequality  $\operatorname{ord} \mathscr{C} \leq n+m+1$ , which in turn is a consequence of (11), because among each n+m+1 members of the family  $\mathscr{C}$  there are either n+1 sets of the form  $\operatorname{Fr} G_{i,j} \times \overline{H}_{i,j}$  or m+1 sets of the form  $\overline{G}_{i,j} \times \operatorname{Fr} H_{i,j}$ .

**3.2.14. Theorem.** For every pair X, Y of normal spaces of which at least one is non-empty such that the Cartesian product  $X \times Y$  is strongly paracompact we have

$$\dim(X \times Y) \leq \dim X + \dim Y.$$

**Proof.** From Theorems 3.2.13 and 3.1.25 it follows that

 $\dim(\beta X \times \beta Y) \leq \dim \beta X + \dim \beta Y = \dim X + \dim Y.$ 

To complete the proof it suffices to apply Theorem 3.1.23.  $\Box$ 

### Historical and bibliographic notes

Theorem 3.2.1 and Proposition 3.2.2 were established by Dowker in [1947]. Lemma 3.2.3 and Theorem 3.2.4 were proved by Ostrand in [1971]; special cases of Theorem 3.2.4 were obtained earlier by Ostrand in [1965] (for metrizable spaces) and by French in [1970] (for collectionwise normal spaces). Theorem 3.2.5 was proved by Morita in [1950a], and Theorem 3.2.6-by Hemmingsen in [1946]. Theorem 3.2.10 was obtained independently by Hemmingsen in [1946], Alexandroff in [1947], and Dowker in [1947]; for compact spaces it was proved by Alexandroff in [1940] and by Morita in [1940]. Theorem 3.2.9, which is a simple consequence of Theorem 3.2.10, was noted by Alexandroff in [1947]. As stated in the notes to Section 1.10, Theorems 3.2.11 and 3.2.12 for metric spaces were established by Kuratowski in [1933a] (who generalized a characterization of dimension of compact metric spaces discovered by Alexandroff in [1928]); Kuratowski's proof extends without substantial changes to normal spaces. The fact that the inequality  $\dim(X \times Y) \leq \dim X + \dim Y$  holds for every pair X, Y of spaces satisfying either (i) or (ii) was proved by Filippov in [1979] (announcement [1973]); (ii) part was announced independently by Pasynkov in [1973]). In the original formulation of (ii) one assumes that the Cartesian product  $X \times Y$  is normal and countably paracompact (cf. the commentary to (ii) in the notes to Section 2.4). Theorem 3.2.13 was proved by Hemmingsen in [1946], and Theorem 3.2.14—by Morita in [1953]. Further information on Cartesian product theorems for dim can be found in Morita [1953], Kodama [1969] (a simplified proof in Engelking [1973]), Nagami [1970], and Pears [1975].

## **Problems**

**3.2.A.** Show that a normal space X satisfies the inequality dim  $X \le n$  if and only if every locally finite open cover of the space X has a locally finite closed refinement of order  $\le n$  or—equivalently—if every locally finite open cover of the space X has a closed shrinking of order  $\le n$ .

**3.2.B** (Ostrand [1971]; for collectionwise normal spaces French [1970]; for metrizable spaces Ostrand [1965]). Prove that a normal space X satisfies the inequality dim  $X \leq n$  if and only if for every locally finite open cover  $\mathcal{U} = \{U_s\}_{s \in S}$  of the space X there exists such a sequence  $\mathcal{V}_1, \mathcal{V}_2, ...$ of discrete families of open subsets of X, where  $\mathcal{V}_i = \{V_{i,s}\}_{s \in S}$ , that  $V_{i,s} \subset U_s$  for  $s \in S$  and i = 1, 2, ... and the union of each n+1 families  $\mathcal{V}_i$ constitutes a cover of the space X. Show that for every locally finite open cover  $\mathcal{U} = \{U_s\}_{s \in S}$  of a normal space X such that dim  $X \leq n$  besides the sequence  $\mathcal{V}_1, \mathcal{V}_2, ...$  with the above properties there also exists such a sequence  $\mathcal{W}_1, \mathcal{W}_2, ...$  of discrete families of open subsets of X, where  $\mathcal{W}_i = \{W_{i,s}\}_{s \in S}$ , that  $\overline{V}_{i,s} \subset W_{i,s} \subset U_s$  for  $s \in S$  and i = 1, 2, ...

*Hint.* Define the families  $\mathscr{V}_i$  and  $\mathscr{W}_i$  for  $i \leq n+1$  by applying Theorem 3.2.4, then define inductively the families  $\mathscr{V}_i$  and  $\mathscr{W}_i$  for i > n+1. Assume that the families  $\mathscr{V}_i$  and  $\mathscr{W}_i$  are defined for  $i \leq m-1 \geq n+1$ , denote by  $\mathscr{T}$  the family of all subsets of the set  $\{1, 2, ..., m-1\}$  which have exactly *n* elements and, for every  $T \in \mathscr{T}$ , define  $F_T = X \bigcup_{i \in T} V_i$ , where  $V_i = \bigcup_{s \in S} V_{i,s}$ . Consider families  $\{V_T\}_{T \in \mathscr{T}}$  and  $\{W_T\}_{T \in \mathscr{T}}$  of open subsets of X such that

 $F_T \subset V_T \subset \overline{V}_T \subset W_T$  for  $T \in \mathcal{F}$  and  $\overline{W}_T \cap \overline{W}_{T'} = \emptyset$  for  $T \neq T'$ ,

for every  $T \in \mathcal{T}$  choose arbitrarily an  $i(T) \leq m-1$  such that  $i(T) \notin T$ , note that  $F_T \subset V_{i(T)}$  and let

$$V_{m.s} = \bigcup_{T \in \mathcal{T}} V_T \cap V_{i(T),s}$$
 and  $W_{m.s} = \bigcup_{T \in \mathcal{T}} W_T \cap W_{i(T),s}$ .

**3.2.C** (Arhangel'skii [1963]). A family  $\mathscr{A}$  of subsets of a set X is *independent* if  $A \setminus B \neq \emptyset \neq B \setminus A$  for each pair A, B of distinct members of  $\mathscr{A}$  By the *rank* of a family of sets  $\mathscr{A}$  we mean the largest integer n such that the family  $\mathscr{A}$  contains n+1 sets with a non-empty intersection which form an independent family; if no such integer exists we say that the family  $\mathscr{A}$  has rank  $\infty$ . Clearly, the rank of a family of sets does not exceed the order of that family.

Prove that a normal space X satisfies the inequality  $\dim X \leq n$  if and only if every finite open cover of the space X has a finite open refinement of rank  $\leq n$ .

*Hint.* For an open cover  $\mathscr{U} = \{U_i\}_{i=1}^k$  of the space X and its finite open refinement  $\mathscr{V}$  of rank  $\leq n$  define  $\mathscr{V}_i = \{V \in \mathscr{V} : V \subset U_i\}$  and consider the families

 $\mathscr{W}_1 = \{ V \in \mathscr{V}_1 : V \text{ is not contained in any member of } \mathscr{V}_2 \cup \mathscr{V}_3 \cup \ldots \cup \mathscr{V}_k \}, \\ \mathscr{W}_2 = \{ V \in \mathscr{V}_2 : V \text{ is not contained in any member of } \mathscr{W}_1 \cup \mathscr{V}_3 \cup \ldots \cup \mathscr{V}_k \}, \\ \ldots \\ \mathscr{W}_k = \{ V \in \mathscr{V}_k : V \text{ is not contained in any member of } \\ \mathscr{W}_1 \cup \mathscr{W}_2 \cup \ldots \cup \mathscr{W}_{k-1} \}.$ 

Check that the family  $\{W_i\}_{i=1}^k$ , where  $W_i = \bigcup \mathcal{W}_i$ , is a cover of X and has order  $\leq n$ .

**3.2.D.** (a) (Wallace [1945], Pupko [1961]) Deduce Theorems 3.1.8 and 3.1.10 from Theorem 3.2.10.

(b) (Hemmingsen [1946]) Deduce Theorem 3.2.13 from Theorem 3.2.11.

*Hint.* Show that every finite open cover of the Cartesian product of compact spaces X and Y has a refinement of the form  $\{U \times V: U \in \mathcal{U}, V \in \mathcal{V}\}$ , where  $\mathcal{U}$  and  $\mathcal{V}$  are finite open covers of X and Y, respectively.

**3.2.E** (Aleksandroff [1947]). Show that a normal space X satisfies the inequality dim  $X \le n \ge 0$  if and only if no continuous mapping  $f: X \to B^{n+1}$  is essential (see Problem 1.9.A).

**3.2.F** (Aleksandroff [1947]). A compact space X such that dim  $X = n \ge 1$  is an *n*-dimensional Cantor manifold if no closed subset L of X satisfying the inequality dim  $L \le n-2$  separates the space X (cf. Definition 1.9.5).

Ch. 3, § 2]

### Problems

(a) Let  $f, g: X \to S^n$  be continuous mappings of a compact space X to the *n*-sphere  $S^n$ . Show that if dim  $Z \le n-1$  for every closed subspace Z of the space X contained in the set  $D(f, g) = \{x \in X: f(x) \ne g(x)\}$ , then the mappings f and g are homotopic.

(b) Prove that every compact space X such that  $\dim X = n \ge 1$  contains an *n*-dimensional Cantor-manifold.

Hint. See the proof of Theorem 1.9.8.

**3.2.G** (Zolotarev [1975]). (a) Show that a normal subspace M of a normal space X satisfies the inequality dim  $M \le n$  if and only if for every open set  $U \subset X$  which contains the set M there exist a normal space Z, a normal subspace  $A \subset Z$  satisfying dim  $A \le n$  and continuous mappings  $f: M \to Z$  and  $g: Z \to X$  such that  $f(M) \subset A \subset g^{-1}(U)$  and gf(x) = x for  $x \in M$ .

(b) Prove that for every completely paracompact subspace M of a normal space X we have dim  $M \leq \dim X$  (see Problem 2.4.B).

Hint. One can assume that  $\dim X = n < \infty$  and X is a compact space (see Theorem 3.1.25). Consider an open set  $U \subset X$  which contains the set M and for every point  $x \in M$  choose a neighbourhood  $V_x$  such that  $x \in V_x \subset \overline{V_x} \subset U$ . Let  $\mathscr{V}_1, \mathscr{V}_2, \ldots$  be a sequence of star-finite open covers of the space M such that the union  $\bigcup_{i=1}^{\infty} \mathscr{V}_i$  contains a refinement  $\mathscr{V}$  of the cover  $\{M \cap V_x\}_{x \in M}$ . For  $i = 1, 2, \ldots$  consider the decomposition  $\{\mathscr{V}_s\}_{s \in S}$ of the cover  $\mathscr{V}_i$  into the components where  $S_i \cap S_j = \emptyset$  whenever  $i \neq j$ ; for every  $x \in M$  denote by  $f_i(x)$  the unique  $s \in S_i$  such that  $x \in \bigcup \mathscr{V}_s$ , and for each  $s \in S_i$  let  $\mathscr{V}_s \cap \mathscr{V} = \{V_{s,j}\}_{j=1}^{\infty}$ . Note that the space  $Z = X \times$  $\times \prod_{i=1}^{\infty} S_i$ , where  $S_i$  has the discrete topology, is normal, and apply Problem 2.4.A and Theorem 3.2.14 to show that dim  $Z \leq n$ . Apply (a) to the continuous mapping f:  $M \to Z$  defined by letting  $f(x) = (x, f_1(x), f_2(x), \ldots)$ for  $x \in M$ , the projection g:  $Z \to X$  and the set  $A = \bigcup_{i,j=1}^{\infty} \bigcup_{s \in S_i} \overline{f(V_{s,j})} \subset Z$ .

**3.2.H.** (a) Prove that every locally finite functionally open cover  $\{U_s\}_{s\in S}$  of a topological space X has such shrinkings  $\{F_s\}_{s\in S}$  and  $\{W_s\}_{s\in S}$ , respectively functionally closed and functionally open, that  $F_s \subset W_s \subset \overline{W_s} \subset U_s$  for  $s \in S$ .

*Hint*. For every  $s \in S$  choose a continuous function  $f_s: X \to I$  such that  $U_s = f_s^{-1}((0, 1])$  and consider the function  $f: X \to I$  defined by letting  $f(x) = \sup_{s \in S} f_s(x)$  for  $x \in X$ .

(b) (Pasynkov [1965]) Prove that a Tychonoff space X satisfies the inequality dim  $X \le n$  if and only if every locally finite functionally open cover of the space X has a functionally open refinement of order  $\le n$  (see notes to Section 3.1).

Hint. Let  $\{U_s\}_{s\in S}$  be a locally finite functionally open cover of the space X. Consider a functionally closed shrinking  $\{F_s\}_{s\in S}$  of  $\{U_s\}_{s\in S}$  and continuous functions  $f_s: X \to I$  such that  $f_s(X \setminus U_s) \subset \{0\}$  and  $f_s(F_s) \subset \{1\}$ . Verify that the formula  $\varrho(x, y) = \sum_{s\in S} |f_s(x) - f_s(y)|$  defines a pseudometric on the set X and consider the metric space  $(Y, \varrho)$  obtained by identifying each pair x, y of points in X such that  $\varrho(x, y) = 0$ . Check that by letting f(x) = [x] one defines a continuous mapping of X to Y and consider the mapping  $F: \beta X \to \beta Y$  such that F(x) = f(x) for  $x \in X$ . Apply the fact that every Hausdorff space which can be mapped onto a paracompact space by a perfect mapping is itself paracompact (see [GT], Theorem 5.1.35) and use Problem 3.1.1(b) and Theorem 3.2.1.

**3.2.I.** (a) Observe that if M is a functionally open subspace of a space X, then a set  $A \subset M$  is functionally open in X if and only if it is functionally open in M. Give an example of a functionally closed subspace M of a completely regular space X and of a set  $A \subset M$  which is functionally closed in M and yet is not functionally closed in X.

(b) Prove that a completely regular space X satisfies the inequality dim  $X \le n \ge 0$  if and only if for every finite functionally open cover  $\mathscr{U} = \{U_i\}_{i=1}^k$  of the space X there exists a functionally open cover  $\mathscr{V}$  of the space X which can be represented as the union of n+1 families  $\mathscr{V}_1, \mathscr{V}_2, \ldots, \mathscr{V}_{n+1}$ , where  $\mathscr{V}_j = \{V_{j,i}\}_{i=1}^k$ , such that  $\operatorname{ord} \mathscr{V}_j \le 0$  and  $V_{j,i} \subset U_i$  for  $i = 1, 2, \ldots, k$  and  $j = 1, 2, \ldots, n+1$ .

**3.2.J.** (a) Prove that a completely regular space X satisfies the inequality  $\dim X \leq n$  if and only if every (n+2)-element functionally open cover  $\{U_i\}_{i=1}^{n+2}$  of the space X has a functionally open shrinking  $\{W_i\}_{i=1}^{n+2}$  of order  $\leq n$ , i.e., such that  $\bigcap_{i=1}^{n+2} W_i = \emptyset$ .

Hint. See the proof of Theorem 1.6.10.

(b) Prove that a completely regular space X satisfies the inequality  $\dim X \leq n \geq 0$  if and only if for every finite family  $\{U_i\}_{i=1}^k$  of functionally open subsets of X and every family  $\{F_i\}_{i=1}^k$  of functionally closed subsets of X such that  $F_i \subset U_i$  for i = 1, 2, ..., k there exist such families  $\{E_i\}_{i=1}^k$ 

and  $\{V_i\}_{i=1}^k$  of functionally closed and functionally open subsets of X, respectively, that  $F_i \subset V_i \subset E_i \subset U_i$  for i = 1, 2, ..., k and  $\operatorname{ord}(\{E_i \setminus V_i\}_{i=1}^k) \leq n-1$ .

(c) Prove that a completely regular space X satisfies the inequality  $\dim X \leq n \geq 0$  if and only if for every sequence  $(A_1, B_1), (A_2, B_2), \ldots$ ...,  $(A_{n+1}, B_{n+1})$  of n+1 pairs of disjoint functionally closed subsets of X there exist functionally closed sets  $L_1, L_2, \ldots, L_{n+1}$  such that  $L_i$  is a partition between  $A_i$  and  $B_i$  and  $\bigcap_{i=1}^{n+1} L_i = \emptyset$ .

*Hint.* Observe that if L is a functionally closed subset of a space X and U,  $W \subset X$  are disjoint open sets satisfying the equality  $X \ L = U \cup W$ , then U and W are functionally open.

**3.2.K.** (a) (Smirnov [1956a]) Prove that a completely regular space X satisfies the inequality dim  $X \le n \ge 0$  if and only if for every closed subspace A of the space X and each continuous mapping  $f: A \to S^n$  which is the restriction of a continuous mapping  $\overline{f}: X \to B^{n+1}$  there exists a continuous extension  $F: X \to S^n$  of f over X.

(b) Prove that a completely regular space X satisfies the inequality  $\dim X \leq n$  if and only if for every finite functionally open cover  $\mathscr{E}$  of the space X there exists an  $\mathscr{E}$ -mapping of X to a polyhedron of dimension  $\leq n$ .

## 3.3. The compactification and the universal space theorems for the dimension dim. The dimension dim and inverse systems of compact spaces

In the proofs of the compactification and the universal space theorems we shall apply Mardešić's factorization theorem, which asserts that every continuous mapping  $f: X \to Y$  of a compact space X to a compact space Y can be let through an intermediate space Z such that dim  $Z \leq \dim X$  and  $w(Z) \leq w(Y)$ . We start with a simple lemma on normal spaces. Let us recall that a cover  $\mathscr{B}$  is a *star refinement* of another cover  $\mathscr{A}$  of the same space if for every  $B \in \mathscr{B}$  there exists an  $A \in \mathscr{A}$  such that  $St(B, \mathscr{B}) \subset A$ , where  $St(B, \mathscr{B})$  denotes the *star of the set B with respect to the cover*  $\mathscr{B}$ , i.e., the set  $\bigcup \{B' \in \mathscr{B}: B \cap B' \neq \emptyset\}$ .

**3.3.1. Lemma.** Every finite open cover of a normal space has a finite open star refinement.

**Proof.** Let  $\mathscr{U} = \{U_i\}_{i=1}^k$  be a finite open cover of a normal space X. Consider a closed shrinking  $\{F_i\}_{i=1}^k$  of  $\{U_i\}_{i=1}^k$ . By Urysohn's lemma, for i = 1, 2, ..., k there exists a continuous function  $f_i: X \to I$  such that  $f_i(X \setminus U_i) \subset \{0\}$  and  $f_i(F_i) \subset \{1\}$ . Define  $I_0 = [0, 1/2), I_1 = (1/4, 3/4)$  and  $I_2 = (1/2, 1]$ . The family  $\mathscr{V}$  of all sets

(1) 
$$V_{j_1, j_2, \dots, j_k} = f_1^{-1}(I_{j_1}) \cap f_2^{-1}(I_{j_2}) \cap \dots \cap f_k^{-1}(I_{j_k}),$$

where  $j_i \in \{0, 1, 2\}$  for i = 1, 2, ..., k, is a finite open cover of the space X. For every non-empty  $V = V_{m_1, m_2, ..., m_k} \in \mathscr{V}$  there exists an  $i \leq k$  such that  $V \cap F_i \neq \emptyset$ , and we clearly have  $m_i = 2$ . If a set of form (1) intersects V, then  $j_i$  is equal to 1 or 2, so that  $\operatorname{St}(V, \mathscr{V}) \subset U_i$ . Thus  $\mathscr{V}$  is a star refinement of  $\mathscr{U}$ .  $\Box$ 

**3.3.2. Mardešić's factorization theorem.** For every continuous mapping  $f: X \to Y$  of a compact space X to a compact space Y there exist a compact space Z and continuous mappings  $g: X \to Z$  and  $h: Z \to Y$  such that dim  $Z \leq \dim X$ ,  $w(Z) \leq w(Y)$ , g(X) = Z and f = hg.

**Proof.** If dim  $X = \infty$  or  $w(Y) < \aleph_0$ , then Z = f(X), g = f and  $h = id_Z$  satisfy the theorem. Thus one can suppose that dim  $X = n < \infty$  and  $w(Y) = m \ge \aleph_0$ . We shall define inductively a sequence  $W_0, W_1, \ldots$  of classes of finite open covers of the space X. Consider a base  $\mathscr{B}$  for the space Y such that  $|\mathscr{B}| = m$  and denote by  $W_0$  the class of all finite covers of the space X by members of the family  $f^{-1}(\mathscr{B})$ . Clearly, for each  $x, y \in X$ 

(2) if  $f(x) \neq f(y)$ , then there exists a  $\mathscr{W} \in W_0$  such that  $y \notin St(x, \mathscr{W})$ .

Assume now that the classes  $W_i$  are defined for all i < k. By virtue of Lemma 3.3.1 and the inequality dim $X \le n$ , for each pair of covers  $\mathscr{W}$ ,  $\mathscr{W}' \in W_{k-1}$  we can choose a finite open star refinement of  $\mathscr{W} \land \mathscr{W}'$  which has order *n*; let  $W_k$  be the class of all covers of the space X thus obtained. In this way the sequence  $W_0, W_1, \ldots$  is defined. Since  $|W_0| \le m$ , we have  $|W_i| \le m$  for  $i = 1, 2, \ldots$  Let  $W = \bigcup_{i=1}^{\infty} W_i$ ; clearly,  $|W| \le m$ . For  $x, y \in X$  define

(3) xEy if and only if for every cover  $\mathscr{W} \in W$  there exists a set  $U \in \mathscr{W}$  such that  $x, y \in U$ .

We shall show that E is an equivalence relation on the space X. It follows directly from the definition that the relation E is reflexive and symmetric, so that it remains to show that if xEy and yEz, then xEz. To this end, it suffices to note that

(4) if  $\mathscr{W}'$  is a star refinement of  $\mathscr{W}$ ,  $y \in St(x, \mathscr{W}')$  and  $z \in St(y, \mathscr{W}')$ , then  $z \in St(x, \mathscr{W})$ ,

because for every  $\mathscr{W} \in W$  there exists a  $\mathscr{W}' \in W$  which is a star refinement of  $\mathscr{W}$ . The relation E determines a decomposition of the space X into equivalence classes; from (3) it follows that

(5) 
$$[x] = \bigcap_{\mathscr{W} \in \mathscr{W}} \operatorname{St}(x, \mathscr{W}) \quad \text{for } x \in X,$$

where [x] denotes the equivalence class that contains x.

We shall show now that the equivalence relation E is closed. Thus, we have to show that for every open set  $U \subset X$  the union of all equivalence classes that are contained in U is an open subset of X. In view of a subsequent application, we shall show a little more, viz., that

(6) for every open set U ⊂ X and each equivalence class [x] ⊂ U there exist an open set V ⊂ X and a cover W ∈ W such that [x] ⊂ V ⊂ ∪ [y] ⊂ St(V, W) ⊂ U.

To begin with, let us note that if  $\mathscr{W}'$  is a star refinement of  $\mathscr{W}$ , then  $\overline{\operatorname{St}(x, \mathscr{W}')} \subset \operatorname{St}(x, \mathscr{W})$ , so that from (5) it follows that  $[x] = \bigcap_{\mathscr{W} \in W} \overline{\operatorname{St}(x, \mathscr{W})}$ . Now, the space X being compact, there exists a finite number of covers  $\mathscr{W}_1, \mathscr{W}_2, \ldots, \mathscr{W}_k \in \mathscr{W}$  such that  $\bigcap_{i=1}^k \operatorname{St}(x, \mathscr{W}_i) \subset U$ . Consider a cover  $\mathscr{W}_0 \in \mathscr{W}$  which refines all the covers  $\mathscr{W}_i$  and a star refinement  $\mathscr{W} \in \mathscr{W}$ of the cover  $\mathscr{W}_0$ . Clearly, the open set  $V = \operatorname{St}(x, \mathscr{W})$  satisfies the relation  $[x] \subset V \subset \bigcup_{y \in V} [y]$ ; the penultimate inclusion in (6) follows from (5), and the last inclusion is a consequence of the relation  $\operatorname{St}(x, \mathscr{W}_0) \subset U$ and (4) with  $\mathscr{W}' = \mathscr{W}$  and  $\mathscr{W} = \mathscr{W}_0$ .

Let Z be the quotient space X/E and  $g: X \to Z$  the natural quotient mapping. As E is a closed equivalence relation, the space Z is compact (see [GT], Theorem 3.2.11). By virtue of (2), f(x) = f(y) whenever xEy,

so that by letting h([x]) = f(x) we define a mapping h of Z to Y; from the relation hg = f it follows that  $h: Z \to Y$  is a continuous mapping.

For every open set  $U \subset X$  let  $U^* = Z \setminus g(X \setminus U)$  and for every  $\mathscr{W} \in W$ let  $\mathscr{W}^* = \{W^* \colon W \in \mathscr{W}\}$ . Clearly,  $g^{-1}(U^*) \subset U$  and  $U^*$  is open in Z. If  $\mathscr{W}'$  is a star refinement of  $\mathscr{W}$ , then for each  $x \in X$  there exists a  $W \in \mathscr{W}$ such that  $[x] \subset St(x, \mathscr{W}') \subset W$ ; thus for every  $\mathscr{W} \in W$  the family  $\mathscr{W}^*$ is a finite open cover of the space Z and  $\operatorname{ord} \mathscr{W}^* \leq n$ .

Now we shall show that every finite open cover  $\{U_i\}_{i=1}^k$  of the space Z has a refinement of the form  $\mathscr{W}^*$ , where  $\mathscr{W} \in \mathscr{W}$ . For each  $x \in X$  there exists an  $i \leq k$  such that  $[x] \subset g^{-1}(U_i)$ , and by virtue of (6) there exist a neighbourhood  $V_x \subset X$  of the point x and a cover  $\mathscr{W}(x) \in \mathscr{W}$  such that  $\operatorname{St}(V_x, \mathscr{W}(x)) \subset g^{-1}(U_i)$ . The open cover  $\{V_x\}_{x \in X}$  of the space X has a finite refinement  $\{V_{x_i}\}_{i=1}^m$ . Consider a cover  $\mathscr{W} \in \mathscr{W}$  which refines all the covers  $\mathscr{W}(x_i)$ . For every  $W \in \mathscr{W}$  there exists an  $i \leq k$  such that  $W \subset g^{-1}(U_i)$ . Since the last inclusion implies that  $W^* \subset U_i$ , it follows that the cover  $\mathscr{W}^*$  is a refinement of  $\{U_i\}_{i=1}^k$ .

Thus we have shown that  $\dim Z \leq n$  and that the family  $\mathscr{D} = \bigcup_{\mathcal{W} \in \mathcal{W}} \mathscr{W}^*$ is a base for the space Z. To complete the proof it suffices to observe that  $w(Z) \leq |\mathscr{D}| \leq \mathfrak{m} \cdot \aleph_0 = \mathfrak{m} = w(Y)$ .  $\Box$ 

**3.3.3. The compactification theorem for** dim. For every normal space X there exists a compactification preserving both the dimension dim and weight, *i.e.*, a compact space  $\tilde{X}$  which contains a dense subspace homeomorphic to X and satisfies the inequalities dim $\tilde{X} \leq \dim X$  and  $w(\tilde{X}) \leq w(X)$ .

**Proof.** We can suppose that  $\dim X = n < \infty$  and  $w(X) = m \ge \aleph_0$ . Consider a homeomorphic embedding  $i: X \to I^m$  of the space X in the Tychonoff cube  $I^m$  of weight m; let  $f: \beta X \to I^m$  be the extension of i over  $\beta X$ . By virtue of Theorem 3.3.2 there exist a compact space  $\tilde{X}$  and continuous mappings  $g: \beta X \to \tilde{X}$  and  $h: \tilde{X} \to I^m$  such that  $\dim \tilde{X} \le \dim \beta X = \dim X$ ,  $w(\tilde{X}) \le w(I^m) = m$  and f = hg. The composition  $h_0g_0$  of the restrictions  $g_0 = g|X: X \to g(X) \subset \tilde{X}$  and  $h_0 = h|g(X): g(X) \to i(X) \subset I^m$  is a homeomorphism, so that  $g_0$  is also a homeomorphism. Thus  $\tilde{X}$  is the required compactification of the space X.  $\square$ 

**3.3.4. The universal space theorem for** dim. For every integer  $n \ge 0$  and every cardinal number  $m \ge \aleph_0$  there exists a compact universal space  $P_m^n$ 

for the class of all normal spaces whose covering dimension is not larger than n and whose weight is not larger than m.

**Proof.** Let  $\{X_s\}_{s\in S}$  be the family of all normal subspaces of the Tychonoff cube  $I^m$  whose covering dimension is not larger than n, and let  $i_s: X_s \to I^m$ be the embedding of  $X_s$  in  $I^m$ . Consider the sum  $X = \bigoplus_{s\in S} X_s$  and the mapping  $i: X \to I^m$  defined by letting  $i(x) = i_s(x)$  for  $x \in X_s$ ; let  $f: \beta X \to I^m$  be the extension of i over  $\beta X$ . By virtue of Theorem 3.3.2 there exist a compact space  $P^n_m$  and continuous mappings  $g: \beta X \to P^n_m$  and  $h: P^n_m \to I^m$  such that dim  $P^n_m \leq \dim \beta X = \dim X = n$ ,  $w(P^n_m) \leq w(I^m) = m$  and f = hg.

Consider now an arbitrary normal space Y such that dim  $Y \leq n$  and  $w(Y) \leq m$ . Since Y is embeddable in  $I^m$ , there exists an  $s \in S$  such that  $X_s$  is homeomorphic to Y. The composition  $h_0g_0$  of the restrictions  $g_0 = g|X_s: X_s \to g(X_s) \subset P_m^n$  and  $h_0 = h|g(X_s): g(X_s) \to X_s \subset I^m$  is a homeomorphism, so that  $g_0$  is also a homeomorphism. Thus  $P_m^n$  is the required universal space.  $\Box$ 

We now turn to a study of inverse systems of compact spaces from the dimensional standpoint. To begin with, let us recall that Theorem 1.13.2 established in Chapter 1 states that for every compact metric space X such that dim  $X \leq n$  there exists an inverse sequence  $\{K_i, \pi_j^i\}$  consisting of polyhedra of dimension  $\leq n$  whose limit is homeomorphic to X. In Example 3.3.8 below we show that the compact space X described in Example 2.2.13 cannot be represented as the limit of an inverse system of polyhedra of dimension 1, although dim X = 1. Hence, Theorem 1.13.2 in its original form does not extend to arbitrary compact spaces; yet we have the following

**3.3.5. Theorem on expansion in an inverse system.** For every compact space X such that dim  $X \leq n$  there exists an inverse system  $S = \{X_{\sigma}, \pi_{\varrho}^{\sigma}, \Sigma\}$ , where  $|\Sigma| \leq w(X)$ , consisting of metrizable compact spaces of dimension  $\leq n$  whose limit is homeomorphic to X.

**Proof.** We can suppose that  $w(X) = \mathfrak{m} \ge \aleph_0$ . Let  $h: X \to I^{\mathfrak{m}} = \prod_{s \in S} I_s$ , where  $I_s = I$  for  $s \in S$  and  $|S| = \mathfrak{m}$ , be a homeomorphic embedding of the space X in the Tychonoff cube  $I^{\mathfrak{m}}$  of weight  $\mathfrak{m}$ . For i = 1, 2, ... denote by  $\Sigma_t$  the family of all subsets of S that have exactly *i* elements; the union

 $\Sigma = \bigcup_{i=1}^{\infty} \Sigma_i$  is directed by inclusion, i.e., the relation  $\leq$  defined by letting  $\varrho \leq \sigma$  if and only if  $\varrho \subset \sigma$ , and has cardinality m = w(X).

Applying induction with respect to *i*, we shall now define for each  $\sigma \in \Sigma_i$  a metrizable compact space  $X_{\sigma}$  such that dim  $X_{\sigma} \leq n$  if i > 1 and continuous mappings  $\pi_{\varrho}^{\sigma} \colon X_{\sigma} \to X_{\varrho}$ , where  $\varrho \leq \sigma$ , such that

(7) 
$$\pi_{\tau}^{\varrho} \pi_{\varrho}^{\sigma} = \pi_{\tau}^{\sigma}$$
 whenever  $\tau \leq \varrho \leq \sigma$  and  $\pi_{\sigma}^{\sigma} = \mathrm{id}_{X_{\sigma}}$ ;

at the same time we shall define continuous mappings  $g_{\sigma}: X \to X_{\sigma}$  satisfying

(8) 
$$g_{\sigma}(X) = X_{\sigma}$$
 and  $\pi_{\varrho}^{\sigma}g_{\sigma} = g_{\varrho}$  whenever  $\varrho \leq \sigma$ .

For i = 1 all conditions are satisfied if for each  $\sigma = \{s\} \in \Sigma_1$  we let  $X_{\sigma} = p_s h(X) \subset I_s$ , where  $p_s: \prod_{s \in S} I_s \to I_s$  is the projection,  $\pi_{\sigma}^{\sigma} = \operatorname{id}_{X_{\sigma}}$  and  $g_{\sigma}^{\gamma} = p_s h$ . Assume that the spaces X and the mappings  $\pi_{\varrho}^{\sigma}$  and  $g_{\sigma}$  satisfying (7) and (8) are defined for all  $\sigma \in \bigcup_{i=1}^{k-1} \Sigma_i$ , where k > 1, and consider a set  $\sigma \in \Sigma_k$ . Let  $\sigma_1, \sigma_2, \ldots, \sigma_k$  be all (k-1)-element subsets of  $\sigma$  and let  $f_{\sigma}$ :  $X \to X_{\sigma_1} \times X_{\sigma_2} \times \ldots \times X_{\sigma_k}$  be the continuous mapping defined by the formula  $f_{\sigma}(x) = (g_{\sigma_1}(x), g_{\sigma_2}(x), \ldots, g_{\sigma_k}(x))$ . By virtue of Theorem 3.3.2 there exist a compact space  $X_{\sigma}$  and continuous mappings  $g_{\sigma}: X \to X_{\sigma}$  and  $h_{\sigma}: X_{\sigma} \to X_{\sigma_1} \times X_{\sigma_2} \times \ldots \times X_{\sigma_k}$  such that  $\dim X_{\sigma} \leqslant n$ ,  $w(X_{\sigma}) \leqslant \aleph_0$ ,  $g_{\sigma}(X) = X_{\sigma}$  and  $f_{\sigma} = h_{\sigma}g_{\sigma}$ . The last equality means that

(9) 
$$\pi_{\sigma_i}^{\sigma}g_{\sigma} = g_{\sigma_i} \quad \text{for } i = 1, 2, ..., k,$$

where  $\pi_{\sigma_i}^{\sigma}: X_{\sigma} \to X_{\sigma_i}$  is the composition of  $h_{\sigma}$  and the projection of  $X_{\sigma_i} \times X_{\sigma_a} \times \ldots \times X_{\sigma_k}$  onto  $X_{\sigma_i}$ . As the space  $X_{\sigma}$  is compact and has a countable weight, it is a metrizable space. For each  $\varrho \in \bigcup_{i=1}^{k-1} \Sigma_i$  satisfying  $\varrho \leq \sigma$  there exists at least one  $i \leq k$  such that  $\varrho \leq \sigma_i \leq \sigma$ . Let us observe that the composition  $\pi_{\varrho}^{\sigma_i} \pi_{\sigma_i}^{\sigma}$  does not depend on the choice of a particular  $\sigma_i$  satisfying  $\varrho \leq \sigma_i$ . Indeed, if for a  $j \leq k$  we also have  $\varrho \leq \sigma_j$ , then

$$\pi_{\varrho}^{\sigma_i}\pi_{\sigma_i}^{\sigma}g_{\sigma}=\pi_{\varrho}^{\sigma_i}g_{\sigma_i}=g_{\varrho}=\pi_{\varrho}^{\sigma_j}g_{\sigma_j}=\pi_{\varrho}^{\sigma_j}\pi_{\sigma_j}^{\sigma}g_{\sigma},$$

which implies that

(10) 
$$\pi_{\varrho}^{\sigma_{i}}\pi_{\sigma_{i}}^{\sigma}=\pi_{\varrho}^{\sigma_{j}}\pi_{\sigma_{j}}^{\sigma},$$

because  $g_{\sigma}(X) = X_{\sigma}$ . In accordance with the above observation, for each  $\varrho \in \bigcup_{i=1}^{k-1} \Sigma_i$  satisfying  $\varrho \leq \sigma$  we define  $\pi_{\varrho}^{\sigma} = \pi_{\varrho}^{\sigma_i} \pi_{\sigma_i}^{\sigma}$  where  $\varrho \leq \sigma_i \leq \sigma$ . More-

over, we let  $\pi_{\sigma}^{\sigma} = id_{X_{\sigma}}$ . From (9), (10) and the inductive assumption it follows that the space  $X_{\sigma}$  and the mappings  $\pi_{e}^{\sigma}$  and  $g_{\sigma}$  satisfy (7) and (8) for  $\sigma \in \Sigma_{k}$ . Thus, we have defined metrizable compact spaces  $X_{\sigma}$  such that  $\dim X_{\sigma} \leq n$  if i > 1 and continuous mappings  $\pi_{e}^{\sigma}$  and  $g_{\sigma}$  satisfying (7) and (8) for  $\sigma \in \Sigma$  and  $\varrho \leq \sigma$ .

It follows from (7) that  $S = \{X_{\sigma}, \pi_{\sigma}^{\sigma}, \Sigma\}$  is an inverse system; we shall show that X is homeomorphic to the limit  $\lim_{\sigma \in \Sigma} S$ . In view of the second equality in (8), for every  $x \in X$  the point  $\{g_{\sigma}(x)\} \in \prod_{\sigma \in \Sigma} X_{\sigma}$  is a thread of S; by assigning this thread to x we define a mapping  $g: X \to \lim_{\sigma \in S} S$ . Since  $\pi_{\sigma}g = g_{\sigma}$  for every  $\sigma \in \Sigma$ , where  $\pi_{\sigma}: \lim_{\sigma \in S} S \to X_{\sigma}$  is the projection, the mapping g is continuous. For each  $\sigma = \{s\} \in \Sigma_1$  we have  $\pi_{\sigma}g = g_{\sigma} = p_s h$ , so that, the mapping h being one-to-one, the mapping g is also one-to-one. Finally, as  $g_{\sigma}(X) = X_{\sigma}$  for every  $\sigma \in \Sigma$ , the mapping g maps X onto  $\lim_{\sigma \in S} S$  (see [GT], Corollary 3.2.16). Thus g is a homeomorphism of X onto  $\lim_{\sigma \in S} S$ .

If  $n \ge 1$ , the system S satisfies all the required conditions; if n = 0, it has to be replaced by the system  $\{X_{\sigma}, \pi_{\varrho}^{\sigma}, \Sigma \setminus \Sigma_1\}$ , because, in general, the inequality dim  $X_{\sigma} \le 0$  does not hold for  $\sigma \in \Sigma_1$  (cf. Problem 3.3.A).

Let us observe that from (8) and (9) it follows that the bonding mappings in the inverse system S in Theorem 3.3.5 are mappings onto.

We shall now prove the following

**3.3.6. Theorem on the dimension of the limit of an inverse system.** If the inverse system  $S = \{X_{\sigma}, \pi_{\varrho}^{\sigma}, \Sigma\}$  consists of compact spaces  $X_{\sigma}$  such that  $\dim X_{\sigma} \leq n$  for  $\sigma \in \Sigma$ , then the limit  $X = \lim_{t \to \infty} S$  satisfies the inequality  $\dim X \leq n$ .

**Proof.** Consider a finite open cover  $\mathscr{U}$  of the space X. The space X being compact (see [GT], Theorem 3.2.13), the cover  $\mathscr{U}$  has a finite refinement of the form  $\{\pi_{\sigma_i}^{-1}(U_i)\}_{i=1}^k$ , where  $\pi_{\sigma_i}: X \to X_{\sigma_i}$  is the projection and  $U_i$ is an open subset of  $X_{\sigma_i}$  for i = 1, 2, ..., k. Let  $\sigma$  be an arbitrary element of  $\Sigma$  such that  $\sigma_i \leq \sigma$  for i = 1, 2, ..., k and let  $W_i = (\pi_{\sigma_i}^{\sigma})^{-1}(U_i)$ . One readily sees that the family  $\{\pi_{\sigma}^{-1}(W_i)\}_{i=1}^k$  is an open refinement of the cover  $\mathscr{U}$ . Since  $\pi_{\sigma}(X)$  is a closed subspace of  $X_{\sigma}$ , dim $\pi_{\sigma}(X) \leq n$  and the open cover  $\{\pi_{\sigma}(X) \cap W_i\}_{i=1}^k$  of the space  $\pi_{\sigma}(X)$  has an open shrinking  $\{V_i\}_{i=1}^k$  of order  $\leq n$ . The family  $\{\pi_{\sigma}^{-1}(V_i)\}_{i=1}^k$  is an open cover of the space X which refines  $\mathscr{U}$  and has order  $\leq n$ . Thus dim $X \leq n$ .  $\Box$ 

Theorems 3.3.5 and 3.3.6 yield the following

**3.3.7. Theorem on inverse systems.** A compact space X satisfies the inequality  $\dim X \leq n$  if and only if X is homeomorphic to the limit of an inverse system consisting of metrizable compact spaces of dimension  $\leq n$ .

We conclude this section with the above-mentioned example of a compact space X which satisfies the equality  $\dim X = 1$  and yet is not homeomorphic to the limit of an inverse system of polyhedra of dimension  $\leq 1$ .

**3.3.8. Example.** The space X described in Example 2.2.13 is compact and satisfies the relation dimX = 1 < indX (see Example 3.1.31). Thus, to show that X has the required property it suffices to prove that for every inverse system  $S = \{K_{\sigma}, \pi_{\varrho}^{\sigma}, \Sigma\}$  consisting of polyhedra of dimension  $\leq 1$  the limit  $K = \lim S$  satisfies the inequality ind  $K \leq 1$ .

Consider a point  $x \in K$  and a neighbourhood  $V \subset K$  of the point x. There exists a  $\sigma_0 \in \Sigma$  and such a neighbourhood  $U_{\sigma_0}$  of the point  $\pi_{\sigma_0}(x)$  in the space  $K_{\sigma_0}$ , where  $\pi_{\sigma_0} : K \to K_{\sigma_0}$  is the projection, that the set  $U = \pi_{\sigma_0}^{-1}(U_{\sigma_0})$  satisfies the relation  $x \in U \subset V$ . Define  $\Sigma_0 = \{\sigma \in \Sigma : \sigma_0 \leq \sigma\}$  and let  $U_{\sigma} = (\pi_{\sigma_0}^{\sigma})^{-1}(U_{\sigma_0})$  and  $F_{\sigma} = \operatorname{Fr} U_{\sigma}$  for  $\sigma \in \Sigma_0$ . Since for each  $\sigma, \varrho \in \Sigma_0$  satisfying  $\varrho \leq \sigma$  we have

$$\begin{aligned} \pi_{\varrho}^{\sigma}(F_{\sigma}) &= \pi_{\varrho}^{\sigma}(\overline{U_{\sigma}} \cap \overline{K_{\sigma} \setminus U_{\sigma}}) \subset \overline{\pi_{\varrho}^{\sigma}(U_{\sigma})} \cap \overline{\pi_{\varrho}^{\sigma}(K_{\sigma} \setminus U_{\sigma})} \\ &= \pi_{\varrho}^{\sigma}(\pi_{\sigma_{0}}^{\sigma})^{-1}(\overline{U_{\sigma_{0}}}) \cap \overline{\pi_{\varrho}^{\sigma}(\pi_{\sigma_{0}}^{\sigma})^{-1}(K_{\sigma_{0}} \setminus U_{\sigma_{0}})} \\ &\quad \subset (\overline{\pi_{\sigma_{0}}^{\varrho})^{-1}(U_{\sigma_{0}})} \cap (\overline{\pi_{\sigma_{0}}^{\varrho})^{-1}(K_{\sigma_{0}} \setminus U_{\sigma_{0}})} = \overline{U_{\varrho}} \cap \overline{K_{\varrho} \setminus U_{\varrho}} = F_{\varrho}, \end{aligned}$$

the family  $S_0 = \{F_{\sigma}, \tilde{\pi}_{\varrho}^{\sigma}, \Sigma_0\}$ , where  $\tilde{\pi}_{\varrho}^{\sigma}: F_{\sigma} \to F_{\varrho}$  is defined by letting  $\tilde{\pi}_{\varrho}^{\sigma}(x) = \pi_{\varrho}^{\sigma}(x)$ , is an inverse system of compact spaces. Now, from Theorems 1.8.12 and 1.3.1 it follows that dim  $F_{\sigma} \leq 0$  for  $\sigma \in \Sigma_0$  so that dim  $\lim S_0 \leq 0$  by virtue of Theorem 3.3.6. One readily checks that  $\pi_{\sigma}(\operatorname{Fr} U) \subset F_{\sigma}$  for every  $\sigma \in \Sigma_0$ ; thus  $\operatorname{Fr} U \subset \lim S_0$  and by virtue of Theorem 3.1.30 we have ind  $\operatorname{Fr} U \leq 0$ . Hence we have proved that  $\operatorname{ind} K \leq 1$ .  $\Box$ 

Let us note in connection with the last example that there exists a compact space X with a similar property which satisfies the equality ind X = Ind X= dim X = 1.

Let us also note that every compact space is homeomorphic to the limit of an inverse system consisting of polyhedra (see Problem 3.3.D) and that one can define a compact space X such that indX = IndX = dimX = 1 which is not homeomorphic to the limit of an inverse system consisting of polyhedra (or, more generally, of locally connected metrizable compact spaces) whose bonding mappings are mappings onto.
#### Problems

### Historical and bibliographic notes

Theorem 3.3.2 was established by Mardešić in [1960]; the present proof was given by Arhangel'skiĭ in [1967]. Theorem 3.3.3 was proved by Skljarenko in [1958]. Theorem 3.3.4 was established independently by Pasynkov in [1964] and by Zarelua in [1964]. Theorem 3.3.5 was given by Mardešić in [1960]; Theorem 3.3.6 is implicit in Freudenthal's paper [1937]. Example 3.3.8 was given independently by Pasynkov in [1958] and by Mardešić in [1960]. Both examples cited at the end of this section can be found in Pasynkov's paper [1962].

#### Problems

**3.3.A.** Observe that Theorem 3.3.5 for n = 0 easily follows from the fact that every normal space X such that dimX = 0 is embeddable in a Cantor cube (see Remark 1.3.18 and Theorem 1.6.11).

**3.3.B** (Pasynkov [1962]). Let  $S = \{X_{\sigma}, \pi_{\varrho}^{\sigma}, \Sigma\}$  be an inverse system of compact spaces and let  $X = \lim S$ . Prove that dim  $X \leq n$  if and only if for each  $\varrho \in \Sigma$  and every finite open cover  $\{U_i\}_{i=1}^k$  of the space  $X_{\varrho}$  there exists a  $\sigma \in \Sigma$  satisfying  $\varrho \leq \sigma$  and such that the cover  $\{(\pi_{\varrho}^{\sigma})^{-1}(U_i)\}_{i=1}^k$  of the space  $X_{\sigma}$  has a finite open refinement  $\mathscr{V}$  satisfying the inequality  $\operatorname{ord}(\mathscr{V}|\pi_{\sigma}(X)) \leq n$ , where  $\pi_{\sigma}: X \to X_{\sigma}$  is the projection.

**3.3.C** (Pasynkov [1958]). Prove that for every inverse system  $S = \{K_{\sigma}, \pi_{\varrho}^{\sigma}, \Sigma\}$  consisting of polyhedra of dimension  $\leq 1$  the limit  $K = \lim S$  satisfies the inequality  $\operatorname{Ind} K \leq 1$ .

*Remark.* It is not known if the number 1 in Problem 3.3.C can be replaced by an arbitrary natural number.

**3.3.D** (Eilenberg and Steenrod [1952]). Prove that for every compact space X there exists an inverse system  $S = \{X_{\sigma}, \pi_{\varrho}^{\sigma}, \Sigma\}$  consisting of polyhedra whose limit is homeomorphic to X.

*Hint*. Embed the space X in a Tychonoff cube.

## CHAPTER 4

## DIMENSION THEORY OF METRIZABLE SPACES

In the realm of metrizable spaces the dimensions Ind and dim coincide. Thus in metrizable spaces both the theorems which depend on the dimension Ind and the theorems which depend on the dimension dim are valid. It will appear in the course of this chapter that the dimension theory of metrizable spaces is by no means inferior to the classical dimension theory of separable metric spaces developed in the first chapter of this book.

The present chapter can be read almost directly after Chapter 1. The results of Chapter 2 are not used here, except for Lemma 2.3.16 which belongs to general topology rather than to dimension theory. From Chapter 3 we use only the beginning of Section 3.1 up to Theorem 3.1.10 and also Theorems 3.1.28, 3.1.29, 3.2.2 and 3.2.5; the last theorem is not used until Section 4.2.

In Section 4.1 the most important properties of dimension in metrizable spaces are established. We start with the Katětov-Morita theorem on the coincidence of Ind and dim and then prove the counterparts of the theorems obtained for separable metric spaces in Section 1.5.

Section 4.2 begins with two characterizations of the dimension dim in metrizable spaces, one stated in terms of special bases and the other in terms of sequences of covers. Then we discuss briefly some characterizations of dim formulated in terms of special metrics. In the second part of the section, we prove by applying an appropriate factorization theorem the existence of a universal space for the class of all metrizable spaces whose dimension is not larger than n and whose weight is not larger than m.

Section 4.3 resumes the considerations of Section 1.12. We generalize in it the theorems on dimension-raising and dimension-lowering mappings established in Chapter 1 and prove two theorems of more special character on the relations between the dimensions of the domain and the range of a closed mapping.

Let us add that the theorems on partitions and on extending mappings

to spheres established in Sections 1.7 and 1.9 extend to all normal, and, *a fortiori*, to all metrizable spaces; the proofs were given in Section 3.2 (cf. Problems 4.1.E and 4.3.B).

## 4.1. Basic properties of dimension in metrizable spaces

We start with one of the most important results in dimension theory, viz., with the theorem on the coincidence of the dimensions Ind and dim in metrizable spaces. In the proof we shall apply two characterizations of the dimension dim in the class of metrizable spaces which are established in Proposition 4.1.2 below (cf. Problem 4.1.A(b)).

**4.1.1. Lemma.** Let X be a normal space. If there exists a sequence  $\mathcal{W}_1, \mathcal{W}_2, ...$  of open covers of the space X such that  $\operatorname{ord} \mathcal{W}_i \leq n$  and  $\mathcal{W}_{i+1}$  is a refinement of  $\mathcal{W}_i$  for i = 1, 2, ..., and the family  $\{\operatorname{St}(W, \mathcal{W}_i) : W \in \mathcal{W}_i, i = 1, 2, ...\}$  is a base for X, then  $\dim X \leq n$ .

**Proof.** For i = 1, 2, ... let  $f_i^{i+1}$  be a mapping of  $\mathscr{W}_{i+1}$  to  $\mathscr{W}_i$  such that  $W \subset f_i^{i+1}(W)$  for each  $W \in \mathscr{W}_{i+1}$ ; let  $f_i^k = f_i^{i+1}f_{i+1}^{i+2} \dots f_{k-1}^k$  for i < k and let  $f_i^i = \operatorname{id}_{\mathscr{W}_i}$  for  $i = 1, 2, \dots$  Obviously

(1) 
$$W \subset f_i^k(W)$$
 for each  $W \in \mathscr{W}_k$  and  $i \leq k$ .

Consider a finite open cover  $\{H_j\}_{j=1}^l$  of the space X. The sets  $X_1, X_2, ...,$  where

(2) 
$$X_k = \bigcup \{ W \in \mathscr{W}_k \colon \operatorname{St}(W, \mathscr{W}_k) \subset H_j \text{ for a } j \leq l \},$$

form an open cover of the space X. For k = 1, 2, ... define the subfamilies

$$\mathscr{U}_{k} = \{ U \in \mathscr{W}_{k} : U \cap X_{k} \neq \emptyset \} \text{ and } \mathscr{V}_{k} = \{ V \in \mathscr{U}_{k} : V \cap (\bigcup_{j < k} X_{j}) = \emptyset \}$$

of the cover  $\mathscr{W}_k$ , and for every  $U \in \mathscr{U}_k$  denote by i(U) the largest integer  $\leq k$  satisfying

(3) 
$$f_{i(U)}^k(U) \in \mathscr{V}_{i(U)};$$

such an integer does exist because  $f_1^k(U) \cap (\bigcup_{j < 1} X_j) = \emptyset$  and  $f_k^k(U) \cap X_k = U \cap X_k \neq \emptyset$ .

For every  $V \in \mathscr{V}_i$  consider the open set

(4) 
$$V^* = \bigcup_{k=i}^{\infty} \left[ \bigcup \left\{ U \cap X_k \colon U \in \mathscr{U}_k, f_i^k(U) = V \text{ and } i(U) = i \right\} \right];$$

as  $V \cap X_i = \emptyset$ , by virtue of (2) there exist a  $W \in \mathcal{W}_i$  such that  $V \cap W \neq \emptyset$ and a  $j(V) \leq l$  satisfying  $V \subset St(W, \mathcal{W}_i) \subset H_{j(V)}$ . From (1) it follows that  $V^* \subset V$ , so that  $V^* \subset H_{j(V)}$ . Since  $\mathcal{V}_i \cap \mathcal{V}_j = \emptyset$  whenever  $i \neq j$ , for every  $V \in \mathcal{V} = \bigcup_{i=1}^{\infty} \mathcal{V}_i$  the set  $V^*$  and the integer j(V) are well defined.

To complete the proof it suffices to show that the family  $\{V_j\}_{j=1}^j$ , where  $V_j = \bigcup \{V^* \colon V \in \mathscr{V} \text{ and } j(V) = j\} \subset H_j$ , is a cover of the space X and has order  $\leq n$ , or—equivalently—that the family  $\mathscr{V}^* = \{V^* \colon V \in \mathscr{V}\}$ is a cover of X and ord  $\mathscr{V}^* \leq n$ .

Let x be an arbitrary point of X. Consider an integer k such that

$$(5) x \in X_k \setminus \bigcup_{j < k} X_j$$

and a set  $U \in \mathscr{W}_k$  which contains the point x; since  $U \cap X_k \neq \emptyset$ ,  $U \in \mathscr{U}_k$ . It follows from (3) and (4) that  $x \in U \cap X_k \subset (f_{i(U)}^k(U))^* \in \mathscr{V}^*$ , so that  $\mathscr{V}^*$  is a cover of X.

It remains to show that  $\operatorname{ord} \mathscr{V}^* \leq n$ . Consider a non-empty intersection  $V_1^* \cap V_2^* \cap \ldots \cap V_h^*$ , where  $V_i \in \mathscr{V}_{m_i}$  and  $V_i \neq V_j$  whenever  $i \neq j$ ; let  $x \in V_1^* \cap V_2^* \cap \ldots \cap V_h^*$ . From the definition of  $\mathscr{V}_{m_i}$  it follows that for the integer k satisfying (5) we have  $m_i \leq k$  for  $i = 1, 2, \ldots, h$ . By (4) there exist sets  $U_i \in \mathscr{U}_{k_i}$  such that  $f_{m_i}^{k_i}(U_i) = V_i$ ,  $i(U_i) = m_i$  and  $x \in U_i \cap X_{k_i}$ . Since  $x \in X_{k_i}$  it follows from (5) that  $k \leq k_i$ . The sets  $W_1, W_2, \ldots, W_h$ , where  $W_i = f_k^{k_i}(U_i) \in \mathscr{W}_k$ , all contain the point x, so that—as  $\operatorname{ord} \mathscr{W}_k \leq n$ —it suffices to show that  $W_i \neq W_j$  whenever  $i \neq j$ . Let us note that the sets  $W_i$  belong to  $\mathscr{U}_k$  and that  $i(W_i) = i(U_i) = m_i$ ; hence  $W_i \neq W_j$  whenever  $m_i \neq m_j$ . When  $m_i = m_i$ , we also have  $W_i \neq W_j$ , because then

$$f_{m_i}^k(W_i) = f_{m_i}^{k_i}(U_i) = V_i \neq V_j = f_{m_j}^{k_j}(U_j) = f_{m_j}^k(W_j). \square$$

**4.1.2. Proposition.** For every metrizable space X the following conditions are equivalent:

- (a) The space X satisfies the inequality  $\dim X \leq n$ .
- (b) For every metric *Q* on the space X there exists a sequence 𝔄<sub>1</sub>, 𝔄<sub>2</sub>, ... of locally finite open covers of the space X such that for *i* = 1, 2, ... ord𝔅<sub>i</sub> ≤ n, δ(Ū) < 1/i for U ∈ 𝔅<sub>i</sub>, and for each U ∈ 𝔅<sub>i+1</sub> the set Ū is contained in a V ∈ 𝔅<sub>i</sub>.
- (c) There exist a metric ρ on the space X and a sequence W<sub>1</sub>, W<sub>2</sub>, ... of open covers of the space X such that for i = 1, 2, ... ord W<sub>i</sub> ≤ n, δ(W) < 1/i for W ∈ W<sub>i</sub>, and W<sub>i+1</sub> is a refinement of W<sub>i</sub>.

**Proof.** The implication (b)  $\Rightarrow$  (c) is obvious and the implication (c)  $\Rightarrow$  (a) follows from Lemma 4.1.1, so that it suffices to prove that (a)  $\Rightarrow$  (b). Consider a metrizable space X such that dim  $X \leq n$  and an arbitrary metric  $\varrho$  on the space X. We shall define inductively a sequence  $\mathscr{U}_1, \mathscr{U}_2, \ldots$  of open covers of X. Assume that k = 1 or that k > 1 and the covers  $\mathscr{U}_i$  are defined for all i < k. For every point  $x \in X$  there exists a neighbourhood  $U_x$  such that  $\delta(U_x) < 1/k$  and the set  $\overline{U}_x$  is contained in a member of  $\mathscr{U}_{k-1}$  if k > 1. Since every metrizable space is paracompact (see [GT], Theorem 5.1.3), it follows from Proposition 3.2.2 that the open cover  $\{U_x\}_{x \in X}$  of the space X has a locally finite open refinement  $\mathscr{U}_k$ . The sequence  $\mathscr{U}_1, \mathscr{U}_2, \ldots$  thus obtained satisfies all the conditions in (b), so that (a)  $\Rightarrow$  (b).  $\Box$ 

**4.1.3. The Katětov–Morita theorem.** For every metrizable space X we have IndX = dim X.

**Proof.** In view of Theorem 3.1.28 it suffices to show that  $\operatorname{Ind} X \leq \dim X$ . We can suppose that  $\dim X < \infty$ . We shall apply induction with respect to  $\dim X$ . If  $\dim X = -1$ , we clearly have  $\operatorname{Ind} X \leq \dim X$ . Assume that our inequality holds for all metrizable spaces with covering dimension  $\leq n-1$  and consider a metrizable space X such that  $\dim X = n \geq 0$  and a pair A, B of disjoint closed subsets of the space X. It suffices to define open sets K,  $M \subset X$  which, together with the set  $L = X \setminus (K \cup M)$ , satisfy the conditions

 $A \subset K$ ,  $B \subset M$ ,  $K \cap M = \emptyset$  and  $\dim L \leq n-1$ ; indeed, the set L is then a partition between A and B and  $\operatorname{Ind} L \leq n-1$ by virtue of the inductive assumption.

Let  $\sigma$  be an arbitrary metric on the space X and let  $f: X \to I$  be a continuous function satisfying  $f(A) \subset \{0\}$  and  $f(B) \subset \{1\}$ . One readily checks that the formula  $\varrho(x, y) = \sigma(x, y) + |f(x) - f(y)|$  defines a metric  $\varrho$  on the space X. From now on we shall consider on X only the metric  $\varrho$ . By virtue of Proposition 4.1.2 there exists a sequence  $\mathscr{U}_1, \mathscr{U}_2, \ldots$  of locally finite open covers of the space X such that for  $i = 1, 2, \ldots$  ord  $\mathscr{U}_i \leq n, \delta(U)$ < 1/i for  $U \in \mathscr{U}_i$  and for each  $U \in \mathscr{U}_{i+1}$  the set  $\overline{U}$  is contained in a  $V \in \mathscr{U}_i$ .

Let  $K_0 = A$ ,  $M_0 = B$ , and for  $i \ge 1$  let  $K_i = X \setminus H_i$  and  $M_i = X \setminus G_i$ , where

and

 $G_{i} = \bigcup \{ U \in \mathcal{U}_{i} \colon \overline{U} \cap M_{i-1} = \emptyset \}$  $H_{i} = \bigcup \{ U \in \mathcal{U}_{i} \colon \overline{U} \cap M_{i-1} \neq \emptyset \};$ 

in this way two sequences,  $K_0, K_1, K_2, \dots$  and  $M_0, M_1, M_2, \dots$ , of sub-

Ch. 4, § 1]

sets of the space X are defined.

Let us observe that

(6) if 
$$U \in \mathcal{U}_i$$
 and  $\overline{U} \cap M_{i-1} \neq \emptyset$ , then  $\overline{U} \cap K_{i-1} = \emptyset$ .

The validity of (6) for i = 1 follows from the definition of  $\rho$ , because no set of diameter less than 1 meets both A and B. If  $U \in \mathcal{U}_i$  where i > 1and  $\overline{U} \cap M_{i-1} \neq \emptyset$ , then for any  $V \in \mathscr{U}_{i-1}$  that contains the set  $\overline{U}$  we have  $V \cap M_{i-1} \neq \emptyset$ , so that V is not contained in  $G_{i-1}$ ; this implies that  $V \subset H_{i-1}$ , which gives the equality  $\overline{U} \cap K_{i-1} = \emptyset$ .

From the local finiteness of  $\mathscr{V}_i$ , the definitions of  $G_i$  and  $H_i$ , and (6) it follows that  $\overline{G}_i \cap M_{i-1} = \emptyset = \overline{H}_i \cap K_{i-1}$  for i = 1, 2, ..., which implies that  $K_{i-1} \subset X \setminus \overline{H_i} = \operatorname{Int} K_i$  and  $M_{i-1} \subset X \setminus \overline{G_i} = \operatorname{Int} M_i$ ; moreover, as  $G_i \cup H_i = X$ , we have  $K_i \cap M_i = \emptyset$ . Hence, the sets  $K = \bigcup_{i=0}^{i} K_i$  and  $M = \bigcup_{i=0}^{\infty} M_i$  are open, disjoint and contain respectively A and B.

Let  $L_i = X \setminus (K_i \cup M_i) = G_i \cup H_i$  for i = 1, 2, ...; clearly  $L = \bigcap_{i=1}^{\infty} L_i$ . The family  $\mathscr{W}_i = \{ U \cap L \colon U \in \mathscr{U}_i \text{ and } \overline{U} \cap M_{i-1} \neq \emptyset \}$  is, for i = 1, 2, ...,an open cover of the space  $L \subset H_i$  and ord  $\mathcal{W}_i \leq n-1$ , because each point  $x \in L \subset L_i \subset G_i$  belongs to at least one  $U \in \mathscr{U}_i$  satisfying  $\overline{U} \cap M_{i-1} = \emptyset$ . If  $U \in \mathscr{U}_{i+1}$  and  $\overline{U} \cap M_i \neq \emptyset$ , then for any  $V \in \mathscr{U}_i$  that contains  $\overline{U}$  we have  $V \cap M_i \neq \emptyset$ , so that V is not contained in  $G_i$ , which implies that  $\overline{V} \cap M_{i-1} \neq \emptyset$ , i.e., that  $V \cap L \in \mathcal{W}_i$ . Thus  $\mathcal{W}_{i+1}$  is a refinement of  $\mathcal{W}_i$ . Since, clearly,  $\delta(W) < 1/i$  for  $W \in \mathcal{W}_i$ , we have dim  $L \leq n-1$  by virtue of Proposition 4.1.2.

From the coincidence of the dimensions Ind and dim in metrizable spaces it follows that some results in the dimension theory of metrizable spaces, such as the subspace and sum theorems, are particular cases of both a theorem on Ind and a theorem on dim. However, the proofs of those particular cases are usually much simpler than the proofs of the corresponding general theorems. Moreover, what is more important, the number of theorems in the classical dimension theory which can be extended to metrizable spaces is larger than that of the theorems hitherto generalized in Chapters 2 and 3.

We are now going to list the counterparts of the theorems established in Section 1.5 for separable metric spaces; we shall always point out the theorems in Chapters 2 and 3 of which those counterparts are particular cases and, when possible, supply a simpler proof. The theorems will be formulated in terms of Ind; obviously, they could as well be formulated in terms of dim.

Let us begin, however, with a brief discussion of the status of the dimension ind in arbitrary metrizable spaces.

Theorems 2.4.4 and 4.1.3 (or 1.6.3, 3.1.29 and 4.1.3) yield the following

**4.1.4. Theorem.** For every strongly paracompact metrizable space X we have indX = IndX = dimX.  $\Box$ 

As a special case of the above theorem we obtain the following important fact, stated above as Theorem 1.7.7.

**4.1.5. Theorem.** For every separable metrizable space X we have indX = IndX = dim X.

Let us note that Theorem 4.1.5 can also be deduced directly from Theorem 4.1.3 and Lemma 1.7.4.

**4.1.6. Remark.** Let us state once more that there exists a completely metrizable space X, known as Roy's space, such that indX = 0 and yet IndX = dim X = 1. The definition of that space and the computation of its dimensions ind and Ind is too difficult to be included in this book.

As we ascertained in Chapter 2, the dimension ind develops pathological properties and is practically of no importance outside the class of separable metric spaces; suffice it to say that in metrizable spaces even the finite sum theorem for ind does not hold (see Problem 4.1.B). Therefore the dimension ind will not be discussed further in this book. It should be stressed, however, that the historical role of the small inductive dimension can hardly be overestimated. The dimension function ind was the first formal setting of the concept of dimension and a good base for the dimension theory of separable metric spaces. Besides, the dimension ind has a great intuitive appeal and yields quickly and economically the classical part of dimension theory.

We now turn to a list of the basic properties of dimension in metrizable spaces. First of all, let us observe that, since every subspace of a space X which satisfies condition (c) in Proposition 4.1.2 also satisfies this condition, from Proposition 4.1.2 and Theorem 4.1.3 we obtain the following theorem (which is a particular case of Theorems 2.3.6 and 3.1.19).

**4.1.7. The subspace theorem.** For every subspace M of a metrizable space X we have  $\operatorname{Ind} M \leq \operatorname{Ind} X$ .

The following simple theorem is a particular case of Theorem 2.2.2.

**4.1.8. Theorem.** If X is a metrizable space and  $\operatorname{Ind} X = n \ge 1$ , then for k = 0, 1, ..., n-1 the space X contains a closed subspace M such that  $\operatorname{Ind} M = k$ .

From Theorems 2.3.8 and 2.3.10 (or 3.1.8, 3.1.10 and 4.13) we obtain the countable and the locally finite sum theorems.

**4.1.9. The countable sum theorem.** If a metrizable space X can be represented as the union of the sequence  $F_1, F_2, \ldots$  of closed subspaces such that  $\operatorname{Ind} F_i \leq n$  for  $i = 1, 2, \ldots$ , then  $\operatorname{Ind} X \leq n$ .  $\Box$ 

**4.1.10.** The locally finite sum theorem. If a metrizable space X can be represented as the union of a locally finite family  $\{F_s\}_{s\in S}$  of closed subspaces such that  $\operatorname{Ind} F_s \leq n$  for  $s \in S$ , then  $\operatorname{Ind} X \leq n$ .  $\Box$ 

The next theorem is a common generalization of the last two theorems; it follows from Theorem 2.3.15 (or 3.1.15) and the fact that every metrizable space is paracompact.

**4.1.11. Theorem.** If a metrizable space X can be represented as the union of a locally countable family  $\{F_s\}_{s\in S}$  of closed subspaces such that  $\operatorname{Ind} F_s \leq n$  for  $s \in S$ , then  $\operatorname{Ind} X \leq n$ .  $\Box$ 

Another common generalization of the countable and the locally finite sum theorems is the following theorem, which is stronger than similar results in Chapters 2 and 3.

**4.1.12. Theorem.** If a metrizable space X can be represented as the union of a transfinite sequence  $K_1, K_2, ..., K_{\alpha}, ..., \alpha < \xi$  of subspaces such that  $\operatorname{Ind} K_{\alpha} \leq n$  and the union  $\bigcup_{\beta < \alpha} K_{\beta}$  is closed for  $\alpha < \xi$ , then  $\operatorname{Ind} X \leq n$ .

**Proof.** Let  $\varrho$  be a metric on the space X. For each  $\alpha < \xi$  define  $F_{\alpha} = \bigcup_{\beta < \alpha} K_{\beta}$  and consider the sets

 $F_{i,\alpha} = K_{\alpha} \setminus B(F_{\alpha}, 1/i) = F_{\alpha+1} \setminus B(F_{\alpha}, 1/i)$  for i = 1, 2, ..., 3

where B(A, r) denotes the open *r*-ball about A with respect to the metric  $\varrho$ . The sets  $F_{i,\alpha}$  are closed and  $\operatorname{Ind} F_{i,\alpha} \leq n$  for  $\alpha < \xi$  and i = 1, 2, ... Since  $F_{i,\alpha} \cap B(F_{i,\beta}, 1/i) = \emptyset$  whenever  $\beta < \alpha$ , the family  $\{F_{i,\alpha}\}_{\alpha < \xi}$  is discrete, so that the set  $F_i = \bigcup_{\alpha < \xi} F_{i,\alpha}$  is closed and satisfies the inequality  $\operatorname{Ind} F_i \leq n$  for i = 1, 2, ...

It remains to show that  $X = \bigcup_{i=1}^{i} F_i$ . Consider an arbitrary point  $x \in X$ ; let  $\alpha$  be the smallest ordinal number less than  $\xi$  such that  $x \in K_{\alpha}$ . Since  $x \notin F_{\alpha}$ , there exists an integer *i* such that  $F_{\alpha} \cap B(x, 1/i) = \emptyset$ , which implies that  $x \notin B(F_{\alpha}, 1/i)$ . Thus  $x \in F_{i,\alpha} \subset F_i$ .  $\Box$ 

From Lemma 1.2.9 and Remark 1.2.10 one easily obtains (cf. the proof of Theorem 1.2.11) the following result, which is a particular case of Theorem 2.2.4.

**4.1.13. The separation theorem.** If X is a metrizable space and M is a subspace of X such that  $\operatorname{Ind} M \leq n \geq 0$ , then for every pair A, B of disjoint closed subsets of X there exists a partition L between A and B such that  $\operatorname{Ind}(L \cap M) \leq n-1$ .  $\Box$ 

We shall now characterize the dimension of subspaces of metrizable spaces in terms of  $\sigma$ -locally finite bases for the space (cf. Proposition 1.5.15); the characterization will be applied in the proofs of the decomposition, enlargement and Cartesian products theorems.

**4.1.14. Proposition.** A subspace M of a metrizable space X satisfies the inequality  $\operatorname{Ind} M \leq n \geq 0$  if and only if X has a  $\sigma$ -locally finite base  $\mathscr{B}$  such that  $\operatorname{Ind}(M \cap \operatorname{Fr} U) \leq n-1$  for every  $U \in \mathscr{B}$ .

**Proof.** Consider a subspace M of a metrizable space X which satisfies the inequality  $\operatorname{Ind} M \leq n \geq 0$ ; let  $\varrho$  be an arbitrary metric on the space X. The space X being paracompact, for i = 1, 2, ... there exists a locally finite open cover  $\mathscr{V}_i = \{V_s\}_{s \in S_i}$  of the space X such that  $\operatorname{mesh} \mathscr{V}_i < 1/i$ ; let  $\{F_s\}_{s \in S_i}$  be a closed shrinking of the cover  $\mathscr{V}_i$  (see [GT], Theorem 1.5.18). By virtue of Theorem 4.1.13 for every  $s \in S_i$  there exists a partition  $L_s$  between  $F_s$  and  $X \setminus V_s$  such that  $\operatorname{Ind}(L_s \cap M) \leq n-1$ . Consequently there exist open sets  $U_s, W_s \subset X$  such that

 $F_s \subset U_s, \quad X \setminus V_s \subset W_s, \quad U_s \cap W_s = \emptyset \quad \text{and} \quad X \setminus L_s = U_s \cup W_s.$ 

As  $\operatorname{Fr} U_s \subset L_s$ ,  $\operatorname{Ind}(M \cap \operatorname{Fr} U_s) \leq n-1$ , and as  $U_s \subset X \setminus W_s \subset V_s$ ,  $\delta(U_s) < 1/i$  for  $s \in S_i$ . The family  $\mathscr{B}_i = \{U_s\}_{s \in S_i}$  is for i = 1, 2, ... a locally

finite open cover of the space X such that mesh $\mathscr{B}_i \leq 1/i$ , so that the union  $\mathscr{B} = \bigcup_{i=1}^{\infty} \mathscr{B}_i$  is a  $\sigma$ -locally finite base for the space X such that  $\operatorname{Ind}(M \cap \operatorname{Fr} U)$  $\leq n-1$  for every  $U \in \mathscr{B}$ .

Conversely, if M is a subspace of a metrizable space X and X has a  $\sigma$ -locally finite base  $\mathscr{B}$  such that  $\operatorname{Ind}(M \cap \operatorname{Fr} U) \leq n-1$  for every  $U \in \mathscr{B}$ , then the family  $\{M \cap U: U \in \mathscr{B}\}$  is a  $\sigma$ -locally finite base for the subspace M whose members have boundaries of large inductive dimension  $\leq n-1$ ; by virtue of Lemma 2.3.16 and Theorems 4.1.9, 4.1.10 and 4.1.7, this implies that  $\operatorname{Ind} M \leq n$ .  $\Box$ 

The next theorem (cf. Theorem 1.1.6) is a particular case of Proposition 4.1.14.

**4.1.15. Theorem.** A metrizable space X satisfies the inequality  $\operatorname{Ind} X \leq n \geq 0$  if and only if X has a  $\sigma$ -locally finite base  $\mathscr{B}$  such that  $\operatorname{Ind} \operatorname{Fr} U \leq n-1$  for every  $U \in \mathscr{B}$ .  $\Box$ 

Let us note that in Proposition 4.1.14 and Theorem 4.1.15 the  $\sigma$ -local finiteness can be replaced by  $\sigma$ -discreteness (cf. [GT], Theorem 4.4.1).

**4.1.16. The first decomposition theorem.** A metrizable space X satisfies the inequality  $\operatorname{Ind} X \leq n \geq 0$  if and only if X can be represented as the union of two subspaces Y and Z such that  $\operatorname{Ind} Y \leq n-1$  and  $\operatorname{Ind} Z \leq 0$ .

**Proof.** Consider a metrizable space X such that  $\operatorname{Ind} X \leq n \geq 0$ . By virtue of Theorem 4.1.15, the space X has a  $\sigma$ -locally finite base  $\mathscr{B}$  such that  $\operatorname{Ind} \operatorname{Fr} U \leq n-1$  for every  $U \in \mathscr{B}$ . From Theorems 4.1.9 and 4.1.10 it follows that the subspace  $Y = \bigcup \{\operatorname{Fr} U: U \in \mathscr{B}\}$  satisfies the inequality  $\operatorname{Ind} Y \leq n-1$ , and from Proposition 4.1.14 it follows that the subspace  $Z = X \setminus Y$  satisfies the inequality  $\operatorname{Ind} Z \leq 0$ .

If X is a metrizable space and  $X = Y \cup Z$ , where  $\operatorname{Ind} Y \leq n-1$  and  $\operatorname{Ind} Z \leq 0$ , then  $\operatorname{Ind} X \leq n$  by virtue of Theorems 4.1.13 and 4.1.7.  $\Box$ 

From the first decomposition theorem we obtain by easy induction 4.1.17. The second decomposition theorem. A metrizable space X satisfies the inequality  $\operatorname{Ind} X \leq n \geq 0$  if and only if X can be represented as the union of n+1 subspaces  $Z_1, Z_2, \ldots, Z_{n+1}$  such that  $\operatorname{Ind} Z_i \leq 0$  for  $i = 1, 2, \ldots$  $\ldots, n+1$ .  $\Box$ 

From Theorem 4.1.17 immediately follows the addition theorem, which is a particular case of Theorems 2.2.5 and 3.1.17.

**4.1.18. The addition theorem.** For every pair X, Y of subspaces of a metrizable space we have

$$\operatorname{Ind}(X \cup Y) \leq \operatorname{Ind} X + \operatorname{Ind} Y + 1. \square$$

We now turn to the enlargement theorem.

**4.1.19. The enlargement theorem.** For every subspace M of a metrizable space X satisfying the inequality  $\operatorname{Ind} M \leq n$  there exists a  $G_{\delta}$ -set  $M^*$  in X such that  $M \subset M^*$  and  $\operatorname{Ind} M^* \leq n$ .

**Proof.** Consider first the particular case of a subspace Z of X such that  $\operatorname{Ind} Z \leq 0$ . By virtue of Proposition 4.1.14 the space X has a  $\sigma$ -locally finite base  $\mathscr{B}$  such that  $Z \cap \operatorname{Fr} U = \emptyset$  for every  $U \in \mathscr{B}$ . The union  $F = \bigcup \{\operatorname{Fr} U: U \in \mathscr{B}\}$  is an  $F_{\sigma}$ -set, and its complement  $Z^* = X \setminus F$  is a  $G_{\delta}$ -set which contains the set Z. From Proposition 4.1.14 it follows that  $\operatorname{Ind} Z^* \leq 0$ .

To complete the proof it suffices to use Theorem 4.1.17 and apply the particular case established above (cf. the proof of Theorem 1.5.11).

Since every compact metrizable space is separable, no non-separable metrizable space has a metrizable compactification. The next theorem is a substitute for the compactification theorem in the realm of metrizable spaces; it follows from Theorem 4.1.19, Lemma 1.3.12, and the fact that each metrizable space is homeomorphic to a subspace of a completely metrizable space (see [GT], Corollary 4.3.15).

**4.1.20. The completion theorem.** For every metrizable space X there exists a completely metrizable space  $\tilde{X}$  which contains a dense subspace homeomorphic to X and satisfies the equalities  $\operatorname{Ind} \tilde{X} = \operatorname{Ind} X$  and  $w(\tilde{X}) = w(X)$ .

We now pass to the Cartesian product theorem. It is a particular case of Theorem 2.4.6; the proof of this particular case is much simpler than the proof given in Section 2.4.

**4.1.21. The Cartesian product theorem.** For every pair X, Y of metrizable spaces of which at least one is non-empty we have

$$\operatorname{Ind}(X \times Y) \leq \operatorname{Ind} X + \operatorname{Ind} Y.$$

**Proof.** The theorem is obvious if  $\operatorname{Ind} X = \infty$  or  $\operatorname{Ind} Y = \infty$ , so that we can suppose that  $k(X, Y) = \operatorname{Ind} X + \operatorname{Ind} Y$  is finite. We shall apply induction

-

with respect to that number. If k(X, Y) = -1, then either  $X = \emptyset$  or  $Y = \emptyset$ , and our inequality holds. Assume that the inequality is proved for every pair of metrizable spaces the sum of large inductive dimensions of which is less than  $k \ge 0$  and consider metrizable spaces X and Y such that  $\operatorname{Ind} X = n \ge 0$ ,  $\operatorname{Ind} Y = m \ge 0$  and n+m = k. By virtue of Theorem 4.1.15 the space X has a base  $\mathscr{C} = \bigcup_{i=1}^{\infty} \mathscr{C}_i$ , where the families  $\mathscr{C}_i$  are locally finite and  $\operatorname{Ind} \operatorname{Fr} U \le n-1$  for every  $U \in \mathscr{C}$ . Similarly, the space Y has a base  $\mathscr{D} = \bigcup_{j=1}^{\infty} \mathscr{D}_j$ , where the families  $\mathscr{D}_j$  are locally finite and  $\operatorname{Ind} \operatorname{Fr} V \le m-1$  for every  $V \in \mathscr{D}$ . For i, j = 1, 2, ... the family  $\mathscr{B}_{i,j} = \{U \times V: U \in \mathscr{C}_i \text{ and } V \in \mathscr{D}_j\}$ 

consists of open subsets of the Cartesian product  $X \times Y$  and is locally finite. Since

$$\operatorname{Fr}(U \times V) \subset (X \times \operatorname{Fr} V) \cup (\operatorname{Fr} U \times Y),$$

by virtue of the inductive assumption and Theorem 4.1.9 we have Ind  $\operatorname{Fr}(U \times V) \leq k-1$ . The family  $\{\mathscr{B}_{i,j}\}_{i,j=1}^{\infty}$  is a base for the Cartesian product  $X \times Y$ . From Theorem 4.1.15 it follows that  $\operatorname{Ind}(X \times Y) \leq k$  and the proof is completed.  $\Box$ 

We shall now prove the theorem on dimension of the limit of an inverse sequence of metrizable spaces. In consideration of the context of inverse systems and the character of the proof given below, the theorem is formulated in terms of the dimension dim.

**4.1.22.** Theorem on dimension of the limit of an inverse sequence. If the inverse sequence  $S = \{X_i, \pi_j^i\}$  consists of metrizable spaces  $X_i$  such that  $\dim X_i \leq n$  for i = 1, 2, ..., then the limit  $X = \lim S$  satisfies the inequality  $\dim X \leq n$ .

**Proof.** For i = 1, 2, ... consider a metric  $\varrho_i$  on the space  $X_i$  bounded by 1. On the Cartesian product  $\prod_{i=1}^{\infty} X_i$  and on its subspace X we shall consider the metric  $\varrho$  defined by letting

(7) 
$$\varrho(\{x_i\}, \{y_i\}) = \sum_{i=1}^{\infty} \frac{1}{2^i} \varrho_i(x_i, y_i) \text{ for } \{x_i\}, \{y_i\} \in \prod_{i=1}^{\infty} X_i.$$

For i, k = 1, 2, ... let  $\mathscr{U}_{i,k}$  be an open cover of the space  $X_i$  such that mesh  $\mathscr{U}_{i,k} < 1/2k$ . We shall define inductively a sequence  $\mathscr{U}_1, \mathscr{U}_2, ...$  of families of sets satisfying the conditions:

- (8) The family  $\mathcal{U}_i$  is an open cover of the space  $X_i$  and  $\operatorname{ord} \mathcal{U}_i \leq n$ .
- (9) For each  $U \in \mathcal{U}_i$ , where i > 1, there exists a  $V \in \mathcal{U}_{i-1}$  such that  $U \subset (\pi_{i-1}^i)^{-1}(V)$ .

(10) For every  $j \leq i \operatorname{mesh}(\{\pi_j^i(U): U \in \mathcal{U}_i\}) < 1/2i$ .

Conditions (8)-(10) are satisfied for i = 1 by an arbitrary open refinement  $\mathcal{U}_1$  of the cover  $\mathcal{U}_{1,1}$  of the space  $X_1$  such that  $\operatorname{ord} \mathcal{U}_1 \leq n$ ; such a refinement exists by virtue of Proposition 3.2.2. Assume that the families  $\mathcal{U}_i$  satisfying (8)-(10) are defined for all i < k > 1. Let  $\mathcal{U}_k$  be an arbitrary open refinement of the cover

$$[(\pi_{k-1}^k)^{-1}(\mathscr{U}_{k-1})] \wedge [(\pi_1^k)^{-1}(\mathscr{U}_{1,k})] \wedge [(\pi_2^k)^{-1}(\mathscr{U}_{2,k})] \wedge \dots \wedge [(\pi_k^k)^{-1}(\mathscr{U}_{k,k})]$$

of the space  $X_k$  such that  $\operatorname{ord} \mathcal{U}_k \leq n$ . From the definition it easily follows that  $\mathcal{U}_k$  satisfies (8)-(10) with i = k; thus the construction of the families  $\mathcal{U}_i$  is completed.

For i = 1, 2, ... the family  $\mathscr{W}_i = \pi_i^{-1}(\mathscr{U}_i)$ , where  $\pi_i \colon X \to X_i$  denotes the projection, is an open cover of the space X. For every  $W \in \mathscr{W}_{i+1}$  there exists a  $U \in \mathscr{U}_{i+1}$  such that  $W = \pi_{i+1}^{-1}(U)$ ; by virtue of (9) one can find a  $V \in \mathscr{U}_i$  such that  $U \subset (\pi_i^{i+1})^{-1}(V)$ . Thus  $W \subset \pi_{i+1}^{-1}(\pi_i^{i+1})^{-1}(V)$  $= (\pi_i^{i+1}\pi_{i+1})^{-1}(V) = \pi_i^{-1}(V) \in \mathscr{W}_i$ , which shows that  $\mathscr{W}_{i+1}$  is a refinement of  $\mathscr{W}_i$ . From (10) it follows that  $\delta(W) < 1/2i + 1/2^i \le 1/i$  for every  $W \in \mathscr{W}_i$ , so that dim $X \le n$  by virtue of Proposition 4.1.2.  $\Box$ 

We conclude this section by proving that for every cardinal number  $m \ge \aleph_0$  the Baire space B(m) defined below is a universal space for the class of all metrizable spaces whose large inductive dimension is not larger than 0 and whose weight is not larger than m. In the following section this result will be extended to higher dimensions; as the reader will see, the proof of the general theorem is much more difficult than that of the particular case discussed here.

We start with the definition of the Baire space.

**4.1.23. Example.** For i = 1, 2, ... let  $X_i = D(\mathfrak{m})$  be the discrete space of cardinality  $\mathfrak{m} \ge \aleph_0$  with the metric  $\varrho_i$  defined by

$$\varrho_i(x, y) = 1$$
 if  $x \neq y$  and  $\varrho_i(x, x) = 0$ .

Ch. 4, § 1]

The Cartesian product  $\prod_{i=1}^{\infty} X_i = [D(\mathfrak{m})]^{\aleph_0}$  is a metrizable space; it is well known that formula (7) defines a metric  $\varrho$  on that space. One can readily verify that by letting

(11) 
$$\sigma(\{x_i\}, \{y_i\}) = \begin{cases} 1/k, & \text{if } x_k \neq y_k \text{ and } x_i = y_i \text{ for } i < k, \\ 0, & \text{if } x_i = y_i \text{ for } i = 1, 2, \dots \end{cases}$$

one defines another metric on the set  $\prod_{i=1}^{\infty} X_i$ . The metrics  $\varrho$  and  $\sigma$  are equiv-

alent. Indeed, a sequence  $\{x_i^1\}, \{x_i^2\}, \dots$  in the Cartesian product  $\prod_{i=1}^{\infty} X_i$  converges to a point  $\{x_i\}$  if and only if for every *i* there exists a k(i) such that  $x_i^j = x_i$  whenever  $j \ge k(i)$ , and the same condition is necessary and sufficient for the convergence of the sequence  $\{x_i^1\}, \{x_i^2\}, \dots$  to the point  $\{x_i\}$  with respect to the metric  $\sigma$  defined by (11).

The Cartesian product  $[D(m)]^{\aleph_0}$  with the metric  $\sigma$  defined in (11) is called the *Baire space of weight* m and is denoted by B(m). The reader can easily check that the weight of the space B(m) is really equal to m. Let us note that by virtue of Proposition 1.3.13 the Baire space  $B(\aleph_0)$  is homeomorphic to the space of irrational numbers.

We shall show that  $\operatorname{Ind} B(\mathfrak{m}) = 0$ . Consider a pair  $x = \{x_i\}, y = \{y_i\}$ of points of  $B(\mathfrak{m})$  and a real number r satisfying  $0 < r \leq 1$ . If the intersection  $B(x, r) \cap B(y, r)$  is non-empty, then there exists a point  $z = \{z_i\}$  $\in B(\mathfrak{m})$  such that  $x_1 = z_1 = y_1, x_2 = z_2 = y_2, \dots, x_k = z_k = y_k$ , where kis the integer satisfying  $1/k + 1 < r \leq 1/k$ ; thus we have B(x, r) = B(y, r). Hence in  $B(\mathfrak{m})$  two r-balls either are disjoint or coincide. In particular, for  $i = 1, 2, \ldots$  the family  $\mathscr{B}_i = \{B(x, 1/i): x \in B(\mathfrak{m})\}$  is an open cover of  $B(\mathfrak{m})$  which consists of pairwise disjoint sets. The union  $\mathscr{B} = \bigcup_{i=1}^{\infty} \mathscr{B}_i$ is a  $\sigma$ -locally finite base for  $B(\mathfrak{m})$  which consists of open-and-closed sets, so that  $\operatorname{Ind} B(\mathfrak{m}) \leq 0$  by virtue of Theorem 4.1.15.  $\Box$ 

**4.1.24. Theorem.** For every cardinal number  $\mathfrak{m} \ge \aleph_0$  the Baire space  $B(\mathfrak{m})$  is a universal space for the class of all metrizable spaces whose large inductive dimension is not larger than 0 and whose weight is not larger than  $\mathfrak{m}$ .

**Proof.** By virtue of the last example it suffices to show that every metrizable space X such that IndX = 0 and w(X) = m is embeddable in X.

[Ch. 4, § 1

Let  $\varrho$  be an arbitrary metric on the space X and  $\mathscr{B}$  an arbitrary base for X such that  $|\mathscr{B}| = \mathfrak{m}$ . From Theorem 1.6.11 (or 4.1.3) and Proposition 3.2.2 it follows that the cover  $\{U \in \mathscr{B}: \delta(U) < 1/i\}$  of the space X has an open shrinking  $\mathscr{B}_i$  consisting of pairwise disjoint sets. Adjoining to  $\mathscr{B}_i$ , if necessary, an appropriate number of copies of the empty set, we can assume that  $\mathscr{B}_i = \{U_{i,s}\}_{s \in X_i}$ , where  $X_i = D(\mathfrak{m})$  is the discrete space of cardinality  $\mathfrak{m}$  used in Example 4.1.23 to define the space  $B(\mathfrak{m})$ .

By assigning to each point  $x \in X$  the element  $s \in X_i$  such that  $x \in U_{i,s}$ we define a continuous mapping  $f_i: X \to X_i$ . For every  $x \in X$  and every closed set  $F \subset X$  such that  $x \notin F$  there exists a natural number *i* such that  $\varrho(x, F) > 1/i$ . The set  $U_{i,s}$  that contains the point *x* is disjoint from *F*, so that  $f_i(x) = s \notin f_i(F) = \overline{f_i(F)}$ . Thus the family  $\{f_i\}_{i=1}^{\infty}$  separates points and closed sets, which implies that the mapping  $F: X \to \prod_{i=1}^{\infty} X_i$  $= B(\mathfrak{m})$  defined by letting  $F(x) = (f_1(x), f_2(x), \ldots)$  is a homeomorphic embedding (see [GT], Theorem 2.3.20).  $\Box$ 

Since the Cartesian product of  $\aleph_0$  copies of  $B(\mathfrak{m})$  is homeomorphic to  $B(\mathfrak{m})$ , Theorem 4.1.24 yields the following

**4.1.25. Theorem.** The Cartesian product  $X = \prod_{i=1}^{\infty} X_i$  of a countable family  $\{X_i\}_{i=1}^{\infty}$  of metrizable spaces satisfies the equality  $\operatorname{Ind} X = 0$  if and only if  $\operatorname{Ind} X_i = 0$  for i = 1, 2, ...

### Historical and bibliographic notes

Lemma 4.1.1 was established by Nagami and Roberts in [1967]. The equivalence of conditions (a) and (b) in Proposition 4.1.2 was proved by Dowker and Hurewicz in [1956]; the equivalence of conditions (a) and (c) was proved by Vopěnka in [1959]. Theorem 4.1.3 was established independently by Katětov in [1952] (announcement in [1951]) and by Morita in [1954]. Theorem 4.1.4 was given by Morita in [1950a]. References concerning Roy's space cited in Remark 4.1.6 are given in the notes to Section 2.4. Theorems 4.1.7–4.1.11, 4.1.13 and 4.1.18 are special cases of the theorems established in Chapters 2 and 3. Theorem 4.1.12 was given by Nagami in [1957]. Theorems 4.1.14 and 4.1.15 were proved by Morita in [1954]. All the remaining theorems in this section, except for Theorem 4.1.22, which was proved by Nagami in [1952] and by Morita in [1954].

## Problems

**4.1.A** (Engelking [1973], Przymusiński [1974]). Let X be a metrizable space and  $\rho$  a metric on the space X; let  $ds(X, \rho) \leq n$  denote that the space X has a sequence of open covers  $\mathscr{U}_1, \mathscr{U}_2, \ldots$  with the properties stated in condition (b) in Proposition 4.1.2.

(a) Show by modifying the proof of Theorem 4.1.3, that if  $ds(X, \varrho) \le n \ge 0$  then for every pair A, B of closed subsets of X satisfying  $\varrho(A, B) > 0$  there exists an open set  $U \subset X$  such that  $A \subset U \subset X \setminus B$  and  $ds(\operatorname{Fr} U, \varrho) \le n-1$ .

(b) Apply (a) to prove the Katětov-Morita theorem without using Lemma 4.1.1.

*Hint.* Observe first that if dim  $X \leq n$ , then ds $(X, \varrho) \leq n$  for every metric  $\varrho$  on the space X; then apply (a) to prove by induction that the inequality ds $(X, \varrho) \leq n \geq 0$  implies that X has a  $\sigma$ -locally finite base  $\mathscr{B}$  such that dim Fr  $U \leq n-1$  for every  $U \in \mathscr{B}$  and show that the existence of such a base implies the inequality Ind $X \leq n$ .

One can also apply (a) to prove by induction that if  $ds(X, \varrho) \le n \ge 0$ then, for every closed set  $A \subset X$  and each open set  $V \subset X$  that contains the set A, there exists an open set  $U \subset X$  such that  $A \subset U \subset V$ and Ind  $\operatorname{Fr} U \le n-1$ . To this end, for  $i = 1, 2, \ldots$  define  $A_i = B(A, 1/i)$ and  $A'_i = B(X \setminus V, 1/i)$ , consider open sets  $W_i$ ,  $W'_i \subset X$  such that  $A_{i+1} \subset W_i \subset A_i$ ,  $ds(\operatorname{Fr} W_i, \varrho) \le n-1$  and  $A'_{i+1} \subset W'_i \subset A'_i$ ,  $ds(\operatorname{Fr} W'_i, \varrho) \le n-1$ , and let  $U = \bigcup_{i=1}^{\infty} (W_i \setminus W'_i)$ .

**4.1.B** (van Douwen [1973], Przymusiński [1974]). Applying the existence of a metrizable space X with the properties described in Remark 4.1.6, define a metrizable space Y with  $\operatorname{ind} Y = 1$  which can be represented as the union of two closed subspaces  $Y_1$  and  $Y_2$  such that  $\operatorname{ind} Y_1 = \operatorname{ind} Y_2 = 0$  and which contains a point p such that  $\operatorname{ind}(Y \setminus \{p\}) = 0$ .

*Hint.* Consider a pair A, B of disjoint closed subsets of X which cannot be enlarged to disjoint open-and-closed sets, and replace the set B by a point p in such a way as to obtain a metrizable space.

**4.1.C.** Let Y be the space considered in Problem 4.1.B and let  $M = Y \setminus \{p\}$ . Show that, though  $\operatorname{ind} M = 0$ , there exists a neighbourhood V of the point p in the space Y such that for every open set  $U \subset Y$  satisfying  $p \in U \subset V$  we have  $M \cap \operatorname{Fr} U \neq \emptyset$  (cf. Propositions 1.2.12 and 1.5.14).

**4.1.D** (R. Pol [1979]). Prove that if a metrizable space X can be represented as the union of a family  $\{F_s\}_{s\in S}$  of closed subspaces such that Ind  $F_s \leq n$  for  $s \in S$  and if there exists a point-countable open cover  $\{U_s\}_{s\in S}$  of the space X such that  $F_s \subset U_s$  for  $s \in S$ , then Ind  $X \leq n$ .

Hint (Hansell [1974]). Let  $\mathscr{B} = \bigcup_{i=1}^{\mathcal{B}} \mathscr{B}_i$  be a base for the space X, where the families  $\mathscr{B}_i$  are locally finite. For every non-empty  $U \in \mathscr{B}$ consider a one-to-one transformation  $j_U$  of the set  $\{s \in S: U \subset U_s\}$  to the integers and let  $U_{s,i,j} = \bigcup \{U \in \mathscr{B}_i: U \subset U_s \text{ and } j_U(s) = j\}$ . Show that the family  $\{U_{s,i,j}\}_{s\in S}$  is locally finite for i, j = 1, 2, ... and  $U_s = \bigcup_{i,j=1}^{\infty} U_{s,i,j}$  for  $s \in S$ .

**4.1.E.** Deduce from Theorems 4.1.3 and 4.1.13 and Remark 1.7.10 that a metrizable space X satisfies the inequality  $\operatorname{Ind} X \leq n \geq 0$  if and only if for every sequence  $(A_1, B_1), (A_2, B_2), \ldots, (A_{n+1}, B_{n+1})$  of n+1 pairs of disjoint closed subsets of X there exist closed sets  $L_1, L_2, \ldots, L_{n+1}$  such that  $L_i$  is a partition between  $A_i$  and  $B_i$  and  $\bigcap_{i=1}^{n+1} L_i = \emptyset$ .

**4.1.F** (Levšenko [1969]; for n = 0 Levšenko and Smirnov [1966]; for separable spaces implicitly Popruženko [1931]). (a) Prove that for every non-empty closed subset A of a metrizable space X satisfying the inequality  $\operatorname{Ind} X \leq n \geq 0$  there exists a closed subset B of the subspace  $X \setminus A$  of X such that  $\operatorname{Ind} B \leq n-1$  and A is a retract of  $X \setminus B$ .

Hint. Consider a metric  $\varrho$  on the space X and define a sequence  $U_1, U_2, ...$ of open subsets of X such that  $A \subset U_i \subset B(A, 1/i)$ ,  $\overline{U_i} \subset U_{i-1}$ , where  $U_0 = X$ , and  $\operatorname{Ind} \operatorname{Fr} U_i \leq n-1$  for i = 1, 2, ...; consider for i = 1, 2, ...a locally finite open cover  $\mathscr{U}_i$  of the subspace  $\overline{U_{i-1}} \setminus U_i$  of the space X such that  $\operatorname{mesh} \mathscr{U}_i \leq 1/i$  and  $\operatorname{Ind} \operatorname{Fr} U \leq n-1$  for every  $U \in \mathscr{U}_i$ . Define  $B = \bigcup_{i=1}^{\infty} \operatorname{Fr} U_i \cup \bigcup \{\operatorname{Fr} U: U \in \bigcup_{i=1}^{\infty} \mathscr{U}_i\}$  and see the hint to Problem 1.3.C(a).

(b) Show that if a metrizable space X has the property that for every non-empty closed set  $A \subset X$  there exists a closed subset B of the subspace  $X \setminus A$  of X such that  $\operatorname{ind} B \leq n-1$  and A is a retract of  $X \setminus B$ , then  $\operatorname{Ind} X \leq n$ .

**4.1.G** (Hausdorff [1934], de Groot [1956]). A metric  $\varrho$  on a set X is called *non-Archimedean* if  $\varrho(x, z) \leq \max[\varrho(x, y), \varrho(y, z)]$  for all  $x, y, z \in X$ .

Let X be a metrizable space; show that on the space X there exists a non-Archimedean metric if and only if  $\operatorname{Ind} X \leq 0$ .

*Hint*. There exists a non-Archimedean metric on the Baire space B(m).

# 4.2. Characterizations of dimension in metrizable spaces. The universal space theorem

In the first part of this section we shall establish four characterizations of the dimension dim in metrizable spaces formulated in terms of the existence of special bases and sequences of covers, and we shall review characterizations of dim in terms of special metrics.

We start with characterizations in terms of  $\sigma$ -locally finite bases with special properties. First we shall prove a lemma related to Theorem 3.2.5.

**4.2.1. Lemma.** If a normal space X satisfies the inequality dim  $X \le n \ge 0$ , then for every  $\sigma$ -locally finite family  $\{U_s\}_{s\in S}$  of open subsets of X and every family  $\{F_s\}_{s\in S}$  of closed subsets of X such that  $F_s \subset U_s$  for  $s \in S$  there exists a family  $\{V_s\}_{s\in S}$  of open subsets of X such that  $F_s \subset V_s \subset \overline{V}_s \subset U_s$  for  $s \in S$  and  $\operatorname{ord}(\{\operatorname{Fr} V_s\}_{s\in S}) \le n-1$ .

**Proof.** Let  $S = \bigcup_{i=1}^{\infty} S_i$ , where  $S_i \cap S_j = \emptyset$  whenever  $i \neq j$  and the family  $\{U_s\}_{s \in S_i}$  is locally finite for i = 1, 2, ... Applying Theorem 3.2.5, one can easily define by induction open sets  $V_{s,i}, W_{s,i} \subset X$ , where  $s \in T_i$ =  $S_1 \cup S_2 \cup ... \cup S_i$  and i = 1, 2, ..., satisfying the following conditions:

(1) 
$$F_s \subset V_{s,i} \subset \overline{V}_{s,i} \subset W_{s,i} \subset \overline{W}_{s,i} \subset U_s \quad \text{for } s \in S_i.$$

(2) 
$$\overline{V}_{s,i-1} \subset V_{s,i} \subset \overline{V}_{s,i} \subset W_{s,i} \subset \overline{W}_{s,i} \subset W_{s,i-1}$$
 for  $s \in T_{i-1}$  and  $i > 1$ .

(3) 
$$\operatorname{ord}(\{\overline{W}_{s,i} \setminus V_{s,i}\}_{s \in T_i}) \leq n-1.$$

For every  $s \in S$  consider the open set

$$V_s = \bigcup_{j=i}^{\infty} V_{s,j}, \quad \text{where } s \in S_i.$$

Conditions (1) and (2) imply that  $F_s \subset V_s \subset \overline{V}_s \subset U_s$  for  $s \in S$  and that  $\operatorname{Fr} V_s = \overline{V_s} \setminus V_s \subset \overline{W_{s,j}} \setminus V_{s,j}$  for  $j \ge i$ , where  $s \in S_i$ . The last inclusion together with (3) yield the inequality  $\operatorname{ord}(\{\operatorname{Fr} V_s\}_{s \in S}) \le n-1$ .  $\Box$ 

**4.2.2. Theorem.** For every metrizable space X and each integer  $n \ge 0$  the following conditions are equivalent:

- (a) The space X satsifies the inequality  $\dim X \leq n$ .
- (b) The space X has a  $\sigma$ -locally finite base  $\mathscr{B}$  such that  $\operatorname{ord}({\operatorname{Fr} U: U \in \mathscr{B}}) \leq n-1$ .
- (c) The space X has a  $\sigma$ -locally finite base  $\mathscr{B}$  such that dim Fr  $U \leq n-1$  for every  $U \in \mathscr{B}$ .

**Proof.** We shall show first that (a)  $\Rightarrow$  (b). Consider a metrizable space X such that dim  $X \leq n$ ; let  $\varrho$  be a metric on the space X. The space X being paracompact, for i = 1, 2, ... there exists a locally finite open cover  $\mathscr{U}_i$ =  $\{U_s\}_{s\in S_i}$  of the space X such that mesh  $\mathscr{U}_i < 1/i$ ; obviously, one can assume that  $S_i \cap S_j = \emptyset$  whenever  $i \neq j$ . Let  $\{F_s\}_{s\in S_i}$  be a closed shrinking of the cover  $\mathscr{U}_i$  and let  $S = \bigcup_{i=1}^{\infty} S_i$ . Applying Lemma 4.2.1 to the families  $\{U_s\}_{s\in S}$  and  $\{F_s\}_{s\in S}$  we obtain a family  $\mathscr{B} = \{V_s\}_{s\in S}$  which has the properties stated in (b).

The implication (c)  $\Rightarrow$  (a) follows from Lemmas 2.3.16 and 3.1.27 and the sum theorems for dim.

To prove that (b)  $\Rightarrow$  (c) we shall apply induction with respect to n. Condition (b) and (c) are equivalent if n = 0, because then they both mean that all members of  $\mathscr{B}$  are open-and-closed. Assume that the implication (b)  $\Rightarrow$  (c) is proved for all metrizable spaces and every n < m and consider a metrizable space X which has a  $\sigma$ -locally finite base  $\mathscr{B}$  such that ord({Fr U:  $U \in \mathscr{B}$ })  $\leq m$ . For every  $U_0 \in \mathscr{B}$  the family  $\mathscr{B}_0 = \{X_0 \cap U: U \in \mathscr{B}\}$ , where  $X_0 = \operatorname{Fr} U_0$ , is a  $\sigma$ -locally finite base for the space  $X_0$ . Since the family of boundaries of members of  $\mathscr{B}_0$  in the space  $X_0$  has order  $\leq m-1$ , it follows from the inductive assumption and implication (c)  $\Rightarrow$  (a) that dim  $X_0 \leq m$ . Thus the space X satisfies (c) with n = m+1 and the proof that (b)  $\Rightarrow$  (c) is completed.  $\Box$ 

Let us note that the equivalence of conditions (a) and (c) in Theorem 4.2.2 follows immediately from Theorems 4.1.3 and 4.1.15. In our proof, however, we applied Lemma 3.1.27, rather than those two theorems, in order to prepare the ground for another proof of the Katětov-Morita theorem (cf. Problem 4.2.A(a)).

We shall now introduce two topological notions which are applied in the next theorem. A sequence  $\mathscr{W}_1, \mathscr{W}_2, \ldots$  of covers of a topological space X is called a *development* for the space X if all covers  $\mathscr{W}_i$  are open and for every point  $x \in X$  and each neighbourhood U of the point x there exists a natural number *i* such that  $St(x, \mathscr{W}_i) \subset U$ . One easily observes that a sequence  $\mathscr{W}_1, \mathscr{W}_2, \ldots$  of open covers of a topological space X is a development for X if and only if for every point  $x \in X$  each family  $\{W_i\}_{i=1}^{\infty}$ of open subsets of X such that  $x \in W_i \in \mathcal{W}_i$  for i = 1, 2, ... is a base for X at the point x. A sequence  $\mathcal{W}_1, \mathcal{W}_2, ...$  of covers of a topological space X is called a *strong development* for the space X if all covers  $\mathcal{W}_i$  are open and for every point  $x \in X$  and each neighbourhood U of the point x there exist a neighbourhood V of the point x and a natural number i such that  $St(V, \mathcal{W}_i) \subset U$ . Clearly, every strong development is a development.

**4.2.3. Theorem.** For every metrizable space X the following conditions are equivalent:

- (a) The space X satisfies the inequality  $\dim X \leq n$ .
- (b) The space X has a development  $\mathscr{W}_1, \mathscr{W}_2, ...$  such that  $\operatorname{ord} \mathscr{W}_i \leq n$ and  $\mathscr{W}_{i+1}$  is a star refinement of  $\mathscr{W}_i$  for i = 1, 2, ...
- (c) The space X has a strong development  $\mathscr{W}_1, \mathscr{W}_2, ...$  such that  $\operatorname{ord} \mathscr{W}_i$ ,  $\leq n$  and  $\mathscr{W}_{i+1}$  is a refinement of  $\mathscr{W}_i$  for i = 1, 2, ...

**Proof.** We begin with the implication (a)  $\Rightarrow$  (b). Consider a metrizable space X such that dim  $X \le n$ . We shall define inductively a sequence  $\mathscr{W}_1, \mathscr{W}_2, \ldots$  of open covers of X which has the properties stated in (b). Assume that k = 1 or that k > 1 and the covers  $\mathscr{W}_i$  are defined for all i < k. The space X being paracompact, the open cover  $\mathscr{W}_{k-1} \land \{B(x, 1/k)\}_{x \in X}$  of the space X, where  $\mathscr{W}_0 = \{X\}$ , has an open star refinement (see [GT], Theorem 5.1.12) which by virtue of Proposition 3.2.2 has in turn an open shrinking  $\mathscr{W}_k$  such that  $\operatorname{ord} \mathscr{W}_k \le n$ . The sequence  $\mathscr{W}_1, \mathscr{W}_2, \ldots$  thus obtained has the required properties, so that (a)  $\Rightarrow$  (b).

The implication (b)  $\Rightarrow$  (c) follows from the fact that if  $St(x, \mathscr{W}_i) \subset U$ and  $\mathscr{W}_{i+1}$  is a star refinement of  $\mathscr{W}_i$  then  $St(V, \mathscr{W}_{i+1}) \subset U$  for any  $V \in \mathscr{W}_{i+1}$  such that  $x \in V$ .

Finally, the implication (c)  $\Rightarrow$  (a) follows from Lemma 4.1.1, because as one readily sees by applying the definition twice—if  $\mathscr{W}_1, \mathscr{W}_2, \ldots$  is a strong development for a space X and  $\mathscr{W}_{i+1}$  is a refinement of  $\mathscr{W}_i$  for  $i = 1, 2, \ldots$ , then the family {St( $\mathcal{W}, \mathscr{W}_i$ ):  $\mathcal{W} \in \mathscr{W}_i, i = 1, 2, \ldots$ } is a base for X.  $\square$ 

Let us note that by virtue of the Nagata-Smirnov theorem (see [GT], Theorem 4.4.7), Theorem 4.2.2 holds for every regular space X if condition (a) is replaced by the following condition:

(a') The space X is metrizable and satisfies the inequality dim  $X \leq n$ .

Similarly, by virtue of appropriate metrization theorems (see [GT], Corollary 5.4.10 and Theorem 5.4.2), Theorem 4.2.3 holds for every  $T_0$ -space X if condition (a) is replaced by condition (a').

As follows from Problem 4.1.G, the class of all metrizable spaces X such that  $\dim X \leq 0$  can be characterized in terms of the existence of special metrics. In this connection it is natural to ask whether the class of all metrizable spaces X such that  $\dim X \leq n$  can be characterized in a similar manner. Investigations in this direction led to a group of interesting theorems, which are reviewed below; the proofs of those theorems are too difficult to be reproduced in this book. Thus, a metrizable space X satisfies the inequality  $\dim X \leq n \geq 0$  if and only if on the space X there exists a metric  $\varrho$  which satisfies any of the following conditions:

- (i) For every point  $x \in X$  and each positive number r we have dim Fr  $B(x, r) \leq n-1$ ; moreover,  $\bigcup_{x \in X_0} \overline{B(x, r)} = \bigcup_{x \in X_0} \overline{B(x, r)}$  for every  $X_0 \subset X$ .
- (ii) For every closed set  $A \subset X$  and each positive number r we have dim  $\operatorname{Fr} B(A, r) \leq n-1$ .
- (iii) For every point  $x \in X$ , each positive number r and every sequence  $y_1, y_2, ..., y_{n+2}$  of n+2 points of X satisfying the inequality  $\varrho(y_i, B(x, r/2)) < r$  for i = 1, 2, ..., n+2, there exist natural numbers  $i, j \leq n+2$  such that  $i \neq j$  and  $\varrho(y_i, y_j) < r$ .
- (iv) For every point  $x \in X$  and every sequence  $y_1, y_2, ..., y_{n+2}$  of n+2 points of X there exist natural numbers  $i, j \leq n+2$  such that  $i \neq j$  and  $\varrho(y_i, y_j) \leq \varrho(x, y_i)$ .

One easily verifies that every metric  $\varrho$  which satisfies (i) satisfies also (ii), and that if on a space X there exists a metric  $\varrho$  which satisfies (ii), then dim  $X \leq n$  (see Problem 4.2.D). Let us note that a separable metrizable space X satisfies the inequality dim  $X \leq n \geq 0$  if and only if on the space X there exists a metric  $\varrho$  which satisfies the first part of (i); it is not known whether the second part of (i) can be omitted also in the case of an arbitrary metrizable space.

An interesting problem is connected with conditions (iii) and (iv), namely it is not known whether the inequality  $\dim X \leq n$  follows from the existence, on the space X, of a metric  $\varrho$  which satisfies the following condition (v) For every point  $x \in X$ , each positive number r and every sequence  $y_1, y_2, ..., y_{n+2}$  of n+2 points of X satisfying the inequality  $\varrho(x, y_i) < r$  for i = 1, 2, ..., n+2, there exist natural numbers  $i, j \le n+2$  such that  $i \ne j$  and  $\varrho(y_i, y_j) < r$ .

One easily verifies that every metric  $\varrho$  which satisfies (iii) satisfies also (v), so that the problem stated above is equivalent to the question whether the existence on a space X of a metric  $\varrho$  which satisfies (v) means that dim  $X \leq n$ . Similarly, one can readily check that condition (v) is equivalent to the following condition:

(v') For every point  $x \in X$  and every sequence  $y_1, y_2, ..., y_{n+2}$  of n+2points of X there exist natural numbers  $i, j, k \leq n+2$  such that  $i \neq j$  and  $\varrho(y_i, y_j) \leq \varrho(x, y_k)$ ,

and that every metric  $\rho$  which satisfies (iv) satisfies also (v') and (v). Let us note that if on a separable metrizable space X there exists a totally bounded metric  $\rho$  which satisfies (v), then dim  $X \leq n$  (see Problem 4.2.F); it is not known if the assumption of total boundedness can be omitted.

We now turn to the universal space theorem. As in the case of normal spaces discussed in Section 3.3, we shall deduce this theorem from a factorization theorem. We start with a simple lemma on paracompact spaces.

**4.2.4. Lemma.** Every locally finite open cover  $\mathcal{U}$  of a paracompact space X has a locally finite open star refinement  $\mathscr{V}$  such that  $|\mathscr{V}| \leq \max(|\mathscr{U}|, \aleph_0)$ .

**Proof.** Let  $\mathscr{W}$  be an open star refinement of  $\mathscr{U}$  (see [GT], Theorem 5.1.12). For every pair  $\mathscr{U}_0, \mathscr{U}_1$  of finite subfamilies of the cover  $\mathscr{U}$  denote by  $\mathscr{W}(\mathscr{U}_0, \mathscr{U}_1)$  the family of all  $W \in \mathscr{W}$  such that  $\mathscr{U}_0 = \{U \in \mathscr{U} : W \subset U\}$ and  $\mathscr{U}_1 = \{U \in \mathscr{U} : \operatorname{St}(W, \mathscr{W}) \subset U\}$ . From the local finiteness of  $\mathscr{U}$ it follows that all the sets of the form  $\bigcup \mathscr{W}(\mathscr{U}_0, \mathscr{U}_1)$  constitute an open cover  $\mathscr{V}'$  of the space X such that  $|\mathscr{V}'| \leq \max(|\mathscr{U}|, \aleph_0)$ . We shall show that  $\mathscr{V}'$  is a star refinement of  $\mathscr{U}$ .

Consider a set  $V = \bigcup \mathscr{W}(\mathscr{U}_0, \mathscr{U}_1) \in \mathscr{V}'$ . Let U be an arbitrary member of  $\mathscr{U}_1$ ; clearly

(4)  $\operatorname{St}(W, \mathscr{W}) \subset U$  for every  $W \in \mathscr{W}(\mathscr{U}_0, \mathscr{U}_1)$ .

For every set  $V' = \bigcup \mathscr{W}(\mathscr{U}_0, \mathscr{U}_1) \in \mathscr{V}'$  which intersects V there exist  $W \in \mathscr{W}(\mathscr{U}_0, \mathscr{U}_1)$  and  $W' \in \mathscr{W}(\mathscr{U}_0, \mathscr{U}_1)$  such that  $W \cap W' \neq \emptyset$ . From

(4) it follows that  $W' \subset U$ , so that  $U \in \mathscr{U}'_0$ , which implies that  $V' \subset U$ . Hence,  $St(V, \mathscr{V}') \subset U$ .

To complete the proof it suffices to consider a locally finite open shrinking  $\mathscr{V}$  of the cover  $\mathscr{V}'$ .  $\Box$ 

**4.2.5.** Pasynkov's factorization theorem. For every continuous mapping  $f: X \to Y$  of a metrizable space X to a metrizable space Y there exist a metrizable space Z and continuous mappings  $g: X \to Z$  and  $h: Z \to Y$  such that  $\dim Z \leq \dim X$ ,  $w(Z) \leq w(Y)$ , g(X) = Z and f = hg.

**Proof.** If dim  $X = \infty$  or  $w(Y) < \aleph_0$ , then Z = f(X), g = f and  $h = \operatorname{id}_Z$  satisfy the theorem. Thus one can suppose that dim  $X = n < \infty$  and  $w(Y) = m \ge \aleph_0$ . For i = 1, 2, ... consider a locally finite open cover  $\mathscr{U}_i$  of the space Y such that mesh $\mathscr{U}_i < 1/i$  and  $|\mathscr{U}_i| \le m$ . Applying Lemma 4.2.4 and Proposition 3.2.2 one can easily define by induction a sequence  $\mathscr{V}_1, \mathscr{V}_2, ...$  of locally finite open covers of the space X such that for i = 1, 2, ...

(5) 
$$\operatorname{ord} \mathscr{V}_i \leq n, \quad |\mathscr{V}_i| \leq \mathfrak{m}$$

and

(6) 
$$\mathscr{V}_{i+1}$$
 is a star refinement of  $\mathscr{V}_i \wedge f^{-1}(\mathscr{U}_{i+1})$ .

We shall now consider another topology on the set X, coarser than the original one, which is defined by declaring that a set  $U \subset X$  is open if for every  $x \in U$  there exists a natural number *i* such that  $St(x, \mathscr{V}_i) \subset U$ . The set X with this new topology will be denoted by X'; let us note that generally X' is not a  $T_0$ -space. We shall show that for every  $A \subset X'$  the interior of the set A in the space X' coincides with the set

$$A^* = \{x \in X: \text{ there exists an } i \text{ such that } St(x, \mathscr{V}_i) \subset A\}.$$

Obviously, it suffices to verify that  $A^*$  is an open subset of X'. Consider an arbitrary point  $x \in A^*$  and an *i* such that  $St(x, \mathscr{V}_i) \subset A$ . For every point  $y \in St(x, \mathscr{V}_{i+1})$  we have  $St(y, \mathscr{V}_{i+1}) \subset St(x, \mathscr{V}_i) \subset A$ , so that  $y \in A^*$ . Thus  $St(x, \mathscr{V}_{i+1}) \subset A^*$ , which shows that  $A^*$  is an open subset of X'.

Let  $\mathscr{V}_i^* = (V^*: V \in \mathscr{V}_i)$  for i = 1, 2, ... As for every  $W \in \mathscr{V}_{i+1}$  there exists a  $V \in \mathscr{V}_i$  such that  $\operatorname{St}(W, \mathscr{V}_{i+1}) \subset V$  and consequently  $W \subset V^* \in \mathscr{V}_i^*$ , the family  $\mathscr{V}_i^*$  is an open cover of the space X'. Since  $A^*$  is the interior of A in X', from (6) it follows that for i = 1, 2, ...

(7)  $\mathscr{V}_{i+1}^*$  is a star refinement of  $\mathscr{V}_i^*$ .

Note that by letting f'(x) = f(x) for  $x \in X'$  we define a continuous mapping  $f': X' \to Y$ ; indeed, it follows from (6) that  $\delta(f'(V^*)) < 1/i+1$  for every  $V^* \in \mathscr{V}_{i+1}^*$ .

For  $x, y \in X'$  define

*xEy* if and only if for every *i* there exists a  $V \in \mathscr{V}_i$  such that  $x, y \in V$ . One easily checks that *E* is an equivalence relation on the space X'. Clearly

$$[x] = \bigcap_{i=1}^{\infty} \operatorname{St}(x, \mathscr{V}_i) \quad \text{for } x \in X',$$

where [x] denotes the equivalence class that contains x. Let Z be the quotient space X'/E and  $g: X \to Z$  the composition of the identity mapping  $i: X \to X'$  and the natural quotient mapping  $g': X' \to X'/E$ . By virtue of (6), f'(x) = f'(y) whenever xEy, so that by letting h([x]) = f'(x) we define a mapping h of Z to Y; from the relation hg' = f' it follows that h:  $Z \to Y$  is a continuous mapping. Obviously, g(X) = Z and f = hg. Let us note that

(8) 
$$g'^{-1}g'(A^*) = A^*$$
 for every  $A \subset X'$ ,

so that

(9) the set 
$$g'(A^*)$$
 is open in Z for every  $A \subset X'$ .

We shall show that Z is a metrizable space. To begin with, observe that Z is a  $T_1$ -space. Indeed, if  $[x] \neq [y]$  then there exists an *i* such that  $y \notin St(x, \mathscr{V}_i)$ , and then the set  $g'(\mathscr{V}^*)$ , where  $V = St(x, \mathscr{V}_i)$ , is a neighbourhood of the point [x] which does not contain the point [y]. Let  $\mathscr{W}_i$  $= \{g'(V^*): V \in \mathscr{V}_i\}$  for i = 1, 2, ...; from (9) it follows that  $\mathscr{W}_i$  is an open cover of the space Z. Now, from the definition of topology on X' it follows that the sequence  $\mathscr{V}_1^*, \mathscr{V}_2^*, ...$  is a development for the space X', so that by virtue of (8) the sequence  $\mathscr{W}_1, \mathscr{W}_2, ...$  is a development of  $\mathscr{W}_i$ for i = 1, 2, ... Hence, the space Z is metrizable (see [GT], Corollary 5.4.10).

From the first part of (5) it follows that  $\operatorname{ord} \mathscr{W}_i \leq n$  for i = 1, 2, ..., so that  $\dim Z \leq n$  by virtue of Theorem 4.2.3. To complete the proof it suffices to show that  $w(Z) \leq m$ . The family  $\mathscr{B}$  consisting of all sets  $\operatorname{St}(z, \mathscr{W}_i)$ , where  $z \in Z$  and i = 1, 2, ..., is a base for the space Z. Since

the covers  $\mathscr{V}_i$  of the space X are all locally finite, it follows that the covers  $\mathscr{V}_i^*$  and  $\mathscr{W}_i$  are point-finite, so that  $|\mathscr{B}| \leq \mathfrak{m}$ .  $\Box$ 

Let us note that in the proof of Pasynkov's factorization theorem only the paracompactness of the space X, and not its metrizability, was used; it turns out that the theorem holds under the even weaker assumption of normality of X (see Problem 4.2.G).

**4.2.6. The universal space theorem.** For every integer  $n \ge 0$  and every cardinal number  $m \ge \aleph_0$  there exists a universal space  $J_m^n$  for the class of all metrizable spaces whose covering dimension is not larger than n and whose weight is not larger than m.

**Proof.** Let  $\{X_s\}_{s\in S}$  be the family of all subspaces of the Cartesian product  $[J(\mathfrak{m})]^{\aleph_0}$  of  $\aleph_0$  copies of the hedgehog  $J(\mathfrak{m})$  (see [GT], Example 4.1.5) whose covering dimension is not larger than n, and let  $i_s\colon X_s \to [J(\mathfrak{m})]^{\aleph_0}$  be the embedding of  $X_s$  in  $[J(\mathfrak{m})]^{\aleph_0}$ . Consider the sum  $X = \bigoplus_{s\in S} X_s$  and the mapping  $i\colon X \to [J(\mathfrak{m})]^{\aleph_0}$  defined by letting  $i(x) = i_s(x)$  for  $x \in X_s$ . Since dim $X \leq n$ , by virtue of Theorem 4.2.5 there exist a metrizable space  $J^n_{\mathfrak{m}}$  and continuous mappings  $g\colon X \to J^n_{\mathfrak{m}}$  and  $h\colon J^n_{\mathfrak{m}} \to [J(\mathfrak{m})]^{\aleph_0}$  such that dim $J^n_{\mathfrak{m}} \leq \dim X = n$ ,  $w(J^n_{\mathfrak{m}}) \leq w([J(\mathfrak{m})]^{\aleph_0}) = \mathfrak{m}$  and f = hg.

Consider now an arbitrary metrizable space Y such that dim  $Y \le n$ and  $w(Y) \le m$ . Since Y is embeddable in  $[J(m)]^{\aleph_0}$  (see [GT], Theorem 4.4.9), there exists an  $s \in S$  such that  $X_s$  is homeomorphic to Y. The composition  $h_0g_0$  of the restrictions  $g_0 = g|X_s: X_s \to g(X_s) \subset J_m^n$  and  $h_0$  $= h|g(X_s): g(X_s) \to X_s \subset [J(m)]^{\aleph_0}$  is a homeomorphism, so that  $g_0$  is also a homeomorphism. Thus  $J_m^n$  is the required universal space.  $\square$ 

## Historical and bibliographic notes

Theorem 4.2.2 was established by Morita in [1954]. The equivalence of conditions (a) and (b) in Theorem 4.2.3 was proved by Nagata in [1956a]; the equivalence of conditions (a) and (c) was proved by Nagami and Roberts in [1967]. Characterizations of dimension in terms of metrics satisfying (i) or (ii) were established by Nagata in [1963], the characterization in terms of metrics satisfying (iii) was obtained by Nagata in [1956], and the characterization in terms of metrics satisfying (iv) was obtained independently by Nagata in [1964] and by Ostrand in [1965a]. Let us add

#### Problems

that in [1936] Marczewski proved that a separable metric space X satisfies the inequality dim  $X \le n \ge 0$  if and only if on the space X there exists a metric  $\rho$  such that for every point  $x \in X$  we have dim Fr B(x, r) $\leq n-1$  for almost all (in the sense of Lebesgue measure) positive numbers r. The question whether the inequality dim  $X \leq n$  follows from the existence on the space X of a metric  $\rho$  which satisfies (v) was raised by de Groot in [1957]. Theorem 4.2.5, under the weaker assumption of normality of X(see Problem 4.2.G), was proved by Pasynkov in [1967a] (announcement in [1964]); the present proof is obtained by amalgamating the proofs given by Arhangel'skil in [1967] and by Morita in [1975]. Theorem 4.2.5 for a separable Y and an analogous theorem with no evaluation of w(Z)were given by Pasynkov in [1963]. Theorem 4.2.6 was established by Nagata in [1960a]; the present proof was given by Pasynkov in [1964]. In Nagata's original proof of Theorem 4.2.6 a universal space is explicitly defined (cf. Pears [1975]). Let us mention, in connection with Theorem 4.2.6, that in [1975] Lipscomb defined for every cardinal number  $m \ge \aleph_0$ a metrizable space L(m) with dim L(m) = 1 and w(L(m)) = m such that each metrizable space X satisfying the inequalities  $\dim X \leq n$  and w(X) $\leq$  m is embeddable in the Cartesian product  $[L(m)]^{n+1}$ , and proved that the Cartesian product  $[L(m)]^{n+1}$  contains an easily definable subspace which is a universal space for the class of all metrizable spaces whose covering dimension is not larger than n and whose weight is not larger than m. The fact that each metrizable space X with  $\dim X = n$  is embeddable in the Cartesian product of n+1 metrizable spaces whose covering dimension is equal to 1 was discovered by Nagata in [1958]; Borsuk proved in [1975] that the two-sphere  $S^2$  cannot be embedded in the Cartesian product of two one-dimensional spaces.

## Problems

4.2.A (Morita [1954]). (a) Prove the Katětov-Morita theorem by applying only Lemmas 4.2.1 and 2.3.16 and Theorems 4.1.9, 4.1.10 and 3.1.28.

*Hint*. Adjoin to conditions (a)-(c) in Theorem 4.2.2 condition (d) stating that  $Ind X \le n$ .

(b) Note that in Theorem 4.2.2 one can replace the  $\sigma$ -local finiteness of  $\mathscr{B}$  by  $\sigma$ -discreteness.

**4.2.B.** (a) Show that every locally finite family  $\{F_s\}_{s \in S}$  of closed subsets of a metrizable space X has a locally finite open swelling  $\{U_s\}_{s \in S}$ .

*Hint*. For i = 1, 2, ... consider the set  $A_i$  consisting of all points  $x \in X$ such that the set  $\{s \in S: B(x, 1/i) \cap F_s \neq \emptyset\}$  is finite, let  $V_s = \bigcup_{i=1}^{\infty} [B(F_s, 1/i) \cap \operatorname{Int} A_i]$  and apply Problem 3.1.A.

(b) (Morita [1955], Nagami [1960]) Prove that a metrizable space X satisfies the inequality dim  $X \le n$  if and only if there exists a sequence  $\mathscr{F}_1, \mathscr{F}_2, \ldots$  of locally finite closed covers of the space X such that  $\operatorname{ord} \mathscr{F}_i \le n$  and  $\mathscr{F}_{i+1}$  is a refinement of  $\mathscr{F}_i$  for  $i = 1, 2, \ldots$  and for every point  $x \in X$  and each neighbourhood U of the point x there exists a natural number i such that  $\operatorname{St}(x, \mathscr{F}_i) \subset U$ .

*Hint.* Let  $\mathscr{F}_i = \{F_s\}_{s \in S_i}$  for i = 1, 2, ... Apply part (a) to define a sequence  $\mathscr{W}_1, \mathscr{W}_2, ...$  of open covers of X, where  $\mathscr{W}_i = \{W_s\}_{s \in S_i}$  is a swelling of the cover  $\mathscr{F}_i$  and  $W_s \subset B(F_s, 1/i)$  for  $s \in S_i$ , such that  $\mathscr{W}_{i+1}$  is a refinement of  $\mathscr{W}_i$  for i = 1, 2, ...; check that  $\mathscr{W}_1, \mathscr{W}_2, ...$  is a strong development for the space X.

**4.2.C** (Nagata [1963] (announcement [1961])). Prove that a metrizable space X satisfies the inequality dim  $X \le n$  if and only if X has a base of rank  $\le n$  (see Problem 3.2.C).

*Hint* (Arhangel'skii [1963]). When proving that X has a base of rank  $\leq n$  use the inequality Ind $X \leq n$ ; apply Theorem 4.1.17.

**4.2.D** (Nagata [1963]). (a) Note that every metric  $\rho$  which satisfies condition (i) satisfies also condition (ii).

(b) Show that if on a space X there exists a metric  $\rho$  which satisfies condition (ii), then dim  $X \leq n$ .

*Hint*. See the second part of the hint to Problem 4.1.A(b).

**4.2.E.** Note that for n = 0 conditions (iii), (iv) and (v) reduce to the condition that the metric  $\rho$  is non-Archimedean; check that each non-Archimedean metric  $\rho$  satisfies conditions (i) and (ii) for n = 0.

**4.2.F** (de Groot [1957]). Applying the fact that on every compact metrizable space of dimension  $\leq n$  there exists a metric  $\rho$  satisfying condition (iii), show that a separable metrizable space X satisfies the inequality dim  $X \leq n$  if and only if on the space X there exists a totally bounded metric  $\rho$  which satisfies condition (v).

*Hint.* Note that if  $\varrho$  satisfies (v), then the metric  $\tilde{\varrho}$  on the completion  $\tilde{X}$  of the space X also satisfies (v). For an arbitrary finite open cover  $\mathscr{U}$  of the completion  $\tilde{X}$  consider a set  $A \subset \tilde{X}$  maximal with respect to the property that  $\varrho(y, y') \ge \delta/2$  for distinct  $y, y' \in A$ , where  $\delta$  is a Lebesgue number for the cover  $\mathscr{U}$ .

**4.2.G** (Pasynkov [1967a] (announcement [1964])). Prove that for every continuous mapping  $f: X \to Y$  of a normal space X to a metrizable space X there exist a metrizable space Z and continuous mappings  $g: X \to Z$  and  $h: Z \to Y$  such that dim  $Z \leq \dim X$ ,  $w(Z) \leq w(Y)$ , g(X) = Z and f = hg.

Hint. Generalize Lemma 4.2.4 to normal spaces; make use of the mapping of X to the metrizable space assigned to the cover  $\mathcal{U}$  as described in the hint to Problem 3.2.H(b).

## 4.3. Dimension and mappings in metrizable spaces

In this section we shall study the behaviour of dimension of metrizable spaces under continuous mappings. We start with extending to arbitrary metrizable spaces the theorems on dimension-raising and dimension-lowering mappings established for separable metric spaces in Section 1.12. The proofs of these extensions closely follow the pattern of the proofs in Chapter 1 and differ from them only in technical details.

**4.3.1. Lemma.** A metrizable space X satisfies the inequality  $\operatorname{Ind} X \leq n \geq 0$  if and only if X has a  $\sigma$ -locally finite network  $\mathcal{N}$  such that  $\operatorname{Ind} \operatorname{Fr} M \leq n-1$  for every  $M \in \mathcal{N}$ .

**Proof.** By virtue of Theorem 4.1.15, it suffices to show that if a metrizable space X has a  $\sigma$ -locally finite network  $\mathcal{N}$  such that  $\operatorname{Ind} \operatorname{Fr} M \leq n-1$  for  $M \in \mathcal{N}$ , then  $\operatorname{Ind} X \leq n$ . Let  $Y = \bigcup {\operatorname{Fr} M: M \in \mathcal{N}}$  and  $Z = X \setminus Y$ . Since the family  ${\operatorname{Fr} M: M \in \mathcal{N}}$  is  $\sigma$ -locally finite, it follows from Theorems 4.1.9 and 4.1.10 that  $\operatorname{Ind} Y \leq n-1$ . The family  ${Z \cap M: M \in \mathcal{N}}$  is a  $\sigma$ -locally finite network for the subspace Z; as the members of this family are open-and-closed in Z, it is a base for Z, so that  $\operatorname{Ind} Z \leq 0$  by virtue of Theorem 4.1.15. The inequality  $\operatorname{Ind} X \leq n$  now follows from Theorem 4.1.16.  $\Box$ 

Let us recall that a continuous mapping  $f: X \to Y$  defined on a Hausdorff space X is *perfect* if f is a closed mapping and for every  $y \in Y$  the fibre  $f^{-1}(y)$  is a compact subspace of X. Obviously, every closed mapping with finite fibres defined on a Hausdorff space is perfect.

**4.3.2. Lemma.** If  $f: X \to Y$  is a perfect mapping, then for every locally finite family  $\mathcal{A}$  of subsets of X the family  $\{f(A): A \in \mathcal{A}\}$  is locally finite in Y.

**Proof.** Since the fibres of f are compact, for every  $y \in Y$  there exists an open set  $U \subset X$  which contains  $f^{-1}(y)$  and meets only finitely many members of  $\mathscr{A}$ . The mapping f being closed, the set  $V = Y \setminus f(X \setminus U)$  is a neighbourhood of the point y. From the inclusion  $f^{-1}(V) \subset U$  it follows that V meets only finitely many members of the family  $\{f(A): A \in \mathscr{A}\}$ .  $\Box$ 

**4.3.3. Theorem on dimension-raising mappings.** If  $f: X \to Y$  is a closed mapping of a metrizable space X onto a metrizable space Y and there exists an integer  $k \ge 1$  such that  $|f^{-1}(y)| \le k$  for every  $y \in Y$ , then  $\operatorname{Ind} Y \le \operatorname{Ind} X + (k-1)$ .

**Proof.** We can suppose that  $0 \leq \text{Ind}X < \infty$ . We shall apply induction with respect to the number n+k, where n = IndX. If n+k = 1, we have k = 1, so that f is a homeomorphism and the theorem holds. Assume that the theorem holds whenever  $n+k < m \ge 2$  and consider a closed mapping  $f: X \to Y$  such that f(X) = Y and n+k = m.

Let  $\mathscr{B}$  be a  $\sigma$ -locally finite base for X such that  $\operatorname{Ind} \operatorname{Fr} U \leq n-1$  for every  $U \in \mathscr{B}$ . Consider an arbitrary  $U \in \mathscr{B}$ ; by the closedness of f we have

(1) 
$$\operatorname{Fr} f(U) = \overline{f(U)} \cap \overline{Y \setminus f(U)} \subset f(\overline{U}) \cap f(X \setminus U)$$
$$= [f(U) \cup f(\operatorname{Fr} U)] \cap f(X \setminus U) \subset f(\operatorname{Fr} U) \cup B$$

where  $B = f(U) \cap f(X \setminus U)$ . Since the restriction  $f | \operatorname{Fr} U$ :  $\operatorname{Fr} U \to f(\operatorname{Fr} U)$  is a closed mapping, it follows from the inductive assumption that

$$\operatorname{Ind} f(\operatorname{Fr} U) \leq (n-1) + (k-1) = n+k-2.$$

Assume that  $B \neq \emptyset$ . Consider the restriction  $f_B: f^{-1}(B) \to B$  and the restriction  $f' = f_B|(X \setminus U): (X \setminus U) \cap f^{-1}(B) \to B$ ; both  $f_B$  and f' are closed, and the fibres of f' all have cardinality  $\leq k-1$ , because  $f^{-1}(y) \cap U \neq \emptyset$  for every  $y \in B$ . It follows from the inductive assumption that

Ind 
$$B \le n + (k-1) - 1 = n + k - 2$$
.

As U is an  $F_{\sigma}$ -set in X, both f(U) and B are  $F_{\sigma}$ -sets in Y; applying Theorems 4.1.7 and 4.1.9, we obtain the inequality  $\text{Ind}[f(\text{Fr }U)\cup B] \leq n+k-2$ .

From the last inequality and from (1) it follows that  $\operatorname{Ind} \operatorname{Fr} f(U) \leq n+k-2$ for every  $U \in \mathscr{B}$ ; the same inequality holds if  $B = \mathscr{O}$ . One readily checks that the family  $\mathscr{N} = \{f(U): U \in \mathscr{B}\}$  is a network for the space Y. In view of Lemma 4.3.2 the network  $\mathscr{N}$  is  $\sigma$ -locally finite, so that  $\operatorname{Ind} Y$  $\leq n+k-1 = \operatorname{Ind} X+k-1$  by virtue of Lemma 4.3.1.  $\Box$ 

We now pass to the theorem on dimension-lowering mappings. To begin with, let us note that reproducing the proof of Lemma 1.9.6 with the application of Theorems 3.2.9, 4.1.3, 4.1.7 and 4.1.21 one obtains

**4.3.4. Lemma.** Let  $f, g: X \to S^n$  be continuous mappings of a metrizable space X to the n-sphere  $S^n$ . If the set

$$D(f,g) = \{x \in X \colon f(x) \neq g(x)\}$$

satisfies the inequality  $\operatorname{Ind} D(f, g) \leq n-1$ , then the mappings f and g are homotopic.  $\Box$ 

Now we shall establish a counterpart of Lemma 1.12.3.

**4.3.5. Lemma.** If a metrizable space X has a closed cover  $\mathcal{K}$  such that  $\operatorname{Ind} K \leq m \geq 0$  for each  $K \in \mathcal{K}$  and a  $\sigma$ -locally finite open cover  $\mathcal{U}$  such that for every  $K \in \mathcal{K}$  and each open set  $V \subset X$  that contains K there exists a  $U \in \mathcal{U}$  satisfying

 $K \subset U \subset \overline{U} \subset V$  and  $\operatorname{Ind} \operatorname{Fr} U \leq m-1$ ,

then  $\operatorname{Ind} X \leq m$ .

**Proof.** By virtue of Theorem 4.1.3 and Remarks 1.7.10 and 1.9.4 it suffices to show that for every closed subspace A of the space X and each continuous mapping  $f: A \to S^m$  there exists a continuous extension  $F: X \to S^m$  of f over X. To begin with, let us observe that

(2) for every  $K \in \mathscr{K}$  there exists a  $U_K \in \mathscr{U}$  such that  $K \subset U_K$ , Ind Fr  $U_K \leq m-1$  and f is continuously extendable over  $A \cup \overline{U}_K$ .

Indeed, it follows from Theorem 3.2.9 (cf. Problem 4.3.B(a)) that the mapping f is continuously extendable over  $A \cup K$ , so that there exists an open set  $V \subset X$  containing  $A \cup K$  such that f is continuously extendable over V; the existence of a set  $U_{\kappa}$  satisfying (2) now follows from the properties of the family  $\mathcal{U}$ .

The subfamily  $\mathcal{U}_0$  of the family  $\mathcal{U}$  which consists of all sets of the form  $U_{\kappa}$  can be represented as the union of a sequence  $\mathcal{U}_1, \mathcal{U}_2, \ldots$  of locally finite families. Let us arrange the members of  $\mathcal{U}_0$  into a transfinite sequence  $U_1, U_2, \ldots, U_{\alpha}, \ldots, \alpha < \xi$  placing first all members of  $\mathcal{U}_1$ , then all members of  $\mathcal{U}_2$  etc. Clearly, for every  $\alpha_0 < \xi$  the family  $\{U_{\alpha}\}_{\alpha < \alpha_0}$  is locally finite.

We shall inductively define a transfinite sequence  $F_1, F_2, ..., F_{\alpha}, ..., \alpha < \xi$  of continuous mappings, where  $F_{\alpha}: A \cup \bigcup_{\beta \leq \alpha} \overline{U}_{\beta} \to S^m$ , such that

(3) 
$$F_{\alpha}|(A \cup \bigcup_{\beta \leqslant \gamma} \overline{U}_{\beta}) = F_{\gamma} \text{ for every } \gamma \leqslant \alpha.$$

Let  $F_1$  be an arbitrary continuous extension of f over  $A \cup \overline{U}_1$ . Assume that the mappings  $F_{\alpha}$  satisfying (3) are defined for  $\alpha < \alpha_0$ . The set  $A \cup \bigcup_{\alpha \leqslant \alpha_0} \overline{U}_{\alpha}$  can be represented as the union of two closed sets

$$A' = A \cup \bigcup_{\alpha < \alpha_0} \widetilde{U}_{\alpha}$$
 and  $A'' = A \cup (\widetilde{U}_{\alpha_0} \setminus \bigcup_{\alpha < \alpha_0} U_{\alpha})$ 

which, by virtue of the local finiteness of the family  $\{U_{\alpha}\}_{\alpha < \alpha_0}$ , satisfy the relation

(4) 
$$A' \cap A'' \subset A \cup \operatorname{Fr}\left(\bigcup_{\alpha < \alpha_0} U_{\alpha}\right) \subset A \cup \bigcup_{\alpha < \alpha_0} \operatorname{Fr} U_{\alpha}.$$

Since the family  $\{A\} \cup \{\overline{U}_{\alpha}\}_{\alpha < \alpha_0}$  is locally finite and  $\{f\} \cup \{F_{\alpha} \mid \overline{U}_{\alpha}\}_{\alpha < \alpha_0}$  is a family of compatible mappings, one can define a continuous mapping  $F': A' \to S^m$  which is a common extension of all mappings  $F_{\alpha}$  with  $\alpha < \alpha_0$ . By virtue of (2) the mapping f is extendable to a continuous mapping f'': $A'' \to S^m$ , and in view of (4)

$$D = \{x \in A' \cap A'': F'(x) \neq f''(x)\} \subset \bigcup_{\alpha < \alpha_0} \operatorname{Fr} U_{\alpha},$$

so that  $\operatorname{Ind} D \leq m-1$  by virtue of (2) and Theorem 4.1.10. It follows from Lemma 4.3.4 that the mappings  $F'|A' \cap A''$  and  $f''|A' \cap A''$  are homotopic. Since the mapping  $f''|A' \cap A''$  is continuously extendable over A'', it follows from Lemma 1.9.7 that the mapping  $F'|A' \cap A''$  is extendable to a continuous mapping  $F'': A'' \to S^m$ . Letting

$$F_{\alpha_0}(x) = \begin{cases} F'(x) & \text{for } x \in A', \\ F''(x) & \text{for } x \in A'', \end{cases}$$

we define a continuous mapping  $F_{\alpha_0}$  of  $A' \cup A'' = A \cup \bigcup_{\alpha \leq \alpha_0} \overline{U}_{\alpha}$  to  $S^m$ , which satisfies (3) for  $\alpha = \alpha_0$ .

As  $X = \bigcup_{\alpha < \xi} U_{\alpha}$ , the formula

 $F(x) = F_{\alpha}(x)$  for  $x \in U_{\alpha}$ 

defines a continuous mapping  $F: X \to S^m$ , which is the required extension of f over X.  $\Box$ 

**4.3.6. Theorem on dimension-lowering mappings.** If  $f: X \to Y$  is a closed mapping of a metrizable space X to a metrizable space Y and there exists an integer  $k \ge 0$  such that  $\operatorname{Ind} f^{-1}(y) \le k$  for every  $y \in Y$ , then  $\operatorname{Ind} X \le \operatorname{Ind} Y + k$ .

**Proof.** We can suppose that  $\operatorname{Ind} Y < \infty$ . We shall apply induction with respect to  $n = \operatorname{Ind} Y$ . If n = -1, we have  $Y = \emptyset$  and  $X = \emptyset$ , and so the theorem holds. Assume that the theorem holds for closed mappings to spaces of large inductive dimension less than  $n \ge 0$  and consider a closed mapping  $f: X \to Y$  to a space Y such that  $\operatorname{Ind} Y = n$ .

Let  $\mathscr{B}$  be a  $\sigma$ -locally finite base for Y such that  $\operatorname{Ind} \operatorname{Fr} U \leq n-1$  for every  $U \in \mathscr{B}$ . Applying the inductive assumption to the restriction  $f_{\operatorname{Fr} U}$ :  $f^{-1}(\operatorname{Fr} U) \to \operatorname{Fr} U$ , we obtain the inequality  $\operatorname{Ind} f^{-1}(\operatorname{Fr} U) \leq n+k-1$ . Since

$$\operatorname{Fr} f^{-1}(U) = \overline{f^{-1}(U)} \setminus f^{-1}(U) \subset f^{-1}(\overline{U}) \setminus f^{-1}(U) = f^{-1}(\operatorname{Fr} U),$$

we have  $\operatorname{Ind} \operatorname{Fr} f^{-1}(U) \leq n+k-1$  for every  $U \in \mathscr{B}$ . One readily checks that the covers  $\mathscr{K} = \{f^{-1}(y)\}_{y \in Y}$  and  $\mathscr{U} = f^{-1}(\mathscr{B})$  satisfy the conditions of Lemma 4.3.5 with m = n+k, so that  $\operatorname{Ind} X \leq n+k = \operatorname{Ind} Y+k$ .  $\Box$ 

We now turn to two more specialized theorems on the relations between the dimensions of the domain and the range of a closed mapping. In our considerations an important role will be played by the sets  $C_k(f)$  and  $D_k(f)$  which for each closed mapping  $f: X \to Y$  of a normal space X to a normal space Y and for k = 1, 2, ... are defined by the formulas

and  

$$C_{k}(f) = \{ y \in Y : |f^{-1}(y)| \ge k \}$$

$$D_{k}(f) = \{ y \in Y : \dim f^{-1}(y) \ge k \}$$

**4.3.7. Lemma.** For every closed mapping  $f: X \to Y$  of a metrizable space X to a metrizable space Y and for  $k = 1, 2, ..., C_k(f)$  and  $D_k(f)$  are  $F_{\sigma}$ -sets in Y.

**Proof.** Consider first the sets  $C_k(f)$ . For i = 1, 2, ... let  $\mathcal{U}_i$  be an open cover of the space X such that mesh $\mathcal{U}_i < 1/i$  and let

[Ch. 4, § 3

 $U_i = \{ y \in Y : \text{ there exists a } \mathcal{U} \subset \mathcal{U}_i \text{ such that } |\mathcal{U}| < k \text{ and } f^{-1}(y) \subset \bigcup \mathcal{U} \}.$ 

From the closedness of the mapping f it follows that the sets  $U_i$  are open.

One easily checks that  $Y \setminus C_k(f) = \bigcap_{i=1}^{\infty} U_i$ , so that  $C_k(f)$  is an  $F_{\sigma}$ -set in Y.

Now consider the sets  $D_k(f)$ . We start with a particular case where f is a perfect mapping. For i = 1, 2, ... let  $V_i$  be the set of all points  $y \in Y$  for which there exists a finite family  $\mathcal{U}$  of open subsets of X such that  $\operatorname{mesh}\mathcal{U} < 1/i$ ,  $\operatorname{ord}\mathcal{U} < k$  and  $f^{-1}(y) \subset \bigcup \mathcal{U}$ . From the closedness of the mapping f it follows that the sets  $V_i$  are open. Applying the compactness of fibres of the mapping f and Theorems 1.6.12, 3.1.3 and 3.1.2, one easily

checks that  $Y \ D_k(f) = \bigcap_{i=1}^{\infty} V_i$ , so that  $D_k(f)$  is an  $F_{\sigma}$ -set in Y.

For an arbitrary closed mapping  $f: X \to Y$  consider the subspaces  $X_0 = \bigcup_{y \in Y} \operatorname{Int} f^{-1}(y)$  and  $X_1 = X \setminus X_0 = \bigcup_{y \in Y} \operatorname{Fr} f^{-1}(y)$  of the space X and the restrictions  $f_0 = f | X_0: X_0 \to Y$  and  $f_1 = f | X_1: X_1 \to Y$ . From the countable sum theorem it follows that  $D_k(f) = D_k(f_0) \cup D_k(f_1)$ , because  $\operatorname{Int} f^{-1}(y)$  is for every  $y \in Y$  an  $F_{\sigma}$ -set in X. Being the image of an  $F_{\sigma}$ -set in X under a closed mapping, the set

$$D_k(f_0) = f(\bigcup \{\operatorname{Int} f^{-1}(y) \colon \operatorname{dim} \operatorname{Int} f^{-1}(y) \ge k\})$$

is an  $F_{\sigma}$ -set in Y. From Lemma 1.12.9 it follows that  $f_1$  is a perfect mapping, so that the set  $D_k(f_1)$  is an  $F_{\sigma}$ -set in Y by virtue of the particular case of our theorem established in the preceding paragraph. Thus  $D_k(f)$  is an  $F_{\sigma}$ -set in Y.  $\Box$ 

**4.3.8. Lemma.** If X is a metrizable space and  $M_1, M_2, ..., M_k$  is a sequence of subspaces of X such that  $M_i$  is an  $F_{\sigma}$ -set in X and  $\operatorname{Ind} M_i \leq n_i \geq 0$  for i = 1, 2, ..., k, then for every pair A, B of disjoint closed subsets of X there exists a partition L between A and B such that  $\operatorname{Ind}(M_i \cap L) \leq n_i - 1$  for i = 1, 2, ..., k.

**Proof.** By the first decomposition theorem  $M_i = Y_i \cup Z_i$ , where  $\operatorname{Ind} Y_i \leq n_i - 1$  and  $\operatorname{Ind} Z_i \leq 0$ . From the enlargement theorem it follows that there exists a  $G_{\delta}$ -set  $Y_i^*$  in  $M_i$  such that  $Y_i \subset Y_i^*$  and  $\operatorname{Ind} Y_i^* \leq n_i - 1$ . The set  $Z_i^* = M_i \setminus Y_i^* \subset Z_i$  is an  $F_{\sigma}$ -set in  $M_i$  and consequently an  $F_{\sigma}$ -set in X. By virtue of the countable sum theorem the set  $Z = Z_1^* \cup Z_2^* \cup \ldots \cup Z_k^*$  satisfies the inequality  $\operatorname{Ind} Z \leq 0$ , so that from the separation theorem it follows that there exists a partition L between A and B such that  $L \cap Z$ 

= Ø. Hence for i = 1, 2, ..., k we have  $M_i \cap L \subset Y_i^*$ , which implies that  $\operatorname{Ind}(M_i \cap L) \leq n_i - 1$  for i = 1, 2, ..., k.

If for a continuous mapping  $f: X \to Y$  of a metrizable space X to a metrizable space Y there exists an integer  $n \ge 0$  such that  $\operatorname{Ind} f^{-1}(y) \le n$ for every  $y \in Y$ , then we say that f is *finite-dimensional*; the smallest n with this property will be denoted by  $\operatorname{Ind} f$ . In order to simplify the statements of the two theorems we are now going to prove, for every finitedimensional mapping f we define

and 
$$D(f) = \max\{\operatorname{Ind} D_i(f) + i: i = 1, 2, ..., \operatorname{Ind} f\} \text{ if } \operatorname{Ind} f \ge 1$$
$$D(f) = -1 \text{ if } \operatorname{Ind} f = 0.$$

**4.3.9.** Vainštein's first theorem. Let  $f: X \to Y$  be a finite-dimensional closed mapping of a metrizable space X to a metrizable space Y. If  $\operatorname{Ind} Y \leq n \geq 0$  and  $\operatorname{Ind} D_i(f) \leq n-i$  for i = 1, 2, ..., n+1, then  $\operatorname{Ind} X \leq n$ ; in other words,  $\operatorname{Ind} X \leq \max \{\operatorname{Ind} Y, D(f)\}$ .

**Proof.** We shall apply induction with respect to n. If n = 0, we have  $\operatorname{Ind} D_1(f) = -1$ , so that  $\operatorname{Ind} f = 0$  and consequently  $\operatorname{Ind} X \leq n$  by virtue of Theorem 4.3.6. Assume that the theorem holds for all natural numbers less than  $n \geq 1$  and consider a closed mapping  $f: X \to Y$  which satisfies the assumptions of the theorem.

From Lemma 4.3.7 it follows that  $D_1(f)$ ,  $D_2(f)$ , ...,  $D_n(f)$  are  $F_{\sigma}$ -sets in Y. Applying Lemma 4.3.8 one can easily define (cf. the proof of Proposition 4.1.14) a  $\sigma$ -locally finite base  $\mathscr{B}$  for the space Y such that for each  $i \leq n$ 

 $\operatorname{Ind}(D_i(f) \cap \operatorname{Fr} U) \leq \operatorname{Ind} D_i(f) - 1$  for every  $U \in \mathscr{B}$ ,

provided that  $D_i(f) \neq \emptyset$ , and that  $\operatorname{Ind} \operatorname{Fr} U \leq n-1$  for every  $U \in \mathscr{B}$ . Consider the restriction  $f_{\operatorname{Fr} U}$ :  $f^{-1}(\operatorname{Fr} U) \to \operatorname{Fr} U$ , where  $U \in \mathscr{B}$ ; clearly,  $D_i(f_{\operatorname{Fr} U}) \subset D_i(f) \cap \operatorname{Fr} U$ . Since  $\operatorname{Ind} \operatorname{Fr} U \leq n-1 \geq 0$  and  $\operatorname{Ind} D_i(f_{\operatorname{Fr} U}) \leq (n-1)-i$  for i = 1, 2, ..., n, applying the inductive assumption to  $f_{\operatorname{Fr} U}$  we obtain the inequality  $\operatorname{Ind} f^{-1}(\operatorname{Fr} U) \leq n-1$ . Hence, by virtue of the inclusion  $\operatorname{Fr} f^{-1}(U) \subset f^{-1}(\operatorname{Fr} U)$  we have  $\operatorname{Ind} \operatorname{Fr} f^{-1}(U) \leq n-1$ for every  $U \in \mathscr{B}$ . Applying the equality  $D_{n+1}(f) = 0$  and the Katětov-Morita theorem, one readily checks that the covers  $\mathscr{K} = \{f^{-1}(y)\}_{y \in Y}$ and  $\mathscr{U} = f^{-1}(\mathscr{B})$  satisfy the conditions of Lemma 4.3.5 with m = n, so that  $\operatorname{Ind} X \leq n$ .  $\Box$ 

The reader can easily verify that Theorem 4.3.9 is a generalization of Theorem 4.3.6.

We now pass to Vaĭnšteĭn's second theorem. It will be preceded by two lemmas of which the second is an important particular case of the theorem.

**4.3.10. Lemma.** If a metrizable space X satisfies the inequality  $\operatorname{Ind} X \leq n \geq 0$ , then for every  $F_{\sigma}$ -set  $C \subset X$  such that  $\operatorname{Ind} C \leq n-1$  there exists an  $F_{\sigma}$ -set  $M \subset X$  satisfying the conditions

Ind  $M \leq 0$ , Ind $(X \setminus M) \leq n-1$  and  $M \cap C = \emptyset$ .

**Proof.** By the first decomposition theorem  $X = Y \cup Z$ , where  $\operatorname{Ind} Y \leq n-1$ and  $\operatorname{Ind} Z \leq 0$ . Obviously, one can assume that  $Y = X \setminus Z$ , and by virtue of the enlargement theorem one can assume that Z is a  $G_{\delta}$ -set in X. Thus Y is an  $F_{\sigma}$ -set and the countable sum theorem yields the inequality  $\operatorname{Ind}(Y \cup C)$  $\leq n-1$ . By virtue of the enlargement theorem there exists a  $G_{\delta}$ -set  $K \subset X$ such that  $Y \cup C \subset K$  and  $\operatorname{Ind} K \leq n-1$ . One easily checks that the set  $M = X \setminus K$  satisfies the required conditions.  $\Box$ 

**4.3.11. Lemma (Freudenthal's theorem).** Let  $f: X \to Y$  be a closed mapping of a metrizable space X onto a metrizable space Y such that  $\operatorname{Ind} f^{-1}(y) \leq 0$  for every  $y \in Y$ . If  $\operatorname{Ind} X \leq n$  and  $\operatorname{Ind} C_2(f) \leq n-1$ , then  $\operatorname{Ind} Y \leq n$ ; in other words,  $\operatorname{Ind} Y \leq \max \{ \operatorname{Ind} X, \operatorname{Ind} C_2(f) + 1 \}$ .

**Proof.** We shall apply induction with respect to n. If n = 0, we have  $C_2(f) = \emptyset$ , so that f is a homeomorphism and  $\operatorname{Ind} Y = \operatorname{Ind} X \leq 0$ . Assume that the theorem holds for all natural numbers less than  $n \ge 1$  and consider a closed mapping  $f: X \to Y$  which satisfies the assumptions of the theorem.

By virtue of Theorem 4.3.6 the set  $C = f^{-1}(C_2(f))$  satisfies the inequality Ind  $C \le n-1$ . Lemma 4.3.7 implies that C is an  $F_{\sigma}$ -set in X, so that Lemma 4.3.10 implies that there exists an  $F_{\sigma}$ -set  $M \subset X$  such that Ind M

 $\leq 0$ ,  $\operatorname{Ind}(X \setminus M) \leq n-1$  and  $M \cap C = \emptyset$ . Let  $M = \bigcup_{i=1}^{\infty} F_i$ , where the sets  $F_i$  are closed in X. For i = 1, 2, ... the restriction of f to  $F_i$  is a homeomorphism of  $F_i$  onto  $f(F_i)$ , so that  $\operatorname{Ind} f(F_i) \leq 0$ . Thus, by the countable sum theorem,  $\operatorname{Ind} f(M) \leq 0$ .

Consider an arbitrary pair A, B of disjoint closed subsets of Y. From Lemma 4.3.8 with k = 2,  $M_1 = f(M)$  and  $M_2 = C_2(f)$  it follows that there exists a partition L between A and B such that  $L \cap f(M) = \emptyset$  and  $\operatorname{Ind}(L \cap C_2(f)) \leq n-2$ . The equality  $L \cap f(M) = \emptyset$  implies that  $f^{-1}(L) \subset X \setminus M$ , so that  $\operatorname{Ind} f^{-1}(L) \leq n-1$ . Consider the restriction  $f_L: f_L^{-1}(L) \to L$ ; since  $C_2(f_L) \subset L \cap C_2(f)$ , we have  $\operatorname{Ind} C_2(f_L) \leq n-2$ , so that by the inductive assumption  $\operatorname{Ind} L \leq n-1$ . Thus  $\operatorname{Ind} Y \leq n$ .  $\Box$ 

**4.3.12.** Vainštein's second theorem. Let  $f: X \to Y$  be a finite-dimensional closed mapping of a metrizable space X onto a metrizable space Y. If  $\operatorname{Ind} X \leq n \geq 1$ ,  $\operatorname{Ind} C_2(f) \leq n-1$  and  $\operatorname{Ind} D_i(f) \leq n-(i+1)$  for i = 1, 2, ..., n, then  $\operatorname{Ind} Y \leq n$ ; in other words,  $\operatorname{Ind} Y \leq \max \{\operatorname{Ind} X, \operatorname{Ind} C_2(f)+1, D(f)+1\}$ .

**Proof.** We shall apply induction with respect to n. If n = 1, we have  $\operatorname{Ind} D_1(f) = -1$ , so that  $\operatorname{Ind} f = 0$  and the theorem reduces to Lemma 4.3.11. Assume that the theorem holds for all natural numbers less than  $n \ge 2$  and consider a closed mapping  $f: X \to Y$  which satisfies the assumption of the theorem.

By virtue of Theorem 4.3.9 the set  $C = f^{-1}(C_2(f))$  satisfies the inequality Ind  $C \leq n-1$ . Hence from Lemmas 4.3.7 and 4.3.10 it follows that there exists an  $F_{\sigma}$ -set  $M \subset X$  such that  $\operatorname{Ind} M \leq 0$ ,  $\operatorname{Ind}(X \setminus M) \leq n-1$  and  $M \cap C = \emptyset$ ; by the countable sum theorem  $\operatorname{Ind} f(M) \leq 0$ .

Consider an arbitrary pair A, B of disjoint closed subsets of Y. From Lemma 4.3.8 with k = n+1,  $M_i = D_i(f)$  for i = 1, 2, ..., n-1,  $M_n = f(M)$  and  $M_{n+1} = C_2(f)$  it follows that there exists a partition L between A and B such that  $L \cap f(M) = \emptyset$ ,  $Ind(C_2(f) \cap L) \leq n-2$  and for each  $i \leq n-1$ 

$$\operatorname{Ind}(D_i(f) \cap L) \leq \operatorname{Ind} D_i(f) - 1,$$

provided that  $D_i(f) \neq \emptyset$ . The equality  $L \cap f(M) = \emptyset$  implies that  $\operatorname{Ind} f^{-1}(L) \leq n-1$ . Consider the restriction  $f_L: f^{-1}(L) \to L$ ; since  $C_2(f_L) \subset C_2(f) \cap L$ and  $D_i(f_L) \subset D_i(f) \cap L$  for i = 1, 2, ..., n-1, we have  $\operatorname{Ind} C_2(f_L) \leq n-2$ and  $\operatorname{Ind} D_i(f_L) \leq (n-1)-(i+1)$  for i = 1, 2, ..., n-1, so that by the inductive assumption  $\operatorname{Ind} L \leq n-1$ . Thus  $\operatorname{Ind} Y \leq n$ .  $\Box$ 

To conclude, we shall state an important theorem which permits us to generalize part of the results obtained in this section to the case of a closed mapping f of a normal space X onto a normal space Y. The proof of this theorem is too complicated to be reproduced in this book; it involves a construction similar to but much more elaborate than the construction used in the proof of Theorem 3.3.2. The symbol  $rd_{x}A$  which appears below denotes the *relative dimension* of a subspace A of a normal space Ywith respect to Y, i.e., the smallest integer n such that  $\dim Z \leq n$  for every closed subspace Z of the space Y contained in A.

Thus, one proves that
(F) For every closed mapping  $f: X \to Y$  of a normal space X onto a normal space Y with dim  $Y < \infty$  there exist compact metrizable spaces  $X_0, Y_0$  and a closed mapping  $f_0$  of  $X_0$  onto  $Y_0$  such that dim  $X_0 = \dim X$ , dim  $Y_0$ = dim Y and dim  $C_k(f_0) \leq \operatorname{rd}_Y C_k(f)$  for k = 1, 2, ... If, moreover, the space Y is weakly paracompact, then dim  $D_k(f_0) \leq \operatorname{rd}_Y D_k(f)$  for k = 1, 2, ...

Let us note that the assumption of weak paracompactness in the second part of (F) cannot be omitted. Indeed, one can define a closed mapping fof a normal space X onto a normal space Y such that dim X = 1, dim Y = 0and dim  $f^{-1}(y) = 0$  for every  $y \in Y$ .

# Historical and bibliographic notes

Theorem 4.3.3 was given by Morita in [1955]; Theorem 4.3.6 was proved independently by Morita in [1956] and by Nagami in [1957]. Theorems 4.3.9 and 4.3.12 were established by Vaĭnšteĭn in [1952] for separable X and Y. Extensions to arbitrary metrizable spaces were given by Skljarenko in [1962] and [1963]. Lemma 4.3.11 was proved by Freudenthal in [1932]. As shown by Lelek in [1971], from Vaĭnšteĭn's theorems many further results about the behaviour of dimension under mappings can be deduced. Lelek's paper gives a comprehensive survey of the topic considered in this section and provides a good bibliography. Theorem (F) stated at the end of the section as well as the example cited after this theorem were given by Filippov in [1972a]. As we have already noted, Filippov's theorem leads to extensions of the results obtained in this section to larger classes of spaces. Some of these extensions were obtained earlier in the abovementioned papers by Morita, Nagami and Skljarenko.

# Problems

**4.3.A** (Suzuki [1959]). Prove that if  $f: X \to Y$  is a closed mapping of a metrizable space X onto a metrizable space Y and  $|f^{-1}(y)| = k < \infty$  for every  $y \in Y$ , then  $\operatorname{Ind} X = \operatorname{Ind} Y$ .

Hint. See the hint to Problem 1.12.A.

**4.3.B.** (a) Following the pattern in Section 1.9, prove that for every continuous mapping  $f: A \to S^n$  defined on a closed subspace A of metri-

zable space X such that  $\operatorname{Ind}(X \setminus A) \leq n \geq 0$  there exists a continuous extension  $F: X \to S^n$  of f over X (cf. Problem 2.2.B).

(b) Deduce from (a), Theorem 4.1.3 and Remarks 1.7.10 and 1.9.4 that a metrizable space X satisfies the inequality  $\operatorname{Ind} X \leq n \geq 0$  if and only if for every closed subspace A of the space X and each continuous mapping  $f: A \to S^n$  there exists a continuous extension  $F: X \to S^n$  of f over X.

**4.3.C** (Morita [1955]). Prove that a metrizable space X satisfies the inequality dim  $X \le n$  if and only if there exists a closed mapping  $f: Z \to X$  of a subspace Z of the Baire space  $B(\mathfrak{m})$ , where  $\mathfrak{m} = w(X)$ , onto the space X with fibres of cardinality at most n+1.

Hint. For a metrizable space X with dim  $X \leq n$  consider a sequence  $\mathscr{F}_1, \mathscr{F}_2, \ldots$  of locally finite closed covers satisfying the conditions in Problem 4.2.B(b) and such that  $|\mathscr{F}_i| \leq m$  for  $i = 1, 2, \ldots$  Define a mapping  $\pi_i^{i+1}$  of  $\mathscr{F}_{i+1}$  to  $\mathscr{F}_i$  such that  $F \subset \pi_i^{i+1}(F)$  for every  $F \in \mathscr{F}_{i+1}$ , let  $\pi_i^j = \pi_j^{j+1}\pi_j^{j+2}\dots\pi_{i-1}^i$  for j < i and  $\pi_i^i = \mathrm{id}_{\mathscr{F}_i}$ . Consider the inverse sequence  $S = \{\mathscr{F}_i, \pi_j^i\}$ , where  $\mathscr{F}_i$  has the discrete topology. Note that  $S = \lim S$  is a subspace of  $B(\mathfrak{m})$  and consider the subspace Z of S consisting of all sequences  $\{F_i\} \in S$  such that  $\bigcap_{i=1}^{\infty} F_i \neq \emptyset$ ; show that the last intersection contains exactly one point of the space X and assign this point to the sequence  $\{F_i\}$ . Prove that the mapping  $f: Z \to X$  obtained in this way has all the required properties. When proving that f(Z) = X apply the fact that the limit of an inverse sequence consisting of finite discrete spaces is non-empty. When proving that f is closed apply the equality  $f^{-1}(x) = Z \cap \prod_{i=1}^{\infty} \{F \in \mathscr{F}_i: x \in F\}$  and show that for every open set  $U \subset Z$  which

contains  $f^{-1}(x)$  there exists a neighbourhood  $V \subset X$  of the point x such that  $f^{-1}(V) \subset U$ .

**4.3.D** (Nagami [1960]). Prove the Katětov–Morita theorem by applying only Theorems 1.6.11, 4.1.13 and 3.1.28 and the facts established in Problems 4.3.A and 4.3.C.

*Hint.* Applying 4.3.A, deduce from 4.3.C that every metrizable space X which satisfies the inequality  $\dim X \leq n$  can be represented as the union of n+1 subspaces  $Z_1, Z_2, \ldots, Z_{n+1}$  such that  $\dim Z_i \leq 0$  for  $i = 1, 2, \ldots$   $\ldots$  n+1.

**4.3.E.** (a) (Hodel [1963]) Show that if  $f: X \to Y$  is an open mapping of a metrizable space X onto a metrizable space Y such that for every  $y \in Y$  the fibre  $f^{-1}(y)$  is a discrete subspace of X, then  $\operatorname{Ind} X \leq \operatorname{Ind} Y$ .

Hint. Apply Lemma 1.12.5.

(b) (Nagami [1960]) Show that if  $f: X \to Y$  is an open mapping of a metrizable space X onto a metrizable space Y such that  $|f^{-1}(y)| < \aleph_0$  for every  $y \in Y$ , then  $\operatorname{Ind} X = \operatorname{Ind} Y$ .

(c) (Hodel [1963]) Show that if  $f: X \to Y$  is an open-and-closed mapping of a metrizable space X onto a metrizable space Y such that for every  $y \in Y$  the fibre  $f^{-1}(y)$  has an isolated point, then  $\operatorname{Ind} Y \leq \operatorname{Ind} X$ .

*Remark.* As shown by R. Pol in [1979], in the realm of metrizable spaces there exist open mappings with discrete fibres which raise dimensions (cf. Theorem 1.12.7).

(d) (Hodel [1963]) Show that if  $f: X \to Y$  is an open mapping of a locally compact metrizable space X onto a metrizable space Y such that  $|f^{-1}(y)| \leq \aleph_0$  for every  $y \in Y$ , then  $\operatorname{Ind} X = \operatorname{Ind} Y$ .

(e) (Arhangel'skiĭ [1966]) Show that if  $f: X \to Y$  is an open-and-closed mapping of a metrizable space X onto a metrizable space Y such that  $|f^{-1}(y)| \leq \aleph_0$  for every  $y \in Y$ , then  $\operatorname{Ind} X = \operatorname{Ind} Y$ .

# **BIBLIOGRAPHY**

- J. M. AARTS
- [1971] A characterization of strong inductive dimension, Fund. Math. 70 (1971) 147-155.
- J. M. AARTS and T. NISHIURA
- [1972] The Eilenberg-Borsuk duality theorem, Indag. Math. 34 (1972) 68-72.
- Р. ALEXANDROFF (П. С. Александров)
- [1925] Über stetige Abbildungen kompakter Räume, Proc. Akad. Amsterdam 28 (1925) 997–999.
- [1926] Sur la dimension des ensembles fermés, C. R. Acad. Paris 183 (1926) 640-642.
- [1927] Une définition des nombres de Betti pour un ensemble fermé quelconque, C. R. Acad. Paris 184 (1927) 317-319.
- [1927a] Über stetige Abbildungen kompakter Räume, Math. Ann. 96 (1927) 555-571.
- [1928] Über den allgemeinen Dimensionsbegriff und seine Beziehungen zur elementaren geometrischen Anschauung, Math. Ann. 98 (1928) 617-636.
- [1928a] Zum allgemeinen Dimensionsproblem, Nachr. Ges. Wiss. Göttingen (1928) 25-44.
- [1930] Analyse géometrique de la dimension des ensembles fermés, C. R. Acad. Paris 191 (1930) 475-477.
- [1932] Dimensionstheorie. Ein Beitrag zur Geometrie der abgeschlossenen Mengen, Math. Ann. 106 (1932) 161-238.
- [1936] Über abzählbar-fache offene Abbildungen, C. R. (Doklady) Acad. Sci. URSS
   13 (1936) 295-299.
- [1940] Über die Dimension der bikompakten Räume, C. R. (Doklady) Acad. Sci. URSS 26 (1940) 619-622.
- [1941] Теорема сложения в теории размерности бикомпактных пространств, Сообщ. Акад. Наук Груз. ССР 2 (1941) 1-6. (The sum theorem in the dimension theory of bicompact spaces)
- [1947] On the dimension of normal spaces, Proc. Roy. Soc. London 189 (1947) 11-39.
- [1951] П. С. Урысон и его место в математической науке (P. S. Urysohn and his place in Mathematics), Introduction to: П. С. Урысон, Труды по топологии и другим областям математики, Том I (Москва 1951). (P. S. URY-SOHN, Works in Topology and other branches of Mathematics, vol. I)
- P. ALEXANDROFF and H. HOPF
- [1935] Topologie I (Berlin 1935).
- P. ALEXANDROFF and B. A. PASYNKOV (II. C. AJIEKCAHAPOB, E. A. IIACLIHKOB)
- [1973] Введение в теорию размерности (Москва 1973). (Introduction to Dimension Theory)

P. ALEXANDROFF and P. URYSOHN

[1928] Über nulldimensionale Punktmengen, Math. Ann. 98 (1928) 89-106.

- R. D. ANDERSON
- [1958] A characterization of the universal curve and a proof of its homogeneity, Ann. of Math. 67 (1958) 313-324.
- R.D. ANDERSON and J.E. KEISLER

[1967] An example in dimension theory, Proc. Amer. Math. Soc. 18 (1967) 709-713.

- A. ARHANGEL'SKII (А. Архангельский)
- [1963] O рангах систем множеств и размерности пространств, Fund. Math. 52 (1963) 257-275. (Ranks of families of sets and dimension of spaces)
- [1966] Отображения открытые и близкие к открытым. Связи между пространствами, Труды Моск. Мат. Общ. 15 (1966) 181–223. (Open and near open mappings. Connections between spaces; English translation: Trans. Mosc. Math. Soc. 15 (1966) 204–250)
- [1967] О факторизации отображений по весу и размерности, ДАН СССР 174 (1967) 1243-1246. (Factorization of mappings according to weight and dimension; English translation: Soviet Math. Dokl. 8 (1967) 731-734)
- R. BENNETT
- [1972] Countable dense homogeneous spaces, Fund. Math. 74 (1972) 189-194.
- C. BESSAGA and A. PEŁCZYŃSKI
- [1970] The estimated extension theorem, homogeneous collections and skeletons and their applications to the topological classification of linear metric spaces and convex sets, *Fund. Math.* 69 (1970) 153-190.

# R. H. BING

- K. BORSUK
  - [1936] Ensembles dont les dimensions modulaires de Alexandroff coincident avec la dimension de Menger-Urysohn, Fund. Math. 27 (1936) 77-93.
  - [1937] Un théorème sur les prolongements des transformations, Fund. Math. 29 (1937) 161-166.
- [1951] Cartesian products of Cantor-manifolds, Fund. Math. 38 (1951) 55-72.
- [1975] Remark on the Cartesian product of two 1-dimensional spaces, Bull. Acad. Pol. Sci. Sér. Math. 23 (1975) 971-973.

H. G. BOTHE

- [1963] Eine Einbettung m-dimensionaler Mengen in einem (m+1)-dimensionalen absoluten Retrakt, Fund. Math. 52 (1963) 209-224.
- L. E. J. BROUWER
  - [1910] On the structure of perfect sets of points, Proc. Akad. Amsterdam 12 (1910) 785-794.
  - [1911] Beweis der Invarianz der Dimensionenzahl, Math. Ann. 70 (1911) 161-165.

<sup>[1951]</sup> Snake-like continua, Duke Math. Journ. 18 (1951) 553-663.

- [1913] Über den natürlichen Dimensionsbegriff, Journ. für die reine and angew. Math.
   142 (1913) 146–152.
- [1913a] Some remarks on the coherence type  $\eta$ , Proc. Akad. Amsterdam 15 (1913) 1256-1263.
- [1923] Über den natürlichen Dimensionsbegriff, Proc. Akad. Amsterdam 26 (1923) 795-800.
- [1924] Bemerkungen zum natürlichen Dimensionsbegriff, Proc. Akad. Amsterdam 27 (1924) 635-638.
- [1924a] Berichtigung zur Abhandlung "Über den natürlichen Dimensionsbegriff", dieses Journal Bd. 142 (1913), S. 146-152, Journ. für die reine und angew. Math.
   153 (1924) 253.
- [1924b] Zum natürlichen Dimensionsbegriff, Math. Zeitschr. 21 (1924) 312-314.
- [1928] Zur Geschichtschreibung der Dimensionstheorie, Proc. Akad. Amsterdam 31 (1928) 953-957.
- [1976] Collected works, vol. II (Amsterdam 1976).

#### J. W. CANNON

- [1973] A positional characterization of the (n-1)-dimensional Sierpiński curve in  $S^n (n \neq 4)$ , Fund. Math. 79 (1973) 107-112.
- Е. Čесн
- [1931] Sur la théorie de la dimension, C. R. Acad. Paris 193 (1931) 976-977.
- [1932] Sur la dimension des espaces parfaitement normaux, Bull. Intern. Acad. Tchèque Sci. 33 (1932) 38-55.
- [1933] Přispěvek k theorii dimense, Časopis Pěst. Mat. Fys. 62 (1933) 277-291. (Contribution to dimension theory; French translation: E. ČECH, Topological papers (Prague 1968) 129-142)
- G. CHOGOSHVILI
  - [1938] On a theorem in the theory of dimensionality, Comp. Math. 5 (1938) 292-298.

# J. VAN DALEN

[1972] Finite products of locally compact ordered spaces, Thesis, Vrije Universiteit te Amsterdam, 1972.

E. K. VAN DOUWEN

[1973] The small inductive dimension can be raised by the adjunction of a single point, Indag. Math. 35 (1973) 434-442.

#### C. H. DOWKER

- [1947] Mapping theorems for non-compact spaces, Amer. Journ. of Math. 69 (1947) 200-242.
- [1953] Inductive dimension of completely normal spaces, Quart. Journ. of Math. Oxford 4 (1953) 267-281.
- [1955] Local dimension of normal spaces, Quart. Journ. of Math. Oxford 6 (1955) 101-120.
- C. H. Dowker and W. HUREWICZ
- [1956] Dimension of metric spaces, Fund. Math. 43 (1956) 83-88.

### V. I. Egorov (В. И. Егоров)

[1959] О метрической размерности точечных множеств, Mam. Có. 48 (1959) 227-250. (On the metric dimension of points of sets)

## S. EILENBERG

[1936] Un théorème de dualité, Fund. Math. 26 (1936) 280-282.

- S. EILENBERG and E. OTTO
- [1938] Quelques propriétés caractéristiques de la théorie de la dimension, Fund. Math. 31 (1938) 149-153.

#### S. EILENBERG and N. STEENROD

[1952] Foundations of algebraic topology (Princeton 1952).

#### **R.** ENGELKING

- [1960] Sur la compactification des espaces métriques, Fund. Math. 48 (1960) 321-324.
- [1973] Some new proofs in dimension theory, Proceedings of the Symposium on General Topology, Rome 1973, Ist. di Alta Mat. Symposia Math. 16 (1975) 83-91.
- [1977] General Topology (Warszawa 1977).

# P. ERDÖS

- [1940] The dimension of the rational points in Hilbert space, Ann. of Math. 41 (1940) 734-736.
- V. V. Fedorčuk (B. B.  $\Phi$ едорчук)
- [1973] Бикомпакты без промежуточных размерностей, ДАН СССР 213 (1973) 795-797. (Bicompacta without intermediate dimensions; English translation: Soviet Math. Dokl. 14 (1973) 1808-1811)
- V. V. FILIPPOV (В. В. Филиппов)
- [1969] Бикомпакт с несовпадающими индуктивными размерностями, ДАН СССР
   184 (1969) 1050–1053. (A bicompactum with noncoinciding inductive dimensions; English translation: Soviet Math. Dokl. 10 (1969) 208–211)
- [1970] О бикомпактах с несовпадающими индуктивными размерностями, ДАН СССР 192 (1970) 289–292. (On bicompacta with noncoinciding inductive dimensions; English translation: Soviet Math. Dokl. 11 (1970) 635–638)
- [1970a] О бикомпактах с несовпадающими размерностями ind и dim, ДАН СССР
   192 (1970) 516-519. (On bicompacta with noncoinciding dimensions ind and dim; English translation: Soviet Math. Dokl. 11 (1970) 687-691)
- [1970b] Решение одной задачы П. С. Александрова (Бикомпакт с несовпадающими индуктивными размерностями), *Mam. C6.* 83 (1970) 42–60. (Solution of a problem of P. S. Alexandroff (A bicompactum with noncoinciding inductive dimensions); English translation: *Math. USSR-Sb.* 12 (1970) 41–57)
- [1972] Об индуктивной размерности произведения бикомпактов, ДАН СССР 202 (1972) 1016–1019. (On the inductive dimension of the product of bicompacta; English translation: Soviet Math. Dokl. 13 (1972) 250–254)
- [1972а] О размерности замкнутых отображений, ДАН СССР 205 (1972) 40-43. (On the dimension of closed mappings; English translation: Soviet Math. Dokl. 13 (1972) 895-900)

- [1973] О размерности нормальных пространств, ДАН СССР 209 (1973) 805-807.
   (On the dimension of normal spaces; English translation: Soviet Math. Dokl. 14 (1973) 547-550)
- [1979] О размерности произведений топологических пространств, Fund. Math., to appear. (On the dimension of products of topological spaces)
- B. FITZPATRICK and R. M. FORD
- [1967] On the equivalence of small and large inductive dimension in certain metric spaces, *Duke Math. Journ.* 34 (1967) 33-37.
- A. FLORES
- [1935] Über *n*-dimensionale Komplexe die im  $R_{2n+1}$  absolut selbstverschlungen sind, Ergebnisse eines math. Koll. 6 (1935) 4-6.
- A. B. FORGE
- [1961] Dimension preserving compactifications allowing extensions of continuous functions, Duke Math. Journ. 28 (1961) 625-627.
- M. K. FORT JR.
- [1962] Homogeneity of infinite products of manifolds with boundary, *Pacific Journ.* of Math. 12 (1962) 879-884.
- F. FRANKL
- [1930] Charakterisierung der (n-1)-dimensionalen abgeschlossenen Mengen des R<sup>n</sup>, Math. Ann. 103 (1930) 784-787.
- F. FRANKL and L. PONTRJAGIN
- [1930] Ein Knotensatz mit Anwendung auf die Dimensionstheorie, Math. Ann. 102 (1930) 785-789.
- M. FRÉCHET
- [1910] Les dimensions d'un ensemble abstrait, Math. Ann. 68 (1910) 145-168.
- J. A. FRENCH
- [1970] A characterization of covering dimension for collectionwise normal spaces, Proc. Amer. Math. Soc. 25 (1970) 646–649.
- H. FREUDENTHAL
- [1932] Über dimensionserhöhende stetige Abbildungen, Sitzungsberichte Preuss. Akad. Wiss. Phys.-Math. Klasse (1932) 34-38.
- [1937] Entwicklungen von Räumen und ihren Gruppen, Comp. Math. 4 (1937) 145-234.
- J. de Groot
- [1956] Non-archimedean metrics in topology, Proc. Amer. Math. Soc. 7 (1956) 948-953.
- [1957] On a metric that characterizes dimension, Canad. Journ. of Math. 9 (1957) 511-514.
- J. DE GROOT and R. H. MCDOWELL
- [1960] Extension of mappings on metric spaces, Fund. Math. 48 (1960) 251-263.
- J. DE GROOT and J. NAGATA
- [1969] Lectures on dimension theory for infinite dimensional spaces, University of Florida, Gainesville 1969.
- R. W. HANSELL
- [1974] On characterizing non-separable analytic and extended Borel sets as types of continuous images, Proc. London Math. Soc. 28 (1974) 683-699.

- F. HAUSDORFF
- [1914] Grundzüge der Mengenlehre (Leipzig 1914).
- [1927] Mengenlehre (Berlin 1927).
- [1934] Über innere Abbildungen, Fund. Math. 23 (1934) 279-291.
- E. HEMMINGSEN
- [1946] Some theorems in dimension theory for normal Hausdorff spaces, *Duke Math.* Journ. 13 (1946) 495-504.
- D. W. HENDERSON
- [1965] Finite dimensional subsets of infinite dimensional spaces, Topology Seminar Wisconsin 1965, Ann. of Math. Studies 60 (Princeton 1966) 141-146.
- [1967] An infinite-dimensional compactum with no positive-dimensional compact subsets—a simpler construction, Amer. Journ. of Math. 89 (1967) 105-121.
- A. HILGERS
- [1937] Bemerkung zur Dimensionstheorie, Fund. Math. 28 (1937) 303-304.
- R. E. HODEL
- [1963] Open functions and dimension, Duke Math. Journ. 30 (1963) 461-468.
- W. HUREWICZ
- [1926] Über stetige Bilder von Punktmengen, Proc. Akad. Amsterdam 29 (1926) 1014-1017.
- [1927] Normalbereiche und Dimensionstheorie, Math. Ann. 96 (1927) 736-764.
- [1927a] Über stetige Bilder von Punktmengen (Zweite Mitteilung), Proc. Akad. Amsterdam 30 (1927) 159-165.
- [1927b] Über das Verhältnis separabler Räume zu kompakten Räumen, Proc. Akad. Amsterdam 30 (1927) 425-430.
- [1928] Über unendlich-dimensionale Punktmengen, Proc. Akad. Amsterdam 31 (1928) 916-922.
- [1930] Einbettung separabler Räume in gleich dimensional kompakte Räume, Monatsh. für Math. und Phys. 37 (1930) 199–208.
- [1931] Dimensionstheorie und Cartesische Räume, Proc. Akad. Amsterdam 34 (1931) 399-400.
- [1932] Une remarque sur l'hypothèse du continu, Fund. Math. 19 (1932) 8-9.
- [1933] Über Abbildungen von endlich dimensionalen Räumen auf Teilmengen Cartesischer Räume, Sitzungsberichte Preuss. Akad. Wiss. Phys.-Math. Klasse (1933) 754-768.
- [1935] Sur la dimension des produits Cartésiens, Ann. of Math. 36 (1935) 194-197.
- [1935a] Über Abbildungen topologischer Räume auf die n-dimensionale Sphäre, Fund. Math. 24 (1935) 144-150.
- [1937] Ein einfacher Beweis des Hauptsatzes über Cantorsche Mannigfaltigkeiten, Prace Mat. Fiz. 44 (1937) 289-292.
- W. HUREWICZ and K. MENGER

- W. HUREWICZ and H. WALLMAN
- [1941] Dimension theory (Princeton 1941).

<sup>[1928]</sup> Dimension und Zusammenhangsstuffe, Math. Ann. 100 (1928) 618-633.

#### J. R. ISBELL

- [1959] Embeddings of inverse limits, Ann. of Math. 70 (1959) 73-84.
- [1964] Uniform spaces (Providence 1964).

# Z. JANISZEWSKI

- [1912] Sur les continus irréductibles entre deux points, Journ. École Polyt. 16 (1912) 79-170.
- M. KATĚTOV (M. KATETOB)
- [1950] A theorem on the Lebesgue dimension, *Časopis Pést. Mat. Fys.* **75** (1950) 79–87.
- [1951] О размерности метрических пространств, ДАН СССР 79 (1951) 189–191. (On the dimension of metric spaces)
- [1952] О размерности несепарабельных пространств. I, Czech. Math. Journ. 2 (1952) 333-368. (On the dimension of non-separable spaces. I)
- [1958] О соотношении между метрической и топологической размерностью, *Czech. Math. Journ.* 8 (1958) 163-166. (On the relation between the metric and topological dimensions)

# Υ. ΚΑΤUTA

- [1966] A note on the inductive dimension of product spaces, Proc. Japan Acad. 42 (1966) 1011-1015.
- [1967] A theorem on paracompactness of product spaces, Proc. Japan Acad. 43 (1967) 615-618.

### L. Keldyš (Л. Келдыш)

[1954] Пример одномерного континуума, нульмерно и открыто отображающегося на квадрат, ДАН СССР 97 (1954) 201-204. (An example of a one-dimensional continuum which can be mapped onto a square by a zero-dimensional open mapping)

#### N. KIMURA

- [1963] On the inductive dimension of product spaces, Proc. Japan Acad. 39 (1963) 641-646.
- [1967] On a sum theorem in dimension theory, Proc. Japan Acad. 43 (1967) 98-102.

### B. KNASTER and K. KURATOWSKI

[1921] Sur les ensembles connexes, Fund. Math. 2 (1921) 206-255.

# Y. KODAMA

- [1969] On subset theorems and the dimension of products, Amer. Journ. of Math. 91 (1969) 486-497.
- [1970] Cohomological dimension theory, Appendix to: K. NAGAMI, Dimension theory (New York 1970).

#### A. KOLMOGOROFF

[1937] Über offene Abbildungen, Ann. of Math. 38 (1937) 36-38.

### K. KURATOWSKI

[1932] Sur l'application des espaces fonctionnels à la théorie de la dimension, Fund. Math. 18 (1932) 285-292.

- [1932a] Une application des images de fonctions à la construction de certains ensembles singuliers, *Mathematica* 6 (1932) 120-123.
- [1933] Topologie I (Warszawa 1933).
- [1933a] Sur un théorème fondamental concernant le nerf d'un système d'ensembles, Fund. Math. 20 (1933) 191-196.
- [1937] Coupures irréductibles et multiplicités Cantoriennes, Ann. Soc. Pol. Math. 16 (1937) 220.
- [1937a] Sur les théorèmes du "plongement" dans la théorie de la dimension, Fund. Math. 28 (1937) 336-342.
- [1968] Topology, vol. II (New York 1968).
- K. KURATOWSKI and K. MENGER
- [1930] Remarques sur la théorie axiomatique de la dimension, Monatsh. für Math. und Phys. 37 (1930) 169-174.
- K. KURATOWSKI and E. OTTO

- V. I. KUZ'MINOV (В. И. Кузьминов)
- [1968] Гомологическая теория размерности, *Успехи Мат. Наук* 23 (1968), вып. 5, 3-49. (Homological dimension theory; English translation: *Russian Math. Surveys* 23 (1968), No. 5, 1-45)
- H. LEBESGUE
- [1911] Sur la non applicabilité de deux domaines appartenant respectivement à des espaces, de n et n+p dimensions, Math. Ann. 70 (1911) 166-168.
- [1921] Sur les correspondences entre les points de deux espaces, Fund. Math. 2 (1921) 256-285.
- S. LEFSCHETZ
- [1931] On compact spaces, Ann. of Math. 32 (1931) 521-538.
- А. LELEK (А. ЛЕЛЕК)
- [1965] О размерности наростов при компактных расширениях, ДАН СССР
   160 (1965) 534-537. (On the dimension of remainders in compactifications; English translation: Soviet Math. Dokl. 6 (1965) 136-140)
- [1971] Dimension inequalities for unions and mappings of separable metric spaces, Coll. Math. 23 (1971) 69-91.
- В. Т. Levšenko (Б. Т. Левшенко)
- [1969] Размерность метрических пространств и ретракция, Fund. Math. 66 (1969) 1-5. (The dimension of metric spaces and the retraction)
- B. T. LEVŠENKO and JU. M. SMIRNOV (B. T. ЛЕВШЕНКО, Ю. М. СМИРНОВ)
- [1966] Об одном свойстве метрических нульмерных пространств, Proceedings of the Second Prague Topological Symposium, 1966 (Prague 1967) 241-242. (On a property of zero-dimensional metric spaces)
- I. K. LIFANOV (И. К. ЛИФАНОВ)
- [1968] О размерности произведения бикомпактов, ДАН СССР 182 (1968) 1274-1277. (On the dimension of the product of bicompacta; English translation: Soviet Math. Dokl. 9 (1968) 1280-1283)

<sup>[1939]</sup> Sur les espaces à connexité n-dimensionelle, Fund. Math. 32 (1939) 259-264.

- I. K. LIFANOV and V. V. FILIPPOV (И. К. Лифанов, В. В. Филиппов)
  - [1970] Два примера к теории размерности бикомпактов, ДАН СССР 192 (1970) 26-29. (Two examples to the theory of the dimension of bicompacta; English translation: Soviet Math. Dokl. 11 (1970) 573-576)
- I. K. LIFANOV and B. A. PASYNKOV (И. К. ЛИФАНОВ, Б. А. ПАСЫНКОВ)
- [1970] О двух классах пространств и двух видах размерности, Вестник Моск. Унив. Сер. Mam. (1970), No. 3, 33–37. (Two classes of spaces and two forms of dimension; English translation: Mosc. Univ. Bull. 23 (1970), No. 3, 26–29)
- S. L. LIPSCOMB
  - [1975] On imbedding finite-dimensional metric spaces, Trans. Amer. Math. Soc. 211 (1975) 143-160.
- L. A. LJUKSEMBURG (Л. А. ЛЮКСЕМБУРГ)
- [1973] О трансфинитных индуктивных размерностях, ДАН СССР 209 (1973) 295-298. (On transfinite inductive dimensions; English translation: Soviet Math. Dokl. 14 (1973) 388-393)
- [1973а] О компактах с несовпадающими трансфинитными размерностями, ДАН СССР 212 (1973) 1297–1300. (Compacta with noncoinciding transfinite dimensions; English translation: Soviet Math. Dokl. 14 (1973) 1593–1597)
- О. V. LOKUCIEVSKIĬ (О. В. Локуциевский)
- [1949] О размерности бикомпактов, ДАН СССР 67 (1949) 217-219. (On the dimension of bicompacta)
- [1973] Аксиоматическое определение размерности бикомпактов, ДАН СССР 212 (1973) 813-815. (An axiomatic definition of the dimension of bicompacta; English translation: Soviet Math. Dokl. 14 (1973) 1477-1480)
- А. L. LUNC (А. Л. Лунц)
- [1949] Бикомпакт индуктивная размерность которого больше, чем размерность, определенная при помощи покрытий, ДАН СССР 66 (1949) 801-803. (A bicompactum whose inductive dimension is larger than the covering dimension)
- E. MARCZEWSKI (E. SZPILRAJN)
- [1937] La dimension et la mesure, Fund. Math. 28 (1937) 81-89.
- S. MARDEŠIĆ
- [1960] On covering dimension and inverse limits of compact spaces, Illinois Journ. of Math. 4 (1960) 278-291.
- S. MARDEŠIĆ and J. SEGAL

[1963] e-mappings onto polyhedra, Trans. Amer. Math. Soc. 109 (1963) 146-164.

- S. MAZURKIEWICZ
  - [1915] O punktach wielokrotnych krzywych wypełniających obszar plaski, Prace Mat. Fiz. 26 (1915) 113-120. (On the multiple points of curves which fill out a plane region; French translation: S. MAZURKIEWICZ, Travaux de topologie (Warszawa 1969) 48-53)
  - [1917] Teoria zbiorów  $G_{\delta}$ , Wektor 6 (1917–18) 129–185. (Theory of  $G_{\delta}$ -sets)

- [1920] Sur un ensemble  $G_{\delta}$  punctiforme qui n'est homéomorphe à aucun ensemble linéaire, *Fund. Math.* 1 (1920) 61-81.
- [1927] Sur les problèmes  $\varkappa$  et  $\lambda$  de Urysohn, Fund. Math. 10 (1927) 311-319.
- [1929] Sur les ensembles de dimension faible, Fund. Math. 13 (1929) 210-217.
- [1933] Ein Satz über dimensionelle Komponenten, Fund. Math. 20 (1933) 98-99.
- M. C. McCord
- [1966] Universal P-like compacta, Michigan Math. Journ. 13 (1966) 71-85.
- K. MENGER
- [1923] Über die Dimensionalität von Punktmengen I, Monatsh. für Math. und Phys. 33 (1923) 148-160.
- [1924] Über die Dimension von Punktmengen II, Monatsh. für Math. und Phys. 34 (1926) 137-161. (Reprints are dated 1924; the whole volume appeared only in 1926)
- [1926] Allgemeine Räume und Cartesische Räume, Zweite Mitteilung: "Über umfassendste n-dimensionale Mengen", Proc. Akad. Amsterdam 29 (1926) 1125–1128.
- [1928] Dimensionstheorie (Leipzig 1928).
- [1929] Über die Dimension von Punktmengen III. Zur Bergründung einer axiomatischen Theorie der Dimension, Monatsh. für Math. und Phys. 36 (1929) 193-218.
- [1929a] Zur Dimensions- und Kurventheorie, Unveröfentlichte Aufsätze aus den Jahren 1921–1923, Monatsh. für Math. und Phys. 36 (1929) 411–432.
- [1930] Antwort auf eine Note von Brouwer, Monatsh. für Math. und Phys. 37 (1930) 175-182.
- [1933] Eine Zuschrift von K. Menger an H. Hahn, betreffend Mengers Antwort auf eine Note von Brouwer in Bd. 37 der Monatshefte, Monatsh. für Math. und Phys. 40 (1933) 233.
- K. MORITA
- [1940] On uniform spaces and the dimension of compact spaces, Proc. Phys. Math. Soc. Japan 22 (1940) 969-977.
- [1950] On the dimension of normal spaces I, Jap. Journ. of Math. 20 (1950) 5-36.
- [1950a] On the dimension of normal spaces II, Journ. of Math. Soc. Japan 2 (1950) 16-33.
- [1953] On the dimension of product spaces, Amer. Journ. of Math. 75 (1953) 205-223.
- [1954] Normal families and dimension theory for metric spaces, Math. Ann. 128 (1954) 350-362.
- [1955] A condition for the metrizability of topological spaces and for n-dimensionality, Sci. Rep. Tokyo Kyoiku Daigaku Sec. A 5 (1955) 33-36.
- [1956] On closed mappings and dimension, Proc. Japan Acad. 32 (1956) 161-165.
- [1975] Čech cohomology and covering dimension for topological spaces, Fund. Math. 87 (1975) 31-52.
- K. NAGAMI
  - [1955] The dimension of paracompact Hausdorff spaces, Nagoya Math. Journ. 8 (1955) 69-70.
  - [1957] Some theorems in dimension theory for non-separable spaces, Journ. Math. Soc. Japan 9 (1957) 80-92.
  - [1959] Finite-to-one closed mappings and dimension. II, Proc. Japan Acad. 35 (1959) 437-439.

- [1960] Mappings of finite order and dimension theory, Jap. Journ. of Math. 30 (1960) 25-54.
- [1960a] On the dimension of product spaces, Proc. Japan Acad. 36 (1960) 560-564.
- [1965] Monotone sequences of 0-dimensional subsets of metric spaces, Proc. Japan Acad. 41 (1965) 771-772.
- [1969] A note on the large inductive dimension of totally normal spaces, Journ. Math. Soc. Japan 21 (1969) 282-290.
- [1970] Dimension theory (New York 1970).
- K. NAGAMI and J. H. ROBERTS
- [1965] A note on countable-dimensional metric spaces, Proc. Japan Acad. 41 (1965) 155-158.
- [1967] A study of metric-dependent dimension functions, Trans. Amer. Math. Soc. 129 (1967) 414-435.
- J. NAGATA
- [1956] On a relation between dimension and metrization, Proc. Japan Acad. 32 (1956) 237-240.
- [1956a] Note on dimension theory, Proc. Japan Acad. 32 (1956) 568-569.
- [1957] On imbedding theorem for non-separable metric spaces, Journ. Inst. Polyt. Osaka City Univ. 8 (1957) 9-14.
- [1958] Note on dimension theory for metric spaces, Fund. Math. 45 (1958) 143-181.
- [1960] On countable sum of zero-dimensional spaces, Fund. Math. 48 (1960) 1-14.
- [1960a] On a universal n-dimensional set for metric spaces, Journ. für die reine und angew. Math. 204 (1960) 132-138.
- [1961] On dimension and metrization, Proceedings of the Symposium on General Topology Prague 1961 (Prague 1962) 282-285.
- [1963] Two theorems for the *n*-dimensionality of metric spaces, Comp. Math. 15 (1963) 227-237.
- [1964] On a special metric and dimension, Fund. Math. 55 (1964) 181-194.
- [1965] Modern dimension theory (Groningen 1965).
- [1967] Product theorems in dimension theory. I, Bull. Acad. Pol. Sci. Sér. Math. 15 (1967) 439-448.
- T. NISHIURA
- [1966] Inductive invariants and dimension theory, Fund. Math. 59 (1966) 243-262.
- [1977] A subset theorem in dimension theory, Fund. Math. 95 (1977) 105-109.

### G. Nöbeling

- [1931] Über eine *n*-dimensionale Universalmenge im  $\mathbb{R}^{2n+1}$ , Math. Ann. 104 (1931) 71-80.
- [1932] Die Projektionen einer kompakten, n-dimensionalen Menge im R<sub>k</sub>, Ergebnisse eines math. Koll. 4 (1932) 24-25.
- P. A. OSTRAND
- [1965] Dimension of metric spaces and Hilbert's Problem 13, Bull. Amer. Math. Soc. 71 (1965) 619-622.
- [1965a] A conjecture of J. Nagata on dimension and metrization, Bull. Amer. Math. Soc. 71 (1965) 623-625.

- [1971] Covering dimension in general spaces, General Topology and its Appl. 1 (1971) 209-221.
- В. А. РАЗУМКОУ (Б. А. ПАСЫНКОВ)
  - [1958] О полиэдральных спектрах и размерности бикомпактов, в частности бикомпактных топологических групп, *ДАН СССР* 121 (1958) 45-48. (On polyhedral spectra and the dimension of bicompacta, in particular bicompact topological groups)
  - [1962] О спектрах и размерности топологических пространств, Mam. Có. 57 (1962) 449-476. (On spectra and the dimension of topological spaces)
  - [1963] Οδ ω-οτοδραжениях и обратных спектрах, *ДАН СССР* 150 (1963) 488-491.
     (On ω-mappings and inverse spectra; English translation: Soviet Math. Dokl.
     4 (1963) 706-709)
  - [1964] Об универсальных бикомпактах данного веса и данной размерности, ДАН СССР 154 (1964) 1042–1043. (On universal bicompacta of given weight and dimension; English translation: Soviet Math. Dokl. 5 (1964) 245–246)
  - [1965] О спектральной разложимости топологических пространств, Mam. C6. 66 (1965) 35-79. (On the spectral decomposition of topological spaces; English translation: Amer. Math. Soc. Transl. Ser. 2, 73 (1968) 87-134)
  - [1966] Об универсальных компактах, *Успехи Мат. Наук* 21 (1966), вып. 4, 91–100. (On universal compacta; English translation: *Russian Math. Surveys* 21 (1966), No. 4, 77–86)
- [1967] Об открытых отображениях, ДАН СССР 175 (1967) 292-295. (On open mappings; English translation: Soviet Math. Dokl. 8 (1967) 853-856)
- [1967а] Об универсальных бикомпактах и метрических пространствах данной размерности, *Fund. Math.* **60** (1967) 285-308. (On universal bicompacta and metric spaces of given dimension)
- [1969] Об индуктивных размерностях, ДАН СССР 189 (1969) 254-257. (On inductive dimensions; English translation: Soviet Math. Dokl. 10 (1969) 1402-1406)
- [1970] О бикомпактах с несовпадающими размерностями, ДАН СССР 192 (1970)
   503-506. (On bicompacta with noncoinciding dimensions; English translation: Soviet Math. Dokl. 11 (1970) 671-675)
- [1971] О размерности нормальных пространств, ДАН СССР 201 (1971) 1049–1052.
   (On the dimension of normal spaces; English translation: Soviet Math. Dokl. 12 (1971) 1784–1787)
- [1973] О размерности произведений нормальных пространств, ДАН СССР 209 (1973) 792-794. (On the dimension of products of normal spaces; English translation: Soviet Math. Dokl. 14 (1973) 530-533)

B. A. PASYNKOV and I. K. LIFANOV (E. A. ПАСЫНКОВ, И. К. ЛИФАНОВ)

[1970] Примеры бикомпактов с несовпадающими индуктивными размерностями, ДАН СССР 192 (1970) 276-278. (Examples of bicompacta with noncoinciding inductive dimensions; English translation: Soviet Math. Dokl. 11 (1970) 619-621)

# A. R. PEARS

<sup>[1975]</sup> Dimension theory of general spaces (Cambridge 1975).

#### H. POINCARĖ

[1912] Pourquoi l'espace a trois dimensions, Revue de Metaph. et de Morale 20 (1912) 483-504.

E. Pol

- [1976] Some examples in the dimension theory of Tychonoff spaces, Bull. Acad. Pol. Sci. Sér. Math. 24 (1976) 893-897.
- [1978] Some examples in the dimension theory of Tychonoff spaces, Fund. Math., to appear.
- [1978a] Strongly metrizable spaces of large dimension each separable subspace of which is zero-dimensional, Coll. Math., to appear.
- E. POL and R. POL
- [1977] A hereditarily normal strongly zero-dimensional space with a subspace of positive dimension and an N-compact space of positive dimension, Fund. Math. 97 (1977) 45-52.
- [1979] A hereditarily normal strongly zero-dimensional space containing subspaces of arbitrarily large dimension, Fund. Math., to appear.
- R. Pol
- [1973] There is no universal totally disconnected space, Fund. Math. 79 (1973) 265-267.
- [1979] On category-raising and dimension-raising open mappings with discrete fibers, *Coll. Math.*, to appear.
- L. S. PONTRJAGIN
- [1930] Sur une hypothèse fondamentale de la théorie de la dimension, C. R. Acad. Paris 190 (1930) 1105-1107.
- L. PONTRJAGIN and G. TOLSTOWA
- [1931] Beweis des Mengerschen Einbettungssatzes, Math. Ann. 105 (1931) 734-747.

G. Poprużenko

[1931] Sur la dimension de l'espace et l'extension des fonctions continues, Monatsh. für Math. und Phys. 38 (1931) 129-138.

T. Przymusiński

- [1974] A note on dimension theory of metric spaces, Fund. Math. 85 (1974) 277-284.
- [1977] On the dimension of product spaces and an example of M. Wage, preprint 1977.
- V. I. Рирко (В. И. Пупко)
- [1961] Монотонность размерности одного класса подмножеств нормальных пространств, Вестник Моск. Унив. Сер. Мат. (1961), No. 2, 41-45. (The monotonicity of the dimension of a class of subsets of normal spaces)

# J. H. ROBERTS

- [1947] Open transformations and dimension, Bull. Amer. Math. Soc. 53 (1947) 176– 178.
- [1948] A problem in dimension theory, Amer. Journ. of Math. 70 (1948) 126-128.
- [1956] The rational points in Hilbert space, Duke Math. Journ. 23 (1956) 489-492.

- [1962] Failure of equivalence of dimension concepts for metric spaces, Bull. Amer. Math. Soc. 68 (1962) 609-613.
- [1968] Nonequality of dimensions for metric spaces, Trans. Amer. Math. Soc. 134 (1968) 117-132.
- M. E. RUDIN and M. STARBIRD
- [1975] Products with a metric factor, General Topology and its Appl. 5 (1975) 235-248.
- T. B. RUSHING

[1973] Topological embeddings (New York 1973).

- S. Sakai
- [1968] An axiomatic characterization of the large inductive dimension for metric spaces, *Proc. Japan Acad.* 44 (1968) 782-785.
- Е. ŠČЕРІМ (Е. ЩЕПИН)
- [1972] Аксиоматика размерности метрических пространств, ДАН СССР 206 (1972) 31-32. (An axiomatics of the dimension of metric spaces; English translation: Soviet Math. Dokl. 13 (1972) 1177-1179)

### A. W. SCHURLE

[1969] Strongly countable-dimensional spaces, Trans. Amer. Math. Soc. 136 (1969) 25-32.

## Z. SHMUELY

[1972] Some embeddings of infinite-dimensional spaces, Israel Journ. of Math. 12 (1972) 5-10.

# R. M. SHORI

[1965] A universal snake-like continuum, Proc. Amer. Math. Soc. 16 (1965) 1313-1316.

W. SIERPIŃSKI

- [1915] Homeomorfizm przestrzeni wymiernych, Wektor 4 (1914–15) 215–221. (Homeomorphism of rational spaces)
- [1916] Sur une courbe cantorienne qui contient une image biunivoque et continue de toute courbe donneé, C. R. Acad. Paris 162 (1916) 629-632.
- [1920] Sur un ensemble ponctiforme connexe, Fund. Math. 1 (1920) 7-10.
- [1920a] Sur une propriété topologique des ensembles denombrables denses en soi, Fund. Math. 1 (1920) 11-16.
- [1921] Sur les ensembles connexes et non connexes, Fund. Math. 2 (1921) 81-95.
- [1922] Sur une propriété des ensembles frontières, Fund. Math. 3 (1922) 7-13.
- [1928] Sur les projections des ensembles complémentaires aux ensembles (A), Fund. Math. 11 (1928) 117-122.
- К. SITNIKOV (К. СИТНИКОВ)
- [1953] Пример двумерного множества в трехмерном евклидовом пространстве, допускающего сколь угодно малые деформации в одномерный полиэдр, и некоторая новая характеристика размерности множеств в евклидовых

P. Roy

пространствах,  $\mathcal{A}AH$  CCCP 88 (1953) 21-24. (An example of a two-dimensional set in three-dimensional Euclidean space allowing arbitrarily small deformations into a one-dimensional polyhedron and a certain new characterization of the dimension of sets in Euclidean spaces)

- [1954] Пример двумерного множества в трехмерном евклидовом пространстве, не разрезающего никакой области этого пространства, ДАН СССР 94 (1954) 1007–1010. (An example of a two-dimensional set in three-dimensional Euclidean space which does not separate any region of that space)
- Е. G. Skljarenko (Е. Г. Скляренко)
- [1958] О вложении нормальных пространств в бикомпакты того же веса и той же размерности,  $\mathcal{J}AH$  CCCP 123 (1958) 36-39. (On the embedding of normal spaces into bicompacta of the same weight and dimension)
- [1962] Теорема об отображениях понижающих размерность, Bull. Acad. Pol. Sci. Sér. Math. 10 (1962) 429-432. (A theorem on dimension-lowering mappings)
- [1963] О теореме Вьеториса-Бегла, ДАН СССР 149 (1963) 264-267. (On a theorem of Vietoris and Begle; English translation: Soviet Math. Dokl. 4 (1963) 371-374)
- JU. M. SMIRNOV (IO. M. СМИРНОВ)
- [1951] Некоторые соотношения в теории размерности, *Mam. Cb.* 29 (1951) 151-172. (Some relations in the theory of dimension)
- [1956] О метрической размерности, в смысле П. С. Александрова, Изв. Акад. Наук СССР Сер. Мат. 20 (1956) 679-684. (On the metric dimension in the sense of P. S. Alexandroff)
- [1956a] О размерности пространств близости, Mam. Cb. 38 (1956) 283-302. (On the dimension of proximity spaces; English translation: Amer. Math. Soc. Transl. Ser. 2, 21 (1962) 1-20)
- [1958] Пример нульмерного пространства имеющего бесконечную размерность в смысле покрытий, *ДАН СССР* 123 (1958) 40-42. (An example of a zero-dimensional space which has infinite covering dimension)
- J. C. SMITH and L. L. KRAJEWSKI
- [1971] Expandability and collectionwise normality, Trans. Amer. Math. Soc. 160 (1971) 437-451.
- E. H. SPANIER

- M. A. Štan'ko (M. A. Штанько)
- [1971] Решение задачи Менгера в классе компактов, ДАН СССР 201 (1971) 1299–1302. (Solution of Menger's problem in the class of compacta; English translation: Soviet Math. Dokl. 12 (1971) 1846–1849)
- J. Suzuki
- [1959] Note on a theorem for dimension, Proc. Japan Acad. 35 (1959) 201-202.
- А. D. Таїманов (А. Д. Тайманов)
- [1955] Об открытых образах боролевских множеств, Mam. Cb. 37 (1955) 293-300. (On open images of Borel sets)

<sup>[1966]</sup> Algebraic topology (New York 1966).

- J. TERASAWA
- [1977] Spaces N∪ R need not be strongly 0-dimensional, Bull. Acad. Pol. Sci. Sér. Math. 25 (1977) 279-281.
- **B.** TOMASZEWSKI
  - [1979] On weakly n-dimensional spaces, Fund. Math., to appear.
- L. A. TUMARKIN
- [1925] Zur allgemeinen Dimensionstheorie, Proc. Akad. Amsterdam 28 (1925) 994-996.
- [1926] Beitrag zur allgemeinen Dimensionstheorie, Mam. C6. 33 (1926) 57-86.
- [1928] Sur la structure dimensionelle des ensembles fermés, C. R. Acad. Paris 186 (1928) 420-422.
- P. URYSOHN
  - [1922] Les multiplicités Cantoriennes, C. R. Acad. Paris 175 (1922) 440-442.
- [1925] Mémoire sur les multiplicités Cantoriennes, Fund. Math. 7 (1925) 30-137.
- [1926] Mémoire sur les multiplicités Cantoriennes (suite), Fund. Math. 8 (1926) 225-359.
- I. A. VAINŠTEIN (II. A. ВАЙНШТЕЙН)
- [1949] О повышающих размерность отображениях, ДАН СССР 67 (1949) 9-12. (On dimension-raising mappings)
- [1952] Об одномерных отображениях, ДАН СССР 83 (1952) 175-178. (On onedimensional mappings)
- N. VEDENISSOFF (H. E. BEZEHUCOB)
- [1939] Замечания о размерности топологических пространств, Уч. Записки Моск. Унив. 30 (1939) 131-140. (Remarks on the dimension of topological spaces)
- [1940] Généralisation de quelques théorèmes sur la dimension, Comp. Math. 7 (1940) 194-200.
- [1941] О размерности в смысле E. Čech'a, Изв. Акад. Наук СССР Сер. Мат. 5 (1941) 211-216. (On the dimension in the sense of E. Čech)
- **P.** VOPĚNKA ( $\Pi$ . BOITEHIKA)
- [1959] Замечание о размерности метрических пространств, Czech. Math. Journ.
   9 (1959) 519-552. (A remark on the dimension of metric spaces)
- M. L. WAGE
- [1977] The dimension of product spaces, preprint 1977.

A. D. WALLACE

[1945: Dimensional types, Bull. Amer. Math. Soc. 51 (1945) 679-681.

H. WALLMAN

[1938] Lattices and topological spaces, Ann. of Math. 39 (1938) 112-126.

- J. J. WALSH
- [1978] Infinite dimensional compacta containing no *n*-dimensional  $(n \ge 1)$  subsets, Bull. Amer. Math. Soc. 84 (1978) 137-138.

#### G. T. WHYBURN

- [1958] Topological characterization of the Sierpiński curve, Fund. Math. 45 (1958) 320-324.
- D. C. WILSON
- [1972] Open mappings of the universal curve onto continuous curves, Trans. Amer. Math. Soc. 168 (1972) 497-515.
- A. V. ZARELUA (A. B. ЗАРЕЛУА)
- [1961] О теореме Гуревича, ДАН СССР 141 (1961) 777-780. (On a theorem of Hurewicz; English translation: Soviet Math. Dokl. 2 (1961) 1534-1537)
- [1963] О теореме Гуревича, *Mam. C6.* 60 (1963) 17-28. (On a theorem of Hurewicz)
- [1963а] О равенстве размерностей, *Mam. Cб.* **62** (1963) 295-319. (On the equality of dimensions)
- [1964] Универсальный бикомпакт данного веса и данной размерности, ДАН СССР 154 (1964) 1015–1018. (A universal bicompactum of given weight and dimension; English translation: Soviet Math. Dokl. 5 (1964) 214–218)
- [1972] О наследственно бесконечномерных пространствах, Theory of sets and topology (In honour of Felix Hausdorff) (Berlin 1972) 509-525. (On hereditarily infinite-dimensional spaces)
- V. P. ZOLOTAREV (B. II. ЗОЛОТАРЕВ)
- [1975] О размерности подпространств, Вестник Моск. Умив. Сер. Мат., No. 5, (1975) 10–12. (On the dimension of subspaces; English translation: Mosc. Univ. Bull. 30 (1975), No. 5-6, 7-8)

# LIST OF SPECIAL SYMBOLS

# Symbols defined in the text

3	$p+q, \lambda p$	100
9	$p_0 p_1 \dots p_n$	100
16, 100	$ \mathscr{K} $	101
22	$\operatorname{St}_{\mathscr{K}}(p_t)$	102
44	$\mathcal{N}(\mathcal{U}), \mathcal{N}(\mathcal{U})$	103
52	$\hat{\varrho}(f,g)$	118
54	YX	118
54, 222	$M_n^m$	122
56	$\operatorname{St}(x, \mathscr{U})$	167
57	w(X)	175
61	St( <i>B</i> , <i>A</i> )	241
61	<i>B</i> (m)	263
61 <sup>;</sup>	$C_k(f), D_k(f)$	281
95	D(f)	283
	3 9 16, 100 22 44 52 54 54, 222 56 57 61 61 61 95	$\begin{array}{rrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrr$

# Symbols from set theory and general topology

 $x \in A, x \notin A - x$  belongs to A, x does not belong to A  $\emptyset$  - empty set  $A \subset B, B \supset A - A$  is contained in B  $\{x \in X: \varphi(x)\}, \{x: \varphi(x)\}$  - the set of x's which satisfy  $\varphi$   $A \cup B, \bigcup_{s \in S} A_s, \bigcup_{i=1}^{\infty} A_i, \bigcup \mathscr{A}$  - union of sets  $A \cap B, \bigcap_{s \in S} A_s, \bigcap_{i=1}^{\infty} A_i, \bigcap \mathscr{A}$  - intersection of sets  $A \cap B$  - difference of sets  $X \times Y, \prod_{s \in S} X_s, \prod_{i=1}^{\infty} X_i$  - Cartesian product of sets and spaces  $\{x_1, x_2, ..., x_k\}$  - set consisting of points  $x_1, x_2, ..., x_k$   $x_1, x_2, ..., \{x_i\}$  - infinite sequence  $\{x_1, x_2, ...\}$  - set consisting of points  $x_1, x_2, ..., x_k$   $f(A), f^{-1}(B)$  - image and inverse image of a set under a mapping gf - composition of mappings |A| - cardinality of a set  $\aleph_0$  - cardinality of the set of natural numbers (aleph zero) c — cardinality of the set of real numbers (continuum)  $A_1, A_2, ..., A_{\alpha}, ..., \alpha < \xi$  — transfinite sequence of sets R, I — the real line and the closed unit interval  $R^n$  — Euclidean *n*-space  $I^n$  — unit *n*-cube in  $R^n$  $S^{n-1}$ ,  $B^n$  — unit (n-1)-sphere and unit *n*-ball in  $B^n$ IN0 --- Hilbert cube  $\varrho(x, y)$  — distance between points  $\varrho(x, A)$  — distance from a point to a set  $\rho(A, B)$  — distance between sets B(x, r), B(A, r) — balls in a metric space  $\delta(A)$  — diameter of a set f:  $X \rightarrow Y$  — continuous mapping  $f|A, f_B$  — restrictions of a mapping  $\overline{A}$ , IntA, FrA — closure, interior and boundary of a set  $A^{d}$  — set of accumulation points of a set  $X^{\mathfrak{m}}$  — Cartesian power of a space X/E — quotient space  $\{X_i, \pi_j^i\}, \{X_\sigma, \pi_{\varrho}^\sigma, \Sigma\}$  — inverse sequence and inverse system of spaces limS — limit of an inverse sequence or an inverse system  $X \oplus Y, \bigoplus_{s \in S} X_s$  — sum of spaces

 $\beta X$  — Čech-Stone compactification

# **AUTHOR INDEX**

Aarts, J. M. 156, 181, 289 Alexandroff, P. 9, 29, 58, 59, 84, 86, 96, 99, 115, 116, 117, 140, 156, 157, 181, 222, 236, 238, 289 Anderson, R. D. 49, 99, 129, 290 Arbangel'skii, A. 84, 86, 238, 249, 275, 276, 288, 290 Bennett, R. 86, 290 Bessaga, C. 86, 290 Bing, R. H. 151, 290 Borsuk, K. 44, 97, 98, 275, 290 Bothe, H.G. 129, 290 Brouwer, L. E. J. 7, 8, 9, 29, 30, 48, 58, 84, 85, 290

Cannon, J. W. 129, 291 Cantor, G. 6 Čech, E. 58, 167, 180, 195, 221, 222, 291 Chogoshvili, G. 116, 291

van Dalen, J. 204, 291
van Douwen, E. K. 265, 291
Dowker, C. H. 98, 168, 169, 180, 181, 182, 194, 195, 221, 224, 236, 264, 291

Egorov, V. I. 59, 60, 292 Eilenberg, S. 69, 84, 97, 249, 292 Engelking, R. 69, 237, 265, 292 Erdös, P. 18, 49, 292

Fedorčuk, V. V. 221, 292
Filippov, V. V. 180, 203, 221, 236, 286, 292, 297
Fitzpatrick, B. 205, 206, 293
Flores, A. 132, 293

Ford, R. M. 205, 206, 293 Forge, A. B. 69, 293 Fort, M. K. 86, 293 Frankl, F. 86, 293 Fréchet, M. 30, 84, 85, 293 French, J. A. 236, 237, 239 Freudenthal, H. 9, 96, 150, 152, 153, 249, 286, 293 de Groot, J. 49, 69, 265, 276, 293 Hansell, R. W. 265, 293 Hausdorff, F. 29, 36, 141, 265, 294 Hemmingsen, E. 58, 236, 237, 238, 294 Henderson, D. W. 48, 294 Hilgers, A. 40, 294 Hodel, R. E. 140, 288, 294 Hopf, H. 117, 289 Hurewicz, W. 9, 18, 28, 48, 49, 59, 69, 70, 71, 96, 97, 98, 128, 129, 130, 140, 141, 264, 291, 294 Isbell, J. R. 150, 151, 295 Janiszewski, Z. 36, 295 Katětov, M. 60, 221, 222, 225, 264, 295 Katuta, Y. 203, 206, 295 Keisler, J. E. 49, 99, 290 Keldyš, L. 140, 295 Kimura, N. 195, 204, 295 Knaster, B. 38, 39, 295 Kodama, Y. 49, 96, 237, 295 Kolmogoroff, A. 140, 295 Krajewski, L. L. 196, 303

Kuratowski, K. 8, 19, 28, 38, 39, 70, 71, 96, 98, 115, 128, 129, 130, 151, 158, 236, 295, 296 Kuz'minov, V. I. 96, 157, 296 Lebesgue, H. 7, 8, 58, 84, 296 Lefschetz, S. 128, 129, 296 Lelek, A. 84, 140, 286, 296 Levšenko, B. T. 265, 296 Lifanov, I. K. 168, 195, 196, 203, 204, 221, 222, 296, 297, 300 Lipscomb, S. L. 275, 297 Ljuksemburg, L. A. 84, 297 Lokucievskiĭ, O. V. 157, 181, 222, 297 Lunc, A. L. 222, 297 Marczewski, E. 274, 297 Mardešić, S. 150, 152, 249, 297 Mazurkiewicz, S. 8, 29, 36, 40, 50, 51, 84, 99, 297 McCord, M.C. 151, 298 McDowell, R. H. 69, 293 Menger, K. 8, 9, 18, 28, 36, 48, 49, 50, 69, 84, 96, 128, 129, 131, 156, 158, 159, 294, 296, 298 Morita, K. 203, 204, 221, 222, 223, 236 237, 264, 274, 275, 276, 286, 287 298 Nagami, K. 59, 84, 86, 140, 150, 195 204, 205, 221, 237, 264, 274, 276, 286, 287, 288, 297, 298, 299 Nagata, J. 49, 71, 84, 196, 204, 223, 274, 275, 276, 293, 299 Nishiura, T. 156, 157, 168, 181, 222, 289, 299 Nöbeling, G. 85, 128, 129, 195, 299 Ostrand, P. A. 223, 236, 237, 274, 299 Otto, E. 69, 84, 98, 292, 296 Pasynkov, B. A. 84, 140, 151, 153, 156, 157, 168, 181, 195, 196, 203, 204, 211, 222, 236, 240, 249, 275, 277, 289, 297, 300

Peano, G. 6 Pears, A. R. 203, 204, 237, 275, 300 Pełczyński, A. 86, 290 Poincaré, H. 7, 301 Pol, E. 39, 180, 225, 301 Pol, R. 40, 180, 266, 288, 301 Pontrjagin, L. S. 49, 86, 128, 293, 301 Poprużenko, G. 266, 301 Przymusiński, T. 180, 265, 301 Pupko, V. I. 222, 238, 301 Roberts, J. H. 36, 38, 84, 130, 142, 264, 274, 299, 301 Roy, P. 203, 302 Rudin, M. E. 203, 302 Rushing, T. B. 117, 302 Sakai, S. 156, 302 Ščepin, E. 156, 157, 302 Schurle, A. W. 84, 302 Segal, J. 150, 152, 297 Shmuely, Z. 84, 302 Shori, R. M. 151, 302 Sierpiński, W. 17, 18, 28, 30, 36, 50, 84, 129, 131, 302 Sitnikov, K. 59, 86, 115, 117, 302 Skljarenko, E. G. 249, 286, 303 Smirnov, Ju. M. 59, 86, 118, 180, 182, 195, 203, 221, 222, 241, 265, 296, 303 Smith, J.C. 196, 303 Spanier, E. H. 98, 133, 303 Štan'ko, M. A. 129, 303 Starbird, M. 203, 302 Steenrod, N. 249, 292 Suzuki, J. 286, 303 Švedov, I. 157 Taimanov, A. D. 140, 303 Terasawa, J. 225, 304 Tolstowa, G. 128, 301 Tomaszewski, B. 51, 304 Tumarkin, L. A. 9, 18, 28, 48, 96, 304

Urysohn, P. 8, 9, 18, 28, 29, 36, 48, 50, 69, 84, 96, 181, 290, 304 Vainštein, I. A. 140, 286, 304
Vedenissoff, N. 18, 58, 180, 181, 203, 222, 304
Vopěnka, P. 264, 304
Wage, M. 180, 304
Wallace, A. D. 238, 304
Wallman, H. 96, 97, 129, 130, 140, 222, 294, 304
Walman, H. 96, 97, 129, 130, 140, 222, 294, 304
Walman, H. 96, 97, 129, 130, 140, 222, 294, 304
Walman, H. 96, 97, 129, 130, 140, 222, 294, 304
Walman, H. 96, 97, 129, 130, 140, 222, 294, 304

# SUBJECT INDEX

addition theorem 45, 260 -- theorem for dim 215 -- theorem for Ind 172 Alexandroff's axioms 154 -- theorem 138 Antoine set 117

Baire space 263 barycentric coordinates 101 Borsuk homotopy extension theorem 92 boundary of a simplex 101 bounded subset of an ordered set 166 Brouwer-Čech dimension 52

Cantor set 11 Cantor-manifold, n-dimensional 91, 238 -- theorem 93 Cartesian product theorem 46, 260 - product theorem for dimension 0 22 Čech-Lebesgue dimension 54 closed mapping 133 --- shrinking 54 - swelling 208 cofinal subset of an ordered set 166 cohomological dimension 95 coincidence theorem 65 compactification theorem 61 --- theorem for dim 244 - theorem for dimension 0 27 completely paracompact space 204 completion theorem 260 complex 101 component of a family of sets 197 - of a space 32 continuous extension 87 continuously extendable mapping 87 continuum hypothesis 82 convex components of a set of ordinal numbers 166

convex set of ordinal numbers 165
subset of Euclidean n-space 85
countable sum theorem 257
sum theorem for dim 212
sum theorem for Ind 188
countable-dimensional space 83
covering dimension 54
dimension for completely regular spaces 222
cut 5
cutting set 5, 77

decompositon theorem, first 43, 259 — theorem, second 43, 259 development 268 dimension at a point 9 dimensional components 99 — kernel 50 Dimensionsgrad 7 discontinuous space 32 Dowker space 168 Dowker's example 176 — theorem 226

embedding theorem 120 — theorem for dimension 0 27 enlargement theorem 45, 260 — theorem for dimension 0 15 Erdös' example 16 essential mapping 96  $\varepsilon$ -mapping 106  $\varepsilon$ -translation 110  $\mathscr{E}$ -mapping 106

face 100 finite-dimensional mapping 283 Freudenthal's theorem 284 function space 118 functionally closed family of sets 222
closed set 222
open family of sets 222
open set 222
fundamental theorem of dimension theory 73
F<sub>er</sub>set 15

general position, system of points in 102 geometric dimension 101  $G_{\delta}$ -set 15

hereditarily disconnected space 31 — normal space 162 homotopic mappings 91 homotopy 91

independent family of sets 238 interior of a simplex 101 irreducible mapping 152

Katětov-Morita theorem 254 Knaster-Kuratowski fan 38 *x*-mapping 105

large inductive dimension 52
Lebesgue's covering theorem 78
Lindelöf space 12
linearly independent system of points 100
locally countable family of sets 191
finite sum theorem 257
finite sum theorem for dim 213
finite sum theorem for Ind 191
strongly paracompact space 218
Lokucievskii's example 178
long line 178
segment 178

Mardešić's factorization theorem 242 Mazurkiewicz's theorem 80 Menger universal space 121 Menger's axioms 155 Menger-Urysohn dimension 3 mesh of a family of sets 56 metric dimension 57 — on a space 24 modification of a mapping 152 Morita's theorem 229

nerve 103 network 133 Nishiura's axioms 155 Nöbeling universal space 121 non-Archimedean metric 266 norm of a point in Hilbert space 16

open mapping 133 -- shrinking 54 -- swelling 208 open-and closed mapping 139 order of a family of sets 54 -- totally paracompact space 205 origin of  $R^m$  100 Ostrand's theorem 228

partition 4 Pasynkov's factorization theorem 272 perfect mapping 277 perfectly normal space 164 point-countable family of sets 163 — sum theorem for dim 213 — sum theorem for Ind 192 polyhedron 101 product of a point by a number 100 punctiform space 32 *II*-like space 150

quasi-component of a space 31 quasi-simplicial mapping 148

rank of a family of sets 238 refinement 54 refining cover 54 313

relative dimension 285 retract 28 Roy's space 197 separated sets 12 eparating set 4, 77 separation theorem 258 - theorem, first 45 - theorem, second 45 theorem for dimension 0, first 12  $\sim$  theorem for dimension 0, second 14 - theorem for Ind 172 separator 4 shrinking 54 Sierpiński's theorem 112 - universal curve 129 simplex 100 simplicial complex 101 Sitnikov's example 113 small inductive dimension 3 snake-like continuum 150 stable value 97 star of a point 167 - of a set 241 - of a vertex 102 - refinement 241 star-finite family of sets 197 strong development 269 strongly hereditarily normal space 163 - paracompact space 197 subcomplex 101 subspace theorem 4, 257 - theorem for dim 216 - theorem for Ind 188 sum of points 100 - theorem 42 - theorem for dimension 0 20 super normal space 168 sweeping out theorem 108 swelling 208 σ-totally paracompact space 205

theorem on dimension of the limit of an inverse sequence 149, 261

theorem on dimension of the limit of an inverse system 247 dimension-lowering — on mappings 136, 281 - on dimension-raising mappings 134, 278 - on expansion in an inverse sequence 144 - on expansion in an inverse system 245 - on extending mappings to spheres 90, 233 - on  $\varepsilon$ -mappings 108 — on  $\varepsilon$ -translations 111 - on &-mappings 107, 233 - on inverse sequences 149 - on inverse systems 248 - on partitions 67, 230 Tietze–Urysohn theorem 87 totally disconnected space 31 - normal space 168 underlying polyhedron 101 universal space 23 --- space theorem 274 - space theorem, first 120 - space theorem, second 126 - space theorem for dim 244 — space theorem for dimension 0 26 unstable value 97 Urysohn lemma 87 Vainštein's lemma 139

theorem, first 283
theorem, second 285
vertex of a simplex 100
of a simplicial complex 101

weakly *n*-dimensional space 50 — paracompact space 164 weight of a space 175

zero-dimensional space 10